A Corpus-Based Critical Discourse Analysis of the Ideological Representations and Legitimation in the Salafi Discourse in Saudi Arabia (1980-2000)

RAJAB GAMAAN MUHSEN ALZAHRANI

A THESIS SUBMITTED FOR THE DEGREE OF DOCTOR OF PHILOSOPHY DEPARTMENT OF LINGUISTICS AND ENGLISH LANGUAGE LANCASTER UNIVERSITY UNITED KINGDOM

September 2013

ProQuest Number: 11003477

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 11003477

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

{يَا أَيُّهَا الَّذِينَ آمَنُوا كُونُوا قَوَّامِينَ لِلَّهِ شُهَدَاءَ بِالْقِسْطِ (وَلَا يَجْرِمَنَّكُمْ شَنَآنُ قَوْمِ عَلَىٰ أَلَّا تَعْدِلُوا أَ اعْدِلُوا هُوَ أَقْرَبُ لِلتَّقْوَىٰ أَ وَاتَّقُوا اللَّهَ أَ إِنَّ اللَّهَ حَبِيرٌ بِمَا تَعْمَلُونَ} قران كريم. سورة المآندة، أية ٨.

{O you who believe! Stand out firmly for Allah as just witnesses; and let not the enmity and hatred of others make you avoid justice. *Be just*: that is nearer to piety; and fear Allah. Verily, Allah is Well-Acquainted with what you do} Holy Qur'an; Chapter 5: 8.

Dedication

This work is dedicated to those who seek justice for the way to justice is only one whereas truth may have different ways.

To my father Gamaan and my mother Alyiaa, who supported me and taught me to avoid dealing with others unjustly.

To my wife: Muneerah who, despite her health problems shared happy and sad moments with me. She is now staying with my elderly parents helping them.

To my son Khalid who is taking care of the whole family (my parents, his mother and my other children) while I am away working on this thesis.

To my daughters Sumayyiah and Sarah and my son Sultan who are waiting for me anxiously to bring them glad tidings about my work on this thesis.

AUTHOR'S DECLARATION

I hereby declare that this thesis is my own work and has not been submitted in substantially the same form for the award of a higher degree elsewhere.

ABSTRACT

This thesis examines the ideological representations of the Salafi discourse in Saudi Arabia from 1980 to 2000. It attempts to answer the following overarching question: To what extent and how is Salafi discourse in Saudi Arabia in the 1980s-1990s homogenous?

Utilizing a composite of methodological tools from CDA; DHA (Reisigl & Wodak, 2001, 2009; Wodak, 1996, 1999, 2001, 2007), social actor theory (van Leeuwen 1996, p. 32-70; 2008, pp. 23-54), corpus linguistics (Baker *et al.* 2008) and argumentation theory (Van Eemeren and Grootendorst 2004), I examine the most salient discursive features of semantic foci, social actor representation and argumentation schemes in religious discourses of two Saudi Salafi strands. I have built two corpora composed of the writings of two prominent Saudi Salafi writers that reflect religious discourses in Saudi Arabia of two Salafi strands; religiously based Salafi discourse as exemplified in the writings of AbdulAziz Ibn Baz and politically-oriented Salafi discourse as exemplified in the writings of Safar Alhawali.

The whole thesis is based on analysing keywords utilizing collocates, word clusters and concordance-based analysis to identify similarity and/or differences in the ideological representations of the two aforementioned Salafi strands. Results show that the two Salafi writers share some ideological representation (both theological and political). However they have different ideological representations that are politically motivated such as Alhawali's negative representations of the West and the American-led Allied forces during the Gulf Crisis.

ACKNOWLEDGEMENTS

(مَن لا يشكرُ النَّاسَ لا يشكرُ اللهَ)

حديثكريم رواهابو هريرة

Abu Hurrairah narrated that the Prophet of Islam said: (He who does not show his gratitude to people and thank them (for their help) is not thankful to Allah).

I would like to express my thanks and gratitude to my supervisor and mentor Professor Paul Baker who provided invaluable advice, incisive remarks and comments, guidance and suggestions (whether in our regular meetings or in his classes and other talks and presentations) that were a catalyst for doing more and writing more in this thesis. I'm extremely grateful to him because without his academic support this work would not have come to light. My thanks also extend to my previous supervisor Veronika Koller.

I am grateful to Distinguished Professor Tony McEnery for his invaluable suggestions and comments on my upgrade document and for his great efforts as the internal examiner of this thesis. I am also grateful to Johnny Unger for his comments in the post-confirmation panel. Further I thank my external examiner Professor Dawn Archer for reading and examining this work. I am extremely grateful to my colleague and Brother Dr. Abdullah Makhayish Alzahrani, the Vice Chancellor of Albaha University, whose brotherly helpful encouragement worked as a catalyst for my work in this thesis.

I am extremely grateful to my father and mother and my brother Khamis Alzahrani. They supported me throughout the years that I spent in the UK. Without their support this work would have been a dream only. I am also so grateful to my wife Muneerah, my daughters Sumayyiah and Sarah and my sons Khalid and Sultan who spent nights and days supporting me and during the last two years they waited patiently for my visit to Saudi Arabia to celebrate the moments I spent with them.

Last but not the least, I would like to thank his Excellency Professor Saad Alhariqi, the Chancellor of Albaha University, and the respectful members of the council in Albaha University who trusted me and granted me an exception to study a PhD abroad as the law in the Saudi universities does not allow people of my age to pursue their higher studies.

Table of Contents

DedicationII			
Abstract			
AcknowledgementIV			
Table of ContentsVI			
List of TablesXI			
List of FiguresXI			
List of ConcordancesXI			
List of AbbreviationsXV			
Part one1			
CHAPTER 1: Introduction1			
1.1 Overview1			
1.2 Research problem1			
1.3 Political and religious context			
1.3.1 The main tenets of the Najdi religious movement			
1.3.2 Religious and political unrest in Saudi Arabia			
1.3.3 The siege of the Grand Mosque in Makkah			
1.3.4 The Iraqi invasion of Kuwait			
1.3.5 The rise of the Sahwah10			
1.3.6 Main tenets of Sahwah11			
1.3.7 Salafism and prominent Saudi Salafis13			
1.3.7.1 Abdul Aziz Ibn Baz13			
1.3.7.2 Safa Alhawali14			
1.4 Aim and scope of the study15			
1.5 Position of the analyst			
1.6 Approach of the study17			
1.7 Rationale of the study18			
1.8 Research questions19			
1.9 Thesis structure19			
CHAPTER 2: Critical Review of Literature21			
2.1 Introduction			
2.2 Studies on religion			
2.2.1 Religious studies			
2.2.2 Studies on Salafism			
2.2.3 Critical religious studies			
2.3 Critical Discourse analysis			
2.3.1 The representation of social actors: Sociosemantic categories			
2.3.2 Discourse Historical Approach (DHA)			
2.3.3 CDA: Voicing some concerns			
2.4 Corpus Linguistics and its contribution to CDA studies			
2.5 Conclusion			
CHAPTER 3: Methodological framework			
3.1 Introduction40			

3.2 Methodological framework40		
3.2.1 Corpus linguistics	41	
3.2.1.1 Keyness3.2.1.2 Collocation and word clusters3.2.1.3 Concordance	42	
3.2.2 Aspects of CDA applied in my study	43	
 3.2.2.1 Social actor theory	44 45 45 46 46 48	
3.2.3 Argumentation		
3.2.3.1 Fallacies and topoi3.2.3.2 Argumentation schemes3.2.3.3 Legitimation3.2.3.4 Recontextualization	52 53	
3.3 Data	54	
3.3.1 Collecting the corpus3.3.2 Translation and transliteration of Arabic data	57	
3.4 Methods	61	
 3.4.1 Computational identification: Analysing keywords 3.4.2 Collocational analysis	63 64 64 65 66	
3.5 Conclusion	67	
Part Two	69	
CHAPTER 4: Examining Ibn Baz's semantic foci keywords across both corpora	69	
4.1 Introduction		
4.2 Theological semantic foci in the Salafi Corpora	73	
4.2.1 Divinity	74	
4.2.1.1 The construction of 'Allah' in both corpora4.2.1.2 The collocates of the key term 'ALLAH' in Ibn Baz4.2.1.3 The collocates of the key term 'ALLAH' in Alhawali	77 83	
4.2.2 Polytheism	93	
4.2.3 Polytheism in Alhawali 4.2.4 Monotheism	.104	
4.2.4.1 Unity of worship 4.2.4.2 Unifying Allah in His names and attributes	105 107	

4.2.5 Islam
4.3 Conclusion
Chapter 5: Examining Alhawali's politically-oriented semantic foci keywords across both corpora
5.1 Introduction
5.2 Politically oriented semantic foci in the Salafi Corpora
5.3 External threats 1: Locations, nationalities and politics
 5.3.1 The key term 'WEST' in Alhawali's corpus: Western civilization as an enemy of Islam
5.4 External threats 2: Military invasion and the Crusader States
5.4.1 Wars and military invasion in Alhawali
5.5 The construction of the past149
 5.5.1 The construction of the past in Alhawali: Europe as a source of evil
5.6 Mind in the Salafi corpora159
5.6.1 'Mind' in Alhawali's corpus160 5.6.2 'Mind' in Ibn Baz's corpus162
5.7 Religion in the Salafi corpora
5.7.1 'Religion' in Alhawali's corpus167 5.7.2 'Religion' in Ibn Baz's corpus170
5.8 Conclusion
Part Three176
Chapter 6: Social Actor Representation of "the Other" in both Salafi Corpora176
6.1 Introduction176
6.2 Social actor representation: Ibn Baz's keywords
6.2.1 The representation of the 'Other' in Ibn Baz181
 6.2.1.1 The representation of 'دعاة' (do ʿāh; teachers/propagandists) in Ibn Baz and Alhawali
6.3 Social actor representation: Alhawali's keywords
6.3.1 The representation of the 'Other' in Alhawali196
6.3.1.1 The representation of 'رجال' (<i>riğāl</i> ; (clergy) men) in Alhawali and Ibn Baz198

6.3.1.2 The representation of 'الشيوعية' (<i>'aššyo 'iyyah</i> ; communism/communist) in Alhawali and Ibn Baz
6.4 Conclusion
Chapter 7: Argumentation in the Salafi corpora: Ideological legitimation
7.1 Introduction
7.2 Identifying key argumentation indicators in the Salafi corpora
7.2.1 Argumentation indicators in Ibn Baz
7.3 Argumentation Schemes in the Salafi Corpora:
7.4 Ideological legitimation (1): Argumentation schemes in Ibn Baz
7.4.1 True beliefs and worship practices
7.4.1.1 Critical evaluation of argumentation thread 1 in Ibn Baz
7.4.2 Arab nationalism: Rejection of the Sharia
7.4.2.1 Critical evaluation of argumentation thread 3 in Ibn Baz
7.4.3 Particles of faith: Sacred Books of Judaism and Christianity
7.5 Ideological legitimation (2): Argumentation schemes in Alhawali
7.5.1 Western woman and social life: A Salafist perspective
7.5.1.1 Critical evaluation of argumentation thread 1 in Alhawali240
7.5.2 The Church and the Salvation 'theory'
7.5.3 Political enmity: Invasion of the Gulf and the Arabian Peninsula246
7.5.3.1 Critical evaluation of argumentation thread 3 in Alhawali250
7.6 Conclusion
Part Four
CHAPTER 8: Explanations of the results in Context257
8.1 Introduction
8.2 Summary of findings 1: Ibn Baz
8.3 Ibn Baz: Perfection of religious duties (Religiously based Salafism)
 8.3.1 Innovation of religious rituals: Oneness of Worship vs. Associationism260 8.3.2 Unifying Allah in His names and attributes
8.4 Summary of the findings 2: Alhawali
8.5 The West and western enigma: Politically oriented Salafism
8.5.1 Religious and Social life in the West
8.6 Morality in the West
8.7 Conclusion
Chapter 9: Conclusion
9.1 Introduction

9.2 Saudi Salafists: Ibn Baz vs. Alhawali	278
9.3 Implications	282
9.4 Limitations	284
9.5 Recommendations and Future Research	286
9.6 Concluding remarks	287
References	288
Appendix A	294
Appendix B	320
Appendix C	330
Appendix D	331
Appendix F	337

•

List of Tables

1

	Table 3.1 The data used for this study	56
	Table 4.1 Semantic foci keywords in Ibn Baz	7 0
4	Table 4.2 Semantic foci keywords in Alhawali	7 1
Contraction of the local division of the loc	Table 4.3 Categories of semantic foci keywords in Ibn Baz	7 3
	Table 4.4 Selected theological semantic foci keywords in Ibn Baz	7 4
Contract Million	Table 4.5 Keywords that refer to 'الله', 'Allah' in Ibn Baz	75
	Table 4.6 Word clusters that contain the different forms of the key term 'ALLAH' and the keywords 'سبحانه'	
	(subhanahu; the Glorified) and 'تعالى' (taʿāla; the Exalted) in Ibn Baz	7 5
1	Table 4.7 Collocates of 'ALLAH' in both corpora	77
	Table 4.8 Word clusters that contain ' \forall ' ($l\bar{a}$; no) and the key term 'ALLAH' in both corpora	8 6
4	Table 4.9 Semantic foci keywords that that indicate 'polytheism' ('ašširk; الشرك) in Ibn Baz	9 3
	Table 4.10 Collocates of 'polytheism' (<i>'ašširk</i> ; الشرك) in Ibn Baz	9 4
	Table 4.11 Collocates of the keyword 'الإسلام' ('a'islām; Islam) in Ibn Baz and Alhawali	109
	Table 4.12 Keywords that indicate the 'final return' in Ibn Baz	114
	Table 5.1 Semantic foci keywords in Alhawali's corpus	124
	Table 5.2 Categories of semantic foci keywords in Alhawali's corpus	124
	Table 5.3 Political semantic foci keywords in Alhawali's corpus	126
	Table 5.4 Collocates of the key term 'WEST' in Alhawali's corpus	128
	Table 5.5 Number of occurrences of the key term 'WAR' in both corpora	143
	Table 5.6 Clusters that contain the singular definite noun 'الحرب' ('alharb; the war) In Alhawali's corpus	144
	Table 5.7 Word forms of the key term 'CENTURY' in both Salafi corpora	151
	Table 5.8 Word forms of the key term 'MIND' in the Salafi corpora	160
	Table 5.9 Word forms of the key term 'RELIGION' in the Salafi corpora	166
	Table 5.10 Word forms of the key term 'RELIGION' that are keywords in both corpora	167
	Table 5.11 Collocates of the key term 'RELIGION' in Alhawali	168
	Table 5.12 Collocates of the word forms of the key term 'RELIGION' in Ibn Baz	171
	Table 5.13 Examples of clusters that contain the key term 'RELIGION' in Ibn Baz	172
	Table 6.1 Categories of social actors in Ibn Baz (derived from keywords)	178
	Table 6.2 Collocates of 'teachers/propagandists' (do 'āh; دعاة) in Ibn Baz	184
	Table 6.3 Word forms of 'دعاة' (do ʿāh; teachers/propagandists) in Alhawali	187
	Table 6.4 Collocates of 'أهل' ('ahl; the people of) in Ibn Baz	190
	Table 6.5 Categories of social actors in Alhawali	195
	Table 6.6 Keywords that indicate 'the Other' in Alhawali	198
	Table 7.1 Argumentation indicators in Ibn Baz	213
	Table 7.2 Categories of keywords that indicate argumentation in Ibn Baz	214
	Table 7.3 Argumentation indicators in Alhawali	216
	Table 7.4 Categorized collocates of 'قال' (gala; He/he said) in Ibn Baz	2 20
	Table 7.5 Word clusters that contain the keyword 'قال' (qala; He/he said) in Ibn Baz's corpus	2 23

L	List of Figures	
	Figure 1.1 Map showing Saudi Arabia and neighbouring countries	6
	Figure 2.1 Van Leeuwen's (1996: 66) social actor representation diagram	32
i i	Figure 4.1 Expanded version of Concordance line 2 in 4.2	80
1	Figure 4.2 Extended version of Concordance line 3 of the collocate ' V ' ($l\bar{a}$; no)	89
-	Figure 4.3 Expanded version of concordance line 6 in Concordance 4.9	98
	Figure 4.4 Expanded version of concordance line 1 in Concordance 4.11	102
1.1.2	Figure 4.5 Expanded version of line 3 in Concordance 4.18	117
	Figure 4.6 Expanded version of concordance line 2 in Concordance 4.19	119
	Figure 5.1 Expanded version of Line 3 of the collocate 'التي، (<i>'allati</i> : that) in Alhawali's corpus	130
1.00	Figure 5.2 Extended version of Concordance line 3 of the key term 'WAR' in Ibn Baz	149
and a second second	igure 6.1 Two extracts from Alhawali containing the keyword 'رجال' (riğāl; men)	200
	igure 6.2 Expanded version of concordance line 3 in Concordance 6.7	204
	igure 7.1 Argumentation thread 1(beliefs and ritual practices)	226
1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	igure 7.2 Argumentation thread 2 (the Salafi definition of worship)	228
	igure 7.3 Arguments structures in argumentation threads 1 and 2 in Ibn Baz	229
	igure 7.4 Argumentation thread 3 (Arab nationalism)	231
	igure 7.5 Reconstruction of argumentation thread 3 in Ibn Baz	232
	igure 7.6 Argumentation thread 4 (faith in Sacred Books)	235
	igure 7.7 Reconstruction of Ibn Baz's arguments in argumentation thread 4	236
	igure 7.8 Argument 1 in argumentation thread 1 (Western women and social life)	238
	igure 7.9 Argument 2 in argumentation thread 1 (Western women and social life)	239
	igure 7.10 A report and statistics used by Alhawali as warrants (argument 3: Western women and social life)	240
	igure 7.11 Reconstruction of Alhawali's arguments about the values of western societies (thread 1)	241
	igure 7.12 The Church and the original sin (argumentation thread 2)	244
	igure 7.13 Reconstruction of Alhawali's argumentation of the original sin (argumentation thread 2)	246
5.	igure 7.14 United States Invades the Gulf and the Arabian Peninsula (argument 1 in argumentation thread 3).	247
1	igure 7.15 American and British press and the war against Iraq (argument 2 in argumentation thread 3)	248
4	igure 7.16 Screenshot of the first paragraph of the original article from the 'EIR'	248 249
100	igure 7.17 Reconstruction of Alhawali's arguments about political enmity (argumentation thread 3)	249

ないのであるというないないのであるとう

1

List of Concordances

į

Concordance 4.1 Concordance of 'لَيَنْصُرَنَّ (layanşuranna; verily (Allah) will aid)	78
Concordance 4.2 Concordance of the collocate النيوم (<i>'alyawm</i> ; (the Last) Day)	78 79
Concordance 4.3 Concordance of the collocate 'الْخَالِصُ' (<i>'alhāliş</i> ; the pure)	81
Concordance 4.4 Concordance of the collocate 'سلم' (sallam; (may Allah) send His peace (upon him))	84
Concordance 4.5 Concordance lines of the collocate 'צ' (<i>lā</i> ; no) in Alhawali	87
Concordance 4.6 Concordance lines of the collocate' الحمد' ('alḥamd; praise) in Alhawali	91
Concordance 4.7 Concordance lines of 'الخرافات' ('alhurafat; superstitions), 'الخرافات' ('alkufr: infidelity) and ': البدع' ('albida'; impermissible innovations in religious practices) in Ibn Baz	95
Concordance 4.8 Concordance of 'الشرك' ('ašširk; polytheism) in Ibn Baz	95 96
Concordance 4.9 Concordance of شرك (<i>širk</i> ; polytheism/associationism) in Ibn Baz	90 97
Concordance 4.10 Concordance of 'الشرك' ('ašširk; the polytheism/associationism) in Alhawali	99
Concordance 4.11 Concordance of شرك (<i>širk</i> ; polytheism) in Alhawali	101
Concordance 4.12 Concordance of 'الاستغاثة' ('al'istigatah; beseeching for help) in Ibn Baz	106
Concordance 4.13 Concordance of 'صفاته' (<i>sifātoho</i> ; His attributes) in Ibn Baz's corpus	107
Concordance 4.14 Concordance of the collocate 'رحمه' (<i>raḥimahu</i> ; may (Allah) have mercy on him) in Ibn Baz	110
Concordance 4.15 Concordance of 'الإسلام' ('al'Islam; Islam) in Ibn Baz	111
Concordance 4.16 Concordance of 'الإسلام' ('al'islam; Islam) in Alhawali	112
Concordance 4.17 Concordance of the keyword 'الأخرة' ('al'āhirah; the Hereafter) in Ibn Baz	115
Concordance 4.18 Concordance of the keyword 'النار' ('annar; the Hell-fire) in Ibn Baz	116
Concordance 4.19 Concordance of the keyword 'النار ('annar; the Hell-fire) in Alhawali	118
Concordance 4.20 Concordance of the keyword 'الأخرة' ('al'āhirah; the Hereafter) in Alhawali	120
Concordance 5.1 Concordance of the collocate 'التي' (<i>'allati</i> : that/which) in Alhawali's corpus	129
Concordance 5.2 Concordance of the word form 'الغربية' ('algarbiyyah; western -adj. fem.) and its	
collocate 'الحضارة' ('ahadarah; civilization) in Alhawali's corpus	130
Concordance 5.3 Concordance of the word form 'الغربية' ('alġarbiyyah; western –adj. fem.) and its collocate 'الغربية' ('alġarbiyyah; western –adj. fem.) in Alhawali's corpus	132
Concordance 5.4 Concordance lines of the word form 'الغرب' ('algarb; the West) in Ibn Baz's corpus	134
Concordance 5.5 Concordance lines of the word forms 'الغربي' (<i>'alġarbi</i> ; Western –adj. masc.) and (<i>'alġarbiyyah</i> ; western –adj. fem) in Ibn Baz's corpus	135
Concordance 5.6 Concordance lines of the keyword 'اسرائيل' ('isra'īl; Israel) in Alhawali's corpus	137
Concordance 5.7 Concordance lines of the key word دي بالإسرائيلية ('al'isra'īliyyah; Israeli) in Alhawali's corpus	139
Concordance 5.8 Concordance lines of اسرائیل 'isra'īl; Israel) in Ibn Baz's corpus	140
Concordance 5.9 Concordance of the keyword 'الحرب' ('alharb; the war) in Alhawali's corpus	144
Concordance 5.10 Concordance of the keyword 'الصليبية' (' <i>alşalebiyyah</i> ; Crusades) in Alhawali's corpus.	145
Concordance 5.11 Concordance of the key term 'WAR' in Ibn Baz's corpus	147
Concordance 5.12 Concordance of the word form 'القرن' (<i>'alqarn</i> ; the century) and its collocate 'مطلع' (<i>matla</i> '; (the) beginning (of)) in Alhawali's corpus	151
Concordance 5.13 Concordance of the cluster 'القرون الوسطي) '('alqorūn 'alwostā; the Middle Ages) in	
Alhawali's corpus	153

Concordance 5.14 Concordance of the word form 'القرن) ('alqarn; the century) in Ibn Baz	155
Concordance 5.15 Concordance of 'القرون' 'alqorūn; the generations) in Ibn Baz's corpus	157
Concordance 5.16 Concordance of the word forms of the key term 'MIND' in Alhawali	160
Concordance 5.17 Concordance of the word forms of the key term 'MIND' in Ibn Baz	163
Concordance 5.18 Concordance of the word forms of the key term 'RELIGION' in Alhawali	168
Concordance 5.19 Concordance of the key term 'RELIGION' in Ibn Baz's corpus	173
Concordance 6.1 Concordance of دعاة' (do 'āh; propagandists) in Ibn Baz	184
Concordance 6.2 Concordance of 'teachers/ propagandists' (do 'āh; دعاة) in Alhawali	187
Concordance 6.3 Concordance of 'أهل' ('ahl; the people of) in Ibn Baz's corpus	191
Concordance 6.4 Concordance of 'أهل' ('ahl; the people of) in Alhawali	192
Concordance 6.5 Concordance of رجال الدين' (riğālo 'addeen; clergymen) in Alhawali	198
Concordance 6.6 Concordance of 'رجال) (<i>riğāl</i> ; men) in Ibn Baz	200
Concordance 6.7 Concordance of 'الشيوعية' ('aššyo 'iyyah; communism/communist) in Alhawali	202
Concordance 6.8 Concordance of 'الشيوعية' ('aššyo 'iyyah; communism/communist) in Ibn Baz's	
corpus	205
Concordance 7.1 Concordance of the collocate (أفَاعْبُدُونَ (fa bodūn; therefore worship Me) in Ibn Baz's	
corpus	219
Concordance 7.2 Examples of noun collocates of the keyword 'قال' (qala; He/he said) in Ibn Baz Concordance 7.3 Examples of concordance of word clusters that contain the keyword 'قال' (qala; He/he	220
said) in Ibn Baz	223

List of Abbreviations

acc.	accusative
adj.	adjective
adv.	adverb
CDA:	Critical discourse analysis
def.	definite
DHA:	Discourse historical
DIIA.	approach
fem.	feminine
gen.	genitive
imp.	imperative
indef.	indefinite
masc.	masculine
n.	noun
nom.	nominative
pl.	plural
sing.	singular
voc.	vocative

Part one

CHAPTER 1: Introduction

1.1 Overview

This study is concerned with the context of modern Saudi Arabia in the period 1980-2000 focussing particularly on the Salafist discourses of two influential Saudi writers, AbdulAziz Ibn Baz and Safar Alhawali (see 1.3.7 below). I use a combination of critical discourse analysis and corpus linguistics techniques in order to compare the discourses contained within the writings of these two influential men. I am particularly interested in how the two writers used language to represent the 'other' in a way that legitimates their own position and delegitimates the position of their opponents (for more information about 'othering' see Said 1978). This chapter offers an introduction to the topic of my study. I shall start by stating the research problem (in section 1.2) followed by section 1.3 which provides some relevant historical and politico-religious background with a special focus on the main tenets and the evolution and rise of the General Islamic awakening (Salafist) trends in Saudi Arabia. In section 1.4, I delineate the aim and scope of the study. Section 1.7 gives the rationale for the study. I state my research questions in section 1.8 and, finally, I give a brief account of the thesis structure in section 1.9.

1.2 Research problem

This research is intended to problematize what I refer to as Salafist discourses in modern Saudi Arabia (1980-2000). In Arabic, the word 'سلف' (salaf) means one's predecessors. In Islamic Law, the concept of Salafism has a variety of meanings, all of which refer to the early form of Islam as practiced by the so-called *pious predecessors* (considered to be superior in

religious status). The appellation *pious predecessors* refers to the companions of the prophet Mohammad, their successors and the immediate followers of the successors (Haddad, 2004), who lived in the first two centuries of the Hijra calendar¹ (Salafism is further detailed in section 1.3.7 below). However, reading through modern Salafi writings such as those of Ibn Baz (see Section 1.3.7 below), this appellation may indicate most of the medieval religious leaders. This difference in interpretation of the term leads us to the problem addressed in this research (Salafism).

First though, a distinction between the concepts 'Salafi' and 'Salafist' needs to be drawn academically. Whereas the concept 'Salafi' denotes the actual teachings of the Salaf, 'Salafist' strictly designates the ideological process of recontextualizing these teachings towards political, religious or social ends. Salafism is a movement among Sunni Muslims and is sometimes seen as related to or even synonymous to Wahhabism (an extremely conservative branch of Islam), although many Salafists object to the association (Roy, 1994, pp. 118-119).

Unfortunately, the dominant type of research on Salafism is confined to content analysis which is focused only on historical accounts of Salafist trends in Saudi Arabia or thematic descriptions of religious beliefs of the Salafist factions with no or at best slight reference to the discursive aspects of the texts produced by modern Saudi Salafist writers.

Since the terrorist attacks of $9/11^2$ many western writers such as Oliveti (2001) and Arab writers such as Haddad (2004) have blamed those tragic events on Saudi Salafism, which has been seen as a collective puritanical movement in Islam. I would argue that this is

^{1.} The Hijrah calendar or Islamic calendar is used in reference to the emigration of the Prophet of Islam from Makka to Medina in 622 AD. This emigration is taken as the starting point of the Hijrah calendar.

² On September 11, 2001, members of the militant Sunni Islamist terrorist group al-Qaeda hijacked four American Airlines passenger planes, flying two of them into the Twin Towers in New York City and another into the Pentagon. Passengers in the fourth plane fought the terrorists and the plane crashed into a field. Almost 3,000 people died as a result of the attacks. The leader of al-Qaeda was Osama Bin Laden, a member of a wealthy Saudi family, who had been raised as a Salafi Muslim. He was killed by American forces in Pakistan on May 2, 2011.

a problem because such writers tend not to make reference to distinctions within the Salafist tradition itself and thus make an over-generalisation which negatively characterizes innocent people. Part of my aim in this thesis is to raise awareness of heterogeneity within Salafism by contrasting the discourse of two Salafi leaders from Saudi Arabia.

1.3 Political and religious context

In this section I provide information about Saudi Arabia, my home country from which my data originates from. Saudi Arabia is a country in the Middle East, which contains the world's largest oil reserves. It was founded in 1932 by Abdul-Aziz, although its creation began in 1902 with the capture of the city of Riyadh by King Abdul-Aziz and his followers (Al-Farsy, 1990).

The Saudi government is an Islamic religious monarchy. The king complies with Sharia (Islamic) law based on the Qur'an (the religious text of Islam) and the Sunnah (the sayings and doings of the Prophet of Islam). There are no political parties or national elections, and power is handed down through the descendants of the Royal family, with approval of religious leaders who are known as the 'alust' ('olama; ulama or religious leaders).³

Following the teachings of Mohammad ibn AbdulWahhab, the majority of Saudis exercise a strictly monotheistic norm of worship. Ibn AbdulWahhab is a religious reformer (1703 -1792) who is considered to be the founder of the Najdi⁴ religious movement that mostly conforms to the teachings of Ahmad Ibn Hanbal⁵ and Ibn Taymiyya⁶. In collaboration with Mohammad Ibn Saud –the founder of the first Saudi state (1745-1811) (Vassiliev, 1998), the Najdi movement flourished in Najd (Niblock, 2006) despite successive attacks

^{3. &#}x27;ulama' is the Arabic word for those who possess knowledge and, in this context, religious knowledge (see the glossary).

^{4.} The Arabic adjective 'Najdi' is derived from the noun Najd which denotes the central region of Saudi Arabia.

^{5.} Ibn Hanbal, Ahmad is the founder of a religious-legal school called Hanbalism in the ninth century.

^{6.} Ibn Taymiyya, Ahmad (1263-1328) is a medieval jurist and theologian (Lacroix, 2009) who adhered to the Hanbali School.

promulgated by the Ottomans and their allies in the area (Vassiliev, 1998). Such collaboration between the religious and the political elements can also be traced in the second (1843-1865), and then the modern (established in 1932), Saudi states. The teachings of Mohammad Ibn AbdulWahhab and his followers will be sketched out in the following section.

1.3.1 The main tenets of the Najdi religious movement

The followers of Mohammad Ibn AbdulWahhab assert that they follow the 'السلف الصالح' ('assalaf 'assālih; pious predecessors)⁷ (see Section 1.3.7), in all aspects of their faith and ritual practices such as prayers, performing 'حج' (hağğ; pilgrimage), paying 'زكاة' (zakat; alms) etc. Following the teachings of Ibn Taymiyya and some aspects of the 'حنبلی' (hanbali; Hanbali) School⁸ of jurisprudence, the teachings of the Najdi movement were mostly based on avoiding 'جاهلية' (ğahilivyah; the situation of ignorance) that closely resembled the life of pre-Islam Arabs.

Mohammad Ibn AbdulWahhab and his followers instruct that any valid religious judgment is to be based on the Qur'an and the Sunnah, and on the 'إجماع' ('iğmā'; the consensus) of the pious predecessors. Additionally, he and his followers advocate 'التوحيد' ('attawheed; Oneness of Allah -or Islamic Monotheism) as perceived from 'القرآن' ('alqur'ān; Our'an) and 'الحديث' ('alhadith; Prophetic traditions).

The concept of Islamic monotheism consists of three tenets. The first tenet is 'unity of Lordship'. The original Arabic term for 'Lordship' is 'الربوبية' (Arrabūbiyyah) which is derived from the word 'رب' (rab; Lord). The followers of Mohammad Ibn AbdulWahhab, as well as all other Muslim factions such as the Sufis and Salafis, stress the unity of Lordship which indicates that Allah is the creator of the world.

^{7.} Translation and transliteration schemes are detailed in Section 3.3.2.

^{8.} The Hanbali School is one of major four canonical schools of jurisprudence in Sunni Islam; the other three are 'شافعي' (šafi 'i; Shafii), 'مالكي' (maliki; Maliki) and 'حنفي' (hanafi; Hanafi).

The second tenet is 'توحيد العبادة' (*tawheed 'al'luhiyyah*; unity of worship). According to Mohammad Ibn AbdulWahhab and his followers, this indicates that all Muslims should single out Allah as the only one to worship, i.e. not to include other figures and objects alongside Allah in their worship. Any innovative worshipping practice, which is not mentioned in the Qur'an or the Sunnah, is considered 'بدعة' (*bid'ah*; impermissible religious practice), that could lead to 'شرك' (*širk*; polytheism) (Abdul-Wahhab, 1996). In this respect, it is considered that those (mostly Sufis and Shiites) who include other figures and even objects alongside Allah in their worship are not only 'مبتدعة' (*mobtadi 'ah*; innovators of religious practices), but also practicing 'شرك' (širk; polytheism), and more specifically violators of the concept of the Unity of Worship.

The third tenet of monotheism is 'توحيد الأسماء والصفات' (tawhīd 'ala'sama' waşifat; unifying Allah in His names and attributes). This category of monotheism is based on the conviction that Allah has the excellent and perfect names and attributes. In order to adhere to this tenet, Muslims should call on Allah by His names and attributes taking into consideration the 'الإلى 'ithāt; confirmation) of these names and attributes i.e. attesting to all Allah's names and attributes as reported in the Qur'an and confirmed by the prophet. The followers of Mohammad Ibn AbdulWahhab confirm that the 'ithāt' of Allah's names and attributes requires neither 'ize names and attributes i.e. denying any of them. They also confirm that Muslims should neither compare those names and attributes to any human characteristics by assigning to Allah the attributes that belong to His⁹ creatures, 'tašbīh; similitude) nor modify their essence by 'ize (takyyīf; assigning them to a certain form) (Abdul-Wahhab, 1996). 'تحديف 'ialigning' 'i

⁹ As a practising Muslim I use the conventional way of referring to Allah with a first initial capital when using the third person pronoun.

very important religious concept in the Najdi religious movement. It is a well established concept¹⁰ in Islam and forms one of the official religious institutions in Saudi Arabia.

After sketching some of the main tenets of the Najdi religious reforms, I move on to outline the religious and political uprisings that took place in Saudi Arabia in the period between the 1920s and the 1970s as a means of providing a historical contextualisation of my own period of analysis.

1.3.2 Religious and political unrest in Saudi Arabia

Beginning in 1901, AbdulAziz (1876-1953) attempted to reclaim the Najd territory from which his family had been displaced since 1890 when the Al Rashid conquered Riyadh. He was successful in 1902, recapturing the city and becoming King of Saudi Arabia (Figure 1.1 shows a political map of the region).



Figure 1.1 Map showing Saudi Arabia and neighbouring countries (source Google Maps 2013).

The 'اخوان نجد' ('ihwān nağd; the brethren of Najd)11, provided AbdulAziz with ascetic military force (Habib, 1978). They were nomadic supporters and advocates of the Najdi

^{10.} For more information about the 'الحسبة' ('alhisbah; enforcement of religious laws) in Islam, see Cook (2000).

religious movement who were encouraged by AbdulAziz to move to 'مجر' (hiğar; settlements) (Habib, 1978). They were part of King AbdulAziz's army during his wars that resulted in annexing the different areas of modern Saudi Arabia to Najd. However, the alliance between AbdulAziz and the 'ihwān was not to last.

Though the '*iḥwān* of Najd adhered to the teachings of Mohammad Ibn AbdulWahhab, which maintain that it is the prerogative of the ruler to determine when conditions warrant jihad (holy war) (Commins, 2006), their call for continuous *jihad* did not halt. The '*iḥwān*'s loyalty to AbdulAziz started to decrease as he prohibited them from plundering areas they considered to be the abodes of infidels. The conflict between AbdulAziz and the '*iḥwān* was elevated as they accused him of increased contact with the British and introducing modern 'satanic' inventions such as the telegraph and automobile (Habib, 1978). The '*iḥwān* were eventually subdued by AbdulAziz and his forces in the Battle of Sibila in March 1929 and subsequently the chieftains of the rebels who had their own political agendas were imprisoned (Habib, 1978; Hegghammer & Lacroix, 2007).

The period from the 1930s until the mid 1970s did not witness any religious uprisings in Saudi Arabia except that of the puritanical protest which was led by Prince Khalid bin Musaid in the mid-1960s, in opposition to King Faisal's¹² policies of incremental modernization (Dekmejian, 1994). Some episodes of secular nationalist unrest appeared in the 1950s and 1960s where the most noticeable was that of the Saudi nationalist uprising among the workers in the oil sector.

The Saudi nationalist movements (referred to as 'Labour Movements') started to influence the workers of the then joint American-Saudi petroleum company (ARAMCO) in

^{11.} In order to differentiate the 'الإخوان المسلمون' ('*Iḥwan*; the brethren) from the Egyptian 'almoslimūn; Muslim Brotherhood), they are sometimes called the 'الجوان نجد' ('iḥwān nağd; the brethren of Najd.

^{12.} AbdulAziz died in 1953 and his second son, Saud became King in 1953 until 1964. In 1964, AbdulAziz's third son Faisal became the king and remained in power until 1975.

the 1950s and 1960s (Commins 2006; Fandy, 2001). They were inspired by the socialist reforms of Abdel Nasser, the president of Egypt between 1956 and 1970, who espoused Pan-Arabism, wanting to unite the people and countries of the Arab World. Pan-Arabism is closely connected to the idea of Arab Nationalism, or the idea that Arabs constitute a single nation. Other movements, such as the communist 'بعث' (*ba't*; renaissance –Baath) Parties in Syria and Iraq had also inspired Saudi nationalist movements (Hegghammer & Lacroix, 2007; Mordechai, 1993; Sager, 2005). However, the Saudi nationalist movement was not successful as the Saudi rulers relied on religious beliefs (incorporated within social structures) to thwart such movements and retain social control.

The conflict between Egypt's president Abdel Nasser and King Faisal –the third king of Saudi Arabia between 1964-1975– made the latter, who relied on religious forces, host members of Egyptian and Syrian Muslim *Brotherhood* parties after they were expelled from their countries (Vassiliev, 1998). Most of these members were well trained academics and job seekers (see Section 1.2.3 below). After arriving in Saudi Arabia, they were employed at universities and schools where they constituted the backbone of the newly established Saudi education system (Alotaibi, 2007; Hegghammer & Lacroix, 2007).

Under the influence of the aforementioned religious activists alongside the traditional and official religious institution, two religious Salafi-oriented¹³ trends developed; one was pragmatic, political and elitist, representing the mainstream Islamist movement called ' الصحوة (*'aṣaḥwah 'al'islamiyyah*; the Islamic awakening) or simply 'الإسلامية (*'aṣaḥwah*; the awakening) –it is also written as Sahwah in some western sources such as Hegghammer and Lacroix (2007). The other trend was an isolationist, pietistic and low-class Islamist movement (referred to as rejectionists). The first *Islamic Awakening* trend emerged clearly during the 1980s (Hegghammer & Lacroix, 2007) whereas the second type of Islamist movement played

^{13.} The term 'salafi' refers to the early form of Islam that was practiced by the pious predecessors (see Section 1.3.7 below).

an important role in the constitution of a militant group that seized the Muslims' holiest mosque of Makkah in 1979 (Commins, 2006).

1.3.3 The siege of the Grand Mosque in Makkah

On 20 November 1979 (shortly after the Islamic revolution in Iran) a group of Islamic militants led by Juhayman Alotaibi (born in 1940) seized the Grand Mosque in Makkah (the Haram) for two weeks (Cummins 2006; Hegghammer, 2008). His main claims were more or less the same as those of the early '*ihwān*; anger at the increased and continuous contact with what his group considered as infidels, and he also criticized television programs shown in Saudi Arabia that he claimed were against the values of Muslims. The siege of the Grand Mosque ended on 4 December 1979 and as a result Juhayman and the other militants were imprisoned. He and some others were executed on 9 January 1980. Then, almost 10 years after the siege of the Grand Mosque, a new episode of political and military turmoil in the Gulf region occurred.

1.3.4 The Iraqi invasion of Kuwait

Saddam Hussein the president of Iraq, who had been at war with neighbouring Iran (see Figure 1.1) for almost a decade, launched an attack on Kuwait shortly after the war with Iran ended. On August 2, 1990, he invaded and occupied Kuwait. Subsequently, Iraqi troops were forced out of Kuwait by American-led allies, in what was known as the Gulf War, code-named Operation Desert Storm by the Americans. Despite the fatwa issued by AbdulAziz Ibn Baz (see below) allowing American-led troops to come to Saudi Arabia, the presence of the American military in Saudi Arabia angered some religious leaders who believed that they had come to occupy the land of Muslims. Safar Alhawali (see below) was one of the most prominent religious leaders who opposed the presence of the American-led troops in Saudi Arabia (Fandy, 1998). These religious leaders adhere to the tenets of the religious awakening

trends in Saudi Arabia. In the following section I shall sketch out the Islamic awakening trend and its main tenets.

1.3.5 The rise of the Sahwah

The trend of the Islamic awakening prevailed just after Juhayman's movement was thwarted. The 1980s witnessed a clear emergence of Islamic awakening as a response that could be guided by both religious and political opposition to the Khomeini's revolution in Iran and the Afghani '*jihad*' against Russia. It flourished among the students in the different Saudi schools and universities under the influence of Islamic activists who included either the advocates of Mohammad Ibn AbdulWahhab's teachings or the Egyptian and Syrian Muslim Brothers who started to come to Saudi Arabia in the 1960s. Despite the fact that the religious teachings and the official curricula of religious courses in Saudi Arabia are mostly based on the teachings of Sayyid Qutb. It is worth noting that both Egyptian and Syrian Muslim Brothers adhere to the teachings of Sayyid Qutb.

Sayyid Qutb (1906-1966) is an Egyptian intellectual whose writings range from literary to religious dimensions. His religious writings were prominent in moulding the Sahwah movement. His influence on the modern Islamic movements, not only in his home country, Egypt, (Hanafi, 1988), but the Muslim world at large, is profound (Naseerah, 2007). In Saudi Arabia, adherents to Sayyid Qutb (referred to as Qutbis) drew attention to his books such as (المستقبل لهذا الدين) (*maʿālim fi 'aṭṭarīq*; Milestones), المستقبل لهذا الدين) (*almustaqbal li hāda 'addīn*; The Future is for this Religion) نعي (*fi zilal 'alqur'ān*; Under the Patronage of Qur'an) etc. These books were given high priority to be celebrated and distributed even in the schools of the Ministry of Education.

Although Sayyid Qutb did not go to Saudi Arabia, his younger brother Mohammad Qutb (1919-present) is another important figure who did settle in Saudi Arabia. Mohammad worked as a teacher in the Um al-Qura University in Makkah. He, along with his brother Sayyid, published and theorized the main tenets of the modern Islamic awakening movement. Mohammad had great influence on his students including one of the writers who is the main focus of this thesis, Safar Alhawali (see below) who was his advisee in both MA and PhD programs in the 1980s. And, as Alotaibi (2007) notes, Muslim Brotherhood activists in Saudi Arabia incorporated the ideas of Sayyid Qutb' into the Salafi Najdi religious backgrounds. In the following section I delineate the main tenets of the Sahwah.

1.3.6 Main tenets of Sahwah

The major tenets of the Sahwah movement are composed of the aforementioned tenets of the Najdi religious reforms (section 1.3.1) that were advocated by the official religious leaders as well as the advocates of Ibn AbdulWahhab's teachings and the ideas of Sayyid Qutb and his younger brother Mohammad Outb that were incorporated into the teachings of Mohammad Ibn AbdulWahhab. The two important concepts, which were primarily advocated by the adherents of Sayvid Qutb, are 'الحاكمية' (alhākimiyyaha; sovereignty of Allah) - or unifying Allah in His sovereignty – and 'الجاهلية' ('alğāhliyyah; ignorance or the religious state of Arabs before Islam). The first of these two concepts appeared in the writings of Abul Ala Maududi (1903-1979), a famous Indian Islamist author, who is considered by some modern researchers to be the founder of modern radical political Islamic movements (Aldayyani, 2007a). The concept of 'gahliyyah' refers to the status of religious ignorance and barbarism of the Arabs in the pre-Islamic era. Sayyid and Mohammad Qutb developed these two concepts in a number of their books. In a book called 'Milestones', Sayyid Qutb defines sovereignty as not limited to mere legal injunction, but includes the principle of administration, its system and its modes (S. Qutb, 2001). In this respect, most of those who quoted him considered all Muslim states to be not bearing the mark of Islam, and hence are states of 'ردة' (riddah; apostasy). Most religious leaders who adhere to the tenets of the

religious awakening trend use the Arab-Israeli conflict and the ongoing situation in Palestine as sign of weak Arab and Muslim countries that should be changed into more powerful states.

The concept 'جاهلية' (*ğāhiliyyah*) means ignorance or barbarism which is usually used to describe the state of the Arabs in the pre-Islamic period. Mohammad Qutb published a book using the term 'جاهلية' (*ğāhiliyyah*) in its title; 'جاهلية' (*ğāhiliyyat 'alqarn 'al'išrīn*; The *Ignorance* of the Twentieth Century) (1980). This concept is one of the common concepts that the writers of the Sahwah have used to denounce all modern Muslim and non-Muslim communities (Alnadi, 2007). This concept means that if these communities are indulging in ignorance and barbarism, then serious corrective action needs to be taken by the 'believers' using whatever means necessary to restore the previous status of the '¹/₂ ¹/₂ ¹/₂

The concept of 'الولاء و البراء' ('alwalā' wa albrā'; loyalty and disavowal) is also repeatedly mentioned in the books of Sahwah writers. It means that believers should have a friendly relationship amongst themselves and at the same time should show hostility against 'others' who are either infidels or not true believers i.e. Muslims that are considered to have deviated from the right path (Alotaibi, 2007).

As indicated above the Sahwah movement is Salafi-oriented. The writers of this movement (adherents to Ibn AbdulWahhab' teachings and the advocates of Sayyid Qutb's ideas) tried to base their teachings on the early writings of the Salaf¹⁴ thinkers such as the aforementioned Ahmad Ibn Hanbal and Ibn Taymiyya. In the following sections I shall write about the concept of Salafism in Saudi Arabia.

^{14. &#}x27;Salaf' is the Arabic word for predecessors.

1.3.7 Salafism and prominent Saudi Salafis

As I have stated in section 1.2 above, the appellation *Salafis* refers to *pious predecessors* (superior in religious status) who belong to the three generations of the early Muslims. They are also called 'القرون المفضلة' ('alqorūn 'al'olā; the early generations) or 'القرون المفضلة' ('alqorūn 'almofaḍalah; the best generations). Modern Salafis usually claim that they strictly follow the religious ritualistic practices and beliefs of the first three generations of Muslims (or the pious predecessors). They always refer to the Qur'anic verses and the Prophetic traditions and adhere to the literal interpretations and explications of these sacred texts as mentioned in the writings and sayings of the early religious leaders who belong to the first three generations. This is based on Prophetic traditions (Hadiths) such as:

Narrated Zahdam bin Mudrab: I heard Imran bin Husain saying, "The Prophet said, 'The best people are those living in my generation, then those coming after them, and then those coming after (the second generation) [...]" narrated in Sahih Bukhari, *Volume 3, Book 48, Number 819.*¹⁵

This form of religious practice in Islam is called Salafism (also see Section 5.5.2).

In Saudi Arabia, the Salafis are the followers of Mohammad Ibn AbdulWahhab (see Section 1.3 above). These include religious leaders and Sheikhs such as AbdulAziz Ibn Baz and Safar Alhawali.

1.3.7.1 Abdul Aziz Ibn Baz

One important traditionalist Salafi is AbdulAziz Ibn Baz (1912-1999) (henceforth Ibn Baz) who was one of the most prominent Saudi official religious leaders. He was active in publishing books and giving religious lectures in different Saudi cities. He worked in several official positions and under his leadership in the Islamic University of Medina, Ibn Baz encouraged his students to preach the main tenets of the Najdi religious movement (see

^{15.} Sahih Bukhari is a voluminous work by Albukhari (commonly referred to as Imam Albukhari 810 – 870). It is a collection of Prophetic Hadiths that are called 'sound' Hadiths The translation of this Hadith is available on: http://www.usc.edu/org/cmje/religious-texts/hadith/bukhari/048-sbt.php Accessed on 31/03/2013.

Section 1.3.1 above) and engage in 'الحسبة' ('alḥisbah; enforcement of religious laws) (Hegghammer & Lacroix, 2007). He became the Grand Mufti¹⁶ in 1993 and engaged himself with issuing 'فتوى' (fatwā; religious dicta). Before that, he issued many fatwas (including allowing American troops to be stationed in Saudi Arabia during the second Gulf War in 1991, and banning women from driving, which may be his most well-known ruling). He published many books regarding theological issues such as monotheism, pilgrimage and other pillars of Islam. The publications of Ibn Baz are made accessible on a website run by his advocates.¹⁷

1.3.7.2 Safar Alhawali

Safar Alhawali (born 1950) is a professor in the Muslim creed ('aqidah; العقيدة) at the department of Creed at Um Al-Qura University. Fandy (2001) characterizes Safar Alhawali (henceforth Alhawali) as being concerned with protecting his country and his culture from foreign domination and defending Islam against what he believes is the onslaught of secular Euro-American traditions. Alhawali became a well-known intellectual and distinguished figure not only in Saudi Arabia but the Arab world at large in the late 1980s and 1990s. His five cassettes about the Sufis in the Hijaz area (western region of Saudi Arabia) were widely distributed – 'الرد على الخرافيين' ('arrad 'ala 'alhorafiyyīn; Refutation of Mysticism) (Alotaibi, 2007). His previous academic work is of great importance to modern Salafis as he discussed the concept of secularism and this was the focus of his MA thesis that was supervised by Mohammad Qutb. One of his books, which has been highly quoted by religious leaders, is 'al'aālam' 'al'izāda' fi 'al'ālam 'al'islāmi; The Phenomenon of

^{16.} The Grand Mufti is the highest official religious leader who is entitled to issue religious dicta (fatwa; فترى). 17. The website of ibn Baz: <u>http://www.binbaz.org.sa/books/list</u>.

'Irjā' in the Islamic World)¹⁸. This book was his PhD thesis which was also supervised by Mohammad Qutb in 1985.

The aforementioned book alongside Alhawali's 'مقدمة في تطور الفكر الغربي والحداثة' (moqaddimah fi taṭawwor 'alfikr 'alġarbi walḥadaṯah; An Introduction to the development of Modern Western Thought and Modernism) are considered to be one of the early revival attempts to confront the secular and modernization process in Saudi Arabia. Alhawali's political views about the American-Israeli phenomenon occupied most of his books specifically during the second Gulf war; the Iraqi invasion of Kuwait which led to the war between Saddam Hussein's troops and the allied forces led by the United States (Fandy, 1998) whereas Ibn Baz (the Grand Mufti in Saudi Arabia –see above) issued a fatwa allowing such troops to defend the country. Alhawali's writings were made accessible on his website¹⁹ and other websites devoted to religious teachings such as Saaid.net.²⁰

1.4 Aim and scope of the study

The aim of this study is to investigate Salafist discourses during the period 1980-2000. As I have noted earlier, Salafism is often seen in simple terms as a united religious movement, as well as being associated with violent acts. I wish to problematize this hegemonic representation by comparing the discourse of two Salafi leaders, Ibn Baz and Alhawali. Towards this end the study draws on a composite discourse analytic approach that combines quantitative and qualitative methods. Having stated the aim and scope of studying Saudi-Salafi discourses, I now turn to my personal position.

^{18.} The concept of ''Irjā'' is linked to 'ايمان' ('iman; faith) in Islam. It means both delay and hope. Faith in Islam is a tripartite concept. It is composed of the following elements; believe with intention, vocalizing believe (i.e. accompanied with utterance/s) and action or performance of duties of worship -prayers, observing fasting during the days of Ramadan... etc (Awaji, 2001). Morjia'ah (the plural of Morj'a) are those who believe that it suffices for an individual to be a believer if s/he accomplishes the first two elements. But Salafis instruct that all the three elements of faith are supposed to be accomplished.

^{19.} Alhawali's website: http://www.alhawali.com/index.cfm?method=home.cat&categoryID=2.

^{20.} Alhawali's page on Islam.net: http://saaid.net/Warathah/safar/k.htm.

1.5 Position of the analyst

Baxter (2003, pp. 59-60) advocates the employment of self-reflexivity in research, which involves being clear about one's own position. Researchers should also be self-aware about the fictionality and textuality of the research process. Therefore, in this section I provide some relevant background information for the reader to explain my own interest in this topic, and how my experiences have given me certain insights which potentially can aid my research but may also influence my perspective.

I am a Sunni Muslim who was born and raised in Saudi Arabia. I was brought up to believe that Islam is connected to peace and human welfare. In 1979, when I was a student at intermediate school, armed Islamist dissidents took over the Grand Mosque at Makkah (Mecca) in Saudi Arabia.²¹ Despite (or perhaps due to) the fact that those who carried out the siege were captured and publicly beheaded, this incident was followed by incremental Islamisation in the country (as well as extremist attacks aimed at toppling the Saudi monarchy) that resulted in the so-called Islamic general awakening or simply the Sahwah (detailed in 1.3.5 above). As part of this, new terms started to appear such as 'Laudi' (*'assalafiyyah*; Salafism). I witnessed the rise of Sahwah in the 1980s.

I have worked for almost 18 years in a college that was nearly controlled by some Salafi individuals during the 1990s and several years of the first decade of this century. During these years, I experienced difficulties with Salafis for two reasons; to some Salafis, I was a graduate of a 'non-Islamic' department (English and European Languages) and I have no beard. Most Salafis present themselves as Allah's proxies, arguing that their opinions are derived directly from the word of Allah. I have always sought to understand the Salafis' arguments and position towards the 'Other'. In this study I do not intend to judge the Salafi

^{21.} Their goal was to institute a theocracy in preparation for an apocalypse. Some of their followers were students of religion at the Islamic University of Medina.

writings²² as right or wrong. I instead want to show how recent writers in Saudi Arabia have recontextualised Allah's words, particularly in relation to ideologies, argumentation and representations of the 'Other'.

1.6 Approach of the study

In this study, I intend to carry out a critical discourse analysis of the key texts produced by two prominent Saudi Salafis in the period before 9/11. It is through the analysis of these writings in terms of their ideological representations that variation in Salafist discourse in contemporary Saudi Arabia can be pinpointed. For the purposes of my research, I use the term *discourse* to mean:

"a cluster of context-dependent semiotic practices that are situated within specific fields of social action, socially constituted and socially constitutive, related to a macro-topic and linked to the argumentation about validity claims such as truth and normative validity involving several social actors who have different points of view" (Reisigl & Wodak, 2009, p. 89).

Following van Dijk (1998, p. 8), I define Ideology as 'the basis of the social representations shared by members of a group'. 'Ideologies help establish and maintain asymmetric power relations that exist through discourse', and 'function as a means of transferring power relations more or less radically' (Reisigl & Wodak, 2009, p. 88).

I use the triangulatory approach of Discourse Historical Analysis (Reisigl & Wodak, 2009) which is based on a broad and multi-faceted concept of 'context' that considers the immediate language or text-internal cotext²³ and co-discourse, intertextual and interdiscursive relationships between texts, genres and discourses, as well as the context of situation and the broader socio-political (and religious) and historical context. I shall be focusing mainly on

^{22.} Here I refer to the sacred Salafi texts, the Qur'an and the Sunnah.

^{23.} Cotext 'is used to refer to the linguistic environment which precedes or follows a concrete text location (Titscher, Meyer, Wodak, & Vetter, 2000).

how social actors are represented in Salafi texts, and how various argumentation schemes are used to legitimize particular ideologies.

In order to maintain critical distance from my data, I am using corpus methods that incorporate software to determine salient argumentative and/or linguistic aspects that contribute to the negative or positive presentations of social actors. This is described in more detail in chapters 4, 5, 6 and 7.

1.7 Rationale of the study

The devastating 9/11 terrorist attacks set off a chain of events that resulted in further devastation on a wide scale. Two countries were invaded and suffered huge structural and human losses; viz. Afghanistan and Iraq. A large amount of literature on Islam and fundamentalism has appeared, and above all, books and writings that link terrorism to Islam in general and Saudi Salafis/'Wahhabi' in particular were published. These books range from polemics which link extremism to Salafism and Saudi Arabia such as Haddad (2004) and Oliveti (2001) to the more academic writtings that try to explain the main tenets and history of Salafism such as Hegghammer (2008, 2009) and Fandy (2001) (see Chapter 2 section 2.2.2 for more details).

On the one hand, none of these writings have been conducted from a CDA perspective which indicates the importance and rationale for this study. On the other hand, the texts under investigation belong to a context that is of major importance for the contemporary world: Saudi Arabia contains the two most sacred Mosques in Makkah and Medina which make Saudi Arabia in the eyes of most Muslims to be the abode of Islam. And on a different plane, Saudi Arabia contains one of the World's largest Oil reserves; an important economic means for the whole world. Saudi Arabia is also associated with the 9.11 attacks, due to the fact that this was Bin Laden's birthplace. It is thus important that academic

research attempts to understand the complex discourses that influence the direction that Saudi Arabia takes, as well as presenting a clearer picture of the nation to the outside world.

The methods I use for this study, to the best of my knowledge, have never been applied to study Salafism in Saudi Arabia. These methods are designed to answer the research questions that I outline in the following section.

1.8 Research questions

This study responds to the following overarching question: To what extent and how is Salafi discourse in Saudi Arabia in the 1980s-1990s homogenous? In order to address this question, I shall consider the following sub-questions:

- 1. What do concordance and collocational analyses of keywords and frequent words in the two corpora of the Salafi discourse reveal about:
 - a. Themes/macro topics;
 - b. The representations of social actors (nominational strategies);
 - c. Argumentation indicators?
- 2. What argumentation schemes are employed in the texts to legitimize certain hegemonic meanings?
- 3. How can the findings in 1 and 2 be explained in relation to Saudi social and politicoreligious contexts?

1.9 Thesis structure

After setting the scene and the context of the study and giving a general introduction to the thesis in this introductory chapter, I carry out a critical review of relevant and related research in Chapter 2. Chapter 3 outlines the methodological framework that I utilize to answer the research questions. Chapter 4 is the first of five chapters of analysis. It is devoted to examining the semantic foci keywords in Ibn Baz's corpus and comparing their usages to that

of Alhawali's. In turn, in Chapter 5, I examine Alhawali's semantic foci keywords compared to Ibn Baz. Chapter 6 is devoted to investigating the representations social actors in both Ibn Baz and Alhawali's corpora. Chapter 7 examines arguments in both corpora. Chapter 8 concludes the analyses of my data where I provide explanations of the results obtained in the previous analysis chapters utilizing the social, political and religious contexts. Finally, Chapter 9 concludes the thesis.
CHAPTER 2: Critical Review of Literature

2.1 Introduction

It is useful to start this chapter by stating the main difference between two relevant types of religious writings: first, writings about religion that focus on religion(s) and faith communities without taking a religious perspective, which I label as 'meta-religious discourse'. Second, writings via which religious authors represent their semantic foci, ideological representations and legitimations. The second one I call religious discourse proper.

This chapter aims to describe and justify the quantitative and qualitative approaches deployed in analysing my research data. It also brings together critical discourse studies (CDA) and corpus linguistics (CL) as a way of analysing discourse and ideological representations. First, it (in section 2.2) evaluates (critical) religious studies identifying the gaps that such studies have failed to bridge. In section 2.3, I introduce critical discourse analysis approaches that could be used in analysing religious discourses. Section 2.4 introduces the field of corpus linguistics that will be employed in order to identify analytical foci, argument indicators and social actor representations. In section 2.4, I outline previous studies that have already attempted to combine CL and CDA, which have helped to provide frameworks for the way I intend to carry out the analysis of this study. Finally, section 2.5 concludes this chapter pointing out the points of departure for conducting this study. Let us first start with studies on religion.

2.2 Studies on religion

It is possible to view some types of writing about religion as oscillating between two extremes, one that advocates and preaches the teachings of a religion such as AbdulWahhab (1996). The other extreme is established on the rejection of the religion, and includes atheist writings such as Dawkins (2006) and Hitchens (2007). However, some texts on religion do not aim (explicitly at least) to convince readers of the existence (or not) of God. Instead, they aim to investigate religion from a more objective or distanced perspective²⁴. One such field is Religious Studies, and it is thus worthwhile examining what Religious Studies scholars have had to say about Salafism.

2.2.1 Religious studies

Religious Studies encompasses a wide range of approaches, and Segal (2006) introduces nine possible ways of studying religion that emanate from eight disciplines; anthropology, economics, literature, phenomenology, philosophy, psychology, sociology and theology whereas the ninth approach is the comparative method. Such research tries only to describe and explain the beliefs of the religion under consideration with no obvious critical stances.

Some researchers in religious studies have attempted to introduce critical aspects into their work. For example, Mittelstaedt (2002) proposes a framework called *religious authority* that could account for the relationships between religion and religious institutions and economic markets. Mittelstaedt concludes that the effects of religion on markets are multiple, including direct and indirect effects such as when theology becomes public policy which could impose market regulations.

Additionally, Fitzgerald (2003) has tried to investigate the ideological status of religion in academia. Although such studies and approaches could yield fruitful results in understanding how religions could affect, in one way or another, people's lives, they do not offer a perspective which is more closely based on critical analysis of discourse from a

^{24.} However, post-structuralists would reject the claim that an 'objective' perspective is ideologically neutral, as the unbiased research is in itself a 'discourse of science through which a particular version... of human life is constructed'. Thus, the desire to be objective could be seen as a 'position' or stance in itself (Burr, 1995, p. 160).

linguistically-oriented perspective in order to examine representations of religions and religious communities.

As I intend to study the Salafi discourses, it is useful here to outline the studies that considered Salafism (see Section 1.3.7 in Chapter 1). The following section delineates the type of studies that tackle Salafism and Saudi Salafis.

2.2.2 Studies on Salafism

Many writers have studied Salafism and Saudi Salafis including Aldayyani (2007a, 2007b), Allen (2006), Alotaibi (2007), Curtin Winsor (2007), DeLong-Bas (2004), Fandy (1998, 2001) Haddad (2004), Hegghammer (2008, 2009), Hegghammer and Lacroix (2007), Oliveti (2001) and Schwartz (2002). However, rather than being text-based analyses of Salafism, these writers tend to give more or less historical accounts of Salafism, although some of them also try to explain Salafi main tenets and beliefs. With the exception of those writers who are Arabs such as Fandy and Haddad or those who can read Arabic such as Delong-Bas, the writers of the above studies have depended largely on translations and previous works about Salafism and Sunni Islam or interviews with some Salafis interceded with interpreters.

Some Muslim and Arab writers, such as Haddad (2004), who takes up a Sufi (mystic) perspective, aims to deprecate the Salafi traditions, having a sectarian ideology. Such writers tend to focus on controversial aspects of Salafism, such as the veneration and seeking the help of dead holy men. For example, Haddad (2004, pp. 27-30) takes the case of a Salafi fatwa, presenting it out of context, in order to denounce the Salafi tradition. Another deprocatory action of Haddad is his attempt to denounce Abu Bakr al-jazairi, a Salafi relegious leader who worked in the Islamic University of Almadenah, Saudi Arabia, and taught in the Prophet's mosque in Almadenah.²⁵ Salafism also attracted the attention of

^{25.} For example, Haddad (2004, p. 88) mentions that al-jazairi wrote a book about the Prophet and confirmed a false story about the Prophet being misled by Satan. However when I read the original Arabic book, it appears that al-jazairi (2003, p. 99) denies that the Prophet was misled by the Satan.

western writers during the 1990s when it started to spread in Europe, but their writings were general in scope and focussed on radicalism where much of the research was carried out by anthropologists (Meijer, 2009). Meijer (ibid.) notes that the scope and focus of the research on Salfism changed after 9/11. He adds that 'a lot has been said and written on Salafism and Wahhabism, but much of this has been through the prism of "security studies" or books that play on the popular view that equates Wahhabism with violence' (2009, p. 2).

The writings on Salafism have thus tended to be based either on descriptive accounts of its history or have focussed on controversial issues surrounding its belief-system, and thus there is little research which has been carried out with a critical analysis of religious Salafist discourse; this holds for both the so-called 'sacred/canonical texts' created by members of a faith community and other meta-religious texts. The second type of texts may explain the 'canonical texts', which are thought to legitimize certain practices of individuals and religious/political institutions, or even define the relationships between members of a faith community and outsiders.

Generally, research under the remit of religious studies does not use a critical discourse analysis (hereafter CDA) perspective, considering issues such as power relations in religious discourses or the way that religious leaders represent the 'other' or the 'self' in their teachings (in spoken or written forms). However, one group which does attempt to consider the role of discourse in religion is based in Manchester (Green & Searle-Chatterjee, 2008), using an approach called Critical Religious Studies.

2.2.3 Critical religious studies

The so-called Manchester approach to religious studies is used to critically study religious discourse. In their introduction, Green and Searle-Chatterjee (2008) write that their main aim is to assess the global consequences of (mis)understanding political practice and social identities wrought through the language of religion. Based on the Foucauldian notion of the term *discourse*, they consider the language of 'religion' as discourse. They address issues of (a) power, giving an 'inclusive', but general in essence, 'concept of power as the ability to influence the bodies and minds of other men and women, typically by means of superior access to physical, cultural, or human resources', and (b) classifying 'others' and naming the 'self' (2008, pp. 8-12).

Researchers using the Manchester approach have addressed topics ranging from traditional to modern presentations of the 'self'/'insiders' and the 'other'/'outsider' as realized linguistically in the discourses of (sub)faith communities (Alexander, 2008; Christmann, 2008; Green, 2008; Gregory, 2008; Hirst, 2008; Klutz, 2008). However, despite taking a critical perspective, the writers of the articles collected in Green and Searle-Chatterjee (2006) do not conduct their analysis from the perspective of critical linguistics, critical discourse analysis or their forefathers, systemic linguistics or Bakhtinian discourse analysis etc., although Green and Searle-Chatterjee mention these approaches in their introduction. There is only a small reference in Klutz (2008) that acknowledges the use of some analytical tools from CDA along with collocational analysis.

The concept of critique, in the Manchester approach, remains vague; neither critical theories, ideologies, nor the analysts' positions, for example, are clearly addressed; a problem that may continue to persist in the critical religious studies that could be attributed to the sensitivity of criticizing religious discourses and, also may be because some researchers adhere, or at least belong to, the religions under investigation (see Section 2.3 below).

Moreover, writers in Green and Searle-Chatterjee (2008) only describe the way that social actors are/were presented in religious discourses. They neither explain nor critique nominations or hegemonic religious representations in the texts they analyzed. Additionally, none of the studies presented in Green and Searle-Chatterjee (ibid.) addresses Salafi discourse or modern Islamic religious trends. In this study, I wish to take an alternative

25

approach to the analysis of religious discourse, by examining a topic (Salafi discourse) which has not been subjected to critical analysis, and by using a critical discourse analysis framework. The following section therefore considers critical discourse analysis and its contribution to the study of social actors.

2.3 Critical Discourse analysis

Critical discourse analysis (CDA) is a well-established field (Fairclough, Mulderrig, & Wodak, 2011) in studying discriminatory/racist, prejudiced, anti-Semitic discourses (Reisigl & Wodak, 2001; van Dijk, 1993; Wodak, 1996) and political discourse (P. A. Chilton & Schaffner, 2002; Wodak, 2009). It emerged in the early 1990s (Wodak & Meyer, 2009a). CDA has developed an important agenda that potentially has social significance to the world today (Michael Stubbs, 1997). It is a collective effort that aims at emancipation and enlightenment. A key aim of CDA is to show how unequal and unfair power relations are maintained, how certain voices or positions are legitimated and others are ignored or oppressed. CDA is not against 'power' per se, but is concerned with abuses of power.

There are a number of 'schools' within CDA which differ from each other in terms of their actual methods used and their analytical foci, but generally all the schools stress the importance of combining an analysis of linguistic features within a text in order to identify discourses or ideologies with consideration of various types of social context in order to fully explain and evaluate the findings. Analysis in CDA takes place on multiple inter-connected levels and may involve examination of processes of text production (under what conditions was a text created and for what reasons) and reception (who consumed the text and how was it responded to). Furthermore, analysis can involve examination of intertextuality (the ways that texts can refer, explicitly or not, to other texts) and interdiscursivity (the ways that discourses can relate to or refer to each other, e.g. by supporting or competing).

26

The term 'critical' is considered by many scholars as a key concept in CDA. Fairclough (1992, p. 9) considers that the term 'critical' involves 'showing connections and causes which are hidden'. He (ibid.) also stresses that the term 'critical' implies 'intervention, for example providing resources for those who may be disadvantaged through change' (1992, p. 9). Wodak and Meyer (2009b, p. 7) state that 'critical theories, thus CDA, want to produce and convey critical knowledge that enables human beings to emancipate themselves from forms of domination through self-reflection'.

CDA has tended to focus on politically-oriented texts e.g. political speeches and debates, policy documents, newspaper editorials, advertising etc. and, hence, CDA studies on religious discourses in general and more specifically on Islam are rare. This may perhaps simply reflect the interests of CDA practitioners, although Salama (2011, p. 49) writes that within discourse communities critiquing religious discourse topics could be socially sensitive, if not risky'. He also adds that 'members of certain religious discourse communities cannot question the sacred texts of religion since their belief system is enshrined in those texts that are in turn above any critique' (2011, p. 49). And in order to account for intertextuality and interdiscursivity (from religious texts and discourses), I would add that analysing religious discourses per se critically requires immense knowledge about the history and origins of beliefs and rituals of the religion(s) and faction(s) involved. Readings in the different developments of faiths and beliefs of the different factions during different historical epochs is required in order to explore intertextuality and interdiscursivity. Thus religiousoriented CDA requires a great deal of knowledge about religious context and history, yet people involved in CDA tend to work in the field of applied linguistics, (or to a lesser extent sociology or politics) rather than religious studies.

Also, CDA originated within Europe and tends to be practised by westernised scholars in English-speaking contexts, although there is also CDA work in other European

languages like German. However, CDA studies on Arabic data (religious or otherwise) are presently very rare.

The above points could possibly explain the rarity of CDA studies on religious discourse in general and Islamic discourses in particular. However, rather than investigating religious discourse per se, there are a few studies that investigated the representation of religion and religious groups (i.e. investigating meta-religious discourses i.e. discourses that revolve around religion and religious communities rather than religious writings produced by religious leaders in any of the religious communities). But these attempts have tended to not incorporate CDA perspectives though (see below).

Hukam Sing and Thuraisingam (2011) have tried to investigate the role of language, as a means for reconciliation, in keeping religion relevant to both individuals and societies in the context of postmodernity. They studied religious sermons from three major faiths in Malaysia (Islam, Christianity and Buddhism), two sermons from each religion. They investigated the language used in sermons in terms of modes, quotations (intertextuality), vocabulary and phrases, personal narratives, religious meaning systems, norms and values and social context. Their eclectic approach is a mash of tools from both CDA adopted from Fairclough's framework of analysing language and power (2001) and the sociological theory of human activity (activity theory). However, they have tried to generalize their results to all three religions ignoring the ideological role of different sectarian beliefs that may have moulded the ideological stances of the preachers.

Rizwan (2011) has investigated the Pakistani Shiites' liturgical gatherings (called the 'majlis' which is derived from the Arabic word 'مجلس' (mağlis) which means gathering) to commemorate the death of Imam Hussein (the son of Imam Ali and the grandson of the Prophet of Islam). The commemoration is conducted in Urdu. Rizwan, utilizing a CDA methodology to explore the ideology behind the sermons in liturgical eulogies, has concluded

that in such sermons, preachers 'discursively produce the ideological mental paradigms which echo basic injunctions of Shi'ah [Shiite] Islam. The pathos of Imam Hussain's [Hussein] tragedy issues out of Shi'ahs' [Shiites] belief that an imam' is infallible (2011, p. 26) [the spellings of Hussain's and Shi'ahs are in original].

Recently, Hjelm (2013), on a purely theoretical basis, suggests implementing CDA methods in the critical sociology of religions. In this article, Hjelm (ibid., p. 1) discusses the relevance of CDA to a critical sociology of religion in three major areas: first, 'the concept of discourse and its different meanings'. Second, he 'examines what being 'critical' means in the context of discourse analysis and constructs a framework for doing practical CDA'. Finally, he discusses CDA as a foundation for a critical sociology of religion'.

In the Arabic context, Alharasi (2003) advocates implementing new approaches to Arabic texts, trying to open the way for studying discourses in Arabic texts utilizing CDA approaches. Alharasi has two objectives. First, he aims at shedding light on some basic concepts in western critical theory. Second, he proposes a thesis that could develop Arabic critical theories. Alharasi' paper, to the best of my knowledge, is the first paper written in Arabic that draws on CDA methodology. He first problematizes modern Arabic theories of criticism, adding a new perspective that takes into consideration marginalized sub-cultures which, according to him, incorporate religious and ethnic minorities in Arab countries. Alharasi (2003) suggests that CDA approaches could be usefully operationalized in some areas of investigations that include religious discourses and the representation of women in Arabic colloquial proverbs. But Alharasi does not provide any sort of analysis that could support his claims.

Al-Ali (2006) has studied religious affiliations and displays of masculine power in the wedding invitation genre in Jordan. Drawing on two analytic frameworks: genre analysis, and critical discourse analysis, he (ibid., p. 691) concludes that 'religious affiliation and

masculine kinship authority not only construct and shape text component selection but also color the lexical choices and naming practices.'

Another set of studies have investigated the representation of Islam in the media, using a CDA perspective. One notable example is Richardson's (2004) study that focussed on the representation of Islam and Muslims in British broadsheet newspapers. His main goal is to analyse 'the ways in which' broadsheet newspapers 'reproduce anti-Muslim racism' (2004, p. xvi). Additionally, the edited collection of papers in Poole and Richardson (2006, p. 1) have examined 'the role and representation of Muslims in the news media in both British and international contexts' drawing 'on a variety of research methods' and analysing 'diverse national and international media to make more general comments on the manner which Islam and Muslims are represented, and *misrepresented*, in the news media' [italics in original] (see Section 2.6 below for other studies which have investigated the representation of Islam and Muslims or faith communities in Islam using corpus tools).

Moving away from the few CDA studies on religious discourse for the moment, other CDA studies have focused on the representation of social actors within certain discourse communities such as Reisigl and Wodak (2001). Their objective was to investigate the discursive production, enactment and reproduction of racist and antisemitic discourses. They have provided 'a critical synopsis of various scientific explanations that point to economical, social, cultural, social psychological and cognitive roots of racist, antisemitic and ethnicits prejudices' (2001, p. xi) (for more information about the DHA of Wodak and her colleagues in the Vienna school see Section 2.3.2 below).

Two related CDA approaches that were operationalized to investigate the representation of social actors include van Leeuwen's sociosemantic categories (van Leeuwen, 1996, 2008) and the discourse-historical approach (henceforth DHA) developed by Wodak and her colleagues in the Vienna school (Reisigl and Wodak, 2001, 2009; Wodak,

1996, 1999, 2001, 2007, 2009) (for more information on van Leeuwen's socisemantic categories and the DHA see Chapter 3, section 3.2.2). As stated above, CDA involves the linguistic analysis of texts, but it also takes analysis beyond the text to consider issues of political and social context in order to give full interpretations and explanations of the findings obtained. Let us start with the studies that utilized van Leeuwen's sociosemantic categories (1996, 2008) of representing social actors.

2.3.1 The representation of social actors: Sociosemantic categories

In my own analysis I am particularly interested in how social actors are represented within Salafi discourse. This focus on social actors is inspired by my concern that the way that humans are represented through language can have extremely important consequences, inspiring people to behave in certain ways, sometimes violently. In particular I am concerned with negative representations or 'othering' whereby an 'out' group is represented as possessing negative qualities ('they are less than human'), internally consistent qualities ('they are all the same') and as being distinct from an in group ('they are not like us'). I wish to see whether Salafi discourse contains such types of social actor representation and if this is consistent across the two different Salafi leaders whose texts I am examining.

The approach taken by van Leeuwen (1996, 2008) focusses on how language can be used to represent social actors (participants in clauses) who can be represented as subjects or objects or excluded altogether via the use of grammatical systems. Van Leeuwen shows various types of representation via a hierarchically structured diagram of different sociosemantic categories, that can be grouped together (1996: 66). See Figure 2.1 below.





For example, at the first level of the diagram a distinction is made between representations that include (actually refer to) social actors and those that exclude (e.g. background or suppress) them. Inclusion is divided into categories of personalisation and impersonalisation, while impersonalisation is divided into abstraction and objectivation, and so on. Each of those categories has different linguistic realisations. For example, a category like relational identification is realised by referring to a social actor through the use of nouns like *brother*, *co-worker*, etc – the person is represented through their relation to someone else. On the other hand a category like physical identification involves language that refers only to some sort of physical aspect of a person e.g.: *a blonde*, *a hunk*, *a cripple*.

Social actor representation has been incorporated into CDA approaches used by Fairclough (2003) and Reisigl and Wodak (2001, 2008). For Fairclough (2001, pp. 145-6), the available choices in the representation of social actors are: inclusion/exclusion,

pronoun/noun, grammatical role, activated/passivated, personal/impersonal, named/classified and specific/generic. Reisigl and Wodak (2001, 2008) refer to the referential or nomination discursive strategy as one of several discursive strategies that could be used by discourse producers in representing social actors (several of their referential or nomination categories appear to be borrowed from van Leeuwen's (1996) approach).

Many other studies draw on van Leeuwen's (1996, 2008) sociosemantic categories. For example, Caldas-Coulthard (2003) has investigated cross-cultural representation of 'otherness' in media texts. She utilized some sociosemantic categories of van Leeuwen (1996). Salama (2011) has borrowed two oppositional paradigms from van Leeuwen's socialactors theory, namely, specification vs. genericization and nomination vs. categorization. He has used an antonymy-based classification scheme of his analysis of some ideological collocations in the meta-Wahhabi discourses (see pp. 37-8 in this thesis).

The prominent approach that incorporated some aspects of the sociosemantic categories in their approach is the Viennese school that developed the discourse historical approach (DHA). This leads us to the second approach that I utilize in analysing the research data (see Chapter 3 for more information about the methods employed in this thesis).

2.3.2 Discourse Historical Approach (DHA)

DHA is one of the salient approaches that provide a toolkit for researchers to investigate the positive self- and negative other-presentation within discourse. The DHA has been operationalised to analyse antisemitic discourses as in Reisigl and Wodak (2001) and more recently to analyse discourses on climate and global warming (Reisigl & Wodak, 2009). DHA combines a number of other approaches together. For example, it has a particular focus on the consideration of historical context when attempting to interpret and explain discourses found in texts. Richardson and Wodak (2009) have also utilized the DHA in tracing the

histories of discourses of the far right wing groups that support 'jobs for the natives' in the UK and Austria. As well as this, it relies on the identification of topoi and argumentative fallacies (taken from rhetorical analyses) in texts. Additionally, as described above, DHA has drawn on social representation theory, to include reference to referential or nomination strategies. It also considers predication, perspectivation and mitigation/intensification (for more information about the DHA, see Section 3.2.2.7).

Other critical discourse analysts have successfully extended it to examine discourses in different political and cultural contexts. For example, Unger (2009) has used the DHA to investigate the discursive construction of Scots in educational and bureaucratic discourses as well as the discourses of ordinary people.

Also, Brindle (2009) has studied some racist representations in a white supremacist web forum using the DHA. He investigated argumentation employed by the white supremacists utilising certain words such as '*ago*' and '*remember*' to locate argumentation in his data. But he does not refer to such words as argumentative indicators, per se, as helping to indicate arguments, nor does he refer to argumentation schemes in his analysis (See Section 3.2.3 in Chapter 3 for more information about arguments and argumentation indicators and schemes). More recently KhosraviNik (2010) has attempted to systematise CDA analytical categories in the representation of social actors by incorporating 'some of the widely applied CDA analytical categories used in the discourse-historical approach (DHA), including referential, predicational, and argumentative strategies' as well as using 'some elements from the socio-semantic approach of van Leeuwen on the representation of social actors'(ibid. p. 57).

In the Arabic context, Suleiman (2008) has extended the DHA and applied it to Arabic data. He has utilised DHA as a "transdisciplinary" problem-oriented theoretical approach (Suleiman 2008, p. 39) in his analysis of Arab-Palestinian political discourse. He has augmented the DHA with functionalising the religious concept of 'at-Taqiyya' as part of his culture-centred approach that, according to him, is the best access point to further understand the 'East' and the 'self'. The 'at-Taqiyya' or 'التقية' ('*attaqiyyah*; safeguarding the self in eminent dangers) is a religious concept, which could be associated with lying, that is mostly employed in real life situations by the Shiite branch of Islam. However, employing it in a purely Sunni context is not always a successful tool for analysis as it always associated with Shiite form of Islam.

In what follows I introduce some concerns that have been raised against CDA and the efforts of CDA scholars to ameliorate this field of social research utilizing corpus tools. After that, I present some studies that investigate the representation of Islam combining CDA and CL.

2.3.3 CDA: Voicing some concerns

Despite CDA being an academic perspective rather than engaging in polemic, it could still be criticised. CDA researchers tend to eschew scientific objectivity and are instead interested in carrying out socially-committed research, based on real-life problems of power inequality that they feel strongly about. However, drawing on CDA alone could be problematic in the sense that bias on the part of researchers may impact on the way that the analysis is carried out. Critics of CDA, such as Widdowson (1995) and Stubbs (1997), outline a number of potential problems associated with it. Stubbs (1997) introduces 11 questions related to data and theory in CDA. The most relevant issue to the present study is that which relates to the collection and analysis of data.

Stubbs (1997) asserts that the methods of data collection and analysis are inexplicit and restricted to text fragments that could be biased. This could result in CDA researchers 'cherry-picking' texts or parts of texts which demonstrate clearly disempowering discourses in order to demonstrate a social problem, while perhaps ignoring other similar texts or parts of the same texts that offer a contesting perspective. For example, we could claim the British press is biased against a certain social group by selecting 100 articles which show clear bias. But we could have ignored 10,000 other articles which present a more balanced or even positive picture. While negative bias exists, it is not the full story.

Additionally, Widdowson (1998, p. 148) claims that in CDA, 'Your analysis will be the record of whatever partial interpretation suits your own agenda'. Such criticisms may have been a catalyst for some CDA scholars and researchers, such as Fairclough (2000), to use a corpus linguistics approach in order to resolve the problem of collecting and analysing representative data.

2.4 Corpus Linguistics and its contribution to CDA studies

Corpus linguistics involves the analysis of large collections of naturally occurring texts or corpora (see 3.2.1) which is aided by the use of certain software packages such as WordSmith tools (Scott, 1999) which are able to carry out complex calculations on language data. This often involves making statistical comparisons based on word frequency or word order, as well as presenting linguistic data in ways that make it easier for a human researcher to identify linguistic patterns and trends. Important concepts within corpus linguistics are wordlists, keywords, collocations and concordances (described in more detail in the following chapter). These are techniques of analysis which facilitate the work of researchers and allow them to investigate linguistic phenomena in a large amount of data.

Recently corpus linguistics has offered quantitative methods that have contributed to reducing 'researcher bias' (Paul Baker, 2006, p. 12). One of the earliest studies that combined CL and CDA is Hardt-Mautner (1995). Corpus tools are also used and advocated by a number of researchers such as Baker et al. (2008), and Mautner (2007, 2009). Combining

CDA with corpus linguistics helps researchers to handle large amounts of data as well as enabling critical discourse analysts to broaden their empirical base (Mautner, 2009, p. 123). Corpus linguistics software, Mautner adds, (ibid.) offers both quantitative and qualitative perspectives on textual data, computing frequencies and measures of statistical significance, as well as presenting the data extracts in such a way that the researcher can assess individual occurrences of search words, qualitatively examine their collocational environments, describe salient semantic patterns and identify discourse functions. The methodological synergy between CL and CDA has been successful in a number of studies including Baker et al. (2008) who have used such synergy in the examination of discourses about refugees and asylum seekers in the UK press. In what follows I introduce some relevant studies that incorporate corpus linguistics tools in analysing discourses about religion.

Studies that use corpus linguistics tools to investigate discourses about Islam include Baker (2010) who used the corpus-driven approach to investigate news stories about Islam before and after the 9/11 attacks on America. He investigated the representation of Islam and Muslims in an 87 million word corpus of British newspaper articles covering the period between 1999 and 2005. He compared broadsheet and tabloid news using keywords, collocates and concordance analyses. The results (Baker 2010, p. 310) indicate that the 'tabloids tended to focus more on British interests, writing about Muslims in a highly emotional style, in connection with terrorist attacks and religious extremism, focussing on a small number of high-profile Muslim "villains" whereas 'the broadsheet covers more stories about Muslims engaged in wars.'

Salama (2011) has studied ideology by examining collocation in meta-Wahhabi discourse in the period after the 9/11 attacks on America. He utilized corpus linguistics and CDA methodologies in order to propose a linguistic model for explicating the ideological nature of collocation between two clashing discourses. He selected a small number of

keywords for further in-depth investigation of their collocates relying on the concepts of textual synonymy, oppositional paradigms and argumentative fallacies. In addition to his use of the socio-cognitive approach to investigate the ideological coherence and socio-religious schemas that could motivate the ideological use of collocations in his research data, Salama has also utilized a social-semiotic perspective to investigate interdiscursive meanings and the symbolic power invested with the collocating words as religious or political signs.

And more recently, Baker et al. (2013) have expanded Baker's (2010) study on the role that the British national press has played in representing Muslims and Islam particularly in the years following the 9/11 attacks on America utilizing corpus tools. The early analysis chapters of their book focus on keywords which characterise the whole corpus, then individual newspapers, then change over time, while later chapters focus on various aspects made salient in light of certain keywords. This includes the use of collectivisation strategies like *Muslim community* and *Muslim world*, gender differentiating strategies (e.g. Muslim women as wearing the veil, Muslim men as prone to radicalisation), the representation of different levels of belief (e.g. *extreme* vs. *moderate* vs. *devout*) and the demonization of Muslims who were seen to be unjustly deserving of government benefits.

However, as I have stated earlier, Salafi discourses have never been studied, to the best of my knowledge, from a CDA perspective, nor have they come under any sort of investigation utilizing corpus tools. This leads us to the final section of this chapter, namely, the conclusion.

2.5 Conclusion

The aim of this chapter was to introduce a survey of research that are concerned with religious studies as well as some studies on Salafism that are aimed at accounting for this

religious phenomenon in Islam. I have also covered studies that have been aimed at reading religious discourses critically.

Apart from Rizwan (2011) and Hukam Sing and Thuraisingam (2011) none of the studies, that I have outlined above, has ever investigated Arabic or English Salafi or other religious discourses. The studies on religious discourse proper must navigate many pitfalls as I have indicated above. Rizwan (2011) and Hukam Sing and Thuraisingam (2011) have not addressed the issues of semantic foci that distinguish religious discourses, representing 'the Other', or argumentation schemes that are mostly utilized to legitimate authorial stances. They have not also utilized corpus linguistics tools as their data are limited which would raise the problem of their possible bias in selecting data that serve their preconceived agendas. This constitutes the following point of departure: I shall be studying Salafist discourses (religious discourses per se) critically utilizing corpus tools to locate the most salient semantic foci, representations of social actors and argumentation schemes in the research data.

In order to achieve the previously mentioned goal I use an eclectic approach utilizing corpus linguistics tools combined with some aspects of CDA methodology that I aim to analyse an Islamic religious discourse proper (Salafism). The next chapter will address the methodological framework that I intend to use in this thesis which is hoped to bridge the gaps in CDA studies of religious discourses.

CHAPTER 3: Methodological framework

3.1 Introduction

This chapter aims to shed light on the methods and data employed in this thesis in order to analyze the Salafi ideological representation in two corpora designed for this purpose. The analysis in this thesis has two major parts. First, I base my analysis on the most salient ideological representations via keyword analysis in both corpora which would enable me to capture the major differences in semantic foci, representations of social actors and argumentation schemes as well as points of agreements of both Salafi writers. Second, on a macro level, this thesis seeks to provide an explanation of results produced. Thus, this chapter also sheds light on how I shall explain and relate my results to socio-political and religious contexts. In section 3.2 I outline the relevant aspects of the two methodological frameworks that I use; corpus linguistics and critical discourse analysis. As the form of CDA that I employ is based on the Discourse Historical Approach, I also include a sub-section on argumentation theory as this is an area from which DHA has borrowed. After outlining the frameworks employed, I then describe how I collected and transcribed my data in section 3.3. Then in section 3.4 I describe how I shall apply various techniques and procedures to my data in order to answer my research questions. Finally, section 3.5 concludes the chapter.

3.2 Methodological framework

In this study I shall also combine corpus linguistics (CL) and critical discourse analysis (CDA) methodologies. As shown in section 2.4 the synergy between them is encouraging as it involves triangulating quantitative and qualitative methods. I shall first expand on the description of corpus linguistics given in chapter 2, paying greater attention to the techniques within CL that I shall be using in this thesis.

3.2.1 Corpus linguistics

A corpus can be defined as a collection of naturally occurring examples of language, although more recently it has been used to refer to collections of (or parts of) texts that are stored and accessed electronically (Hunston, 2002). Corpora are collected in order to be representative of a particular language variety²⁶, and corpus builders attempt to ensure that they create corpora that are sampled from a range of representative sources, with no single text skewing the data. For that reason, similar-sized samples of texts are often collected, rather than whole texts. Corpora that aim to represent a whole language variety may be hundreds of millions of words in size such as the British National Corpus (BNC) and The Corpus of Contemporary American English (COCA). Corpora which are based on more restricted types of language (such as British weather reports) may, quite legitimately, be much smaller.

Corpus linguistics involves the analysis of corpora which is aided by the use of software packages which are able to carry out complex searches and calculations on language data. This often involves making statistical comparisons based on word frequency or word order, as well as presenting linguistic data in ways that make it easier for human researchers to identify linguistic patterns and trends. Important concepts within corpus linguistics are wordlists, keywords, collocations and concordances. These are techniques of analysis which facilitate the work of researchers and allow them to investigate linguistic phenomena in a large amount of data.

3.2.1.1 Keyness

A keyword is simply 'a word that appears in a text or a corpus statistically significantly than would be expected by chance when compared to a corpus which is larger or of equal size'

^{26.} This is true of a representative corpus. However other corpora such as the two Salafi corpora under investigation in this thesis are not designed to be representative of a language variety. They are rather designed to be representative of the type of the writings of each Salafi writer chosen for critical discourse analysis.

(Paul Baker, Hardie, & McEnery, 2006, pp. 97-98). In order to obtain keywords, wordlists are generated first for two corpora (the second corpus is considered as the comparator or reference corpus), and then frequencies of all of the words in the two wordlists are compared against each other using a statistic such as the chi-squared or log likelihood statistical test in order to determine which words occur significantly more often in the target corpus (Paul Baker, 2006, p. 125). Keywords can be useful in revealing the 'aboutness' of a particular corpus, and can provide an initial focus for analysts which is based on actual salience rather than what an analyst may 'feel' is important. In my study, I use keywords to identify different lexical foci between the two types of Salafi discourse I have collected. This gives an indication of the different topics that are prevalent in each type of discourse, as well as giving me an idea of which social actors are referred to, and the sorts of words which are used as part of argumentation strategies. However, these keywords, once identified, will be examined in more detail via collocation and concordance analyses (see the following sections).

3.2.1.2 Collocation and word clusters

When words tend to co-occur next or near each other in naturally occurring language use, they are called collocates (Firth, 1957). The frequent co-occurrences of two words in a text may indicate a relationship that can be ideological, as Stubbs (1996, p. 172) points out: 'words occur in characteristic collocations, which show the associations and connotations they have, and therefore the assumptions which they embody.' Stubbs (ibid., p. 195) argues that 'if collocations and fixed phrases are repeatedly used as unanalysed units in media discussion and elsewhere, then it is very plausible that people will come to think about things in such terms.' In my own study, I shall examine collocates of the keywords I have identified, in order to get a better idea of the meanings attached to them. It is worth noting here that in cases where a keyword is very frequent, occurring say, thousands of times, it can be helpful to also examine frequent word clusters. A word cluster is simply a fixed sequence of words

(sometimes called an n-gram, where n stands for the number of words in the sequence). While collocates can often identify salient relationships between words, word clusters can help to show their most frequent contexts.

Corpus software can quickly identify lists of collocates or clusters in a text or corpus, although such lists may not tell us anything about the actual use of the collocates or clusters and what they achieve in the research data. This would need to be identified through concordance-based analysis.

3.2.1.3 Concordance

A concordance is a list (usually in a tabular form) of all of the occurrences of a particular word, phrase or collocational pair within its immediate context – usually a few words to the left and the right of the search term (Paul Baker, 2006, p. 71). Concordances can be sorted alphabetically to the left or right of the search term in order to enable similar linguistic structures to be more easily identified. Additionally, concordance lines can be expanded if more context is required. I shall therefore rely on the analysis of concordance lines in order to investigate the context that keywords and their collocates occur in. This will help me to better explain why certain patterns appear in the writings of one author, compared to the other.

Having described some of the techniques from corpus linguistics that I shall be employing in this study, next I shall outline the aspects of CDA that I shall use.

3.2.2 Aspects of CDA applied in my study

The representation of social actors is one of the main themes in this study. I shall utilize two major aspects of the sociosemantic categories proposed by writers within CDA such as Fairclough, van Leeuwen and Wodak, namely personalization and impersonalization as nominational strategies, and argumentation schemes that are used as legitimating strategies.

43

3.2.2.1 Social actor theory

Following Halliday (2004) social actors are the Participants in social processes (Fairclough, 2003) or social practices (van Leeuwen, 2008). According to Fairclough (2003, p. 145) social actors are usually Participants in the clause but they may be located within the Circumstances²⁷, in which he confirms that "not all Participants are social actors as in 'the car hit Mary', 'the car hit a rock' - both 'Mary' and 'a rock' are objects of the verb (i.e. Participants), but only 'Mary' is a social actor". In this study I use some of the sociosemantic categories of van Leeuwen (1996, p. 32-70; 2008, pp. 23-54) (detailed below) to focus on the representation of 'the Other' as opposed to in-group representations. The aim of CDA analysis is to investigate social inequality which could be 'expressed, signalled, constituted, legitimized and so on by language use" (Wodak, 2001b, p. 2). A focus on 'the Other' then, is in keeping with the aims within CDA. And while both of the religious leaders I am examining are Salafis, during my early examinations of keywords I found it notable how each one tended to focus on a different type of 'Other', and further investigation indicated that an analysis of 'the Other' would help to reveal the ideological stances of each leader. Therefore, the boundaries of the two Saudi Salafi strands could be realized more obviously by examining representations of social actors who belong to the out-group rather than the Salafi in-group. In what follows I shall outline aspects of van Leeuwen's social actor representation framework that will be used in my own analysis.

3.2.2.2 Sociosemantic categories

As described in section 2.3.1, Van Leeuwen (1996 pp. 32-70; 2008, pp. 23-54) proposes an approach²⁸ that tackles the representation of social actors in terms of inclusion and exclusion. His main aim is to establish a sociological and critical relevance of his linguistic categories by developing an inventory that could account for the representation of social actors in

^{27.} This follows Halliday's concept of 'Circumstance' (Halliday, 2004, pp. 259-280)

^{28.} This has been slightly changed from his previous approach (van Leeuwen, 1996).

English. He identifies two reasons for drawing up such inventory; first, 'the lack of biuniqueness of language' where 'the agency, for instance, as a sociological concept, [...] is not always realized by linguistic agency, by the grammatical role of 'agent'[...] it can also be realized in many other ways, for instance, by possessive pronouns' (van Leeuwen, p. 23). Van Leeuwen's second reason for his sociosemantic inventory is based on the assumption that meanings are subject to cultural and historical linguistic change.

3.2.2.3 Nominational strategies

Van Leeuwen's approach is composed of twenty one possible ways (organised hierarchically) in which social actors can be represented in discourse. The first two levels of sub-division are inclusion and exclusion. Exclusion involves processes like passivisation and backgrounding – or ways in which social actors become 'unmentioned' in various ways. In the early stages of my data analysis I was open to including any or all of van Leeuwen's categories which appeared in my analysis. However, it became clear that certain categories occurred more often than others, and due to reasons of time and word limits I have instead focussed on those which appeared most frequently and appeared to be most clearly related to the construction of 'the Other', namely the inclusion category which is further subdivided into personalization and impersonalization. These two sub-categories actually cover the majority of the other categories in van Leeuwen's system (see Figure 2.1), and are described in more detail in the following two sub-sections.

3.2.2.4 Personalization

Personalization is the representation of social actors as human beings by means of abstract or concrete nouns which have the semantic feature +human. This category of the sociosemantic inventory can involve representing social actors by *indetermination* which refers to the representation of social actors as unspecified, anonymous individuals or groups.

45

Indetermination could also be realized by aggregation (i.e. collectivization which represent social actors as groups) as in 'many believe' or 'someone says'.

Though nomination involves representation of social actors 'in terms of their unique identities, by being nominated', it can also be 'titulated either in the form of honorification, the addition of standard titles, ranks, etc., or in the form of affiliation, the addition of personal or kinship relation terms'. Various levels of formalisation can thus be used to nominate social actors.

Identification occurs when social actors are referred to 'in terms of what they, more or less permanently, or unavoidably, are'. Van Leeuwen distinguishes three types of identification: a. Classification (which is socially and culturally variable) occurs when social actors are identified according to the different classes of people in a given society (according to wealth, age, religion etc.); b. Relational identification is the representation of social actors in terms of their personal, kinship, or work relations to each other. Finally, physical identification relates people to their physical features.

3.2.2.5 Impersonalization

Impersonalization is the representation of social actors by means of abstract or concrete nouns which lack the semantic feature +human. Van Leeuwen identifies two types of impersonalization: a. Abstraction which involves the representation of social actors in terms of a quality assigned to them; and b. objectivation that involves representation of social actors by means of reference to a place (spatialization), or a thing closely associated either with their person (somatization) or with the action in which they are represented as being engaged (utterance autonomzation, instrumentalization).

In the following subsection I address issues that relate to the circumstances of this thesis – using a categorisation system developed with the English language grammatical system on Arabic data.

46

3.2.2.6 Applicability of the sociosemantic categories to Arabic data

Though Van Leeuwen (2008, p. 23) states that his approach is designed to investigate how the Participants of social practice can be represented in English discourse, his sociosemantic inventory can also be applied to Arabic texts. It is also worth noting that despite the fact that lexicogrammatical and discourse-level linguistic systems, and nominal group may slightly differ in the case of Arabic, i.e. in some specific instances of linguistic realizations, the application of this approach as a whole can be extended to Arabic texts.

One example of the differences between English and Arabic in the linguistic realization of the sociosemantic categories and the categories themselves is nomination of social actors by affiliation. In English, representation of social actors by nomination in terms of affiliation (relational identification) involves the addition of personal or kinship markers to the designated social actor as in, quoting the example of Van Leeuwen, "Auntie Barbara" while linguistic realization of this category in Arabic occurs by adding the word 'بن/این' (*'ibn /bin*; the son of) or 'بن' (*bint*; the daughter of) between the two names (see Section 3.3.2 for translation and transliteration conventions in this thesis). This is in the case of adding the father's name (or even the mother's name in rare, mostly negative, instances as in the following vocative case ' μ_{i} (μ_{i} '*ibn 'assawdā*'; oh you, the son of the black mother), but if the affiliation is familial or tribal, then the addition of '*ili*' (*'al*; the), or the modern usage of '*ili*' (*'al*; the family or the house of) is necessary before the last name, '*'lial' otaibi*; the Otaibi –member of the tribe of 'otaibah), or '*ili*' (*'al' a member of the family of Saud)*.

In the first case, there are two types of changes: first the addition of the Arabic definite article 'ل' (*'al*; the) at the beginning of the tribe's name, then it undergoes slight linguistic change (morphophonemic); the addition of the vowel /i/ and the deletion of /ah/ in the case of 'عتيبة' (*'otaibah*), both morphophonemic changes occur at the final segment of the name of the tribe. But the addition of *'āl* involves no changes at all. Moreover, nomination in

Arabic can also involve names of places where the social actor(s) originally come from or even having lived in that specific place for sometime or a lengthy period; a person who lived or came from the city of Makkah could be nominated as '(almakki) as in: Mohammad ibn Sa'īd almakki; almakki is a reference to the city of 'abcaccondot almakki, where the last two phonemes, /a/ and /h/, were omitted and a morpheme that indicates affiliation which is composed of the long vowel / ī / is attached at the end. Van Leeuwen's approach considers this last point as a type of identification by classification which involves reference to the social actors in terms of age, gender or provenance and so forth.

In Arabic people can also be nominated with reference to their children; ' i_{μ} / i_{μ} , ' i_{μ} / i_{μ} / i_{μ} / i_{μ} , ' i_{μ} / i_{μ} , ' i_{μ} , '

For my approach, I am interested in examining the representation of 'othered' social actors in Salafi discourse, and therefore van Leeuwen's approach is useful to me. However, it is also important to be able to show how such representations are legitimated, based on the arguments that authors use to justify them. Therefore, I need to combine van Leeuwen's approach with the Discourse Historical Approach.

3.2.2.7 Discourse Historical Approach (DHA)

DHA is a well-known approach to critical discourse analysis developed by Wodak and her colleagues in the Vienna school (Reisigl & Wodak, 2001, 2009; Wodak, 1996, 1999, 2001a,

2007). Reisigl and Wodak (2001, 2009) state that DHA is a three-dimensional approach; firstly, identifying the topics of the discourse under consideration, then investigation of discursive strategies follows and finally examination of linguistic means (as types) and the specific context-dependant linguistic realization (as tokens) takes place.

Reisigl and Wodak (2001) mention five important discursive strategies that 'are involved in the positive self- and negative other-presentation'; referential or nomination (of which several strategies are borrowed from van Leeuwen's approach (1996) mentioned above), predication, argumentation, perspectivation and mitigation/intensification strategies upon which the following five questions are constructed:

1- How are persons, objects, phenomena/events, processes and actions named and referred to linguistically?

2- What characteristics, qualities and features are attributed to social actors, objects, phenomena/events and processes?

3- What arguments are employed in the discourse in question?

4- From what perspective are these nominations, attributions and arguments expressed? And,

5- Are the respective utterances articulated overtly, are they even intensified or are they mitigated? (Reisigl & Wodak, 2009, p. 93)

The approach (DHA) revolves, more or less, around the aforementioned questions which resonate in all the authors' publications about DHA (Reisigl & Wodak 2001; Wodak, 1996, 1999, 2001a, 2007).

According to Reisigl and Wodak (2001, p. 44), a discursive strategy is 'a more or less accurate and a more or less intentional plan of (discursive) practices adopted to a particular social, political, psychological, or linguistic aim' i.e. 'a systemic way of using language'.

49

Following the pragma-dialectical approach of van Eemeren and Grootendorst (1992), Reisigl and Wodak introduce argumentation in their own approach. They define the notion of discourse as linked to, inter alia, 'the argumentation about validity claims such as truth and normative validity involving several social actors who have different points of view' (2009, p. 89). They provide rules for rational disputes and constructive arguing. When these rules are violated, fallacies occur. Thus they have provided a (non-exhaustive) list of argumentative fallacies which are based upon Aristotle's rhetoric (Reisigl & Wodak, 2001) (e.g. straw man fallacy and ad arguments; argumentum ad hominum and argumentative fallacies and topoi e.g. in order to legitimate a particular negative representation of a social actor. In the following section I outline the aspects of argumentation theory which occur in DHA and are thus relevant to my own thesis.

3.2.3 Argumentation

It is useful to start this section by defining argumentation. According to Eemeren and Grootendorst (2004, p. 1), argumentation is 'a verbal, social, and 'rational'²⁹ activity aimed at convincing a reasonable critic of the acceptability of a standpoint by putting forward a constellation of propositions justifying or refuting the proposition expressed in the standpoint'. The aim of using argumentation in this analysis is to identify the arguments via which Salafi writers try to legitimate their stances (see Section 3.2.2.3 below). As I have mentioned in the previous section, following Reisigl and Wodak (2001), argumentation can be seen as one of the discursive strategies that text producers may use in their different representations. I also use the concept of argumentation schemes as an analytical tool via which arguments can be discerned (see Section 3.2.3.2 below). In order to assess the

^{29.} Defining arguments as rational could be problematized as not all arguments are rational. In addition, we might query what 'rational' means in every culture; as what might be rational in one culture might not be in another.

acceptability of an argument, we need to identify fallacies and topoi which I shall write about below.

3.2.3.1 Fallacies and topoi

A topos (pl. topoi) is a persuasive strategy or rule which connects an argument to a claim or conclusion. Topoi are thus broad beliefs which help to maintain an argument without actually constituting the argument itself.

A topos can be further described as being part of an argumentation scheme that belongs to the premises and it is the formal warrant that connects the argument with the conclusion (Reisigl & Wodak, 2001; Rubinelli, 2009; Walton, Reed, & Macagno, 2008). It could also be defined as an arrangement of knowledge, as a discursive resource where arguments for satisfying a conclusion can be discerned without the topoi themselves being considered as part of the argument (Brindle, 2009; van Der Valk, 2000). For example, a topos of authority (Reisigl and Wodak 2001, p. 79) 'is based on the conclusion rule: X is right ... because A (=an authority) says that it is right). A topos of responsibility (ibid. p. 78) states 'because a person... is responsible for... specific problems, it or they should act in order to find solutions to these problems.'

On the other hand, a fallacy is a form of argumentation which appears convincing but is logically flawed because it uses poor reasoning (see Reisigl and Wodak (2001, pp. 71-74). An argument can be fallacious even if its conclusion is true. For example, the fallacy of unqualified authority occurs in cases where a person with insufficient expertise on a matter is quoted as being a legitimate authority on a subject (such as when famous sports-persons are used in adverts to endorse non-sports-related products as being better than their rivals, or when an expert physicist is asked to make moral pronouncements, because of their expert status even if such pronouncements are reasonable). Reisigl and Wodak (2001, p. 79) note that it can sometimes be difficult to distinguish between topoi and fallacies. Identifying a topos or a fallacy in argument(s) could be handled easily, yet, a need to satisfactorily evaluate and analyze the whole argument –whether fallacious or not–, taking into account the argument structure, seems more adequate and could do justice to arguments intended to present standpoints in religious discourse. And in case of multiple/complex arguments when a text producer may either introduce a constellation of arguments that may constitute a conclusive defence of a standpoint – coordinative argumentation –, or arguments supporting other arguments – subordinative argumentation– (van Eemeren et.al., 2002), the need to present the argumentation structure schematically is even greater. Moreover, we may encounter arguments with unstated premises or conclusions (enthymemes) which deductive logic may fail to account for (Walton & Reed, 2005). In order to account for such argumentation complexity, argumentation schemes are introduced as tools that can best identify topoi and evaluate and analyze arguments (Perelman & Olbrechts-Tyteca, 1969; Frans van Eemeren, et al., 2002; F. H. van Eemeren & Grootendorst, 2004; Walton, 2005; Walton, et al., 2008; Walton & Reed, 2005).

3.2.3.2 Argumentation schemes

Argumentation schemes are forms of argument that capture stereotypical patterns of human reasoning, especially defeasible ones, like argument from expert opinion, that have proved troublesome to view deductively or inductively (Walton, 2005). They are more or less conventionalized ways of achieving or promoting a transfer between an explicit premise and the standpoint (van Eemeren and Grootendorst, 2004). These argumentation schemes help to 'identify and evaluate common types of argumentation in everyday discourse' (Walton, et al., 2008, p. 11). In this respect, Van Eemeren and Grootendorst (2004) distinguish between three major categories of argumentation schemes: causal argumentation, symptomatic (sign) argumentation and argumentation based on comparisons (argument from analogy). Though

this classification is broad, it can help identify the main argumentation scheme(s) in the research data.

Religious discourse is argumentative in nature. The aim of using argumentation schemes in my analysis is to critically analyze argumentations that permeate the data of this study as well as identify patterns of reasoning in the Salafi discourse. As Walton et al. (2008) note, most argumentation schemes are defeasible even after an argument has been accepted, and they might later be rejected as new evidence is taken into consideration. Argumentation schemes, thus, can help researchers to critically analyze religious texts and identify fallacies and topoi. Arguments can be traced in a corpus by means of certain lexical elements that serve as indicators. Argumentative indicators (phrases or words) introduce or put forward a standpoint. They may, also, refer to argumentative moves which may include 'to my mind', 'the way I see it' etc. (Frans van Eemeren, Houtlosser, & Henkemans, 2007; F. H. van Eemeren & Grootendorst, 2004).

To give a brief example from my data, an argumentation scheme by sign could be used by Salafi writers who are utilizing Qur'anic verses as the words of Allah, which according to all Muslims, are inherently correct and true. Based on a preliminary corpus analysis, I found that some of these Qur'anic verses and other Prophetic traditions (the highest divine and human authorities in Islam) have been quoted by both Salafi writers whose writings are under examination in this thesis. Additionally they include some of the writings of medieval Salafi religious leaders. Such quotes are used for the purpose of legitimating their stances.

3.2.3.3 Legitimation

According to Van Leeuwen (2007, p. 91) there are four categories of legitimation: '1) 'authorization', legitimation by reference to the authority of tradition, custom and law, and of persons in whom institutional authority is vested; 2) 'moral evaluation', legitimation by

reference to discourses of value; 3) rationalization, legitimation by reference to the goals and uses of institutionalized social action, and to the social knowledges that endow them with cognitive validity; and 4) mythopoesis, legitimation conveyed through narratives whose outcomes reward legitimate actions and punish non-legitimate actions.' He (ibid.) also distinguishes between subcategories of authority legitimation: custom (conformity and tradition), authority (personal and impersonal such as laws) and commendation (expert and role model).

Particularly relevant for my thesis, in terms of legitimation, is the way that the Salafi authors have used religious quotes in their writings. This leads to a discussion of recontextualisation in the following subsection.

3.2.3.4 Recontextualization

Fairclough defines recontextualization as 'the appropriation of elements of one social practice within another' (Fairclough, 2003, p. 32). Reisigl and Wodak (2009, p. 90) point out that transferring given elements to a new context involves two processes; first a decontextualization process, then a recontextualization process takes place. They also indicate that the extracted elements are inserted in a new context and can gain new meanings. Linell (1998) gives three types of recontextualization which are intratextual, intertextual and interdiscursive while Bernstein states that 'recontextualizing principles' 'selectively appropriate, relocate, refocus and relate to other discourses' which causes semantic shifts to occur (Bernstein, 1990, p. 184). Recontextualization includes interdiscursivity that, as Reisigl and Wodak (2009, p. 90) say, connotes reference to topics or subtopics of other discourses. They (ibid., p. 90) add that intertextual connections to past and present texts 'are established in different ways: through explicit reference to a topic or main actor; through reference to the same events; by allusions or evocations; by the transfer of main arguments from one text to the next and so on.'

Towards the end of this chapter I shall discuss how I shall bring the corpus and CDA methods mentioned so far together in order to answer my research questions. But before that, this is a good place to discuss the data itself in more detail.

3.3 Data

Using the corpus analysis tool WordSmith 4 (Scott 2004)³⁰, I shall quantitatively analyze the data under investigation from a corpus linguistic perspective in order to identify salient linguistic features which will then be subjected to qualitative analysis using techniques and categories from CDA. First, I shall describe the data.

3.3.1 Collecting the corpus

The data that will be analyzed in this study are books and booklets written by two famous Saudi-Salafi writers in the 1980s and 1990s. I have chosen to focus on AbdulAziz Ibn Baz and Safar Alhawali. These writers are two of the most prominent religious leaders from that period, belonging to two different strands of Salafism (Fandy, 1998, 2001; Meijer, 2009).

I have chosen this period because, as described in Chapter 1, important political and religious changes took place during this time, beginning with the siege of the Grand Mosque in 1979, which could be seen as the beginning of the rise of the Sahwah and the increasing power of Salafism in Saudi Arabia. I choose the end of the century as my 'cut-off' point because the 9/11 attacks resulted in an increased global scrutiny of Salafism, which has generally been interpreted (at least in the West), as responsible for the 9/11 attacks. I therefore want to focus on the period leading up to 9/11 in order to investigate what the nature of Saudi Salafi discourse actually was at that point, whether it was unified or conflicted, and whether it advocated the sorts of violent attacks which resulted in 9/11.

^{30.} WordSmith was chosen for its flexibility and my own familiarity with its use. While newer versions of WordSmith Tools are available, I found that version 4 works best with Arabic corpora.

I base my selection of the data for the present study on the following criteria:

1- All the selected books or booklets are written by two prominent Saudi-Salafi writers representing the two types of Salafi strands under investigation;

2- All the selected books or booklets address socio-religious issues and also make reference to in and out groups in various ways; and

3- Accessibility: the texts selected for this study were either used as sources for other booklets and brochures that were distributed for free in mosques, public health care units, barber saloons and other public places and/or have been made accessible via the internet on popular websites exclusively designed for religious teaching.³¹ All of the texts are written in Arabic, and the analysis has been carried out on the original Arabic texts rather than translations. However, I provide English translations as part of my analysis, for the purposes of clarity.

Using these criteria, I collected all electronically accessible texts for each text producer. I chose five books and booklets for each author that are available on the authors' websites. Table 3.1 shows the text producers, the titles of their books and word counts:

N	Safar Alhawali	Words	AbdulAziz Ibn Baz	Words
1	An introduction to the Western intellect and Modernism; مقدمة في تطور الفكر الغربي والحداثة	7472	Criticizing Arab Nationalism; نتد القومية العربية	12028
2	Unraveling the Ordeal to Enlighten the Scholars of the (Islamic) Nation: كشف الغمة عن علماء الأمة	30729	Importance of Religious Knowledge in Defeating Destructive Ideas ; أهمية العلم في محاربة الأفكار الهدامة	5248
3	A statement about the incidents (the second Gulf Crisis),بيان للأمة عن الأحداث	5918	The True and the Erroneous Creed; العقيدة الصحيحة وما يضادها	4223
4	Secularism: its Origin, Development and affects on the Islamic World; العلمانية نشاتها وتطورها في الحياة الإسلامية	147931	A Collection of fatwas (part one) Monotheism and Other Related Issues; مجموع فتارى (الجزء الأول) التوحيد وما يتعلق به	77413
5	Nationalism; القومية	3984	A Collection of fatwas (part two) Monotheism and Other Related Issues; مجموع فتاوى (الجزء الثاني) التوحيد وما يتعلق به	31629
Total		196034		130541

Table 3.1: The data used for this study

³¹ The books of Safar Alhawali were made accessible on his website: http://www.alhawali.com/, and the books of AbdulAziz ibn Baz are downloadable from his website: http://www.binbaz.org.sa/.
This has resulted in two Arabic corpora. Alhawali's collection of texts is about 196,000 words and Ibn Baz's is 130,500 words in size. The individual texts vary somewhat in size (from between ca. 4000-15000 words). Using equal-sized samples would have reduced the size of the corpus and made patterns harder to spot and generalisations more difficult to make. Therefore, I am using the full texts, but have tried to take into account the differing sizes into my analysis by considering dispersion. For example, if a particular linguistic pattern occurs due to its appearance in one large text only, then this is noted accordingly and accounted for as being due to that specific text rather than being generalized across the whole writings of that author. As I have indicated above, my data is Arabic which necessitates translations of analysed parts into English.

Once the files were gathered, they needed to be prepared for use with the corpus tool. All the books and booklets were initially Microsoft Office Word documents that were changed into Plain Text files (*.txt files) in order to make them readable to WordSmith. Next I needed to make some adjustments to the Arabic texts to take account of conjoined words. I separated the one-letter word ' $_{J}$ ' (*wa*; and) from the words that follow it. There were almost 35,000 (17,867 in Ibn Baz's corpus and 17,215 in Alhawali's corpus) occurrences of the oneletter word ' $_{J}$ ' (*wa*; and) all of which were written with no spacing separating it from the following words and would thus be identified by WordSmith as one word. However, there were some cases when ' $_{J}$ ' (*wa*) is part of a word and comes in the initial position. Thus, each case had to be analysed manually and separated only if required. The following section describes in more detail issues surrounding translation and transliteration of the Arabic data that I was working with.

3.3.2 Translation and transliteration of Arabic data

The analysis in this thesis is concerned with Arabic data. I carry out the analysis on the Arabic data, although for English readers who do not know Arabic well I provide a

translation of the analysed data. Translation involves communicating the meaning of one language by representing it in a second language. All keywords, collocates and word clusters are translated into their English equivalents based on preliminary concordance analyses of the Arabic texts. As is the case with most religious writings, quotations (sometimes lengthy) from sacred texts, and Qur'an and Prophetic traditions, are used for different authorial purposes. These texts have undergone many different interpretations in the past centuries. Most of these quotations are utilised by and sometimes interpreted differently by different authors to suit different beliefs. It is noteworthy here to indicate that there are some translations published by reliable sources such as the Ash'arite version of Our'an by Yusuf Ali (2009) which was first published in 1934. Another translation of the meanings of the Qur'an that reflects the Salafi interpretation is the Saudi sponsored translation by Al-Hilali and Khan (1997) which is also available online.³² As there are multiple possible ways of translating these sayings, I have chosen the Salafi translation of Al-Hilali and Khan's (1997) of the Qur'anic verses found in Salafi writers' texts. For example, I chose Al-Hilali and Khan's (1997) Salafi translation of the following Qur'anic verse {الرَّحْمَانُ عَلَى الْعَرْشِ اسْتَوَى } which, according to the explications of the Salafis, could be translated literally as {The Most Gracious (Allah) rose over (Istawa) the (Mighty) Throne (in a manner that suits His Majesty). However, other translations such as that by Yusuf Ali (2009) bear the Ash'arites' ideology. Thus according to the Ash'arites, as reflected by Yusuf Ali's translation, the previous Our'anic verse is translated as {(Allah) Most Gracious is firmly established on the throne (of authority)} (see Section 8.3.2 for more information about the Salafi-Ash'arites differences).

Additionally, I have tried to keep the original Arabic terms that indicate some Salafi concepts which do not have any cultural equivalents in the target language (English). I have thus transliterated and translated them into English. Transliteration involves the conversion of

^{32.} This Saudi sponsored translation is available on <u>http://www.qurancomplex.org/default.asp?l=eng</u> Among the many websites that provide the translations of Qur'an is the following: http://www.searchtruth.com/search.php

a written text from one script into another. An example of how I combine transliteration and translation is the term 'الحاكمية' ('alḥakimiyyaha; sovereignty of Allah). Here I first give the Arabic term, then in italics, the transliterated term employing letters used in the English writing system in order to represent the sound that the Arabic word would make if spoken. Finally, after the semicolon I give the English translation or meaning of the word. For the transliteration, I follow the DIN 31635 standard for the transliteration of the Arabic alphabet (see appendix C).³³

It is noteworthy that Alhawali utilized translated quotes from different English sources. In this case I refer to the original English text then compare the Arabic version to that of the original to examine any noticeable differences that could be utilized by the (anonymous) translator(s) (Alhawali is notably unable to read or speak English) to achieve certain ideological ends. It is also noteworthy here to give some notes about my use of the different forms of brackets that appear in the translation of analysed texts.

I write three dots enclosed in square brackets [...] to indicate that the concordance line is taken from a lengthy and verbose paragraph(s) as I omit some repeated sentences and phrases in the original Arabic version. I also use square brackets to write my own comments on the translation from English into Arabic when the Salafi author directly quotes from English sources as we shall see in the analysis of Alhawali's writings in Chapters 4, 5 and 7. Additionally I use square brackets to provide some words from the co-text that could help relate the concordance line or excerpts to the texts they belong to e.g. 'we [the missionaries]'. These words or phrases mostly appear in square brackets at the start of some concordance lines in both versions, Arabic and English.

^{33.} The DIN 31635 standard for the transliteration of the Arabic alphabet is available on the following link: <u>http://transliteration.eki.ee</u>.

I use round brackets to either give some hints or definitions of difficult religious or Arabic concepts to help readers to understand concordance lines and other excerpts such as the word 'Zakat' (regular charity). I also use round brackets to provide words and phrases that could explain the meanings of keywords, collocates or word clusters that appear in tables, figures or concordances. An example from the research data is the word 'صلى' (salla) which literally means 'prayed', however it occurs in my data as part of an optative phrase which means 'may Allah send His blessings on' (detailed below). As for the brackets that exist in the original texts I prefer to keep them as they appear in the translation. Curly brackets {...} always indicate Qur'anic verses and a small asterisk indicates the end of a Qur'anic verse and the beginning of another one. The Arabic writing system is another issue that needs special attention. Unlike the English writing system which shows each word separately, Arabic may include two or more words together. This is because the Arabic writing system reflects the morphological structure of Arabic which is substantially more complex than that of English, and as a result some pronouns as well as the definite article are always attached to verbs and nouns (some of which are keywords in this thesis). I call such phrases 'one-word phrases', for example, the definite article '-1' ('al; the), is always attached to its following noun as in the Arabic word 'الرسول' ('alrasūl: the messenger).

There are some other examples that show the morphological complexity of the structure of Arabic words such as 'illine illine illine illine illine illine illine illine illine illine into a whole phrase containing 7 words if translated into English as in 'and then we made you drink it'. WordSmith deals with such one-word Arabic phrases as single lexical units, although I have translated them into English as multi word units. Some of these words may occur as keywords or collocates. When translating keywords, and collocates or any other salient words, I provide some other grammatical information that could give their meanings clearly in English. Such information is put between brackets, so for example 'Line' or 'Line' '.

would be translated as (O... –vocative) or 'محمدا' (mohamadan; Mohammad –acc.) and so forth.

It is also noteworthy here that some of the keywords and collocates in the research data are homographic in nature such as 'سلم' (*silm*; peace, *salima*; be saved, *sullam*; ladder, or *sallam*; to say peace be with/(may Allah) send His peace (on him)) and 'صلى' (*sala*; exposed to the blaze of/*salla*; prayed/(may Allah) send His blessings (on him)). In Arabic writing system, the short vowels /a/, /o/ and /i/ are indicated by diacritics. Also, gemination (or elongation of a consonant sound) in the Arabic writing system is indicated by the diacritic which is placed on the letter representing the geminated consonant. However, these writing system rules were not always followed by the authors who created my data.

In analysing polysemous words, I point out the different meanings of such words therein as in Chapter 6 section 6.2. The homographs as well as polysemous words in Arabic show one of the shortcomings of WordSmith (and probably other software packages) as they are not identified by such software packages as distinctive words. However, readers of Arabic should be able to successfully recognize the geminated consonants or the other diacritics that represent the short vowels depending on the linguistic context. Another difficulty of translating search words or concordances is that some morphological changes in some Arabic words are not obvious as these changes are indicated by diacritics which are mostly not used in the two corpora. However speakers of Arabic can successfully identify them as in the social actor indicating keyword 'apple' ('*ibadoh/ibadah/ibadih*; His servants/worshippers – nom., acc. and gen. respectively)³⁴ in Ibn Baz. The three forms of this keyword can only be identified from the co-text (see criterion 5 on p. 72 below).

^{34.} The case markers in Arabic are normally indicated by short vowel suffixes; -o for nominative, -i for genitive and -a for accusative, but there are substantial exceptions to this rule (Ryding, 2005, p. 54) -for more information about case inflection in Arabic morphology, see chapter 7 in Ryding (2005).

Having identified the nature of my data, I now turn to describe the procedures of analysis in this thesis.

3.4 Methods

This section first covers the corpus techniques that were used in order to identify keywords. I then go on to show how they will be analysed using other corpus techniques (namely collocation and concordance analyses), and how I shall identify features pertinent to CDA and argumentation theory. I end the chapter by discussing how I shall carry out the analysis of the religious and socio-political context in order to explain my findings.

3.4.1 Computational identification: Analysing keywords

In the following section I outline the method that I shall use to computationally identify the semantic foci that reflect the topics, social actor representation and the argumentation markers that indicate argumentation schemes in the two Saudi-Salafi corpora. I take keywords as the starting point of my analysis in this thesis. For the purpose of comparing the writings of the Salafi authors in these two corpora, using Mike Scott's WordSmith suite of tools (2004), wordlists of both corpora were generated and their relative frequencies were compared against each other and subjected to log-likelihood tests using Wordsmith's default cut-off point for statistical significance, to generate keyword lists for each corpus. As I have indicated above, I use frequency information as a way of identifying potential argumentation indicators. I also derive keywords in order to compare semantic foci and social actor representations between the two sets of Salafi texts. A detailed analysis of the keywords via collocates and concordance lines will help me to identify self and other representations in my data, as well as pinpoint argumentation schemes and indicators.

A preliminary investigation of collocates and concordance lines of these keywords enabled me to group them according to three broad types:

- 1. Keywords which reveal the semantic foci of the texts
- 2. Keywords which refer to social actors
- 3. Keywords which are argumentation indicators

Some keywords can have multiple functions so may be grouped under more than one category. Carrying out a detailed analysis of all the keywords found is beyond the scope of this thesis. Therefore, I utilized some criteria to reduce these keywords to a manageable number (the criteria are mentioned in the beginning of the relevant chapters). In analysing the three types of keywords I use collocational and concordance-based analyses as well as word clusters.

3.4.2 Collocational analysis

After generating keywords and selecting a smaller number for analysis, I first attempt to consider their collocates in order to get a better idea of the meanings attached to them. In order to limit the number of collocates obtained for each keyword, and to focus on those which are most salient, I use the following criteria in identifying collocates : a) collocates of the keywords are generated (utilizing the default settings of Wordsmith) within the span of ± 5 and MI score of 3 or higher which can be 'taken as evidence that' the 'two items are collocates' (Hunston, 2002, p. 71); b) all collocates should occur in at least three texts or more in each corpus to be included in the analysis; and c) if keywords have other wordforms, their collocates are grouped under each word-form and only one collocate that has the highest MI score of each word-form is selected for the analysis.

After selecting the collocates based on the criteria above, I then try to investigate the sort of association (positive or negative) held between the keywords and their collocates. However, in some cases there may be hundreds or even thousands of occurrences of keywords such as the keyword 'الله' (*'allah*; Allah) (4253 times in Ibn Baz and 994 times in Alhawali) (see Chapter 4) and a collocate such as 'سلم' (*sallam*; may (Allah) send His peace (upon him)) which occurs 872 in Ibn Baz and 97 times in Alhawali's corpus. This huge number of occurrences of keywords and collocates turns concordance-based analysis into a daunting experience. Thus, before concordance-based analysis of such search words I examine how keywords occur in their most frequent word clusters which will enable me to capture the most prominent use of the search words in the respective corpus. The next step is concordance based analysis.

3.4.3 Concordance-based analysis

Although I use collocates and word clusters, these techniques do not really give the contexts that words occur in, rather, they focus on the most salient and/or frequent patterns. Therefore I use concordance based analysis as the first qualitative step in analysing keywords and their related collocates and word clusters. Due to the fact that the concordance lines need to be translated into English, and often each one is several lines long (in order to give enough context for the meaning to be understood), it is not possible to go through each concordance line in detail. So I first attempt to classify similar concordance lines together, based upon the relationship between a word and its collocate. Then I choose a smaller number of concordance lines that are intended to be representative of either the most typical relationship between a collocational pair (or cluster), or, in cases where there is a wider range of relationships, I choose a variety of concordance lines in order to demonstrate this.

3.4.4 Semantic foci

As described in section 3.4.1, the keywords are categorised into three different groups: semantic foci, social actors and argument indicators. The semantic foci keywords help me to identify the 'discourse type' (Partington, 2010, p. 88) of each corpus. Here I use discourse type to refer to the ways in which texts can be labelled according to their topic or theme. For example, Chilton (2004) refers to political discourse, Williams and Chrisman (1993) talk of colonial discourse, while Hajer (1997) mentions environmental discourse. I examine a small

set of these semantic foci keywords from each corpus, using the corpus techniques described above. Relative frequency is only one (important) aspect of understanding the differences between the two authors though. I thus examine concordances, collocates etc. of the chosen keyword as it is used by both authors, including the author who under-uses it (although in such cases I may need to directly rely on concordance analyses as there may not be many collocates due to the word's low frequency in that corpus). In some cases, concordance lines need to be further expanded in order to fully understand the meaning behind the appearance of particular keywords in certain contexts.

The analysis of semantic foci keywords could lead to the identification of topics and the different argumentation schemes with which social actors are being associated for legitimation purposes in the corpus data, and this is noted where it occurs.

3.4.5 Social actor representation

Second, the keywords which refer to social actors will be examined in order to identify ingroup and out-group representations. The analysis of social actor representation will focus on identifying particular nominational strategies. I especially focus on two major categories; personalization and impersonalization. The first can be linguistically identified through the linguistic choices made by text producers such as 'personal or possessive pronouns, proper names, or nouns and adjectives. It can also be realized in nominating social actors using titles of honorification.

Classification, which is socially and culturally variable, is another form of nomination in my data. Social actors can be identified according to their religions (i.e. religionym (Reisigl & Wodak, 2001, p. 116) such as Muslims, Christians or Jews), their religious factions (Sunni, Shiite, Sufis etc.) or their intellectual affiliation to (mostly modern literary) theories and movements – 'حداثي' (*hadati*; modernist); '*vata*' (*ba'ti*; an advocate of Baath socialist party/of or pertaining to Baath party). The second type of nominational strategies is impersonalization. It involves reference to social actors by abstraction and objectivation. Abstraction is the representation of social actors in terms of a quality assigned to them. Objectivation is the representation of social actors by means of reference to a place (spatialization) or with the action in which they are represented as being engaged (for example, utterance autonomization as in 'the report says'). Nomination by impersonalization could be used to 'background the identity/role of social actors'. It can also add 'impersonal authority to an action or quality of social actor and positive/negative connotations to an action or utterance of social actor' (van Leeuwen, 2008, p. 47).

The previously delineated discursive strategies are forms of positive selfrepresentation and negative other-representation that contribute to the religious discrimination between individuals within the same faith communities and instigating the enmity between adherents of other religions/religious factions preventing dialogues that could lead discussants to mutual understanding of the diverse positions of each other. It is the intention of this thesis to focus on the different Salafi representations of the 'Other'.

3.4.6 Argumentation schemes

Thirdly, the keywords which refer to argument indicators will be examined in order to identify particular arguments, argumentation schemes, topoi and fallacies. After locating arguments utilizing the argumentation indicators, identification of argumentation schemes then takes place. I shall first show the structure of the argument under consideration (premises, warrants, conclusions or even defeasible arguments such as those with unstated premises or conclusions – enthymemes), and then argumentation schemes can be identified. Evaluation of argumentation schemes via critical questions will be the next step which may result in the identification of topoi or fallacies (see Chapter 3 section 3.2.3). In religious discourse, arguments might be used ideologically to introduce and legitimize standpoints vis-

à-vis representations of social actors, or to refute the standpoints of others. However explanation of such representations can be done via analyzing religious and socio-political contexts, which is discussed in the following section.

3.4.7 Analyzing religious and socio-political contexts

Having identified how the various argumentative schemes and social actor representations in the texts contribute towards Salafi discourses, an important aspect of CDA is to both explain and evaluate these discourses, by relating the discourses to the wider context. The historical, religious (and sometimes social and political) context that modern Salafi groups draw on is vast. Using the triangulatory approach of the DHA (Reisigl & Wodak, 2009) the context of situation is elaborated on by considering interlinks between texts and discourses the two Salafi text producers draw on when providing arguments utilized in negative/positive representations of the 'Other' – as well as utilizing previous canonical, religious, texts and discourses about topics i.e. taking into account interdiscursivity and intertextuality. Biographies of the two text producers and their institutions are also considered as context-relevant information. I also draw on my knowledge of religious sacred texts (Qur'an and Prophetic traditions) and other religious writings that pertain to the findings in Chapters 4-7. Finally, I consider historical factors as part of context, based partly around elaboration of some of the historical context of Saudi Arabia that was provided in Chapter 1.

3.5 Conclusion

So far I have delineated the quantitative and qualitative methodologies proposed for the analysis in this thesis in terms of methods and data. I have shown how I shall begin the analysis at the corpus level in order to identify salient words that are most distinctive of the differences between the two authors under examination. The remaining analyses are involved in explaining why such words appear as keywords, by first considering co-text via collocates,

67

clusters and concordances, and then to show how such words relate to social actor representation and argumentation, using more qualitative measures that require closer reading of texts. A final level of explanation involves relating the findings in the texts to contexts outside the corpus – the identities of the authors, their religious and political beliefs and Saudi Arabia's history and politics.

This section ends Part 1 of the thesis. Part 2 consists of the analysis itself, starting with Chapter 4.

Part Two

CHAPTER 4: Examining Ibn Baz's semantic foci keywords across both corpora

4.1 Introduction

This chapter is the first of five quantitative and qualitative analysis chapters of the research data. Chapters 4 and 5 are devoted to deriving the keywords in the two corpora of the writings of Sheikh AbdulAziz Ibn Baz and Sheikh Safar Alhawali, as well as analysing keywords which relate to topics or semantic foci. Chapters 6 and 7 focus on social actor representation and argumentation respectively. Chapter 8 is devoted to explaining the findings in chapters 4, 5, 6 and 7 in relation to Saudi social and politico-religious contexts. This chapter and the next one respond to research question 1 (a): What do concordance and collocational analyses of keywords and frequent words in the two corpora of Salafi discourse reveal about themes/macro topics? Answering this question contributes to answering the overarching question: To what extent and how is Salafi discourse in Saudi Arabia in the 1980s -1990s homogenous?

As described in Chapter 3, each of the Salafi corpora is comprised of five texts consisting of the writings of the text producers. For the purpose of comparing the two sets of data, using Mike Scott's WordSmith suite of tools (2004), wordlists of both corpora were generated and their relative frequencies were compared against each other and subjected to log-likelihood tests using Wordsmith's default cut-off point for statistical significance, to generate keyword lists for each corpus. The comparison resulted in 224 (Table A1 in Appendix A) keywords for Ibn Baz's corpus and 199 (B1 in Appendix B) keywords for Alhawali's corpus. In order to focus on keywords that were more widely representative of authorial choice rather than being due to a topic briefly but repetitively mentioned in a single specific text, I only focus on keywords that occurred in at least three texts from one author.

69

However, if any of the selected keywords in Tables A1 and B1 (Appendices A and B respectively) occurs more prominently in any of the texts, a notice will be provided therein.

Tables 4.1 and 4.2 below show only the semantic foci keywords (ordered by keyness score) in both Salafi corpora. Both Tables are an abridged version of two more detailed tables (appendices A and B: Tables A2 and B2 respectively) that show the keywords along with their translations in English, number of texts that contain the keyword and other important statistical details such as frequencies, percentages, keyness, and p values. The two sets of keywords are selected according to the procedure described in Chapter 3 section 3.4.1.

N	keyword Eng. trans.	F	F. Alhawali	Ν	Keyword Eng. trans.	F.	F. Alhawali
1	لم Allah	4253	994	28	(Allah has) ordained نرع	47	8
2	the Glorified سبحاته	809	8	29	the Exalted in power, العزيز	46	8
3	the Majestic جل	424	18	30	truly/ with the truth بالحق	37	4
4	the Exalted in Mighty عز	363	11	31	beseeching for help الاستغاثة	28	1
5	the blessings' the prayer الصبات	289	18	32	the Hell-fire التار	58	17
6	the Exalted تعالى	477	151	33	the Sharia Law الشرع	34	4
7	لله to Allah	267	74	34	inten testimony	22	0
8	the call (to) الدعوة	216	45	35	warning التحذير	31	3
9	the Peace the peace Islamic salutatio السلام	n 255	78	36	العلم knowledge (about Allah)	225	183
10	associationism, polytheism الشرك	131	15	37	obeying Him طاعته	25	1
11	in with by Allah من الم	173	41	38	arbitration (according to Islamic Law) التحاكم	30	3
12	the True/ right الحق	242	94	39	ولى Wali (Lord, God, Protector, Helper)	36	6
13	≯≖ the Most High	76	0	40	the most Generous کریم	20	0
14	the guidance/ the right path المهدى	67	3	41	security السلامة	24	1
15	the falsehood البنطل	71	5	42	blessed) end) العائبة	29	3
16	followed them exactly (in faith) بإحسان	51	0	43	العليب His/his path/way	23	1
17	the great/ the Great العظيم	84	15	44	جواد Magnanimous	19	0
18	His his religion دينه	73	11	45	رحنه (He) Alone	91	50
19	His attributes صفائه	42	0	46	religion نين	126	85
20	the religiously impermissible innovative practice and beliefs	63	8	47	evil عر	38	9
21	the worship العبادة	120	51	48	call (to Allah) دعوة	55	21
22	۲ Islam الإسلام	337	271	49	via the call (to Allah) بالدعوة	24	2
23	the (Day of) Resurrection القيامة	45	4	50	the Hereafter الأخرة	63	28
24	رب Lord, may	61	13	51	Al-Munkar (polytheism and disbelief and all that is wrong and Islam has forbidden)	33	7
25	his Lord ربه	40	3	52	idols الأصنام	23	2
26	the attributes (of Allah) المستقات	46	6	53	guidance هدی	34	8
27	straight (way) المستقيم	34	2	54	obedience (submission to Allah) طاعة	42	14

Table 4.1 Semantic foci keywords in Ibn Baz

Table 4.2 Semantic foci keywords in Alhawali

N		Eng. trans.	F.	F.Baz	N	Key word	Eng. trans.	F.	F.Baz
1		Europe	263	4	24		the military (def. adj. fem.)	47	0
2	الخليج	the Gulf	161	1	25		contemporary	47	0
3	-	the West	181	6	26		the modern	54	1
4	النظرية	the theory	120	0	27	الماضىي	the past	44	0
5	1	the revolution	115	0	28	الدينى	the religious (def. adj. masc.)	57	2
6		the intellect/ the thinking	120	3	29		the Crusades	51	1
7		the religious (def. adj. fem./pl)	104	1	30	الدين	the religion	554	222
8	الغريبة	the western	110	2	31	العالمية	the () World War/the international	55	2
9		the European	92	0	32	تاريخ	history (indef.)	59	3
10		the war	112	9	33	العصبور	ages	48	1
11	المتراع	the conflict	65	0	34	المعاصرة	the modern	40	0
12	عصبر		83	4	35	السياسي	the political (masc. adj./the politician	39	0
13		politics	82	4	36		region	49	2
14		mind	116	13	37		military (indef. adj. fem.)	36	0
15		European (sing. adj. masc.)	55	0	38		the military (def. adj.masc.)	35	0
16		the political (adj. pl.)	67	2	39		religious (indef. adj. fem./pl.)	33	0
17		the western (pl. adj. masc.)	67	2	40	النصرانية	Christianity	51	4
18	-	the history (def.)	107	12	41	الوطنية	patriotism	30	0
19		the middle (ages)	59	1	42	المادية	materialistic/materialism	42	2
20		movement	58	1	43	التاريخي	the historical	29	0
21	الحركة	the movement	57	1	44	الفكري	intellectual	36	1
22		the social (contract)	56	1	45	الحضارة	civilization	45	3
23	إسرائيل	Israel	83	7					

Taking into consideration that the corpora are drawn from the writings of two prominent Saudi Salafi authors written in the 1980s and 1990s, the semantic foci keywords tabulated in Tables 4.1 and 4.2 above may give insights into the interests of the two Salafi writers. The first striking result is that Ibn Baz's keywords appear to be more religiously focussed in nature, referring to Allah, Islam and related concepts. On the other hand, Alhawali's keywords appear to be more politically-based, focussing on countries, concepts like politics and the military. Second, many of the semantic foci keywords in Ibn Baz are also mentioned, though on a smaller scale, in Alhawali whereas Alhawali's semantic foci keywords are rarely mentioned in Ibn Baz. For example, of the 54 Ibn Baz keywords, only six of them (11%) are never mentioned by Alhawali, whereas of the 45 Alhawali keywords, 14 of them (31%) are never mentioned by Ibn Baz. This result suggests that while both authors refer to religion to various degrees in their writing, Alhawali appears to be much more politically oriented than Ibn Baz. I shall address the semantic foci in both corpora as indicated by the keywords tabulated in Tables 4.1 and 4.2. As stated above, I shall focus my analysis on a small representative number of keywords that reveal the most salient ideological differences between the two corpora.

In order for a semantic foci keyword to be included in the analysis, it should satisfy the following quantitative and qualitative criteria:

- 1- Based on a preliminary concordance analysis, all the semantic foci keywords are classified into categories according to their meanings. The top three keywords (ordered by keyness score in the main corpus in order to compare the actual use of the selected keywords in the two corpora) in each category are selected for the analysis provided they satisfy criteria 2-5 below.
- 2- A keyword must occur at least five times in both corpora before it can be considered for detailed analysis, so that I am able to compare its usage across both corpora.
- 3- If there were more than three keywords in a single category, then the keywords that occur most frequently in the reference corpus are selected for the analysis in order to allow for a fuller qualitative comparison between the two corpora (8 keywords were selected for analysing theological semantic foci).
- 4- Keywords that belong to the same lemma or word-form are tabulated and grouped under a key term (written in small capitals) and their frequencies are added together. If they occur in the reference corpus and have other word-forms, they are, also, referred to and included in the analysis.
- 5- Arabic words that are homographic or polysemous are not identified by Wordsmith as such. Thus, they needed to be first discerned via concordance lines. If more than 50% of the concordance lines do not indicate a single meaning (e.g. a religious or political meaning), they will not be included in the analysis as their keyness is questionable.

72

Having stated the criteria for selecting keywords, I shall analyze Ibn Baz's theological semantic foci in Chapter 4 followed by Chapter 5 which considers Alhawali's politically oriented semantic foci.

4.2 Theological semantic foci in the Salafi Corpora

Table 4.1 above shows a set of mainly theological³⁵ keywords³⁶ in Ibn Baz's corpus. Based on a preliminary concordance-based analysis, I have carried out a classification of these keywords as follows: Firstly, more than one-third of Ibn Baz's keywords indicate divinity in Islam. These keywords refer to Allah who is (seen as) the ultimate authority and the core concept of divinity in Islamic teachings and theology. The second set of theological keywords indicates monotheism and what Ibn Baz deems to be impermissible religious practices. The third type of keywords in Ibn Baz is related to Islam, as the true faith, and Islamic religious practices. The final set of the theological keywords refers to the 'final return'. Table 4.3 shows this classification of the theological semantic foci keywords in Ibn Baz (for simplicity I have only provided the English translations of words, but see Table 4.1 for the original Arabic words).

Category Keywords Allah, the Glorified, the Majestic, the Exalted in Mighty, the Exalted, to Allah, the Peace, in/with Allah, the Most High, the Great, His attributes, Lord, his Lord, the attributes (of Allah), the Exalted in power, Wali (Lord, God, Protector, Helper), Magnanimous, (He) Allah Alone, His attributes, truly, with the Truth, (Allah has) ordained, the True/ right, the call to (Allah), call (to Allah), obedience (submission to Allah), Allah's knowledge, (Allah has) guided, (Allah's) guidance, knowledge (about Allah), Monotheism and associationism, polytheism, the religiously impermissible innovative practice and beliefs, impermissible beseeching for help, arbitration (according to Islamic Law), idols, the falsehood, the religious practices worship, His attributes, the attributes (of Allah), followed them exactly (in faith), and beliefs Islam, God's decree (Sharia), the guidance, the right path, arbitration (according to Islam Islamic Law), straight (way), His/his path/way, obeving Him, testimony, the (Day of) Resurrection, the Hell-fire, warning, (blessed) end, the Hereafter The final return

Table 4.3 Categories of semantic foci keywords in Ibn Baz

^{29.} Here I define theological as opposed to political keywords as those terms that explicitly refer to religious concepts and bear no apparent political or otherwise meanings. The classificatory schemes of the lexical keywords in Ibn Baz are based on a preliminary concordance.

^{30.} The English translations of all the keywords in this chapter and the following chapters are provided based on a preliminary concordance analysis.

Having stated the most prominent theological semantic foci in Ibn Baz, I now give a detailed analysis of a selection of the theological semantic foci in Ibn Baz as compared to Alhawali's.

Application of the criteria in Section 4.1 above resulted in the selection of 8 theological semantic foci keywords. Table 4.4 below shows the selected theological semantic foci keywords in Ibn Baz and their frequencies in both corpora.

	N	Key word	Eng. trans.	Freq.	F. Alhawali
	1	بالأم/لله/الله	Allah/ to Allah/ in, with, by Allah	4693	1109
Divinity	2	مبحانه	the Glorified	809	8
	3	ئعلى	the Exalted	477	151
Monotheism and impermissible	4	المترك	(the) polytheism/ associationism	131	15
religious practices and beliefs	5	البدع	the religiously impermissible innovative practice	63	8
Islam	6	الإسلام	Islam	337	271
The final return	7	الأخرة	the Hereafter	63	28
	8	النار	the Hell-fire	58	17

Table 4.4 Selected theological semantic foci keywords in Ibn Baz

First, let us first consider the core concept of divinity in Islam: Allah.

4.2.1 Divinity

The theme of divinity in Islam revolves around the concept of Allah. Table 4.5 below contains three forms of the key term 'ALLAH': 'لله' (*'allah*; Allah), 'له' (*lillah*; to Allah) and 'لله' (*billah*; in/with/by Allah) which occurred 4253, 267 and 173 times respectively (4693 times collectively). ³⁷ Alhawali also uses these forms of the key term 'ALLAH': 'لله' (*'allah*; Allah) 994 times, 'له' (*lillah*; to Allah) 74 times and 'بالله' (*billah*; in/with/by Allah) 41 times (amounting to 23.37%, 27.71% and 23.69% of the occurrences in Ibn Baz). The larger number of occurrences of the key term 'ALLAH' in Ibn Baz's writing reveals a greater theological emphasis by Ibn Baz as compared to Alhawali's (see Section 3.3.2 in Chapter 3 for more information on the transliteration scheme for Arabic words used in this thesis).

^{31.} I write the key term 'ALLAH' in small capitals in order to denote all various forms of the keyword 'Allah': الله' ('allah; Allah), 'بالله' (billah; in/with/by Allah) and 'ش' (lillah; to Allah).

Table 4.5 Keywords that refer to 'الله', 'Allah' in Ibn Baz

	Ν	Key word	Eng. trans.	Freq.	F. Alhawali
Div	1	بالله/لله/الله	Allah/ to Allah/ in, with, by Allah	4693	1109
Divinity	2	سبحانه	the Glorified	809	8
Y	3	ئعلى	the Exalted	477	151

Table 4.3 also shows two other theological keywords that are closely linked to the key term 'ALLAH'. These keywords are 'سبحانه' (*subḥanahu*; the Glorified)³⁸ (809 occurrences), '*izella*; (*ta'āla*; the Exalted) (which collocate with each other – MI score 5.043). The second keyword '*iu'* '*subḥanahu*; the Glorified) also collocates with the keyword '*iu'* (*lillah*; to Allah) (MI score 3.893). In terms of critical discourse analysis, these other two keywords indicate positive predicational strategies, although it is interesting that the terms themselves are also frequently used alone to refer to Allah, their collocation with Allah being so strong that they actually become synonymous for Allah. As further indication of the interchangeability of these three keywords, Table 4.6 shows how they also appear in clusters, reinforcing each other.

N	Clusters	Eng. trans.	Freq.	Texts	<u> </u>
1	الله بيحانه	Allah, the Glorified	232	5	2-word Clusters
2	الله تعالى	Allah, the Exalted	67	4	ord(
3	بالله سبحائله	in with Allah, the Glorified	13	4	ไนร
4	شم بيبحائله	to Allah, the Glorified	13	4	ters
5	سيحاته و تعالى	the Glorified and the Exalted	114	5	
6	الله سبحاته و	Allah, the Glorified and	76	5	ω-γ-
7	الله سيحاته في	Allah, the Glorified in	14	3	ord(
8	الله بيدانية قد	verily, Allah, the Glorified	13	4	3-word Clusters
9	لله سيحاته و	to Allah, the Glorified and	6	3	ers
10	بالله سيحانه و	in Allah, the Glorified and	5	3	
11	الله سبحاته و تعالى	Allah, the Glorified and the Exalted	39	4	4-word Clusters
12	ىبىماتە و ئىمالى و	the Glorified and the Exalted and	21	3	
13	قال سيحانه و تعالى	the Glorified and the Exalted said	17	3	lust
14	کما گال اللہ سیجاتہ	as Allah, the Glorified, said	14	5	ers

Table 4.6 Word clusters that contain the different forms of the key term 'ALLAH' and the keywords 'سبحانه' (*subhanahu*; the Glorified) and 'نتعالى' (*taʿāla*; the Exalted) in Ibn Baz

^{32.} I capitalize all names and attributes of Allah, following the traditional way of referring to Him in Islamic discourse.

As these three keywords all refer to Allah, in order to avoid redundancy, I shall only analyze the key term 'ALLAH' and its variant forms. Let us now move to the analysis of the key term 'ALLAH'.

4.2.1.1 The construction of 'Allah' in both corpora

It is beyond the scope of this thesis to do justice to all the occurrences of the key term 'ALLAH' due to time and space restrictions. Therefore I shall first consider the collocates of the term 'ALLAH', then word clusters and finally, I shall carry out a concordance-based analysis of some representative examples of the term.

To reiterate from Chapter 3 (section 3.4.2), I use the following criteria in identifying collocates of keywords in this analysis: a) collocates of the keywords are generated (utilizing the default settings of Wordsmith) within the span of ± 5 and MI score of 3 or higher which can be 'taken as evidence that' the 'two items are collocates' (Hunston, 2002, p. 71); b) all collocates should occur in at least three texts or more in each corpus to be included in the analysis; and c) if keywords have other word-forms, their collocates are grouped under each word-form and only one collocate that has the highest MI score of each word-form is selected for the analysis.

Wordsmith initially identified 130 collocates of the key term 'ALLAH' in Ibn Baz and 20 collocates in Alhawali. However applying the above criteria resulted in the identification of 6 collocates of 'لله' ('allah; Allah) across both corpora (3 collocates in each corpus) that could reveal the most prominent uses of this key tem and what semantic foci are associated with it. Table 4.7 below is a simplified version of two more elaborated tables that contain the collocates of the key term 'ALLAH' in both corpora along with their translations in English and other important statistical details (see Tables A6 and B5 in Appendices A and B respectively). The Joint column indicates how often the pair of words co-occur in the relevant corpus.

Table 4.7 Collocates of 'ALLAH' in both corpora

Ν	keyword Eng. trans.	Collocates Eng. trans.	Joint	MI	Corpus
1	کا Allah	verily (Allah) will aid لَشِعْسُرَنْ	7	5.652	=
2	in/ with Allah المنتقد	نَبُوَحَ (the Last) Day	5	12.743	lhn B
3	ئە to Allah	the pure الْخَالِصُ	11	8.646	8
4	۵) Allah	(may Allah send His) peace (upon him) سنم	106	7.751	≥
5	in with Allah بنت	^y no	6	4.004	dhaw
6	to Allah ش	praise الحمد	9	10.453	ali

Some of the collocates in Table 4.7 indicate positive association between these collocates and the different forms of the key term 'ALLAH' (e.g. the concepts relating to aid, pure, peace and praise). However, collocates 2 and 5 are not immediately suggestive of any positive semantic association. Analysis of clusters and concordances will reveal more about these two collocates. Let us first consider the collocates of the keyword 'ش' ('allah; Allah) in Ibn Baz.

4.2.1.2 The collocates of the key term 'ALLAH' in Ibn Baz

The first collocate of 'ALLAH' in Ibn Baz tends to occur within a phrase which is a direct quote from Qur'anic verses. The verb 'لينصر' (*yanşur*; aid) in the phrase 'لينصرُنَ' (*layanşuranna*; verily (Allah) will aid) starts with the assertive particle 'L' (*la*) and ends with the geminated 'نُ' (*nna*) that indicates confirmation which occurs 7 times. This construction is equivalent to 'verily... will' in English. Initially I could claim that this collocational environment indicates the certainty and confirmation of the 'aid' that is expected from 'Allah'. However, the contextual environment in the following concordance lines (Concordance 4.1) reveals more (note that I shall choose only a sample of the concordance lines that indicate the most typical function of a linguistic construction in the corpus):

Concordance 4.1 Concordance of 'ليَنصرَنْ' (layanşuranna; verily (Allah) will aid)39

N

Concordance [...] فند وعد الله سبحانه بالتصبر للعاملين في النائيا و التواب في الأخرة ، قال عز و جل : (و <u>لمتصرّن ا</u>لله من يتُصرُد إنّ الله للتويُّ عَزِيزُ * النين إنْ هَكَنَّاهُمْ في الأرض أنّاهوا الصَّلاة و أنّوًا الزُّكَاة و أمرَّرا بِالْمَنْطُرُوفِ و تَبَوَّنَا عَنِ الْمَنْكَرِ * و لُمُ عَائِيَةُ الأُمُورِ)، [...]، وعدهم جل و علا بالنصر ، و هو يعم النصر في النائيا و التمكين فيها و النصر و الرضي من الله سبحانه يوم القيامة

[...] indeed Allah, the Glorified, has promised to aid those (Muslims) who do good deeds in this world as well as reward them (Paradise) in the Hereafter. Allah, the Exalted in Mighty and the Majestic, said: {Verily, Allah will aid those who aid His (Cause). Truly, Allah is All-Strong, All-Mighty * Those (Muslim rulers) who, if We give them power in the land, (they) enjoin the five compulsory congregational prayers, pay the Zakat (regular charity), enjoin Al-Ma'ruf (i.e. Islamic Monotheism and all that Islam orders one to do), and forbid Al-Munkar (i.e. disbelief, polytheism and all that Islam has forbidden)* And with Allah rests the end of (all) matters (of creatures)}, [...]. The Majestic and the most High has promised to provide them with aid which includes victory and power in land and on the Day of Resurrection they win the satisfaction of Allah, the Glorified.

[...] بأبيها النين أمنوا إنْ تتُصرُوا الله بتُسرُكُمُ ويَثَبَّتُ ألَّا تكُمُ) وقال تعالى : (و لَيَتَصَرُفُ الله من بتُسرَد [...] وقال تعالى : (و عذاله النين أمنوا منكم و عَمِلُوا الصَّرَخَة كَ لَتَنْ تَخْلَقُهُمُ في الأرض كُمَّا التُخْلَف الذين من قَبْنِهُ [...] بِعُبُونَتِي لا يَشُرِكُونَ بِي مَنِيَّنَ [...] قال ماتك بن أنس رحمة الله عليه في هذا المعنى : (لن يصلح آخر هذا لأمة إلا ما أصلح أوليه)

[...] O you who believe! If you aid (in the cause of) Allah, He will aid you, and make your foothold firm) and (Allah) the Exalted said {Verily, Allah will aid you, and make your foothold firm) and (Allah) the Exalted said {Verily, Allah will aid those who aid His (Cause) [...] and the Exalted said: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them [...] (provided) they (believers) worship Me and do not associate anything (in worship) with Me.}. In this respect Malik Ibn Anas, may Allah have mercy on him, said: (The last (people) of this nation will not be granted success unless they follow (in all good deeds) the early Salaf)

[...] بأبها الذين آمتوا إنَّ تتُصرُوا الله يتُمنزكُمُ ويُبَيَّتُ ألْمَاهكُمُ و مَن عز وجل : {و لِمَتَعنونَ الله من يتَصرُ [...] و قال سبحته :{ و اعتميفوا بخبّل الله جميعًا و لا تترَقُوا } [...] و الأيت في هذا المعنى كثيرة. ولكن ما دام أن الذاذ إلا من نساء الله منهم ، يطليون الهدى و التوجيه من خير كتاب الله و منة رسوله صلى الل عليه و سلم [...] فيتهم تن يجدوا طريقاً للخروج مما هم فيه من التخلف و التلاح فيها

[...] O you who believe! If you aid (in the cause of) Allah, He will aid you, and make your foothold firm} and (Allah) the Exalted in Mighty and the Majestic said {Verily, Allah will aid those who aid His (Cause) [...] and the Glorified said {And hold fast, all of you together, to the Rope of Allah (i.e. this Qur'an), and be not divided among yourselves}[...] But so long as Muslim leaders, except those who Allah has guided, seek guidance and directions from sources other than the Book of Allah (Qur'an) and the Sunnah of His messenger, may peace and blessings of Allah be upon him, [...], they will not find means to get rid of backwardness and disputes between them

As indicated by the above concordance lines (Concordance 4.1), the collocate 'لَيَتْصُرُنَّ (*layanṣuranna*; verily (Allah) will aid) is part of Qur'anic verses that are utilized by Ibn Baz to support his Salafi views regarding success (victory and succession (to the present rulers)) in life on Earth (Lines 1 and 3) and winning satisfaction and paradise in the Hereafter (Line 2), which, as Ibn Baz claims, is granted to those who follow the commandments of Allah. Thus, a positive religious semantic relation links the collocate 'لَيَتْصُرُنَّ (*layanṣuranna*; verily (Allah) will aid) with the keyword 'Allah'. This collocate indicates that Ibn Baz characterises Allah as a powerful God who can grant success (victory and dominance in life on Earth and Paradise in the Hereafter provided Muslims follow His commandments). Let us now move to delineate the collocate of 'بالله (*billah*; in/with/by Allah).

^{39.} For the translation and transliteration conventions and other symbols used in thesis, see Section 3.3.2 in Chapter 3 and Appendix C.

The second (as well as the third) form of the key term 'ALLAH' is preceded with a one-letter preposition. The preposition that is attached at the beginning of the keyword 'النَّهُ' (*'allah*; Allah) is 'ب' (*bi*; in/with). The collocate of (*billah*; in/with/by Allah) is 'ب' (*'al'āḥir*; the Last (*'alyawm*; (the Last) Day) which also co-occurs with the word 'الأَخِرِ '*'al'āḥir*; the Last (Day)) (MI score 12.799). They are mostly used together as shown in Concordance 4.2 which again gives the most typical cases of the collocation.

Concordance 4.2 Concordance of the collocate النوم ('alyanwm; (the Last) Day)

N	Concordance
	[] و يقول سبحانه : (و ما الحُمَّلُنَّةُ فيه من شيء فَحُكُمُه إلى الله نَائِكُم الله رَبَّي علَّذِهِ تَرَكَّلُتُ و الَيْهِ أُبِيبُ) [] و يقول سبحانه : (و ما الحُمَّلُنَّةُ فيه من شيء فَحُكُمُه إلى الله نَائِكُم الله رَبَّي علَّذِهِ تَرَكَّلُتُ و الَيْهِ أُبِيبُ) [] و يقول سبحانه : (و ما الحُمَلُنَّةُ فيه من شيء فَحُكُمُه إلى الله نَائِكُم الله رَبِّي علَّذِهِ تَرَكُلُتُ و اللَّهِ أُبِيبُ) [] و يقول سبحانه : (و ما الحُمَلُنَّةُ فيه من شيء فَحُكُمُه إلى الله نَائِكُم الله رَبِّي علَّذِهِ تَرَكُلُتُ و اللَّهِ أُبِيبُ) [] و يقول سبحانه : (إنَّ كنتُم قُرُولُونَ بِالله و ا ليَوْم الأخِي) تم ذكر فيه نحكم خير للعباد في العاجل و الراح و أحسن عاقبه [] فقل سبحانه : (إنَّ كنتُم قُرُولُونُ بِالله و اليَوْم الأخيرُ) تم ذكر أنه ذكر خير للعباد في العاجل و الأجل و أحسن عاقبه [] Ind the Glorified says: {And in whatsoever you differ, the decision thereof is with Allah (He is the ruling Judge). (And say O Muhammad to these polytheists:) Such is Allah, my Lord in Whom I put my trust, and to Him I turn (in all of my affairs and) in repentance.} []
2	to them [the Qur'an and the Sunnah] to judge in any dispute, then He mentioned that as a condition of faith as He, the Glorified, said: { if you believe in Allah and in the Last Day.} then He mentioned that sooner or later this is better and good end (for the worshippers) [] [] قال تعلى : (لا تُجدُ قُوْمًا يُؤْمِنُونَ بِانَهُ و اليَوْمِ الأَحْمِ يُوَادَّرِنَ مِن حَدَّ اللهُ و رسوله و لو كُمُوا آيَانَهُمُ أو أَخْرَائِهُمُ أو أَخْرَائِهُمُ أو أُخْرَائِهُمُ أو القومية] يقول : أَهصوا [] [] الله روية إلا مصالحكم وتستردوا أُمجنكم ، و كأن الإسلام وقف في طريقيم ، وحل بينيم و بين أمجدهم ، هذا و الله هو الجهل و التلبيس [] [] الله ويقد المحكم وتستردوا أُمجنكم ، و كأن الإسلام وقف في طريقيم ، وحل بينيم و بين أمجدهم ، هذا و الله هو الجهل و التلبيس [] [] الله ويقد المحكم وتستردوا أُمجنكم ، و كأن الإسلام وقف في طريقيم ، وحل بينيم و بين أمجدهم ، هذا و الله هو الجهل و التلبيس [] [] الله ويقد المحمول مع الحمل الله ويقد المحمور أو معنا مع من مع الله من الحمل و التلبيس [] [] الذين عن القومية ، [] مع من المحمور المحمور مع الحمل وقف في طريقيم ، وحل بينيم و بين أمجدهم ، هذا و الله والتلبيس [] [] الله ويقد المحمول مع الحمل إلى الله ويقد المحمور مع الحمل و التلبيس [] الله ويقد المحمول مع من المحمول إلى الله ويقد المحمور ، إلى الله ويقد مع من المحمول و التلبيس إلى إلى الله و الله ويقد مع مع المحمول إلى الله ويقد مع مع المحمول و المحمول و التلبيم ويقد مع المحمول و التلبيس [] [] الم والم مع مع المحمول و المحمول والم المحمول ويقد مع مع المحمول و التلبيم والم إلى إلى إلم والم إلى والم إلم والم إلى والم إلم والم إلى إلم والم إلم وال
	المقدد المحدجة تتلخص في : الإيمان بالله ، و ملائكته ، و كنه ، و رسله ، و اليوم الأخر ، و بالقدر خيره و نشره ، فهذه الأمور السنة هي أصول العقيدة الصحيحة التي نزل بها كتاب الله العزين ،[]. و أنلة هذه الأصول السنة في الكتاب و السنة كثيرة جداً ، فمن ذلك قول الله سبحانه : (ليس ألي أن تُوَلُوا وُجُو هُكُم قِبَلَ الْمَشْرِقِ و المغرب و لكِن ألير من أمن بالله و المؤمر الملائكة و الكتاب و النبيين) The correct creed is embodied in the following: belief in Allah, His Angels, His Books (scriptures and revelations of Allah to all His messengers), His messengers, the Last Day and the Divine Preordainment (good and bad). The previous six particles are the fundamentals of the true creed that were sent down in the honourable Qur'an [] there are many indications of these six fundamentals in the Qur'an and the Sunnah such as what the Glorified said: (It is not Al-Birr (piety, righteousness, and each and every act of obedience to Allah, etc.) that you turn your faces towards east and (or) west (in prayers); but Al-Birr is (the quality of) the one who believes in Allah, the Last Day, the Angels, the Book, the Prophets}

The phrase 'اليَوْمِ الْآخِرِ' (*'alyawm 'al'āḥir*; the Last Day) is mentioned by Ibn Baz in Line 3 as one of the six articles of faith (belief in Allah, His Angels, His Books, His messengers, the Last Day and the Divine Preordainment - good and bad). According to Ibn Baz the six articles of faith epitomize the correct creed (Line 3). The phrase 'اليَوْمِ الْآخِرِ' (*'alyawm 'al'āḥir*; the Last Day) appears in the Qur'anic 4-word cluster 'بِاللهُ وَ الْآخِرِ' (*billah wa 'alyawm 'al'āḥir*; in Allah and the Last Day) that occurs 5 times in 3 texts of Ibn Baz's corpus and is always preceded by words derived from the triliteral root 'آ م ن' (*ā-ma-na*', to believe/have faith in)⁴⁰. This cluster (in Lines 1 and 2) could be interpreted as being ideologically deployed by Ibn Baz to legitimate his stances vis-à-vis two other themes. In Line 1 Ibn Baz mentions that Muslims should, in matters of disputes, refer to the Qur'an and the Sunnah of the Prophet of Islam (i.e. Islamic Laws (Sharia) as opposed to man-made laws) which according to his explication of the Qur'anic verses (Line 1) is a requisite of faith in Allah and the Last Day. And Line 2 indicates Ibn Baz's negative representation of nationalism which, according to him, opposes belief in Allah and the Last Day. His negative representation of Arab nationalism shows that the semantic foci in Ibn Baz's corpus are not always theological. Here Ibn Baz engages himself in politics (for more information about Ibn Baz's representations of social actors see Section 6.2). Let us expand Line 2 for further clarification:

Figure 4.1 Expanded version of Concordance line 2 in 4.2

هؤلاء القوميون يدعون إلى التكل حول القومية العربية مسلمها وكافرها [...] و يقولون: إن نظامها لا يفرق بين عربي و عربي ، و ان تقرقت أديانهم ، فهل هذا إلا مصادمة لكتاب الله ، و مخالفة لشرع الله ، و تحد لحدود الله [...] ونظام القومية يقول : كلهم أولياء مسلمهم و كافرهم و الله يقول : (شرّع لكم من الأين ما و صُمى يه نُوحًا و الزي أوُخليًا إليُهم إليك و ما و صُبِّنًا به إبْرَاهِيمَ و مُوسَى و عيسَ أَنْ أَقِيمُوا الدين و لا تُتَثَرُقُوا فيه) [...] و قال تعالى : (لا تُجدُ قَوْمًا يُؤمِنُونَ بالله و الق<mark>يّق الأخري</mark> يُوَادُونَ من حَدًّا الذين و لا تُتَثَرُقُوا فيه) [...] و قال تعالى : (لا تُجدُ قَوْمًا يُؤمِنُونَ بالله و الق<u>يقيم المعلم و لم</u> قاله العرب و من عرب أن أقيم الذين من حَدًّا الله و لو كانُوا آياة هُمُ أو صُبِّنًا به إبْرَاهِيمَ و مُوسَى و عيسَ أَنْ أقيمُوا الدين و لا تُتَثَرُقُوا فيه) [...] و قال تعالى : (لا تُجدُ قَوْمًا يُؤمِنُونَ بالله و الق<u>يقيم الأخو</u> يُوَادُونَ من حَدًّا الله و رسوله و لو كانُوا آياة هُمُ أو أَنْنَاءَهُمُ أو إخْوَانَهُمُ أو عُمَيبَرِيَهُمُ) وسّرع [...] دعاتها [لقومية] يقول: أقصوا الذين عن القومية ، و افصلوا الذين عن الدولة ، وتكالوا حول أنفسكم وقوميتكم ، حتى تدركوا مصالحكم وتستردوا أمجادكم ، و كان الإسلام وقف في طريقيم ، وحال بينهم و بين أمجادهم ، هذا و الله هو التيبس [...] و قدهم على الفسكم وقوميتكم ، ولى يتكلوا تحت راية الإسلام ،[...] و عدهم على ذلك النصر و العاقية الحيدة [...]

These (Pan Arab) nationalists propagate that all Muslim or non-Muslim Arabs should be gathered as one race (under the banner of Pan Arab nationalism) [...] they say that nationalism views all Arabs to be equal despite the fact that they have different religions. This is a sheer contradiction to Allah's Book (Qur'an) and His Sharia and transgressing the limits that Allah has ordained [...]. Additionally, it (nationalism) considers all Arabs to be friends supporting each other whereas Allah said: {He (Allah) has ordained for you the same religion (Islamic Monotheism) which He ordained for Noah, and that which We have revealed to you (O Muhammad), and that which We ordained for Abraham, Moses and Jesus saying you should establish religion and be not divided therein}[...] and the Exalted, also said: {You (O Muhammad) will not find any people who believe in Allah and the Last Day, making friendship with those who oppose Allah and His Messenger (Muhammad), even though they were their fathers or their sons or their brothers or their kindred (people)}. And the law of the advocates [of (Arab) nationalism] determines that separation between the religion [Islam] and nationalism will guarantee well-being and regain your [Arabs] glorious and honourable past as if Islam is an obstacle that obstruct them [Arabs] to regain their past glory. By Allah, this is ignorance and mystification of the truth [...] and Allah has made it obligatory for Muslims to gather and support each other under the banner of Islam, [...] and consequently He (Allah) has promised to render them victorious and powerful and will grant them blessed end [...]

In the above extract Ibn Baz continues to present Arab nationalism and its advocates negatively. He claims that Arab nationalists want to separate religion from the state and make Muslims and non-Muslims as brothers neglecting the Islamic faith which are secular concepts

^{40.} I follow the traditional grammar text books in writing the roots of Arabic words in disconnected letters.

that all Salafis reject. Ibn Baz thus utilizes a Qur'anic verse in order to legitimate his rejection of Arab nationalism.

The last word form of the key term 'ALLAH' is preceded with the preposition 'لَنَّزُانَ to). This one-letter preposition usually 'predicates the concept of belonging in both concrete and abstract senses' (Ryding, 2005, p. 371). The collocate that co-occurs with the word form 'أَنْ (*lillah*; to Allah) is the adjective 'الْخَالِصُ' (*'alhālis*; the pure) (MI score 8.646). Concordance 4.3 shows 3 concordance lines of the contextual environment of the collocate 'أَلْخَالِصُ' (*'alhālis*; the pure).

Concordance 4.3 Concordance of the collocate 'الْخَالِصُ' ('alhālis; the pure)

_	('alhalis; the pure)
Ν	Concordance
1	و الحبادة : هي صاعته سبحانه و صاعة رسوله محمد صلى الله عليه و سلم [] و قال عز و جل : {قَاطَبُو الله مخْلِصُا له الدين * الا لله الذينُ المخاصرُ } [] و الأيات في
1	هذا المعنى كثيرة ، و كلها نتل على : وجوب إفراد الله بالعبادة ، و معلوم أنَّ الدَّعاء بقواعه من العبادة ، فلا يجوز لأحد من الناس أن يدعو إلا ربه ، و لا يستحين و لا
	[]، بن باز به از الله الله الله الله الله الله الله
	And the worship is obeying Him, the Glorified, and His messenger, Muhammad, may Allah send His blessings and peace upon
1	him [] and the Exalted in Mighty and the Majestic said: {So worship Allah (Alone) by doing religious deeds sincerely for
	Allah's sake only. Surely the pure religion (i.e. the worship and the obedience) is for Allah only [] and the Qur'anic verses
	about this issue are many that all indicate the obligation of worshipping Allah alone. And it is known that all types of
	supplication are part of worship, thus, according to these Qur'anic verses, it is impermissible for all people to supplicate to and
	neither seek nor beseech the help from other than Allah []
	[] معنى لا إنه إلا الله [] لا معبود بحق إلا الله [] و هو الذي دل عليه كتاب الله سبحانه في مواضع من القرآن الكريم [] و في قوله عز و جل: {نتَزَيْلُ الْكَتَاب من الله الْحَرَيْزِ الْحَكِيم * إنا أنزلنا إليك الكتاب بالحق فاعْبُر الله مُغْلِصًا له الدين * الا سُر الذين التُخُو من دونه أوليا، ما تُغَبَّدُهُمُ إلا ليتَزَبُونَ إلى اللهُ لَفَقَ
	المعنى أنهم يقولون ما تعدهم إلا أيقربونا إلى الله زلني، فرد الله عليهم تلك يقوله سبحاته : {إنَّ الله يتخُكُم بينهم في ما هم فيه يُخْتَبُونَ إنَّ الله لا يهدي من هو كُذِبٌ كُقُارُ }.
	[] 'no God but Allah' [] means that no one deserves to be worshiped except Allah [] and this what the Book of the Allah,
	the Glorified, indicates in many verses in the Glorious Qur'an [] and in the following utterance of the Exalted in Mighty and
2	the Majestic: {The revelation of this Book (the Qur'an) is from Allah, the All-Mighty, the All-Wise. Verily We have sent down
	the Book to you (O Muhammad) in truth: So worship Allah (Alone) by doing religious deeds sincerely for Allah's sake only.
	Surely the pure religion (i.e. the worship and the obedience) is for Allah only. And those who take Auliya (protectors, helpers,
	lords, gods) besides Him (say): "We worship them only that they may bring us near to Allah." } that is they say that we
	worship them only to bring us near to Allah, but Allah, the Glorified, rejected that and said: {Verily Allah will judge between
	them concerning that wherein they differ. Truly, Allah guides not him who is a liar, and a disbeliever}.
	حتيقة هذه العبادة : هي إفراد الله سيحاته بجميع ما تعبد العباد به من دعامٍ ، و خوفٍ ، و رجاءٍ ، و صلامٍ ، و ضوم ، وذبح ، و نذي ،[] و غالب القرآن الكريم نزل في
	هذا الأصل العظيم، كتوله سبحاته: (فاغبُر الله مخلصًا له الدين الالله الذين الخالص) [] فشهادة أن لا إله إلا الله تتنضى: إخلاص العبادة لله وحدد والقبها عما سواد، و
	اهذا معصف لا المالا الأملاح
	The real meaning of this worship is to single out Allah, the Glorified, in all forms of worship such as supplication, fear (of
3	Allah), hope (for His mercy), prayer, fasting, sacrifice (slaughtering animals for His sake) and making pledges (to make
	charitable acts for the sake of Allah), [] and most of the Qur'anic verses that were sent down about this great principle such
	as the following utterance of the Glorified: {So worship Allah (Alone) by doing religious deeds sincerely for Allah's sake only.
	Surely the pure religion (i.e. the worship and the obedience) is for Allah only [] and the testifying that there is no God but
	Allah' necessitates the sincerity of dedicating the worship for Allah alone. That is the meaning of 'there is no God but Allah' []

All the lines in Concordance 4.3 indicate the concept of 'عبادة' (*'ibadah*; worship) which is modified by the adjective 'الْخَالِصُ' (*'alḫāliṣ*; the pure) which occurs in Ibn Baz's corpus as he quotes the following Qur'anic verses:

{So worship Allah (Alone) by doing religious deeds sincerely for Allah's sake only* Surely <u>the pure</u> religion (i.e. the worship and the obedience) is for Allah only. And those who take Auliya (protectors, helpers, lords, gods) besides Him (say): "We worship them only that they may bring us near to Allah." Verily Allah will judge between them concerning that wherein they differ. Truly, Allah guides not him who is a liar, and a disbeliever.} Holy Qur'an 39:2-3.⁴¹

The above Qur'anic verse is utilized by Ibn Baz to delineate the concept of 'worship' (Lines 1 and 3) which he links to the meaning of essential Islam with the phrase 'no God but Allah' (Line 2). This phrase constitutes the first half of the first pillar of Islam (there is no God but Allah and Mohammad is His messenger).⁴² Thus the third form of the key concept 'ALLAH' is linked to other semantic foci, viz., the first pillar of Islam and the concept of 'pure worship'. In Line 2, Ibn Baz recontextualizes the Qur'anic verses above which indicate that people should sincerely adopt the servitude of Allah and avoid polytheism. Such recontextualization contributes to the Salafi representation of divinity as related to worship.

So far, we have seen the three forms of the key concept 'ALLAH' as used by Ibn Baz. The key term 'ALLAH' in all its three forms are used in recontextualized different theological semantic foci that contribute to Ibn Baz's Salafi construction of divinity in Islam. Allah is represented as helping people who help his cause, rewarding them both in their life and in death. Ibn Baz also creates a strong association between belief in Allah and belief in 'the last day', implying that the two are mutually inclusive but separate from nationalist ideology, which he views negatively. Finally, Ibn Baz describes worship of Allah as the pure religion

^{41. (39:2-3)} indicates that these Qur'anic verses are number 2-3 in Chapter 39.

^{42.} In the Sunni Islam the pillars of Islam are as follows: 1- Testimony; to witness that there is no god but Allah and Mohammad is his messenger, 2- Performing the five daily prayers, 3- Fasting during the days of the month of Ramadan (of the Muslim calendar), 4- Alms-giving (zakat), and 5- Performing pilgrimage by bodily and financially able adult Muslims.

(thus implying that anything else is impure). Now let us look at how 'Allah' is represented by Alhawali.

4.2.1.3 The collocates of the key term 'ALLAH' in Alhawali

Table 4.7 above contains three collocates for the three forms of the key term 'ALLAH' in Alhawali. These collocates are 'سلم' (*sallam*; (may Allah send His) peace (upon him) (MI score 7.751), 'Y' ($l\bar{a}$; no) (MI score 4.004)) and 'الحمد' (*'alḥamdu*; praise) (MI score 10.453). While these collocates are different to the collocates of 'ALLAH' for Ibn Baz, an initial glance at how they are used in the Alhawali texts suggests that like the collocates for Ibn Baz, they are used in formulaic phrases that represent Allah positively as looking kindly on people who are favoured by the authors. So while Ibn Baz uses phrases like 'verily Allah will aid', Alhawali uses phrases like 'may Allah send His peace upon him'.

Let us first consider the collocate 'سلم' (*sallam*; (may Allah) send His peace (upon him). This collocate is of polysemous and homographic nature.⁴³ It occurs 97 times where 90 occurrences indicate the religious meaning 'سلم' (*sallam*; may (Allah) send His peace (upon him)). This is always mentioned by Alhawali after referring to the Prophet of Islam as shown in Concordance 4.4.

^{43. &#}x27;ملم' could either, without gemination, be transliterated as 's<u>alima</u>', (be saved), or 's<u>ilm</u>', (peace n.). However, the consonant /l/ could be geminated as in 's<u>ollam</u>' (ladder) or 's<u>allam</u>a' which could be used as a verb that indicates Muslim salutation (to say 'assalāmo 'alaikom', 'peace be with you'). Also, 'sallam' could be the past tense of 'yosallim', to submit. In addition, 'sallam' could be used parenthetically with other words including 'salla' (alaihi 'alaihi wa sallam'; may Allah send His blessings and peace upon him) (for more information on homographs and polysemous words in my Data see Chapter 3, Section 3.3.2).

Concordance 4.4 Concordance of the collocate 'سلم' (sallam; (may Allah) send His peace (upon him))

N	Concordance
	[] أنها [العلمانية] باختصار: نظام طاعوتي جاهلي يتنافى مع لا إله إلا الله [] و قال ابن كثير رحمة الله عند تفسير قوله تعالى: ((فلا و رَبَّكَ لا يُؤْمِنُونَ حَتَّى يُحَكَّمُوكَ
	فيمًا تُتَجَز بَيْنَهُمُ [] النساء:65. يتسم الله بنفسه الكريمة المقدسة أنه لا يؤمن أحد حتى يحكم الرسول صلى الله عليه و سلم في جميع الأمور [] ، مراده من ذلك بيان أن
	الجاهلية صفة تلحق كل من حكم بغير ما أنزل الله، و ليست فترة تاريخية انتهت بظهور الإسلام []
	In short, indeed it [secularism] is a despotic (a tyrant worshiped as an idol besides Allah) and ignorant (pertaining to pre-Islamic ignorant
	Arab) system that contradicts 'no God but Allah' (Islamic monotheism), [] and this is indicated in Ibn Kathir's explication -may Allah
1	have mercy on him- of the following utterance of the Glorified: [But no, by your Lord, they can have no Faith, until they make you (O
	Muhammad) judge in all disputes between them [] Holy Qur'an 4:65. Ibn Kathir said that Allah is swearing by His Glorious and Holy
	Self that none can claim faith unless s he subordinates him herself in all affairs to the judgement of the messenger -may the blessings and
ĺ	peace of Allah be upon him- []. Ibn Kathir's aim of his explications is to show that 'aljahiliyyah' (pre-Islamic status of ignorant Arabs) is
	an appellation that is used to describe all those who refer to other than Allah's Laws that He sent down. And it is not a description of a
	period that was before Islam []
	و يقارن المستشرق الأمريكي بودلي بين النصرانية و الإسلام في ذلك قائلًا: (تو أن القديس بطرس عاد إلى روما الإمتلاً عجباً من الطوس الضخمة و ملابس الكهنوت
	المزركتية و الموسيقي الغريبة في المعبد المقرونة باسمة []، و لكن إذا ما عاد محمد صلى الله عليه و سلم إلى أي مسجد من المساجد المنتشرة بين لندن و زنزيار، فإنه
	سرجد نفس السُعائر البسيطة التي كانت تقام في مسجده في المدينة).
2	And in his comparison between Christianity and Islam, the American orientalist Bodley says: (If St. Peter returned to Rome, he might be
-	puzzled by the gorgeous ritual, by the gaudy vestments and the strange music in the temple associated with his name [], But if
	Mohammad, 'may the blessings and peace of Allah be upon him', dropped into any Mosque between London or Zanzibar, he would find
	the same simple rites as in his brick and palm-beamed house of worship [my italics: the translator used the word 'mosque'] in Medina).
<u> </u>	م المساعدة بد إعليه ومنه ما المسلمين المنظر من المسلمان والم والمراحة الراح وعنادة من مناسبة والمسلمين وتسعرون بصرورة الولاء [] و الحمد لله أن المسلمين المقوا إلى حد كبير، فنجد ما حدث في الفنانستان أو الغلبين أو البوسنة و الهرسك و غير ها؛ جعل المسلمين يتسعرون بصرورة الولاء
	[] والحمد ندان المشاهرين الماق إلى كما يبير، فجد ما كذك في العائشان والعبين أو البوتشة والبوتشة والعرفي المشاهرين يشعرون بصرورد الوزاء [الإسلامي،[] جملت المسلمين يشعرون بأننا أمة واحدة في أمالها و آلامها؛ لأنها واحدة في عقيدتها و قبلتها و كتابها و سنة نبيها صلى الله عليه واسلم وافي كل الأواصل و
	الروابط، و لا يوجد أمة أوغل وأعمق في التاريخ من هذه الأمة؛ لأنها فوق كل القوميات و العصريات والعصبيات. المحاط محاط منا لمحسب معاد منه فاصل معاد منه المعاد المحالية المحالية من المحسب المحالية منها منها منها منها ال
1	[] and all praise is due to Allah that all Muslims are wide-awake to a large extent. We deem the incidents that happened in Afghanistan
13	Philippines, or Bosnia and Herzegovina and others, made Muslims feel the necessity of being loyal to Islam, [] [these incidents made
	Muslims feel that we are one nation in its hopes and pains, because it (Muslim nation) is one nation in its creed, its qiblah (prayer direction
	towards Makkah), its Book (the Qur'an) and the Sunnah of its Prophet, may the blessings and peace of Allah be upon him, and it is one
	nation in all bonds and relationships. There is no other nation that is deeply rooted in history than this Muslim nation, because it rejects
	nationalism, racism and all sorts of bigotry.

In all the occurrences, the collocate 'سلم' (*sallam*; (may Allah) send His peace (upon him)) is used with the keyword 'الله' (*'allah*; Allah) in the optative construction 'سلم عليه و سلع ' (*salla 'allahu 'alaihi wa <u>sallam</u>'*; may Allah send His blessings and <u>peace</u> upon him) which, according to the teachings of Islam, is said in paying due respect to the Prophet of Islam as the following Qur'anic verse tells:

{إِنَّ اللَّهَ وَمَلَائِكَتَهُ يُعْمَلُونَ عَلَى النَّبِيِّ يَا أَيُّهَا الَّذِينَ آمَنُوا حَلُّوا عَلَيْهِ وَسَلِّمُوا تَسْلِيمًا } سورة الأحزاب اية ٥٦.

{Allah sends His Salat (Blessings) on the Prophet (Mohammad), and also His angels (ask Allah to bless and forgive him). O you who believe! Send your Salat on (ask Allah to bless) him (Mohammad), and (you should) greet (salute) him with the Islamic way of greeting (salutation i.e. 'السلام عليكم' ('assalāmo 'alaikom; peace be on you)} (33:56).

In these concordance lines, Alhawali asks Allah to bestow peace upon the Prophet of Islam for reasons which go beyond mere worship for its own sake. In Line 1 (Concordance 4.4), Alhawali declares that secularism, as a non-Islamic system, is opposed to the teachings of the Qur'an that made it clear for Muslims to subordinate themselves in all affairs to the judgment of the Messenger of Islam (i.e. to the Laws of Islam –Sharia). Here the collocate 'سلم', which is, also, an indicator of argumentation and a social actor, is ideologically recontextualized to refute secularism calling Muslims to adhere to the Sharia Laws that were sent down to the Prophet of Islam instead. In this respect, we can start to see the different dispositions of both writers regarding 'adherence to the Laws of Sharia'.

الله Baz has recontextualized following the Sharia of Islam in all disputes with no reference to secularism per se, utilizing faith in Allah and 'اليَوْمِ الْأَخِر' ('alyawm 'al'āḥir; the Last Day) to legitimate his stance of rejecting the man-made laws, instead arguing that Muslims should refer to the Islamic Laws (Sharia) instead. On the other hand, Alhawali's recontextualization of adherence to the Sharia is further extended by recontextualizing Ibn Kathir's explication of the Qur'anic verse 4: 65 who, according to Alhawali, stated that the 'الجاهلية' ('alğahiliyyah; ignorance of the pre-Islamic days) is an appellation that describes adherence to other than the Sharia of Allah in all times rather than a description of a specific period before Islam. This is done by Alhawali to include secularism as opposing the Sharia in order to legitimate his stance and refutation of secularism.

In Line 2, Alhawali utilizes Bodley's (1946, p. 343) comparison between the ritualistic practices of modern Muslims and Christians. Alhawali adds the collocate 'سلم' (*sallam*; (may Allah) send His peace (upon him)) to the quote from Bodley (1946) which may give the impression that Bodley himself used this collocate. In this line we find two important religious social actors (St Peter and Mohammad) in two religions (Christianity and Islam), who belong to two different historical epochs, and are hypothetically projected to modern

time by means of anachronism that, according to van Leeuwen (2008: 50), is used to naturalize ideological discourses. Bodley's argument is ideologically recontextualized by Alhawali to legitimize his negative stance towards the change of rites in Christianity made by the Church and the practice of Islamic rites by modern Muslims that 'resembles' Islamic ritualistic practice during the life of the messenger of Islam. Such recontextualization helps Alhawali to construct the idea of an ideal Muslim community (or nation) which is further developed in Line 3 in Concordance 4.4 above.

In concordance line 3, Alhawali gives a list of examples of how Islam constitutes one nation, including the Sunnah of the Prophet of Islam, Mohammad, who is positively represented via the phrase 'may Allah send His peace upon him'. Thus the phrase 'may Allah send His peace upon him' is generally used by Alhawali as a way of showing respect for various figures, but mainly Mohammad.

The second collocate I shall consider in this section is ' \mathcal{Y} ' ($l\bar{a}$; no). The Arabic phrases that contain the negative particle ' \mathcal{Y} ' ($l\bar{a}$; no) and the key term 'ALLAH' are usually used in religious expressions such as the 4 and 7-word clusters in Table 4.8, which occur in both corpora that compose the research data of this study.

N	Clusters	Eng. trans.	Freq	Texts	Corpus
1	لا إله إلا الله	there is no God but Allah	25	2	Alhawali
2	لا إلم إلا الله	there is no God but Allah	40	3	Ibn Baz
3	حول و لا قرة	(no) strength and no power (save by Allah)	17	4	Ibn Baz
4	لاحول و لاقوة إلا بـالله	no strength and no power, save by Allah	12	2	Ibn Baz

Table 4.8 Word clusters that contain 'V' ($l\bar{a}$; no) and the key term 'ALLAH' in both corpora

In all the clusters in Table 4.8, the particle ' \mathcal{V} ' ($l\bar{a}$; no) that indicates absolute negation is used in close proximity with the key term 'ALLAH'. In clusters 1 and 2, it is used with the word form 'ilia' ('allah; Allah) as in ' \mathcal{V} ($l\bar{a}$ 'ilaha 'illa 'allah; there is **no** God but Allah) and with the word form 'iliah; in/with/by Allah) as in clusters 3 and 4 that contain the exceptive construction that has two-part formation of the negative particle 'Y' ($l\bar{a}$; no) as in 'Y Y و ...' ($l\bar{a}$... wa $l\bar{a}$; no ... and no). This two-part negative construction is usually used in religious expressions indicating absolute negation (Ryding, 2005, p. 646) which could be translated into English as 'neither... nor' or literally as 'no... and no' as in 'Y حول و لا قوة إلا ' $L\bar{a}$ hawla wa <u>la</u> qowwata 'illa billah; there is <u>no</u> strength and <u>no</u> power, save by Allah). The negative particle 'Y' ($l\bar{a}$; no) in the two-part construction is the target in this analysis, because it is the collocate of the second word form 'بالش' (*billah*; in/with/by Allah). Concordance 4.5 below shows lines that repeat the same function of the collocate 'Y' as used with the word form 'بالش' (*billah*; in/with/by Allah) in Alhawali.

Concordance 4.5 Concordance lines of the collocate ' \mathcal{Y} ' ($l\bar{a}$; no) in Alhawali.

	Concordance 4.5 Concordance lines of the conocate 3 (12; no) in Alhawaii.
N	Concordance
1	و الخلاصة [] أن الواجب على الأمة أن تعرف الواقع [في السعودية الثاء از مة الخليج عامي 1990-1991] على حقيقته [] لاسيما و قد جاء على لسان أكثر من مسؤول في هذه البلاد و غيرها أن المنطقة دخلت في نفق مظلم لا يعلم نهايته إلا الله و حفاً نطق، فقد كنا نظن أن الأمة قد استقرت في الفاع و أنه ليس وراء واقعها من سقوط ، فإذا بهذا اللغاع السحيق يفتح فلا لتدخل في نفق عميق و لا حول و لا قوة إلا بالله. in short [] the [Muslim] nation should realize the actual state of affairs [in Saudi Arabia during the Gulf Crisis in 1990- 1990] and other countries who truly stated that the upon and other countries who truly stated that the whole region has entered into a dark tunnel that no one knows its end but Allah. We thought that the [Muslim] nation has resided in a deep abyss and will not fall any further, but suddenly this abyss opened into a further deeper tunnel and there is no strength and no power, save by Allah
2	إنَّ أُمَّة تعيش حالة الحرب الشاملة يجب أن تكون أبعد الناس عن اللهو و الثرف [] فهي في رباط دائم و تغور متوالية ، و لا قرة لها إلا باش [] ، و على ألم العلم و الدعوة أن يكونوا قدوة للناس [] فإن الله سبحانه و تعالى لم يعلق و عده بالنصر و النجاة و الإعلام و الدعوة أن يكونوا قدوة للناس [] فإن الله سبحانه و تعالى لم يعلق و عده بالنصر و النجاة و الإعلام و العزة لمن اتصف بالإسلام بل خص به أهل العلم و الدعوة أن يكونوا قدوة للناس [] فإن الله سبحانه و تعالى لم يعلق و عده بالنصر و النجاة و الإعلام و العزة لمن اتصف بالإسلام بل خص به أهل الإيمان كما في قوله تعالى : (إنَّا للنصُرُ رُسُلنا وَالَّذِينَ آعلُوا في الْحَيَاةِ اللَّذَينَ و يؤم يَقُولُم الأُمَّابَة) A nation [Muslim nation] that is under a total war should eschew entertainment and luxury [] it is in a continuous territorial wars, and has <u>no</u> power, save by Allah [] it is a requisite on the people of [religious] knowledge and call to Allah to set themselves as good examples to ordinary people [] because Allah, the Glorified and the Exalted, did not promise victory, safety of Hell-fire, superiority and glory to those who are Muslims, indeed He dedicated them [safety of Hell-fire, superiority and glory] to people of faith [among Muslims] as the Exalted said: Verily, We will indeed make victorious Our Messengers and those who believe (in the Oneness of Allah - Islamic Monotheism) in this world's life and on the Day when the witnesses will stand forth, (i.e. Day of Resurrection)
3	و هذا [جهل اهداف التبشير] ما اضطر القس زويمر حرئيس مؤتمر القدس التبشيري. إلى إيضاح ذلك فقال: "أيها الإخوان الأبطال، و الزملاء الذين كتب الله لهم الجهاد في سبيل المسيحية و استعمار ها لبلاد الإسلام، [] و إنما مهمتكم أن تخرجوا المسلم من الإسلام ليصبح مخلوقاً لا صلة له بالله، و بائتائي لا صلة تربطه بالأخلاق التي تعتمد عليها الأمم في حياتها and [the unawareness of the goals of the missionaries] obliged priest Zwemer- the Chairman of the Missionary Council in AlQuds [Jerusalem]- to say: Dear brave brothers and colleagues whom God chose to struggle for Christianity and its colonization of the Muslim land [] your only mission is to drive a Muslim out of Islam in order to make him a faithless creature that has <u>no</u> relation with Allah, and consequently will be with no morals that nations depend on in their lives
4	إن الواقع السياسي المعاصر الذي تتعكّس عليه الصورة الحقيقية للجاهلية الأوروبية ليزخر بالدلائل الفاطعة و البراهين الفوية و يعج بالمتناقضات الصارخة و الظواهر الغربية التي تنذر مجتمعة بالمصير المتئوم والنهاية المروعة لعالم لا يؤمن بالله و لا يحتكم إلى شريعته indeed the reality of modern politics that reflects the real picture of the Europeans jahilyyiah (ignorance) that is replete with strong evidence and cogent proofs, and is full of gross contradiction and manifestations of Western life that foretell the inauspicious destiny and the terrifying end of a world that has <u>no</u> faith in Allah and does not enjoin His Sharia

Line 1 shows that the collocate ' \underline{V} ' ($l\bar{a}$; no) is used twice in the 7-word cluster ' \underline{V} equal to \underline{V}

الا بالله (<u>lā hawla wa lā qowwata</u> 'illa billah; <u>no</u> strength and <u>no</u> power, save by Allah). This

religious two-part absolute negative formation (no ... and no....) usually occurs towards the end of describing (and a rejection of) a negative situation/action which indicates a concession of the human weakness as compared to the Divine strength and power. In Line 1, Alhawali uses this absolute negation after describing the state of affairs in Saudi Arabia during the Gulf Crisis (1990-1991) where he, from a political Salafi perspective, includes all Muslim countries as one nation that 'has no power except by Allah'. In line 1, he quotes more than one Saudi official who claim the whole region has entered a 'dark tunnel'. This idea of the 'Muslim nation' in trouble is further developed in Line 2.

In Line 3, the collocate 'Y' ($l\bar{a}$; no) and the word form 'بالله' (*billah*; in/with/by Allah) are used together with the word 'مسلة' (*silah*; relation) forming the phrase 'لا صلة له بالله' ($l\bar{a}$ *silat laho billah*; [creature that] has no relation with Allah) which occurs in a reported speech. Alhawali uses a direct quote from a speech of the American missionary Zwemer (1867-1952) who, is quoted as declaring that the main purpose of the missionaries among Muslims is to

make Muslims faithless creatures that have *no* relation with 'Allah'; a further development of this theme of victimizing Islam and Muslims. Line 3 occurs in a book written by Alhawali that discusses Secularism. Under the subtitle 'Missionaries', Alhawali concurs that the process of secularization of Islamic communities started with such missionaries as indicated by the extended version of Concordance line 3.

Figure 4.2 Extended version of Concordance line 3 of the collocate ' \forall ' (*lā*; no)

[...] بدأت المخططات لإخراج الأمة الإسلامية من دينها، وتعريتها من متومات وجودها، و حملها [...] على العلمانية وانتَعْلَمت جيوش الغزو في ثلثة لجنحة كُبري هي: أمين الأحداث المباشر . 2- المستشرقون . 3- المبشرون كما يسمون . [...] 3- المبشرون: كما أن للمستشرقين و المبشرين أهداناً مشتركة، فإن لهم وسائل متناخلة، و يمكن القول بأن مبدان المُستَتْرِقِينِ الأساسي هو التقافة و الفكر، بينما يركز المبسرون جهودهم في النواحي الاجتماعية و التربوية. و ليس غزيبا أن يجهل المسلمون الدوافع الحقيقية للتبشير، فقد كان يجهلها بعض أتباع الإرساليات التبشيرية أنفسهم، إذ لم يكن الجميع يدركون أبعاد الخطة الجديدة و مراميها [...]. و هذا ما اضطر القس زويمز مرئيس مؤتمر القدس الفبشيري- إلى إيضاح ذلك فتال: "أيها الإخوان الأبطال، و الزمَّلاء الذين كتب الله لهم الجهاد في سبيل المسبحية و استعمارها لبلاد الإسلام، [...] و إنما مهمتكم أن تخرجوا المسلم من الإسلام ليصبح مخلوفاً لا صلة له بالله، و بالتألي لا صلة تربطه بالأخلاق التي تعتمد عليها الأمم في حياتها [...] لقد قبصنا لجيها الإخوان في هذه الحقبة من الدهر من تلت القرن التاسع عشر إلى يومنا هذا [وقت ما من النصف الأول من القرن المشرين] على جميع برامج التعليم في الممالك الإسلامية و نشرنا في تلك الربوع [...] والكنائس و الجمعيات و المدارس المسيحية الكثيرة، لتى تهيمن عليها الدول الأوروبية و الأمريكية [...] [...] the plans, that aim to drive the Muslim nation out of Islam, deprive it from the fundamentals of its existence and force it to adopt secularism, started. And the armies of invading [the Muslim World] united in three main wings: 1- Forces of direct occupation [of Muslim land]. 2- Orientalists. 3- The so called 'missionaries'. [...] 3- Missionaries: As the orientalists and missionaries have shared interests, they also share means. It could be said that the arena of orientalism is culture and intellect whereas the missionaries exert their efforts in the social and educational aspects. It is not surprising that Muslims are unaware of the real objectives of missionaries; indeed some of the missionaries are also unaware of these objectives because few of them know the dimensions and objectives of the new plan. And this obliged priest Zwemer- the Chairman of the Missionary Council in AlQuds [Jerusalem]- to say: Dear brave brothers and colleagues whom God chose to struggle for Christianity and its colonization of the Muslim land [...] your only mission is to drive a Muslim out of Islam in order to make him a faithless creature that has no relation with Allah, and consequently will be with no morals that nations depend on in their lives. [...] Dear brothers, we [the missionaries] --- controlled all educational programs in all Islamic States during the period from the last 30 years of the 19th century till today [sometime in the first half of the 20th century]. We distributed [...] Churches and many Christian Associations and schools that are controlled by European and American countries [...]

Thinking about interpretations of Alhawali's description of how missionaries aim to turn Muslims away from Allah, it is possible that this passage could exacerbate and perpetuate antagonism between his readers and the Western countries as colonial powers. The theme of 'the evil West' continues to be projected in Alhawali as in Line 4 in Concordance 4.5.

The last Line in Concordance 4.5 shows that the collocate 'Y' (*lā*; no) is used with verb 'بوامن' (*vo'min*; believe/have faith) followed by the word form 'بالله' (*billah*; in/with/by Allah). Alhawali uses the phrase 'لا يوامن بالله' (*lā yo'mino billah*; has no faith in Allah) to

indicate a faithless world which, according to him, does not enjoin the Sharia of Allah. This 'faithless world', Alhawali continues to negatively project, 'reflects the real picture of the European ignorance' which eventually will have 'inauspicious destiny' and a 'terrifying end'. The word 'قامان (*ğahiliyyah*: ignorance) in the phrase 'جاهلية الأوروبية' (*šahiliyyah*: ignorance) is significant as it is a recontextualization of a Qur'anic concept indicating the judgement and norms of life of the pre-Islamic days which are rejected by Islam:

{أَنَحُكُمَ الجُّامِلِيَّةِ يَبْعُونَ وَمَنْ أَحْسَنُ مِنَ اللَّهِ حُكْمًا لَقَوْمٍ يُوتِنُونَ } سورة المائدة اية ٥٠. {Do they then seek the judgement of (the days of) Ignorance (the conventions of the pre-Islamic period of Ignorance)? And who is better in judgement than Allah for a people who have firm Faith} Holy Qur'an 5:50.

Using the word '*alğahiliyyiah*' (ignorance of pre-Islamic days) in this context to describe modern Europe consolidates and perpetuates Alhawali's negative image of Europe.

Alhawali uses the collocate 'Y' ($l\bar{a}$; no) with the words ' $\bar{s}_{\bar{s}}$ ' (*qowwah*; power), ' $\bar{s}_{\bar{s}}$ ' (*silah*; relation) and ' $\bar{s}_{\bar{s}}$ ' (*yo'min*; believe/have faith) in order to construct the relationship between humans and God as one where God is all-powerful and humans need to cement their relationship with God by always believing in Him. These descriptions of 'Allah' are, from a political Salafi perspective, ideologically recontextualized to reject the status quo of the 'Muslim nation' which is, according to Alhawali, weak and being dominated by the West. Alhawali accentuates the war and conspiracy against Islam and the Muslim world (Lines 2 and 3), and he negatively represents European culture as being in a state of ' $\bar{g}ahiliyyah$ ' (ignorance) (Line 4).

The last collocate of the key term 'ALLAH' that will be considered here is 'الحمد' ('alḥamdo; praise) which collocates with the word form 'ش' (lillah; to Allah) creating a positive representation of 'Allah' as a God who is to be praised. This collocation is an intertextual reference in that it is derived from many Qur'anic verses such as the second verse

in the first Chapter in the Qur'an: 'الحمد شرب العالمين' ('alḥamdo Lilahi rabbi 'al'alamīn; all praise is due to Allah the Lord of the 'al'alamīn (the worlds: mankind, jinn and all that exists). It is also used at the beginning of the Muslim daily prayers, so will be familiar to all Muslims. It is an acknowledgement of the favours of Allah and a satisfactory indication about a present or previous status or a condition based on the belief that everything that happens is caused by Allah. It is usually uttered in conversation between Muslims such as asking a Muslim 'How are you?' The expected and the most usually used response is 'الحمد ش' ('alḥamdo lillah; all praise is due to Allah) which indicates that this Muslim is well. However, this phrase could be recontextualized in the different situations as shown by the concordance lines (Concordance 4.6) of 'الحمد' ('alḥamdo; praise) in Alhawali.

Concordance 4.6 Concordance lines of the collocate' ('alhamd; praise) in Alhawali

Ν	Concordance
	فالبعتيون السوريون الذين يربطهم بالعراق رابطة التومية و البعتية هم أيضاً. نخلوا ضمن الجيوش التي دخلت إلى العراق وقتلت من العراقيين ما
	تشت؛ فهذه تعمة . و ا لحدد شد فبحما كانوا يتقاتلون في لبنان، و منهم من يؤيد عون، و منهم من يؤيد شمعون أصبحوا يتقاتلون بأتضبهم و جهاً لوجه،
	<u>فاقصد</u> لله لأن القومية العربية في حالة احتضار.
	The Syrian Baathis (adherents of the Socialist Baath Party) who have nationalist and Socialist bonds with Iraq
1	were also among the armies (American-led coalition force in the Operation Desert Storm) that invaded Iraq and
1	killed many Iraqis. This is a grace (from Allah) -and praise is due to Allah. After they fought each other in Lebanon
	when some of them supported Aoun (the leader of the Free Patriotic Movement and a Lebanese Prime Minister
	1988-1990 who is backed by Saddam Hussein in the Civil war that started 1975 and ended in 1990) and some others
	supported Chamoun (Lebanese Christian leader -supported by Syria), they were face to face in a battle killing each
	other. So all <u>praise</u> is due to Allah, because Arab Nationalism is dying
	إن احتلال الكويت، تلك التشة التي قصمت ظهر البعير، سيتتهي بشكل ما و حينها ستلتت هذه القوى لتصحيح و ضعنا نحن كما المحوا مراراً و صرحوا،
	المان أحددنا تذلك؟ و هذا الإعداد أن يكفقا و أن يضيرنا فإن كفي الله المؤمنين التشل فلله الجمع و التسكر، و أن منه الأجر إن شاء الله و إن كانت الأخرى
	کن الجواب حاضراً و الرد سريعاً.
	The invasion of Kuwait -the straw that broke the camel's back- will end soon in a way or another, then these
2	Forces (American-led coalition force in the Operation Desert Storm) will start to rectify our status as they alluded
	to and declared many times. What have we prepared for that? Indeed taking the measures (for the coming war
	against us) will neither cost much nor harm us. If Allah sufficed for the believers in the fighting, to Allah all <u>praise</u>
	is due, and, Allah willing, we expect His reward (in the Hereafter). And if they attack us, then our response will be
	fast and efficient.
	و نتيجة نلك [رفض الاسلام للكهنوت] الطبيعية هي أن الطغيان الفظيع الذي مارسته الكنيسة، و كان أحد أسباب العلمانية، لا وجود له في تاريخ الإسلام.
	ا فالطعيان الديني، ذلك الذي [] و يتيم محاكم التتنيش لتصيدهم [المخالتين]، لا يوجد له – و ش <u>الحمد</u> . نظير في تاريخنا الإسلامي
3	and one of a natural consequences for [the rejection of priesthood by Islam] is that the horrible tyranny of the
	Church, which was one of the reasons of the evolution of secularism, does not exist in Islam. And praise is due to
	Allah that the religious tyranny, that [] held the Inquisition to hunt them [people who have different religious
	views/faiths], did not exist in the history of Islam.

All the concordance lines show that the collocate 'الحمد' ('alḥamdo; praise) is used parenthetically along with the word form 'ش' (*lillah*; to Allah) indicating that Alhawali is describing a satisfactory condition or state of affairs. In Line 1, Alhawali recontextualizes the phrase a satisfactory condition or state of affairs. In Line 1, Alhawali recontextualizes the phrase '('alḥamdo lillah; all praise is due to Allah) to show his satisfaction when two opposing Socialist countries that adhere to the Pan-Arab nationalistic ideologies; namely, Syria and Iraq (see Chapter 1) are involved in a direct deadly combat (during the Gulf Crisis 1990-1991) after they were using the conflicting forces in Lebanon as proxies (in the Lebanese civil war 1975-1990). To Alhawali, such dramatic killing is a grace from 'Allah' and an indication of the fall of the Pan-Arab nationalist movement. He views the Gulf crisis as both bad and good which is a sheer contradiction which may indicate his rejection of both the Allied forces and Arab nationalists.

In Line 2 Alhawali warns against the American-led coalition force in Operation Desert Storm, which, according to him, after the war ends, will turn to practice pressure on 'Us' to rectify '*Our* status'. In response to such a 'threat', Alhawali suggests a type of preparation that, according to him, will empower the country (Saudi Arabia) against any aggressive attempt by the Western Allies (and more prominently the USA) to hinder the Islamization of the Saudi society. The phrase '*Leak*' ('*alḥamdo lillah*; all praise is due to Allah) is preceded by a recontextualization of the following Qur'anic verse:

{وَكَفَى اللَّهُ الْمُؤْمِنِينَ الْقِتَالَ } سورة الاحزاب، اية ٢٥.

{Allah sufficed for the believers in the fighting} Holy Qur'an 33:25.

Thus, Alhawali uses the collocate 'الحمد' ('alḥamd; praise) in a hypothetical situation expecting Allah's help. The last concordance line in Concordance 4.6 shows that Alhawali recontextualized 'الحمد' ('alḥamd; praise) in a negative representation of the 'Other' (the Christian Church), reminding Muslims that in the past the Church used Inquisitions to hunt people who opposed it and had different religious views/faiths.
So far I have delineated the collocates of the key term 'ALLAH' in both corpora. The overall picture of the collocates in Ibn Baz helps to reference a powerful 'God' to whom the pure religion is to be dedicated, along with the idea that Muslims should refer to Qur'an and the Sunnah of His prophet (Sharia) in all disputes. However, Ibn Baz also refers to Allah as part of a strategy to justify a negative representation of Arab nationalism.

Similarly, though on a different scale, Alhawali uses the collocates of the key term 'ALLAH' in different theological and political contexts as well as in negative representations of the 'Other' (see Chapter 6 Section 6.3) to legitimate his stance towards Secularism, Arab nationalism and the American-led coalition force in the Operation Desert Storm (see p. 92). Let us now turn to examine the other semantic foci as indicated by the keywords in Ibn Baz.

4.2.2 Polytheism

Polytheism or associationism in Islam means that worshipping Allah involves the association of other deities with Him. The semantic foci keywords in Ibn Baz that indicate this theme are shown in Table 4.9 below.

Table 4.9 Semantic foci keywords that that indicate 'الشرك' ('ašširk; polytheism) in Ibn Baz

	N	Key word	Eng. trans.	Freq	F. Alhawali
Monotheism and impermissible	1	المترك/ شرك	the polytheism, the associationism/ polytheism, associationism	131	15
religious practices and beliefs	2	البدع	the religiously impermissible innovative practice	63	8

The keyword 'الشرك' ('ašširk; the polytheism) and its variant word form 'شرك' (širk; polytheism) are two forms of one noun that indicates one of the predominant themes in Ibn Baz: (the) polytheism. The first of these two word forms is preceded with the Arabic definite article 'l' ('al; the) which, according to the Arabic writing system, is attached at the beginning of the noun. The /l/ sound in 'l' ('al; the) has undergone a regressive assimilation where this sound is assimilated to the /š/ in 'širk' and in effect the /š/ sound is geminated as indicated by the highlighted letters ('ašširk). Both forms are used in Ibn Baz. The first form

(*'ašširk*; the polytheism) is predominantly used by Ibn Baz (131 times) in all five texts, whereas the word form 'شرك' (*širk*'; polytheism) is less popular, occurring 28 times in 3 texts only (collectively 159). First, I shall consider the keyword 'الشرك' (*'ašširk*'; the polytheism).

Polytheism as the opposite concept of 'التوحيد' (*'attawhīd*; monotheism) is seen by Ibn Baz as a contradiction to the message of Islam that is based on monotheistic beliefs. In order to identify why and how this concept is used by Ibn Baz, I shall first examine its context of usage by considering its collocates. Table 5.10 below shows that the keyword 'الشرك' (*'ašširk*; polytheism) has 5 collocates.

Table 4.10 Collocates of 'الشرك' ('ašširk; polytheism) in Ibn Baz

N	Collocates	Eng. trans.	F.	Joint F.	MI
1	الأكير	the major	27	20	9.527
2	الخرافات	superstitions/ unrealities	19	6	8.297
3	الكنر	infidelity/rejection of Islam	30	5	7.375
4	اليدع	impermissible innovations in religious practices	63	6	6.568
5	بلائم	in/ with/by Allah	173	9	5.696

The first collocate 'الأكبر' (*'a'akbar*; the major) is a classificatory theological term which indicates that Ibn Baz distinguishes between different types of polytheism (see Concordance 4.8 below). The last collocate 'بالله '(*billah*; in/with/by Allah) is also a keyword (see Table 4.3 above). In the Arabic writing system the preposition '--- '(*bi*; with/in) is attached at the beginning of the word 'ساله '(*'allah*; Allah) forming a prepositional phrase ('*billah*; in/with/by Allah)) and is consequently identified by WordSmith as one word. This keyword comes in the prepositional phrase that indicates either faith as in 'الإيصان (*'al'iman billah*; belief/faith in Allah) or polytheism as in 'الشرك بالله '(*'alfunafat*; superstitions) and '*'liabida*'; (*'albida*'; *'impermissible* innovations in religious practices).

Concordance 4.7 Concordance lines of الخرافات ('*alhurafat*; superstitions), 'الكفر' ('*alkufr*: infidelity) and 'البدع' ('*albida*'; impermissible innovations in religious practices) in Ibn Baz

Ν	Concordance
	[] و لكن الشيخ و أنصاره يدعون الناس إلى الحق .[] ينهونهم عن الباطل .[] و يزجر ونهم عنه حتى يتركوه ، وكذلك
	جد في إنكار ا لبدع و الخرافات حتى أز الها الله سبب دعوته []
1	[] but the Sheikh [Mohammad Ibn Abdul Wahhab] and his proponents call people to the right path
1	[] and dissuade them from believing in falsehood, as well as reproach them till they abandon it
	[falsehood]. In addition, he made every effort in disapproving of impermissible religious practices
	and superstitions till Allah eradicated them because of his [Mohammad Ibn Abdul Wahhab] call []
	و من العقائد المصادة للعقيدة الصحيحة في باب الأسماء و الصفات عقائد أهل البدع : من الجهمية ، و المحتزلة ، و من سلك
	سيبِلهم في نفي صفات الله عن و جل []
2	There are also other erroneous beliefs that contradict the true creed in aspect of Allah's names and
-	attributes. These are the beliefs of the people of impermissible religious practices such the Jahmites,
1	the Mu'tazilites and those who followed their way in nullifying the names and attributes of Allah, the
	Majestic and Exalted in Mighty, which resulted in not ascribing perfection to Him, the Glorified []
	و من أبرز هؤلاء الدعاة المصلحين الإمام التبيخ محمد بن عبد الوهاب [] أعادت للإسلام في الجزيرة العربية قوته و صفاءه و
	نفوذه و طهر الله به الجزيرة من الشرك و البدع ، و هناهم به إلى الصبر اط المستقيم.
3	One of the most prominent reformers of Islam is the Imam and the Sheikh Mohammad Ibn Abdul
5	Wahhab []. [His call to monotheism] rendered Islam, in the Arab Peninsula, to be powerful, clear from
	polytheism and victorious. And because of him, Allah has purified the Peninsula from polytheism and
	impermissible religious practices, and guided them [people in the Arab Peninsula] to the straight path.

The lines in Concordance 4.7 show how Ibn Baz positively represents monotheists and negatively represents polytheists. For example, he describes Sheikh Mohammad Ibn Abdul Wahhab, who led a call to monotheism, as a 'prominent reformer of Islam' (line 3). Analysis of a concordance of the keyword 'الشرك' ('*ašširk*; polytheism) shows that it is usually used with three adjectives: 'الشرك' ('*akbar/'al'akbar*; major/the major) which appears as one of the collocates of 'الشرك' ('*ašširk*; polytheism), 'أكبر /الأكبر ' (*'ašģar/'al'aṣġar*; minor/the minor) and '*i aśširk*; polytheism) could be classified into three subcategories; 'الشرك' ('*ašširk*; '*al'akabr*; major polytheism) could be classified into three subcategories; '*itmac'' 'al'akirk' 'al'akabr*; major polytheism), 'الشرك الأصغر ' (*'ašširk 'al'aṣġar*; minor polytheism) and '*itmac'' 'albafiy*; concealed polytheism). Concordance 4.8 below gives three examples of concordance lines of the three different categories of polytheism.

Concordance 4.8 Concordance of 'الشرك' ('ašširk; polytheism) in Ibn Baz

	(ussink, polyticisii) in foir Baz
N	Concordance
1	[] فمن على به [حكم المركد] حق العذاية عرف نواقض الإسلام، وأسباب الردة، وأنواع الكفر و الصلال. و النوع الثاني: الشرك الأصفر، و هو ما ثبت بالنصوص تسميته شركاً، لكنه لم يبلغ درجة الشرك الأكبر [] and whosoever study it [the religious ruling of apostasy] diligently would know all issues that contradict Islam and lead to apostasy, the types of disbelief and error. And the second subcategory is the lesser polytheism as confirmed by the sacred texts and identified as polytheism, but is not as grave as the major polytheism []
2	[] فلين سبحانه في هذه الأية أن عبادة غيره من الأنبياء و الأولياء أو غيرهم. هي الشرك الأكبر. و إن سماها فاعلوها بغير ذلك و قال تعالى: (و الزينَ التُخَدُّوا مِن دونه أولياء [] Allah then clearly delineated in this verse that worshiping others such as prophets and pious men and the like is <u>major polytheism</u> even if those who practice such worship name it otherwise, the Glorified said: {And those who take Auliyaa' (protectors, helpers, lords, gods) other than Allah []
3	[] قالوا بلى يا رسول الله قال (الشرك الحقم يقوم الرجل فيصلي قيزين صلاته لما يرى من نظر رجل إليه) خرجه الإمام أحمد [] [] they said: O, messenger of Allah tell us, he said (the <u>concealed polytheism</u> is when a man stands up to pray then he starts to make his prayer as perfect as possible (showing off) because he sees another man looking at him), this Hadith is narrated by Imam Ahmad []

The concordance analysis of the word 'الشرك' (*'ašširk*; the polytheism/associationism) shows one aspect of Ibn Baz's representation of the 'other'. Ibn Baz considers those who do not align themselves with core concepts of monotheism as deviating from the right path (the notion of social actor representation will be discussed in Chapter 6).

Ibn Baz, also, uses the indefinite noun 'شرك' (*širk*; polytheism), though on a smaller scale, to more or less explicate the meanings of polytheism. Despite the fact that the word (*širk*'; polytheism) does not have any collocates, concordance analysis (Concordance 4.9 below) shows that Ibn Baz continues to expound the concept of polytheism. Concordance 4.9 Concordance of شرك (širk; polytheism/associationism) in Ibn Baz

Ν	Concordance
	[] و غیره من أنواع المبادة، و عبادتهم شرك بالله عز و جل. فالكاهن من يزعم []
1	[] and any other types of worship. And worshipping them is considered <u>associating others</u> with Allah, the Exalted in Mighty and the Majestic, (i.e practicing polytheism). Thus, a priest who claims [to know some of the unseen] []
	[] في هذه الأية أن دعاءهم غيره شرك به عز و جل كما أوضح سبحانه []
2	[] [Allah has elucidated] in this Qur'anic verse that calling onto any deity other than Him (to cure someone or to bring him/her close to Allah) is <u>associating others</u> with Allah, the Exalted in Mighty and the Majestic, (i.e practicing polytheism) as the Glorified expounded []
	[] و الأنتجار و الأصنام و غيرها ، شرك بالله عز و جل ينافي العبادة []
3	[] [calling onto deities other than Allah (to cure someone or to bring him/her close to Allah) such as] trees, idols and other [dead individuals and objects] is <u>polytheism</u> that contradicts (the pure) worship [of Allah] []
4	[] أو ما أتنبه ذلك ، فهذا تسرك خفى ، لكنه شرك أصغر. فاتضبح بهذا أن الشرك []
7	[] or the like, which is concealed polytheism but it is considered as minor polytheism. Thus, it becomes clear that polytheism is []
	[] أنواع تكثه ، و الحقيقة أنه نوعان : شرك أكبر ، وشرك أصغر . فالشرك []
5	[] D68[polytheism] is classified into three subcategories, but, in fact, it is of two main categories: Major <u>polytheism</u> and minor polytheism. Thus the [major] polytheism [is][]
	[] النوعين السابقين : سَرك أكبر ، و شرك أصغر ، و إن سمي خفيا. فالسَرك بكون []
6	[] [is either of] the two previously mentioned subcategories: Major polytheism and minor polytheism, though it is called concealed [polytheism]. Thus polytheism could be []

In his explanation of polytheism, Ibn Baz attempts to outline what contradicts the pure worship that should be dedicated to Allah alone (concordance lines 1, 2 and 3).

As I have illustrated above (see Concordance 4.8), Ibn Baz uses the adjectives 'أكبر ' ('a'kbar; major) 'أصغر' ('asġar; minor) and 'خفي' (*hafiy*; concealed) to denote three subcategories of polytheism, the fifth concordance line indicates that Ibn Baz limits these subcategories to two broad categories: major and minor. This is further explicated by the expanded version of concordance line 6 in Concordance 4.9 which sums up the classificatory scheme of polytheism that Ibn Baz advocates. Let us consider the expanded version of this concordance line in Figure 4.3: Figure 4.3 Expanded version of concordance line 6 in Concordance 4.9

و بما ذكرنا يعلم أن الشرك المفقي لا يخرج عن النوعين السابقين: شرك أكبر، و شرك أصغر، و إن سمي خلوا. فانشرك يكون خلياً و يكون جلياً. فالجلي: دعاء الأسوات و الاستغاثة بالأموات و النذر لهم، و نحو ذلك. و الخفي: ما يكون في قلوب المنافقين يصلون مع الناس، و يصومون مع الناس، و هم في الباطن كفار يعتقدون جواز عبادة الأوثان و الأصنام، و هم على دين المشركين. فهذا هو الشرك الخفي الأكبر [...] الشرك الخفي الأصغر، كانذي يقصد بقراءته تناء الناس، أو بصلاته أو بصدقته

And according to what I have mentioned earlier, polytheism is limited to the two previously mentioned subcategories: Major polytheism and minor polytheism, though it is called concealed [polytheism]. Thus polytheism could either be concealed or observable. An example of the observable (polytheism) might be calling onto dead individuals, beseeching their help or making vows – to do charitable acts devoted to the dead individuals near their graves – (to cure someone or to bring success or to bring him/her close to Allah) and so forth. Concealed polytheism is what resides in the hearts of hypocrites; they pray with Muslims and observe fasting (during Ramadan), but in fact they are infidels as they believe that it is permissible to worship idols or they believe in the religion of polytheists. This is called concealed major polytheism. [...] Concealed minor polytheism happens when someone, for example, recites Qur'an, performs a prayer or gives charity in order that people praise him/her

The above expanded version of concordance line 6 indicates that Ibn Baz differentiates between observable and concealed subcategories of polytheism. He uses the adjective 'خفي' (*hafiy*; concealed) to denote both major and minor types of polytheism: 'Concealed major polytheism' is an equivalent term for hypocrisy. It describes the behaviour of people who are infidels, according to Ibn Baz, whereas the observable major polytheism is the practice of the polytheists of worshipping idols. The lesser degree of polytheism is further classified into subcategories; either concealed or observable minor subcategories of polytheism, which, as Ibn Baz continues to describe, are the insincere performance of Islamic ritual duties (i.e. all the forms of worship should be devoted to Allah alone).

So far I have delineated the perspective of polytheism in Ibn Baz, who has quite a complex understanding of the concept, splitting it into various categories (concealed or observable major and minor categories of polytheism). Notably, all are represented negatively. These negative representations are limited to religious practices within Islam (see Chapter 8 section 8.5 for an explanation of the political views of Ibn Baz). However, Alhawali extends these categories, using different categorical terms to discuss polytheism in Christianity and western societies, as shown below.

4.2.3 Polytheism in Alhawali

The keyword in Ibn Baz 'شرك' ('ašširk; the polytheism) and its indefinite form 'شرك' (širk; polytheism) are also used by Alhawali, though on a smaller scale: 'الشرك' ('ašširk; the polytheism) occurs 15 times in one text and 'شرك' (širk; polytheism) occurs 11 times in two texts as compared to the higher occurrences of these words in Ibn Baz: 131 times in 5 texts and 28 times in 3 texts respectively. Despite the limited number of occurrences of these two terms that refer to polytheism, it is useful to examine how Alhawali constructs the concept of polytheism – does he construct it in the same way as Ibn Baz (just less frequently), or does he employ a different construction?

The words 'الشرك' (*'ašširk*; the polytheism) and its indefinite form 'شرك' (*širk*; polytheism) do not collocate with any other word in Alhawali's corpus, however, below are two Concordances (4.10 and 4.11) that show some selected concordance lines that indicate the use of 'شرك' (*'ašširk*; the polytheism) and its indefinite form 'شرك' (*širk*; polytheism) by Alhawali.

N	Concordance
	[] و من نواقض الإسلام الحسرة - غير ا لشرك الذي هو الناقض الأكبر والذي لا تسك أن العلمانية نوع منه كما سيأتي- ناقصان []
1	[] and in addition to polytheism, which is the greatest invalidator and, no doubt, it includes secularism as I will explain
	later, there are 10 invalidators of Islam, two of them are as follows []
	[] و حسب هذا الميدأ نصب رجال الدين انفسهم انداداً لله تعالى و أوقعوا أنباعهم في الشرك الأكبر (التُخَذُوا أخْبَارَ هُمْ وَرُ هُبَائَهُمْ أَرْبَابًا مَّن نُونِ اللَّهِ
	[]
1	[] and according to this principle, the clergymen set up themselves rivals to Allah, the Exalted, and made their followers
4	commit major polytheism : (They (Jews and Christians) took their rabbis and their monks to be their lords besides Allah
	(by obeying them in things which they made lawful or unlawful according to their own desires without being ordered by
	Allah) []
	[] الخ. و الواقع أن كلا النوعين من الشرك مردهما إلى أصل واحد ، وهو تحكيم غير الله []
3	[] etc But in fact, both types of polytheism have the same root which is arbitration to the man-made laws rather than
	to the Sharia of Allah []
	[] النتائج السيئة للعلمانية على الإنسان: وليست عاقبة هذا الشرك هي الخلود المؤبد في النار فحسب بل إن أتباعه []
4	[] bad [results of secularism] on humans: The final return of this type of polytheism is not only dwelling in Hell-fire
	forever but []
	[] المهم أن النتيجة المنطقية لذلك هي ا لشرك بالله، و هل هذاك صورة من صور []
5	[] the important issue is that the logical end for that is polytheism (associating others with Allah). Is there any other
	manifest form of polytheism other than []
L	

Concordance 4.10 Concordance of 'الشرك' ('ašširk; the polytheism/associationism) in Alhawali

Alhawali's above definition of polytheism goes along with the general Salafi interpretation. However, he extends the concept of polytheism which is the invalidator of Islam to include secularism as indicated by concordance line 1. Thus, secularism is seen as infidelity that not only leads to the destruction of the humans but makes them dwell in hell forever (line 4). Alhawali (in Line 2) quotes a Qur'anic verse (9:31) that delineates the practices of the followers of the rabbis and monks in both Judaism and Christianity: The rabbis and monks make things lawful or unlawful according to their desires that contradict Allah's revelations (the Holy Scriptures). Thus their followers who obey these religious leaders in such matters take them as their Lords besides Allah as the Qur'anic verse quoted by Alhawali tells:

{الَّخَذُوا أَحْبَارَهُمْ وَرُهْبَانَهُمْ أَرْبَابًا مِّن دُونِ اللَّهِ} سورة التوبة ، اية ٣١.

{They (Jews and Christians) took their rabbis and their monks to be their lords besides Allah (by obeying them in things which they made lawful or unlawful according to their own desires without being ordered by Allah)} (Holy Qur'an 9:31).

So while both Ibn Baz and Alhawali view polytheism as bad, Alhawali connects this trait to two other religious groups that he constructs as the Other – Jews and Christians. Thus, for Alhawali, reference to polytheism helps to legitimate his opposition to these other religions. For Ibn Baz, polytheism is only discussed in relation to Muslims. In concordance line 3, Alhawali claims that there are two types of polytheism (described in more detail below in Concordance 4.11) which are the logical consequence (Line 5) of taking the man-made Laws as a means of arbitration rather than the Sharia of Allah.

Alhawali continues to explicate his Salafi views of polytheism using the indefinite noun form 'شرك' (*širk*; polytheism):

Concordance 4.11 Concordance of شرك (širk; polytheism) in Alhawali

N	Concordance
	[] (الإرادة و القصد) و (الطاعة و الآتباع). أما شرك الإرادة و القصد فهو النّوجه []
1	[] (will and intention [to worship other than Allah]) and (obeying [other than Allah] and following
	[other than His Sharia]). Polytheism by will and intention in worship is dedicating []
	[] أعلى تدرك الطاعة و الإتباع - إنه شرك في عبادة الله و أن كان الذين يمارسونه []
2	[] I mean polytheism by obeying [other than Allah] and following [other than His Sharia] is
	associating others in worshipping Allah even if those who practice it []
	[] أمرا و نهياً في العقيدة أو الفروع هو شعرك أكبر بالله تعالى، سواءُ أجاء ذلك []
3	[] altering the Sharia of Allah, ordering people to believe or not in creed matters or in any other
	religious aspects, is major polytheism i.e. associating partners with Allah, the Exalted, whether such
	altering comes in the form of []
	[] يقع حتماً في شرك الطاعة والاتباع و هو شرك أعظم، لتنافيه مع توحيد الألوهية []
4	[] [and whosoever does so] eventually commits polytheism by obeying [other than Allah] and
7	following [other than His Sharia] which is the most evil type of polytheism that contradicts the unity
	of worship []

The classificatory scheme of polytheism in Ibn Baz (major and minor whether concealed or observable) indicates pure Salafi religious perspective (Abdul-Wahhab, 1996). However, Alhawali, from a different Salafi perspective, uses other terms to classify polytheism. Line 1 in Concordance 4.11 reveals that Alhawali differentiates between two types of polytheism: Polytheism of 'الإرادة و القصد' ('al'iradah wa 'alqaşd; 'will and intention' [to worship other than Allah]) and 'ladlast e ligitation ('ita'ah wa 'alittiba'; obeying [other than Allah] and following [other than His Sharia]). In such a classificatory scheme, Alhawali rephrases the previously mentioned classification of major polytheism in Ibn Baz to broaden the scope of polytheism to encompass any possible human activity. Let us consider the expanded version of concordance line 1 that indicates Alhawali's classificatory scheme:

Figure 4.4 Expanded version of concordance line 1 in Concordance 4.11

[…] و على هذا تستطيع القول بأن الشرك ـ نتب البشرية الأكبر و مدار الصراع بين الأمم و الرسل ـ هو عيدة الطاغوث مع الله أو من دونه في أمرين ستلازمين: (الإرادة و القصد) و (الطاعة و الاتباع).

اً أَمَا شَرِّكَ الإِرْادَةُ وَ النصد فَهُو النَّوجَةِ إلى غير الله تعالى ينسىء من تنعاش النجد كالصلاة و النزابين و النتور و الدعاء و الاستغاثة تبعاً للسناجة الجاهلية القائلة: ((مَا تَعْبُدُهُمُ إِلا يُبُكُرُونَا إلى اللهُ زُلْقَى)) [الزمر: 3] و طاغوت هذا النوع هو الصنم أو الوتن أو الجني أو الطوطي.. إلخ إذا تتخذ من النائية من الاستحداث إلى من النائية عام النوع هو الصنم أو الوتن أو الجني أو الطوطي.. إلخ

(رما تحديم : بعريون بني العارسي)) (الرامن. وإو عاصوت هو الموع مو العنم أو الومن أو العبي والسومعد... بعن و أما تبرك الطاعة و الاتياع فهو التمرد على تبرع الله تعالى و حتى تحكيمه في تبلون الحياة بعضها أو كلها، و هو منزق الطريق بين الإسلام و الجاهلية، كما أنه السمة المتنزكة بين الجاهلية كلها على هدار التثريغ، و به استحقت أن تسمى جاهلية مهما بلغ تشتها في المحسنرة و المعرفة: ((أفحكمَ الْجَاهَيَّةُ بَنْقُونَ)) [المتكةر50] ((أم ليم تبركة تنزعُوا ليم من الثين ما لمَّ بدنَ به الشرع) [التمورى:21] و طاغوت هذا التوع هو الزعماء و الكهان و المعرفة: ((أفحكمَ الْجَاهَيَّةُ بَنْقُونَ)) و التقليد و الأعراف و التوانين و النستين و الأمواء... إلغ.

And consequently, we can say that polytheism – the humans' gravest sin and the cause of the conflict between the messengers of Allah and their nations – is the worship of 'طاغرت' (*tağuut*, false deities / false leaders / false judges) either along with Allah or without Him in two related issues: (will and intention) and (obeying and following).

Polytheism by 'will and intention' is dedicating some of the rites of worship to other than Allah, the Exalted, such as slaughtering animals, making vows (to do charitable acts) supplication and beseeching help, which according to the naivety of the ignorant nations (in the previous eras) who say: (We worship them only that they may bring us near to Allah) (39:3). The أطلاف (taguut; false deities / false leaders / false judges) of this type of polytheism could be idols, the Jinni or the totem ... etc.

Polytheism by 'obeying and following' (obeying other than Allah and following other than His Sharia) is the rejection of the Sharia of Allah, the Exalted, or even the rejection of the supremacy of it over some or all aspects of life. Thus, polytheism is the marking difference between Islam and ignorance and the distinguishing feature of all ignorant nations throughout history. It is the cause that such nations deserved to be called ignorant no matter how civilized and knowledgeable they are, [Allah said]: (Do they then seek the judgement of (the days of) Ignorance?) (5:50) and (Or have they partners with Allah (false gods) who have instituted for them a religion which Allah has not ordained?) (42:21). The discut false deities / false leaders / false judges) of this type of polytheism could beleaders, priests, great men of a society, systems, traditions, costums, man-made laws, constitutions and desires... etc.

After quoting Ibn Qayyim (1292-1350) and Mohammad Ibn Abdul Wahhab (1703-1792), two Salafi leaders, Alhawali defines 'polytheism' as the worship of 'degut; false deities, false leaders or false judges) which, according to him, is the greatest invalidator of Islam and caused by arbitration to the man-made laws rather than the Sharia of Allah.

Polytheism by 'will and intention' means holding a firm will and intention to worship deities other than Allah which is seen as infidelity (a major polytheism according to Ibn Baz). This is a purely religious concept that both writers (Ibn Baz and Alhawali) condemn. However, Alhawali's second type of polytheism, i.e. polytheism by 'obeying and following', which could also be considered a major type of polytheism as described by Alhawali himself, is, nevertheless, widening the scope of polytheism to include issues other than religious rituals.

The word 'طاغوت' (tagūt; false deities/false leaders/false judges) is a key concept in the expanded concordance line 1 (Figure 4.4). Alhawali claims that each type of polytheism has a different 'الطاغوت' (taġūt) that polytheists worship. The 'طاغوت' (taġūt) of the first type of polytheism, Alhawali continues to explicate, is idols, the Jinni, the totem etc..., which are deities of polytheism in religions other than Islam. The word 'totem indicates' Alhawali's inclusion of Totemism (Ancestors and animal worship) as a non-Salafi definition of polytheism. The first two types of the 'طاغوت' (taġūt), viz., idols and Jinni, are also mentioned in the Qur'an:

{وَإِذْ قَالَ إِبْرَاهِيمُ رَبِّ اجْعَلْ هَاذَا الْبَلَدَ آمِنًا وَاجْنَبْنِي وَبَنِيَّ أَن نَّعْبُدَ الْأَصْنَامَ} سورة ابراهيم اية ٤٥.

{And (remember) when Ibrahim (Abraham) said: "O my Lord! Make this city (Makkah) one of peace and security, and keep me and my sons away from worshipping (idols)} (14:45)

{قَالُوا سُبْحَانَكَ أَنتَ وَلِيُّنَا مِن دُونِهِم بَلْ كَانُوا يَعْبُدُونَ الْجِنَّ أَكْثَرُهُم بَهِم مُؤْمِنُونَ} سورة سبأ اية ٤١.

{They (the angels) will say: "Glorified be You! You are our Wali (Lord) instead of them. Nay, but they used to worship the jinn⁴⁴; most of them were believers in them."} (34:41).

Indeed, Alhawali's explication of the 'طاغوت' ($ta\dot{g}ut$) of polytheism by 'will and intention' is not incongruent with the Salafi tradition of examining major polytheism as all these are forms of dedicating rituals to other than Allah.

The 'طاغوت' (taġūt) of polytheism by 'obeying [other than Allah] and following [other than His Sharia] is a religious concept that is politically oriented to include forms of any dominating powers in any society; leaders, priests, great men of a society, systems, traditions, costumes, man-made laws, constitutions and desires. Thus, the concept 'طاغرت' (taġūt) is recontextualized by Alhawali to include all forms of dominating powers in a society, that could lead to encompass every aspect of social life under the umbrella of the Sharia, creating a rift between what Alhawali considers as 'ignorance' and Islam, i.e. Islamization of every social aspect of human life. It might be sarcastic to say that Alhawali wants his readers to obey and follow him in all his teachings which may make him included under what he labelled as 'طاغوت'. (taġūt).

^{36.} The Arabic noun 'Jinn' is the plural of 'Jinni'. This is similar to the English 'genie'.

Both Ibn Baz and Alhawali represent polytheism negatively. In addition, they, from different Salafi perspectives, recontextualize it utilizing different arguments. Ibn Baz's representation and classification of polytheism are oriented towards the rituals and aspects of worship that contradicts Islamic monotheistic beliefs; refocusing the Salafi explications of the types of polytheism. Ibn Baz's projection of the monotheistic beliefs does not open a dialogue with other Saudis (Sufis and Shiites) who have different views concerning polytheistic practices in Islam, which could result in suppression to such minorities. Alhawali recontextualizes polytheism to include secularism, also linked to Judaism and Christianity. He represents ' $d_1d_2u_1$ ' to include all dominating powers in a society i.e. following other than the religiously guided Sheikhs which might lead to the imposition of Salafi political interpretations of Islam on the society.

As indicated above, whenever 'polytheism' is mentioned, it could also indicate the opposite concept 'monotheism' in the sense that negating the former affirms the latter. The second semantic focus in Ibn Baz that I want to look at is the concept of monotheism.

4.2.4 Monotheism

As indicated in Chapter 1, the concept of Islamic monotheism consists of three subcategories or principles; first, the unity of Lordship which refers to the belief that Allah is the Creator and the Sustainer of the world; second, unity of worship indicates that all Muslims should see Allah as the only one to worship, and third, unifying Allah in His names and attributes which indicates that Muslims should call on Allah by His names and attributes as they appear in the Qur'an without changing their meanings (Abdul-Wahhab, 1996).

The first of these three subcategories of monotheism is not always a problematic issue in Islam as all Muslims agree that Allah is the Creator and the Sustainer of the world. However the other two types of monotheism have caused divisions in Islam. Part of the concept of the unity of worship in Islam is the concept of ' c_{2} ' (do ' \bar{a} '; supplication). In this respect, Sufis and Shiites visit graves of pious men to make supplications and seek their help to intercede with Allah. This has caused fissures and conflicts between the Salafis and their opponents; the Shiites and Sufis.

The last of the three monotheistic categories i.e. unifying Allah in His names and attributes, has also caused conflicts among the different medieval Sunni schools of theology such as 'المعترلة' ('almu'tazilah; the Mu'tazilites) and their descendants; 'المعترلة' ('al'aša'irah; Ash'arites) and 'الماتريدية' ('almaturidiyyah; Maturidites) who sought to understand the principles of المعتردة الإسلامية' ('al'aāqīdah 'al'islamiyyah; Islamic creeds) through a dialectical method: for example, they use arguments rather than literal interpretations of sacred texts to understand the names and attributes of Allah. It is noteworthy here that these differences between the Salafi and Ash'arites' conceptions in the names and attributes are not taught to ordinary people. However, they occasionally appear in the religious writings of both Sunni factions (Ash'arites and Salafis).

The major conflict between the Salafis and the previously mentioned theological schools occurs in the monotheistic category of 'Allah's names and attributes' where these schools reject the Salafi literal interpretations of Allah's names and attributes (the Salafi concepts of names and attributes of Allah are detailed in section 4.2.3.2 below). The following sub sections address these two thematic foci in Ibn Baz.

4.2.4.1 Unity of worship

A keyword that relates to the concept of unity of worship in Ibn Baz is 'الاستغاثة' ('al'istigatah; beseeching for help) (28 occurrences). Let us consider the concordance lines of the keyword 'الاستغاثة' ('al'istigatah; beseeching for help) in Concordance 4.12.

Concordance 4.12 Concordance of 'الاستغاثة' ('al'istigatah; beseeching for help) in Ibn Baz

N	Concordance
	[] و السجود و الطواف والذبح و النذر و الخوف و الرجاء و الاستغاثة و الاستعانة والاستعادة و سائر أنواع الدعاء، و []
1	[] and prostration, circumambulation, sacrifice (slaughtering an animal for the sake of Allah), making pledges (to make
	charitable acts for the sake of Allah), fear (of Allah), hope (for His mercy), beseeching for (His) help, asking for (His)
	help and guidance, seeking (His) protection and all other sorts of supplication, and []
	[] كما يفعل المشركون في كل زمان مع اصدامهم و أوتانهم، في التعظيم و الدعاء و الإستغاثة ، و التمسح و التبرك وطلب المدد.
2	[] and as the polytheists do in all eras for their idols: adoring them, calling onto them, beseeching for (their) help and
	seeking their blessing and support.
	خلق [الله] التقلين ليعبدوه، دون كل ما سواه ، و ليخصوه بالدعاء و الاستغالة ، و الذبح و النذر وسائر العبادات، وقد بعث []
3	[Allah] has created mankind and jinn to worship Him alone, and to single Him out in supplication, beseeching for help
	(from Him alone), sacrifice (slaughtering animals for His sake), making pledges (to make charitable acts for the sake of
	Allah) and all other worships, and [He] has sent []
	[والعبادة هي طاعته و طاعة] رسوله صلى الله عليه وسلم كما سلف ، و يدخل قيها الدعاء و الاستغاثة ، و الخوف ، و الرجاء والذبح والنذر كما
	يدخل فيها []
4	[and worship means obeying Him and] His messenger, may Allah's peace and blessings be upon him, as I indicated
	earlier. Worship includes supplication, beseeching for help (from Allah), fear (of Allah), hope (for the mercy of Allah),
	sacrifice (slaughtering an animal for the sake of Allah) and making pledges (to make charitable acts for the sake of
	[] الملاذ من كل تس وهذا كله خطأ و ضلال و تسرك أكبر فإن الدعاء و الاستغاثة و الاستجارة و جميع العبادات يجب إخلاصها لله []
5	[] seeking refuge [with the Prophet] of every evil which is wrong and manifest error and major polytheism. Indeed
	supplication, beseeching for help, seeking refuge and all other forms of worship should be dedicated to Allah alone []

In all the concordance lines in Concordance 4.12 above, the keyword '(al'istigatah;beseeching for help) is constructed as a form of worship which, according to Ibn Baz, is supposed to be dedicated to Allah alone as indicated ostensibly in concordance lines 3, 4 and 5. Concordance line 1 reveals that the keyword '(al'istigatah; beseeching for help) is one sort of '(al'a; supplication) which, according to Ibn Baz, is '(al'a; (beseeching for help) is one sort of '(al'a; supplication) which, according to Ibn Baz, is '(al'a; (beseeching for help) worship). Thus, beseeching for help from pious dead individuals including Prophets, either by calling onto them or going to their graves and beseech their help, is considered by Ibn Baz to be a violation of monotheism and precisely a sheer contradiction of the unity of worship; 'manifest error and major polytheism' (concordance line 5 in Concordance 4.12).

The thematic focus discussed in this section is linked to the concept of polytheism presented in section 4.2.2 above in the sense that confirming monotheistic beliefs indicates the rejection of polytheism. So far I have delineated the Salafi concept of polytheism as opposed to monotheism, in the following section I focus on the third category of monotheism which is unifying Allah in His names and attributes.

4.2.4.2 Unifying Allah in His names and attributes

The Arabic word for monotheism does not exist as a keyword in Ibn Baz's corpus, nor is it particularly frequent in either corpus⁴⁵; however, there are a number of other keywords including 'صفاته' (*sifātoho*; His attributes), which when examined in detail, are connected to monotheism. Concordance 4.13 below shows 3 selected lines of the keyword 'صفاته' (*sifātoho*; His attributes).

Concordance 4.13 Concordance of 'صفاته' (sifātoho; His attributes) in Ibn Baz's corpus

N Concordance و من الإيمان بالله أيضا : الإيمان بأسمائه الحسني و صفاته الطي الواردة في كتابه العزيز ، و التابتة عن رسوله الأمين من غير تحريف و لا تعطيل و لا تكييف و لا تمتيل ، بل يجب أن تمر كما جاءت بلا كيف ، مع الإيمان بما دلت عليه من المعاني المظيمة التي هي أوصداف لله - عل و جل- يجب و صفه بها على الوجه اللائق به من غير أن يشابه خلقه في شيء من صفائه ، كما قال تعالى : - (ليس كمثله شيء و. هو السميع البصير.) . and part of the faith in Allah as well is the belief in His excellent names and His perfect attributes that are mentioned in His glorious 1 Book and reported in the Sunnah of His honest Prophet without altering, nullifying or comparing His attributes to human characteristics. It is mandatory [on Muslims] to believe in them as they were reported [in the sacred texts] without questioning, and they should believe in their [names and attributes] great meanings which describe Allah - the Exalted in Mighty and the Majestic-m and He should be described with these attributes in a manner that suits Him, where He does not resemble any humans in His attributes as the Exalted said: {There is nothing like Him; and He is the All-Hearer, the All-Seer} [...] و الواجب على المسلم أن يسلك في هذه الآيات و ما في معناها من الأحاديث المسحيحة الدالة على أسماء الله و صفاته مسلك أهل السنة و الجماعة و هو الإيمان بها ، و اعتقاد صحة ما دلت عليه و إتباته له سبحانه على الرجه اللائق به من غير تحريف و لا تعطيل و لا تكبيف و لا تمثيل ، و هذا هو المسلك الصحيح الذي سلكه انسلف المسالح و اتفقوا عليه 2 [...] and a Muslim should follow the path of the people of Sunnah [Prophetic Traditions] and the Community [consensus of the majority of Muslims] who believe that these Qur'anic verse and sound Hadiths that indicate the names and attributes of Allah and confirm that they [names and attributes] are for Him, the Glorified, in a manner that suits Him without altering, nullifying or comparing His attributes to human characteristics. And this is the correct belief that the pious Salaf followed and agreed upon و النوع الثاني : توحيد الأسماء و الصفات ، و هو أيضا من جنس توحيد الريوبية ، قد أقروا به و عرقود,و توحيد الريوبية يستلزمه ؛ لأن من كان هو الخلاق الرزاق و المالك لكل شيء ، فهو المستحق لجميع الأسماء الحسلي و الصفات العلى ، و هو الكامل في ذاته و أسمائه و صفاته و أفعاله ، لا شريك له ، و لا شبيه له ، و لا تدركه الأبصار و هو السميع العليم ، كما قال سيحانه: {لَيْسَ كُمِتَّلِهِ سَيَّةً و هو السَّمِيمُ الْبَصِيرَ } and the second type [of monotheism] is unifying Allah in His names and attributes which is a sub-category of the unity of Lordship that They [Arab atheists] acknowledged and recognized. The unity of Lordship entails it [unifying Allah in His names and attributes] because 3 whoever becomes the Creator, the All-Provider and the Possessor of everything, is the one who deserve all the excellent names and perfect attributes. So, He is absolutely perfect in Himself, names and attributes, and His actions. He has no partner, and there is nothing like Him. No vision can grasp Him and He is the All-Hearer, the All-Knower, as the Glorified said: {There is nothing like Him; and He is the All-Hearer, the All-Seer}

Concordance 4.13 shows three concordance lines via which Ibn Baz expounds on the Attributes of Allah which is a sub-category of monotheism in Islam. This sub-category of monotheism is based on the conviction that Allah has the excellent and perfect names and attributes where Muslims are ordered by Allah to call on Him by His names as indicated by the following Qur'anic verse taken from the same text that contains concordance line 2:

{وَلَهُ الأَسْمَاءِ الْحُسْنَى فَادْعُوهُ بِمَا وَذَرُواْ الَّذِينَ يُلْحِدُونَ فِي أَسْمَآئِهِ}

^{37.} The word 'monotheism' ('attawheed; التوحيد) occurs 26 times in ibn Baz and 7 times in Alhawali.

{And (all) the Most Beautiful Names belong to Allah, so call on Him by them, and leave the company of those who belie or deny (or utter impious speech against) His Names} (7:180).⁴⁶

In Lines 1 and 3, Ibn Baz, repeats the statement in the Qur'an that Allah has the excellent names and perfect attributes and none can excel Him. Line 2 indicates that the monotheistic belief in Allah's names and attributes requires 'أبلات' (*'itbāt'*; the confirmation) of Allah's names and attributes without 'تحريف' (*taḥrīf*; altering) their expression or meaning or '*ta 'tīl*; nullifying them) i.e. denying all or some of them. He also states that Muslims should neither compare those names and attributes to any human characteristics by assigning to Allah the attributes that belong to His creatures, '*itakyyīf*; framing).

So far I have delineated two semantic foci (polytheism and monotheism) in Ibn Baz that are reflected and indicated by the keyword 'polytheism' ('ašširk; (الشرك). These two themes constitute an integral part of the Salafi creed which were recontextualized by Ibn Baz as they were mentioned by many Salafi leaders including Ibn Taymiyya⁴⁷ (2004) who wrote a large amount of literature on these issues refuting 'المعتزلة' ('almu'tazilah; the Mu'tazilites) and their descendants; 'الأشاعرة' ('al'aša'irah; Ash'arites) and 'المعتزية' ('almaturidiyyah; Maturidites). Ibn Baz, on the one hand, appears to have refocused attention on the Salafi concepts of the names and attributes of Allah. Ibn Baz's major concern is to refute the polytheists' practices of using intercessors with Allah in their worship. The practice of the Shiites who take the twelve Imams and the Sufis who take Auliya (friends of Allah according to the Sufi's interpretations of Auliya) as intercessors with Allah are refuted by Ibn Baz. On the other hand, Alhawali's recontextualization of polytheism mainly includes reference to secularism and obeying those who are in power. I now turn to sketch out how the two writers represent Islam.

^{46. (7:180)} indicates that this Qur'anic verse is number 180 in Chapter 7.

^{47.} Ibn Taymiyya, Ahmad (1263-1328) is a medieval Salafi jurist and theologian (Lacroix, 2009) who adhered to the Hanbali School of jurisprudence.

4.2.5 Islam

Words which refer directly to Islam are used by both writers, though each from a different Salafi perspective. The keyword 'الإسلام' ('al'Islam; Islam) in Ibn Baz (337 occurrences) is also considerably used by Alhawali (271 occurrences). Both writers use this word in all five texts in each corpus. However, Alhawali uses other words that belong to the lemma 'إسلام' ('islām; Islam): 'بالإسلام' (bil'islām; with Islam), 'إلاسلامي)' ('islām; Islam): 'بالإسلام' ('al'islāmi: the Islamic -sing. masculine) and 'بالإسلام' ('al'islāmyyah; the Islamic -sing. fem.). Table 4.11 below shows the collocates of the key concept 'ISLAM' and other related words in both corpora. To reiterate from the start of this Chapter, only collocates that occur in 3 texts and more will be included in the analysis and if the keyword has other forms, only one collocate for each word form is included in the analysis.

Ν	Keywor	d Eng. trans	F.	Collocates	Eng. trans	F.	Joint	MI	\square
1		Islam		این	the son of	116	17	5.539	н
2	الإمبلام		337	نيّمية	Taymyyia	19	16	8.062	Ibn E
3	1 2			رحمه	may (Allah) have mercy on him	88	8	4.850	Baz
4	بالإسلام	with Islam	16	ان	indeed, verily	1605	5	4.380	
5	الإسلام	Islam	271	كما	as	583	11	3.770	Alha
6	الإنبلامي	Islamic (sing. Masc.)	178	العالم	World	305	76	8.100	Alhawali
7	الإمىلامية	Islamic (sing. Fem.)	210	الأمة	nation	131	33	7.877	-

Table 4.11 Collocates of the keyword 'الإسلام' ('a'islām; Islam) in Ibn Baz and Alhawali

The collocates of 'الإسلام' ('al'Islam; Islam) in Ibn Baz are all limited to the nomination of the Sheikh of Islam Ibn Taymiyya, a well-known medieval Salafi scholar, which could indicate a link between Ibn Baz and Ibn Taymiyya (analysis of social actors will be carried out in Chapter 6). The collocate 'رحمه' (raḥimahu; may (Allah) have mercy on him) also indicates the positive use of a parenthetical (optative) phrase that is usually used after the nomination of people who have died. This collocate always occurs as part of the above fixed phrase. After generating a concordance of the collocate 'رحمه' (raḥimahu; may (Allah) have mercy on him), all the collocates of Islam in Ibn Baz appear in one phrase as in Concordance 4.14 below; viz., نشيخ الإسلام ابن تيمية رحمه الله ' (šayh 'al'Islam 'ibn Taymiyya

rahimahu allah; the Sheikh of Islam Ibn Taymiyya, may Allah have mercy on him):

N	Concordance
	[] ما أوضحه سَيخ الإسلام ابن تيمية رحمه الله في تفسير الاستواء بالعلو فوق العرش []
1	[] as the Sheikh of <u>Islam Ibn Taymiyya <i>may (Allah) have mercy on him</i>, explicated "استواء" (istwaa';) as</u>
	'He rose over the Throne (really in a manner that suits His Majesty)' []
	[] ما ذكره شيخ الإسلام ابن تيمية ر <u>حمه</u> الله إذ يقول : العبادة : اسم جامع لكل ما يحبه الله و يرضاه من الأقوال و الأعمال الظاهرة و
	الْبِنْطْنَةَ ، و هذا يدل على أن العبادة تقتضي : - الانقياد المتام لله تحالى ، أمرأ و نهيأ []
2	[] what the Sheikh of Islam Ibn Taymiyya may (Allah) have mercy on him, mentioned that 'worship is a
2	comprehensive term that comprises all that Allah loves and pleases Him including apparent and hidden sayings
	and deeds'. This indicates that 'worship requires the complete submission to Allah in obeying His orders,
	avoiding what He has prohibited []
	[] و لا ريب أن الدعوة إلى القومية العربية من أمر الجاهلية ، لأنها دعوة إلى غير الإسلام [] و عقبتها تمزيق الشمل و غرس العداوة و
	السَحْناء في القلوب ، و التفريق بين القبائل و السّعوب قال سَيخ الإسلام ابن تيمية رحمه الله : (كل ما خرج عن دعوى من نسب أو بلد أو
	جنس [] ، فهو من عزاء الجاهلية []
2	[] and no doubt that calling to Arab nationalism is as sort of jahiliyyah (ignorance of the pre-Islamic days).,
2	because it is a call to other than Islam [] which consequently results in dividing [the Muslim nation]
	encouraging hostility [] and discrimination among tribes and nations. The Sheikh of Islam Ibn Taymiyya
	may (Allah) have mercy on him, said: Any call to other than (Islam) such as the call to familialism, racism,
	patriotism [] is a form of <i>jahiliyyah</i> (ignorance of the pre-Islamic days) []

Concordance 4.14 Concordance of the collocate 'رحمه' (rahimahu; may (Allah) have mercy on him) in Ibn Baz

All the lines in Concordance 4.14 show that Ibn Baz is referring to Ibn Taymiyya's works. In Line 1, Ibn Baz quotes Ibn Taymiyya as explaining the Salafi, literal concept of the Qur'anic word 'استوى' (*'istawa*; 'He (Allah) rose over (the Throne really in a manner that suits His Majesty). This is a Salafi stance that is opposed by the Ash'arites' interpretation of 'استوى' (*'istawa*) which is '(Allah) established Himself firmly', and consequently they believe that Allah exists everywhere (for more explanation on this disagreement between the Salafis and Ash'arites see Chapter 8 section 8.3.2).

Line 2 indicates that Ibn Baz recontextualizes Ibn Taymiyya's Salafi definition of 'worship' in Islam. Additionally, Ibn Baz recontextualizes Ibn Taymiyya's views concerning the call to familialism, racism and patriotism, in order to refute Arab nationalism (Line 3).

Despite the fact that the keyword 'الإسلام' ('al'Islam; Islam), in Ibn Baz, collocates only with the nomination of Ibn Taymiyya, a look at some representative lines (Concordance 4.15) reveals further aspects of the actual use of this keyword.

Concordance 4.15 Concordance of 'الإسلام' ('al'Islam; Islam) in Ibn Baz

1.7	
N	Concordance
	[] و تعانيمه، فأواجب على أبناء الإسلام بدلا من التحمس للتومية والمناصرة لدعاتها : أن يكرسوا جهدهم []
1	[] and its teachings, and instead of enthusiastically advocating pan Arab nationalism and supporting its advocates, the
L	sons of Islam should devote their efforts to []
	[] مما وقع فيه الكثيرون من المنتسبين إلى الإسلام من الغلو في الأنبياء و المساحين و البناء على قبور هم و اتخاذ المساجد []
2	[] the practices of those who claim Islam such as glorifying prophets and saints and construct building on their graves
L	taking them as places of worship (mosques) []
	[] و أن تشرحم عليهم ، و أن تعرف أنهم أئمة الإسلام و دعاة الهدى ، و لكن لا يحملك ذلك على التعصب و التثليد الأعمى []
3	[] and you should call onto Allah to have mercy on them, you should also know that they are the Imams of Islam who
	guided people to the right path. But this should not lead you to fanaticism and blindly imitate them []
	[الأيمان] و بالتدر خبره و شره على طريقة أئمة الإسلام في توحيد الله ، و إخلاص العبادة له جل و علا ، و في الإيمان بأسماء الله و صفاته []
4	[believe] in the divine will and decree of Allah (good and bad) following the monotheistic steps of the Imams of Islam, and
	dedicating the worship to Allah, the Majestic and the Most High, alone, and believe in His names and attributes []
	[…] أن الاحتفال بالموالد أيس من دين ا لإسلام ، بل هو من البدع المحدثات التي امرن الله و رسوله بتركها و الحذر منها […]
5	[] indeed celebrating the birthday (of the Prophet of Islam) is not part of the obligatory duties of Islam. It is an
	innovation of religious practices that Allah and His messenger ordered us to abandon and avoid []
	[] و ابي حنيفة النعمان و غير هم و من بعدهم من أئمة الهدى ، كمتيخ ا لإسلام ابن تيمية ، و العلامة ابن القيم و الحافظ ابن كثير []
6	[] and Abu Hanifah Alnuman and others, and their followers the Imams who guided people to the true path of Islam such
	as the Sheikh of Islam Ibn Taymyyia, the scholar Ibn Alqayyim, and the Hafidh Ibn Kathir []
	[] ترك الإسراك كله: صغيره وكبيره فالشرك الأكبر ينافى التوحيد ، و ينافى ا لإسلام كلياً. و الشرك الأصغر ينافى كماله الواجب فلا بد من ترك []
7	[] abandoning all forms of (major and minor) polytheism. Because major polytheism negates monotheistic beliefs and
	Islam, and minor polytheism negates the perfection of monotheism. Thus we should avoid []

Ibn Baz represents Islam as a family (Line 1) whose sons are instructed to reject pan Arab nationalism and support what Ibn Baz believes to be the cause of Islam, following the guidance of the Salafi Imams (Lines 3, 4 and 6). Among others, Ibn Baz positively nominates the following Salafi Imams (Line 6): Abu Hanifah (699-767), the founder of the Hanafi School of Islamic jurisprudence, the Sheikh of Islam, Ibn Taymyyia (1263-1328), Ibn Alqayyim (1292-1350), and Alhafidh Ibn Kathir (1301-1373). Nomination of such medieval Imams is a theme that reflects Ibn Baz's adherence to the Salafi monotheistic beliefs, viz., avoiding polytheism (Line 7). In addition, Ibn Baz states that Muslims should believe in Allah's names and attributes (Line 4) following those Salafi medieval Imams. This supports the previously delineated theme of monotheism in section 4.2.4 above.

As regards Alhawali's construction of Islam, he uses other words that relate to the lemma 'إسلام' (*'islam*; Islam) considerably: 'الإسلام' (*'al'islam*; Islam – 271 occurrences), (*'ili'islam*; with Islam – 16 occurrences), 'الإسلامي' (*'alislami*: the Islamic –adj. sing. masc. - 178 occurrences) and 'الإسلامية' (*'al'islamyyiah*: the Islamic –adj. sing. fem. - 210 occurrences). Table 4.11 also shows these words and their collocates in Alhawali. The collocates of the words 'الإسلام' ('al'islam; Islam) and 'بالإسلام' (bil'islam; with Islam) are grammatical words that are classified in this thesis as argumentation indicators which may not reveal any semantic foci (argumentation is discussed in Chapter 7). The collocates of the other two adjectives (that stem from the same lemma: 'اسلام') ('islam; Islam)) are nouns that indicate the Islamic world and the identity of Muslim nation as opposed to the non-Islamic world or non-Muslims, the West for example; thus constructing two opposing worlds. This could be further investigated using concordance-based analysis of the word 'لإسلام'

Concordance 4.16 Concordance of 'الإسلام' ('al'islam; Islam) in Alhawali

[المبشرون و المست المفاجئة، و وجد الس تتوم على إبادة الإسا e military
ل كترم على إبادة الإسا
e military
sudden uprising
nking; thinkers]
te <u>Islam</u> and
ر افَنَنی هذین [محه
ر النبي منين رامت افصل الدين عن الس
vriters
Syrian author],
5
هل من الإستعانة أز
lam (Saudi
[المستشرقون أدمغا
يديرون الصواع الم
ral institutions
mage the
حیت ان حیازه کتب روب
اب∆فرار محمد مئیست
Faymyyia as
ما بزك أنها كانت .
11 to use the
n to use the
بل تقوم بين حضار
tianity and

Islam is constructed in Alhawali as the victim of external threats (particularly a Judeo-Christian attempt to eradicate Islam) in Lines 1, 3, 4 and 6. In addition to external threats, Alhawali reports other examples of 'internal threats': some suspected writers and journalists, who allegedly claim adherence to Islam, call for the separation between religion and politics (Line 2) as well as abolition of classical Arabic. They advocate using Arabic slang instead (Line 6). Thus, according to him, the religious enmity between Christianity and Islam is continuous (Line 7).⁴⁸ The word 'الحوار' ('alḥiwār; the dialogue) is followed by the bracketed word 'الصراع' ('alṣirā'; the conflict). Alhawali thus draws the attention of his audience to what he considers to be an inevitable confrontation between the West and the 'Muslim world' (see Sections 5.3 and 5.4 in Chapter 5).

Alhawali refers to the 1991 war against Iraq (Line 3) drawing the attention of his audience to what he believes is the major purpose of the American troops that came to 'the land of Islam', Saudi Arabia: their purpose is not to fight back the Iraqi troops and free Kuwait, but, according to Alhawali, it is to protect the 'Jewish State' (Israel).' Although as indicated in Line 5, Alhawali opposes the Iraqi regime, confirming that they made it unlawful for their citizens to obtain the Salafi books of both Ibn Abdul Wahhab and Ibn Taymyyia, he abhors the idea that King Fahad (the fourth king of Saudi Arabia between 1982-2005) sought the help of the American troops against Iraq (the ambivalent stances of Alhawali towards Iraq is further investigated in Chapter 5, Section 5.3). This description of Saudi Arabia as the land of Islam having non-Muslim military troops whose aim is to protect the Jewish State (Israel) could be seen as an attempt to instigate his audience against both the US troops and the Saudi leaders, utilizing the enmity between the Palestinians and Israel.

^{48.} Taking the co-text into our consideration, we find Alhawali is quoting a report by the senior advisor to the American President Johnson in 1964 (see Section 5.2.1 in Chapter 5).

This more politically-oriented construction of Islam as under attack, from both without and within, is different from that of Ibn Baz whose construction of Islam revolves more around monotheistic beliefs and practices.

A final theme emerging from the keywords list of Ibn Baz is the 'Hereafter'. Both authors write about this theme and, in what follows, I shall describe how it is used in Ibn Baz and Alhawali.

4.2.6 The final return

The last semantic focus revealed by the keyword list in Ibn Baz is the 'final return'. Table 4.12 shows 2 keywords that indicate this theme in Ibn Baz along with their frequencies in both corpora.

Table 4.12 Keywords that indicate the 'final return' in Ibn Baz

	N	Key word	Eng. trans.	Freq	F. Alhawali
The final	1	الأخرة	the Hereafter	63	28
return	2	الذار	the Hell-fire	58	17

Both writers use the same set of words to indicate the concept of life after death. However, Alhawali uses them relatively less often.

For Ibn Baz, the keyword 'الأخرة' (*'al'āḥirah*; the Hereafter) collocates with the antonym (*'addoniya*; this world) (MI score: 9.790, joint frequency: 48). Using such an antonym may indicate Ibn Baz's representation of two opposing worlds; the ephemeral world of doing; 'الدنيا' (*'adduniya*; this world) and the eternal world of reward/punishment 'الأخرة' (*'al'āḥirah*; the Hereafter). Concordance 4.17 presents 5 concordance lines of Ibn Baz's actual use of the keyword 'الأخرة' (*'al'āḥirah*; the Hereafter).

Concordance 4.17 Concordance of the keyword 'الأخرة' ('al'āḥirah; the Hereafter) in Ibn Baz

N

Concordance

1.	Concordance
	[] والجزير بشلب العلم، أيتما كان أن يقل على كتب الله، []. و من تعبر القرآن طالبا لليدي أعزه الله .[] و هذا السنة المطيرة []، فقد وعدالله سجته بالنصر العملين في النقل والقراب في الأخرة، قل [] والجزير بشلب العلم، أيتما كان أن يقل على كتب الله، []. و هن تعبر القرآن طالبا لليدين القرآن من القرآن ع
	اعز و جل (و للتِصْرُنُ الله من يتصُرُدُ إن الله للوي غزيز + النين إن مكناهم في الأرض الأمثوا الصاحد و التوا الزكة […] و لل عانية الافزير)، فقد وعد الله سبحنه و تعدل على هانين ٢ ليليل ، لذيل يغملون للحق ، و ليليلون
	المستنة ، ويزدون الزكاة لمستحقيها [] بالنصر ، و هو بعم النصر في الدنيا و النمكين فيها و النصر و الرضي من الم سبحته بوم القيامة []
	[] and any student of [religious] knowledge should, wherever he is, commence willingly to study the Book of Allah []. Whoever studied the Book of Allah
	seeking [Allah's] guidance. Allah will endue him with power and honour [] and so for the purified Sunnah [] indeed Allah has promised to help those who help to
	establish Islam in this life, and reward them in the Hereafter, for He, the Exalted in Mighty and the Majestic, said: {Verily, Allah will help those who help His to
	establish (Cause). Truly, Allah is All-Strong, All-Mighty}. Those (Muslim rulers) who, if We give them power in the land, (they) enjoin prayers [i.e. to perform the
	five compulsory congregational Salat (prayers) (the males in mosques)], to pay the Zakat []. And with Allah rests the end of (all) matters (of creatures). Thus, in
	these two Qur'anic verses. Allah has promised to endue power and honour to those who work to establish Islam, enjoin prayers and pay the Zakat [alms] to those in
	need both in this life (give them power and honour) and in the Day of Judgement (Allah will be pleased with them) []
	[] فأبان سبسته في هذه الأبات أن من اتبع هذاه لم يضل و لم يشق بل له البدى و السعادة في النشا و الأخرة و من أعرض عن نكره فله المعيشة الضناك []
₁	[] Allah shows in these Quranic verses that whoever follow His guidance he will neither go astray, nor will be distressed. Indeed, he will be guided to the true path
2	and gain happiness in this life and the Hereafter. And whosoever turns away from His message (i.e. neither believes in Qur'an nor acts on its teachings) verily, for
	him is a life of hardship []
	[] و من حد عن سبله باء بالحبية و الندامة ، وخسر النتيا و الأخرة ، فلو كان دعاة التومية بقصدون بدعوتهم إليها تعظيم الإسلام وخدمته ، و رفع تستمه ، لما [] ، وغضبوا من صوت دعة الإسلام إذا دعوا إليه []
3	[] and whosoever does not followed His right path, will draw on himself unsuccess and remorse, and he loses both this world and the Hereafter. Thus, if the
	advocates of nationalism aim at glorifying and serving Islam, they would not be displeased with those who call for Islam []
	[] لأن الكتر و المعاصبي هما سبب كل بلاء وشر في الدنيا و الأخرق و أما توحيد الله و الإيمان به و برسله و صاعته و صاعته و ساعته و الدنيا و الأخرة، و النجاة من كل سكروه []
4	[] because infidelity and sins are behind every distress and evil both in this world and the Hereafter, but monotheism and belief in Allah and His messengers and
	obeying Him and His messengers [] [lead to] success in this life and in the Hereafter and salvation in calamities []
	[] فاحاروا أبها المسلمون ما حتركم الله منه، و بادروا إلى ما أمركم به بإخلاص و تسلموا من كل شر في النينو و الأخرق و من أعظم الطاعة لله التحاكم الى شريعته و الرضي بحكمها [] ، عملا بتول الله عز و جل:
	[[]] [قاد ر زايُدُ لا يُؤْمِنُونَ حتى يُحَمُّدُونَا فيها تَدَجَرَ بِيتِهم لا يَجِدُوا في أَنْشَيِعُ حربُها تشتيداً و يُتَأْمُوا تشانيداً []
	[] O Muslims, beware of what Allah has warned against, and set out to sincerely obey His commandments, you will be save of all evil both in this world and the
2	Hereafter. And part of the greatest obedience to Allah is to refer to His Sharia for judgement (in all dispute) and become pleased with any Sharia's judgement
	following the saying of the Exalted in Mighty and the Majestic [Qur'anic verse]: {[But no, by your Lord, they can have no Faith, until they make you (O Muhammad)
	judge in all disputes between them, and find in themselves no resistance against your decisions, and accept (them) with full submission) []
	The collocate 'الأخرة' (<i>'al'āḥirah</i> ; the Hereafter) is represented in Ibn Baz as the ultimate goal

of any Muslim who should strive and please Allah to gain success in both this world and the Hereafter. Line 3 indicates that Ibn Baz's recontextualizes criticism of (Pan Arab) nationalists who, according to him, do not follow the right path of Allah. Thus, they will bring on themselves failure and remorse and eventually lose both this world and the Hereafter. Line 4 indicates Ibn Baz's negative representation of infidelity and sins, as opposed to monotheistic beliefs, which as he claims 'are the cause of distress and eveil'. In Line 5, Ibn Baz encourages Muslims to avoid man-made laws, and instead refer to the Sharia of Allah for judgment (in all disputes). In this last line, he warns Muslims that dissuasion from the Sharia of Allah is the cause of evil in both this world and the Hereafter.

The second keyword in Ibn Baz that refers to the 'Hereafter' is 'النار' ('annar: the Hell-fire). Again it collocates with an oppositional term 'الجنة' ('alğannah; Paradise) (MI score: 9.263). Concordance 4.18 shows 5 concordance lines of Ibn Baz's use of the keyword 'النار' ('annar; the Hell-fire).

Concordance 4.18 Concordance of the keyword 'النار' ('annar; the Hell-fire) in Ibn Baz

 [] قال صلى الله عليه و سلم : (من منت و هو يدعو لله تذا حتل التالر) خرجه البخاري في صحيحه، و الأحاديث في هذا المعلى كثيرة ، و هذا هو معنى لا إله إلا الله، فإن معتده لا معبود حق إلا الله [] [] the messenger of Allah -may the blessings and peace of Allah be upon him- said: (He who dies associating anything with Allah would enter the Hell-fire). This Hadith is narrated by Albukhari in His collection of Sound Hadiths. There are many Hadiths in this respect which is meaning of 'there is no God but Allah' i.e. there is no diety that deserves worship except Allah [] [] فاتواجب على كل مؤمن : أن يحتر ذلك ، [] و قال فيه سيحانه أيضنا : {إن الله لا يُغيرُ أن يَشرَك به و يَغيرُ ما دُون ذلك لهمَ يُثناء}. فمن مات عليه فيو من أهل النار جزماً ، الجنة عليه حرام ، و [] Thus every believer should avoid that [polytheism], [] Allah the Glorified said {Verily, Allah forgives not that partners should be set up with Him (in worship), but He forgives except that (anything else) to whom He wills}. So whoever believes in deities other than Allah and dies will certainly be in Hell-fire. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. 	I	N Concordance	
 إ] the messenger of Allah -may the blessings and peace of Allah be upon him- said: (He who dies associating anything with Allah would enter the Hellfrey. This Hadith is narrated by Albukhani in His collection of Sound Hadiths. There are many Hadiths in this respect which is meaning of there is no God but Allah' ie. there is no diety that deserves worship except Allah. [] [] Fuit, and the inter is no diety that deserves worship except Allah. [] [] Thus every believer should avoid that [polytheism], [] Allah the Glonfied said {Verily, Allah forgives not that partners should be set up with Him (in worship), but He forgives except that (anything else) to whom He wills). So whoever believes in deities other than Allah and dies will certainly be in Hell-fire. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. [] Paice and the infective of the set of the s		ن صلى الله عليه و سلم: (من ملت و هو يدعو للمتار خل الشار) خرجه البخاري في صحيحه، و الأحديث في هذا المعنى كثيرة، و هذا هو معنى لا إله إلا الله، فإن معتاها لا معبود حق إلا الله []	[] دَز
 الطلال freq. 1. Inis Hadrih is narrated by Albukhan in His collection of Sound Hadrihs. There are many Hadrihs in this respect which is meaning of there is no God but Allah i.e. there is no diety that deserves worship except Allah [] [] Fiti (=, =, dot in the interiation of the except Allah (] [] Thus every believer should avoid that [polytheism]. [] Allah the Glofifed said (Venly, Allah forgives not that partners should be set up with Him (in worship), but He forgives except that (arything else) to whom He wills). So whoever believes in deities other than Allah and dies will certainly be in Hell-fire. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. [] The (Hourd) that add the due to the the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. [] and it [the will of Commandments] is fabricated and ascribed to the Messenger -may the blessings and peace of Allah be upon him It includes rituals that Allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him- said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occup his seat in Hell-fire) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger (I] (] (Take (Hourd) and the set of the dottines of the modem atheists that allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him- said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occups his seat in Hell-fire) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger (Malh di not say [] [] CuEic [Dail] Allah di not say [] [1 [] the messenger of Allah -may the blessings and peace of Allah be upon him- said; (He who dies associating anything with Allah would enter the	he
 is no God but Allah i.e. there is no diety that deserves worship except Allah [] ا] دلتو اخب على كل مؤمن: أن يترز الله. (] و قال في سيحله ليضا: {إنَّ اللَّه لا يَعْنُوْ أَنْ يُتُرَاكَ به و يَعْنُوْ مَا قَوْنَ يَسْأَعُل. بن مات عليه نيو مر أهل النز جزماً ، الجنة عليه حرام. (] در محف ني العل أبه الإيد نعرد يشمن نلله. 2 [] Thus every believer should avoid that [polytheism]. [] Allah the Glorified said (Verily, Allah forgives not that partners should be set up with Him (in worship), but He forgives except that (anything else) to whom He wills). So whoever believes in deities other than Allah and dies will certainly be in <u>Hell-fire</u>. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. [] و أنه [أرصبة] مكونية على رسول أه صلى أم عليه و سلوم أديل ، و كنت عليه كيو من ولي أنفي أ متحد من العلى)), و ق قال [] and it [the will of Commandments] is fabricated and ascribed to the Messenger -may the blessings and peace of Allah be upon him. It includes intuals that Allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him. Said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occupy his seat in <u>Hell-fire</u>) and, indeed, the liar who fabricated the Sommandment, has ascribed something that the Messenger of Allah did not say [] [] (أراب الأربي كله ، و من في كل مواني ما مواني ينه ، و للمواني و عربرها مراه الأبن السواي في الأبل و الأبل و الأبل المواني و الأبل المواني و الأبل و الأبل و الأبل و الأبل و الأبل و الله الله و الله والله و الله و المواني و مواني في مواني و المواني و مواني المواني و المواني و مواني المواني و مواني المواني و مواني الموا	'	Hell-fire). This Hadith is narrated by Albukhan in His collection of Sound Hadiths. There are many Hadiths in this respect which is meaning of th	iere
 2 [] Thus every believer should avoid that [polytheism], [] Allah the Glorified said {Verly, Allah forgives not that partners should be set up with Him (in worship), but He forgives except that (anything else) to whom He wills). So whover believes in deities other than Allah and dies will certainly be in <u>Hell-fire</u>. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. 2 (] The forgives except that (anything else) to whom He wills). So whover believes in deities other than Allah and dies will certainly be in <u>Hell-fire</u>. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. 3 (] and it [the will of Commandments] is fabricated and ascribed to the Messenger – may the blessings and peace of Allah be upon him It includes intuals that Allah has never allowed! The Prophet – may the blessings and peace of Allah be upon him It includes intuals that Allah has never allowed! The Prophet – may the blessings and peace of Allah be upon him It includes that the Messenger of Allah did not say [] (] or use (intertionally) let him occupy his seat in <u>Hell-fire</u>) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say [] (] or use (intertional did not say [] (] or use (intertional did not say [] (] and he infidelious doctrines (the doctrines of infidels) that oppose the true Islamic creed [are] the doctrines of the modem atheists who are the followers of Marx and Lenin and other propagandists of atheim and infidelity [] and part of their [atheists] tenets are the rejection of the final return. Paradise, <u>Hell-fire</u>, and all religons and wousever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life	L	is no God but Allah' i.e. there is no diety that deserves worship except Allah []	
 2 []Thus every believer should avoid that [polytheism], [] Allah the Glorified said {Verily, Allah forgives not that partners should be set up with Him (in worship), but He forgives except that (anything else) to whom He wills). So whoever believes in deities other than Allah and dies will certainly be in <u>Hell-fire</u>. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. [] ر آب [آب [آر صيه على رسول الله صلى الله صلى الله صلى الله صلى الله صلى الله عليه ر سلم : (((مر فل على ما نه الل فليقوا متعدمن السلم)) و فال ((مر فل على رسول الله صلى الله صلى الله صلى الله عليه ر سلم : (((مر فل على ما نه الل فليقوا متعدمن السلم)) و فال عليه ر سلم : ((مر فل على رسول الله صلى الله صلى الله صلى الله عليه ر سلم : ((مر فل على رسول الله صلى الله صلى الله صلى الله عليه ر سلم : ((مر فل على رسول الله صلى الله صلى الله عليه و سلم الله الله الله والله صلى الله عليه وسلم . (فل على رسول الله صلى الله صلى الله صلى الله عليه وسلم . (فل عليه الله و على ماله الله الله الله الله الله الله الله			
 (in worship), but He forgives except that (anything else) to whom He wills). So whoever believes in deities other than Allah and dies will certainly be in <u>Hell-fire</u>. He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. [] و آن [آن	.		
 in <u>Hell-fire</u>, He (polytheist) will not enter the Paradise, and will dwell forever in Hell-fire, we seek refuge in Allah from that. [] (آب [الرصية] مكاربة على رسول الله صلى الله عليه و سلم متصمنة لتشريع نين لم يأذن به الله". ((من قال على ما لم اقل قليشوا متحد من الطر)) و قد قال مندر من الم عليه و سلم. and it [the will of Commandments] is fabricated and ascribed to the Messenger -may the blessings and peace of Allah be upon him It includes intuals that Allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him- said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occupy his seat in <u>Hell-fire</u>) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say [] [] ([] و انعذا الكثرية المصنية المطبقة المصحيفة. [] ما يعتقد المدحدة في هذا العصر من أتباع ماركثر رئينين و غيرهما، من دعاة الإلحد و الكثر رألحد و الحر العرف ولله مالية عليه و لله عليه المالغ عليه و العليه و الغيو (لأخرة [] [] و انعذا الكثرية المصحيفة. [] ما يعتقد المدحدة في هذا العصر من أتباع ماركثر رئينين و غيرهما، من دعاة الإلحد و الكثر و العلي العليه و العليه و العليه و العليه و العليه و الغيو (لأخرة [] [] و انعذا الكثرية المصحيفة. [] ما يعتقد المدحدة في هذا الحصر من أينا عاركثر رئينين و هزار من ما عليه و العلي و الأخرة [] [] و انعذا الكثرية المصديفة الصحيفة. [] ما يعتقد المدالغذا لله عليه العليه و العليه و العليه و العليه و العلي و الأخرة [] [] و انعذا الكثرية المصديفة إلى عليه و العلي و الأخرة [] [] و انعذا الكثرية المصديفة و العلي ماله الله عليه و العلي و الأخرة و [] [] و الألمان و الله عليه و العلي و الأخرة [] و العلي و الأخرة [] [] و الله المالغلية الله معنية العلي الله و العلي علي و الله و العلي و الأخرة []	12	2 [] Thus every believer should avoid that [polytheism], [] Allah the Glorified said {Verily, Allah forgives not that partners should be set up with]	Him
 [] و تُبد الوصية] مكوبة على رسول الأصلى المُ عليه و سلم ماد يلل، و مُعتمدة لتشريد نين لم يكن به الثم أو قد قد النصية على رسول الأصلى الم عليه و سلم ماد يلل، و كذب صريح خطيراً [] and it [the will of Commandments] is fabricated and ascribed to the Messenger -may the blessings and peace of Allah be upon him. It includes rituals that Allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him- said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occupy his seat in <u>Hell-fire</u>) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say [] [] و تعتد الكثرية المصدية الصحيحة. [] ما يعتده المحمدة في هذا المصر من أتماع ماركس ولنيين و غيرهما: من دعاة الإلحاد و الكثر			be
مالتركي هذه أنوصية على رسول الله صلى الله عليه و سلمها لديقل ، و كذب عليه كناً صريحاً خطيراً [] and it [the will of Commandments] is fabricated and ascribed to the Messenger -may the blessings and peace of Allah be upon him It includes rituals that Allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him- said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occupy his seat in <u>Hell-fire</u>) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] or it is that the Messenger of Allah did not say [] [] are the infidelious doctrines (the doctrines of infidels) that oppose the true Islamic creed [are] the doctrines of the modern atheists who are the followers of Marx and Lenin and other propagandists of atheism and infidelity [] and part of their [atheists] tenets are the rejection of the final return, Paradise, <u>Hell-fire</u> , and all religions and whosoever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] [] [] artic atter atter (and would lead atheists to the most awful ends in this iffe and in the Hereafter [] [] artic atter atter (and would lead atheists to the most awful ends in this life and in the Hereafter [] [] artic atter atter atter atter atter atter atter atter			
 [] and it [the will of Commandments] is fabricated and ascribed to the Messenger – may the blessings and peace of Allah be upon him It includes rituals that Allah has never allowed! The Prophet – may the blessings and peace of Allah be upon him- said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occupy his seat in <u>Hell-fire</u>) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say [] [] or itsel ''''''''''''''''''''''''''''''''''''			
 rituals that Allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him- said: (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occupy his seat in <u>Hell-fire</u>) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say [] [] (Liele Libro, Allah did not say [] [] and the infidelious doctines (the doctrines of infidels) that oppose the true Islamic creed [are] the doctrines of the modem atheists who are the followers of Marx and Lenin and other propagandists of atheism and infidelity []. and part of their [atheists] tenets are the rejection of the final return, Paradise, <u>Hell-fire</u>, and all religions and whosoever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awfue and in this life and in the Hereafter [] [] [in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remores, and the <u>Hell-fire</u> will be their abode and destination, and in this respect, the Glorified and the Exalted says: (Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a the surface at the surface and the text of the prosent rulers) in the land, as He granted it to those before them, and the the will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a 		هذه الوصية على رسول الله صلى الله عليه و سلم ما لم يقل ، و كذب عليه كذبا صريحا خطيرا []	منتري
 what I have not said then (surely) let him occupy his seat in Hell-fire) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say [] [] والمدف الكفر (المعنية) المصححة، [] ما يعتده المحمدة في هذا المصححة، [] من يعتده المحمدة إلى أسرائيد الكفر (المعنية) والمحمدة المحمدة إلى أسرائيد المحمدة [] ما يعتده المحمدة إلى المحمدة إلى أسرائيد المحمدة إلى أسرائيد المحمدة إلى أسرائيد الكفر ([] والمدف الكفر (المعنية) والمعنية والمعنية العربة المحمدة المحمدة إلى أسرائيد المحمدة المعنية المعنية المحمدة المعنية معام المحمدة المحمدة المحمدة المحمدة المعنية المحمدة المح			
 that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Allah did not say [] that the Messenger of Marx and Lenin and other propagandists of atheism and infidelity [] and part of their [atheists] the doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] the Messenger of Marx and Lenin and the most awful ends in this life and in the Hereafter [] the Theorem of resurction] the believers will win Paradise and there appear on the infidels' faces the signs o			
 [] والعدف الكثرية المصديدة للمعتبدة المحدجة، [] ما يعتده المحددة في هذا العصر من أتباع ماركس ولينين و غير هما، من دعاة الإلحاد و الكثر، [] و من أصوليم إنكار المعد، و إنكار الجذة و [] والعدف الكثرية المصديدة المحدجة، [] ما يعتده المحددة في هذا العصر من أتباع ماركس ولينين و غير هما، من دعاة الإلحاد و الكثر، و من نظر في كتبهم ودرس ما هم عليه علم ذلك يقيناً، و لا ريب أن هذه العقيدة مصادة لجميع الأديان السدوية ، و منضية بأهلها إلى أسوأ العواقب في الذنبا و الأخرة [] [] and the infidelious doctrines (the doctrines of infidels) that oppose the true Islamic creed [are] the doctrines of the modern atheists who are the followers of Marx and Lenin and other propagandists of atheism and infidelity []. and part of their [atheists] tenets are the rejection of the final return, Paradise, <u>Hell-fire</u>, and all religions and whosoever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] [] إلى يوه التبامية إذ المارضين يتوزون بالجنة ، و الكفرون تعلو وجر هيم الذاته و الندامة ، و القار تكون مؤلهم و مسير هم. و في هذا المعني يتول سبحانه و تعاني: (منوا منكم و غيلوا الكثر في المناز في تعرفون المواقع و عنولوا المواقع و عنولوا المواقع و عنولوا المعاد و علي المعاد و علي الله و الندوا منكم و غيلوا المعاد و المعاد و المواقع و علي المعاد و المعاد و المواقع و علي المعاد و المواقع و علي المعاد و المواقع و علي المعاد و المواقع و علي المعاد و المواقع و علي المواقع و علي المواقع و علي المعاد و المواقع و علي المعاد و المواقع و علي المعاد و المواقع و علي أول المعاد و المواقع و علي المعاد و المواقع و علي أول المواقع و علي المواقع و علي أول المواقع و علي المواقع و علي أول المواقع و علي المواقع و المواقع و علي أول و المواقع و			hing
 التار ، و الكتر بالأدين كلها ، و من نظر في كتبهم وترس ما هم عليه علم ذلك يوندا ، و لا ريب آن هذه المعقيدة مصادة لجميع الأدين السمارية ، و متصرية بأهله إلى أسوا العواق في الدين و الأخرة [] and the infidelious doctrines (the doctrines of infidels) that oppose the true Islamic creed [are] the doctrines of the modern atheists who are the followers of Marx and Lenin and other propagandists of atheism and infidelity []. and part of their [atheists] tenets are the rejection of the final return, Paradise, <u>Hell-fire</u>, and all religions and whosoever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] [] [] in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the <u>Hell-fire</u> will be their abode and destination, and in this respect, the Glorified and the Exalted says: [Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a 			
 4 [] and the infidelious doctrines (the doctrines of infidels) that oppose the true Islamic creed [are] the doctrines of the modern atheists who are the followers of Marx and Lenin and other propagandists of atheism and infidelity []. and part of their [atheists] tenets are the rejection of the final return, Paradise, <u>Hell-fire</u>, and all religions and whosoever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] [في يوم أنتيامة] فللمؤمنون يتوزون بالجنة، و الكافرون تعلو وجومهم النالة و الندامة، و الندار متكور متواهم و مصيرهم. و في هذا المعنى يتول سيحانه و تعلق. (رحفا الله النين آمنوا منكو و خوليم ألف يتبكرون من المعنى يتول سيحانه و تعلق. (رحفا الله النين آمنوا منكو و خوليم ألف يتبكرون من المعنى يتول سيحانه و تعلق. (رحفا الله النين آمنوا منكو و خوليم ألف يتبكرون من المعنى يتول سيحانه و تعلق.) [in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the <u>Hell-fire</u> will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a 			
followers of Marx and Lenin and other propagandists of atheism and infidelity []. and part of their [atheists] tenets are the rejection of the final return, Paradise, <u>Hell-fire</u> , and all religions and whosoever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] [في بود التوامل المعنى يقول سبحانه و تعالى: (وطا الله الناية و النامة ، و القار تكون مقواهم و مصير هم و في هذا المعنى يقول سبحانه و تعالى: (وطا الله الناين آمنوا منكم و عَمِلُوا [في بود التوامية] فالمؤمنون يقوزون بالجنة ، و الكافرون تعلو وجوهيم الناية و النامة ، و القار تكون مقواهم و مصير هم و في هذا المعنى يقول سبحانه و تعالى: (وطا الله الناين آمنوا منكم و عَمِلُوا المسترحمين في الأرض كانا النتفلات النون من تقييم و ليتكلّن لهم بينيكم الذي التعلي كون مؤولهم أله الني يكون بي تبيئ [in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the <u>Hell-fire</u> will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a		. و الكفر بالأديان كلها ، و من نظر في كتبهم ودرس ما هم عليه علم نلك يقينا ، و لا ريب أن هذه المقيدة مضادة لجميع الأديان السماوية ، و مقضية بأهلها إلى أسوا المواقب في الذنيا و الأخرة []	النار ،
followers of Marx and Lenin and other propagandists of atheism and infidelity []. and part of their [atheists] tenets are the rejection of the final return, Paradise, <u>Hell-fire</u> , and all religions and whosoever reads in their books will realize that, and no doubt that their doctrines oppose all divine religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] [في بود التوامل المعنى يقول سبحانه و تعالى: (وطا الله الناية و النامة ، و القار تكون مقواهم و مصير هم و في هذا المعنى يقول سبحانه و تعالى: (وطا الله الناين آمنوا منكم و عَمِلُوا [في بود التوامية] فالمؤمنون يقوزون بالجنة ، و الكافرون تعلو وجوهيم الناية و النامة ، و القار تكون مقواهم و مصير هم و في هذا المعنى يقول سبحانه و تعالى: (وطا الله الناين آمنوا منكم و عَمِلُوا المسترحمين في الأرض كانا النتفلات النون من تقييم و ليتكلّن لهم بينيكم الذي التعلي كون مؤولهم أله الني يكون بي تبيئ [in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the <u>Hell-fire</u> will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a		1 [] and the infidelious doctrines (the doctrines of infidels) that oppose the true Islamic creed [are] the doctrines of the modern atheists who are th	le
religions, and would lead atheists to the most awful ends in this life and in the Hereafter [] [في يود النيامة] فالمؤمنون يتوزون باجنة، و الكافرون تعلو وجرمهم النالة و الندامة، و الندار تكون متواهم و مصيرهم و في هذا المعنى يقول سبحانه و تعالى: (وَحَدَّ اللهُ النين آمنوا متكم و حَمِلُوا المصابحات المنازية في الأرض كلنا المنتخلية أو النيكتر في بينية الذي التعلق و الندامة، و الندارة، و في هذا المعنى يقول سبحانه و تعالى: (وَحَدَّ اللهُ النين آمنوا متكم و حَمِلُوا المصابحات المنازية و المنازية و الندائة، و الندامة، و الندارة، و الندارة، و الندارة، و الندارة، و الندارة، و الندارة، و الندائة و الندائة، و المنازية، و مصيرهم و في هذا المعنى يقول سبحانه، و المعار، و حرمهم الذلة، و الندائة، و الندائينية، و حصيرهم و في هذا المعنى يترض من قالية و المنازية، و الندائة، و الندائينينية، و الندائة، و و الندائة، و الندائة، و المعادة، و المعادة، و الندائة، و الندائة، و الندائة، و الندائة، و الندائة، و الندائ و المنائة، و الندائة، و و الندائة، و الندائة، و		followers of Marx and Lenin and other propagandists of atheism and infidelity []. and part of their [atheists] tenets are the rejection of the final	
[في بود التيمة] فالمؤمنون يتوزون بالجنة، و الكافرون تعلو وجومهم الذالة و التدامة، و التاريخ، و القار تكون متواهم و مصبورهم و في هذا المعنى يتول سبحانه و تعالى: (وَحَدَاللَّهُ الذَينِ آمنوا متكمر عَمِلُوا المستجدب الالتخليلية لم في الأرض كلما المتخلفة الذين من تبريغ و للككُنُنُ فيم بينهم الذي التعنى ليم و تبيئاً لمن يتبر خوانية ألما يتبترتني لا يتبركون بي شيئاً) [in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the <u>Hell-</u> 5 fire will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a			
المتَرَيْنَةُ لَيْنَا يَتَبَدُونَهُمْ فَي الأَرض كُمَّا النَّقُلُتُ النين من تَبْيَهُمْ وَلَيَحْمَّنُ لَهم بينَهُمُ النَّعْنَى لَهم ر لَيَكَلَّنُهُ مِن بَعْرِ خَوْلِيهُمْ أَمَّا يَتَبَدُونَيهُمْ أَمَّا يَتَبَدُونَي عَيدًا [in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the <u>Hell-</u> <u>fire</u> will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a			
[in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the <u>Hell-</u> 5 <u>fire</u> will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a		د انتوامة] فالمؤملون يقوزون بالجلة، و الكافرون تعلو وجوهيم النابة و الندامة، و القال تكون متواهم و مصيرهم و في هذا المعتى يقول سبحانه و تعالى: ﴿وَحَدَّ اللَّهُ النَّابِينَ آمنوا منكم و عَمِلُوا	[في بو
5 fire will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe and do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a			
do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a		[in the Day of resurrection] the believers will win Paradise and there appear on the infidels' faces the signs of humiliation and remorse, and the Hell	<u>.l-</u>
that He will grant them the authority to practise their religion which He has chosen for them (i.e. Islam). And He will surely give them in exchange a	1	5 fire will be their abode and destination, and in this respect, the Glorified and the Exalted says: {Allah has promised those among you who believe a	and
		do righteous good deeds, that He will certainly grant them succession to (the present rulers) in the land, as He granted it to those before them, and	1
safe security after their fear (provided) they (believers) worship Me and do not associate anything (in worship) with Me}.			1
		safe security after their fear (provided) they (believers) worship Me and do not associate anything (in worship) with Me).	
Hell-fire is constructed by Ibn Baz as the abode of the wrong-doers in all lines in			

Hell-fire is constructed by Ibn Baz as the abode of the wrong-doers in all lines in Concordance 4.18. The wrong-doers in Ibn Baz are polytheists (Line 2) (also described as people who associate anything with Allah in their worship (Line 1), those who lie and fabricate Prophetic Hadiths (Line 3), atheists (Line 4) and infidels and those who associate others with Allah in worship i.e. polytheists (Line 5). In Line 3, Ibn Baz recontextualizes a Hadith (Prophetic Tradition) that includes the keyword 'النار 'annar: the Hell-fire) to refute certain mystic beliefs and reject a leaflet called 'the Will of Commandment' that was ascribed to the Messenger of Islam. Considering more co-text of concordance line 3, we find Ibn Baz's refutation of this leaflet which was written by an anonymous writer of a Prophetic Commandment attributed to someone Called Sheikh Ahmad, the Custodian of the Prophet's Mosque in Medina (second Islamic holiest city in Saudi Arabia):

Figure 4.5 Expanded version of line 3 in Concordance 4.18

فقد اطلعت على كلمة منسوية إلى الشيخ أحمد خلام الحرم النبوي القدريف بعنوان: (هذه وصية من المدينة المنورة عن الشيخ أحمد خلام الحرم النبوي الشريف) [...] قال فيها [...] و من لم يكتبها و يرسلها حرمت عليه شفاعتي يوم التيامة ، و من كتبها و كان فقيرا أغناه الله ، أو كان مديوناً قضى الله دينه [...] و لو فرضنا أن الشيخ أحمد المذكور [...] زعم أنه رأى النبي صلى الله عليه و سلم في النوم أو اليقظة ، و أوصاد بهذه الوصية . لعلمنا يقينا أنه كلاب ، [...] لموجود كثيرة منها [-أن الرسول صلى الله عليه و سلم لا يرى في اليوم أو اليقظة ، و أوصاد بهذه الوصية زعم من جهلة المسوفية أنه يرى النبي صلى الله عليه و سلم في اليقط يه و سلم لا يرى في اليتفلة بعد وفاته صلى الله عليه و سلم ، و من مكتوية على رسول الله صلى الله عليه و سلم في اليقظة ، أو أنه يحضر الموك أو ما شابه نلك ، فقد غلط أقيح الغلط [...] مكتوية على رسول الله صلى الله عليه و سلم في اليقظة ، أو أنه يحضر الموك أو ما شابه نلك ، فقد غلط أقيح الغلط [...] و أنها لمكتوية على رسول الله صلى الله عليه و سلم في اليقظة ، أو أنه يحضر الموك أو ما شابه نلك ، فقد غلط أقيح الخلم مكتوية على رسول الله صلى الله عليه و سلم عن اليقظة ، أو أنه يحضر الموك أو ما شابه نلك ، فقد علما أقيح الخلط [...] و أنها مكتوية على رسول الله صلى الله عليه و سلم و من على الله عليه و سلم ما لم يقل عليه و منام ، و من على ما لم الما ليتور منذ الله عليه و سلم و متصنماته لتشريع دين لم يأذن به الله إو ف قال النبي صلى الله عليه و سلم : و التصديق يأمتال هذه المفتريات ، و أن يكون لها رواج فيما يبلكم ، فإن الحق عليه نور لا يلتبس على طليه، فاطلبوا الحق بدليله ، و البكم الملم عما أمكل علكم

Verily, I have read a letter attributed to Sheikh Ahmad, the Custodian of the Prophet's Mosque in Medina. It was given the following title: (This is a Commandment from the Prophet, Narrated on the Authority of Sheikh Ahmad, the Custodian of the Holy Mosque in Medina). [...]. The writer of this leaflet said: '[...] and whoever does not write it [the Commandment] or send it to others, will be deprived of my [the Prophet's] intercession with Allah to save him in the Day of Resurrection, and anyone, who is poor and in debt, writes it, Allah will make him rich, and Allah will pay all his debts. [...] and if we assumed that the above mentioned Sheikh [...] claimed that he, in a dream or in reality, saw the Prophet, may peace and blessings of Allah be upon him, and gave him this Commandment, certainly we know that he is a liar [...]. The case is so, because of many reasons such as: 1- The Prophet, may peace and blessings of Allah be upon him, cannot be seen in reality after his death, and anyone who is destitute of knowledge amongst the Sufis, claims that he sees the Prophet or asserts that the Prophet attends the celebrations of his birth and so forth, makes a serious error [...]. And it [the will of Commandments] is fabricated and ascribed to the Messenger -may the blessings and peace of Allah be upon him-. It includes rituals that Allah has never allowed! The Prophet -may the blessings and peace of Allah be upon him- said; (Whoever (intentionally) ascribes to me what I have not said then (surely) let him occupy his seat in Hell-fire) and, indeed, the liar who fabricated this Commandment, has ascribed something that the Messenger of Allah did not say. Thus, dear readers and brothers: Beware [these liars], and I urge you not to believe in such lies, and do not communicate them to others. Indeed, the path is illuminated for the seekers of truth. You should accept only religious knowledge that is based on evidence [from Qur'an and the sound Prophetic sayings] and ask the scholars of religion about any religious issue you are uncertain of

Ibn Baz disagrees with the mystic beliefs of some Sufis who claim that they can see

and communicate with the Prophet of Islam after his death and that they can beseech him for help. Ibn Baz recontextualizes the Hadith that contains the keyword 'Hell-fire' to legitimate his stance towards these Sufi beliefs. The phrase 'من جهلة الصوفية' (*min ğohalat 'aşofiyyah*; some ignorant individuals amongst the Sufis) reveals one aspect of social actor representation by aggregation. In Lines 4 and 5, Ibn Baz continues to delineate the in/out group construction (i.e. the believers vs. infidels and modern atheists who are the followers of Marx and Lenin and other propagandists of atheism and infidelity) (see Chapter 6 for more detailed analysis of Ibn Baz's representation of social actors). Line 4 of Concordance 4.18 shows Ibn Baz's stance towards other religions. He uses the term 'الأديان السماوية' ('al'adyān 'assamāwiyyah; divine religions) to indicate the three monotheistic (Abrahamic) religions i.e. Judaism, Christianity and Islam, making them as one front opposed by modern atheism that would lead atheists to the most awful ends in this life and in the Hereafter. In positively representing Islam, Christianity and Judaism as being connected and in opposition to atheism, Ibn Baz shows a very different stance towards Christianity and Judaism to Alhawali, who views such religions as wanting to destroy Islam.

Moving on to the other author, as noted above, Alhawali also uses these 'final return' keywords, though on a smaller scale. The 'النار' ('*annar*; the Hell-fire) and 'الأخرة' ('*al'āḥirah*; the Hereafter) occur 17 and 28 respectively. These two words are mostly used in one of Alhawali's books discussing secularism. Concordance 4.19 shows 4 concordance lines that reflect the actual use of the word' 'النار' ('*annar*; the Hell-fire) in Alhawali.

N	Concordance
	[] عاقبة هذا الشرك هي الخلود المؤيد في الثان فحسب، بل إن أتباعه ليصلون نار الضياع و التمزق في هذه النانيا []
1	[] the result of this polytheism (secularism) is not only dwelling in Hell-fire forever, but
	secularists will suffer from being lost and ripped apart []
İ	[] فإنه خير لك أن تنقد عضوا من أعضائك من أن يلقى جسدك كله في ا لثار. و انطلاقاً من ذلك حرمت الزواج على
	رجال الدين []
2	[] and it is better for you (in order to save yourself from God's punishment) to lose one of your
	organs, rather than being put in Hell-fire. Based on that, the [Church] made it unlawful for
	clergymen to get married []
1	[] و قصد حرماتهم من كل خير في الدنيا، وابتلاهم بحرماتهم من <u>الثان</u> التي هي ضرورية جداً للإنسان.
3	and he (Zeus) intended to deprive them from all good things in this world, he also punished them
	by hiding the fire from them which is a necessity for all humans.
	[] التي هي ضرورية جدا للإنسان، و لكن (بروميتوس) سرق الثار، من السماء أو من مصنع (هيقايستوس) المه النار و
	الحرف و بخاصبة الحدادة []
4	[] which is a necessity for all humans, but Prometheus stole fire from heaven or from the
	factory of Hephaestus the god of fire and craftsmen, especially blacksmiths []

Alhawali uses the word 'النار' (*'annar*) in two senses: the first is religious and pertaining to the punishment in the Hereafter (Hell-fire) and the second is to indicate the fire that humans use. As indicated by Line 1 in Concordance 4.19, Alhawali uses the word 'النار' (*'annar*; the Hell-fire) to claim that secularism is a type of polytheism which 'not only leads secularists to dwell in Hell-fire forever, but they will suffer from being lost and ripped apart' in this life. In

the second concordance line, Alhawali criticizes the Church's decision that obliged clergymen not to get married as in the expanded version of Line 2:

Figure 4.6 Expanded version of concordance line 2 in Concordance 4.19

فاتحفة حمتلاً- خلق إنساني نبيل [...] و دعت إليه الأديان كافة، لكن الكنيسة بالغت في تصور ها لهذا الخلق، حتى حرمت ما أحل الله، [...] و ذلك بابتداعها الرهبانية و تنفير ها التديد من المرأة لذاتها، فتعاليمها تقول عن النظر المجرد: إذا نظرت عينك إلى معصية فاقلعها، فإنه خير لك أن تفتد عصواً من أعضائك من أن يلتي جسدك كله في اللهار. و انطلاقاً من ذلك حرمت الزواج على رجال الدين معتقدة أن رجل الدين لا يجوز له أن يهبط إلى مستوى أخلاق المتعب، و يتاركهم في السار. و انطلاقاً من ذلك الوجهة النظرية، أما الواقع العملي فتىء مختلف تماماً، فقد كانت الأديرة مناوعين على على الوجهة النظرية، أما الواقع العملي فتىء مختلف تماماً، فقد كانت الأديرة مباءات للفجور و مواخير للدعارة، و كان للباوات و الآمان الما يعنيات و الحظايا ما لا يكون لذى الملوك الدنيويين، و تولى منصب البابوية عدد من الأبناء غير الما عين ال الأباء و الكرادلة [...]

Chastity, for example, is a noble human trait that all religions invited people to have. However, the Church exaggerated in conceptualizing this human trait. It [the Church] forbade what Allah made lawful [...] because it invented monasticism and extremely estranged the woman for only being a female. The teachings of the Church say the following about the mere looking at a woman: if your eye looked at a sin (i.e. a woman) pull it out because it is better for you (in order to save yourself from God's punishment) to lose one of your organs, rather than being put in <u>Hell-fire</u>. Based on that, the [Church] made it unlawful for clergymen to get married. This decision is based on the belief that clergymen should not fall to the level of ordinary people and share them the same despicable pleasure! [...] this is from the theoretical perspective. However what happened in reality was completely different. Monasteries were brothels and places for immorality. The Popes and priests had more mistresses and concubines than kings had. Indeed some of those who had Papal position were illegitimate sons of some Popes and Cardinals (senior priests in the Roman Catholic Church) [...]

Monasticism is the target of Alhawali's criticism of the Church. He is alluding to the

following Qur'anic verse that is concerned with Christian monks:

{ئُمَّ قَفَيْنَا عَلَىٰ آثَارِهِم بِرُسُلِنَا وَقَفَّيْنَا بِعِيسَى ابْنِ مَرْمَ وَآتَيْنَاهُ الْإِنجِيلَ وَجَعَلْنَا فِي قُلُوبِ الَّذِينَ اتَّبَعُوهُ رَأْفَةً وَرَحْمَةً وَ رَهْبَانِيَّةُ ابْتَدَعُوهَا مَاكَتَبْنَاهَا عَلَيْهِمْ إِلَّا ابْبْغَاءَ رِضْوَانِ اللَّهِ فَمَا رَعَوْهَا حَقَ رِعَايَتِهَا فَآتَيْنَا الَّذِينَ آمَنُوا مِنْهُمْ أَحْرَهُمْ وَكَثِيرٌ مَنْهُمْ فَاسِقُونَ} سورة الحديد ايه ٢٢.

This Qur'anic verse could be translated as:

{Then, We sent after them Our Messengers, and We sent 'Îsa (Jesus) - son of Maryam (Mary), and gave him the Injeel (Gospel). And We ordained in the hearts of those who followed him compassion and mercy. But the **monasticism which they invented** for themselves, We did not prescribe for them, but (they sought it) only to please Allah therewith, but they did not observe it with the right observance. So We gave those among them who believed their (due) reward; but many of them are Fasiqun (rebellious, disobedient to Allah). Holy Qur'an 57:27.

But Alhawali adds his explication of the monasticism to further denounce the Church (see footnote 49 below). He claims that part of monasticism is to ban marriage for clergymen (which is not true for all Christian denominations). Banning marriage for clergymen, as Alhawali claims, caused moral denigration to the extent that some of those who were appointed as Popes were illegitimate children of other Popes and priests who had sex with the many mistresses and concubines they had. The last two lines (3 and 4) are actually part of the same sentence and are from Alhawali's discussion of 'fire' in Greek Mythology which, according to him, is one of the reasons that led the scholars of the West to firmly believe in the defeat of God as indicated by the following quotes taken from the co-text of concordance line 4:

[...] هكذا رسخ في النفسية الأوروبية شعور متأصل بأن كل طفر للإنسان في مجال العلم والمعرفة إنما هو هزيمة لإرادة الإله [...] ، و حتى بعد أن تخلت أوروبا عن عبادة زيوس و جوبيتر، و رفضت عبادة إله الكنيسة معتنقة عبادة (الطبيعة) . [...] Thus, a feeling is established in the minds of the Europeans that any success in science or knowledge is considered a defeat of the God's will, [...] And even after the Europeans had deserted worshipping Zeus and Jupiter, they refused to worship the God of the Church, and instead they worship nature.

Alhawali continues to criticize the Church as indicated by his use of the word 'الأخرة'

('al'āhirah; the Hereafter) as indicated by the concordance lines in Concordance 4.20:

Concordance 4.20 Concordance of the keyword 'الأخرة' ('al'āhirah; the Hereafter) in Alhawali

N	Concordance
	[] فيمت الكنيسة من قول المسيح: "مملكتي ليست من هذا العالم" -إن كان قالها. أن الدنيا و الأكرة [] ضدان لا يجتمعان، الدنيا مملكة الشيطان [] و عمل الإنسان فيها لتحسين
	أوضاعه المعاتبية [] كليما أعمال دنسة بملويها الشيطان ليصرف الإنسان عن مملكة المسيح الحالدة (الأخرة)، و النتر و شطف العيش - حسب المفهوم الكنسي- هما مفتاح الملكوت
	الضامز، []
1	[] the Church interpreted the following saying of Christ-if he really had said it: "my kingdom is not part of this world" that the life and the
	Hereafter [] are two opposing and conflicting worlds; this life is the kingdom of the Satan [] and the human efforts to improve his living
	[] are all profane that were dictated by Satan to dissuade humans from the eternal Kingdom of Christ (the Hereafter), and, according to the
	Church, poverty and destitution of good means of living are the needed key to enter the Kingdom of Christ []
	[] ما بيمنا الأن هو أن مهزلة صكوك الغفران قد ساعدت بصنة مباشرة على هدم التعاليم النينية من أساسها، و الاستهتار بكل المعتدات و الأصول الإيمانية بجملتها. و أسهدت في انتشار
	فكرة إنكار الأخرة و المبته و النار التي لا يقوم دين بغيرها []
2	[] what concerns us here is the farce named 'Indulgence' (full or partial remission of temporal punishment granted by the Church) which
	directly contributed to destroying the religious teachings (in Christianity) and rediculing all beliefs and the fundamentals of faith. It also
	contributed to the widely spread denial of the Hereafter, Paradise and Hell-fire that all religions are based on []
	[] و نستطيع أن تستعرض بعض ملامح الأدب و القن في تلك العصر (عصر النهضة الأوروبية) معتمدين على بحوت و تحليلات بعض المفكرين الأوروبيين. فنجد أن أبرز ها ما يلي
	:[] 2- الاهتمام بالحياة النتيا والوجود الإنساني فيها: و تلك الاهتمام نشأ رد فعل لتركيل الكنيسة على عالم الأخرة و حصر كل التشاط الفكري و القلى لاتباعها في مجال الحديث عن
	التانوت و القانيسين و الملائكة و المعجزات ، و كيت الفشاص و الأحاسيس الإنسانية أياً كانت ما لم تكن في حدود دائرتها اللاهوتية []
2	[] and we can have a look at some of the features of literature and arts in that era (Renaissance) as indicated by researches and analyses of
	some European intellectuals. These are the most prominent features: [] 2. The interest in worldly life and the existence of human: this
	interest emerged as a reaction to the emphasis on the Hereafter by the Church and the restriction it imposes on the intellectual and artistic
	activities of its followers on Trinity, saints, Angels and miracles. It was also emerged as a reaction to suppression of human feelings
	whatsoever unless they are within the theological boundaries of the Church []
	أنو أن الكنيسة ريانية حقاً لكان أول عمل عملته بعد القضاء عهد الاضطهاد المرير هو البحث عن ذاتها هي. بيعث الإنجيل الأصلي ونشره [] كان في إمكانها أن تقتع الامبراطور []
	غير أن الذي حصل فعلاً هو أنه لا الكنيسة كانت مؤمنة جادة تطمع في هداية النامل ابتغاء مرضاة الله والدار الآخرة، و لا قسطنطين كان مؤمناً جاداً يريد أن يضع عن عنته ربقة الوتنية []
	[] If the Church had been really holy, the first thing that it should have attempted to do, after the period of oppression, is to search for its
+	role in life. This could happen through reviving and publishing the real Bible [] Also, the Church was able to influence the emperor to
	become Christian [] But what happened was different; neither the Church was faithful and serious to guide people to the right path to
	please Allah and gain success in the Hereafter nor Constantine the Great was a good believer who wanted to get nd of paganism []

In all four Concordance lines, Alhawali blames the practices of the people of the Church that, according to him, led to the rejection of Christianity. In Line 1, Alhawali suspects that Jesus had ever said 'my kingdom is not part of this world'. However, he continues to criticize the Church claiming that its explanation of the words of Jesus made Christians think only of the Here-after and consequently live in poverty and destitution. Alhawali's target in Line 2 is Indulgence (full or partial remission of temporal punishment granted by the Church) which, as he claims, contributed directly to the rejection of the fundamental religious concepts of the Hereafter, Paradise and Hell-fire and eventually lead to the rejection of religions. Thus Alhawali blames the Church for the rejection of not only Christianity but all religions. He also blames the Church (Line 3) for the evolution of literary movements (without mentioning any) that made the worldly life (which seems that he is against them) as their core concept. He claims that these movements appeared as a reaction to the Church's emphasis on the Here-after, the restrictions it imposed on intellectual and artistic activities and suppression of human feelings. In Line 4, Alhawali blames both the Church⁴⁹ and the first Roman Emperor (306 to 337), who converted to Christianity, for not being faithful to the true Christianity. Thus, the semantic foci indicated by the keywords that refer to the 'final return' are recontextualized by Alhawali, in order to criticize the Church, Western thinking and secularism.

4.3 Conclusion

So far I have delineated the theological semantic foci in Ibn Baz. I have also examined how Alhawali refers to the same concepts, because he uses some of the theological semantic foci keywords in Ibn Baz with considerable frequency, and it is useful to compare and contrast how each Saudi Salafi writer represents and legitimates his stances towards the

^{49.} Alhawali seems not to differentiate between the different Christian factions and he usually links Christianity to Catholicism.

same semantic foci. The findings in this chapter could be summarized as follows: First, comparing the keywords in the two corpora revealed the accentuation of a major difference regarding the semantic foci, namely, Ibn Baz's theological and Alhawali's political thematic foci. Secondly, both writers seek to use the key term 'ALLAH' in well-known Islamic idiomatic structures which help to legitimate their texts. Thirdly, they also use intertextual quotes from other Islamic authentic texts to align their writings to the Islamic authentic texts – a form of legitimation via conformity (further detailed in sections 7.4, 7.5, 8.3 and 8.5). The two sacred texts, the Qur'an and the Sunnah are extensively used by Ibn Baz and, on a smaller scale, by Alhawali. Additionally, both writers use intertextual quotes from texts written by medieval Salafi religious leaders. All intertextual quotes are recontextualised by both writers to legitimate the arguments that appear therein.

Fourthly, despite the different authorial stances towards polytheism, it is represented negatively by both authors as it, according to the two authors, opposes Islamic monotheistic beliefs. However, from a religious ritualistic authorial stance, Ibn Baz rejects polytheism, whereas Alhawali's dislike of polytheism stems from a politically oriented authorial stance which suggests a step towards the politicizing of Islam. The theme of politicizing Islam will be further investigated in Chapter 5 when we look more closely at Alhawali's keywords.

Chapter 5: Examining Alhawali's politically-oriented semantic foci keywords across both corpora

5.1 Introduction

This chapter continues the analysis that was initiated in the previous chapter. It analyzes a selection of Alhawali's more politically focussed semantic foci keywords and the different ideological legitimation and representation strategies that appear therein. As in the last chapter, for the purpose of comparing Alhawali's writing to that of Ibn Baz's, I shall also consider Ibn Baz's use of some keywords that occur in both Salafi corpora. The selection of such keywords is based on the quantitative and qualitative criteria outlined at the start of Chapter 4. To reiterate from Chapter 4, the present chapter responds to research question 1(a): What do concordance and collocational analyses of keywords and frequent words in the Salafi corpora reveal about themes/macro topics? Answering this question contributes to answering the overall question: To what extent and how is Salafi discourse in Saudi Arabia in the 1980s -1990s homogenous?

As I have indicated in Chapters 3 and 4, I utilized a classificatory scheme, based on a preliminary investigation of collocates and concordance analysis, that enabled me to classify these keywords according to their functions (see Tables B1-4 in Appendix B). To remind the reader, Table 5.1 shows the semantic foci keywords in Alhawali's corpus.

Table 5.1	Semantic foci	keywords in	Alhawali's corpus

N	Key word	Eng. trans.		F.Baz		Kay word	Eng. trans.	F.	T D -
1		Ешторе	263	4	24		the military (def. adj. fem.)	47	F.Baz
2		the Gulf	161	1	24 25			1	0
3		the West	181	6			contemporary	47	0
4		the theory			26		the modern	54	1
5		the revelution	120	0	27		the past	44	0
			115	0	28		the religious (def. adj. masc.)	57	2
6		the intellect/ the thinking	120	3	29		the Crusades	51	1
7		the religious (def. adj. fem./pl)	104	1	30		the religion	554	222
8		the western	110	2	31		the () World War/the international	55	2
9		the European	92	0	32	ئاريخ	history (indef.)	59	3
10	-	the war	112	9	33	التسبور	ages	48	1
11	الصراع	the conflict	65	0	34	المعاصرة	the modern	40	0
12	عصير	era	83	4	35	البعليانسي	the political (masc. adj./the politician	39	0
13	السباسة	politics	82	4	36		region	49	2
14	المعقل	mind	116	13	37		military (indef. adj. fem.)	36	ō
15	الأوزوبي	European (sing. adj. masc.)	55	0	38		the military (def. adj.masc.)	35	0
16	السياسية	the political (adj. pl.)	67	2	39		religious (indef. adj. fem./pl.)	33	0
17	الغربي	the western (pl. adj. masc.)	67	2	40		Christianity	51	4
18	التاريخ	the history (def.)	107	12	41	الوطنية	patriotism	30	0
19	الوسطى	the middle (ages)	59	1	42		materialistic/materialism	42	2
20		movement	58	1	43		the historical	29	ō
21		the movement	57	1	44	-	intellectual	36	1
22	الاجتماعى	the social (contract)	56	1	45		civilization	45	3
23	إسرائيل		83	7					

Let us now consider in greater detail the politically oriented semantic foci in Alhawali.

5.2 Politically oriented semantic foci in the Salafi Corpora

Utilizing preliminary concordance-based analysis, I was able to classify the semantic foci keywords in Alhawali into six categories as shown in Table 5.2. As before, for simplicity, only the English translations are given in this table.

Ν	Categories	Keywords					
1	Locations and nationalities	Europe, the Gulf, the West, the western (pl. adj. fem), the European (pl.adj. fem), European (sing. adj. masc.), the western (pl. adj. masc.), Israel, region					
2	Military and wars	the war, the conflict, the military (def. adj. fem.), the Crusades, the () World War/ the international, military (indef. adj. fem.), the military (def. adj. masc.)					
3	Periods of time	the () century, era, the history (def.), the middle (ages), history (indef.), ages, contemporary, the modern, the past, the historical					
4	Politics	the revolution, politics, the political (adj. pl.), the social (contract), movement, the movement, the political (masc. adj.)/ the politician, patriotism					
5	Mind and abstract terms	the theory, the intellect/ the thinking, mind, materialistic/materialism, intellectual, civilization					
б	Religion	the religious (def. adj. fem. pl), the religious (def. adj. masc. pl.), the religion, religious (indef. adj. fem. pl.), Christianity					

Table 5.2 Categories of semantic foci keywords in Alhawali's corpus

Based on a preliminary concordance analysis, I first observed that the keywords in the first category (Locations and nationality) indicate two different and, according to Alhawali's perspective, opposing worlds (see below). Of the 9 keywords in this category, only 2 refer to Arab countries or areas (the in-group), viz., 'الخليج' ('alḥalīğ; the Gulf) and 'anatiqah; region). The remaining keywords in this category indicate the out-group, namely the West and Israel.

Secondly, the keywords in category 2 (Military and wars) relate to both historical and modern wars that, as Alhawali claims, had or are having negative consequences on the 'Muslim/Arab world'. (*'al'ālamiyyah*; world/international) is used by Alhawali to refer to the First and Second World Wars (25 out of 36 occurrences) and 'الصليبية' (*'asalibiyyah*; pertaining to the Crusades) is also used to refer to the medieval wars on the 'Muslim world'. The other three keywords are all derivatives of the word '*askar*) which could be translated into English as 'army, military or troops'. All the three derivatives indicate (possible) military invasions or confrontations between the West and Arab and/or Muslim countries (detailed below).

The keywords in category 3 (Periods of time) indicate themes that intersect with the themes in the above two categories, namely, 'Locations and nationalities' and 'Military and wars'. For example, according to Alhawali, the Arab (Muslim) lands were scenes of both historical (Crusades) and modern (colonial) wars. The rest of the keywords in categories 4-6 indicate themes that refer to Alhawali's reflections on political, social and religious life in the West. The keyword 'الثورة' ('*altawrah*; the revolution), in category 4 (Politics), often indicates the French revolution (49 times out of 115) or the industrial, scientific and knowledge revolutions against the Church in the West (26 times collectively). It also refers to the Islamic revolution in Iran (6 times). The rest of the occurrences of this keyword refer to communist (3 times) and other revolutions. Category 6 (Religion) contains 4 keywords that are derived from the same root. These keywords are 'الدين' ('*addīn*; the religion) (554 times), 'الديني' ('*addīn*; the religion) (554 times), '

(*'addīni*; the religious – def. adj. masc.) (47 times), 'الدينية' (*'addīniyyah*; the religious – def. adj. fem./pl.) (104 times) and 'دينية' (*dīniyyah*; religious – indef. adj. fem./pl.) (33 times).

As previously noted, due to time and space restrictions, it is beyond the scope of this thesis to do justice to all the keywords. Thus I utilize the criteria I have introduced in Chapter 4, Section 4.1, to narrow down the scope of the analysis to the most prominent political thematic focus in the Salafi corpora. Table 5.3 shows the selected keywords for the analysis.

Table	Fable 5.3 Political semantic foci keywords in Alhawali's corpus ⁵⁰									
	Category	N	Keyword Eng. trans.	Freq.	F. Baz	Keyness	Р			
핏	Locations and	1	the West الغرب	181	6	137.282	7.3E-16			
External	Nationalities	2	Israel إسرائيل	83	7	48.076	2.9E-13			
nal	Politics	3	politics السياسة	82	4	58.38	5.5E-14			
threats		4	the political السياسية	67	2	53.697	1.1E-13			
ats	Military invasion	5	the war الحرب	112	9	66.356	2.4E-14			
Cons	Construction of history		century القرن	134	8	89.417	4.7E-15			
	Mind Religion		mind المعقل	116	13	57.56	6.1E-14			
			the religious الدينية	104	1	96.293	3.3E-15			
			the religion الدين	554	222	42.784	1.2E-12			

Table 5.3 shows 9 of Alhawali's semantic foci keywords classified into 6 categories. The categories 'Locations and nationalities' and 'Politics' could be grouped together to indicate one major theme, namely 'External threats'. Although the category 'Military invasion' could be classified under the theme 'external threats' or 'Construction of history', I prefer to deal with it separately as it requires a more detailed analysis as shown below. It is also noteworthy that the keywords '*Lizqurb*' ('*algarb*; the West) and '*Lizquib*' ('*isra*'*īl*; Israel) indicate Alhawali's representation of social actors using various strategies identified by van Leeuwen (1996). These keywords involve strategies of impersonalization (referring to humans via an abstract concept), spatialization (referring to people via the country or region that they live in) and collectivization (referring to all the individuals of one country/region as being in a single group). I debated whether to investigate these keywords in the following chapter which looks at social actors, but have focused the social actor chapter on less abstract cases of social

^{50.} The keyword 'المسليبية' ('asaleebiyyah; crusades) is also worth investigating here, but it does not satisfy the criteria I used for choosing the semantic foci keywords in the analysis. However, based on collocational analysis, it emerged as a collocate of the keyword key term 'WAR' (MI score 10.997 and joint freq. 25). Thus, I will include it in the analysis of the key term 'WAR' (see Section 5.4.1 below).

actors which involve personalization. Therefore, I discuss these two keywords in this chapter as semantic foci indicators rather than as social actors, although I acknowledge the overlap.

To reiterate from the selection of keywords criteria that I have outlined in Chapter 4, keywords that belong to the same lemma or word-form and occur in the reference corpus (in this Chapter Ibn Baz's corpus) are tabulated and included in the analysis. Thus each of the categories 'Politics' and 'Religion' contains two sets of keywords each of which are derived from one root: 'السياسة' ('assiyasah; politics), 'السياسة' ('assiyasiyyah; political) (occurring collectively 264 times in Alhawali and 6 times in Ibn Baz) and 'الدينية' ('addīniyyah; religious) (occurring collectively 611 in Alhawali and 224 in Ibn Baz). However, these two sets of word forms occur as keywords in Alhawali's corpus. Additionally, the keyword 'الدين' ('addīn; the religion) is also used by Ibn Baz with considerable frequency (222 times). Let us now examine the selected keywords in each category in greater detail.

5.3 External threats 1: Locations, nationalities and politics

As I have indicated in Chapter 4 (section 4.2.1.3), so far, Alhawali's most prominent thematic foci appear to be based on negatively representing the West. Ideologically, this serves to legitimate his stance towards what he believes to be western dominance over Muslim/Arab countries. Alhawali seems to be consistent in his views towards the West as an aggressor towards Muslim countries (see Section 5.4 below). As I have indicated above, based on a preliminary concordance analysis, I found that many keywords indicate one major theme in Alhawali, namely that of external threat. The first two keywords in Table 5.3 that indicate external threat are 'الغرب' ('alġarb; the West) (181 times in Alhawali and only 6 in Ibn Baz) and 'العرب'' ('isra'īl'; Israel) (83 times in Alhawali and 7 in Ibn Baz). Additionally, 'اسرانیل' ('assiyasah; politics) and 'السیاسیة'' ('assiyasiyyah; political) occur collectively 149 times in Alhawali's texts and 6 times in Ibn Baz's. The huge difference of occurrences of these four

keywords in both sets of texts (413 times collectively in Alhawali and 19 in Ibn Baz) shows Alhawali's reliance on and interest in these threat-evoking keywords. Let us take each keyword in turn, starting by sketching collocational profiles for each.

5.3.1 The key term 'WEST' in Alhawali's corpus: Western civilization as an enemy of Islam

The keyword 'الغرب' (*'alġarb*; the West) has no collocates. However, Alhawali uses two related word forms which do have collocates.⁵¹ These word forms are 'الغربي' (*'alġarbi*; western -adj. masc.) and 'الغربية' (*'alġarbiyyah*; western –adj. fem.). Table 5.4 shows the collocates of these word forms.

Table 5.4 Collocates of the key term 'WEST' in Alhawali's corpus

Word forms	Eng. trans.	Freq.	Collocates Eng. tra	ns. Freq.	Texts	Joint	MI
الغربي	western (adj. masc.)	67	which/th الني	at 1230	3	8	4.25
الخربية	westem (adj. fem.)	110	civilizatic الحضارة	on 45	3	12	8.892
			Europe اوروپا	263	3	9	5.93

The collocates of the key term 'WEST' do not, at face value, bear any negative connotations that could mark the association between them and the key term 'WEST'. However concordance based analysis of these collocates reveals more about Alhawali's actual use of them. The word form 'الغربية' ('alġarbiyyah; western -adj. fem.) and its collocate 'الغربية' ('oroba; Europe) constitute the 2-word cluster 'أوروبا الغربية' ('oroba 'alġarbiyyah; Western Europe) that occurs 7 times across 3 texts and suggests that he occasionally is more specific about which countries he means when referring to the West. But examining the collocates of the key term 'WEST' via concordance-based analysis could reveal more about Alhawali's actual use in terms of ideology. I shall choose the lines that combine both the word forms of the key term 'WEST' and their collocates, then I shall choose a small number of representative examples that illustrate how the keyword is most typically used. Let us now examine the collocates of the key term 'WEST' each in turn.

^{51.} It is noteworthy here that I only include in the analysis the collocates that occur in three texts or more.
Concordance 5.1 Concordance of the collocate 'التي' ('allati: that/which) in Alhawali's corpus

N

Concordance نيذة عن المجتمع التيوعي: كان ما سبق لمحة عن اليهيمية الهابطة و المعينة الضائك و الحياة النكدة **التي** يصلّى سعيرها المجتمع **الغربي** المعاصر، و هى سحابة مظلمة تنطى سماء أوروبا كليا،[...] و ترسل صواعتها و تتواظيا على كل ركن من أركانها، و لكن الرفاق في موسكو يزعمون أن ذلك الوجه الكلّح خاص بالمجتمعات اليورجوازية، [...]

1 An overview of the communist society. The previous notes were only a preview about the low subhuman, hard wretched and painful life <u>that</u> excruciates modern <u>western</u> society. It is (life in western society) a gloomy cloud that covers the sky all over Europe, [...] It sends its sudden and severe attacks to every corner of Europe. But the comrades in Moscow claim that such gloomy face of life is peculiar to the bourgeois societies, [...]

و لئن كانت كتب أرسطو بمنبزلة الكرة المسفيرة التي نغنت منها أوروبا في انفلائها من سجن الكنيسة المظلم فإنها لم توصلها إلى بر الأمان، بل إلى نغق الجاهلية الإغريقية التي لم تخرج منه إلا إلى صحراء التلق والضبياع **التي** يصطلي الإنسان ا**لفربي** المعاصر بلهيبها.

² And if the books of Aristotle were a small window that Europe used to escape from the gloomy prison of the Church, nevertheless they did not take them to a safe land. On the contrary, they led them to the tunnel of the Greek ignorance through which it (Europe) went to the desert of anxiety and destruction <u>that</u> excruciate the <u>western</u> individual.
المنازخة في الفكر الأوروبي: إن ولادة أوروبا في ظل الحروب العسليبية وتسعورها بذاتها من خلالها هو الذي يفسر تلك التفقصات المسارخة التي يويتيها الفكر <u>الأفريبي</u> متمثلة في هذه المعادلات المسادخة للعربي المنازخة المعادلات المسادخة في الفكر الأوروبية.
The stark contradictions in western thinking: Indeed the birth of Europe and the realization of itself through the Crusades can

explain the stark contradictions in the <u>western</u> thinking <u>that</u> are resembled in the difficult formula: The Crusaders' hatred of Islam from the days of Peter the Hermit till Klaus that goes along with a complete revolution against the religion of the Cross (Christianity).

The collocate 'للتي ' ('allati: that) is a singular relative pronoun in Arabic. It relates the clause that follows it to an antecedent clause. This collocate and the word form ' الغربي' ('algarbi; Western –adj. masc.) occur 8 times jointly. As appears in the lines above, Alhawali describes negative situations using the relative pronoun 'التي ' ('allati: that) to assign them to the West. In Line 1, Alhawali constructs western society as low, subhuman, hard, wretched and painful. In Line 2 Alhawali claims that people in the West tried to avoid the restrictions of the Church through Aristotle's books but fell into Greek ignorance which led them to anxiety and destruction. He never clarifies what he means by this. Through a series of linked metaphors e.g. 'gloomy cloud', 'gloomy prison', 'tunnel of ignorance', 'desert of anxiety' lines 1 and 2 show how Alhawali negatively represents life in the West. Particularly in line 2, westerners are conceptualised as prisoners although their escape via Greek philosophy leads them into a desert. This view of western society as flawed seems to be a recurrent theme in Alhawali's writings which appears in my analysis of different keywords of Alhawali's corpus (see Chapter 4, Sections 4.2.1.3, 4.2.4 and 4.2.5, Chapter 6, Section 6.2.1.1 and 6.3 and Section 7.5 in Chapter 7).

In Line 3, Alhawali tries to justify the contradictions in his claims about the West being both fanatically religious (Crusaders) and also secular. In both cases Alhawali claims that the West is the enemy of Islam. This is shown in the cotext that immediately precedes Line 3.

Figure 5.1 Expanded version of Line 3 of the collocate 'التي' ('allati: that) in Alhawali's corpus

وإجمالاً: ولدت أوروبا ولادة جديدة، ووجد قديها لأول مرد في تاريخها التسور بأنها أمة واحدة تواجه عدراً أبدياً هو الإسلام. وكانت طفراتها في ذلك العصر الذي سمى "عصر النبحنية" أو الانبعات الذي تعدّنت ألا تجمله يدأ تاريخياً بمعرفة الدين الريائي واكتشاف حضارته العظمي، بل بلحظة الإبحار العكسي إلى الجاهلية الإغريقية واكتشاف أرسطو. To sum up, Europe was reborn. And for the first time in its history, Europe had the feeling that it is one nation that faces one eternal enemy which is Islam. The so called ' the Enlightenment age' is the early stage of the European development or the point of origins that Europe deliberately did not make it historically start by knowing the Godiy religion and discovering its greatest civilization, on the contrary, Europe made it start with the moment of salling against the tide towards Greek ignorance and the discovery of Aristode's books.

Aside from the fallacious hasty generalization about western hatred of Islam, Alhawali continues to reiterate his negative representation of the West and western culture (for detailed discussion about fallacies in Alhawali's argumentation see Section 7.5 in Chapter 7).

The other two collocates in Table 5.4 of the word form 'الغربية' (*'alġarbiyyah*; western –adj. fem.) are 'أوروبا' (*'oroba*; Europe) and 'الحضارة' (*'aḥaḍarah*; civilization) (joint

frequencies 12 and 9 respectively). Let us examine each in turn.

Concordance 5.2 Concordance of the word form 'الغربية' ('algarbiyyah; western –adj. fem.) and its collocate 'الحضارة' ('ahadarah; civilization) in Alhawali's corpus

N	Concordance		
	يقول التقرير (للمستشار الأول للرئيس الأمريكي جونسون سنة 1964م) (يجب أن ندرك أن تلك الخلافات بين إسرائيل و العرب، لا تقوم بين دول أو		
1	تسعوب بل تقوم بين حضارات). [] (إن غلبة الحضارة الغربية في التسرق - و هي العدو القديم للحضارة الإسلامية - قد أورتت العربي المسلم التسعور		
الصنار أمام طنوان تلك الحصارة التي يمقلها و يحترمها في نفس الوقت).			
	The report (by the senior advisor to the American President Johnson in 1964) says: We must realize that the Israeli-Arab		
	conflicts are not between countries or nations but between cultures) [] (the prevalence of the western civilization in the		
	East -which is the old enemy of Islamic civilization- made the Arab Muslim inherit the feeling of inferiority, humiliation and		
	insignificance in the presence of the inundation of that (western) civilization that he abhors and admires at the same time).		
	لقد استسلم المسلمون لنوم طويل محتجين بالقدر، فأم يوقظهم إلا هنير الحضارة الغربية و هي تدك معاقلهم و تقتحم حصونهم، [] و في لحظة الانبهار و		
	الانذهال هذه، قال المستشرقون [] و المبشرون و أذنابهج إن النين [] هو سبب التأخر و الجمود في العالم الإسلامي []		
2	Verily the Muslims had submitted themselves to a very long lethargy, claiming that this is their destiny, however they were		
-	woken up by the uproar of western civilization destroying their fortresses and invading their fortifications. And in this		
	moment of dazzlement and perplexity (because of western power) the orientalists [], missionaries and their tails (henchmen		
	amongst Muslims) say. Islam is the reason for underdevelopment and imperviousness in the Muslim world []		
	و قيما كانت الأوضاع تنحد إلى الهاوية، تلقت الأمة ضربة عليفة من يد نابليون - طليعة الحضارة الغربية الكافرة - أيتظنها هذه الضربة من نومها، و		
	لكنها أنقدتها صرابها.		
3	As (the Muslim nation) was falling to an inferior status, it received a violent strike by Napoleon (French Emperor 1804-		
	1815) -the front row of the irreligious western civilization. This strike awakened it (the Muslim nation) from its lethargy,		
	however this strike made the Muslim nation lose its way.		

After claiming that the West considers Islam to be its eternal enemy (Concordance 5.1), Alhawali constructs what he believes to be the antagonism and war between Islamic and Western civilizations. In Line 1 Alhawali utilizes a report (by the senior advisor to the American President Johnson in 1964) to lend support to his claims about western enmity towards Islamic civilization.⁵² In Line 2, Alhawali claims that western civilization attacked and destroyed Muslim fortresses. Alhawali also claims that Napoleon's military invasion of Egypt and Syria (1798–1801) is the first attack of the irreligious western civilization on the Muslim land (Line 3) that later invaded Arab lands. Alhawali does not provide any further information about the British involvement in helping the Ottomans defeat the French because he is concerned to show the enmity between the West sand the East rather than showing their collaboration (for more information see Matar, 1999).

The word form 'الغربية' ('alġarbiyyah; western –adj. fem) and its collocate 'الغربية' ('aḥaḍarah; civilization) are used by Alhawali in association with events that indicate historical wars and invasions initiated by some western countries on the Arab lands. His major claim here is that those wars and miseries resulted from western civilization. Lines 2 and 3 indicate Alhawali's exposition of the early military contact between some European countries (such as France) and the Arab countries that were under the Ottoman Empire, namely the events that took place during the 18th and 19th centuries.

Moving forward in history (i.e. to the last quarter of the previous century), Alhawali starts to use the word form 'الغربية' ('alġarbiyyah; western -adj. fem) with the collocate أوروبا'

^{52.} Alhawali is not quoting directly from a translation of the report by the senior advisor to the American President Johnson. He quotes from a book (The Battle of Islam; Past and Modern Time –no date) written by Mohammad Mahmoud Alsawwaf (1914-1992) who is an Iraqi religious leader and one of the founders of Muslim Brotherhood in Iraq. It is thus difficult to ascertain whether the senior advisor to President Johnson was accurately represented here which makes this a recontextualization at best –and very possibly a reformulation or construction.

Concordance 5.3 Concordance of the word form 'الغربية' ('alġarbiyyah; western -adj. fem.) and its collocate 'الغربية' ('alġarbiyyah; western -adj. fem.) in Alhawali's corpus

N	Concordance
	لقد حرصت الولايات المتحدة الأمريكية على تطويق الاتحاد السوفيتي و العالم الإسلامي بسلسلة من الأحلاف الحسكرية و السياسية منها حلف (الناتو)
	الذي يتسمل أ وروبا الغربية و حلف (السنتو) الذي شمل دول الشرق الأوسط []
1	The United States of America wanted eagerly to enclose the Soviet Union and Muslim World with a chain of military and
	political allies such as NATO that includes Western Europe and CENTO (Central Eastern Treaty Organisation also
	known as the Baghdad Pact which was dissolved in 1979) that includes the states of the Middle East []
	و في عدد آخر بتاريخ 21/12/1410هـ يقول كاتب آخر هو مدير المركز العربي لبحوت- التنمية والمستقبل بالقاهرة عن تحديات أوروبا: (يتزايد القلق في
	أوروبها الغربية و جنوب أوروبا بتسكل خاص و فرنسا تحديداً من تطور يطلقون عليه المد الإسلامي، و تطور آخر يسمونه التغيير الديمغرافي و التطوران
	حادثان في شمال أفريقيا).
2	And in another issue dated 21/12/1410 H. (15/07/1990) another writer, who is the director of Arabic Centre for Research-
	Development and Future in Cairo, says about the challenges that face Europe: (Concern increases in Western Europe,
	especially in southern Europe and more specifically in France, about the Islamic expansion and the demographic change that
	both exist in north Africa).
	إن أي باحت نزيه لا يستُطيع أن ينكر مآتر الدولة العتمانية و مزاياها التي تستحق التناء و التقدير، فهي التي جعلت المد الإسلامي في أوروبا الشرقية يبلغ
	مناد بعد أن فقد المسلمون بلادهم في أوروبا الغربية بسقوط الأندلس، و هي التي كسرت الكماشة الأوروبية التي كادت تطبق فكيها على العالم الإسلامي،
	و لم تتمكن من ذلته إلا بعد انهيار الدولة العتمانية.
3	Verily any respectable researcher cannot deny glorious deeds and the merits of the Ottoman Empire that deserve to be
	valued and praised because it made the Islamic expansion reach the highest of its range in eastern Europe after the Muslims
	lost their lands in Western Europe when Al-Andalus fell. The Ottoman Empire stopped the European dangers that were
	about to devour the Muslim world which took place after the fall of the Ottoman Empire.

With the rise of United States of America as a superpower in the world after WWII, Alhawali claims that during the Cold War (between the United States and the Soviet Union), the United States started to surround the Muslim World and the Soviet Union with its military allies (such as NATO and CENTO (Line 1)). Western Europe is viewed as one of these allies of America, responsible for threatening Muslim countries via military alliances.

Alhawali (Line 2) tries to legitimate his stance towards Europe utilizing a quote from the director of Arabic centre for Research development and Future in Cairo. However, he does not provide the name of the director of the Arabic centre in any part of the relative text which brings into question the veracity of his claims here (see p. 131 and for more details about Alhawali's argumentation strategies see Chapter 7). Alhawali tries to show that Western Europe countries are worried because of the expansion of Islam and demographic change in northern Africa. This could indicate that Alhawali tries to represent Western Europe countries as monitoring Arab countries.

Again, Alhawali tries to support his claims about Western Europe and their enmity to Islam. Line 3 shows Alhawali's historical accounts of Western Europe. He positively constructs the Ottoman Empire as protecting Islam and at the same time negatively constructs Western Europe as the enemy. Additionally, Alhawali tries to evoke the memories of his audience about the medieval Muslim state, Al-Andalus, in Western Europe (in parts of what are today called Spain, Portugal, Gibraltar, and France).⁵³

To sum up, Alhawali uses the key term 'WEST' to discursively construct a negative image of western civilization claiming that it is irreligious and excruciates (torments) western people. He also claims that western civilization wants to destroy Islam by evoking memories of historical wars and the military invasions of Arab lands by some western countries. In so doing, Alhawali, defames both the West and western civilization which contributes to polarization of the world into two antagonist sides i.e., the west vs. Islam. This, as Baker et al (2013, pp. 130-5) note on the construction of 'the West' and 'the Muslims' by British newspapers, could reify these terms and present them as in conflict with each other.

Having delineated the thematic focus of Alhawali that is indicated by his use of the key term 'WEST', I now turn to see the actual use of the word and its derivatives as used in Ibn Baz's writings.

5.3.2 The key term 'WEST' in Ibn Baz's corpus: Protection of Muslim society

The key term 'WEST' occurs only 10 times in Ibn Baz's corpus: The word form 'الغرب' ('alġarb; the West) occurs 6 times whereas each of the other word forms, 'الغربية' ('alġarbiyyah; western -adj. fem) and 'الغربي' ('alġarbi; Western -adj. masc.), occur twice. Let us examine examples of each word form.

^{53.} For more information about Al-Andalus, see the link below: http://unesdoc.unesco.org/images/0011/001144/114426eo.pdf.

Concordance 5.4 Concordance lines of the word form 'الغرب' ('algarb; the West) in Ibn Baz's corpus

N

3

Concordance

نقد نكرنا من الأدلة الشرعية و الواقع الملموس ما يدل على تحريم الاختلاط و اشتراك المرأة في أعمال الرجال ما فيه كغاية [...] ، و لكن نظرا إلى أن بحض الناس قد يستغيدون من كلمات رجال **الغرب** و الشرق أكثر مما يستغيدون من كلام الله و كلام رسوله صلى الله عليه و سلم و كلام علماء المسلمين ، رأينا أن ننقل لهم ما يتضمن اعتراف رجال **الغرب** و الشرق بمضار الاختلاط و مغامده لعلهم يتتنعون بذلك [...]

Verily, I have mentioned enough Qur'anic verses and Prophetic traditions that prohibit the mixing between women and men (in offices and jobs), but due to that some people (Muslims) may learn from what the men of <u>the West</u> or the East have said more than learning from the words of Allah and the messenger of Allah, may Allah send His blessings and peace upon him, the words of Muslim religious leaders, I felt the need to report some of the writings of the men of the <u>West</u> and the East that include confessions of harming effects and heinous acts of the mixing between men and women that may convince them (some Muslims) [...]

و نتيجة لتلك ألقلاقل التي نشأت في المجتمعات في كل مكان [المشاكل التفسية والاجتماعية] عند غير المسبلمين نتيجة ضعف الأيمان]، و نشأ عنها تصرفات عجيبة من الشباب و غيرهم في **الغرب** و الشرق، بعضها يضحك التكلي، و شر البلية ما يضحك، [...]

And as a result of those disturbances [psychological and social problems among non-Muslims who have no faith] that evolved in every society. Additionally, some strange behaviours evolved among young men and others in the <u>West</u> and the East, some of which may even cause the mother who is bereaved of her child to laugh, and indeed the worse of distress is what may cause laughter [...]

قل عز و جل : {يا أيها الذين آمنوا لا تُأكُوا أمَوَاتُكُمُ بِأَتَاظِلَى [...] فهذا يبين لنا أن نظام الإسلام في المل نظام متوسط ، لا مع رأس المال العائم من <u>الغرب</u> و أثباعه ، و لا مع التيوعيين الملحدين الذين استباحوا الأموال و أهدروا أهلها ، لم يبالوا يهم و استعدوا التعوب و قضوا عليها [...] The Exalted in Mighty and the Majestic said: {O ye who believe! Eat not up your wealth among yourselves in vanity} [...] thus these (rules in the Qur'anic verses and prophetic traditions) indicate that the Islamic financial system is a moderate system. It neither aligns itself with the unjust capitalism of the <u>West</u> and its advocates nor with the communism of the atheists who made lawful to appropriate the wealth of people and ruin them. They even did not care about them, they enslaved nations and ruin them [...]

Ibn Baz utilizes the word form 'الغرب' ('alġarb; the West) to positively represent Islam as a religion that protects society. In Line 1 (Concordance 5.4), Ibn Baz tries to lend support to his argument about his rejection of what he calls 'mixing between men and women in offices and jobs'. He utilizes the writings of 'cell' (riğāl 'alġarb; men of the West) as well as writings of the East (implied to be non-Muslims also), noting that such writers have noted harmful effects when men and women work together. Ibn Baz thus selectively refers to writers who are against sexual equality. The West is constructed here as harmful to people then because of the potential transfer of its practices (such as sexual equality) to the Arab and Muslim countries.

Line 2 shows another way that Ibn Baz conceives of Islam as a guide to stable life whereas, according to him, people in the West or the East (who are not Muslims and lack faith) engage in strange behaviours that can cause 'a bereaved mother to laugh'⁵⁴, for

^{54.} This is an Arabic idiom that is used when strange things happened.

example. However, reading the co-text of concordance line 2, Ibn Baz does not provide any further information about the strange behaviors that he refers to, nor does he give any concrete evidence to back up the claim about such behaviors. On the contrary, Ibn Baz uses the idiomatic phrase 'make bereaved mothers to laugh' which means that even a mother who is so sad (because she lost a child) may laugh at the behaviors of the people in the West that Ibn Baz claims to be strange and funny. In Line 3, Ibn Baz refers to the financial system in Islam. He positively represents the Islamic fiscal system and negatively represents capitalism and communism. He claims that capitalism is unjust and communists confiscate the wealth of people whereas the Islamic financial system protects people's wealth. The other word forms of the key term 'WEST' are also used by Ibn Baz in a defensive mode against what he claims to be possible dangers that threaten Muslims (see Concordance 5.5).

Concordance 5.5 Concordance lines of the word forms 'الغربي' ('alġarbi; Western –adj. masc.) and 'الغربية' ('alġarbiyyah; western –adj. fem) in Ibn Baz's corpus

Ν	Concordance
1	[] بل إن أحد الكتاب من هذه العصابة [التوميين العرب] وجد الوجه الذي يطلع به الناس ليتول : إن الإسلام جنى على العروبة ، و أن اللغة العربية قد انتشرت أبعد مما انتشر الإسلام ، و أن الإسلام - لأنه عالمي - ضار بالتومية العربية, وظاهر أن هذا الكلام بقطع النظر عن بصلائه ، إنما يروج لحساب الاستعمار الغربي منه و الشرقي على السواء ،[] even one of the writers who belongs to this gang [Arab nationalists], was shameless enough to say that Islam incurred bad consequences on Arabism and Arabic had spread widely far more than Islam. He also claimed that Islam -because it is international- is pemicious to Arab nationalism. Aside from the contradictory nature of what he said, this writer makes propaganda for both <u>Western</u> and Eastern colonialism. []
2	[] اختارت جماعة منهم معن انطلى طبيع سحر هذه الحضارة؛ لإكمال تعليمهم في الخارج في الجامعات الأوربية و الأمريكية و غيرها حيث يواجهون هنك بسلسلة من التبيهات و التربيوات على أيدي المستشرقين و الملحبين [] و حيث يواجهون الحياة الغريبية بما فيها من تتسخ و تبذل و خلاعة و نتكك و مجون و إباحية [] و الواجب التحبيبيق من نطاق الابتعات إلى الخارج و حصره في علوم معينة لا تتوافر في الناخل [] و الناخل [] a group of them (Saudis) who were under the fake charming influence of this civilization (Westem civilization), chose to pursue their education abroad in European, American and other universities where they are faced with a chain of lustful desires and doubts created by orientalists and atheists [].
	Additionally, they (Saudi students) will face the <u>Western</u> life that contains degeneration, debasement, wantonness, shamelessness, fragmented social bonds and licentiousness [] thus it is obligatory duty to restrict sending students abroad to a smaller scale and limit it to certain areas of knowledge that are not available inside (Saudi universities) []

According to Ibn Baz, occupation of Muslim lands by Western (and/or Eastern) powers is a threat which made him claim that the writings of Arab nationalists' are mere propaganda that promotes colonialism (Line 1). In Line 2, Ibn Baz presents Western life negatively. He claims

that degeneration, debasement, wantonness, shamelessness, social fragmented bonds and licentiousness are all features of Western life that may affect Saudis who study abroad. Thus he calls for limits on sending students to study abroad, saying they should only study sciences that are not available in Saudi universities (he does not expand on why such subjects are taught in western countries but not Saudi Arabia).

Therefore, the two Salafi writers hold more or less negative conceptions about the West. However, these conceptions are crucially different in one way. While Alhawali, from a politically oriented Salafi perspective, extensively condemns the West and western civilization based on his claims about the eternal enmity between the West (including the Crusaders –see below) and Islam, Ibn Baz's conception (from a religious Salafist perspective) of the West is less obvious and mostly limited to his view that westerners engage in social behaviours or ideologies that he deems to be wrong and against the teachings of Islam. Ibn Baz's pointing out Western/Eastern colonialism (Concordance 5.5, line 1) is an indication that he is also able to construct the West as dominating Arab countries. However, only one case like this was found, compared to the many cases in Alhawali. Let us now turn to examine the other keyword in Alhawali that refers to external threats, i.e. 'ucut' ('isra'īl; Israel).

5.3.3 Israel in Alhawali's corpus

The other keyword that indicates external threat is 'إسرائيل' ('isra'īl; Israel). It occurs 83 times in Alhawali and 7 times in Ibn Baz. This keyword does not have any collocates in Alhawali which may indicate that he uses this keyword to refer to a range of different contexts (fragmented semantic foci). Alhawali also uses other word forms of the key term 'ISRAEL' viz., 'الإسرائيلية' ('al'isra'īliyyah; Israeli – def. adj. fem. 7 times) and 'الإسرائيلية' ('isra'īliyyah; Israeli – indef. adj. fem. 4 times). However, the actual use of these word forms and their thematic function in Alhawali could be revealed via including more co-textual information

using concordance-based analysis. Concordance 5.6 shows some selected examples that

indicate how the keyword is most typically used by Alhawali.

Concordance 5.6 Concordance lines of the keyword 'إسرانيل' ('isra'īl; Israel) in Alhawali's corpus

	(<i>Isra II</i> , Israel) in Alnawali's corpus
N	
	[جريدة التبس الكويتية] إن التقارير التي ترد من واستطن إلى وزارة الخارجية في إحدى النول العربية تؤكد أن الخطة اليهودية تقصى بأن تكون إسرائيل و قبل عام (2010م)
	أكبر دول المنطقة بعد مصر من حيث الإمكانات الديمغرافية (السكانية)، فليس اليهود السوفييت وحدهم هم الذين يتدفقون على الدولة العبرية بل إن الفائض الأمريكي سيتجه
1	[the Kuwaiti 'Alqabas Newspaper'] the reports, that come from Washington to a Ministry of Foreign Affairs of one of the Arab countries,
	confirm that the plan requires that before 2010, Israel should be the largest country in demographic capabilities (population) after Egypt,
	because not only the Russian Jews will flow into the Hebrew State (Israel), the American overpopulation will go there too, []
	[EIR FEETURE] تخطط حكومات أمريكا و روسيا و بريطانيا لحرب عربية إسرانيلية جديدة في الشرق الأوسط و يريد السوفييت و الأمريكان و الإنجليز قيام تحالف
	عسكري سياسي بين إسرائيل و سوريا []، و رغماً عن صدور بعض الإشارات و الدلالات البعيدة عن هذا الهدف إلا أن السياسات الانجلو أمريكية و السوفيتية كانت
	تستهدف منذ مدة طويلة قيام دولتي إسرائيل الكبرى والموريا الكبرى والتهيئتهما لتصبيرا قوى إقليمية كبرى في منطقة التلوق الأوسط.
2	[EIR FEETURE -spelling in the original] The U. S., Soviet, and British governments are planning a new Arab-Israeli war in the Middle
	East. The Soviets and Anglo-Americans want to make a military and political alliance between Israel and Syria [] Despite appearances
	to the contrary, it has been longstanding Anglo-American and Soviet policy to make a Greater Israel and a Greater Syria into the Middle
	East's regional superpowers.
	[] إن أمريكا أن تسمح أبدأً لأعداء إ سرائيل الذين أقسموا على النيل منها بتحقيق أغراضهم. و يتابع نيكسون مفاخراً بإسرائيل كأشد يهودي في العالم تعصباً لقد بهرت
	<u>إسرائيل</u> العالم كله بكل ما أنجزته خلال 40 عاماً من الحرب و سندهت العالم بما تستطيع أن تنجزه في 40 سنة من السلام ص-292.
2	[] United States of America will never allow the enemies of Israel who vowed to attack it (Israel) to achieve their goals, then Nixon
5	continues to boast, as if he were the most zealous partisan Jewish in the world, saying: Israel astonished the world with what it
	accomplished in forty years of war, it will astonish the world with what it can accomplish in forty years of peace (Nixon 1988, p. 292 -
	page number is in the Arabic translation)
	إن أي مزرخ أو باحث يلقى نظرة سريعة فاحصبة على الحقبة التي تسهدت ميلاد الدين النصراني، سيرى أن منطقة حوض البحر الأبيض كانت تموج بعقائد و أفكار متباينة
	نذكر منها: 1- الديانة اليهودية: و هي ديانة مغلقة خاصبة بأسباط بني إ سرائيل؛ لكنها تتميز بأنها ديانة سماوية لها كتاب مقدس، و موطنها فلسطين، حيت ولد المسيح وأرسل.
	Verily any historian or researcher who investigates the era that witnessed the birth of Christianity would clearly see that the
	Mediterranean Basin was replete of heterogeneous faiths and ideas such as: 1- Judaism: It is a closed religion for the Children of Israel,
	but its distinctive feature is that it is Divine religion (an Abrahamic religion) that has a Holy Book. It existed in Palestine where Christ was
L	1

With the exception of Line 4, all the lines indicate that Alhawali, utilizing quotes from different sources, uses the keyword 'اسرائيل' ('isra'īl; Israel) to refer to a possible military threat to Arab countries. In line 1, Alhawali utilizes a quote from the Kuwaiti Alqabas newspaper to lend support to his claim about Israel being a growing threat to the Arab countries. In line 2, Alhawali uses a quote from 'EIR FEETURE'⁵⁵. He also mentions 'a new Arab-Israeli war as being planned by the 'the Russian and the Anglo-American superpowers (see below). In these 3 lines, Alhawali describes a claimed conspiracy against Arab countries. He claims that the superpowers are planning to kindle a new war in the Middle East which

^{55.} The questionable spelling of 'EIR FEETURE' exists in the original text where this concordance line is found. After I searched for this magazine I found that it most likely refers to the Feature articles section in the '*Executive Intelligence Review*' which was founded in 1974 (see also footnote 93 in Chapter 7).

would result in making two superpowers there: Greater Israel and Greater Syria. Alhawali also (Line 3) quotes Nixon (the 37th president of the United States) reporting that 'the United States of America is protecting Israel' which, according to Alhawali as well as most Saudis, is an enemy. In the last line, Alhawali positively refers to the historical appellation 'the children of Israel' representing Judaism as one of the Divine religions (Abrahamic religions).

The keyword 'السرائيل) '(*'isra'īl*; Israel) is used predominantly (74 out of 83 times) in one of Alhawali's texts that was published during the Iraqi invasion of Kuwait in 1990. Alhawali was against the presence of the American-led Allies troops in Saudi Arabia claiming that they came with huge numbers to destroy Islam and Saudi Arabia. Alhawali uses 'السرائيل)' (*'isra'īl*; Israel) as a possible threat that would collaborate with the West to destroy Islam (see Section 4.2.5 in Chapter 4). It is noteworthy here that after the defeat of the Iraqi troops none of the claims of Alhawali has ever happened. However he did not say or write statement showing his false conceptions till the moment of writing this thesis. The theme that Islam is being targeted by the West is recurrent in Alhawali's writings (see Chapter 4, Chapter 6 section 6.3 and Chapter 7 section 7.5.3, as well as below). The definite adjective '(*'al'isra'īliyyah*; Israeli) collocates with '*iliquowāt*; the (military) forces) in one text of Alhawali's writings that I mentioned above (MI score 11.510) which shows that this word form is also used by Alhawali as an indication of an external threat. Let us now consider concordance lines of the definite adjective '*'uuuuuu*' (*'al'isra'īliyyah*; Israeli) (Concordance 5.7).

Concordance 5.7 Concordance lines of 'الإسرائيلية' ('al'isra'iliyyah; Israeli) in Alhawali's corpus

N

Concordance

[EIR FEETURE] لا تتمكن القوات الإسرانيلية من إختراق الخطوط العربية من فترة 2/3 ساعات بالمقارنة مع فترة 6 ساعات التي إستغرقتها القوات الإسرانيلية لإختراق الخطوط المصرية عام 1967م، و ستعد القوات الإسرانيلية في باقي الفترة إلى تطبير الإقليم الأردني،

[EIR FEETURE] Israeli forces would break through Arab lines in two to three hours, compared to the six hours it took the Israeli forces to crack Egyptian lines in 1967. The Israeli forces then, in the rest of the period, will aim to mop up the Jordanian territory,

EIR FEETURE] و لن تحاول القوات الإسرائيلية التجاوز إلى عمق العراق أو دخوله مطلقاً حيث من المتوقع قطع خطوط <mark>القوات الإسرائيلية</mark> في الأردن، إضافة إلى أن إ**سرائيل** ليست لديها القدرة على التعامل مع المناطق العراقية المأهولة بالسكان، و ربما ترى إ**سرائيل** مناسب توجيه ضربة عسكرية لحقول النفط العراقية شمال العراق.. الخ.

2 [EIR FEETUR] Israeli forces will try to penetrate very far into Iraq, if at all, since their (Israeli forces) supply lines could be cut in Jordan. Israel has no capacity to deal with Iraq's populated areas. On the other hand, an Israeli force may choose to strike the oil fields in northern Iraq etc..

[نشرته صحيفة صائدي تايمز في عدها الصادر في 9/2/1975م] أنجز مجلس الأمن القومي في الولايات المتحدة دراسة تفصيلية لخطة سرية للداية وضعتها وزارة الفاع نغزو حقول النفط السعودية في حال نشوب حرب أخرى في الشرق الأوسط ينجم عنها حظر نفطي عربي جديد.[...] تقود هذا الهجوم تسع كتائب متساة محمولة جواً من قاعدتها في ولاية نورت كارولينا و بحماية جوية إلى الظهران في الخليج عن طريق ا<u>لقاعدة الجوية **الإسرانيلية** في</u> "حسريم" و تستولي كتائب المشاة على حقل النفط في الظهران [...]

3 [published in Sunday Times in 9/2/1975] The Security national Council published a detailed study of a top secret plan made by the Department of Defense (the Pentagon) to invade the Saudi oil wells in case another war starts in the Middle East that could lead to a new Arab oil embargo, [...] This attack is lead by a group of nine infantry battalions carried by air from North Carolina to Dhahran (a major administrative center for the Saudi oil industry located in Eastern Province of Saudi Arabia) protected by the <u>Israeli Air Force</u> <u>Base</u> in Hatzerim (an Israeli Air Force Base located in al-Naqab (Negev) desert). The nine infantry battalions will seize the oil field in Dhahran [...]

In the first 2 lines of Concordance 5.7, the collocate 'القوات' ('alqowāt; the (military) forces) precedes the word form 'الإسرائيلية' ('al'isra'īliyyah; Israeli –def. Adj.). These two words as well as 'Israeli Air Force' in Line 3 refer to the Israeli military capabilities that are introduced by Alhawali to indicate what he claims to be a direct threat. Here Alhawali writes about some war scenarios that he claims happened during the presence of the American-led troops in Saudi Arabia (during the Gulf Crisis in 1990). Alhawali also uses the indefinite adjective 'isra'īliyyah; Israeli) 4 times in the same manner. He uses this word 3 times to form the phrase 'isra'īliyyah; Israeli) 4 times in the same manner. He uses this word 3 times to form the phrase 2 in Concordance 5.6 above to instigate the Saudis against what he believes to be Western (and/or American) domination over the 'Muslim world'.

Ibn Baz, on a smaller scale, also uses the key term 'ISRAEL' which will enable me to compare the occurrences of the key term 'ISRAEL' in the two corpora, as demonstrated in the following section.

5.3.4 Israel in Ibn Baz's corpus

The word 'اسرائيل' ('isra'īl; Israel) collocates with 'بني' (bani; the children of) (MI score 12.923). Both words indicate a religious appellation of the Israelites (ancient Hebrew people). Ibn Baz also uses other forms of the word 'اسرائيل' ('isra'īl; Israel). He uses the word ''(isra'īliyyah; of Israelites) (one time) to indicate Jewish narratives. He also uses the word form 'اسرائيلي' ('isra'īli; Israeli) (one time) to indicate a member of old Hebrew people. Unlike Alhawali, Ibn Baz uses the key term 'ISRAEL' mostly in connection with religious issues. However, concordance-based analysis could reveal more about Ibn Baz's use of the term (see Concordance 5.8).

Concordance 5.8 Concordance lines of 'إسرائيل' (*'isra'īl*; Israel) in Ibn Baz's corpus

	Concordance 5.8 Concordance lines of 'إسرائيك' ('isra'il; Israel) in Ibn Baz's corpus		
N	Concordance		
	كتب آخر من هذه العصابة (القوميين العرب) يطلب منا بإلحاح : أن ننسى التاريخ ؛ [] و نسى هذا الغر أن اليهود في كبد الشرق الأوسط ، أقاموا دولتهم بإمداد من التاريخ		
	لموحى ، و أنهم جعلوا اسم إ سرائيل علما عليها ، إنه حلال للناس جميعا أن يستصحبوا كاريخهم في كفاحهم ، أما نحن المسلمين فحرام علينا أن نذكر فصلا من هذا الكاريخ []		
1	Another writer who belongs to this gang (Arab nationalists) insists that we forget the history (of Islam) []. This inexperienced heedless person		
	has forgotten that the Jews, inspired by their history, established their state and made the name Israel to mark it. Is it lawful for all people, in		
	their struggle, to remember their histories whereas it is not allowed for Muslims to remember even one chapter of their history []		
	[] فتال عز وجل: {لُعِنَ الذين كَفَرُوا من بَنِ ين إسْرَائِيلَ على لِسُنَنِ نَاوُدُ وعِيسَى ابْنِ مَرَّئِهَ ذلك بما عَصَوُّا وكَتُوا تِحَدُّونَ * كَنُوا لا بِئَنَاهُوْنَ عن مُنْكَرٍ فَطُوهُ لَبِنْسَ ما كَنَوا تِفْعَلُونَ}.		
	ر لا يرضي مسلم صحيح العقيدة سليم الإيمان بريه أن يتصف بعمل كفار بني إسرائيل في عدم إنكار المنكر []		
	[] and then the Exalted in Mighty and the Majestic said: {Those among the Children of Israel who disbelieved were cursed by the tongue of		
2	Dawud (David) and Isa (Jesus), son of Maryam (Mary). That was because they disobeyed (Allah and the Messengers) and were ever		
	transgressing beyond bounds. They used not to forbid one another from Al-Munkar (wrong, evil-doing, sins, polytheism, disbelief) which they		
	committed. Vile indeed was what they used to do}. Thus no Muslim who has true faith in his Lord to commit the sin made by the disbelievers		
	from among the Children of Israel who did not reject Al-Munkar (wrong, evil-doing, sins, polytheism, disbelief) []		
	[] و صدق فيهم قوله سبحانه فيما ذكر في بني إسرائيل : {و جَعَلْنًا منهُمُ أَنِّمَةٌ تَبَدُّرنَ بِأَمْرِنَ لِمَّا صَبَرُوا و كَتُوا بِآيَتِيَّا يُوقِنُونَ } صدق هذا في أصحاب الرسول صلى الله عليه و		
	سند و فيمن سار على سبيلهم ، صناروا ألمة و هناة و دعاة للحق ، و أعلاما يقتدى بهم []		
,	[] the words of the Glorified about the Children of Israel became true about them (the companion of the Prophet of Islam): {And We made		
1,2	from among them (Children of Israel), leaders, giving guidance under Our Command, when they were patient and used to believe with certainty		
	in Our Signs). This Qur'anic verse is true about the companions of the messenger, may Allah send His blessings and peace upon him, and those		
	who followed them. They became Imams and teachers of the truth (Islam) who were distinguished men who inspired Muslims []		
	[] وتنسير القومية العربية هذا التفسير الكفور (إن الإسلام ليس إلا نهضمة عربية، فاز بها هذا الجنس العظيم في القرون الوسطى)، هو حرب أخرى صد الإسلام ، إنه		
	لجدير أن يتسمى هؤلاء بأتباع التومية الجريبة لا العربية. أليسوا يعملون لمصلحة الإستعمار و <u>إسرائيل ؟</u> []		
4	[] And such rejectionist and ungrateful interpretation [of Islam] by Arab nationalists is another war against Islam (Islam is only an Arabic rise		
1	achieved by this great race [Arabs] in the medieval centuries). Verily these (Arab nationalists) are supposed to be named the advocates of		

Hebrew rather than Arab nationalism. Aren't they work for the interests of colonialism and Israel ? [...]

In Lines 1 and 4, Ibn Baz uses the key term 'ISRAEL' to refer to modern Israel whereas in concordance lines 2 and 3, he uses the key term 'ISRAEL' as an appellation that refers to the ancient Hebrew people as mentioned in the Qur'an. In line 1, he quotes from an Arab nationalist who apparently calls for people to forget the history of Islam. The case of Israel is

held up as an example of how other religions have benefited from remembering their history, while Ibn Baz contrasts Islam, saying that Muslims are not even allowed to remember one chapter of their own history. Islam is thus constructed as oppressed, in comparison to other religions like Judaism.

The semantic focus in Line 2 is about 'الأمر بالمعروف والنهي عن المنكر' ('al'amr bilma'rof' wannahi 'an almunkar; commanding right and forbidding vice) as an Islamic doctrine mentioned in the Qur'an. Ibn Baz uses the Qur'anic verse about the behaviour of the disbelievers among the Children of Israel in order to legitimate his stance about commanding right and forbidding vice. Line 3 praises the companions of the Prophet utilizing what he believes to be the likeness between them and those pious religious leaders among the Children of Israel who were praised in the Qur'an. Finally, in Line 4, Ibn Baz associates modern Israel with colonialism⁵⁶ which is a clear indication that he considers modern Israel as an enemy.

The two Salafi writers use the key term 'ISRAEL' in different political and religious thematic foci. On one hand, Alhawali utilizes the different word forms of 'ISRAEL' to construct what he claims to be a possible external (Israeli as well as American) threat to Saudi Arabia. As I have indicated above, the word forms of 'ISRAEL' occur mostly (74 out of 83 times) in one of his texts that was published and distributed during the Gulf Crises in 1990. On the other hand, Ibn Baz uses the word forms of the key term 'ISRAEL' in different theological semantic foci. Less often, Ibn Baz uses 'ISRAEL' to construct an external threat (Israel and colonialism) using the phrase 'IŞRAEL' to construct an external threat another war on Islam), as in Concordance 5.8, line 4. This leads us the next thematic focus as indicated by the keywords in Table 5.3.

^{46.} Colonialism, in my view, is used as a euphemism for military aggression and occupation of foreign lands which resulted in atrocities, genocides and oppression of natives.

5.4 External threats 2: Military invasion and the Crusader States

The selected keyword that indicates military threat in Alhawali (Table 5.3) is 'الحرب' ('alḥarb; the war), (112 occurrences). This word is also used in Ibn Baz, on a much smaller scale (9 times). There are 8 possible word forms in Arabic of the key term 'WAR': 'الحرب' ('alḥarb; the wars –sing. def. n.), 'الحروب' ('alḥarūb; the wars –pl. def. n.), 'حرب) ' ('alḥarūb; the wars –pl. def. n.), 'حرب) ' ('alḥarūb; war –indef. sing. n.), 'حرب) ' ('alḥarūb; wars –indef. pl. n,), 'حرب) ' (harbi; military/warlike – indef. masc. adj.), 'حربية' (harbiyah; the military/the warlike – def. masc. adj.), 'الحربي) ' ('alḥarbi; the military/the warlike – def. masc. adj.), 'حربية' (harbiyyah; military/warlike – indef. fem. adj.), and 'الحربية' ('alḥarbiyyah; military/warlike – def. fem. adj.). Wordsmith allows for the possibility to search for all these word forms at one time by adding asterisks at the beginning and the end of the root as follows: *حرب * (*harb*; *WAR*). Table 5.5 shows all word forms of the key term 'WAR' and number of occurrences in both Salafi corpora.

N	Word form	Eng. trans	Alhawali	Ibn Baz
	الحرب	the war	112	9
2	الحررب	the wars	47	4
3	حرب	war	45	4
4	حروب ا	wars	б	2
5	الحريبة	the military/ the warlike	3	0
6	حربى ا	military/ warlike	1	0
7	الحربى	the military/ the warlike	11	00
ТО	TAL		215	19

Table 5.5 Number of occurrences of the key term 'WAR' in both corpora

The only word form of the key term 'WAR' that is not used in Alhawali is 'حربية' (harbiyyah; military or warlike –indef. fem. adj.). The total number of the occurrences of the key term 'WAR' is almost 11-fold greater in Alhawali than it is in Ibn Baz (215 and 19 occurrences respectively). The first three word forms in Table 5.5 occur more prominently in Alhawali (and on a smaller scale in Ibn Baz) than the other word forms of the key term 'WAR'. An initial observation of Table 5.5 is that the most frequent word forms of the key term 'WAR' in Alhawali's corpus are the definite⁵⁷ nouns 'الحرب' ('alḥarb; the war – n. sing.) and 'الحروب'

^{57.} Definiteness in Arabic is indicated by attaching the prefix 'l'' ('al; the) to the noun '(harb; war).

('alhor $\bar{u}b$; the wars – n. pl.) which occur 112 and 47 times respectively, whereas the indefinite forms of the key term 'WAR' are less frequent in Alhawali's corpus; ('----') (harb; war) 45 occurrences and '----- (har $\bar{u}b$; wars) 6 occurrences). This could indicate that Alhawali refers to specific war(s). This observation is further supported by investigating collocates and clusters of the key term 'WAR'. Let us first examine threat indicating words in Alhawali's corpus.

5.4.1 Wars and military invasion in Alhawali

The word forms of the key term 'WAR' in Alhawali collocate with 'لعالمية' ('al'alamiyyah; world) (MI score 9.515 and joint freq. 23) and 'الصليبية' ('aşalībiyyah; pertaining to the Crusades) (MI score 10.997 and joint freq. 25).⁵⁸ Both collocates are keywords in Alhawali's corpus (with keyness scores of 42.272 and 43.81 respectively). Having keywords collocating with each other may indicate the pervasiveness of the theme that these keywords together indicate i.e. 'World wars and the Crusades'. The definite plural form of the key term 'WAR', namely 'الحروب الصليبية' ('alḥorūb; the wars) is part of the phrase 'الحروب ('alḥorūb 'aşalebiyyah; Crusades) which is also a 2-word cluster in Alhawali (25 occurrences in four of Alhawali's texts) (note that I only include the clusters that occur in three texts or more). Table 5.6 shows the clusters that include the singular definite noun 'الحروب' ('alḥarb; the war).

Table 5.6 Clusters that contain the singular definite noun 'الحرب' ('alharb; the war) In Alhawali's corpus

N	Cluster	Eng. trans.	Freq.
1	الحرب العالمية التانية	World War II	13
2	الحرب العالمية الأولى	World War I	9
3	بد الحرب العالمية	after the World War	7
4	الحرب العالمية	World War	23

As indicated by Table 5.6, the keyword 'الحرب' ('*alḥarb*; the war) occurs in 4 clusters indicating the first and second World Wars. However, a concordance based analysis shows more about the actual use of this keyword (Concordance 5.9).

^{58.} I applied the criteria in Section 4.2.1.1 and only one collocate is used for each word form of the key term 'WAR'.

Concordance 5.9 Concordance of the keyword 'الحرب' ('alharb; the war) in Alhawali's corpus

N Concordance المحاصة محاملة. [...] تم جاءت الروح الصليبية كامنة في مدافع نابليون، التي تضبعنا على أول محاولة ضخمة من أوروبا الحديثة لاستثمال تدفقه الإسلام [...] كما أن الواقع التاريخي يؤكد أن لاده ا**لحرب** لن تنتهى، و أن الذي خدع بعض المستغلِّنين هو اختلاف فصولها و مضَّاهر ها 1 [...] then the spirit of the Crusades appeared in the army of Napoléon, which shows us the first largest attempt to eradicate Islam [...] the historical reality confirms that this war [between Europe and Islam] will never end, and those who were made inattentive were deceived by the new shape this [war] took. [...] في الحرب العالمية الأولى كانت البداية عندما التق فيما يسمى التاقية سايكس بيكو على تصبيم الخلافة الحكماتية بين دول الغرب، فجاءت الحركة التومية العربية وجيئت جود سُأَ، وحاربت مع الإنجليز ضد الدولة العثمانية.[...] وبانتهاء الحرب العالمية انتهت الخلافة العثمانية تماماً، وتمارق العالم الإسلامي [...] [...] the start was during World War I when the Western countries made the Sykes-Picot Agreement to divide the Ottoman 2 Caliphate between the Western countries, then the Movement of Arab Nationalism started to mobilize people (armies) and assisted the English troops against the Ottoman Empire [...] with the end of the World War, the Ottoman Caliphate vanished completely, and the Muslim World was ripped apart [...] [...] و في شهر مارس 1976م أي قبل 15 سنة إلا تليلاً تشرت جريدة الأتباء الكويتية ترجمة لمقال تشر في إحدى المجانب الأمريكية. [...]". ابتدأ المقال بتأكيد أنه ليس أمام تحكومة الأمريكية إلا خياران: إما الانتهيار الإقتصادي، و إما الاستيلاء على العملكة العربية السعودية [...]"!! . تم يقول: بعد الاستيلاء على السعودية محكمات مجرمي الحرب و سنطيق قرانين العقوبات التقليدية في البلاد [...] 3 [...] nearly 15 years ago, in March 1976, the Al-Anba Kuwaiti newspaper published a translation of an article appeared in one of the American magazines, [...] the article started with a confirmation that the American government has only two choices: either an economic recession or invading Saudi Arabia [...]. Then it [the article] says: after invading Saudi Arabia we will start a legal prosecution of the war criminals and apply the traditional laws in that area [...] [...] قبل أقل من إسبو عين نشرت مجلة (EIR FEE TURE) مقالات في غاية الخطورة ننقل ترجمة بعض ما جاء فيها: [...] إن هذه الدول[أمريكا و روسياً و بريطانيا] تأمل في حدرت أزمة بترونية جديدة كفلك التي حنثت في عام 1976 [...]. و ربما كان أحد أهم أهداف ا<u>لحرب</u> هو خلق سابتة استخدام الأسلحة الكيمارية و التروية. (في حرب تحرير الكويت 1990- 1991) و في هذا الخصوص يجب عدم تجاهل المؤسسات العنصرية السوفيتية و البريطانية و الأمريكية التي تنظر بتلق تديد نحو التكاتر السكاتي العربي [...] [...] we will report the translation of some extremely serious articles published by EIR FEETURE' nearly two week ago: [...] These 4 countries [America, Russia, and Britain] hope that a new Oil Crisis, like the one in 1976, happens [...]. And one of the most important goals of the war might be using nuclear and chemical weapons (in the Gulf War 1990-1991), and in this respect we should not ignore the Soviet, British and American racist institutions that are extremely worried about the increasing growth of Arab population [...]

Alhawali conceives of the modern wars that are apparently intended by the Europeans to dominate the Arab World as a continuation of the medieval Crusades (Line 1) whose aim, according to Alhawali, is to eradicate Islam. Alhawali then claims that the collaboration between the Arab nationalistic movement and English troops resulted in the elimination of the Ottoman Caliphate and consequently the 'Muslim World' was ripped apart (Line 2) and divided between the French and the English colonials according to the Sykes-Picot Agreement but Alhawali does not mention that this was in response to the Ottomans declaring war against Britain and France. In Lines 3 and 4, Alhawali continues to victimize and represent the 'Us' (the Gulf countries) by selectively reporting translated articles (as a form of legitimation). Introducing quotes from such translated articles, Alhawali claims to make clear that the aims of the Gulf war are to invade Saudi Arabia and possibly use

chemical and nuclear weapons by the 'Other' (Allied Western forces). Alhawali thus utilizes the keyword 'الحرب' ('alḥarb; the war) to lend support to his claims about the heedlessness of Muslims and the hostility of the West whose wars are mere continuation of the Crusades to eradicate Islam which is further consolidated by actual use of the other keyword 'الحروب' ('alḥorūb; the wars) and its collocate 'الصليبية' ('aṣalebiyyah; pertaining to the Crusades), which is also a keyword for Alhawali.

The theme of 'the Crusade and Crusader States' becomes obvious in the cluster that combines 'الحروب' ('alḥorūb; the wars) and 'الحروب' ('aṣalebiyyah; pertaining to the Crusades) in Alhawali (both are used in Arabic to indicate 'the Crusades'). Concordance 5.10 shows some representative concordance lines of the collocate 'inclusion' ('aṣalebiyyah; pertaining to the Crusades).

Concordance 5.10 Concordance of the keyword 'الصليبية' ('aşalebiyyah; pertaining to the Crusades) in Alhawali's corpus

N	Concordance
	[] إن ولادة أوروبا في ظل الحروب <u>الصليبية</u> وتسعورها بذاتها من خلالها هو الذي يفسر تلك التناقضات الصارخة التي يعيشها الفكر الغربي متمتلة في هذه المعادلات الصحبة العصب صليبي على الإسلام من بطرس الناسك إلى كلاوس.
	[] indeed the birth of Europe and the realization of itself through the <u>Crusades</u> can explain the stark contradictions in the Western thinking which are resembled in the Crusaders' hatred of Islam from the days of Peter the Hermit till
	Claes.
-	إذا تصورنا ذلك [التغيرات التي تريد أمريكا احداثها] أدركنا خطراً كبيراً يهند المنطقة في حالة تدمير العراق و إحلال التحالف الشيعي محله (إيران-
	سوريا-العراق)[] أما باكستان فكثير من قادة جيسها الكبار شيعة و معهم إخوانهم القاديانية و البريلوية). إنها مصيبة عظمي لو أصبح هذا القوس الكبير قوساً رافضياً يهودياً توجهه ا <u>لصليبية</u> الغربية المتحالفة.
2	If we imagined that [the changes that America want to make] we realize the hazard that threatens the Middle East
	when the Iraqi regime is destroyed and replaced by Shiite allied forces (Iran-Syria-Iraq) [] as for Pakistan, most of its military leaders are Shiite supported by their brothers; the Gadyanis (Ahmadis) and Barelvis (Sufis). Indeed it is a
	Calamity if this huge crescent (surrounding the Sunnis) becomes Rafidi (Shiite) and Jewish controlled by allied
	Western Crusaders' States.
	الحروب الصليبية [] كانت خطئها عسكرية بحثة و هدفها تدمير الكيان الإسلامي بالقوة. و الحروب ا لصليبية الاستعمارية كانت خطئها تقوم على هدف القضاء على الإسلام، و لكن بواسطة احتلال أراضيه احتلالاً مباشراً []
3	The plan of the Crusaders [] was of a military nature and its major goal is to severely destroy Islam. But the plan of
	[modern] colonialist Crusaders was to eradicate Islam by directly occupying the land of Muslims []
	إن العلاقة بين البلاد الإسلامية و بين أوروبا خاصبة جصفتها مهد العلمانية. لا يمكن أن تدرس بغير الحروب الصليبية. فهذه الحروب أضغت على
3	لَتُنَّكَ الحداوة الأبنية الراسخة مظهراً جديداً []
-	Indeed the relationship between Muslim countries and Europe -as the abode of secularism- cannot be studied without
	considering the Crusades. These Crusades gave a new shape to the everlasting, deep-rooted hostility []

In Line 1 (Concordance 5.10), Alhawali claims a connection between the Crusaders and modern Western thinking that embodies the hatred of Islam from the days of Peter the Hermit

until Willy Claes who was Secretary General of NATO in the mid-1990s. In Line 2, Alhawali combines the Shiite and Jewish people as controlled by allied Western Crusaders' States. According to Alhawali, the medieval Crusades and what he sees as 'modern' Crusades (colonial powers) have one common aim which is to destroy/eradicate Islam (Line 3). As we have seen before, Alhawali represents Europe as both the abode of secularism (Line 4) and as being home to religious Crusaders whose aim is to eliminate Islam - a contradiction that has the ideological effect of increasing the amount of distance and difference between 'the Muslim world' and 'the West'. Thus, Alhawali makes a strong link between the medieval Crusaders and modern western foreign policy, characterising the two as the same. The West is thus represented as *always* hostile to Muslim countries, both historically and currently. War is something that the West does towards Muslim countries, in other words. Let us move on to examine the war indicating words in Ibn Baz's corpus.

5.4.2 Wars and military invasion in Ibn Baz

The key term 'WAR' is also used in Ibn Baz. He uses only three forms of this key term, viz., 'بحرب' (*harb*; war), 'الحرب' (*'alḥarū*; the war), and 'الحروب' (*'alḥarūb*; the wars) that are represented and recontextualized differently for different legitimation purposes. The first observation about the key term 'WAR' in Ibn Baz is that it has no collocates and is not used in any frequent clusters. This could be due to the limited number of occurrences of this key term (19 occurrences – see Table 5.5). However, a concordance based analysis of 'WAR' could reveal more about how Ibn Baz uses it (concordance 5.11).

Concordance 5.11 Concordance of the key term 'WAR' in Ibn Baz's corpus

N	Concordance
	[] إن هؤلاء الناس [القوميون العرب] يقولون في صراحة: إن الإسلام ليس إلا نهضمة عربية، فان بها هذا الجنس العظيم في القرون الوسطى، []. وتنصير
	القومية العربية هذا التقسير الكفرر. الكفرد، هو <u>هرب</u> أخرى ضد الإسلام، […]
1	[] verily, these groups [Arab nationalists] declare that Islam is only an Arabic renaissance achieved by this great race
	[Arabs] in the medieval centuries, []. And such rejectionist and ungrateful interpretation [of Islam] by Arab nationalists is
	another war against Islam, []
	[] فعليك [الداعية الى الإسلام] أن تحكم تسرع الله في العبادات و في المعاملات. [] و في السلم و ا لعرب ، و مع العدو و الصديق، و في الجذايات و في
	کل شيء، دين الله يجب أن يحکم في کل شيء […]
2	[] thus you [the caller to Islam] should refer to the Sharia of Allah in the acts of worship, business transactions [] in
	peace, during war, with foes, with friends, in capital offences and in everything. The religion of Allah (Sharia) should be
	referred to in every aspect (of our life) []
	[] و لقد استمر التباب المسلم في عطاء الخير المتجدد في ا لحروب الصليبية في الشام و الأندلس و غيرها من المواقف التي يتصادم فيها الحق بالباطل حتى
3	المورم ، []
1	[] and verily, the Muslim youths continued to benevolently exert their efforts during the Crusades in Syria, Al-Andalus
	(Spain) and other situations where where the truth continuously collides with the falsehood, []
	[] و كان تخريب الدرعية و القضاء على دولة آل سعود في عام 1233 هـ. فمكت الناس في نجد في فوضمي، و قتل [] ثم في عام أربعين بحد المالتين و
	ألف اجتمع سَمل المسلمين في تجد على الإمام تركي بن عبد الله بن محمد بن سعود، [] و الطفأت الفتن التي بيلهم بعد ا لعروب الطويلة التي حصلت على أيدي
	المصريين ر أعوالهم []
1	[] and the destruction of Diriyah (the home town of the Saudi Royal Family located) and the toppling down of the Saudi
4	State was in the year 1233 H. (1818), and people in Najd (central region of Saudi Arabia) were in chaos and fighting each
	other [] then in 1240 H. (1824) the Muslims (people in Najd) reunited and gathered around Imam Turki bin Abdullah bin
	Muhammad bin Saud (the founder of the second Saudi State 1824–1891) [] and consequently the discords between
	them ceased after the long wars, that were caused by the Egyptians and their complicit partners (Mohammad Ali Pasha's
	invasion of Najd), []

Unlike Alhawali's recontextualizations of the key term 'WAR' that promote polarization between Western countries and the Muslim world, Ibn Baz uses it to define different threats that target or have targeted Islam and Muslims (Lines 1, 3 and 4). In his refutation of the Arab nationalistic view of Islam (Line 1), Ibn Baz uses the indefinite noun 'حرب' (*harb*; war) in a metaphorical sense considering such a definition of Islam as a war (among other wars) against Islam. The definite noun 'الحرب' (*'alḥarb*; the war) in Line 2 is used to refer to one of the different aspects of life that, according to Ibn Baz, should be governed by the Sharia (in this context, the translation of the Arabic definite noun 'الحرب' (*'alḥarb*; the war) becomes indefinite in English). In this case then, Ibn Baz is not referring to any specific war.

Ibn Baz recontextualizes the definite plural noun 'الحروب' (*'alḥorūb*; the wars) (Lines 3 and 4) to achieve different ends. First, the definite plural noun 'الحروب' (*'alḥorūb*; the wars) is modified by the adjective 'الصليبية' (*'aṣalebiyyah*; pertaining to the Crusades) (Line 3) where both indicate 'the Crusades' that Ibn Baz considers as one of the situations where falsehood

collides with the truth. Expanding Line 3 could reveal more clues of how Ibn Baz utilizes the

'Crusades'.

Figure 5.2 Extended version of Concordance line 3 of the key term 'WAR' in Ibn Baz [...] و الشباب في أي أمة من الأمم هم العمود الفقري الذي يشكل عنصر الحركة و الحيوية إذ لديهم الطاقة المنتجة، و [...] و الشباب في اي أمة من الأمم هم العمود العلوي الذي يسلس طعلس المراحة (سيري با عيهم المسرم المسرم المعرف اللعطاء المتجدد، و لم تنهض أمة من الأمم غالبا إلا على أكتاف شبابها الواحي و حماسته المتجددة. [...] و لقد استمر الشباب المسلم في عطاء الذير المتجدد في <u>الحروب</u> الصليبية في الشام و الأندلس و غير ها من المواقف التي يتصادم فيها الحق بالباطل حتى اليوم، فغاظت تلك الحماسة أعداء الإسلام، حيث سعوا إلى وضع العراقيل في طريقهم، أو تغيير اتجاهيم، [...] أن بتأليب بعض الحكومات عليهم. كل هذا قد يؤدي بالتالي إلى ظهور حركات تتسم بطابع الوقوف من المجتمع و القيدات، موقفاً قاسياً و مصّداً، قد يصل إلى نوع من المواجهة في بعض الأحيان، أو العمل السري الذي قد يخالطه ما يشينه، أو يغير من مجراه الطبيعي Islamic movements and the role youths: [...] and the youths, that are the backbone of any nation, constitute the [basic] element of activity and vivacity because they possess the productive energy and continuous exertion of efforts. Mostly, all nations rose up on the shoulders of their conscious youths and their rejuvenated enthusiasm. [...] and verily, the Muslim youths continued to benevolently exert their efforts during the Crusades in Syria, Al-Andalus (Spain) and other situations where the truth continuously collides with the falsehood. Such enthusiasm vexed the enemies of Islam who vehemently attempted to impede the progress of the Muslim youths or divert the direction of their attention [...] or incite some governments against them. Consequently all these [are reasons that] may lead to the emergence of new movements [among the Muslim youths] that could hold harsh and opposing attitudes towards the society and the leaders. Such attitudes could lead to confrontation or the evolution of secret movements that could be led astray.

In the above extract, Ibn Baz claims that the enemies of Islam have the malevolent intention of destroying Islam which is an indication of Ibn Baz's construction of external threat. The term 'الحروب الصليبية' ('alḥorūb 'aṣalebiyyah; Crusades) is utilized by Ibn Baz to positively represent Muslim youths who, according to him, always benevolently exert efforts that displease the enemies of Islam. In his representation of the 'Other', Ibn Baz anonymizes a group of people (the enemies of Islam) who, according to him, are blamed for Muslim youths taking extreme attitudes towards society and rulers (the representation of social actors will be detailed in Chapter 6).

In Line 4, Ibn Baz uses the plural definite noun 'الحروب' ('alḥorūb; the wars) modified by the adjective 'الطويلة' ('alṭawīlah; the long) to describe some disruptive historical events in Najd (the central region of modern Saudi Arabia) in the years 1818-1824, after the military invasion led by Mohammad Ali Pasha of Egypt. According to Ibn Baz, such long wars ceased after the Muslims [people of Najd], united around one Imam/ruler (Imam Turki bin Abdullah the founder of the second Saudi State 1819 - 1891). Ibn Baz, from a Salafist political perspective, conceives of the Egyptian attacks as destroying the government that is based on the religion of Islam. He recontextualizes the disruptive historical events (1818-1824) to superficially address the early stage of establishing the second Saudi State reducing it to gathering the people (of Najd) by the Imam/ruler. Here, Ibn Baz clearly shows his support to what he believes to be the legitimate Muslim ruler (see Section 8.3.4 for more explanation of the Salafi creed about the Imam/ruler).

To sum up, despite the fact that Alhawali and Ibn Baz mention 'wars against Islam' both writers represent the concept differently. On the one hand, Alhawali tries to construct (an Israeli/western) danger that surrounds Islam and Arab/Muslim lands. He, from a politically Salafist ideological perspective, claims that the West (as well as Israel) not only aim to invade Arab lands but destroy Islam (i.e. instigating the public against what he claims to be western political and military enmity towards Arabs and Islam). He also links the first Gulf Crisis in 1990 (and any possible coming wars) to colonialism and medieval Crusades. On the other hand, Ibn Baz (less frequently) refers to a less clearly-defined set of enemies of Islam which can refer to Arab Nationalists, Crusaders or the Egyptian army of Mohammad Ali Pasha – during his invasion of Najd - and his collaborates). One aspect that both Salafi writers share is that they utilize the key term 'WAR' in historical contexts, which leads us to the next section that tackles the construction of the past in the Salafi corpora.

5.5 The construction of the past

Table 5.3 shows the keyword 'القرن' ('alqarn; century – def. sing. n.) that indicates the category 'Periods of Times' in the Salafi corpora. It is noteworthy here that the word 'القرن' ('alqarn) is polysemous. It could be translated into English as 'the century', 'the generation' or 'the horn' depending on the cotext. However I shall only include the word forms that

indicate periods of time. There are other possible Arabic word forms of the key term 'CENTURY' such as the indefinite singular noun 'قرنین' (*qarn*; century/generation), the indefinite dual nouns 'قرنین/قرنان' (*qarnān/qarnayn*; two centuries/generations –nom. and acc. respectively) and their definite counterparts 'lقرنین/القرنان' (*algarnān/'algarnayn*; the two centuries/generations –nom. and acc. respectively), the indefinite plural noun 'قرنون' (*qorūn*; centuries/generations) and its definite counterpart 'lقرون' (*aqorūn*; the centuries/generations) and its definite counterpart 'lقرون' (*aqorūn*; the centuries/generations). Table 5.7 shows the word forms of the key term 'CENTURY' that both Salafi writers use in their texts. It is worth noting here that I only include the word forms that exist in three or more of the Salafi texts in each corpus. Also, the English translations are based on a preliminary concordance analysis of these word forms.

Table 5.7 Word forms of the key term 'CENTURY' in both Salafi corpora

N	Word forms	Eng. trans.	F. Alhawali	F. Baz
1	القرن	the () century	134	8
2	القرون	the centuries/generations	57	9
3	قرون 🔰	the centuries/homs	26	0
4	قرن	the centuries/generations	10	0

Table 5.7 contains 4 word forms of the key term 'CENTURY' as used by the two Salafi writers in their texts. The most frequent word form is 'القرن' ('alqarn; the century), occurring 134 times in Alhawali and 8 times in Ibn Baz. Let us now consider these word forms in both Salafi texts and how each writer uses them to achieve ideological ends.

5.5.1 The construction of the past in Alhawali: Europe as a source of evil

One of Alhawali's most prominent thematic foci is demonizing the West (see Section 5.2 above and elsewhere in this thesis). In his writings about the past, Alhawali also represents the West. The only word forms of the key term 'CENTURY' that have collocates in Alhawali are 'القرين' ('algarn; the century) and 'القرين' ('algorun; centuries).

There are 11 collocates of the word form 'القرن' ('alqarn; the century) and 6 collocates of the word form 'القرون' ('alqorūn; centuries) (see Table B6 in Appendix B). I shall consider only the collocates 'مطلع' (maţla'; (the) beginning of) that collocates with the word form 'الوسطي' ('alqarn; the century) (MI score 9.6846, Joint frequency 9) and 'القرن' ('alwosţā; (the middle) that collocates with word form 'القرون' ('alqorūn; centuries) (MI score 11.074, joint frequency 37). The rest of the collocates will not be included in the analysis because they occur in less than 3 texts of Alhawali's corpus. The collocates listed above, at face value, bear no positive or negative meanings. Thus I shall turn to concordancebased analysis of the word forms 'القرون' ('alqorūn; centuries) and their collocates (see Concordance 5.12).

Concordance 5.12 Concordance of the word form 'القرن' ('alqarn; the century) and its collocate 'مطلع' (mațla'; (the) beginning (of) in Alhawali's corpus.

N	Concordance
1	أما الفكر فقد تأثر الشيخ محمد عبده بفلسفة كومت المعقلية [] و معلوم تأثر كل زعماء الفكر في مطلع هذا القرن بالشيخ من
	کریب اُن بعید.
	Regarding his thinking the Sheikh Mohammad Abduh (an Egyptian religious leader 1849 - 1905) was
1	influenced by the rational philosophy (positivism) of Auguste Comte (French philosopher 1798 -
1	1857). And at the beginning of this century (20th), it is known that all (Muslim/Arab) intellectual
	leaders were, more or less, under the influence of the Sheikh (Mohammad Abduh).
[و مرت في مطلع هذا القرن حقبة مظلمة، راجت فيها سوق الأفكار الموبوءة و المذاهب المنحرفة، حتى أظهر أعداء الإسلام
	تفاؤلهم بأن هذه الأمة ستلفظ أنفاسها عما قليل.
2	And the beginning of this century (20th century) was a dark time where the poisoned ideas and
}	distorted movements were in vogue, to the extent that the enemies of Islam were optimistic that this
	nation is dying very soon.
	و نذلك فإن المجتمع الغربي منذ مطلع القرن العشرين قد تخلي حقيقة عن الدين [] و أصبح (الدين) لا يعلي سوى واحد من
	أمرين: الطقوس المتحجرة [] أو اللامبالاة الساخرة []
3	Thus, since the beginning of the twentieth century , the western society, in fact, abandoned religion
]	[] The religion [in the West] means either outmoded rituals [] or sarcastic carelessness []
<u> </u>	وصبع القارة الأوروبية في مطلع القرن العشرين: وعلى المستوى العام بقيت في أوروبا إلى مطلع القرن العشرين بقايا من
.	الأوضاع الاجتماعية الموروتة وتسيء من القيم الشاحبة []
4	Europe at the beginning of the twentieth century: Generally, till the beginning of the twentieth
	century, there (in Europe) remained few traditional principles and faded values []

Philosophy, according to Salafis as well as most Muslim (medieval and modern) religious leaders, 59 is not true guidance. Thus Alhawali accuses the Egyptian religious leader (Sheikh Mohammad Abduh) of being under the influence of the French philosopher Auguste Comte's positivism rather than being influenced by the Qur'an and the Sunnah. Alhawali (Line 1) represents both the Egyptian religious leader and western philosophy negatively. Additionally, he attacks all Arab intellectuals (religious or otherwise) who lived at the beginning of the 20th century as being under the influence of Sheikh Mohammad Abdu who was under the influence of Comte. Line 2 shows Alhawali's resentment of political and social movements in the Arab/Muslim countries at the early stage of the twentieth century. He claims that the period was full of corruption and decadence to the extent that the enemies of Islam (in the West) anticipated the fall of what he calls the 'Muslim nation'. In Line 3, Alhawali represents Europe as non-religious which contradicts his previous representation of Europe as religiously fanatic (see Section 5.4.1 above). In Line 4, Alhawali also claims that (traditional) values faded in Europe. The plural word form of the key term 'CENTURY' 'القرون' ('algorūn: the centuries) collocates with 'الوسطى' ('alwostā; (the middle), which both compose the 2-word cluster 'القرون الوسطى' ('alqorūn 'alwostā; the Middle Ages) that occurs 37 times in 3 of Alhawali's texts (see Concordance 5.13).

^{59.} Debate about philosophy is deep rooted in the history of Islam. Here is one example: Ibn Rushd (also called Averroes, 1126-1198) is one of the Muslim philosophers and religious leaders who wrote summaries and commentaries on both Aristotle and Plato's works. For more information see the following links: <u>http://www.britannica.com/EBchecked/topic/45595/Averroes</u> and <u>http://www.iep.utm.edu/ibnrushd/</u>. Additionally, Algazali (also known in the West as Algazel c. 1058 – 1111) wrote extensively on philosophy and refutation of philosophy. See <u>http://plato.stanford.edu/entries/al-ghazali/</u>.

Concordance 5.13 Concordance of the cluster 'القرون الوسطي) '('alqorun 'alwosta; the Middle Ages) in Alhawali's corpus N

Concordance

[...] انتصر العلم الحديث على خرافات الكنيسة و أساطير القرون الوسطى الأوروبية و هذم أساليب البحت و طرق الاستنباط المدرسية التقليدية كان ذلك بلا ريب نصراً كبيراً انتخسانية في كل مكان، و فتحاً جديداً في عالم المعرفة و النور. لكن هذا النصر و الفتح اختفيا تحت ركام الاستغلال البسّع لما أنجزه الإنسان من تقدم في المعرفة [...] 1 [...] the victory of science over the fables of the Church and the myths of European middle ages and the abolition of the research methods of the traditional school was, undoubtedly, a great victory of humanity and a new horizon of light and knowledge. However, such victory and new horizon of knowledge disappeared underneath heaps of repugnant abuse of human progressive achievements in knowledge [...] ولاً نظرية الكبيسة و نظام الإقطاع - على الرغم من أن الكليسة لم تحاول تتبير بنية المجتمع المسيحي أو تنظيم بعض مجالات الحياة على الأظل، فقد كان لها أثر فعال في قصاد المرون الوسطى من الوجهة النظرية. أقرت الكنيسة النظام الإقطاعي السائد، بل أصبحت مؤسسة من مؤسساته التابتة، و أقرت الاضطهاد الفظيع الذي كان بتحرض ٥ أرقاء المجتمع رغم تناقيه مع تعاليم الإنجيل [...]

First: The Church's theory and feudalism: - Despite that the Church did not try to change the structure of the Christian society or, at least, 2 organize some aspects of life, it, from a theoretical perspective, had effective influence on the economy of the middle ages. The Church acknowledged the prevalent feudal system. Indeed feudalism became one of the Church's established institutions. Additionally, the Church acknowledged the hideous oppression that used to be practiced against the slaves in the society despite the fact it contradicts the teachings of the Bible [...]

[..] دائن عة الإنسانية (في أورويا) مدينة كلياً للحضارة الإسلامية، و لا ينحصر ذلك في الأتر الأدبي -اقتباس أبرز ممتليها و هو دانتي من أبي العلاء و ابن طفيل- بل يشمل المصبر كله، حتى أن الإمبراطور فردريك الثاني و أبو أكبر أباطرة **القرون الوسطي** بإطلاق، ويخبر لدى بعض المفكرين أول المحدتين ورائد النهضة، كان يتكلم العربية [...]

Thus the humanistic attitude (in Europe) is in great debt to Islamic civilization. It is not only limited to literary works - the borrowings of 3 Dante (1308 - 1321), who is the most famous representative literary figure in Europe, from Abi Alalaa (an Arab blind philosopher, writer and poet 973-1058) and Ibn Tufail (also known as Abubacer Aben Tofail, an Islamic philosopher who lived in Andalusia -modern Spain)1105 - 1185) - but indeed it includes the whole era to the extent that Frederick II (1194 - 1250) who is the most prominent and powerful Emperor in the middle ages and considered by some intellectuals to be one of the first founders and pioneers of [European] Renaissance, was able to speak Arabic [...]

Alhawali continues to represent Europe negatively and reminds his audience of the greatness of Islamic civilization during the Middle Ages. In Line 1, Alhawali insinuates that modern Europeans abuse science. In Line 2, he returns to criticise the Christian Church claiming that it supported feudalism (and even slavery) in the Middle Ages. The last line shows Alhawali's conception of European civilization that, according to him, borrowed humanism from Islamic civilization during the Middle Ages (note that this is at odds with his discussion about the West).

To sum up, the thematic focus in this section is about Alhawali's representation of Europe during the Middle Ages. He utilizes the key term 'CENTURY' to construct a negative image of Europe during the Middle Ages which serves as the bases of Alhawali's Salafist negative presentation of modern Europe. At the same time, he positively represents medieval Islamic civilization. Thus, as appears in the concordance lines above, Alhawali's construction

of the past is limited to a negative representation of Europe and his claims about the European enmity to Islam as well as constructing an idealised, advanced image of Islamic civilization in the Middle Ages. This is potentially a powerful strategy as it helps to justify and strengthen current opposition to 'the West'. If the West can be demonstrated as always having being an enemy of Islam, then any current problems are perhaps less likely to be viewed as resolvable, temporary or atypical. Instead, they are simply another manifestation of a continuous pattern.

Ibn Baz also uses the key term 'CENTURY' to construct the past, but his perception of the past differs from Alhawali's. Let us examine the limited occurrences of the key term 'CENTURY' in Ibn Baz's corpus that could help show the difference between the two Salafi writers in their construction of the past.

5.5.2 The construction of the past in Ibn Baz's corpus: The best generations and suitability of Islam in modern times

As I have indicated above, Ibn Baz uses two word forms of the key term 'CENTURY' that indicate the 'Periods of Times' namely, 'القرن' ('alqarn; the century) (8 times) and 'القرون' ('alqorūn; centuries/generations) (9 times). 'القرن' ('alqarn; the century) collocates with the word 'عشر' ('ašar; -teenth) (MI score 12.426 and joint frequency 7) that is used in Arabic to refer to the second part of the compound numbers from 11-19. This could indicate that Ibn Baz refers to centuries from the eleventh to the nineteenth centuries in both Hijra (Islamic) or Gregorian calendars. Let us move on to consider 'القرن' ('alqarn; the century) in Ibn Baz's corpus.

Concordance 5.14 Concordance of the word form 'القرن' ('alqarn; the century) in Ibn Baz

N	
1 IN	Concordance
	(نواقض الإسلام) الرابع : من اعتقد أن هدي غير النبي صلى الله عليه و سلم أكمل من هديه ،[] فهو كافر [] و يدخل في القسم الرابع من اعتقد أن
	الأنظمة و القوانين التي يسلها الناس أفضل من شريعة الإسلام [] أو أن نظام الإسلام لا يصلح تطبيقه في القرن العسرين []
1	(Nullifiers of Islam) Fourth: Whoever believes that guidance other than that of the Prophet, may Allah send His blessings and
	peace upon him, is superior to his [] is infidel [] This include those who believe that man-made laws and legislations are
	better than the Islamic Sharia [] or those who believe that Islamic legislation is not suitable for the twentieth century []
	كلمة تحذيرية حول إنكار رشاد خليفة للسنة المطهرة: [] 4- ادعاؤه أن الأخذ بالسنة و كتابتها و جمع الأحاديث في القرنين التاني والتالت كان سببا في
	سقوط الدولة الإسلامية. [] لذا فقد رأيت من الواجب توضيح أمره و كتبف حقيقته للمسلمين لنلا يغتر أحد بكلامه أو ينخدع بأرائه []
2	A warning letter about Rashad Khalifa's (an American-Egyptian Qur'anist 1935–1990) rejection of the purified Sunnah: [] 4-
-	He claims that relying on the Sunnah and collection of the Hadiths in the second and third centuries was a major reason that
	caused the fall of the Islamic Caliphate [] Thus I realised that it is an obligatory duty to expose him in front of Muslims so that
	no one will be misled or deceived by his ideas []
	و من أبرز هؤلاء الدعاة المصلحين الإمام التنيخ محمد بن عيد الوهاب مجدد ا لقرن التالي عتس الهجري رحمه الله الذي وفقه الله للقيام بدعوة إصلاحية
	عظيمة أعادت للإسلام في الجزيرة العربية قوته و صفاءه و نفوذه و طهر الله به الجزيرة من السَّرك و البدع []
3	One of the most prominent reformers is the Imam Sheikh Mohammad Ibn AbdulWahhab (1703 - 1792). He is the reformer of
	the twelfth Hijra century (18th century), may Allah have mercy upon him. Allah gave him success to start a great reforming call
	that brought back the strength and purity of Islam in the Arabian Peninsula. Additionally, Allah used him (Sheikh Mohammad
	Ibn AbdulWahhab) to purify the Peninsula from polytheism and impermissible religious innovations []
	أن أول من دعا إلى القومية العربية في أواخر القرن التاسع عتس الميلادي ، هم الغربيون على أيدي يعتات التبشيل في سوريا ، ليفصلوا الترك عن العرب ،
	ويفرقوا بين المسلمين ، ولم تزل الدعوة إليها في السّام و العراق ولبنان تزداد وتنمو[]
4	Verily, the first people who called for Arab nationalism in the late nineteenth century are the westerner missionaries in Syria in
	order to separate between Arabs and Turks and divide the Muslims. The call to it (Arab nationalism) is still increasingly
	developing in Cham (Syria) Iraq and Lebanon []

The word form 'القرن' ('alqarn; the century) is used by Ibn Baz for different purposes. In Line 1, he uses it to refer to 'nullifiers of Islam'. Nullifiers of Islam are those beliefs that, according to Muslim religious leaders, go against Islam and whoever amongst Muslims agrees with or believes in them becomes infidel. Ibn Baz, as a Salafist (as well as most Muslim religious leaders), claims that believing that Islamic legislations are not suitable for the twentieth century is infidelity, ergo, Ibn Baz argues that Islamic legislation is relevant for the present-day.

Line 2 indicates Ibn Baz's refutation of the claims of the Qur'anist Rashad Khalifa about the Sunnah of the prophet of Islam which is considered by all Muslims to be the second source (after the Qur'an) of the Sharia.⁶⁰ Ibn Baz also uses the word form 'القرن' (*'alqarn*; the century) in reference to associationism and impermissible religious innovations. He

^{60.} There are four sources of Sharia in Islam: the Qur'an, the Sunnah of the Prophet of Islam, consensus of Muslim jurists and Analogical deduction.

represents Sheikh Mohammad Ibn AbdulWahhab as a religious reformer whose efforts, according to Ibn Baz, helped purify Islam from polytheistic practices. Line 4 shows Ibn Baz's rejection of Arab nationalism. He attributes initiation of Arab nationalism to western missionaries whose aim, according to Ibn Baz, is to fight the Turks.

Although Ibn Baz's use of the word form 'القرن' ('alqarn; the century) (in Line 1) does not signify history as he is writing about the century he lived in, it signifies his conception of the continuity of time as regards the Sharia. Basically all Salafists acquiesce to the idea of the suitability of the Sharia laws to be applicable in all aspects of life in all times and places. However such a conception of continuity neglects some Salafists' rejection of new social, political and technological developments. Such resistance is evidently reflected in some Salafists' rejections of any new social or technological developments and later they reluctantly acquiesce. Examples of such reluctance to acquire new developments include the rejection of girls' education by some Salafists immediately after the royal decree in 1960 as recorded by Alwashmi (2009) and the rejection of satellite television channels in the early 1990s (Alomar, no date). Let us now turn to investigate the second word form of the key term 'CENTURY' used by Ibn Baz.

The definite plural form 'القرون' ('*'alqorūn*; the generations) collocates with the word (*'almofadalah*; the best/preferred) (MI score 13.536 joint frequency 7). The collocate (*'almofadalah*; the best/preferred) is an Arabic passive participle which functions as an adjective, that holds a positive semantic preference, to modify the word form 'القرون' (*'alqorūn*; generations). The collocability of the word form 'القرون' *'alqorūn*; the generations) and the word 'المنفضلة' (*'almofadalah*; the best/preferred) could be explained by the 2-word cluster 'القرون المفضلة' (*'almofadalah' almofadalah*; the best generations) that occurs 7 times in 3 of Ibn Baz's texts. Let us now examine the co-textual environment of the word form '*'alqorūn*; '*'alqorūn*; the generations). Concordance 5.15 Concordance of 'القرون) (' 'alqorūn; the generations) in Ibn Baz's corpus

N Concordance [...] لا يجوز الاحتفال بمولد رسول الله صلى الله عليه و سلم [...] لأن ذلك من البدع المحدثة، لكون رسول الله صلى الله عليه و سلم لم يُعلُّه [...] و لم يفعله أيضا التابعون لهم بإحسان في القرون المفضلة و هم أعلم الناس بالسنة. [...] [...] It is religiously not permissible to celebrate the birth day of the Prophet, may Allah send His blessings and peace upon him [...] because such celebration is an impermissible religious innovation and the Prophet, may Allah send His blessings and peace upon him, did not do it [...]. The followers of the companions, who were from the best generations, who knew the Sunnah better than any other people, did not do it [...] إن دعوة الإمام التبيخ محمدين عبد الوهاب رحمه الله هي الدعوة الإسلامية التي دعا إليها رسول الله صلى الله عليه و سلم و صحابته الكرام و **سلف هذه الأمة الصالح [...]** لأنها أتمرت تمرات عظيمة لم تحصل على يد مصلح قبله بعد ا**لقرون** المفضلة، و ذلك لما ترتب عليها من قيام مجتمع يحكمه الإسلام [...] Verily, the call of the Imam Sheikh Mohammad Ibn AbdulWahhab (1703 - 1792), may Allah have mercy 2 upon him, is the same call that Allah's messenger, may Allah send His blessings and peace upon him, his noble-hearted companions and the pious predecessors of this (Muslim) nation invited people to [...] because it resulted in many great effects that had never happened before by any religious reformer since the best generations. The case is so, it (the call of Mohammad Ibn AbdulWahhab) resulted in establishing a society that is governed by Islam [...] و قال (محمد الغزالي حول القومية) أيضا في كتابه المذكور (مع الله) صفحة 347 ما نصبه [...] لمانا يلقى الإسلام هذا الهوان كله؟ و الجواب : عند الاستعمار الذي يجر خلفه ضغائن ا**لقرون** الأولى ويضم نصب عينه ألا تقوم للإسلام قائمة في بلاده [...] And Mohammad Algazali (an Egyptian religious leader 1917-1996) wrote in his previously mentioned book 3 (With Allah) page 374: [...] Why Islam is so oppressed? The answer is only known by colonialism (colonialists), who are caused by the grudges of the early centuries, decided that Islam should not resist its enemies in its own country [...]

The cluster 'القرون المفضلة' ('almofadalah 'almofadalah; the best generations) appears in Lines 1 and 2. In the Salafi discourses, this cluster is commonly used to refer to the 'السلف الصالح' ('assalaf 'aṣṣāliḥ; the pious predecessors) (see Section 1.3.6 in Chapter 1). It is a recontextualization of many Prophetic traditions (Hadiths) including the following:

Narrated Zahdam bin Mudrab: I heard Imran bin Husain, may Allah be pleased with both of them, saying, "The Prophet said, 'The best people are those living in my generation, then those coming after them, and then those coming after (the second generation) [...]" narrated in Sahih Bukhari, *Volume 3, Book 48, Number 819.*⁶¹

In Line 1, Ibn Baz rejects the celebration of the birthday of the Prophet of Islam which is a common Sufi practice. He claims that such celebration is an impermissible religious

^{61.} Sahih Bukhari is a voluminous work by Albukhari (commonly referred to as Imam Albukhari 810 – 870). It is a collection of Prophetic Hadiths that are called 'sound' Hadiths. The translation of this Hadith is available on: http://www.usc.edu/org/cmje/religious-texts/hadith/bukhari/048-sbt.php. Accessed on 31/03/2013.

innovation because the Prophet of Islam and the best generations did not perform such religious rituals. Thus, he uses reference to the 'best generation' to legitimate his rejection of celebration of the birthday of the Prophet of Islam. Ibn Baz, in Line 2, links the religious reforms of Mohammad Ibn AbdulWahhab to what he and the Salafis believe to be the best generations; 'السلف الصالح' (*'assalaf 'aṣṣāliḥ*; the pious predecessors). Line 3 indicates 'the Other'. Ibn Baz quotes the Egyptian religious leader Alghazali who claims a relation between Islam being oppressed to colonialists and the grudges they carry from the early centuries. Alghazali refers to the Crusades and the early confrontation between Muslims and the people in the West. However, the confrontation between the Muslims and the West goes back to the early wars and the establishment of Al-Andalus (see footnote 53).

The past in Ibn Baz's texts is thus connected to themes such as polytheism, Sufi impermissible religious practices (celebrating the birthday of the Prophet of Islam), and Arab nationalism which seem to be recurrent in Ibn Baz's texts (see Chapter 4 sections 4.2.1.2, 4.2.1.3 and 4.2.4, Chapter 6 sections 6.2.1.1 and 6.3.1.1 and Chapter 7 section 7.4.1.1 and 7.4.2). It is also connected to nullifiers of Islam and the protection of the Sunnah which was collected in the second and third Hijra centuries (corresponding to 8th and 9th centuries). Ibn Baz, in these semantic foci, projects the Salafists' conception of the ritualistic religious practices (such as celebrating the birthday of the Prophet of Islam.) that, as he claims, are not known during the time of 'السلف الصالح' ('assalaf 'asşāliḥ; the pious predecessors). Thus we may conclude that Alhawali's use of the key term 'CENTURY' is limited to demonizing the West and modern Arabic/Muslim culture that he claims to be under the influence of western philosophy, whereas Ibn Baz is mostly concerned with the religious practices of 'asşāliḥ; the pious predecessors). Let us move on to examine the construction of mind and religion in both corpora.

5.6 Mind in the Salafi corpora

The keywords that indicate the category 'Mind' in Alhawali's corpus are 'العقل' ('al'aql; the mind) (116 occurrences in Alhawali and 13 in Ibn Baz). The keyword 'العقل' ('al'aql; the mind) also has other word forms that occur in both corpora. It is derived from the trilateral verb 'at' ('a qa la; to understand). Table 5.8 below contains 7 word forms of the key term 'MIND' that are used in the Salafi corpora. The translation of these word forms are based on a preliminary concordance analysis.

Ν	Word form	Eng. trans	F. Alhawali	F. Baz
1	العقل	the mind	116	13
2	2.6.8	the logical/ the mentality (adj. pl./n. sing.)	31	6
3	العظى	the logical (def. sing. adj.)	29	1
4	العقول	the minds	17	10 ·
5	عتلية	logical/ mentality (adj./n.)	13	0
6	عتثي	logical (indef. Sing. adj.)/ my mind	3	1
7	متأيات	mentalities	1	0
total			210	31

Table 5.8 Word forms of the key term 'MIND' in the Salafi corpora.

Two of the word forms of the key term 'MIND' are used in the two Salafi corpora carrying different meanings. The word form 'العقاية' ('al'aqliyyah; the logical/mentality) is used 7 times by Alhawali as a definite noun that could be translated into English as 'mentality' and 24 times to indicate the adjective 'logical' whereas this word form is used in Ibn Baz as a definite adjective bearing the meaning 'logical'. The indefinite word form 'agliyyah; logical/mentality) is also used twice in Alhawali as an indefinite noun carrying the meaning 'mentality' and 11 times as an indefinite adjective meaning 'mental'. Additionally, the word 'agliy' ('aqli) is used 3 times in Alhawali as an indefinite adjective meaning 'logical' and once in Ibn Baz as a noun meaning 'my mind'. Now, I turn to examine the word forms of the key term 'MIND' in each Salafi corpus.

5.6.1 'Mind' in Alhawali's corpus

The word forms of the key term 'MIND' neither have collocates nor occur in word clusters in

Alhawali, so let us consider the co-textual environment of these words in Alhawali.

Concordance 5.16 Concordance of the word forms of the key term 'MIND' in Alhawali

N	Concordance
a mort	[] قدمت جيوش الاحتلال المسكري إلى العالم الإسلامي تقودها تتقليات غير المقلية البربرية الصليبية، فهي تتمتع بقسط كبير من الدهاء والخبت، و هي تعرف سلفاً أن لها مهمة أعظم من مهمة أجدادها []
	[] The armies of the invaders came to the Muslim world led by mentalities that are different from the barbaric
	Crusaders' mentality. They enjoy much of shrewdness and wickedness. They also know that their mission is greater than that of their forefathers []
2	[] إن بدعة كهذه (صكوك الغفران) لن تمر في التاريخ مرور الكرام [] و تثير ردود فعل بعيدة الآثار ، لاسيما و قد ظهرت في الفترة التي اتصلت فيها أوروبا
	رسان ہے : بنور الإسلام، و أخذت ا لعقول الثالمة تكلمس مكانها في الحياة []
	[] verily, an (impermissible religious) innovation of this kind (Indulgence: full or partial remission of temporal
	punishment granted by the Church) will surely not pass as if nothing had happened [] and it should cause long lasting
	reactions especially it (Indulgence) appeared in the period that Europe was in contact with the light of Islam, and the
	sleeping <u>minds</u> (of the medieval Christians) started to take their places in the course of life []
3	[] و تعالت أصوات الباحثين و القلاسفة، منادية بأن ا لعقل هو الحكم الوحيد و ا لعقل هو كل شيء،[] و القداء والصلب و الرهباتية إلخ، كلها أباطيل مصللة و
	عقائد مر ذوله []
	[] researchers and philosophers started to call that mind is the only arbiter and mind is everything [] and Atonement,
	Crucifixion and Monasticism are false conceptions that mislead people and despicable beliefs []
4	هذا كله غير حداثة ماركس التي حقرت كل ما قبلها [] ثم جاء عصر استانين و تبنى اتحاد الكتاب السوفيت أراءه الأكثر حداثة، و هكذا إنه ا لعقل البشري
	المحدود الضمعيف الذي يتخيل كل مرحلة من مراحله أنها نهائية الكاريخ []
	[] In addition, the modernism of Marx (Karl Marx a German philosopher 1818-1883) that denigrated all previous
	intellectual works [] then came the time of Stalin (Joseph Stalin, one of the leaders of the Soviet Union 1878-1953)
	whose more recent modernist ideas were adopted by The Soviet Writers' Union and so forth. Indeed it is the weak and
	limited human mind which imagines that any of its stages is the end of history []
1	و الرومانسية هي ارتئانا صوفي، و لكن موضوعة ليس الرب كما في رهبانية النصاري بل الطبيعة، و هي لا تهدف إلى التوجيه ا لطلي للناس عن طريق حكمة الاسترب المالية المالية المالية المالية المالية المالية المالية النصاري بل الطبيعة، و هي لا تهدف إلى التوجيه الطلي
5	القدماء، بل إلى الإشباع العاطقي الذي يجعل الذات محور العالم، إنها مزيج من اليأس الرهباني، والهروب من الواقع الذي كلما تقدمت المعرفة المقلية أظهرت أنه أكثر اقتامة وكأبة.
	Romanticism is a a mystic retrogression, however it revolves not around God as is the case in Christian Monasticism on the contrary its topic is nature. Its goal is not mental guidance of people via old wisdom. Its goal is emotional gratification
	the contrary its topic is nature. Its goal is not <u>memoral</u> guidance of people via one wisdom. Its goal is controlling guinearent that makes the self to be the centre of the world. Verily it is a mixture of monastic despair and escape from reality that
	becomes more depressing and gloomy whenever human mind achieved some progress in knowledge.

To reiterate from Chapter 3, I classify all concordance lines according to their functions (mostly themes) in the relevant texts, then I choose one or two lines that are representative of the overall picture, typically being used with the same function. Thus, concordance 5.16 only shows examples of concordance lines that repeat the same function that is indicated by the search words (the different word forms of the key term 'MIND') in Alhawali. Firstly, in Line 1, Alhawali uses the indefinite plural noun 'area '(*aqliyyāt*; mentalities) to negatively represent the western armies (as shrewd and wicked) that invaded the 'Muslim world' in the

18th and 19th centuries. He links them to the medieval Crusades claiming that their mission (to eradicate Islam) this time is more difficult than that of the Crusaders (see Section 5.4.1 above). He also uses the singular definite noun 'العقلية' ('al'aqliyyah). Alhawali uses the indefinite plural (mentalities) to represent anonymous people whom he claims to be leading attacks on Islam whereas he uses the definite singular noun (the mentality) to describe the Crusaders. Alhawali, from a Salafist political perspective, uses these two different word forms (plural vs. singular) in order to accentuate what he claims to be the wickedness of the direct menace threatening Islam in modern times. Secondly, Alhawali, in Lines 2-5, uses different word forms of the key term 'MIND' to represent religious and cultural lives in the West.

Alhawali, in Lines 1 and 2, nominates social actors (medieval Crusaders and some western leaders) via abstraction using word forms of 'MIND' (see Section 6.3 in Chapter 6 for more details about Alhawali's representation of social actor). In Line 2, Alhawali compares the religious life in medieval Europe to Islam recontextualizing the Salafi term 'بدعة' (*bid'ah*; impermissible religious innovation) to represent *Indulgence* (full or partial remission of temporal punishment granted by the Church) negatively. He claims that after Europe contacted the Muslim world the 'sleeping minds' (medieval Christians) saw the light and rejected '*Indulgence*'. Alhawali continues by claiming that the Church's teachings (Atonement, Crucifixion and Monasticism) are false and caused researchers and philosophers to reject religious teachings of the Church and depend mainly on 'mind' (Line 3) (see p. 119 and footnote 49). Alhawali then continues to negatively represent the human mind claiming that it is weak and limited (line 4) as manifested in the developments of modernist economic and political theories from Marx and Stalin. In Line 5, Alhawali turns to attack the literary movements in Europe. He claims that romanticism, according to Alhawali, is a sort

of monastic despair. Alhawali claims that romanticism, as a human achievement in Europe, is a kind of escaping from the gloomy reality that resulted from the achievements of the human mind that he claims to be weak and limited. In all the concordance lines (Concordance 5.16) Alhawali uses the different forms of the key term 'MIND' to represent the West via three major semantic foci: the negative representations of the teachings of the Church, political and economic theories (modernism) and the 18th century literary romanticism, i.e., demonizing the West. Having explicated the semantic foci introduced by the key term 'MIND' in Alhawali, I now turn to examine Ibn Baz's use of this key term.

5.6.2 'Mind' in Ibn Baz's corpus

Ibn Baz' use of the key term 'MIND' is limited (collectively 31 times) as compared to that of Alhawali (collectively 210 times). He uses the word forms 'العقل، ('al'aql; mind) (13 times), the definite plural 'العقول' ('al'oqūl; the minds) (10 times), the adjective 'العقول' ('al'aqliyyah; logical) (6 times) and the adjectives 'العقلي' ('al'aqli; the mental -def. sing. adj.) and 'aqli, 'aqli; my mind) (once each). The word forms of the key term 'MIND' in Ibn Baz do not have any collocates. They also do not occur in any recurrent word clusters. Instead, let us now see how Ibn Baz uses them via concordances.

Concordance 5.17 Concordance of the word forms of the key term 'MIND' in Ibn Baz

N	
	Concordance
	[] و هكنا رسانته الموسومة بـ (التنمرية)[] و بين فيها عقيدة أهل السنة بكلتها النتائية و العظيمة [] و كل من خالف أهل السنة فيما اعتشوا في باب الأسماء
	و الصفات فإنه يقع و لا بد في مخالفة الأدلة النقلية و العقلية []
	[] and so is his book titled Arrisalah Attadmoriyyah (a book written about the names and attributes of Allah by Ibn
	Taymiyya, a medieval Salafi religious leader 1263-1328)) [] He explained in it the beliefs of the Sunnis providing
	evidences from the Qur'an and Prophetic traditions as well as logical proofs [] And whoever opposes the beliefs of the
	Sunni, he/she should contradicts evidence from the Qur'an and Prophetic traditions as well as logical proofs []
2	[] تم بيين (الله) الأدلة في مواضع كثيرة ، عندما يتُّهلها المؤمن يعرف أن الدليل النظي مؤيد بالتاليل العظمي المساهد المصنوس ، و لهذا ذكر سبحاته بعد قوله :
	(يَا نَبِها الناس أُعْبَدُوا رَبِكُمُ المجهُ على ذلك نقل : (انْذِي خَلْقَةُ و النَّبِينَ من قَبْتِكُمُ لعلكم للتَّونَ) ، و المعنى أن هذا المحالق لذا هو المستحق أن نعبده لكونه خلقا []
	[] then (Allah) plainly states evidences in the Qur'an in many Chapters that when a believer takes into consideration
	will realize that the Qur'anic evidence is supported by perceptible and felt logical proofs, and for this reason the Glorified
	mentioned after this Qur'anic verse {O mankind! Worship your Lord (Allah)} the following proof {Who created you and
	those who were before you so that you may that ye may have the chance to learn righteousness} which means that the
	Creator deserves to be worshipped because He created us []
2	[] كما قال سبحانه و تعالى : (و لا تُحْدَنِينُ اللهُ هُقَلًّا عَمَّا يَعْمَلُ الضَّالِمُونَ إِنَّما يُؤَخَّرُ هُمَّ لِيَوْمِ تُشْخَصُ فِيهِ الْأَبْصَارُ) [] هناك جزاء و حسابا ، و قد قامت على هذا
	الأنفة من القرآن و السنة ، و إجماع الأمة ، و العقول الصحيحة []
	[] as the Glorified and the Exalted said {Consider not that Allah is unaware of that which the Zalimun (polytheists,
	wrong-doers) do, but He gives them respite up to a Day (of the Judgment) when the eyes will stare in horror.} there will
	be punishment and reward that are based on evidences from Qur'an and the Sunnah (Prophetic traditions), the
	consensus of the Muslim nation and sound <u>minds</u> []
	[] لا ريب أن الدعوة إلى أن تكون التومية العربية هي الرابطة الأولى بين العرب ، دعوة باطلة لا أساس يؤيدها ، لا من العقل و لا النتل ، بل هي دعوة []
	حاهلية إلحادية ريدف دعائها إلى محاربة الإسلام []
	[] No doubt that advocating Arab nationalism to be the only bond that unites Arabs is false and baseless. Neither
	mind helps to prove it nor evidences from Qur'an and Prophetic traditions support it. On the contrary, it is an atheist
	and Jahili (ignorant i.e. pertaining to the pre-Islamic era) call that aims to fight Islam []

Ibn Baz uses the word forms of the key term 'MIND' referring to the mind as evidence that supports his claims and religious conceptions. In Line 1, Ibn Baz introduces one of the books (*Arrisalah Attadmoriyyah*; A Letter to the People of Tadmor (a city located in northeast Damascus –also called Palmyra) of the medieval Salafi religious leader (Ibn Taymiyya). In his book, Ibn Taymiyya introduces the Islamic Sunni monotheistic concept in the names and attributes of Allah which confirms that all Muslims should believe and use them in their prayers as they were reported in Qur'an and Prophetic traditions.⁶² He claims that the Salafi conception of the names and attributes is supported by evidences from the Qur'an and Prophetic traditions as well as logical proofs and anyone who rejects this Sunni belief is

^{62.} According to the Salafi belief in Allah's names and attributes, Muslims, in all forms of their worship (including supplications and beseeching Allah's help) should call on Him by His names and attributes (see Section 4.2.4.2 in Chapter 4).

contradicting these logical proofs as well as the Qur'an and the prophetic traditions. In Line 2, Ibn Baz introduces two parts of one Qur'anic verse that reports another Islamic belief in worshipping Allah and believing that He is the Creator. Ibn Baz builds a causal relationship: worshipping Allah is based on the belief that He is the Creator. He also continues to say that this belief which emanates from the Qur'an (Line 2) is also supported by perceptible logical proofs that are reported in Qur'an. Ibn Baz uses the word form 'لعتر ('al'oqūl; the minds), in Line 3, to represent believers who have faith in the Hereafter. He says that there will be punishment and reward in the Hereafter which, according to him, is reported in the Qur'an and Prophetic traditions and approved by all Muslims as well as sound minds. Line 4 introduces Arab nationalism as one of the thematic foci of Ibn Baz (see Section 4.2.6 in Chapter 4). He rejects the claim that Arab nationalism unites Arabs claiming that neither mind nor evidence from Qur'an and Prophetic traditions supports it. He also claims that advocating Arab nationalism is a war against Islam.

Both Salafi writers use the key term 'MIND' differently. On the one hand, Alhawali uses the key term 'MIND' to represent the West negatively in different political/religious foci. He uses it to represent what he claims to be the western direct menace to Islam. He also represents western religious (Church) and other cultural aspects negatively. On the other hand, Ibn Baz uses the key term 'MIND' as a legitimation strategy to lend support to his stances in different semantic foci: the Salafi version of faith in Allah's names and attributes, the concepts of creation of humans and their worship of Allah, the reward and punishment in the Hereafter (theological semantic foci) and rejection of Arab nationalism (political semantic focus). These concepts are argued to be supported as logically proven and believed by those who have sound minds. Thus, on the surface, Ibn Baz occasionally seems to incorporate aspects of language perhaps more typically associated with scientific discourse in his
rhetoric. However, these examples tend to *claim* that something is logically proven rather than *showing* that it is so. Let us now move to examine how each writer represents religion.

5.7 Religion in the Salafi corpora

The keywords that indicate religion in Alhawali are the definite noun 'الدين ''addīn; the religion) and the adjective 'الدينية' ('addīniyyah; the religious – sing./pl. fem.). However there are other word forms of the key term 'RELIGION' (Table 5.9) that indicate religion in both corpora.

Ν	Word form	Eng. trans.	F. Alhawali	F. Baz
1	الدين	the religion	554	222
2	الدينية	the religous (def. adj. sing./pl. fem.)	105	1
3	دين	religion	85	126
4	الدينى	the religious (def. Adj. sing. masc.)	59	2
5	دينية	religous (indef. adj. sing. fem/ pl.)	33	0
6	دينهم	their religion	27	38
7	دينى	religious (indef. adj. sing. masc.)	12	0
8	دېنه	His/his religion	11	73
Tot	al		886	462

Table 5.9 Word forms of the key term 'RELIGION' in the Salafi corpora.

There are 8 word forms of the key term 'RELIGION' that are used in the Salafi corpora under investigation. Table 5.9 shows these word forms and their relative frequencies in each corpus. The definite noun 'الدين' (*'addīn*; the religion) is the most frequent word form of the key term 'RELIGION' that both Salafi writers use (554 times in Alhawali and 222 times in Ibn Baz). Notably, Ibn Baz uses the indefinite noun 'دين' (*dīn*; religion) with a higher frequency (126 times) than Alhawali (85 times). Also, both writers use the word form '*cuiter*' (*dīnahom*; their religion) with relatively low frequencies (38 times in Ibn Baz and 27 times in Alhawali).⁶³ The rest of the word forms are either used by Ibn Baz with a very low frequency – once and twice for definite adjectives '(*addīniyyah*; the religious – sing. pl. fem.) and '*lucuia*' (*licuita*')' (*licuita*')' (*licuita*')''

^{63. &#}x27;دينهم' (*dīnahom*; their religion) could indicate the religion of 'the Other'. However when I examined this word form in both corpora I found that Ibn Baz uses it in all the occurrences to indicate Islam (see below) whereas Alhawali uses it to refer to both Islam and more noticeably Christianity.

('addīni; the religious –sing. masc.) respectively - or do not occur in Ibn Baz – the indefinite adjective forms 'دينيه' (*dīniyyah*; religious – sing. pl. fem.) and 'دينيه' ('addīni; religious –sing. masc.). It is noteworthy that some of the word forms in Table 5.9 are keywords in both corpora (detailed more below). This could be due to that the texts in both corpora being composed of religious writings and talking about religion. This result could also indicate that both writers use different word forms to write about religion from different Salafi perspectives to achieve different goals (see below). Additionally, my corpora are not lemmatized which results in having many word forms that belong to the same lemma as keywords in either of the Salafi corpora as shown in Table 5.10 below.

		-			•		•	
N	Key word	Eng. trans.	Texts	Freq.	RC. Freq.	Keyness	Р	\geq
1	دينه	His/his religion	4	73	11	80.31568	8.05E-15	ម្ល
2	دين	religion	5	126	85	33.80208	3.18E-09	R
3	الدينية	the religous (def. adj. sing./pl. fem.)	4	104	1	96.29324	3.31E-15	/
4	الدينى	the religious (def. Adj. sing. masc.)	4	57	2	44.16547	7.62E-13	lha
5	الدين	the religion	5	554	222	42.78454	1.18E-12	Iwal
6	دينية	religous (indef. adj. sing. fem/ pl.)	5	33	0	33.55122	4.02E-09	

Table 5.10 Word forms of the key term 'RELIGION' that are keywords in both corpora

With the exception of the word forms 'ديني،' (*addīni*; religious –sing. masc.) and 'ديني،' (*dīnahom*; their religion), all other word forms are keywords in either Salafi corpus as shown in Table 5.10. The indefinite noun 'دين ' (*dīn*; religion) and 'دينه' (*dīnah*; His/his religion) are keywords in Ibn Baz whereas the rest of the word forms are keywords in Alhawali. However taking the total frequency of the word forms of the key term 'RELIGION' into consideration there exists a significant difference between the collective occurrences of these word forms in both corpora (log likelihood score 19.41) which indicates that 'RELIGION' overall is a key term in Alhawali. Let us now move on to investigate the word forms of the key terms 'RELIGION' in Alhawali.

5.7.1 'Religion' in Alhawali's corpus

Let us first examine the collocational profile of the key term 'RELIGION' in Alhawali. Table 5.11 shows the collocates of the key term 'RELIGION' in Alhawali.

Word form	Eng. trans.	Freq.	Collocates	Eng. trans.	Freq.	Texts	Joint	MI
			رجأل	(clergy)men	123	3	72	7.69441
الدين	the religion	554	ئىي،	thing	1 99	3	10	4.15231
			عن	about	1215	3	54	3.9 7515
1.1	الدينى the religious (adj. masc.)	57	الإصلاح	the reform	33	3	8	9.70346
لىرىسى		1,5	حركة	movement	58	3	8	8.88987
دين	religion	85	بار	or	936	3	7	4.10834

Table 5.11 Collocates of the key term 'RELIGION' in Alhawali

There are 6 collocates (Table 5.11) of the key term 'RELIGION' in Alhawali. The collocate 'رجال, (*riğāl*; men) (MI score 7.694 joint freq. 72) is also a keyword in Alhawali (keyness 66.161). Both this collocate and the keyword 'الدين' (*'addīn*; the religion) compose the Arabic annexation construct 'رجال الدين' (*riğāl 'addīn*; clergymen) which is a 2-word cluster in Alhawali (69 times in 3 texts). This annexation construct has a negative connotation and is always used in the Salafi discourses to nominate religious leaders in religions other than Islam (see Section 6.3.1.1 in Chapter 6).

The collocates of 'للديني' ('addīni; the religious -adj. masc.) are 'حركة' ('al'işlāh; the reform) and 'حركة' (harakah; movement) (MI scores 9.703 and 8.889 respectively). 'حركة' (harakah; movement) is also a keyword in Alhawali (keyness 50.673). Both collocates occur 8 times jointly with the keyword 'الديني' ('addīni; the religious -adj. masc.). The collocability of these words could be explained by the 3-word cluster 'حركة' (harakat 'al'işlāh 'addīni; the movement of religious reform) (8 times) which denote the efforts of Martin Luther (a Christian German monk 1483 – 1546) and other European theologians in the Middle Ages. The previous collocates and word clusters of the key term 'RELIGION' in Alhawali indicate Christianity. This result supports other results (explicated above and elsewhere in this thesis) that indicate the focus of Alhawali's writings on the 'Other'. Let us

now move to explore more of Alhawali's use of the different word forms of the key term

'RELIGION' via selected concordance lines.

Concordance 5.18 Concordance of the word forms of the key term 'RELIGION' in Alhawali

N	Concordance
	[] و لأول مرة في تاريخ الإنسان أصبح النزاع بين الذكاء الصرف و القيم الخاقية قضية حياة أو موت،[] هذا قليل من كثير من النتائج السيئة التي جلبها الصراع المتنوم بين فين أوروبا و علمها، و نفع إليها التحصب المقيت من قبل دعاة اللادينية في مجال مفروض فيه أن يكون أعظم طريق إلى الله و أقوى دافع إلى خشيئة.
1	[] and for the first time in the human history, the conflict between human intelligent mind and moral values became an issue of life and death [] these are only few of the bad consequences that appear as a result of the sinister conflict between the religion of Europe (Christianity) and the science (of Europe) which was caused by the repugnant bigotry of the advocates of irreligion in an aspect that is should be the greatest path to Allah and the most compelling factor to fear Him (i.e. science causes people to know and fear Allah).
	[] و هذا يتودنا إلى معرفة مدى التسخ الذي وصلت إليه البيئة الأوروبية و الانهيار ا لديني و الخلقي الذي اجتاحها؛ فكانت بيئة ملائمة لتتريخ أفكار فرويد و تقبلها مما دفعها إلى تتسخ أعظم
2	[] this leads us to realize the huge debasement and the moral and <u>religious</u> collapse in the Europe. Thus it was suitable abode that allowed the ideas of Freud (Austrian neurologist 1856 – 1939) to germinate which caused more (religious and moral) debasement.
3	[] و ليس تمه تلك في أن مصر الحديثة التي يريدها كرومر هي دولة لا <u>دينية</u> لا صلة لها بالإسلام و حكومتها ستكون على الشرط الذي مر أنة أزالغاء الشريعة الإسلامية) [] and there is no doubt that modern Egypt advocated by Cromer (a British diplomat and colonial administrator 1841–1917) is a non <u>religious</u> and non Islamic state. Additionally, its administration will be based on the condition that I have cited above
	(abolition of Islamic Sharia) [] مواجهة توروبا للإسلام: أما الإسلام فإنه لما كان الرومان عامة يحتون كل ما عناهم من السعوب برابرة، و لما كان البابوات و رجل الكنيسة يحتون الإسلام وتنبة فقد
4	ائتل الموردان في النظر؟ القائمة إلى الحام الإسلامي، و المئزجت العنصرية القائمة بالحقد الميتي الجديد. Europe confronts Islam: As for Islam, all Romans considered all other nations (including Muslims) to be barbarians, and the Popes and clergymen considered Islam to be atheism. They (the Romans and Christians) both had gloomy perspective about the Muslim world, and thus old bigotry is mixed with the new <u>religious</u> grudge.
5	[] أما حركة الإصلاح الديني قلم توك مع "لوتر" و "كالفن"، بل لها جنور عميتة الصلة بالإسلام لا يستطيع أي باحث أوروبي أن يغللها مهما قلل من شائها، و منها (] أما حركة الإصلاح الديني قلم توك مع "لوتر" و "كالفن"، بل لها جنور عميتة الصلة بالإسلام لا يستطيع أي باحث أوروبي أن يغللها مهما قلل من شائها، و منها (حركة تحطيم الصور والتماثيل) التي اجتاحت الأمبراطورية البيزنطية في أوائل القرن الثامن الميلادي [] (حركة تحطيم الصور والتماثيل) التي اجتاحت الأمبراطورية البيزنطية في أوائل القرن الثامن الميلادي [] (حركة تحطيم الصور والتماثيل) التي اجتاحت الأمبراطورية البيزنطية في أوائل القرن الثامن الميلادي [] (حركة تحطيم الصور والتماثيل) التي اجتاحت الأمبراطورية البيزنطية في أوائل القرن الثامن الميلادي [] (حركة تحطيم الصور والتماثيل) التي اجتاحت الأمبراطورية البيزنطية في أوائل القرن الثامن الميلادي [] (حركة تحطيم العار 1483 من من الميلادي [] المع العام العام العام التي التي اجتاحت الأمبراطورية البيزنطية في أوائل القرن التامن الميلادي [] (حركة تحطيم العار و التماثيل) التي احتاجة الأمبراطورية البيزنطية في أوائل القرن التامن الميلادي [] أما حماية الصاح العالم المالة التي التي التي التي التي التي التي التي

As is the case of his writings that defame the West, Alhawali continues to use the key term 'RELIGION' to construct a negative image of Europe (see all the sections above and Chapters 4, 6 and 7). Alhawali, in Line 1, uses the indefinite keyword 'نين' ($d\bar{i}n$; religion) with the keyword 'أوروبا' ('oroba; Europe) (keyness 233.322) to form the annexation construct ' نين ' ($d\bar{i}n$ 'oroba; the religion of Europe) which means the religion of Europeans, i.e. Christianity. The previous annexation construct is also followed by 'e = d a a b' (wa 'ilmiha; and its science) meaning 'and the science of Europe'. As with keywords like 'West' and 'Israel',

Alhawali represents Europeans by 'Europe' which is a collectivising and impersonalising strategy (see Alhawali's representations of social actors in Chapter 6). Alhawali claims that there was a conflict between science and religion in Europe that was, according to him, caused by the repugnant bigotry of the advocates of irreligion. Alhawali here actually sides with the Church, though he defames it. This ambivalent stance of Alhawali could not be due to his rejection of irreligion only, but could be caused by his fondness of the concept of religion in general. This is explained by the Salafi belief about religion as a means to know Allah and fear Him which occurs in the following Qur'anic verse:

{إِنَّمَا يَخْشَى اللَّهَ مِنْ عِبَادِهِ الْعُلَمَاءُ إِنَّ اللَّهَ عَزِيرٌ عَفُورٌ } سورة فاطر، آية ٢٨. {It is only those who have knowledge among His slaves that fear Allah. Verily, Allah is All-Mighty, Oft-Forgiving} Holy Qur'an 35: 28

In Line 2, Alhawali returns to denigrate Europeans claiming that Europe was religiously and morally debased which was suitable abode for the ideas of Freud, which according to Alhawali are against religion, to germinate. Alhawali, in Line 3, reminds his readers of the British occupation of Egypt. He nominates Cromer (the British colonial administrator) as against Islam advocating a non religious state in Egypt and abolishing the Sharia. In another attempt (Line 4) to represent Europeans negatively, Alhawali tries to characterise Europe as an eternal enemy of Islam recalling the racist views of the Byzantine Empire toward others, including Muslims which, according to Alhawali, continues in a form of religious bigotry by the Popes and priests towards Muslims. Line 5 shows Alhawali's attempt to shed light on the genesis of religious reforms in Christianity. He claims that the beginnings of such reforms were not due to efforts of religious reformers such as Martin Luther (a German monk and a religious reformer 1483 – 1546) and John Calvin (a French theologian and religious reformer 1509 –1564) but due to the early contact between Muslims and the Christians which resulted in destroying images and idols (Iconoclasm) in the Byzantine Empire in the early years of the 8th century.

Alhawali uses the different keywords that belong to the lemma 'RELIGION' to further construct a negative image of Europe recalling what he believes to be the historical enmity between Europe and Islam. His negative representation of Europe and claims about such enmity contributes to constructing two opposing worlds (Europe vs. the Muslim world). Let us now see how Ibn Baz uses the key term 'RELIGION'.

5.7.2 'Religion' in Ibn Baz's corpus

As I have noted earlier, the word forms 'دين ' (*dīn*; religion –indef. n.) and 'دينه ' (*dīnah*; His/his religion) are keywords in Ibn Baz (keyness 33.802 and 80.315 respectively). Ibn Baz also uses the word form 'الدين ' (*addīn*; the religion) 222 times. He also uses other word forms of the key term 'RELIGION': 'الدينية' (*addīniyyah*; the religious –def. sing. pl./fem. adj.) and 'الديني (*addīni*; the religious –def. sing. masc. adj.) that occurs only once and twice respectively). Let us first consider the collocational profile of the key term 'RELIGION' in Ibn Baz.

The key term 'RELIGION' collocates with 36 words some of which are keywords in Ibn Baz. Due to time and space restrictions, it is beyond the scope of this thesis to examine all these collocates. Thus based on the criteria I have provided in Chapter 4 (section 4.2.1.1) Table 5.12 shows the selected collocates for analysis. It is an abridged version of a more detailed table (see Table A7 in Appendix A).

N	Word form	Eng. trans.	Freq.	Collocates	Eng. trans.	Freq.	Texts	Joint	MI
1	الدين	the religion	222	الْخَالِصُ	the pure	11	4	11	9.19973
2		religion	126	الإسلام	Islam	337	4	27	6.37515
3		His/his religion	73	يطى	raise the prestige of (His word and make it) superior	б	3	5	10.5413
4	ىين	religion	126	حتى	ษปี	234	3	6	4.73146
5		their religion	38	المسلمين	Muslims	226	3	7	6.73339

Table 5.12 Collocates of the word forms of the key term 'RELIGION' in Ibn Baz

Unlike Alhawali, whose collocates of the key term 'RELIGION' indicate a religion other than Islam (viz. Christianity), Ibn Baz's collocates indicate Islam. The first collocate indicates a specific form of Islam, i.e. 'الخالص' ('alḫaliṣ; the pure) which occurs jointly with 'الدين' ('adīn; the religion) 11 times in 4 texts (MI score 9.199). As indicated by the diacritics, the word is part of a Qur'anic verse (see below). The collocates 'الإسلام' ('al'islam; Islam) and 'المسلمين' ('almoslimīn; Muslims) clearly indicate the religion of Islam. The collocate 'يعلي' (vo 'li; raise the prestige of (His word and make it) superior) bears a positive association with religion (Islam). Some of these collocates are also used in clusters by Ibn Baz.

Ibn Baz uses the key term 'RELIGION' in 54 clusters (see Table A8 in Appendix A). Table 5.13 contains only 6 examples of word clusters that contain the key term 'RELIGION' in Ibn Baz.

N		Cluster	Eng. trans	Freq.	Texts
ې ي	1	في الدين ر	in the religion and	27	3
3-word	2	فې دينه و	in His/his religion and	23	3
3 4	3	في دين الله	in the religion of Allah	18	3
4	3		(So worship) Allah (Alone) by doing religious deeds sincerely for (Allah's sake)	11	4
4-word clusters	4	خبرا يفقهه فی الدين 	(If Allah wants to do good to a person), He makes him comprehend the religion.	б	3
usters	5	دینه و بعلی کلمته	(may Allah render) His religion (victorious) and raise the prestige of (His word) and make it superior	5	3
7-word clusters	б		(So worship) Allah (Alone) by doing religious deeds sincerely for Allah's sake only. Surely (the pure) religion is for Allah (only)	11	4

Table 5.13 Examples of clusters that contain the key term 'RELIGION' in Ibn Baz

Clusters 3 and 6 are parts of one Qur'anic verse tackling the issue of worship in Islam (see Section 4.2.1.2 in Chapter 4). The cluster i_{ij} (*fi dīni 'illah*; in the religion of Allah) (cluster 3 in Table 5.13) bears a positive association between the religion and Allah utilizing the grammatical annexation construct. It associates the religion with Allah which in the Islamic context means the religion of Islam. Cluster 4 introduces the religious theme of knowledge in Salafist texts. It is part of the following prophetic tradition: روى معاوية: سَمِعْتُ النبيَّ صلَّى الله عليهِ وسلَّم يقولُ: {مَن يُرِدِ اللهُ به خيرًا يُفَقَّهُهُ في الدِّينِ، وإنما أنَا قاسِمٌ واللهُ يُعْطِي، ولن تزالَ هذه الأمةُ قائمةً على أمر اللهِ، لا يَضُرُهم مَن خالفهم، حتى ياتيَ أمرُ الله} صحيح البخاري ، كتاب العلم ، المجلد الاول رقم ٣ رقم الحديث ٧١.

Narrated Muawiya: I heard Allah's Apostle saying: (If Allah wants to do good to a person, He makes him comprehend the religion I am just a distributor, but the grant is from Allah. (And remember) that this nation (true Muslims) will keep on following Allah's teachings strictly and they will not be harmed by any one going on a different path till Allah's order (Day of Judgment) is established.".Sahih Bukhari, Knowledge, Volume 1, Book 3, Number 71.⁶⁴

Ibn Baz recontextualizes this prophetic tradition to indicate what Ibn Baz believes to be the knowledge in the religion and its importance in Islam. He encourages his fellow Muslims to learn about the religion. Cluster 5 is 'دينه و يعلي كلمته' (*dīnahu wa yo 'li kalimatahu*; (may Allah render) His religion (victorious) and raise the prestige of (His word) and make it superior). Ibn Baz uses this cluster 5 times in an optative construction which always occurs in Ibn Baz's conclusions. This cluster is also frequently used as a fixed phrase in Friday sermons. The words and phrases between brackets are not shown in the Arabic version of the cluster. However, Arabic readers could infer some of the missing parts if they pertain to grammatical rules (see Chapter 3 for translation conventions used in this thesis). The optative construction that includes cluster 5 is provided above. It constitutes one of the features of any Islamic religious writing that starts with an introduction (proem). The proem starts with the basmallah i.e. saying 'In the name of Allah, Most Gracious, Most Merciful' then is followed by praising Allah which is a recontextualization of the first verse in the Qur'an in the opening Chapter (Praise is due to Allah the Lord of mankind and jinn). The proems usually contain prayers of success mostly pertaining to the themes in the writing followed or are concluded by Calling Allah to send His blessings and peace upon His messenger (Mohammad).

The last cluster I want to consider in this analysis is cluster 6 which is used by Ibn Baz as part of the following Qur'anic verses:

> {إِنا أَنزلنا إليك الكتاب بِالحق فَاعْبُدِ الله مُخْلِصًا له الدين * ألا لله **اللَّينُ** الْحَالِصُ و الذين اتَّخَذُوا من دونه أولياء ما نَعْبُدُهُمْ إلا لِيُقَرِّبُونَا إلى الله زُلْفَى } ايه ٢-٣ سورة الزمر.

^{64.} The translation of this Hadith is available on the following link: <u>http://www.usc.edu/org/cmje/religious-</u>texts/hadith/bukhari/003-sbt.php.

These two Qur'anic verses could be translated as:

{Verily We have sent down the Book to you (O Mohammad) in truth: So worship Allah (Alone) by doing <u>religious</u> deeds sincerely for Allah's sake only* Surely the religion (i.e. the worship and the obedience) is for Allah only. And those who take Auliya' (protectors, helpers, lords, gods) besides Him (say): "We worship them only that they may bring us near to Allah."} Holy Qur'an 39: 2-3.

Ibn Baz tries to legitimate his stance towards certain ritualistic practices. He recontextualizes these Qur'anic verses (11 times) in different topics that have one major theme, namely, reporting the worship of Allah and rejecting what Ibn Baz believes to be associationism or polytheistic practices i.e. refutation of some Sufi and Shiites' worshipping practices such as beseeching help from dead saints (see the subsections in 4.2 in Chapter 4). Additionally, Ibn Baz utilizes the key term 'RELIGION' to achieve different legitimation purposes. Let us consider some concordance lines that repeat the same function.

Concordance 5.19 Concordance of the key term 'RELIGION' in Ibn Baz's corpus

N	
	الأمر الثالث (في الدعوة) : بيان الأمر الذي يدسى إليه [] و هو الإسلام و هو يين الله الحق [] و يدخل في ذلك الدعوة إلى الإيمان بكل ما تخبر الله به و
1	رسله [] و الأخذ بما شرع الله في الطهارة و الصلاة [] و الحرب و السلم و في كل شيء لأن <u>فين</u> الله عز و جل دين شامل []
1	The third issue (in calling for Islam): Callers for Islam should make it clear [] it is Islam the true religion of Allah []
1	Calling to Islam includes inviting people to believe in all that Allah and His apostles told [] and perform what Allah
	has ordained in purification and prayers [] peace and war and in everything, because the religion of Allah is complete
1	فقال : يا شيخ سأبايعك على دين الله و رسوله ، و على الجهاد في سبيل الله [] و تواف التاس إلى الدرعية من كل مكان [] و رنب الدروس في الدرعية في
	العقائد ، و في القرآن الكريم [] العلوم العربية ، و التأريخية ، و خير ذلك من العلوم النافعة
	[] and then he (Mohammad Ibn Saud the founder of the first Saudi state 1744-1818) said: O Sheik (Mohammad Ibn
2	Abdul Wahhab, a Salafi religious reformer 1703-1792) I will give you the oath of allegiance on the basis of the religion
1	of Allah and His messenger and making the Jihad for Allah's sake [] and people from different parts of the region then
	came to Diriyah (in central Saudi Arabia) [] he then started courses in creed, Qur'an [] Arabic, history and other
	beneficial fields of knowledge []
1	[] و قد يقع فيها (الاحتفالات بالموالد) [] اشرك الأكبر ، و نلك بالغلو في رسول الله [] و الاستغاثة به [] و أعتقد أنه يعلم الغيب ، [] و قد صبح عن رسول الله صلى الله عليه و سلم أنه قدل : ((إيكم و الغلو في الدين فايتما أطلك من كان قبلكم الغلو في الدين) []
1	
	[] and Muslims could commit major polytheism in (the celebrations of the prophet's birthday). This could happen in
3	exceeding the proper bounds in the belief of the apostle of Allah [] beseeching his help [] and believing that he knows the unseen []. Truely, it was narrated in a sound Hadith that the apostle of Allah, may Allah send His
	blessings and peace upon him said: (Beware of exceeding the proper boundaries in the religion, indeed those nations
	before you were destroyed by exceeding the proper boundaries in the religion) []
\vdash	الما تقديد الو هايد : هي التمسك بكتاب الله و سنة رسوله صلى الله عليه و سلم [] ها كان عليه السلف الصالح ، و أنمه الدين و الهدى ، أهل النقد و التكوى في
	باب معرفة الله ، و إثبات صفات كماله و تعوت جلاله ، التي نطق بها الكتاب العزيز ، و صحت بها الأخبار النبوية []
	The beliefs of Wahhabis involves the strict adherence to the Book of Allah and the Sunnah of His apostle, may Allah
4	send His blessings and peace upon him [] and following the pious predecessors, the Imams of the religion and
	guidance who are the people of jurisprudence and Fatwas in knowing Allah and confirming the perfect attributes of the
	Majestic that were reported in the glorious Qur'an and sound Prophetic traditions []
	[] ولما كان عمل الفين [] وصبغ الحية بتقالية جامعة و معلومة و واضبحة ، و برص الصفوف طي إحساس مشترك ، و دفعها الي مصبر واحد ، في
	الاستعمار استهدف إقصاء الدين عن أفاق البلاد كلها، و تكوين أجيال غريبة عنه، إن لم تكن كار مة له []
5	[] and when the religion aims to [], mould life with known and comprehensive traditions, unite people and make
	them share the same feelings and destiny, colonialism (colonialists) aims to exclude the religion from life and prepare
	new generations who do not know it or rather hate it []

Ibn Baz utilizes the key term 'RELIGION' in theological and (occasionally) political themes. Line 1 indicates Ibn Baz's concept of Islam as the true religion. Line 2 is about the political theme of 'بيعة' (*bai'ah*; oath of allegiance) to the ruler or religious leader. Line 2 indicates the start of the coalition between the Salafi leader Sheikh Mohammad Ibn AbdulWahhab (1703-1792) and the political leader Mohammad Ibn Saud (the founder of the first Saudi State (1744-1818). Although Ibn Baz's writings are mostly theological, in this Line, he writes about the establishment of the first Saudi State from a Salafist perspective which supports making oaths of allegiance between both religious and political leaders.

In Line 3, Ibn Baz refers to what he believes to be polytheistic practices of the Sufis when they celebrate the birthday of the prophet of Islam. He claims that such practices might fall in the category of greater polytheism by exceeding the religious proper bounds in the belief in the Prophet of Islam. Ibn Baz utilizes a prophetic tradition about exceeding the proper religious bounds and recontextualizes it to include some of the beliefs about the Prophet of Islam such as beseeching his help and believing that he knows the unseen.

In his defence of 'wahhabism', Ibn Baz claims that it is a strict adherence to the teachings of the Qur'an and the Sunnah of the Prophet of Islam. Additionally he claims that 'wahhabism' is also following the teachings of the 'السلف الصالح' ('assalaf 'aṣṣāliḥ; the pious predecessors). Here Ibn Baz legitimates what he calls 'wahhabism' as being Salafist and based on the teachings of Qur'an and the Prophet of Islam. Finally, Ibn Baz opposes colonialism, projecting it as aiming to exclude religion from the lives of Muslims.

In this section we have seen that both Salafi writers (Ibn Baz and Alhawali) utilize the key terms 'MIND' and 'RELIGION' in different semantic foci. Safar Alhawali seems to be concerned with negative representations of Christianity and (medieval and modern colonial) western threats (including Israel) whereas Ibn Baz's representation of both mind and religion

174

is mostly concerned with articulating concepts and belief in and about Islam that are viewed as correct, and more precisely the Salafist form of Islam.

5.8 Conclusion

This chapter ends the analysis of the semantic foci in both Salafi corpora. The main theme in this chapter is semantic foci keywords in Alhawali's writings. Collocational profiles (if there were any) for the semantic foci keywords in both corpora have been sketched out followed by cluster and concordance-based analyses.

In the previous chapter we saw that Ibn Baz's semantic foci keywords appear to be mostly theological which reflect his conception towards ritualistic practices marked by his recontextualizations of Qur'anic verses and Prophetic traditions as well as Salafi leaders' writings. He also tries to refute the Sufi and Shiites' practices that he believes to be against the true teachings of Qur'an and Prophetic traditions. On the other hand, Alhawali's semantic foci are more overtly political. Alhawali from a political Salafist perspective draws the attention of his readers to what he believes to be an external western menace that threatens both Muslim and Arab countries and has been doing so for centuries. Additionally, Alhawali repeatedly defames western political, social and religious lives.

It is also noteworthy, that Ibn Baz writes, on a smaller scale though, about some political themes. He tries to refute Arab nationalism that was supported and perpetuated by Abdul Nassir of Egypt and his allies (the Baath regimes) in Iraq and Syria (for more information on Arab-Israeli Conflict see Rea and Wright (1997)). He also shares (to a lesser extent) some of Alhawali's views concerning colonialism and the West (see Chapter 8 for more explanation).

Both Salafi authors write also about social actors and utilize arguments in their writings which will be the foci of Chapters 6 and 7 respectively.

Chapter 6: Social Actor Representation of "the Other" in both Salafi Corpora

6.1 Introduction

This chapter is a continuation of the analysis of the Salafi corpora. It brings together an aspect of analysis that was touched on in part in the previous two chapters: social actor representations. It responds to research question 1(b): What do concordance and collocational analyses of keywords and frequent words in the two corpora of Salafi discourse reveal about representations of social actors (nominational strategies)?

Based on further examination of keywords and utilizing van Leeuwen's social actor theory, Sections 6.2 and 6.3 tackle the representation of social actors in both Salafi corpora respectively. Sections 6.2.1 and 6.3.1 are dedicated to analysing the representation of 'the Other' in Ibn Baz's corpus and Alhawali's corpus respectively. Finaly, section 6.4 concludes this chapter. Let us first examine the representation of social actors in both Salafi corpora.

6.2 Social actor representation: Ibn Baz's keywords

This section focuses the analysis on the keywords that indicate social actors, some of which were briefly presented in Chapters 4 (pp. 73-93) and 5 (pp. 124-146). Table A3 (Appendix A) shows 75 keywords that indicate social actors in Ibn Baz which I have grouped into five categories as shown in Table 6.1.

Table 6.1 Categories of social actors in Ibn Baz (derived from keywords)

N	Category	Key word
1	The Prophet of Islam and other prophets	(may) the blessings (of Allah be upon him), messenger. His messenger, the messenger, the messengers, His messengers, messenger (accusative), the prophet, (may) peace (of Allah be upon him), Mohammad, our Prophet, his family, His Prophet, Mohammad (accusative)
2	The Salaf 1: The Companions of the Prophet	son of, (may Allah) be pleased (with him), his (the Prophet) companions, the father of (accusative), the father of (nominative), (may Allah be pleased) with him, his (the Prophet) companions, the companions of the Prophet, (may Allah be pleased) with them, Hurairah, Omar
3	The Salaf 2: The successors of the companions of the Prophet	son of, (the pious) ancestors, the father of (accusative), the Imam, (may Allah have) mercy (upon him), Muslim, the father of (nominative), the successors of the companions of the Prophet, the sheikh, Imams, pious (ancestors), Alawzaee, Malik, their followers
4	General Social actors	the people of, (they, those who) believed, O, (those, people) (vocative), O, (vocative), the Sunnah (of the Prophet of Islam)/ the Sunnis, indeed they are, all, His servants, His worshippers, servant, worshipper, the servants of Allah/ His worshippers, to them, the Muslims (accusative), with them, that you may, from you, they, to His servants/worshippers, others, the father of (nominative), 'althaqalain (the humans and the jinn (genies)), who (phural), other, you were, to you, the believers, 'alealameen (mankind, jinn and all that exists), the Jinn, the teachers/ propagandists, His creatures, his patrons' advocates, after them, to the people of, upon you (phural)
5	The Other	teachers/ propagandists, the people of, polytheists (accusative), polytheists (nominative), friends of God (religious status in Sufi orders)

It is worth noting here that when translated into English some of the keywords in Table 6.1 become multi-word units. This is because the Arabic writing system reflects the morphological structure of Arabic which is substantially more complex than that of English, and as a result some pronouns as well as the definite article are always attached to verbs and nouns (some of which are keywords in this thesis). The definite article (al; the), for example, is always attached to its following noun as in the Arabic word (al; al; the), for example, is always attached to its following noun as in the Arabic word (al; al; alwasil: the messenger). In the case of pronouns as in (alwa; (anho; with him), in the first line in category 2 in Table 6.1, the pronoun <math>(a, (h; him)) is preceded with a preposition (alwa; (alkal), (alkal)) (*alali*) which, in all the occurrences, is part of the optative and glorification phrase (alkal), (alkal), (alkal) (*alali*) that contain the preposition (alkal), (alka

^{65.} The case is so with its plural form 'عنهم' ('*anhom*; with them) in the optative phrase 'رضي الله عنهم' (*radiya 'allahu 'anhom*; may Allah be pleased with them).

^{66.} For this specific example see footnote 33.

phrase 'to His servants/worshippers' (see Section 3.3.2 for an overview of the Arabic writing system and the translation conventions of the keywords into English in this thesis).

There are also some keywords that appear in two categories such as 'teachers/propagandists' which is a translation of one Arabic word ' $i = (do'\bar{a}h)$. Such keywords have been placed in two categories because they are used by Ibn Baz to indicate two meanings depending on the noun that follows it. The first meaning indicates those who teach Islam and thus the word ' $i = (do'\bar{a}h)$; teachers) refers to the in-group social actors. Second, if the following noun indicates other teachings such as atheism then the meaning of the word becomes 'propagandist' in English and at face value it indicates 'the Other'. The case is the same with the keyword 'the people of' (see below). I classified the keywords 'other' and 'others' in the category 'General social actors' in Ibn Baz as they do not belong to the category 'The Others'. The case is so, because Ibn Baz uses these two words to refer to anonymous social actors as in 'other Muslims' and 'the other person' which do not refer to the category 'The Other'.

As indicated by Table 6.1, there are five categories of social actors in Ibn Baz. Let us have a brief look at these categories and see how they are sociosemantically represented (van Leeuwen, 1996, 2008). It should also be noted that, although the key term 'ALLAH' could be categorized as a social actor, as I have demonstrated in Chapter 4, in my data it appears more as an important Islamic theme that is associated with different semantic foci and has already been examined.⁶⁷ Thus it will not be included in the analysis in this chapter.

The largest category of social actors (in terms of the number of word types) indicates general social actors who are mostly anonymously nominated by personalization via

^{67.} The case of religious deities raises interesting questions for social actor theory and CDA in general. Such deities are often conceived of as omnipotent, causing everything to happen yet are unseen by humans (requiring faith to believe in their existence). Thus, their actions are relayed second-hand by humans or through religious texts that they are supposed to have dictated to humans. Additionally, some religious people may view it as inappropriate to analyse deities in terms of them being 'social actors'. Indeed, carrying out CDA from a religious stance raises its own issues, particularly if a religion does not condone certain forms of criticism.

categorization (such as 'الغلن ('annas: the people), 'لجن ('alǧinn; genies)) by means of both functionalization (such as 'عباده' ('ibadoh; His worshippers -nom.) and identification by classification such as the religionym (Reisigl & Wodak, 2001, p. 116) ('almoslimīn; the Muslims -acc.).⁶⁸ Here, Ibn Baz refers to general (or ordinary) social actors that are sometimes not given any active role in his texts other than performing what he believes to be ordained by Allah. They are passive and assigned to a certain role as in 'عباده' ('ibadoh; servants/worshippers) whose primary role in this life is to worship Allah. The keyword 'العندة ('assonnah; the Sunnah of the Prophet/Sunnis) has more than one meaning in Ibn Baz's texts. But what concerns us here is the sense that indicates social actors i.e. 'the Sunnis'. It refers to the Sunnis (as opposed to the Shiites or Sufis) which is sometimes preceded by the word 'أهل' ('ahl; the people of). Aggregating a group of social actors within Muslim communities (defining the in-group social actors) is an othering strategy which focuses the attention of Ibn Baz's readers to those who are different from Sunnis which indicates Ibn Baz's polarization of Muslim societies into two opposing groups, namely, the Sunnis as opposed to the Sufis or Shiites.

The second largest category in Table 6.1 indicates the Prophet of Islam and other prophets. The social actors in this category are positively represented by nomination as realized by the proper name 'محمد/محمد' (mohammad/mohammadan; Mohammad -nom. and acc. respectively), and by categorization (such as 'رسول' (rasūl; messenger), 'رسله' (rosolih; His messengers) which all are mostly followed by the honorific optative construction ' معلى الله ' (salla 'allahu 'alaihi/'alaihim wa sallam; may Allah send His blessings and peace upon him/them) (see Section 4.2.1.3).

The categories that indicate the two types of the Salaf, viz., the companions of the Prophet of Islam and their successors, also constitute a large segment of the social actors

^{68.} Another form of categorisation which runs across all the sets of words in Table 6.1 is relational identification (van Leeuwen 1996: 54-55). For example, 'عباده' ('*ibadoh*; His worshippers) also suggests that the worshippers have a relationship to Allah.

represented in Ibn Baz. The social actors in this category are mostly represented by nomination, either by their proper names such as 'ععر' ('*omar*; Omar –a famous companion of the Prophet and the second Caliph in Islam), 'لأوزاعي' ('*al'awza'ī*; Alawzaee a Salafi medieval religious leader) or by honorification by means of affiliation using words that indicate kinship such as 'بن' (*bin*; the son of) or a cognomen such as 'lee(*i'abo/'abi*; the father of – nom. and acc. respectively). Nominating medieval (Salafi) social actors, mostly followed by optative constructions such as 'may Allah have mercy on him' that show respects being paid to them, indicates Ibn Baz's interests in the Salafi teachings that he adheres to (see Section 7.4).

All of the categories mentioned so far designate in-group social actors who, when I examined their concordance lines, were found to be mostly positively represented. However, a smaller set of social actor words (some of which are mentioned in the sacred texts of Islam) in Ibn Baz fall under the category that I have labeled as the 'Other'. All of the social actors in this category are represented by categorization by means of identification viz., they are identified by categorizing them in terms of what they are. Examples of these nomination strategies are 'المشركون/المشركون/المشركين' ('almošrikūn/'almošrikūn; the polytheists –nom. and acc. respectively) and functionalization such as ' $c_2 = \frac{1}{2}$ ' (do 'āh; propagandists). The last strategy indicates that the social actors (the Other) are engaged in an activity that is determined by the meaning of the following noun (see Section 6.2.1.1 below).

It is this last category that I wish to concentrate on, as the construction of 'the Other', I presume, on the one hand, can help to reveal Ibn Baz's ideological stance more clearly to delegitimize certain social (religious) practices associated with 'the Other' (such as those rituals of Sufis and Shiites detailed in Section 6.2.1.1 below). These rituals are practiced by what the two Salafi authors deem to be 'the Other' (Sufis and Shiites), some of whom are Saudis. Additionally, the aim of CDA analysis is to investigate social inequality which could be 'expressed, signaled, constituted, legitimized and so on by language use" (Wodak, 2001b,

p. 2). On the other hand, the aim of the analysis in this thesis is to define the boundaries of the two Saudi Salafi strands which could be realized more obviously in representing social actors who belong to the out-group rather than the in-group.

It is worth noting here that, due to space and time restrictions, I shall detail only the analysis of the top two keywords that indicate the 'Other' utilizing the following criteria. 1- Only keywords that occur at least five times in the reference corpus are considered for detailed analysis, so that I am able to compare the representation of social actors that are indicated by these keywords across both corpora.

2- If there were more than two keywords that indicate 'the Other', then the keywords that occur most frequently in the reference corpus are selected for the analysis in order to include the most frequently used (theological/political) keywords in the reference corpus.

3- Keywords that belong to the same lemma or word-form are tabulated and grouped under a key term (written in small capitals) and their frequencies are added together. And if they occur in the reference corpus and have other word-forms, they are also referred to and included in the analysis.

4- Arabic words that are homographic or polysemous are not identified by Wordsmith as such. Thus, they needed to be first discerned via concordance lines. If more than 50% of the concordance lines do not indicate a single meaning (e.g. a religious or political meaning), they will not be included in the analysis as the keyness of such words might be uncertain.

5- Keywords that I have presented in earlier chapters (such as 'المشركون' ('almošrikūn; polytheists) will not be included in the analysis in order to avoid repetition. Now let us consider the representation of the 'Other' in both corpora.

6.2.1 The representation of the 'Other' in Ibn Baz

There are four social actors in the category 'Other' in Ibn Baz. Applying the criteria for selecting keywords for analysis that I have stated above, I shall delineate the

181

representation of the social actors that are indicated by the keywords that refer to the 'Other' in Ibn Baz, namely, 'دعاة' (*do* 'āh; teachers/propagandists) and 'أهل' (*'ahl*; the people of). And for the purpose of comparing the two corpora that comprise my research data, I shall also examine how they are represented in Alhawali. Let us first start with the representation of 'دعاة' (*do* 'āh; teachers/propagandists).

6.2.1.1 The representation of 'دعاة' (do'āh; teachers/propagandists) in Ibn Baz and Alhawali

As is the case with explicit ideological (religious and political) texts (van Dijk, 2011), there is an explicit representation of both the in-group and out-group social actors. The keyword 'دعاة' (do 'āh; teachers/propagandists –indef.) refers to two types of social actors, the identity of which is distinguished by the noun which follows the word, functioning as part of an annexation/genitive construction.⁶⁹ In Arabic, indefinite words become definite when used in genitive structures (Ryding, 2005, p. 160). The connotation (positive or negative) of the second noun determines whether this keyword means teachers or propagandists. There are four possible Arabic word forms of the 'cale'' (do 'āh; teachers/propagandists –indef.), namely, 'cale'' (da 'i/ da 'yah; teacher/propagandist –indef.), 'cale'' (adda 'i/'adda 'yah; the teacher/propagandist –def.)⁷⁰, 'cale'' (do 'āh; teachers/propagandists –indef.) and 'lcale'' ('addo 'āh; the teachers/propagandists – def.). However, Ibn Baz mainly uses the last two of the possible forms of this keyword, viz., 'cale'' (do 'āh; teachers/propagandists – indef.) and 'lcale'' ('addo 'āh; the teachers/propagandists – def.). Let us first examine the collocates of these two word forms:

^{69.} Genitive (or annexation) constructions in Arabic occur when two nouns are 'linked together in a relationship where the second noun determines the first by identifying, limiting, or defining it, and thus the two nouns function as one phrase or syntactic unit' (Ryding, 2005, p. 205).

^{70.} The word forms 'داعي' (*da'i*; -indef. sing.), 'الداعي' (*'adda'i*; -def. sing.) could also mean to cause or 'calling to'. Also, the word form 'لداعي' (*'adda'i*; the teacher -def. sing.) is mostly used by the Ismalites (a branch of Shiite Islam mostly concentrated in Najran, southern Saudi Arabia) to indicate a high rank religious status whose historical job is to call people for the teachings of Ismaili faction. Thus it is not expected to be used by the Salafis. However, the word forms 'داعية' (*da'yah*; the teacher -indef. sing.) and 'الداعية' (*'adda'yah*; the teacher -def. sing.) are used to nominate a Sunni religious preacher.

Table 6.2 Collocates of 'teachers/propagandists' (do 'āh; دعاة) in Ibn Baz

keyword	Eng. trans	Freq.	Collocate	Eng. trans	Freq.	Joint	MI
دعاة	teachers/	49	الهدى	guidance	67	5	7.347
	propagandists	49	الدق	the truth (Islam)	242	8	6.173
	مؤلاء these هؤلاء	these	59	5	7.686		
الدعاة	propagandists	44	إلى	to	1104	11	6.173
	Propaganations		ان	truly	1605	10	3.920

As shown in Table 6.2, the plural word forms of the key term 'TEACHERS/PROPAGANDISTS' have 5 collocates. The indefinite keyword 'دعاة' (do 'āh; teachers/propagandists) is jointly used in genitive constructions with the definite collocates ('للهدى' ('alhoda; the guidance) and 'اللهدى' ('alhaq; the truth i.e. Islam) making the phrases 'دعاة الهدي' (do 'āt 'alhoda; the teachers of (the) guidance) which occurs 5 times, and 'دعاة الهدي' (do 'āt 'alhaq; the teachers of the truth/Islam) which occurs 8 times. These two collocates suggest positive associations (guidance, the truth i.e. Islam), which are used by Ibn Baz to represent the in-group social actors. The definite form of the key term 'TEACHERS/PROPAGANDISTS' has 3 grammatical words that, at face value, do not appear to carry any positive or negative meanings. It is worth noting here that all the occurrences of 'الدعاة' ('addo 'āh; the teachers/propagandists – def.) are also used to represent in-group social actors which makes the meaning of this keyword in English to be 'the teachers'.

Because this word has two meanings, the collocates in the table above have tended to be used with the positive meaning of the word, rather than contributing towards the construction of 'the Other', and this was backed up when I examined concordances of these collocational relationships. Therefore, a collocational analysis is not appropriate here in order to examine how Ibn Baz constructs the Other with this keyword. Instead, I have focused on a concordance based analysis of the keyword 'دعاة' (do 'āh; propagandists) as it refers to social actors who are considered by Ibn Baz as the 'Other'. An examination of concordance lines showed that 'دعاة' (do 'āh; propagandists) is used in a genitive construction with 'Leas' ('algawmiyyah; nationalism) (9 times), 'الشيوعية' ('aššiyo 'iyyah; communism) (3 times) ('al'ihtilāt; mixing –between men and women in jobs etc.) (2 times), and 'الماسونية' ('al'ihtilāt; mixing –between men and women in jobs etc.) (2 times), and 'الوثنية' ('alwataniyyah; paganism), 'النار 'annār; Hell-fire) and 'الزيغ و الضلال' ('azzayġ wa 'aḍalāl; deviation (from the truth) and error) (once each). Concordance 6.1 below shows five concordance lines of the keyword 'دعاة' ((do ʿāh; propagandists)) that were chosen in order to demonstrate how Ibn Baz constructs the Other.

Concordance 6.1 Concordance of 'دعاة' (do 'āh; propagandists) in Ibn Baz

Ν	Concordance
	[] ما يعتده الملاحدة في هذا العصر من أتباع ماركس وليثبين و غيرهما، من دعاة الإلحاد و الكتر، سواء سموا ذلك الستراكية أو شيوعية أو جير ذلك []
	[] the doctrines of the modern atheists who are the followers of Marx and Lenin and other <u>propagandists</u> of atheism and disbelief, whether they named that Socialism, Communism or Baathism or whatsoever []
	و من تأمل أحوال أهل العلم [] كأبي العباس بن تيمية رحمه الله []. تعم من تأمل أحوالهم، و فتح الله عليه بنهم ما قالوا و ما كثيرا رأى [] العلوم ألصحيحة ، [] و يتضبع له أن من خالتهم من <u>دعاة</u> الزيغ و الضبلال ، ليس عندهم إلا الشبهات الباطلة ، و الحجج الزائلة
2	and whosoever studies the biographies of the people of [religious] knowledge such as Abi Alabaas Ibn Taymyyia, may Allah have mercy on him, []. Indeed whosoever studies their biographies, and Allah guided him to understand what they had written, will realize [] the authentic knowledge' [] and he will also realize that their opponents from among the <u>propagandists</u> of deviation (from the truth) and error, only produce erroneous suspicions and false arguments
	و هنا نبية ينكرها بعض دعاة القرمية أحب أن أكشفها للقارئ، و هي أن بعض دعاة القرمية زعم أن النهي عن الدعوة إلى القومية العربية و التحلير منها يتضمن تنقص العرب وإنكار فضلهم [في نشر الإسلام]
3	and here is a suspicious claim mentioned by the propagandists of (Arab) nationalism which I would like to expose to the readers: some <u>propagandists</u> of (Arab) nationalism claim that banning and warning against nationalism implicitly includes defaming Arabs and denying their contribution [in the spread of Islam]
	[] قد بعته الله بكتابه العظيم، و الرسانة العامة التي أوضح بها كل تسيء، تم يأتي دعاة الماسونية الذين يريدون أن يردوا الناس إلى الأحوال البهيمية، []
4	(Mohammad) whom Allah sent with Qur'an and the general message that includes every aspect of life, whereas the propagandists of Masonry want us to revert to the animalistic life []
	[] قد وفقوا في إظهار الحق وبيان أدلته، وأوضحوا ما يتعلق بدعوة التوحيد، و الرد على دعاة الوتتية، وعباد القيور، وبرزوا في هذا السبيل، وكاتوا على النهج المستقيم، نهج انسلف العسالح []
5	[] (the previous Salafi leaders) were rightly guided to make the truth prevail showing the indicative Qur'anic Sunnah proofs. They explicated the monotheistic beliefs and refuted the claims of the propagandists of paganism and worshippers of graves. They were prominent in this respect. They followed the guidance of the rightly guided pious predecessors []

The genitive constructions in all the lines are the linguistic realization of the representation of 'the Other' by personalization via categorization of those who advocate non-Islamic ideologies. ' $(do'\bar{a}t 'al'ilh\bar{a}d;$ propagandists of atheism) in Line 1 is a representation by categorization of those whom Ibn Baz claims to propagate non-Islamic ideologies, whether Socialist, Communist or the ideologies of the Baath parties in Iraq and Syria which Ibn Baz sees as being coated in Arab nationalism (see below). Here, Ibn Baz

groups many social actors under one very general category (atheists) using hypotactic grammatical structures, i.e. sentences with connective particles. The disjunctive 'j' ('*aw*; or) 'indicates an option between two or more elements' and that option 'may include all elements' (Ryding, 2005, p. 418). 'j' ('*aw*; or) is used by Ibn Baz to include indiscriminately all options he introduces, viz. '*imic lex.*' (*'istarākiyyah*; socialism), 'or' '*inic i yuū 'iyyah*; communism) 'or' '*inic i jiyyah*; Baathism).⁷¹ In doing so, Ibn Baz associates all these social actors with atheism and disbelief. Additionally, he uses the additive particle '*g*' (*wa*; and) in order to represent the atheists in greater detail: '*and* some of their [atheists'] fundamentals include denying the Hereafter (*and*)⁷² Paradise *and* Hell-fire *and* disbelief in all religions'. This is an explicitly negative representation as it associates them with what Ibn Baz claims to be atheists and disbelievers (Marx and Lenin), thus presenting them as 'Others' who are seen as being lost and opposing the true Muslim creed.

Nationalism (Line 3) is the most common word used by Ibn Baz in the genitive constructions with the negative meaning of the keyword 'دعاة' (do 'āh; propagandists). All the occurrences of this genitive construction 'دعاة' (do 'āh; propagandists) + 'القومية' ('alqawmiyyah; nationalism) appear in one text of Ibn Baz carrying the title 'Refuting (Arab) Nationalism'. Arab nationalists are constructed by Ibn Baz as deviating from what he considers to be the right path and calling for erroneous beliefs that oppose the 'pure teaching of Islam' (for a detailed explanation of the representation of Arab nationalism see Chapter 8).

^{71.} In Arabic 'او' ('aw; or) is always put before each noun that is used as an option whereas it appears before the last noun in the English translation of concordance line 1 (see Chapter 3 for the translation conventions used in this thesis).

^{72.} In Arabic, the additive particle '3' (wa; and) is used within sentences to link clauses, phrases, and words which is different from English where a comma is used to separate each item (for more details see Chapter 18 in Ryding (2005).

Propagandist of deviation (from the truth) and error (Line 2), propagandists of Masonry (line 4) and propagandists of paganism and worshippers of graves (the latter implies some Sufi and Shiite practices) (line 5) are all negatively represented as people who have deviated from the right path. Although lines 2 and 5 refer to the 'Others' who are astray, they are also used to indicate the job of the (in-group) pious religious leaders who are viewed as rightly guided in refuting the 'claims' of the worshippers of false gods. In summary, then, Ibn Baz uses the keyword 'c.alis' (do'āh; propagandists) to negatively represent 'the Other' (especially nationalists) who he believes to be against the teachings of Islam. Now let us look at how the other author, Alhawali, uses the same word 'c.alis' (do'āh; teachers/propagandists) to represent social actors.

Alhawali uses the words 'دعاة' (do ' $\bar{a}h$; teachers/propagandists) (33 times) and 'iLcali' ('addo ' $\bar{a}h$; the teachers/propagandists) (13 times). Alhawali also uses other word forms of (calify; teachers/propagandists) as shown in Table 6.3.

Table 6.3 Word forms of 'دعاة' (do 'āh; teachers/propagandists) in Alhawali

Ν	Word		Freq.	Texts
1	دعاة	propagandist	33	3
2	الدعاة	teachers/propagandists	13	5
3	دعاتها	its propagandists	4	1
4	الداعى	the propagandist/causing	2	1
5	لدعاة	for the propagandists	1	1
6	لدعاتها	for its propagandists	1	1

Alhawali uses 6 word forms of 'aligned's (do' $\bar{a}h$; teachers/propagandists) 56 times in total. However, based on a preliminary concordance analysis, it appears that except for the words ' $('ddo'\bar{a}h)$; the teachers/propagandists) and ' (i_{ℓ}) ' (' $add\bar{a}'i$; causing), all the word forms are used to indicated propagandists. The word forms ' $(do'\bar{a}h)$; teachers/propagandists) and ' (i_{ℓ}) ' ($i_{\ell}do'\bar{a}t$; for the propagandists/teachers of) appear in genitive constructions to denote different sorts of social actors whose identities are revealed by the meaning of the second noun. These keywords have no collocates in Alhawali. A concordance-based analysis shows how he uses them though. I shall mainly focus on the most frequently used genitive constructions that denote 'the Other' which is 'دعاة' (do 'āh; propagandist) (33 times). Alhawali uses 'دعاة' (do 'āh; propagandist) to represent different social actors from those in Ibn Baz by personalization via categorization and mostly by means of association. It is used in Alhawali in genitive constructions with 'اللادينية' ('alladiniyyah; irreligion) (5 times), 'التغريب' ('attaġreeb; westernization) (4 times), 'العلمانية' ('al'ilmāniyyah; secularism) (2 times). Alhawali includes westernization) (4 times), 'العلمانية' ('al'ilmāniyyah; secularism) (2 times). Alhawali includes (each of which is mentioned once) in one of his texts that is mainly written to criticize secularism. Additionally, Alhawali uses the keyword 'دعاة' (do 'āh; propagandist) with the nouns 'دعاة' ('attafarnoğ) (once), which is also translated into English as 'westernization' and 'دعاة' ('al'aşriyyah; modernity) (once). Concordance 6.2 shows the range of different ways that Alhawali uses these nouns in genitive constructions with 'دعاة') ('al'āhawali uses these nouns in genitive constructions with 'دعاة') ('al'aşriyyah; besond in genitive constructions with 'دعاة') ('a'ağı besond in the secularish ' دعاة') '('al'aşriyyah; modernity) (once). Concordance 6.2 shows the range of different ways that Alhawali uses these nouns in genitive constructions with 'isa' (do'āh; propagandist).

N	Concordance
	[] أما المهزلة التاريخية العظمي (صحوك الغفران)، [] فلا يستطيع أحد من أعداء الإسلام أو دعاة العلمانية [] أن يزعم أنها وجدت في التاريخ
	الإسلامي []، فهذه المهزلة الأضبحوكة لم يعرفها المنظمون حتى في أحط و أحلك عصبور هم حين فتبا الجهل []
1	[] as for the greatest historical farce (Indulgence: full or partial remission of temporal punishment granted by the
[-	Church), [] no one of the enemies of Islam or propagandists of secularism [] can claim that it has occurred in the
ļ	history of Islam [], Muslims have never known this finniest farce even during their darkest ages when ignorance of
	religion prevailed []
	[] كان الأزهر[] تلعة إسلامية يحسب لها أعداء الإسلام كل حساب، [] و لذلك ظل الأزهر سلين طويلة محط المقت و مصب اللحات من قبل
	دعاة التغريب و الخدينية، حتى جعلوه رأس المتساكل الثقافية في مصدر و العقبة الكود في سبيل النهضية إ
2	the Azhar (a religious educational institution in Cairo, Egypt) was a huge religious educational institution that instigated
-	the fears of the enemies of Islam [] thus (the Azhar) was for many years the target of hatred by the propagandists of
	westernization (of Muslim countries) and irreligion to the extent that they made it the cause of all cultural problems and
	an obstacle that hindered development in Egypt!
	[] تم جاء الجيل المستحد للغرب معاناً عداوته الاسلامية و اللغة العربية. و أشهر زعمائه أحمد لطفي السيد، و زميله و رفيق عمره عبد الاست
	العزيز فهمي، وزوج أخله إسماعيل مظهر، تم صديقه الحميم له حسين.[] لقد كان كل دعاة العامية أناساً مسبو هين و صالتهم بالدوائر. الاستعمارية واضحة. و ذلك ما يؤكد أنها كانت جزءاً من المخطط اليهودي الصليبي للقصاء على الإسلام []
-	[] then the West-enslaved generation appeared declaring their enmity to the Islamic culture and Arabic Language.
ذ	From amongst their leaders are Ahmed Lutfi El-Sayyed (an Egyptian intellectual 1872-1963), his colleague and close
	friend Abdul Aziz Fahmi (an Egyptian politician 1870-1951),and his brother in law Ismaeel Mazhar (1891-1962) and then his close friend Taha Hussein (an Egyptian writer an intellectual 1889-1973) [] indeed all the propagandists of
	using the colloquial [Egyptian Colloquial instead of Classical Arabic] were suspected individuals and their connections
	with the colonial powers are obvious, which confirms that it was part of the Judo-crusaders' plan to eradicate Islam []
	(…) معاد محمد معاد معاد معاد معاد معاد معاد معاد مع
	[] تا ملکو [مسی میں] سی ملکو میں جاری کی و روسو و مسی میں معربیوں ، و مال مع مل دعام ملاقت الدور الج و العصر بدار]
4	[] as for his [Lutfi El-Sayyed (an Egyptian intellectual 1872-1963)] intellectual writings, he was under the influence of
	Darwin, Mill (John Stuart), Rousseau and this sort of Westerners, and he used to support every offensive propagandist
	of westernization and modernity []

Line 1 (Concordance 6.2), which also includes an insinuation of a negative representation of Christianity, indicates that Alhawali uses a hypotactic grammatical structure (sentences with the particle ' j^{i} ('*aw*; or) that functions as an additive (see Ryding, 2005, p. 418) to create an association between the 'enemies of Islam and the propagandists of secularism'. In Line 2, Alhawali constructs the Egyptian religious educational institution 'the Azhar', as a target of the propagandists of westernization of Muslim countries and the propagandists of irreligion.

Lines 3 and 4 indicate a different way of representing social actors by nomination in Alhawali. The social actors in Line 3 are not only the 'propagandists of using the colloquial (Egyptian) instead of classical Arabic' who are negatively represented by Alhawali, but rather, as he claims, the 'enemies of Islamic culture'. He explicitly links them to the European colonial powers. In addition, Alhawali represents them as part of what he calls 'the Judeo-Christian plan to destroy Islam'. Further some social actors become points of identification in Alhawali. Ahmed Lutfi El-Sayyed, an Egyptian intellectual (1872-1963), Abdul Aziz Fahmi (an Egyptian politician 1870-1951), Ismaeel Mazhar (1891-1962) and Taha Hussein (an Egyptian writer and intellectual 1889-1973) are all identified by nomination and presented negatively as examples of a generation enslaved by the West and enemies of Islamic culture and Arabic.

The representation by nomination via association (Line 3) of such social actors is linguistically realized by the use of possessive pronouns (his colleague, his close friend) and by the possessive pronouns and familial relationship (his brother in law) which may suggest that Alhawali wants his audience to think that such social actors as connected to each other by familial relationships and friendships rather than being important writers and intellectuals (i.e. guilt by association). In Line 4, the negative representation of social actors by association is further developed. Ahmed Lutfi El-Sayyed is projected to be under the influence of western thinkers. Additionally Alhawali accuses him of being supportive of every 'offensive' propagandist of westernization and modernity. The differences in usage of 'i = 1' (do ' $\bar{a}h$; propagandists) between the two writers then can be thus summarized. Both writers use the term negatively, to refer to people who they see as ideologically against their own point of view. But Ibn Baz refers to such people more in terms of their role in attacking religion itself (such people are atheists or nationalists), whereas Alhawali tends to construct them more as culturally threatening and thus also "linguistically threatening" – involving westernization or even changing the language that people speak. Additionally, while Alhawali tends to nominate individuals such as Ahmed Lutfi El-Sayyed, it is worth noting here that Ibn Baz is less specific in naming social actors (this point is discussed in more detail in Chapter 8, section 8.5.1). Having stated the different representations of two Salafi writers of social actors indicated by the keyword 'i = 1' (do ' $\bar{a}h$; teachers/propagandists), I now move on to sketch how each writer uses the keyword 'i = 1.

6.2.1.2 The representation of 'أهل' ('ahl; the people of) in Ibn Baz and Alhawali

The keyword 'أهل' (*'ahl*; the people of) is also used in genitive constructions in Ibn Baz to represent many social actors by categorization. Let us first examine the collocates of 'أهل' (*'ahl*; the people of) in Ibn Baz (Table 6.4).

N	Keyword	Eng. trans.	F	Collocate	Eng. trans.	F.	Joint	MI
1				الجماعة	Majority	21	16	8.383
2				العلم	the religious knowledge	225	93	7.501
3				السنة	the Sunnah	142	33	6.670
4		the		الإيمان	the faith	135	20	6.021
5	أهل	people of	244	البدع	the impermissible innovative religious practices and beliefs	63	9	5.968
6				الكتاب	Book	110	9	5.164
7				الباطل	falsehood	71	5	4.948

Table 6.4 Collocates of 'iabl; the people of) in Ibn Baz⁷³

The first four collocates in Table 6.4 refer to the in-group social actors. The first and the third collocates are usually used in the 4-word cluster 'أهل السنة و الجماعة' ('ahlu ssunati

^{73.} Following the traditional way of writing in Islamic texts I capitalized the collocates 'Majority, Sunnah and Book'. These collocates refer to well known categorized social actors.

walğama'ah; the people of the Sunnah and the Majority) which refers to the Sunnis as opposed to the Shiites or any other Sufi order. This cluster is used 16 times across three of Ibn Baz's texts. The collocate 'العلم' (*'al'im*; the religious knowledge) is usually used to indicate the people of religious knowledge or scholars of religion. The social actors categorized as 'أهل الإيمان' (*'ahlu 'al'imān*; people of faith) (the fourth collocate in Table 6.4) belong to an inner circle of Muslims as 'إيمان' (*'imān*; faith) usually refers to a higher status in Islam which is a recontextualization of many Qur'anic verses including the following:

{قَالَتِ الأَعْرَابُ آمَنًا ۖ قُلْ لَمْ تُؤْمِنُوا وَلَكِنِ قُولُوا أَسْلَمْنَا وَلَمَّا يَدْخُلِ الإيمَانُ فِي قُلُوبِكُمْ ۖ وَإِن تُطِيعُوا اللَّهُ وَرَسُولَهُ لَا يَلِنَّكُم مَنْ أَعْمَالِكُمْ شَيْنًا إِنَّ اللَّهَ غَفُورٌ رَحِيمٌ } مسورة الحجرات اية ١٤.

{The desert Arabs say, "We believe (i.e. we have Faith)." Say, (unto them, O Mohammad) "You have no faith; but you (only) say, 'We have submitted our wills to Allah (i.e. we became Muslims), For not yet has Faith entered your hearts. But if you obey Allah and His Messenger, He will not belittle aught of your deeds: for Allah is Oft-Forgiving, Most Merciful} 1 (Holy Qur'an 49:14).

The remaining collocates indicate the 'Other' which is mostly represented negatively. As well as the collocates above, an examination of concordance lines reveals that additionally lbn Baz uses 'أهل' ('ahl; the people of) with nouns such as 'الخياف' ('alğahiliyyah; ignorance of the pre-Islamic days) (10 times) and 'الشرك' ('ašširk; polytheism) (twice). 'الجاهليه' ('alğahiliyyah; ignorance of the pre-Islamic days) is a collocate of أهل' ('ahl; the people of) in only two texts of Ibn Baz (MI score 6.603).⁷⁴ I included it in the analysis of social actor representation because it is associated with polytheism which is one of the major semantic foci in Ibn Baz. It is also used by Alhawali (Chapter 4, Section 4.2.1.2) which makes it more plausible to see Ibn Baz's actual use of this word as compared to that of Alhawali regarding social actor representation. Concordance 6.3 shows these collocates and the social actors designated as the other in Ibn Baz.

^{74.} Table 6.3 shows only the collocates of 'أهل' ('ahl; the people of) in three and more of the texts that comprise the Salafi corpus of Ibn Baz.

Concordance 6.3 Concordance of 'أهل' ('ahl; the people of) in Ibn Baz's corpus

	(<i>ani</i> ; the people of) in Ibn Baz's corpus
N	Concordance
	[…] و من المقاد المصدّد للعقيدة الصحيمة في بأب الأسماء و الصفات عقاد أ هل البدع : من الجهمية ، و المعتزلة ، و من سلك سيليم في نفي صفات الله عز و جل […]
1	[] and some of the erroneous beliefs that contradict the authentic Islamic creed, in the aspect of Allah's names and attributes, are
	the beliefs of the people of impermissible religious innovative practice and beliefs such as the Jahamites, the Mu'tazilites and those
L	who followed their way in nullifying the names and attributes of Allah, the Majestic and Exalted in Mighty []
	[] و قال سبحاته : إن لا يُتُوالله بِعلَّمِ إلا جِنَّناك بِلْحَقَّ و أَخْسَنُ تَشْبِيرًا } قال بحض السلف : هذه الآبة عامة لكل حجة يتَّى بها أهل الباصل إلى يوم التيامة.
	[] and the Glorified said: (And no example or similitude do they bring (to oppose or to find fault in you or in this Qur'an), but We
2	reveal to you the truth (against that similitude or example), and the better explanation thereof} and some of the (religious scholars
	of) the Salaf said: This Qur'anic verse is general and suitable to refute all arguments that are brought up by the people of falsehood
	ull the Day of Resurrection.
	[] التعلق بالأموات و الملائكة و الجن و غير هم من المُخلوقات ، و دعاءهم و الاستعادة بهم و تحو ذلك من عمل أ هل الجاهلية المشركين ، و من النجح الشرك بالله سبحانه ا
	فالراجب تركه و الحذر من ذلله []
3	[] depending on dead people, angels, the Jinn (the genies) and other creatures, calling on to them and seeking their protection
	etc All these are the practices of the people of Jahiliyyah (the pre-Islamic days), who were polytheists, which are the worst
	aspects of associating others with Allah, the Glorified. Thus All Muslims should beware of and avoid these practices []
	[] ر منتها قوله تعالى : {و لا تُجَدِلُوا ألهَلُ الْكِتَابِ إلا بِلْتِي هِي أَخْسَنُ} و أ هل الكتاب : هم الكفرة من اليهود و النصاري ، فلا يجوز جدالهم إلا بالتي هي أحسن []
1	[] and another example: Allah, the Exalted, said: {And argue not with the people of the Book, unless it be in (a way) that is better
1	(with good words and in good manner). The people of the Book are the infidels who are the Jews and Christians. Thus it is not
	permissible to argue with them (invite them to Islamic Monotheism) unless in a good manner []
	[] ر أصبح أهل الحق في ستمر الأمصار الذين عرفوا كتبهم [الشيخ محمد بن عبدالوهاب و تلاميده] .[] ينشرون دعوتهم ، و يستعينون بما ألفوا في هذا الشأن على
1	خصوم الإسلام و أعداء الإسلام في كل مكان، من أهل الشرك و البدع و الخرافات []
5	[] and the people of truth in every country who read their books [Sheikh Muhammad Ibn Abdul Wahhab and his disciples]
	advocated their call (to monotheism) and seek the help of such books to refute the arguments of the enemies of Islam who are the
	people of polytheism, impermissible innovative religious practice and beliefs and superstitions []

The concordance lines in Concordance 6.3 above designate 5 different social actors that are represented by means of personalization via categorization negatively in Ibn Baz. In personalizing social actors, Ibn Baz uses the collocates 'البدع'' (*albida*'; the impermissible innovative religious practices and beliefs) (Line 1), 'البطل'' (*albatil*; the falsehood) (Line 2), 'الجاهلية' (*'alğahiliyyah*; the ignorance of pre-Islamic days) (Line 3), 'الخرك'' (*ašširk*; the polytheism) (Line 5). These nouns carry negative associations such that when they are used with the keyword 'اهل' (*'ahl*; the people of) negative meaning are associated with the social actors that these words designate. In line 1, the Other is constructed negatively as having erroneous beliefs that contradict Islamic creed. Additionally, the Other is referred to as engaging in *innovative* practices, which is seen as *impermissible* (also referred to in line 5). In line 2, a Qur'anic verse is referred to as being suitable in refuting arguments of 'the people of falsehood'. Thus, Ibn Baz advocates that such people need to be defeated, although via rhetorical (rather than physical) means.

As for the noun 'الكتاب' ('alkitāb; the Book) which, in Ibn Baz, refers to any of the Holy Scriptures (in the Abrahamic religions) and when he uses it with 'أهل' ('ahl; the people of) it refers to the Jews and Christians who have varying (positive or negative) Qur'anic representations depending on certain historical and Qur'anic contexts. In Line 3, the people of the Book (Jews and Christians) are represented negatively as disbelievers of Islam. However, according to Ibn Baz who based his argument on the Qur'anic verse 29:46, Muslims should argue with them in a good manner (Line 4), which continues the theme in line 2 of Ibn Baz advocating that his followers defeat 'the Other' via argument (see Section 7.5.2). Now let us investigate the use of the word 'anic' ('ahl; the people of) in Alhawali.

The word 'أهل' (*'ahl*; the people of) has no collocates in Alhawali so again I have relied on concordance based analysis. Concordance 6.4 shows a sample of cases of this word being used by Albawali

being used by Alhawali.

Concordance 6.4 Concordance of 'أهل' ('ahl; the people of) in Alhawali

N	Concordance
	[] و قال رحمه الله في تقسير قوله تعالى: ((أفَحُكْمَ الْجَاهِلِيَّةِ يَبْغُونَ و من أَحْسَنُ من الله حُكْماً لِقَوْمٍ يُوقِبُونَ)) [المائدة:50]: ((ينكر الله تعالى على من
	خرج عن حكم الله [] كما كان أهل الجاهلية يحكمون به من الصلالات و الجهالات، مما يضعونها بأرائهم و أهوائهم []
	[] and in his explication of following Qur'anic verse: {Do they then seek the judgement of (the days of) Ignorance?
1	And who is better in judgement than Allah for a people who have firm Faith}(Holy Qur'an 5:50), Ibn Kathir's (1301-
	1373), may Allah have mercy on him, said: Allah, the Glorified, rejects that people adhere to other than the Sharia of
	Allah [] such as the practice of the people of Jahiliyyah (the pre-Islamic days) who referred to the laws of
	misguidance and ignorance which they invented based on their opinions and desires []
	[] و هذا هو الفارق الجوهري الأول في المستَّلة بين الإسلام و النصرانية المحرفة،[] و لذلك جاءت دعوة النبي صلى الله عليه و سلم أهل
	الكتاب مناسبة لمقتضي الحال التي كانوا عليها، من عبادة الأفراد و تقديس المخلوقين []
2	[] and this is the first essential difference in this issue [authoritarian priesthood] between Islam and the distorted
2	Christianity [] and for that reason [the control of people's lives by priests] the invitation that the Prophet (of Islam),
	may the blessings and peace of Allah be upon him, directed to the people of the Book to Islam was appropriate to
	their situation as they worshipped individuals (priests) and Sanctified the created ones (humans) []
	[] و في بداية الأمر علينا أن نستحضر في أذهاننا الصورة الساخرة التي وصف بها المؤرخون المسلمون دياتة الإفرنج الصليبيين [] و منها:
	نتبرج النساء و خروج غالبهن عن الحسمة و الحياء، [] فمالت إليهم (أي إلى الفرنسيين) نفوس أهل الأهواء من النساء الأسافل []
ļ	[] at the start we should conceptualize the sarcastic description made by the Muslim historians of the cuckolding
3	behaviour of the French Crusaders [] such as allowing their women to flaunt themselves where most of these
	women show themselves indecently and shamelessly [] some Muslim women [in Egypt] liked them (the French
	Crusaders [due to their behaviour with their women]). these women are from among the people of whim (those
	follow their desires) who belong to the lowest class []
	[] و كذلك عرفهم صلى الله عليه و سلم من أدلة التوحيد و النبوة و المعاد و الرد على جميع فرق أهل الكفر و الضلال []
	[] and [the Prophet] may the blessings and peace of Allah be upon him, taught them (Muslims) the indications and
4	proofs of monotheism, prophethood, the final return and refuting all arguments of all the factions of the people of
	infidelity and error []

Both Ibn Baz and Alhawali use 'الجاهلية' ('alğahiliyyah; the ignorance, i.e. the pre-Islamic days) and 'الكتاب' ('alkitāb; the Book) to refer to the same social actors but for different legitimation purposes. In Line 1, the historical أهل الجاهلية' ('ahlu alğahiliyyah; the people of ignorance – in the pre-Islamic days) are also represented negatively as inventing laws based on their opinions and desires. Alhawali therefore is disapproving of people following their own desires in preference of those set out by Allah.

In Islam the 'أهل الكتاب' ('ahlu alkitāb; the people of the Book) are always used to represent both Jews and Christians. However, Alhawali uses 'أهل الكتاب' ('ahlu alkitāb; the people of the Book) in Line 2 to mainly mean the Christians representing Christianity as a distorted religion. Another nomination of the Christians also appears in Line 3, viz., the crusaders. Here Alhawali negatively represents some Muslim women (in Egypt) as people of whim who belong to the lowest class of the society because, according the historians quoted by Alhawali, these women imitated and liked the crusaders. These women are categorically represented and associated with the French crusaders (men) who are also represented negatively as engaging in cuckolding behavior. Another representation by categorization is in Line 4 where Alhawali negatively represents all factions who he sees as against Islam, considering them to be the people of infidelity and error.

Therefore, both Salafi authors use the word 'أهل' ('ahl; the people of) to negatively represent different types of social actors who according to Salafis belong to 'the Other' by personalization via categorization. However, Alhawali and Ibn Baz differ in using the genitive construction 'أهل الكتاب' ('ahlu alkitāb; the people of the Book). While Ibn Baz refers to the Jews and Christians utilizing a Qur'anic verse to state that Muslims should argue with them in a good manner, Alhawali uses the same genitive construction to mainly mean the Christians representing Christianity as a distorted religion and equating them with the Crusaders. So far I have outlined the representation of social actors in Ibn Baz with a special focus on the representation of the 'Other'. Now let us turn to examine the representation of social actors in Alhawali.

6.3 Social actor representation: Alhawali's keywords

This section is devoted to analysing the representation of social actors in Alhawali. Table B3 (Appendix B) shows 37 keywords that indicate social actors in Alhawali. Based on preliminary collocational and concordance-based analyses, I have classified them according to their functions into seven categories as shown in Table 6.5. It is worth mentioning that the category 'the Other' is referred to by different keywords that also fall under other categories in Table 6.5. Some of these social actors have already been discussed in Chapter 5 so will not be examined here.

N	Categories	Keywords	
1	General social actors	the human, mankind, the researchers, the world	
2	Countries and nationalities	the American (feminine)/(Unites States) of America, America, the United, Egypt, Iraq, the regime, the (United) States (of America), the governments/the countries, the American (adj. masculine), Israel, Soviet (Union), France, the government/the country, the union/Union	The
3	Locations	Europe, the West, the (Christian/Western/Crusaders) World, European, the Western (societies/countries/Europe)	9 Other
4	Religion	(clergy)men, the Jewish, the Pope, the Churches] .
5	abstraction	communism/communist	
6	Names and personal pronouns	I. Hussein	
7	In-group social actors	Christ, Egypt, Iraq, Arab/Arab (world)/Arabic/Arabian (Gulf), Turkey, Musl (World/society), the Gulf	lim

Table 6.5 Categories of social actors in Alhawali

Unlike Ibn Baz's (mostly positive) theological categories of social actors, we first observe that nearly all of the keywords in Alhawali refer to social actors anonymously in political terms with the exception of the categories labeled 'Religion' and 'Names and personal pronouns'. Second, the majority of social actors in the categories above are represented by Alhawali as 'the Other' which mostly refer to the West and Western countries, whereas the in-group social actors are far less in number (seven keywords only). Taken collectively, the relative frequency of all keywords relating to 'the Other' is 5.2% in Ibn Baz and 72.8% in Alhawali, indicating that Alhawali is much more concerned with 'the Other' than Ibn Baz.

Some keywords are used by Alhawali to represent two or more types of social actors as in 'the world' which is used to represent opposite and conflicting 'worlds' as in 'Muslim/Arab/Christian/Crusaders' World' or just simply the whole world. Thirdly, most of the keywords carry metonymical references (see below). Despite the fact that 'Christ' is an important Christian figure, I have included him in category 7 that shows in-group social actors. The case is so, because 'Christ' is considered by Alhawali, as is the case in Islam, as a messenger of Allah where all Muslims should believe in him and his message as indicated in many Qur'anic verses such as the following:

{قُولُوا آمَنَّا بِاللَّهِ وَمَا أُنزِلَ إِلَيْنَا وَمَا أُنزِلَ إِلَىٰ إِبْرَاهِيمَ وَإِسْمَاعِيلَ وَإِسْحَاقَ وَيَعْقُوبَ وَالْأُسْبَاطِ وَمَا أُوتِيَ مُوسَىٰ وَ **عِيسَىٰ** وَمَا أُوتِي النَّبِيُّونَ مِن رَّقِمِ لَا نُفَرَّقُ بَبْنَ أَحَدٍ مِّنْهُمْ وَنَحْنُ لَهُ مُسْلِمُونَ } سورة البقرة، اية ١٣٦.

{Say (O Muslims), "We believe in Allah and that which has been sent down to us and that which has been sent down to Abraham, Ishmael, Isāc, Jacob, and to Al-Asbat [the offspring of the twelve sons of Jacob], and that which has been given to Moses and <u>Jesus</u> [Christ], and that which has been given to the Prophets from their Lord. We make no distinction between any of them, and to Him we have submitted (in Islam)}. Holy Qur'an 2:136.

A concordance examination of the keyword 'iu' ('ana; I) in category 7 found that this word was not used by Alhawali to directly project what he thinks or believes.⁷⁵ In all the cases of this word, Alhawali uses 'I' to quote (mostly western) writers writing or speaking about their views. Let us first have a brief look at Alhawali's representations of the social actors, after which I shall select examples that show the most prominently used sociosemantic categories.

In the first category 'General social actors', Alhawali anonymously represents three social actors by personalization via categorization ('the human' and 'mankind') and functionalization (as in 'researchers'). Based on a preliminary concordance analysis,

^{75.} Unlike English, the pronoun 'I', as well as other pronouns that indicate the subject (or object) in Arabic, could be incorporated into the verb inflectional suffixes (Ryding, 2005, p. 63). Such inflections appear as part of the verb that could not be recognized by WordSmith as separate words.

Alhawali also uses 'the researchers' argumentatively (see Section 7.5 below). Most of the social actors in the second (the largest category) and the third categories are metonymically represented by Alhawali. They are represented by impersonalization by means of objectivation via spatialization such as 'the (United States of) America', 'Israel', 'Soviet (Union)', 'Europe' and 'the West'. This particular way of representing social actors indicates a major difference between the two writers as there were no spatialization keywords in Ibn Baz. In the fourth category, Alhawali uses other nomination strategies that represent social actors in other religions, namely, Christianity and Judaism. The sociosemantic representations of the social actors in category 4 (Table 6.5) are categorization ((clergy) men), classification via religionym (the Jewish), functionalization of social actors by impersonalization is shown in category 5 where Alhawali represents communists by means of abstraction (see below).

The only representation of social actors by nomination in the Table 6.5 is 'Hussein'. There are actually three key figures that are represented by this nomination in Alhawali: two political leaders, the Iraqi president Saddam Hussein (1937-2003) and King Hussein of Jordan (1935-1999), who is nominated by honorification (see p. 207), and one main intellectual, Taha Hussein (an Egyptian influential writer 1889-1973). Alhawali's representation of the in-group social actors therefore varies between nomination (Christ) to personalization by means of categorization (Arab) and objectivation via spatialization (Arab world and Muslim world). Now let us consider how Alhawali represents 'the Other' as compared to Ibn Baz.

6.3.1 The representation of the 'Other' in Alhawali

As I have indicated above, the keywords that indicate 'the Other' in Alhawali outnumber those that indicate the in-group social actors (81% of the social actor keywords indicate 'the

Other'). This indicates Alhawali's interest in representing the 'Other' more than in-group social actors, when compared to Ibn Baz. Table 6.6 below shows these keywords.

N	Keyword	Eng. trans.	F	F. Baz	Keyness	N	Keyword	Eng. trans.	F	F. Baz	Keyness
1	اوروبا	Europe	263	4	233.322	14	الدول	the governments' the countries	115	10	65.642
2	الغرب	the West	181	6	142.034	15	الأمريكي	the American	69	1	61.514
3	الأمريكية	the American, (Unites States) of America	137	1	129.309	16	الأورويي	the European	55	0	55.921
4	أمريكا	America	137	4	110.289	17	الغربي	the western	67	2	53.697
5	العالم	the world/ (Christian/ Western/ Crusaders') World	305	52	109.48	18	إسرائيل	Israel	83	7	48.076
6	دول	governments' countries	96	0	97.615	19	البغ	the Pope	42	0	42.702
7	الغريية	the Western (societies/ countries/ Europe)	110	2	95.467	20	السوقيتي	Soviet	40	0	40.668
8	المتحدة	the United/United	93	0	94.564	21	فرنسا	France	40	0	40.668
9	الأورويبة	the European (feminine)	92	0	93.547	22	الدولة	the government, the country	112	20	38.392
10	المتنبوعية	communism ⁷ communist	124	6	88.502	23	الإتحاد	the union, Union	36	0	36.601
11	التظام	the regime	88	1	80.353	24	الكلأس	the Churches	29	0	29.484
12	الولايات	the (United) States (of America)	77	0	78.293	25	أنا	I	29	0	29.484
13	رجال	(ciergy)men	123	12	66.161	26	<u>مين</u>	Hussein	45	3	28.828

Table: 6.6 Keywords that indicate 'the Other' in Alhawali

Some of the keywords in Table 6.6 have already been seen in Table 5.1 when we considered the semantic foci keywords of Alhawali. In Chapter 5, we have already considered keywords, *the West* and *Israel*, which were classified as semantic foci although during the analysis it emerged that Alhawali sometimes constructed them as social actors. As the analysis of these words (and similar words like *America*) has already been undertaken, it is not the intention of this chapter to replicate the contents of Chapter 5.

Therefore, based on the criteria in Section 6.2 above, the keywords 'رجال' (*riğāl*; (clergy) men) and 'الشيوعية' (*'aššyo 'iyyah*; communism/communist) were selected for the analysis. Let us take each in turn.

6.3.1.1 The representation of 'رجال' (riğāl; (clergy) men) in Alhawali and Ibn Baz

The keyword 'رجال' (*riğāl*; men) occurs in 4 of Alhawali's texts (123 times), but he extensively uses it in one text that is devoted to discussion of 'Secularism' (115 times). 'رجال' (*riğāl*; men) is an indefinite noun that is usually used in annexation constructs with other nouns where the second noun identifies the first (in this case 'رجال' (*riğāl*; men). The only collocate of the keyword 'رجال' (*riğāl*; men) in Alhawali is ''الدين' (*addeen*; the religion) (MI score 7.694). Both the keyword 'رجال' (*riğāl*; men) and its co-occurring collocate (joint frequency 72) make the Arabic nominal phrase 'رجال الدين' (*riğālo 'addeen*; the men of religion) which is usually translated into English as 'clergymen'.

Concordance 6.5 Concordance of رجال الدين' (riğālo 'addeen; clergymen) in Alhawali

_	Concordance 6.5 Concordance of رجال الدين (rigālo 'addeen; clergymen) in Alhawali
Ν	Concordance
	[] سبق التول بأن الكنيسة تملك الإتصاعيات برقيتها، و أن بعض رجال الدين كان يملك الألاف من الأرقاء، غير أن ذلك لم ينتع الكنيسة، بل أر غمت أتباعها على العمل المجاني في
	حَقُولُها و في مشرو عاتلها. لاسيما بناء الكنائس و الأضرحة []
1	[] I have said that the Church owns fiefdoms and the slaves who live on them and that some clergymen used to own thousands of
	slaves. But that was not sufficient in the eyes of the Church, it obliged its followers to work as labor in its fields and projects without
	payment, especially in building churches and graves (of holy men) []
	و قد أدرك أتين دينيه هذه الحقيقة و كمانت إحدى دوافع اعتدقه للإسلام يقول: (الوسيلة هي إحدى كبريات المسائل ألتي فاق بها الإسلام جميع الأديان. إذ نيس بين الله و عبده وسيف و نيس
	في الإسلام تساويبة و لا رهبان. []). إن الإسلام ليس فيه شيء اسمه <u>رجال دين</u> أصلاً، بل أن هذه الكلمة المحدثة لا يستعملها إلا مغرض مضلل أو سالاج معدوع.[]
	Étienne Dinet (French Orientalist painter 1861-1929) realized this fact [Muslims do not take human individuals as a means of access to
,	Allah] which is one of the issues that motivated him to embrace Islam. He says: (The Islamic concept of 'Wasilah' [i.e. seeking the way to
2	approach to Allah] is one of the greatest Islamic concepts that distinguish Islam from other religions. There are no intercessors between
	Allah and His worshiper. Additionally, there are neither priests nor monks in Islam, []). Indeed, clergymen do not exist in Islam. This
[newly imported denomination [clergymen] is only used by people who have agendas against Islam or those who blindly follow them []
	و كان من الأسس الباطلة التي بني عليها رجال النين مبررات وجودِهم مبدأ "التوسط بين الله و الخلق"، الذي يتتضي الأيذهب الإنسان إلى رجل الدين ليعلمه كيف يعبد الله، بل ليعبد الله
	بواسطته، []، و حسب هذا العبدأ نصب <u>وجال الدين</u> أنتسهم أنداداً لله تعالى و أوقعوا انتباعهم في الشرك الأكبر (التَّخَذُوا أحْبَارَ هُمَّ و رُهْبَانَهُمَّ أَرْبَدياً من نُونٍ اللهم [التوبة:31]
	And among the invalid principles that clergymen used to legitimate their positions in the Church is the principle of intercession between
3	Allah and His creatures. This principle requires that people do not go to the clergyman to teach him how to worship Allah, but to worship
	Allah through him, [], based on this principle, the clergymen set up themselves rivals to Allah, the Exalted, and made their followers
	commit greater polytheism {They (Jews and Christians) took their rabbis and their monks to be their lords besides Allah (by obeying them
	in things which they made lawful or unlawful according to their own desires without being ordered by Allah)} Holy Qur'an 9:31
	[] فالكاتوليك يقوم بينهم على هذا التوسط فرجل النين هو الواسطة، فلا يمكن الاستغفار إلا عن طريقه، و لا يمكن الصنفة إلا عن طريقه! حتى إنه يباشر كثيراً من الطقوس و أمور
	التناس عند الموت، و عند الفولاد، و عند الزواج، و أمور كثيرة لابد أن يقوم بها القسيس، و إن لم تتعل لأي إنسان فإنه يعد محرومةً من الجنة. و من ذلك احتكار هو لقراءة وتتسبر الانجيل.
	فلا يمكن لأحداش يقرأ الإتجل أو يفسره إلا رجل الدين، ويقرعونه باللغة القديمة، وليس باللغات المعاصرة والتهجات المطبة التي كانت في أوريه بن
1.	[] The Catholics' religion is based on intercession and the clergyman is the intercessor. Thus, seeking forgiveness and charity giving
4	should be through him! (he controls every aspect) to the extent that he initiates many rituals that concern people's life, death, marriage
	and many other aspects that if not initiated by priests for individuals they are considered deprived from God's paradise. Additionally, they
	control reading and explicating the Gospel. No one is able to read or explicate it, but the clergymen. They read it in an old language and
1	
	and many other aspects that if not initiated by priests for individuals they are considered deprived from God's paradise. Additionally, they control reading and explicating the Gospel. No one is able to read or explicate it, but the <u>clergymen</u> . They read it in an old language and not in the modern European languages and dialects.

The men who are represented in Concordance 6.5 are the 'clergymen' in Christianity that Alhawali categorically and ideologically represents negatively. Line 1 represents clergymen as well as the Church as having an ardent desire to enslave people. Utilizing the words of Alphonse-Etienne (aka Nasreddine) Dinet (a French Orientalist painter 1861-1929) about Islam, Alhawali draws a comparison (Line 2) between Islam and Christianity, denouncing clergymen whom he also describes as monopolizing access to the Bible (Line 4). Alhawali's rejection of such appellation to be ascribed to Muslim Sheikhs (Line 2) reveals his negative stance towards Christianity and clergymen. In Line 3 Alhawali quotes a Qur'anic verse to further denounce clergymen who, according to him, set up themselves as rivals to Allah, which he labels as resulting in polytheism. Additionally, Alhawali uses the keyword '(rigāl; men) in annexation constructs with other nouns that denote Christian social actors, namely, '(alkaneesah; the Church) and '(allāhūt; theology) as indicated by the following extracts:

Figure 6.1 Two extracts from Alhawali containing the keyword 'رجال' (riğāl; men)

و من جهة السلوك تلطخت سيرة رجال الكنيسة و أعضاء الأديرة برذائل و أرجاس يترفع عنها الفرد العادي و يتستر عليها الفاجر البذيء [...] Regarding the behaviour of the <u>men</u> of Church and members of Monasteries (monks), their fame was smeared by moral faults and immoral conduct that ordinary individuals refrain from and secretly committed by lewd individuals

و لقد جزعت الكنيسة من هذه الانتقادات و الآراء جزعا شديداً، و لعنت فولتير و أشياعه و كفرتهم، و حرمت قراءة كتبهم و تعرض فولتير للمضايقة و الاضطهاد من قبل <u>رجان</u> اللاهوت،

Verily the Church panicked due to such criticism and the opinions (of Enlightenment writers). It cursed and excommunicated Voltaire (French philosopher 1694–1778) and his advocates. It also banned their books. Indeed, Voltaire was harassed and prosecuted by **men** of theology (theologians).

In the first extract, Alhawali accuses men of the Church as well as members of Monasteries (clergymen) of engaging in immoral conduct. He further negatively intensifies his claim by adding that ordinary (not religious) individuals refrain from doing such acts. Alhawali is vague by not naming the acts that such individuals carry out but instead refers to them as 'immoral conduct'. In Section 6.2.1.2 above, Alhawali is also vague in his projection of the cuckolding behavior of the crusaders. It is likely that he does not describe the 'sexual acts committed by the men of the Church and the crusaders' because revealing such acts is socially and religiously unacceptable in Islam (see Section 8.6 in Chapter 8 for more

information on morality in the West). But the mere mention of such behavior makes it reasonable to his audience to reject not only the teachings of the Church but the West as a whole as being involved in such undesirable (to him) sexual behavior. Additionally, he (in the second extract) uses the keyword ' $(r\check{g}j\bar{a}l;$ men) to denote the people of the Church (theologians) as oppressing Enlightenment intellectuals such as Voltaire (French philosopher 1694-1778), although as shown in other sections (e.g. 5.3.1), Alhawali has also been critical of western philosophical movements which reveals a contradiction in Alhawali's stance towards the Enlightenment intellectuals.

While the keyword 'رجال' (*riğāl*; men) is mostly used by Alhawali to denigrate social actors in Christianity, Ibn Baz uses the word 'رجال' (*riğāl*; men) 12 times to denote different social actors (as shown in the sample of concordance lines in Concordance 6.6).

Concordance 6.6 Concordance of 'رجل' (riğāl; men) in Ibn Baz

N	Concordance
	[] فأفضله [العلم التسرعي] وأعظمه وأتسرفه ما يتعلق بالله وأسمائه وصفاته، و هو علم العقيدة، []. و يلتحق يذلك علم السيرة النبوية، و التاريخ الإسلامي، و تراجم رجال الحديث و أئمة الإسلام، و يلتحق بذلك كل ما له صلة بهذا العلم.
1	[] and the best and the most honourable [amongst the fields of religious knowledge] is what pertains to Allah's names and attributes
	which is the field of Creed [] in addition, the Prophetic hagiography, Islamic history, and the hagiographies of the men of Hadith
	(Prophetic tradition reporters) and Muslim Imams and all fields that pertain to it.
	و أوجه نصبحتي أيضا إلى أقوام من المسلمين بعيشون بينهم [من لا يحكم شرع اللم]، و قد علموا الدين ، و سرّع رب الحالمين ، و مع ذلك لا زالوا يتحاكمون عند النزاع إلى رجال بحكمون بينيم بعادات و أعراف ،[] مشابهين في ذلك صنيع أهل الجاهلية الأولى.
2	and, also, I advice some Muslims who live with those [who, in matters of dispute, do not refer to the Sharia Laws]. Despite they have
	studied Islam and the Sharia of Allah, they still, in matters of dispute, refer to men who refer to costumes and traditions, [] which reflect
	the behaviour of the people of pre-Islamic era.
	[] و يقرؤون تلك الأوراد في مجالين الذكر ، أو في المساجد بح صلاة المعرب ، زاعمين أنها قربة إلى الله ، كقولهم : بحق الله ، أجلل الله ، أعينونا بعون الله ، و كونوا
	عوننا بالله []. أن هذه الأدعية و أنواع الاستغالة التي بينتها في بينتها في سؤالك، كلها من أنواع الشرك الأكبر لأنها عبادة لغير الله، و طلب لأمور لا يقدر عليها سواد
2	[] they read such prayers in a place with others or in Mosques after the Maghrib (the sun set prayer) claiming that such prayers bring
	them close to Allah. In these prayers they say. Oh men of Allah, by the right of Allah, provide us with the help of Allah and be our
	supporters []. Verily, such prayers and the types of beseeching help that you mentioned in your question are all considered different
	kinds of major polytheism, because they are types of worshipping other than Allah and requesting matters that no one can do but Allah
	و لقد أحسن الكاتب الإسلامي السّهير : أبو الحسن اللذوي في رسالته المسّهورة : (اسمعوها مني صريحة : أبها العرب) حيت يقول [] : (فمن المؤسف المحزن المخجل
	أن يقوم في هذا الوقت في العالم العربي ، رجال يدعون إلى القومية العربية المجردة من العقيدة و الرسالة ، []، إنها جريمة قومية تبَّر جميعُ الجرائم القومية ، التي سجلها
Ι.	تاريخ هذه الأمة ،[].
4	Verily, the great Islamic (Indian) writer AbulHasan Nadvi (1913-1999) did well in his famous letter (A Frank Call to the Arabs). He said
	[] : it is shamelessly sad that in this stressful time in the Arab World, some men call for Arab nationalism disregarding the Islamic
	message and Creed, []. It is the most devastating among the nationalistic crimes in the history of this (Muslim) nation, []

Ibn Baz uses the noun 'رجال' (riğāl; men) in annexation constructions to represent both in-

and out-group social actors. Line I (Concordance 6.6) shows Ibn Baz's positive representation

of the medieval reporters of the Prophetic Hadiths whose hagiographies, according to Ibn Baz
as well as the Salafi religious leaders, are an integral part of the religious knowledge. In Line 2, Ibn Baz, from a Salafi standpoint, rejects the reference to Laws other than the Sharia. He uses the indefinite plural noun 'رجال', (*riğāl*; men) as opposed to Allah that connote negative meaning (i.e. the divine vs. the human). In Line 3, the association between men and Allah is not as positive as it appears to be. On the contrary, Ibn Baz rejects beseeching help from individuals nominated by Sufis as ('رجال الله')' (*riğāl 'allah*; the men of Allah). He considers such Sufi practices to be major polytheism (detailed in Chapter 4, Section 4.2.2). In Line 4 in Concordance 6.6, Ibn Baz negatively represents those who call for Arab nationalism quoting Nadvi's (an Indian Islamic writer 1913-1999) call to the Arabs. Nadvi sarcastically blames some (Arab) men who call for Arab nationalism rather than Islam.

Therefore, both authors utilize the word 'رجال) (riğāl; men) to represent different sorts of social actors. On the one hand, Alhawali is mostly concerned with the discursive construction of social actors in other religions, mainly clergymen in Christianity, negatively representing them as engaging in immoral sexual conduct. In his discussion of the Church and its theologians, Alhawali claims that they oppressed Enlightenment intellectuals. Such oppression, according to Alhawali, forced people in the West to abandon religion (Christianity) and adopt secularism (for more discussion of Alhawali's representation of the Church and the West see Sections 4.2.1.3 and 4.2.5 in Chapter 4 and Chapter 5). Thus Alhawali represents the Western world as full of corruption and immorality and consequently, whatever comes from the West is viewed as evil. On the other hand, Ibn Baz uses the word 'رجال' (riğāl; men) to represent both in-group and out-group social actors. The in-group social actors are medieval Salafists (Hadith reporters and the Imams of Muslims) whose hagiographies are recommended by Ibn Baz to be studied as part of what he claims to be the noble (religious) knowledge. The other type of social actors denoted by the word (riğāl; men) in Ibn Baz are negatively represented as rivals to Allah in the Sufi rituals and those who call for Arab nationalism.

Let us now consider Alhawali's use of the other keyword that is eligible for analysis in this section, i.e. 'الشيوعية' ('aššyo 'iyyah; communism/communist).

6.3.1.2 The representation of 'الشيوعية' ('aššyo'iyyah; communism/communist) in Alhawali and Ibn Baz

The keyword 'الشيوعية' ('aššyo 'iyyah) is used to mean either 'communism' (noun: 67 times) or 'communist' (adjective: 48 times) in Alhawali's texts. He uses both word forms to represent the 'Other' by means of impersonalization via abstraction (communism) or by means of personalization via categorization (communist) as in Concordance 6.7.

Concordance 6.7 Concordance of 'الشيوعية' ('aššyo 'iyyah; communism/communist) in Alhawali

С	01	10	or	da	an	c	e
		_		_	_	_	-

N يتول جارودي (فيلسوف وكاتب فرنسي 1913-2012) [...] "هذه الحقيقة التاريخية" (أن الدين انعكاس لتنقاه فعلى) هي التي يلخصها ماركس في تعيير مقتضب "الدين أفيون التسعوب" [...] ترى ا**لشيوعية** أن الفكر البشري انعكاس للواقع المادي. فالمادة هي الأسابن الوحيد و عنها ينبق الفكر و تنبتق المتداعر والأحاسيس، و من هذه المتداعر الدين نضه. Garaudy (French philosopher and writer, 191-2012) says: 'this historical fact (that religion is a mere reflection of a real misery) is epitomized by 1 the following paraphrased statement of Marx "Religion is the opium of the people" [...]. Thus, communism considers human intellectual activity as a reflection of the materialistic world. The material is the sole base of life from which intellect, feelings and emotions emanate. And religion itself emanates from these feelings. [...] و النظرية التبيوعية دين رسمي متحصب لا يقبل الجدل و لا يسمح بالمنافسة. و كل تسئ في الدولة الثبيوعية لابد أن ينبق من المؤدة المار كسبة و يتمتني معها، فالحلو يجب تسخير ه لتتبيت النكر المادي [...] و الأدب و الغن لابد أن يلتزما بما سمي الواقعية الاستر اكية.[...]. و هكذا في كل سي تريد الشيوعية أن تقرض نفسها عقيدة [...] فهي فكرة جاهلية متكاملة تقابل تماماً الدين بمفهومه الحقيقي الشامل 2 [...] and communism is an intolerant official religion that does not accept discussion or competition. Every activity in the communist country should reflect the Marxist creed and should not contradict it as well. Science should be deployed in the furtherance of the materialistic intellect [...]. Additionally, literature and arts should be confined within the boundaries of the so called socialist realism [...] so; communism wants to impose itself in every aspect of life as a creed [...]. it is a complete jahili (ignorant) idea that confronts the real conclusive meaning of religion [...] أما على الصعيد التطبيقي، فإن الشيوعية تحصر الشرور كلها - منذ بدء الخليقة إلى الأن - في علة واحدة هي الملكية الفردية، و لذلك فإتهم يؤمنون إيماناً أعمى بأن القضاء على الملكية النرانية [...] كفيل بتحقيق الجنة الأرضية، وإزالة كل الترور والمساوئ التي يعج بها التاريخ ولا تستتني الشيوعية من ذلك تبياأً حتى المرأة فإن التيوعية الجنسية والإياحية المطلقة هدف صريح من أهداف التورات السَّيوعية في كل مكان [...] 3 [...] practically, communism confirms that all evils - in all ages- exist as a result of 'private property'. Such convection made them blindly believe that abolition of private property [...] will result in a utopia and eliminate all evil and sufferings of humanity. Communism excludes nothing, even women, to the extent that sexual communism and absolute licentiousness are explicit goals of all communist revolutions [...] و أما حلف وارس فيضم الاتحاد السوفيتي وأوروبا الشرقية الشيوعية على اختلاف أجاسها و أعراقها التي أصبحت كتلة واحدة بعد الحرب العالمية التاتية التي انتبت عام (1945م).[...] فلما ظهر المعسكر الشرقي الأشتراكي اندمجت الفكرة الاشتراكية في الحركات التومية و الوطنية (في العالم العربي) - لأنها كلها مستوردة من الغرب- فقامت التورة المصرية و حُرُّلها جمال عبدالناصر من فكرة و طنية إلى فكرة قومية Regarding the Warsaw pact, it includes the Soviet Union and the communist countries in Eastern Europe despite the fact that these countries belong 4 to different races and ethnicities which became one block after the Second World War that ended in 1945 [...]. After the emergence of the Socialist Eastern bloc, communism (in the Arab World) was incorporated into nationalistic and patriotic movements - because these ideas were all imported from the West- and when the Egyptian revolution took place, Abdul Nasser (the president of Egypt between 1956 and 1970) changed it from Egyptian patriotic into Pan Arab nationalistic revolution

In his representation of 'الشيوعية' ('aššyo 'iyyah; communism) by means of impersonalization via abstraction, Alhawali (Line 1) quoting Roger Garaudy (French philosopher and writer 1913-2012), tries to show how 'communism', in opposing the notion of religion, actually ends up being a religion in itself to its followers, thus it is hypocritical. Additionally, he (Line 2) tells his readers that 'Communism' is a religion and the 'communist' countries control every aspect of their citizens' life. Communism, according to him, is intolerant and a *ğahili* concept i.e. ignorant and belongs to pre-Islamic thinking (the concept of ignorance ('*alğahiliyyah*) is detailed in Chapter 4, Sections 4.2.1.3 and 4.2.3). Alhawali continues to represent communism negatively in lines 3 and 4. He claims that communism ignores moral values which, according to him bear the immoral values and disgrace of the *ğahilyyiah* (i.e. values of the pre-Islamic era) and tries to establish absolute immoral societies (Line 3) as opposed to societies that follow religious moral values using a threat indicating word, 'revolution', to legitimate his stance towards communism (see below). Expanding concordance line 3 (Figure 6.2) reveals more about Alhawali's views of communism and what he believes to be a call to sexual communism and the absolute moral decadence of societies.

و أما على الصحيد التطبيقي، فإن التليو عية تحصل التلوول كلُّها -منذ بدء الخليقة إلى الآن - في علة واحدة هي الملكية الفردية. و لذلك فإنهم يؤمنون إيماناً أعمى بأن القضاء على الملكية الفردية. و قبض الدولة على وسائل الإنتاج كغيل بتحقيق الجنة الأرضية، و إز الة كل الشرور و المساوئ التي يعج بها التاريخ، و لا تسنَّتني الشيوعية من ذلك شيئاً حتى المرأة فإن الشيوعية الجنسية و الإباحية المطلَّفة هدف صريح من أهداف التورات التنبوعية في كل مكان، فالزواج ينتج الأسرة، و الأسرة في نظر هم أعدى أعداء المجتمع اللاطبقي، لأنها تحتّم على المرء أن يتملك و يدخّر ، و الملكية الغردية تقَليد إقطاعي استغلالي، أبن لم يتم القضاء عليه انتكس المجتمع إلى طور تاريخي أدني [...] [...] practically, communism confirms that all evils -in all ages- exist as a result of 'private property'. Such conviction made them blindly believe that abolition of private property [...] will result in a utopia and eliminate all evils and sufferings of humanity. <u>Communism</u> includes everything (in the causes of human sufferings), even women, to the extent that sexual communism and absolute licentiousness are explicit goals of all communist revolutions. The case is so, because they believe that marriage establishes family which they consider to be the worst enemy of classless societies. They also believe that marriage demands people to own properties and save money and individual property is an exploitive feudal tradition that if not abolished would cause societies to relapse to a primitive historical epoch [...]

Here Alhawali demonizes communism and represents it as rejecting marriage (a baseless claim) and destroying the basic unit of societies, the family. As in Section 6.3.1.1 above,

Figure 6.2 Expanded version of concordance line 3 in Concordance 6.7

Alhawali stresses moral values in order to denigrate communism and the West (further discussed below and in Chapter 8, section 8.5).

In the last concordance line, Alhawali continues to polarize the world by trying to establish a link between the Arab communist and socialist countries (such as Iraq, Syria, South Yemen)⁷⁶ and the Eastern bloc that includes the Soviet Union and the communist countries in Eastern Europe. Additionally, he claims that communism incorporates nationalistic and patriotic movements because they are all imported from the West. Alhawali's last claim of demonizing communism and communists escalates polarization between the West and the Arab world as well as his negative representation of the West as a source of communism, nationalistic and patriotic movements (for Alhawali's representation of the West, see Section 5.3.1). All the lines in Concordance 6.7 thus indicate that Alhawali, from a political point of view, tries to demonize communism and communist countries as well as the West, while constructing the two as having similarities (both are constructed as sexually amoral) (for more information about sexual morality in communism see Field (2007)). Now, let us move to Ibn Baz's representation of communism.

Ibn Baz also uses the word 'الشيوعية' (*'aššyo 'iyyah*; communism) as a concept that contradicts Islam. Concordance 6.8 shows such 4 cases.

^{76.} South Yemen is also called The People's Democratic Republic of Yemen. It is a socialist state which was established after the British troops left the country in 1967. It was united with north Yemen to form the State of modern Yemen in 1990.

Concordance 6.8 Concordance of 'الشيوعية' ('aššyo 'iyyah; communism/communist) in Ibn Baz

N

Concordance [...] و بما ذكرناه من الأدلة القرآنية ، و إجماع أهل العلم يعلم السائل و غيره ، أن الذين يدعون إلى الاشتراكية أو الشيوعية أو غيرهما من المذاهب الهدامة المناقضة لحكم الإسلام ، كفار ضلال [...]

1 [...] Based on the previous Qur'anic verses, and the consensus of the people of knowledge (religious leaders), the questioner and others will realize that those, who call for <u>communism</u> or any other subversive doctrines that oppose Islamic Sharia, are apostates and astray [...]
2 [...] big 2 [...] big 2 [...] communism or any other subversive doctrines that oppose Islamic Sharia, are apostates and astray [...]
2 What is the Islamic opinion regarding those who call for arbitration according to the principles of socialism or <u>communism</u>? [...] the answer to this question: [...] All religious leaders confirm that anyone claims that any law is better than the Sharia of Allah [...] becomes an apostate [...]
2 What is the Islamic opinion regarding those who call for arbitration according to the principles of socialism or <u>communism</u>? [...] the answer to this question: [...] All religious leaders confirm that anyone claims that any law is better than the Sharia of Allah [...] becomes an apostate [...]
3 The only means to get rid of and eliminate [the calamities that afflict the Muslim nation] is to examine them critically according to the perfect rules of Qur'an and the Sunnah [...] Thus if the propagandists of <u>communism</u> and socialism, the atheists, proceed (to present their argument), you (Muslims) read the Qur'anic verses that guide you to the indicative signs of His existence, the Glorified [...]
4 What is the Qur'anic verses that guide you to the indicative signs of His existence, the Glorified [...]

As the Majestic and the Most High (Allah) informed us in His glorious Book that He is the All-Wise, the All-Knower and has power over all things, the Majestic and the Most High. This is the most profound reply to the propagandists of <u>communism</u>, the *dahriyyah* (a pre-Islamic atheistic doctrine) and socialism [...]

{وَقَالُوا مَا هِيَ إِلَّا حَيَاتُنَا الدُّنْيَا غَوْتُ وَغَيْا وَمَا يُهْلِكُنَا إِلَّا الدَّهْرُ وَمَا لَهُم بِنَالِكَ مِنْ عِلْمٍ إِنَّا مَمْ إِلَّا يَطَنُّونَ } And they say: "There is nothing but our life of this world, we die and we live and nothing destroys us except Ad-Dahr (time). And they have no knowledge of it: they only conjecture Holy Qur'an 45:25. Both Salafi writers thus represent communism and communists negatively, but each one uses different representation strategies. Alhawali uses both impersonalization via abstraction and personalization by means of categorization whereas Ibn Baz only uses personalization via categorization. Another difference is that Ibn Baz seems to be preoccupied with those social actors who call for communism in the Muslim World as he deems them to be apostates, whereas Alhawali's representation is focused on communism and communists from outside Muslim-majority countries as a threat to societies. In addition, Alhawali connects communism, nationalistic and patriotic movements with the West which is part of a tendency of Alhawali to relate all what he believes to be negative with the West (for more details on the representation of the West in Alhawali see Chapter 5, Section 5.3.1). Ibn Baz seems to be offering advice to his readers on how to argue with propagandists of communism (lines 3 and 4) which could be linked to his earlier advice about only arguing "in a good manner" with the people of the Book.

The categorization strategies that both authors used contribute to ideologically constructing two opposing groups of social actors and most notably the out-group that was negatively constructed. This may lead to polarizing the Muslim/Arab societies in the case of Ibn Baz (constructing, via grammatical annexation structures using *the people of* and *men of*, the Shiites and Sufis' practices negatively) and, to a large extent, Alhawali's categorization of the West leads to polarizing the world into mostly the enemies of Islam and the oppressed Muslim/Arab counties which eventually could lead to the rejection of the modernization of Saudi Arabia. Additionally, such polarization (between 'the West' and 'the Muslims') could, as Baker et. al. (2013, pp. 130-5) note, lead to reify these terms and present them 'the West' and 'the Muslims' as in conflict with each other.

The above negative representations of 'the Other' reveal the Salafists' fear and rejection of any western intellectual/industrial or otherwise development. Further, these

206

representations show their inclination to lead the society to live within the boundaries of their ideological interpretations of the Muslims' sacred texts.

6.4 Conclusion

In this chapter I have tried to apply the sociosemantic categories of van Leeuwen (1996; 2008, pp. 23-54) that he utilized in analysing the representation of social actors in English voicing the caveat that social actors representation may differ in some aspects of his categories due to cultural and societal conventions in nominating social actors. However, as I have pointed out in Chapter 3 (section 3.2.2.6) that there may be some differences in the category 'nomination by affiliation', the rest of the categories found in the Salafi corpora fit neatly into van Leeuwen's sociosemantic categorization scheme.

In general, both Salafi writers employ different sociosemantic categories in their representation of social actors. Such representation strategies involve both personalization and impersonalization. They utilize the sociosemantic categories of categorization and classification via religionym (Sunnis vs. Shiites and Sufis which serve as ways of polarizing Saudi society) as well as functionalization which indicates the Salafi writers' perception of the social actors referred to. Additionally, both Salafi writers represent some social actors by nomination using honorification titles. But Ibn Baz utilizes nomination by honorification to represent medieval social actors positively (mainly Salafi religious leaders showing his reverence and adherence to their teachings) whereas Alhawali uses the same sociosemantic category to represent modern political social actors (King Hussein of Jordan, and the Iraqi president 'Saddam Hussein') often in a way that represents them as responsible for political actions that resulted in instability of the Arab and Gulf states.

In their representation of 'the Other', both Salafi writers use categorization that serve their legitimation purposes in representing what they believe to be good or bad (Muslim vs. non-Muslim categories for example). Ibn Baz, on the one hand, represents 'the Other' by categorization selecting theological terms (polytheists, friends of God, and propagandists of doctrines that he conceives of as opposing Islam). On the other hand, Alhawali, additionally, represents 'the Other' by impersonalization by means of objectivation via specialization (metonymical references) and abstraction. The metonymical references in Alhawali further develop the divide between two opposing worlds (the West vs. the Muslim and/or the Arab Worlds). As I have noted earlier (section 5.3.1), this form of categorization of social actors as Baker et al (2013, pp. 130-5) note on the construction of 'the West' and 'the Muslims' by British newspapers, could reify these terms and present them as in conflict with each other. These representations of social actors are all deployed in argumentations that both salafi writers use for different legitimation purposes that I shall explore in the following chapter.

Chapter 7: Argumentation in the Salafi corpora: Ideological legitimation

7.1 Introduction

This chapter is devoted to analysing arguments and argumentation schemes that both Salafi writers deploy to legitimate their stances vis-à-vis their representations of social actors as well as the thematic foci they address in their texts. It responds to the following subquestions:

> 1(c) What do concordance and collocational analyses of keywords and frequent words in the two corpora of Salafi discourse reveal about argumentation indicators? And;

> 2. What argumentation schemes are employed in the Salafi corpora to legitimize certain positions?

Answering these questions contributes to answering the overarching question: To what extent and how is Salafi discourse in Saudi Arabia in the 1980s -1990s homogenous? In identifying arguments I use Van Eemeren and Grootendorst's (2004) typology of schemes (see below). I refer to *ideological legitimation* in the title of this chapter, as I am interested in the ways in which the writers from different Salafist ideological perspectives try to persuade and/or convince their audience of certain standpoints concerning religious/politico-religious issues via appeal to their religiously-moulded (Salafi) collective religious, political and moral conceptions of the world.

To reiterate from Chapter 3, Arguments can be classified according to various argumentation schemes. Argumentation schemes are forms of argument that capture stereotypical patterns of human reasoning, especially defeasible ones, like argument from expert opinion, that have proved troublesome to view deductively or inductively (Walton, 2005). Van Eemeren and Grootendorst (2004) distinguish between three major categories of argumentation schemes: causal argumentation, symptomatic (sign) argumentation and argumentation based on comparisons (argument from analogy). Though their classification scheme is broad, it can help identify the main argumentation schemes in the research data. For example, examples of argumentation scheme by sign could be found by writers utilizing Qur'anic verses as the words of Allah, which according to Muslims, are inherently correct and true.

Therefore, in order to trace arguments in the research data, I shall identify and examine keywords that are likely to indicate arguments as the starting point in the analysis. Additionally, I use collocates and word clusters to further analyse argumentative keywords. The main reason behind using keywords is to search for the most salient arguments which

^{77.} In the field that is called 'the sciences of Hadith' or the study of Prophetic traditions, the Hadith is either accepted or rejected based on the evaluation of the 'سند' (sanad; chain of reporters), i.e. the individuals who report the Hadith and the 'سند' (matn; the text of the Hadith) or the wording of the Hadith utilizing a number of criteria and principles that concern the credibility of the reporters of the Hadiths and the texts authenticity. Based on such principles the Hadiths are then classified into seven categories each of which is further classified into different types. For more information see the following link: http://www.islamic-awareness.org/Hadith/Ulum/.

distinguish the two Salafi authors when they are compared together. I also use word clusters in order to narrow down the scope of analysis to the most commonly used argumentation schemes in the research data. After locating arguments, the identification and evaluation of argumentation schemes will then take place, based on critical evaluation of argumentation schemes (via critical questions, see below) which may result in identifying topoi⁷⁸ or fallacies⁷⁹. Let us now move to examine the keywords that indicate arguments in both corpora.

7.2 Identifying key argumentation indicators in the Salafi corpora

Of the 224 keywords in Ibn Baz and 199 keywords in Alhawali, there seem to be fewer keywords that, at face value, indicate argumentation than those which indicate semantic foci or social actors. Based on my own observation of some keywords and a preliminary investigation of collocates and concordance lines of keywords in both corpora, I was able to identify two sets of keywords that indicate arguments in the research data (see Tables 7.1 and 7.3 below).⁸⁰ In order for a keyword to be selected as an argumentative indicator, I specified that it must be used to indicate argumentation in at least 50% of its occurrences. To avoid repetition of analysis, any keywords that I have presented in earlier analyses (such as 'Allah' and 'messenger', even if they appear to have argumentative functions) will not be included in the analysis of this chapter. It is worth noting here, as I have stated in Chapter 4, that the capitalized 'He' refers to 'Allah' following the traditional way of referring to 'Him' in Islamic discourses. The translations of the keywords are based on a preliminary concordance

^{78.} A topos (pl. topoi) is a persuasive strategy or rule which connects an argument to a claim or conclusion. Topoi are thus broad beliefs which help to maintain an argument without actually constituting the argument itself (see 3.2.3.1).

^{79.} A fallacy is a form of argumentation which appears convincing but is logically flawed –See Reisigl and Wodak (2001, pp. 71–74) and section 3.2.3.1 in this thesis.

^{80.} Tables 7.1 and 7.3 are abridged versions of two more detailed tables (Tables A3 and B3 in Appendices A and B respectively) that show the keywords along with their translations in English, the number of texts that contain the keyword and other statistical details such as percentage frequencies, and p values.

analysis (see Section 3.3.2 for translation conventions followed in this thesis). Let us move to consider the keywords that indicate argumentation in Ibn Baz.

7.2.1 Argumentation indicators in Ibn Baz

Keywords in Ibn Baz that indicate argumentation are tabulated in Table 7.1 below. These keywords are mostly lexical and carry religious meanings. Although these words are key in Ibn Baz, a few are also quite frequent in Alhawali's texts such as 'قال' (*qala*; He/he said) (1,100 times in Ibn Baz and 195 times in Alhawali), 'السنة' (*'assonnah*; the Sunnah of the Prophet of Islam)⁸¹ (142 times in Ibn Baz and 35 times in Alhawali), and '*faqala*; then He/he said) which occurs 85 times in Ibn Baz and 30 times in Alhawali.

N	Key word	Eng. trans.	Freq.	F. Alhawali	Keyness
1	ئ ال	He/he said	1100	195	1129.327
2	قرله	His/his saying	221	1	395.1075
3	الأية	the Qura'anic verse/ till the end of the Qura'anic verse	129	5	199.8372
4	الأيات	the Qura'anic verses	138	14	174.8027
5	الأحاديث	Hadiths (the sayings of the Prophet of Islam)	83	4	124.3767
6	السنة	the Sunnah of the Prophet of Islam	142	35	120.9175
7	فآراجب	therefore it is obligatory (for Muslims) to	46	0	84.65556
8	رواد	(this Hadith is) narrated by	56	3	82.39207
9	فتأل	then (He/he) said	85	30	54. 9 1756
10	الأدلة	the indications (from Qura'an and Prophetic traditions)/ proofs	54	10	54.06848
11	تبت	it became firm (that the Prophet said)	46	6	53.56055
12	حديث	Hadith	48	9	47.76144
13	لتول	because (He'he) said	24	0	44.16569
14	لأن	because	158	110	39.72014
15	المتحيحين	the two most authentic books of Hadiths	25	1	38.5453
16	آيات	Qura'anic verses	31	4	36.23737
17	منح	it was narrated in a sound Hadith	33	5	36.21875
18	دلت	indicated (fem. Sing. and pl.)	26	2	35.4 69 52
19	بذلك	hence/ by that means	147	105	34.95971
20	≟ ل	indicated (masculine Sing.)	27	3	33.23123

Table 7.1 contains different types of keywords that indicate argumentation in Ibn Baz which I

have divided into three categories:

1. Keywords (verbs and nouns) that indicate *narration* of canonical texts (7 keywords)

^{81.} As I have indicated in section 6.2, the keyword 'السنة' ('assonnah; the Sunnah of the Prophet of Islam) could also refer to the social actors 'the Sunnis' if it is preceded with the word 'أهل' ('ahl'; the people of). Both keywords occurred jointly (33 times) referring to the Sunnis. The rest of the occurrences (108 times) indicate the Sunnah of the Prophet and once was used to indicate 'all' 'olamā' 'assonah; the Sunni religious leaders).

- 2. Keywords that refer to canonical texts (8 keywords)
- 3. Grammatical and other keywords (5 keywords)

Table 7.2 shows the categories of the argument-indicating keywords in Ibn Baz (I have only provided the English translations of the words – see Table 7.1 for the original Arabic words).

N	Categories	Keywords
1	Verbs and nouns that indicate narration from canonical texts	He/he said, His/his saying, (this Hadith is) narrated by, then (He/he) said, it became firm (that the Prophet said), because (He/he) said, it was narrated in a sound Hadith,
2	Nouns that denote canonical texts	the Qura'anic verse/ till the end of the Qura'anic verse, the Qura'anic verses, Hadiths (the sayings of the Prophet of Islam), the Sunnah of the Prophet of Islam, the indications (from Qura'an and Prophetic traditions)/ proofs, Hadith, the two most authentic books of Hadiths, Qura'anic verses
3	Grammatical and other keywords	therefore it is obligatory (for Muslims) to, because, indicated (fem. Sing. and pl.), hence by that means, indicated (masculine Sing.)

Table 7.2 Categories of keywords that indicate argumentation in Ibn Baz

Most of the keywords in the categories above show Ibn Baz's focus on using keywords that carry religious meanings. The keyword with the highest keyness score is 'قال' (*qala*; He/he said) (1100 times). In religious texts, this keyword may indicate reporting the sayings of 'Allah', the Prophet of Islam or someone else. It is derived from the root ' $j \in l$ ' (*q* w l; to say)⁸². There are three other keywords that are derived from the same root. These keywords are 'قول' (*qawlaho*; His/his sayings), 'i (*faqāla*; then He/he said) and 'i (*leqawl*; because (He/he) said). It seems that Ibn Baz uses the key term 'SAY' to introduce Qur'anic verses and Prophetic traditions to ideologically legitimate his stances (detailed below).

There are other words such as 'تقول' (taqūl; she says) that are derived from the same root but will not be considered here as they are not keywords and occur in fewer than three texts. Additionally, the keyword 'قل' (qol/qalla; you say –imp./to be rare) (91 occurrences) is also used by Ibn Baz but it is not included in the analysis as it is homographic in nature. This word is used by Ibn Baz in two different senses. The first meaning that this word bears in Ibn

^{82.} I follow the traditional grammar text books in writing the roots of Arabic words in disconnected letters.

Baz is 'i (*qalla*; to be rare) which is not argumentative (6 occurrences). Secondly, the form of this word that is derived from the root 'i (*q w l*; to say) occurs in Ibn Baz as parts of different Qur'anic verses that are introduced by other word forms of the key term 'SAY'. Thus in order to avoid redundancy it will not be included in the analysis. Let us now move to sketch out some other argumentative keywords in Ibn Baz.

The keyword 'فالواجب' (falwāğib; therefore it is obligatory (for Muslims) to), in category 3 in Table 7.2, is derived from the root 'و ج ب' (wa ğa ba) which basically means 'to be necessary'. Ibn Baz also uses some other keywords that are derived from the same root. These keywords are 'أوجب' ('alwāğib; the obligatory (duty)) (keyness score 43.859), 'أوجب' ('awğaba; ((Allah) made ... obligatory) (keyness score 45.708) and 'أوجب' (wağūb; obligation) (keyness score 41.061). All these terms indicate religious 'injunctions' in Islamic Sharia where they are lesser, in degree, than the 'فرض' (fard; an ordinance enjoined by Allah). The denial of the first is a sin that, according to Muslim religious leaders,⁸³ could be forgiven by Allah whereas the denial of the second is infidelity which is a grave sin that is viewed within Islam as going completely against Islam. 'vi' (li'anna; because) and 'i(liqawl; because (He/he) said) indicate causality which argumentatively may indicate cause and effect forms of argument. Let us now consider the keywords that indicate the presence of arguments in Alhawali.

7.2.2 Argumentation indicators in Alhawali

The keywords in Ibn Baz (with the exception of 'فالواجب' (*falwāğib*; therefore it is obligatory to) and 'لقول' (*liqawl*; because (He/he) said) are also used by Alhawali, though on a smaller scale. In contrast to Ibn Baz, the keywords that indicate arguments in Alhawali (Table 7.3) do not overtly appear to bear any religious meanings.

^{83.} An exception in rejecting or not doing the obligatory duties is that the Kharijite faction (modern Kharijite faction is called Ibadite who live in Oman and the Nafusa Mountains in Libya) see it as infidelity.

Table 7.3 Argumentation indicators in Alhawali

N	Keyword	Eng. trans.	F.	F.Ibn Baz	Keyness
1	کمن	as	583	1	580.49
2	ان ا	indeed/if	750	192	163.557
3	-	but	315	60	100.997
4		may	154	17	77.139
5	بل	rather/evem/however	485	140	72.498
6	لذلك	therefore/because of that/based on that/(to/for) that	136	17	62.826
7	نترجة	result	70	6	40.231
8		standpoint/point of view	50	2	37.562
9	الباحتين	the researchers	34	0	34.567
10		I	29	0	29.484
11	ئتول	(she/it/they -fem. pl.) say(s)	57	6	29.367

Preliminary concordance analysis revealed that the keywords 'لذلك' (*lidālik*; therefore/because of that/based on that/(to/for) that)⁸⁴ and 'تَعَبِّهُ' (*natīğah*; result) indicate causality. The keyword 'موقف' (mawqif; standpoint/point of view) is used by Alhawali to introduce the standpoint of people he views as opponents. Of the 50 occurrences of the keyword 'موقف' (mawqif; standpoint/point of view), Alhawali uses it to denote the standpoints of the Church (10 times), America (2 times), Europe (3 times), the Soviet Union (once), Freud (an Austrian psychiatrist 1856-1939) (4 times), Iraq, Iran and Kuwait (altogether 5 times) and modern sciences and philosophy (5 times). He uses these standpoints either to support his own viewpoint or refute the stances of his opponents. The keywords 'الباحثين' ('albaḥitīn; the researchers –acc.) and 'تقول' (taqūl; (she/it/they –fem. pl.) say(s)) indicate that Alhawali seems to have used another form of appeal to authority other than that of Ibn Baz, who has tended to quote Allah (shown below and elsewhere, see Chapter 4).

(*'albaḥitiīn*; the researchers –acc.) is predominantly used by Alhawali in one text refuting secularism (30 times). His use of 'الباحثين' (*'albaḥitiīn*; the researchers –acc.) is not always used in argumentation scheme from authority (symptomatic argumentation). Alhawali generally uses the word 'الباحثين' (*'albaḥitīn*; the researchers –acc.) to refer to

^{84.} The keyword 'لذلك' (*ledālik*; therefore/because of that/based on that/to that) does not always indicate causality. Alhawali uses it also to indicate the meaning 'to that' as indicated by the English translation based on a preliminary concordance analysis of the keyword (see Section 7.5 below). It is worth noting here that even lbn Baz uses the same word to indicate argumentation only 4 times (out of 17 times).

unspecified researchers rather than naming them in his refutations of secularism. In two of his texts, the keyword 'الباحثين' ('albahitīn; the researchers –acc.) is pre-modified by its collocate 'بحض' (ba'd; some) (joint frequency 8 and MI score 7.504) which anonymously aggregates 'researchers'. This type of personalization (i.e. aggregation), van Leeuwen (2008, p. 37) writes, 'is used to regulate practice and manufacture consensus opinion'. The use of the collocate 'بحض' (ba'd; some) could also be considered as a form of the category 'indetermination' which, according to van Leeuwen (ibid, p. 40), 'endows social actors with a kind of impersonal authority, a sense of unseen, yet powerfully felt coercive force'. Alhawali also uses other word forms of the keyword 'الباحثي' ('albahitīn; the researchers –acc.) such as the singular form 'لاباحثي' ('albahitīn; the researchers –acc.) (8 times). These word forms are used in different social actor representations such as aggregation as in the plural forms of the word or anonymously classifying social actors.

ر (taqūl; (she/it/they say(s) – fem. pl.) is a verb derived from the root 'قول' (q w l; to say). It carries a suffix that indicates a feminine grammatical subject. Nouns, in Arabic, are classified as either masculine or feminine. Words like 'reports'⁸⁵, newspaper(s), and 'study/studies' are all treated as feminine nouns. Verbs that either follow or precede such nouns therefore have the feminine affix. Of the 57 occurrences of the keyword 'قول' (taqūl; (she/it/they say(s) - fem. sing./pl.), Alhawali argumentatively uses it 17 times quoting studies and theories, 7 times quoting the media (articles in newspaper and magazines), 7 times non-Islamic religious texts and institutions (such as the Torah, Acts of the Apostles and the Church) and 4 times utilizing quotes from famous westerners including the murder mystery novelist Agatha Christie and the movie star Marilyn Monroe (described in more detail below).

^{85.} The singular form of 'reports' is treated as masculine in Arabic.

Alhawali also uses some other forms of the reporting key term 'SAY' in Ibn Baz as indicated by Table 7.1 above which include 'قَلْنَ' (*qala*; He/he said) (195 times) and 'قَتْلَ' (*faqala*; then He/he said) (30 times) and 'قَلْنَ' (*qol/qalla*; you say –imp./to be rare) (21 times). However, of the 21 occurrences of the word 'قَلْن' (*qol/qalla*; you say –imp./to be rare), Alhawali uses it 9 times denoting the meaning 'to be rare' which is not argumentative. The rest of the occurrences of this word are used by Alhawali as part of Qur'anic verses that are introduced by other reporting words derived from the root 'قَالَ (*q w l*; to say). All these occurrences are used as warrants in Alhawali's arguments. He also uses the keyword 'الباحثين' ('*albaḥitin*; the researchers –acc.) in the same manner. Let us now explore some of these keywords in more detail in order to examine argumentation schemes in both corpora. I shall present the argumentation schemes in both corpora (Section 7.3), and then I shall evaluate them critically in Sections 7.4 and 7.5.

7.3 Argumentation Schemes in the Salafi Corpora:

Both Salafi writers use argumentative keywords that indicate schemes relating to authority and causality. On the one hand, Ibn Baz uses keywords (in categories 1 and 2 in Table 7.2) that refer to Islamic canonical texts which reveal Ibn Baz's focus on religious authority. He also uses words such as 'فالواجب' (*falwāğib*; therefore it is obligatory to) and 'لقول' (*leqawl*; because (He/he) said) that indicate causality. On the other hand, Alhawali uses keywords such as (*lidālik*; therefore/because of that/based on that/(to/for) that) and 'iiii (*natīğah*; result) which could also be translated as 'as a consequence or consequently' that indicate causality. He also uses keywords that indicate arguments from authority, viz., 'الباحثين' (*'albaḥitīn*; the researchers –acc.) and 'iie) '*iieut*'; (*taqūl*; (she/it/they say(s) fem. sing./pl.). These two schemes (causality and authority/argumentation scheme from sign), as they are indicated by keywords, seem to be the most salient forms that are used in the texts of the two Salafī authors revealing their authorial stances. Let us look at each argumentation scheme and then evaluate these arguments based on critical questions adopted from Walton (1995, p. 142) and Walton et. al. (2008, pp. 309-346) that would reveal the (un)acceptability of the arguments and the ideological stances of each Salafi writer. It is worth noting here that I subject the unacceptable arguments only to the critical evaluation. I shall start with the argumentation in Ibn Baz.

7.4 Ideological legitimation (1): Argumentation schemes in Ibn Baz

As I have indicated above, Ibn Baz seems to rely heavily on both Qur'anic verses and Prophetic traditions to warrant his arguments and legitimate his stances vis-à-vis his representations of social actors as well as the thematic foci addressed in his texts. The key term 'SAY' as shown above contains 4 keywords, viz., (qala; He/he said) (frequency 1100, keyness score 1129.327), 'قوله' (qawlaho; His/his sayings) (Freq. 221, Keyness 395.107), 'قول' (faqāla; then He/he said) (Freq. 85, keyness 54.917) and 'قول' (leqawl; because (He/he) said) (Freq. 24, keyness 44.165). Due to space and time restrictions I shall only analyze the keyword 'قول' (qala; He/he said) as it is the highest in both keyness and frequency.

The key concept 'SAY' collocates with 82 words (see Table A9 in Appendix A). Of the 82 collocates, the keyword 'قال' (*qala*; He/he said) collocates with 61 words. These collocates could be categorized as follows:

- Collocates that are apparently parts of Qur'anic verses or indicate the Qur'an. The collocates that are parts of the Qur'an are written with diacritics that are mostly used in writing Qur'anic verses in the texts of Ibn Baz and Alhawali.
- Collocates that indicate three types of grammatical subjects: Allah, the Prophet of Islam and some other names and references to Salafi religious leaders.
- 3) Grammatical and other words that indicate none of the above.

These categories of collocates are presented in Table 7.4 below:

	(quid, field said) in fon Baz
categories	Collocates of 'أفل' (qala; He/he said
Qur'an	therefore worship Me, We have sent down, verily, (what) has been revealed (to you (O Mohammad) from your Lord), We have sent, A Book (revelation), verily (Allah) will aid, (did) you not, (if) you (were to) ask them, (it was) incumbent (upon Us), upon Us, they will certainly say, help, that, if (you) were to, false deities, (that) you (may) obtain mercy, end, believers (accusative), this, I (Allah), All-Hearer, created, with full submission, you (mankind) worship, Jinn (genies), (Allah) wills (that something be done), your Lord, the Qur'anic verse/ till the end of the Qur'anic verse, (those) who believed, O (vocative), O (another form of vocative), that He/he
Allah	the Exalted, the Exalted in Mighty, the Majestic, the Most High, the Glorified, other than Him/him
Prophet	the Prophet, messenger, (may Allah send) His peace (upon him)
Salaf	Alhafidh (a higher status of religious leaders in Sunni Islam), (may Allah) have mercy on him, the father of, Malik, Sheikh, the son of (may Allah) be pleased with him, Imam, (may Allah be pleased) with him
Other collocates	that you may, yes' best, I said, as, you say (imperative), verily, as, indeed, to you, for this reason, those

Table 7.4 Categorized collocates of 'قال' (qala; He/he said) in Ibn Baz

Concordance 7.1 Concordance of the collocate 'ناغندون' (fa 'bodūn; therefore worship Me) in Ibn Baz

N	Concordance
	وقال تعالى: {وَمَا أَرْسَلْنَا مِن قُبْلِكَ مِن رَّسُولٍ إلَّا نُوجى إليه أنَّهُ لا إلنه إلا أنَّا فأغبُلون}، وقال عن نوح
1	and the Exalted said: {And We did not send any Messenger before you (O Muhammad) but We revealed to
1	him (saying): La ilaha illa Ana [none has the right to be worshipped but I (Allah)], therefore worship Me (Alone
	and none else)} and He said about Nuh (Noah)
	و قال تعالى : {وَمَا أَرْسَلْنَا مِن قُبِّلِكَ مِن رَّسُولٍ إِلَّا نُوحِي إلَيْهِ أَنَّهُ لا إِلَـهَ إِلا أَنَا فَاعْبُدُونَ } وَ قال سبحانه :
2	and the Exalted said: {And We did not send any Messenger before you (O Muhammad) but We revealed to
-	him (saying): La ilaha illa Ana [none has the right to be worshipped but I (Allah)], therefore worship Me (Alone
	and none else)}and the Glorified said
	و قال تعالى : {وَمَا أَرْسَلُنَا مِن قَلِيْكَ مِن رَّسُولٍ إِنَّا نُوحِي إِلَيْهِ أَنَّهُ لَا إِلَهُ إِنَّه لا أَنا <u>فَاعْبُدُون</u> } و قال عز و جل
3	and the Exalted said: {And We did not send any Messenger before you (O Muhammad) but We revealed to
3	him (saying): La ilaha illa Ana [none has the right to be worshipped but I (Allah)], therefore worship Me (Alone
	and none else)) and the Exalted in Mighty and the Majestic said
	يقول تعالى : {وَمَا أَرْسَلُنَا مِن قَلْلَكَ مِن رَّسُولٍ إِلَّا نُوجى إِنَيْهِ أَنَهُ لا إِنَّهَ إِنَّه إِنَّه إِنَّه إِنَّهُ لا إِنَّه إِنَّهُ لا إِنَّه إِنَّهُ لا إِنَّهُ إِنَّهُ الْمُ
4	and the Exalted says: (And We did not send any Messenger before you (O Muhammad) but We revealed to
4	him (saying): La ilaha illa Ana [none has the right to be worshipped but I (Allah)], therefore worship Me (Alone
	and none else)}, and the Glorified says

All the lines above indicate that the collocate 'فَا عَبْدُونِ (*fa 'bodūn*; therefore worship Me) is part of a Qur'anic verse that is preceded either by وقال تعالى ' (*wa qala ta 'āla*; and the Exalted said) (Lines 1, 2, 3, 4) or 'يقول تعالى ' (*yqolo ta 'āla*; the Exalted says) which refer to Allah. The noun collocates of the keyword 'قال ' (*qala*; He/he said) in 'Allah', 'the Prophet' and 'Salaf' categories are always grammatical subjects. They hold an argumentative authoritative relation with the keyword 'قال ' (*qala*; He/he said). Below (Concordance 7.2) are examples of these categories. Due to space and time restrictions, I shall present only one concordance line as an example of one collocate that has the highest MI score in each category, i.e. 'الحافظ ' (*'alḥafiz*: Alhafidh: a higher status of the scholars of Hadith) (joint freq. 10 and MI score 5.838), 'النبي ' (*ta 'āla*; the Exalted) (joint freq. 242 and MI score 5.624) and ' *'annabi*; the Prophet) (joint freq. 37 and MI score 3.999):

Concordance 7.2 Examples of noun collocates of the keyword 'قال' (qala; He/he said) in Ibn Baz

_	Concordance 7.2 Examples of noun conceases of the Reyword Co (quid, fic/he said) in foir baz
Ν	Concordance
	[] و لهذا قال عن و جل: {يا أيها الذين آمنوا لا تأكُلُوا أمُؤالكُمْ بَيْنِكُمْ بِٱبْنَطِلَى} و قال النبي علية الصلاة و السلام : ((كل المسلم على المسلم حرام دمه و ماله و عرضه)) []
1	[] and therefore the Exalted in Mighty and the Majestic said: {O ye who believe! Squander not your wealth among
	yourselves in vanity} and the Prophet, may Allah's blessings and peace be upon him, said: (The whole of a Muslim
	for another Muslim is inviolable: his blood, his property, and his honour) []
	قد دل القرآن الكريم و السنة المطهرة على أن العبادة حق لله وحده ، و أنه لا يدعى إلا الله ، و لا يستحان إلا به ، كما قال تعالى : {إِثَاثَ تُعْبُدُ و إِيَّاكَ
	نستُتِعِينُ} و قال تعالى : {و أنَّ الْمُسَاحِدَ لله فلا تَدْعُوا مَعَ الله أَحَدًا} و قال النبي صلى الله عليه و سلم : ((الدعاء هو العبادة))[]
	Verily the Holy Qur'an and the purified Sunnah of the Prophet indicated that 'worship is to be dedicated to Allah
	Alone' and no one can be invoked or beseech his help but Allah, as the Exalted said: { You (Alone) we worship
	and You (Alone) we beseech for help} and the Exalted said: {And the mosques are for Allah (Alone): so invoke
	not anyone along with Allah} and the Prophet, may Allah send His blessings and peace upon him, said: (Supplication
	is (a form of) worship) []
	[] وقال موفق الدين ابن قدامة رحمه الله [] قول رسول الله صلى الله عليه و سلم حجة ، [] وقال الحافظ ابن كثير رحمه الله في تغسير
	قوله تعالى : {لَيَخُذُرِ الذين يُحَالِفُونَ عن أَمْرِدِ أَنَّ تُصِيبَهُمُ فِتْنَهُ أَو يُصِيبَهُمْ عَذَابً أَنِيمُ} أَنِي : عن أمر رسول الله صلى الله عليه و سلم ، و هو سبيله و
	منهاجه و طريقته []
	[] and Mwaffaq Addeen Ibn Qudamah, may Allah have mercy upon him, said [] and the sayings of the
3	messenger of Allah are competent authority, and Alhafidh Ibn Katheer, may Allah have mercy upon him, said in his
	explication of the following saying of the Exalted (i.e. The Qur'anic verse) {And let those who oppose his
	commandment beware, lest some Fitnah (disbelief, trials, afflictions, earthquakes, killing, overpowered by a tyrant)
	should befall them or a painful torment be inflicted on them}, his commandment is the Messenger's (Muhammad)
	Sunnah - legal ways, orders, acts of worship and statements []

Lines 1 and 2 indicate that Ibn Baz is utilizing Qur'anic verses and Prophetic traditions in warranting his arguments. This is also reflected by the keywords in categories 1 and 2 (Table

7.4 above) which denote narrations of Qur'anic verses and Prophetic traditions. According to Muslims the Qur'anic verses are the words of Allah who is the highest divine authority in Islam, whereas the Prophetic traditions are the words that were said by the Prophet of Islam who is the highest human authority in Islam. In order for this warrant to be effective, we have to accept the premise that both the Qur'anic verses and Prophetic traditions are inherently true. This indicates that Ibn Baz, in his arguments, tends to appeal to Allah and the Prophet of Islam as the highest religious authority. As well as referring to Allah and the Prophet, Ibn Baz quotes others. In Concordance 7.2, line 3 nominates one of the influential medieval religious leaders. In this line, Ibn Baz quotes Ibn Kathir (1301-1373 a medieval Salafi religious leader and historian, well known for his explications of the Qur'an). The nomination of Alhafidh is also used by Ibn Baz when quoting other medieval Salafi religious leaders such as Alhafidh Ibn Rajab (1335-1393 a medieval Salafi religious leader who advocated the Hanbali jurisprudence school. He is also known for his commentaries on some books of Hadith and Alhafidh Aldhahabi (1274-1348) a medieval Salafi religious leader, who advocated the Shafi'i jurisprudence school and a leading figure in the study of Prophetic traditions. Additionally, Ibn Baz quotes another medieval religious leader (Line 3), Mwaffaq Addeen Ibn Qudamah (1147-1223) who is an advocate of the Hanbali jurisprudence school and antirationalist and speculative theological views. The inclusion of such names in Ibn Baz's texts indicates his reliance on the books of the medieval Salafi religious leaders. However Ibn Baz's reliance on Qur'anic verses and Prophetic traditions is more frequent than his quoting of medieval Salafi leaders. This is demonstrated by taking word clusters in Ibn Baz's writing into consideration.

There are 143 clusters that include the keyword 'قال' (*qala*; He/he said): 56 2-word clusters, 45 3-word clusters, 36 4-word clusters and 6 7-word clusters. All these clusters indicate that Ibn Baz relies heavily on quoting sacred texts (Qur'an and Prophetic traditions). Table 7.5 below shows the five top frequent clusters in each type:

	N	Word	Eve trans		Trut
Lu			Eng. trans.	Freq.	Texts
5	1	و قان نعانی	and the Exalted said	123	4
3-word	2		the Exalted in Mighty and Said	111	5
	3	قال تعالى و	the Exalted said and	92	4
clusters	4	و قال سبحانه	and the Glorified said	82	4
CIS	5	قال سبحانه و	the Glorified and said	80	5
4	1	قال عز و جل	the Exalted in Mighty and the Majestic said	111	5
4-word	2	و قال عز و	and the Exalted in Mighty and (the Majestic) said	62	4
	3	و سلم أنه قال	(may Allah send His blessings) and peace (upon him) that he said	55	3
dusters	4	و قال سبحانه و	and the Glorified and (the Exalted) said	47	4
ers	5	كما قال عز و	as the Exalted in Mighty and (the Majestic) said	40	4
	1	النبي صلى الله عليه و سلم قال	the Prophet, may Allah send His peace and blessings upon him, said	23	4
7-	2	قال النبي صلى الله عليه و سلم	(said) the Prophet, may Allah send His peace and blessings upon him	22	3
-word	3		(may) Allah send His blessings and peace upon him said	21	3
	4	و قال صلى الله عليه و سلم	and (the Prophet), may Allah send His peace and blessings upon	17	3
clusters	5	لقول النبي صلى الله عليه و سلم	because of the saying of the Prophet, may Allah send His blessings and peace upon him (i.e. Hadith)	7	3

Table 7.5 Word clusters that contain the keyword 'فل ' (qala; He/he said) in Ibn Baz's corpus

The cluster that is the highest in frequency is 'و قال تعالى' (wa qala ta'āla; and the Exalted said) (123 times). Two word clusters were not included in Table 7.5 as otherwise the list of clusters would have been much longer. However, it is worth noting that the most frequent 2-word cluster that contains the reporting verb 'قال' (qala; He/he said) and refers to Allah is also ' تعالى ' (qala ta'āla; the Exalted said) (194 times, see Table A10 in Appendix A) which is similar to the cluster ' وقال تعالى ' (wa qala ta'āla; and the Exalted said). The difference is only the addition of the additive particle ' وَ (wa; and) (see below). All the clusters in Table 7.5 indicate reference to Qur'anic verses and Prophetic traditions which are mostly used by Ibn Baz to warrant his arguments as in Concordance 7.3 below.

Concordance 7.3 Examples of concordance of word clusters that contain the keyword 'قال' (qala; He/he said) in Ibn Baz

N Concordance
[...] کدعائه و الإستغاثة به ، و طلبه المدد و الثنفاء للمرضي ، و النصر على الأعداء ، و نحو ذلك من أنواع الجادة. و الله سيحانه أوجب على عباده أن
[...] عنصود بالحدادة ،[...]، كما قال عن و جل : {و ما خلتُتُ الْجِنَ و الإِنْسَ إِلا لِبَخْدُون}
[...] such as invoking him [the Prophet and he is dead] beseeching his help, invoking him for support, ask him to heal from illnesses and grant victory on enemies and other worshipping practices. Allah made it obligatory for His servants (worshippers) to invoke Him Alone [...], as <u>the Exalted in Mighty and the Majestic said</u>: { And I (Allah) created not the jinn and mankind except that they should worship Me (Alone)}

[الإيمان بأسماء الله وصفائه] هي أوصاف لله - عز و جل- يجب و صفه بها على الوجه اللائق به من غير أن يتنابه خلقه في تسيء من صفائه ، كما قال تعالى : (ليس كمثله تسيء و هو السميع اليصير) و قال عز و جل : (فلا تُعتُر بُوا لله الأمثال إنّ الله يعلم و أنْتُم لا تَتْمَلُمونَ) [Belief in Allah and His names and attributes] they are attributes for Allah, the Exalted in Mighty and the Majestic, which should be used to when describing Him in a manner that suits Him without comparing Him to His creatures, as The Exalted said {There is nothing like Him; and He is the All-Hearer, the All-Seer} and the Exalted in Mighty and the Majestic said forward similitudes for Allah (as there is nothing similar to Him, nor He resembles anything). Truly! Allah knows and you know not}

[دعا الأمواك من الأنبياء و غير هم] و هذه الأعمال هي أعمال من مات على الشرك بالله عن و جل ، و هكنا الأعمال المبتدعة التي لم يأذن بها الله [...] كما ق**ال النبي صلى الله عليه و سلم** : ((من أحدث في أمرنا هذا ما أيس منه فهو رد)) متفق على صحته

[invoking dead prophets and others] and these are worshipping practices of those who were associating other with Allah, the Exalted in Mighty and the Majestic, till they died. And these are the religiously impermissible innovative practices that Allah did not allow (His creatures) to do [...] as the Prophet, may Allah send His blessings and peace upon him, said: (Whosoever innovate sny worshipping practice that is not (in agreement) with Islam, will have it rejected). This is sound Hadith that is narrated by Bukhari and Muslim

أما المسلاة و السلام على رسول الله صلى الله عليه و سلم فيى من أفضل القريات ، و من الأعمال المسللحات ، كما قال الله تعالى : {إِنَّ الله و مَلائِكُهُ يُحتَلونَ على النَّبِيَّ لا أَبِها الذين آمنوا منلُوا عَلَيْهِ و سَلَّمُوا تُسْلِيمُا } و قال النبي صلى الله عليه و سلم : ((من صلى على واحدة صلى الله عليه بها عسراً)) regarding invoking Allah to send His blessings and peace upon the Prophet is one of the best deeds that bring us closer to Allah, and one of the righteous and good deeds as Allah, the Exalted, said {Allah sends His Salat (Graces, Honours, Blessings, Mercy) on the Prophet (Muhammad), and also His angels (ask Allah to bless and forgive him). O you who believe! Send your Salat on (ask Allah to bless) him (Muhammad), and (you should) greet (salute) him with the Islamic way of greeting} and <u>the</u> **Prophet, may Allah send His blessings and peace upon him, said**: (whoever invokes Allah to send His blessings on me once, Allah, in return, blesses him/her ten times)

4

All lines in Concordance 7.3 indicate that Ibn Baz utilizes both Qur'anic verses and Prophetic traditions in his arguments. Lines I and 3 are refutations of ritualistic practices of some Sufi and Shiites (see Chapter 4, Section 4.2.2), such as invoking or beseeching the help of prophets and other dead individuals, utilizing a Qur'anic verse (Line 1) and a Prophetic tradition (Line 3) to legitimate his stance towards such practices that he conceives to be religiously impermissible. In Line 2, Ibn Baz uses two Qur'anic verses to support his stance concerning the names and attributes of Allah. He invites Muslims to describe Allah by His attributes in a manner that suits Him. The last line indicates Ibn Baz's opinion towards invoking Allah to send His blessings and peace upon the Prophet of Islam. He uses a Qur'anic verse and a Prophetic tradition to support his claim that such invocation is one of the best deeds that brings Muslims close to Allah. Let us now take the most frequent cluster

in Table 7.5 to capture the most frequent and salient argumentation scheme that appears in Ibn Baz's writings.

As part of Islam, Muslims believe that the Qur'an is the words of Allah who is the ultimate authority in Islamic faith. Hence all Qur'anic verses are inherently viewed as true in Islam. In Ibn Baz's corpus, the Qur'anic verses are indicated by the reporting keywords such as 'قال' (qala; He said) and the keywords that refer to 'Allah' (see Chapter 4, Section 4.2.1). As I have indicated above, such keywords tend to co-occur with each other ('قال') (qala; He said) and 'نعالى' (ta'āla; and the Exalted). The 3-word cluster 'و قال تعالى' (wa qala ta'āla; and the Exalted said) (123 times) and the 2-word cluster 'قال تعالى (gala ta 'āla; the Exalted said) (194 times) are the most frequent of all the clusters in Ibn Baz that contain the verb 'قل' (*gala*; He/he said). The linguistic structure of these two types of clusters above is simply put forward as follows: reporting verb 'قال' (qala; He said) + the grammatical subject (ta ' \bar{a} la; the Exalted) which functions as the agent in the structure. The addition of the additive particle 'y' (wa; and) signifies that Ibn Baz uses hypotactic structures in order to put forward a constellation of warrants in one argument which could be seen as coordinative argumentation. The word 'تعالى' (ta'āla; the Exalted) is one appellation that is used to talk about 'Allah' who is the ultimate authority in Islam. In order to narrow down the scope of analysis and capture the most frequent and salient form of argumentation schemes in Ibn Baz, let us move to concordance-based analysis of the 3-word cluster 'و قال تعالى (wa qala ta ʿāla: and the Exalted said).

In order to choose the concordance lines for more explanations and further analysis of the argumentation scheme(s), I shall present only the arguments that pertain to the semantic foci discussed in Chapter 4. I shall present each concordance line including more of the cotext in order to show the whole argumentation thread. By argumentation thread I mean the linguistic chunks (whole sentences) that arguments appear in, starting from the premise going through the warrant(s) and ending up with the conclusion. Argumentation threads, in my data, normally occupy larger linguistic chunks than those included in concordance lines. They may also include a constellation of arguments – coordinative or subordinative argumentation (see below and Chapter 3). Let us move on to the first argument I want to consider here.

7.4.1 True beliefs and worship practices

In his defense of what he assumes to be true guidance of Allah, Ibn Baz introduces his arguments either to reject the beliefs and worship rituals practiced by some Muslims (mostly Sufis and Shiites) who do not adhere to the Salafi version of Islam or to support his views regarding certain theological aspects such as the particles of faith (see Section 7.4.3 below). Let us consider his first argument in the expanded version of concordance line 1 (see Appendix D for Concordances D1 and D2 that indicate arguments in both Salafi corpora):

Figure 7.1 Argumentation thread 1 (beliefs and ritual practices)

من العقائد المضادة للحق ما يعقده بعض المتصوفة : من أن بعض من يسمونهم بالأرلياء يتساركون الله في التدبير ، ويتسبر فون في تشون العالم ، ويسمونهم بالأقطاب و الأوتاد و الأغوات ، و غير اذلك من الأسماء التي اخترعوها لألهتهم ، و هذا من أقبح الشرك في الربوبية ، و هو شر من شرك جاهلية العرب؛ لأن كتار العرب لم يشركوا في الربوبية و إنما أشركوا في العبادة ، و كان شركهم في حال الرخاء ، أما في حال التدري فيخلصون فله الحبادة ، كما قال الله سيحانه : (فَإِذَا رَكِبُوا في الْفَائْةِ دَعُوا الله مُخْلِصِينَ له الدين قلمًا نجافة إلى أبَرَ إذا هم يُشْركُونَ) أما الربوبية فكانوا معترفين بها لله وحدد ، كما قال سبحانه (و لَنِنْ سَأَنْهُمْ مِن خَلَقَهُمْ لِيُتُوَلُنُ الله) ، و قال تعالى : (قُلْ من يُرْزُلُقُكُمْ من السُمَاءِ و الأرْض أَمَنْ بِمَلِكُ السُمْعَ و الأَبْصَارُ و من يُخْرِجُ الْحَيَّ من الْمَثِبَ و يُخْرِجُ الْمَثِبَ من الْحَيَّ من الْحَيَّ من الْحَيَ فَتُلُ أفلا تتَقُونَ) و الآيات في هذا المحنى كثيرة. أما المشركون المتأخرون فزادوا على الأولين من جهتين ، إحداهما : شرك بعضهم في الربوبية ، و الثانية : شركهم في الرخاء و الشدة ، كما يعلم نائك من خالطهم و سبر أحوالهم ، و رأى ما يغطون عند قبر الحسين و البنوي و غيرهما في مصر ، وعند قبر الحيدروس في عدن ، و الهادي في اليمن و ابن عربي في الشام ، [...] وغيرها من التبور المشهورة التي غلت فيها العامة و صرفوا لها الكثير من حق الله عز و جل. [...] و نسأله سبحانه أن يردهم إلى رتندهم ، [...] ، و أن يوفق قادة المسلمين و علماءهم لمحاربة هذا الشرك و التصاء عليه [...] Some other beliefs that oppose the true guidance of Allah are that of some Sufis: [Some Sufis] belief that the so called 'Aulliya' (friends of God) share Allah in disposing the affair of all things in the world. They name them 'the poles', the pillars', 'the aid providers' (ranks of Sufi saints) and other names they invented for their gods, which is more repugnant form of polytheism (that opposes the Unity of Lordship). It is even more wicked than the polytheistic beliefs of the ignorant Arabs in the pre-Islamic era. The case is so, because those Arab infidels (in the pre-Islamic period) did not practice polytheism that opposes the Unity of Lordship, but practiced the form of polytheism that opposes the Unity of worship. In addition, they practiced polytheism during times of comfort, but during times of hardships they sincerely worship Allah making their faith pure for Him Alone, as Allah, the Glorified, said: {And when they embark on a ship, they invoke Allah, making their Faith pure for Him only, but when He brings them safely to land, behold, they give a share of their worship to others}. Also, they do not contradict the Unity of Lordship as they confirm that Allah is their Lord, as the Glorified said { And if you ask them who created them, they will surely say: "Allah"}, and the Exalted said: {Say (O Muhammad): "Who provides for you from the sky and the earth? Or who owns hearing and sight? And who brings out the living from the dead and brings out the dead from the living? And who disposes the affairs?" They will say: "Allah." Say: "Will you not then be afraid of Allah's punishment (for setting up rivals in worship with Allah)?"}. And there are plenty of Qur'anic verses that indicate such meanings. However, modern-day polytheists added two aspects to the old form of polytheism: they associate others with Allah in His Lordship and commit polytheism in times of both comfort and hardship. This is known about them (modern-day polytheists] to those who live with them and explored their style of worship and saw what they did near the grave of 'Alhussein', and Albadawi and other graves of Sufi saints in Egypt, Alahdarouse in Aden, Alhadi in Yemen. Ibn Arabi in Cham (Damascus), [...] and some other famous graves that common people exceeded the limits and perform rituals of worship that are His right (i.e. these rituals are supposed to be conducted to Allah Alone), [...] we pray to Allah to help them (Some Sufis) to regain their senses and guid Muslim political and religious leaders to the right path in order to fight and eradicate this type of polytheism [...]

In the above expanded version of concordance line 1, we first observe the premise 'Some other beliefs that oppose the true guidance of Allah are that of some Sufis' which indicates Ibn Baz's negative stance towards some mystic beliefs that he believes are in opposition to the 'true guidance of Allah'. Second, Ibn Baz uses two of the argumentation schemes in Van Eemeren and Grootendorst's (2004) typology: by analogy and by sign. Using analogy to compare the mystic practices of modern-day Sufis to polytheistic practices of the Arabs in the pre-Islamic era, Ibn Baz uses the elative (gradation) form of the adjective 'شر' (šarr; wicked) and the word 'شر' (men; than) that could be translated into English as 'more wicked than)'. He warrants his comparison with a Qur'anic verse that is introduced by the phrase 'من ' (kama qala 'allahu sobhanh; as Allah the glorified said) which is always used in Islamic discourses to introduce Qur'anic verse that indicate appeals to divine authority. This takes us to the second argumentation scheme used by Ibn Baz: argumentation scheme from sign.

In the expanded version of Line 1 (argumentation thread 1), Ibn Baz introduces his argument using 'كما قال الله سبحانه' (*kama qala 'allahu sobhanh*; as Allah, the Glorified said). He also uses another Qur'anic verse which is introduced by 'و قال تعالى' (*wa qala ta 'āla*; and the Exalted said). Ibn Baz concludes his argument by invoking Allah to guide those whom he believes to practice polytheism and the Muslim political and religious leaders to fight and eradicate these Sufi ritualistic practices.

Utilizing a group of argumentation schemes, as I have indicated above, warranting his arguments with Qur'anic verses (refuting the mystic beliefs of the Sufis about their saints), Ibn Baz seeks to legitimate his stance towards such beliefs without introducing the Sufis' arguments and explanations of such practice. In other words, Ibn Baz does not engage with the beliefs of the Sufis or their reasons for their beliefs. He, on the contrary, starts from a pre-conceived assumption, namely, the Salafist ideological perspective which rejects the

previously mentioned mystic practices (see below). Line 2 further develops Ibn Baz's arguments and shows the Salafist perspective toward such practices, though he does not name any Sufi or Shiite social actors.

Figure 7.2 Argumentation thread 2 (the Salafi definition of worship)

و العبادة : هي طاعته سبحانه و طاعة رسوله محمد صلى الله عليه و سلم ، بنعل ما أمر الله به و رسوله ، و ترك ما نهى الله عنه و رسوله و عن إيمان بألله و رسوله ، و إخلاص لله في العمل ، مع غاية الحب لله و كمال الذل له وحدد كما قال تعالى : {و قطتى زبلك ألا تُعْبُدُوا إلا إيَّهُ أي أمر و أوصى بأن يعبد وحدد و قال تعالى : {الْحَمْدُ للّه رَبَّ الْعَانَبِينَ * الرُحْمَنِ الرَّحِيمِ * مَالِكِ يَوْمِ الذَينِ * إيَّكَ نَتْبُدُو إيَّكَ تَسْتِعِينُ} أبان سبحانه بهذه الآيات أنه هو المستحق لأن يعبد وحده و قال تعالى : {الْحَمْدُ للله رَبَّ الْعَانَبِينَ * الرُحْمَنِ الرَّحِيمِ * مَالِكِ يَوْمِ الذَينِ * إيَّكَ نَتْبُدُو إيَّكَ تَسْتَعِينُ} أبان سبحانه بهذه الآيات أنه هو المستحق لأن يعبد وحده و يعن عان به وحده ، و قال تعالى : له الدين * ألا لله الذَينُ الْخَالِصُ في قال تعالى : {فَذَعُوا اللهُ مُؤْلِمِينَ له الذين و لو كره الكَفِرُونَ في قال تعالى : {و أنَّ أَلْمَعَاجَد للله فذا المعنى كثيرة ، و كليان ألخَالِصُ في قال تعالى : {فَذَعُوا اللهُ مُؤْلِمِينَ له الذين و لو كره الكَفِرُونَ في قال تعالى : {و الآيات في هذا المعنى كثيرة ، و كليا تدل على : وجوب إفراد الله بالعبادة ، و معلوم أن الدعاء بأنواعه من العبادة ، فلا يودر إلا ربه ، و لا يستعين و لا يستغيت إلا به ، عملا بهذه الأيات الكريمة ، و ما جاء في معناها و هذا فيما عدا الأمور العادية ، و الأسباب الحسية ، التي يقدر عليها المخلوق الحر الحن ، فإن تلك ليستعين و لا يستغيت إلا به ، عملا بهذه الأيات الكريمة ، و ما جاء في معناها و هذا فيما عدا الأمور العادية ، و الأسباب الحسية ، التي يقدر عليها المخلوق الحى الحاس ، فإن تلك ليستعين و لا يستغيت إلا به ، عملا

And the <u>worship is obeying, Him the Glorified, and obeying His messenger Muhammad</u>, may Allah send His blessings and peace upon him. It is doing what Allah and His messenger ordered us to do and refrain from committing what Allah and His messenger forbid holding faith in Allah and His messenger. It is the sincere worship of Allah making faith pure to Him Alone. Worship is also the ultimate devotion of love to Allah and the complete submissiveness to Him Alone <u>as the Exalted</u>: {And your Lord has decreed that you worship none but Him} which means that He ordered and enjoined on us to worship Him Alone, <u>and the Exalted said</u>: {So worship Allah (Alone) by doing religious deeds sincerely for Allah's sake only. Surely the religion (i.e. the worship and the obdeience) is for Allah only}, <u>and the Exalted said</u>: {So, call you (O Muhammad and the believers) upon (or invoke) Allah making (your) worship pure for Him (Alone) (by worshipping none but Him and by doing religious deeds sincerely for Allah's sake only and not to show off and not to set up rivals with Him in worship), however much the disbelievers (in the Oneness of Allah) may hate (it)}, <u>and the Exalted said</u>: {And the mosques are for Allah (Alone): so invoke not anyone along with Allah}, and the Qur'anic verses about this issue are many that indicate the obligation of worshipping Allah Alone. And <u>it is known that all types of supplications are part of worship</u>, thus, according to these Qur'anic verses, <u>it is</u> <u>impermissible for all people to supplicate to and neither seek nor beseech the help from other than Allah</u> in order to fulfil the obligation indicated by the previously mentioned Qur'anic verses and other verses that bear the same meanings. However, it is not considered as a form of worship if someone takes proper means to seek the help of living human beings who are present in normal issues they can do. On the contrary it is permissible as indicated by both <u>sacred texts (Qur'an and Prophetic traditions) and the consensus of the (Salafi)</u>

Ibn Baz uses a group of Qur'anic verses to warrant his definition of the concept of worship. He then moves on to his conclusion and stance towards the types of supplication that are usually practiced by some Sufis and Shiites to invoke and beseech the help of dead humans such as the Prophet of Islam, his grandson 'Alhussein' or any other saints, considering such practices as forms of worship. He concludes that such practices are impermissible in Islam. Additionally, Ibn Baz uses both the sacred texts and the consensus of religious leaders as an authority to legitimate his Salafi stance towards the religiously permissible beseeching of help from individuals who are alive and present provided that such help is in worldly issues that those individuals are capable to provide. In order to evaluate the arguments that appear in the schemes above, we need first to consider the structures of the arguments that appear therein, and then utilize some critical questions adopted from Walton (1995, p. 142) and Walton et. al. (2008, pp. 309-346). Let us first restructure the arguments in argumentation threads 1 and 2:

		Arguments structures in Ibn Baz (argumentation threads 1 and 2)	Schemes	
Prer	1	Some beliefs of some modern-day Sufis oppose the true guidance of Allah. These beliefs are similar to/even worse than the polytheistic beliefs of the ignorant Arabs in the pre-Islamic era		
Premises	2	Worship is obeying Him, the Glorified, and obeying His messenger Muhammad/All types of supplications are part of worship		
Warrants	1	The similarity between some Sufi beliefs and Arabs infidels in the pre-Islamic era/ Qur'anic verses introduced by 'as Allah, the Glorified said' and 'and the Exalted said'	Analogy / Sign	
unts	2	Qur'anic verses introduced by the phrases 'as the Exalted said' and 'and the Exalted said'	y / Si	
Conc	1	Muslim political and religious leaders should fight and eradicate the Sufi and Shiite polytheistic beliefs and rituals	gn	
Conclusions	2	It is impermissible for all people to supplicate to and neither seek nor beseech the help from other than Allah		
R	Rela	tion in argument 1 from analogy. Similarity between the infidels' practices in the pre-Islamic era and the	ose of	
Cela	$\frac{\widetilde{\alpha}}{\omega}$ modern-day Sufis and Shiites.			
	🖞 Symptomatic relation is that Qur'anic verses are the words of Allah that, according to Muslims' belief, are inherent			
7.0	^{iz} true			

Figure 7.3 Arguments structures in argumentation threads 1 and 2 in Ibn Baz

Ibn Baz uses two types of argumentation schemes (analogy and sign) warranting his argument from analogy by the comparison he draws between some beliefs of some modernday Sufis and Shiites and the beliefs of the Arab infidels of the pre-Islamic era as well as supporting the same argument symptomatically (using Qur'anic verses). In his exposition of the meaning of worship, Ibn Baz utilizes Qur'anic verses to lend support to his argument. Now let us examine Ibn Baz's argument from analogy that seems to be faulty.

7.4.1.1 Critical evaluation of argumentation thread 1 in Ibn Baz

The Qur'anic verses utilized by Ibn Baz in his argumentation scheme by sign are all related to the arguments they intend to support. However this argumentation scheme by analogy is problematic. The argumentation from analogy (in argumentation thread 1 – true belief and worship) could be evaluated using some critical questions adopted from Walton (1995, p. 142) and cited in Walton et. al. (2008, pp. 309-346). These questions are as follows:

- 1. Are there differences between C1 and C2 that would tend to undermine the force of the similarity cited?
- 2. Is A true (false) in C1?

3. Is there some other case C3 that is also similar to C1, but in which A is false (true)?⁸⁶ It could be argued that Ibn Baz's argument from analogy regarding the practices of the pre-Islam infidels and modern Sufis and Shiites undermines the force of the similarity between the two types of the social actors - modern Sufis/Shiites and the pre-Islam infidels. First, regardless of the practices of both types of faith communities (Sufis and Shiites) that might be doubtful from the Salafist perspective⁸⁷, the Infidels rejected the message of the Prophet of Islam whereas the Sufis acquiesce to it and all other Islamic practices and rituals as well as accepting that there is 'no God but Allah and Mohammed is His messenger'. This phrase is the first pillar of Islam and anyone believes in it thus becomes a Muslim. Second, the Arab atheists in the pre-Islamic era used to worship idols whereas modern Sufis (as well as other Muslims) reject such atheist practices. Ibn Baz here could be guilty of basing his argument on a fallacious faulty analogy (Walton et. al. 2008, p. 51). However, it is noteworthy here that it is only the aspect of analogy that could be fallacious in argumentation thread 1 whereas the rest of the argument Ibn Baz uses about singling out Allah in the forms of worship 1 (about the true belief and worship) conforms with the Qur'anic verses cited in Figure 7.1 above.

Now let us move on to the expanded version of line 3 (Concordance D1 in Appendix D) in order to see how causality and symptomatic schemes are also used by Ibn Baz in his refutation of Arab nationalism.

^{86.} C means Case and A means Base premise.

^{87.} These practices may include beseeching help from dead pious individuals at their graves and circumambulation around their graves.

7.4.2 Arab nationalism: Rejection of the Sharia

In what follows Ibn Baz tries to establish a connection between the rejection of Sharia and the Arab nationalistic movement utilizing two forms of argumentation schemes (causality and sign). Figure 7.4 below contains Ibn Baz's arguments from both causality and sign that he uses to legitimate his stance towards Arab nationalism.

Figure 7.4 Argumentation thread 3 (Arab nationalism)

الوجه الرابع: من الوجوه الدالة على بطلان الدعوة إلى القومية العربية أن يقال : إن الدعوة إليها و التكتل حول رايتها يفضى بالمجتمع و لابد إلى رفض حكم القرآن؛ لأن التوميين غير المسلمين لن يرضوا تحكيم القرآن ، فيوجب ذلك لزعماء القومية أن يتخذوا أحكاما وضعية تخالف حكم القرآن ، حتى يستَوي مجتمع التومية في تلك الأحكام ، **و قد صرح الكثير منهم بذلك كما سلف** ، وهذا هو الفساد المطلع ، و الكفر المستبين و الردة السافرة ، **كما قال تعالى** : (فلا و رَبَّكَ لا لِؤُمِنُونَ حَتَى يُحَكِّمُوكَ فيما تَسَجَر بَيْنَهُمْ تَم لا يَجِدُوا في أَنْفُسِهِمْ حَرَجًا مِمًا قَحَدَيْتَ و يُسَلِّمُوا تُسْلِيمًا) في قال تعالى : (أَفَحُتُهُ الْجَاهِلِيَّةُ تِبُنُونَ و من أحسن من الله حُكْمًا لِقَرْم يُوفِئُونَ) و قال تعالى : (و من لم يُحُكُدُ بما أنزل الله فأُولَئِكَ هم الْكافِرُونَ) و قال تعالى : (و من لم يَحْكُم بما أنزل الله فأُولَئِكَ هم الطَّالِمُونَ) و قال تعالى : (و من لم يَحْكُمُ بما أنزل الله فَأُولَئِكَ هم الْفَاسِقُونَ) . The fourth viewpoint. One of the viewpoints that also indicate the rejection of the nationalists' call is that it could be said that calling to it (Arab nationalism) and gathering people under its banner will definitely lead [Muslim] society to reject arbitration to Qur'an (Sharia laws), because the [Arab] nationalists who are not Muslims will not be satisfied with the Sharia laws. Hence, [pan Arab] nationalism leaders will take man-made laws that contradict Sharia in order to make all people in their community equal. Verily a lot of them [leaders of Arab national parties] declared that they will rule according to man-made laws as indicated earlier. Doing so is a tremendous corruption, obvious infidelity and blatant apostasy, as the Exalted said: {But no, by your Lord, they can have no Faith, until they make you (O Muhammad) judge in all disputes between them, and find in themselves no resistance against your decisions, and accept (them) with full submission} and the Exalted said: {Do they then seek the judgement of (the days of) Ignorance? And who is better in judgement than Allah for a people who have firm Faith} and the Exalted said: {And whosoever does not judge by what Allah has revealed, such are the Kafirun (i.e. disbelievers - of a lesser degree as they do not act on Allah's Laws)} and the Exalted said: {And whosoever does not judge by that which Allah has revealed, such are the Zalimun (polytheists and wrong-doers - of a lesser degree)} and the Exalted said: { And whosoever does not judge by what Allah has revealed (then) such (people) are the Fasiqun [the rebellious i.e. disobedient (of a lesser degree)] to Allah}

Ibn Baz combines both causality and sign schemes in his refutation of Arab nationalism. According to Ibn Baz, if nationalist leaders rise to power, then rejection of the Sharia laws will occur. For this argument to be effective, the reader needs to understand and accept that the rejection of Sharia laws would have negative consequences. Yet Ibn Baz does not rely solely on argumentation scheme from causality; he, from a Salafist perspective, depends largely on authoritative Qur'anic verses that obligate Muslims to refer to Sharia Laws. He uses the clusters 'كما قال taulou' (kama qala ta'āla; as the Exalted said) (one time) and ' تعالى (wa qala ta'āla; and the Exalted said) (4 times) to introduce his arguments, using argument by sign.

Ibn Baz claims that adopting Arab nationalism will eventually lead (Muslim/Arab and more specifically Saudi) society to reject Sharia. He explains that by predicting that the non-Muslim Arab nationalists will not approve of Sharia laws, which could consequently force Arab nationalist leaders in Arab and Muslim countries to reject Sharia and adopt man-made laws. It is worth noting here that non-Muslim Arabs are still a minority in the Arab countries and none of them was/is a head/president of any of the Arab states or in a position to force Muslim societies to adopt any type of laws. Additionally, Ibn Baz introduces his argument from causality with a hypothetical phrase that contains the passive form of the verb say, ' أن ' *('an yoqāl*; that it could be said), assuming that others may share the same views with him about Arab nationalism and their aim to reject the Sharia. He introduces the argument from causality to begin another argument (from sign) using the warrant 'Verily many of them [leaders of Arab national parties] declared that they will rule according to man-made laws' to lend support to his claim about the Arab nationalists' rejection of the Sharia. Reconstructing these arguments gives more insights about the argumentation schemes they appear therein.

	Arguments structures in Ibn Baz (Argumentation thread 3)	Schemes
Premises	Adoption of Arab nationalism leads to adoption of man-made laws and rejection of Sharia/ Adoption of man-made laws is a tremendous corruption, obvious infidelity and blatant apostasy	
Warrants	Non-Muslim [Arab] nationalists will not be satisfied with the Sharia laws/ A lot of Arab nationalist leaders declared that they will take man-made laws instead of the Sharia/ Qur'anic verses introduced by 'as the Exalted said' and 'and the Exalted said'	Causality / Sign
Conclusions	(Unstated conclusion) Muslims (Saudis) should reject Arab nationalism	n
Relations	Symptomatic relation: Declarations of Arab nationalist leaders are highest authority in Arab Nationalist lations parties/ Qur'anic verses are the words of Allah that, according to Muslims' belief, are inherently true Causal relation in argument: Adopting Arab nationalism leads to the rejection of Sharia	

Figure 7.5 Reconstruction of argumentation thread 3 in Ibn Baz

In argumentation thread 3, Ibn Baz uses two arguments from sign and an argument from causality in defence of his negative standpoint towards Arab nationalism. His argument from causality is inserted between two arguments from sign which Ibn Baz uses to support his claim that Arab nationalism is a tremendous corruption, obvious infidelity and blatant apostasy. Ibn Baz uses two types of authorities as reliable sources: the alleged declarations of Arab national leaders and Qur'anic verses. Thus, according to Ibn Baz, adoption of Arab nationalism leads to the rejection of Sharia which, according to the Qur'anic verses such as those in Figure 7.4 is infidelity and apostasy. However these arguments need further examination utilizing some critical questions which is the aim of the next subsection.

7.4.2.1 Critical evaluation of argumentation thread 3 in Ibn Baz

As I have indicated in Section 7.4.2 above, the argument from causality (in argumentation thread 3) starts with the hypothetical phrase 'أن يقال' (*'an yoqāl*; that it could be said). Ibn Baz uses this phrase via which he introduces his argument from causality claiming that adoption of Arab nationalism eventually leads to the rejection of Sharia laws. He explains his claim by predicting that non-Muslim Arab nationalists will not approve of Sharia laws, which will force Arab nationalist leaders to adopt man-made laws rather than Sharia. In order to provide more critical analysis of Ibn Baz's argument from causality, I shall use the following critical questions:

- 1. Is the evidence cited strong enough to warrant the causal generalization?
- 2. Are there other causal factors that could interfere with the production of the effect in the given case? (Walton 1995, p. 142) and cited in Walton et. al. (2008, pp. 309-346).

The causal claim that Ibn Baz introduces is as follows: 'because the [Arab] nationalists who are not Muslims will not be satisfied with the Sharia laws'. He concludes that '[pan Arab] nationalism leaders will take man-made laws that contradict Sharia in order to make all people in their community equal'. However, adoption of man-made laws in the Arab countries is not due to Arab nationalist movement alone, but due to the secular movements as well. Additionally, Ibn Baz's causal claim is superficial. Adoption of man-made laws in the Arab countries has a long and complex history. It could be as a result of 'colonialism'.⁸⁸ In Egypt, for example, taking man-made laws started with French occupation of Egypt in 1798 (Alsinaidi, 1986). Man-made laws were further implemented in Egypt during the British occupation till the Egyptian revolution in 1952 (Ibid. 1986). Thus, the causal relation cited by Ibn Baz is not strong enough to warrant that adoption of Arab nationalism leads to the rejection of Sharia. Ibn Baz could be accused of using a slippery slope fallacy (Walton et. al, p. 186), i.e. Ibn Baz uses Arab Nationalists having power in the Arab countries as a slippery slope to the end of Sharia.

Again Ibn Baz continues to symptomatically lend support to his causal claim (mentioned earlier) by quoting Arab nationalist leaders: 'verily many of them [leaders of Arab national parties] declared that they will rule according to man-made laws as indicated earlier.' This is an argumentation by sign i.e. utilizing the declarations of the leaders of Arab National parties as an authentic and reliable source. However, Ibn Baz neither names any of these leaders nor mentions any authentic source of his quote. Instead he collectively aggregates Arab nationalist leaders utilising the pre-modifying phrase 'الكثير منهم' ('alkatīr minhom; a lot of them). The questions that are intended to critically evaluate the symptomatic argumentation scheme are introduced in Walton (1995, p. 142) and cited in Walton et. al. (2008, pp. 309-346):

- 1. What is the strength of correlation⁸⁹ of the sign with the event signified?
- 2. Are there other events that would more reliably account for the sign?

Aggregation of social actors (in our case Arab nationalist leaders) serves the aim of 'regulating practice and manufacturing consensus opinion' (van Leeuwen, p. 37), and 'endows social actors with a kind of impersonal authority, a sense of unseen, yet powerfully

^{88.} The concept of 'colonialism' is a euphemism as it conceals the atrocities that were committed during the military occupations of the third world.

^{89.} Correlation is not intended to indicate statistical correlation in this context.

felt coercive force' (Ibid., 2008, p. 37). But such aggregation mystifies the Arab nationalist leaders whom Ibn Baz quotes. Such aggregation undermines the strength of using them as a reliable source. Additionally Ibn Baz at the end of his quote claims that he mentions such quotes earlier but reading the relevant text he does not mention any of them and/or their quotes throughout the whole text. Thus Ibn Baz could be accused of fallacious appeal to authority (Reisigl and Wodak, 2001, p. 77).

7.4.3 Particles of faith: Sacred Books of Judaism and Christianity

Argument 4 in Figure 7.4 shows Ibn Baz's stance towards the Scriptures. Utilizing the symptomatic scheme, Ibn Baz introduces belief in Sacred texts of other Abrahamic religions:

Figure 7.6 Argumentation thread 4 (faith in Sacred Books)

و هكنا الإيمان بالكتب بعب الإيمان إجمالاً بأن الله سبحاته أنزل كتباً على أنبيانه و رسله ، لدان حقه و الدعوة إليه ، كما قال تعالى : (لقُدْ أَرْسَلْنَا رَسُلْنَا بِالْيَنَاتِ و أنزلنا مَعَهُمُ الكتاب و المِيزَانَ لِنَقُومَ الناس بَلْشَبْلَ الآية ، و قال تعالى : (كَانَ الناس أَمَّةُ و احَدَةُ فَبَمَتَ اللهُ النَّبِيِّنَ مُنتَّرِينَ و مُنْزِرِينَ و أنزل منهُمُ الكتاب بِالْحَقَّ لِنِحْمَ بَيْنَ النَّاسِ فيما اخْتَلُوا فيه) الآيه . و الذعوة إليه ، كما قال بما سمى الله منها كالتوراة و الإنجيل و الزبور و القرآن
and in the same manner is the belief in Scriptures. It is obligatory (on Muslims) to generally believe
that Allah, the Glorified, sent down Books to His prophets and messengers to state His rights (of
true worship) and to call people to them, as the exalted said { Indeed We have sent Our Messengers
with clear proofs, and revealed with them the Scripture and the Balance (justice) that mankind may keep
up justice till the end of the Qur'anic verse, and the Exalted said: {Mankind were one community and
Allah sent Prophets with glad tidings and warnings, and with them He sent down the Scripture in truth to
judge between people in matters wherein they differed till the end of the Qur'anic verse. We should
specifically believe in those Books that Allah named (in the Qur'an) such Torah, Gospel, Psalms
(the Holy Book of Dawud (David)) and the Qur'an

In the above concordance line, Ibn Baz continues to project and explicate his stance towards believing in the Scripture utilizing Qur'anic verses that are the words of Allah, according to Muslims' belief (particles of faith), and thus seen as inherently true. Let us first reconstruct the arguments in argumentation thread 4.

Figure 7.7 Reconstruction of Ibn Baz's arguments in argumentation thread 4

	Arguments structures in Ibn Baz (argumentation thread 4)	Schemes
Premises	Scriptures (Torah, Gospel, Psalms (the Holy Book of Dawud (David)) and the Qur'an) were sent down from Allah to his prophets and messengers/ The Scriptures contain Allah's rights of (true worship)	Sign
	Qur'anic verses introduced by 'as the Exalted said' and 'and the Exalted said'/ Allah named those Holy Books in the Qur'an	
Conclusions	It is obligatory (on Muslims) to generally believe that Allah, the Glorified, sent down Books to His prophets and messengers/ We should specifically believe in those Books that Allah has named in Qur'an): Torah, Gospel, Psalms (the Holy Book of Dawud (David)) and the Qur'an	
Relations	Symptomatic relation: Qur'anic verses are the words of Allah that, according to Muslims' belief, are inherently true.	

The most prominent scheme which is, from a Salafist ideological perspective, used by Ibn Baz is the symptomatic scheme utilizing the authority vested in Qur'an as the words of Allah which, according to Muslims, are inherently true. These argumentation schemes are utilized by Ibn Baz to legitimate his stances towards the semantic foci and social actor representations that these schemes appear in.

To sum up, Ibn Baz's arguments are mainly theological. Despite the fact that Ibn Baz uses all the three types of argumentation schemes in Van Eemeren and Grootendorst (2004), as I have demonstrated above, the symptomatic scheme is most frequently used by Ibn Baz in his arguments. Ibn Baz's use of symptomatic schemes signifies his great reliance on, from a Salafist perspective, powerful and legitimate tools in his arguments, viz., Qur'anic verses which are seen by Muslims as inherently true. The argument threads are shown utilizing the cluster 'و قال *iad aia*; and the Exalted said) that indicate argumentation by sign. However, Ibn Baz uses schemes by analogy and causality. He extensively uses arguments from sign that are based on the power invested in Qur'anic verses: according to Muslims they are inherently true as they are the words of Allah. Thus his use of such arguments is a means of legitimating his stances about his representation of social actors and semantic foci. Ibn Baz's argument from analogy and causality can be shown to be fallacious as I have indicated above. I shall now turn to explicate the argumentation schemes in Alhawali.

7.5 Ideological legitimation (2): Argumentation schemes in Alhawali

I have indicated above (Section 7.2.2), Alhawali uses argumentation indicators that bear meanings of causality and reveal Alhawali's reliance on quoting authority. These argumentative keywords are 'لذلك' (*lidālik*; therefore/because of that/based on that/(to/for) that) and 'لنتيجة' (*natīğah*; result) that indicate causality and ('لنلك'' ('*albaḥitīn*; the researchers –acc.) and 'لتقول' (*she/it/they* fem. pl.) say(s)) that indicate authority. The keyword '*lidālik*; therefore/because of that/based on that/(to/for) that) is not always used by Alhawali to indicate argumentation. Of the 136 occurrences of this keyword, Alhawali uses it 54 times indicating 'to/for that' which does not bear any argumentative meaning (see footnote 83). Additionally, one of the aims of the analysis in this chapter (and this thesis as well) is to compare the use of the argumentative keywords in both corpora that comprise the research data. Thus it will not be included in the analysis.

As I have indicate earlier (Section 7.2.2 and Chapter 4), the reporting keyword 'imes i a v i d v i i d v i i d v i i d v i i d v i d
conceives to be the western aggressive intentions towards the 'Muslim world'. In what follows I shall present the argumentation threads that are indicated by the keyword ' $iaq\bar{u}l$; (she/it/they fem. pl,) say/says) in addition to the other word forms of the word 'SAY' that appear therein. Due to Alhawali's verbose and lengthy exposition as well as the complex nature of his arguments I shall first outline his arguments (starting with premises) for each argumentative thread in Concordance lines D2 (Appendix D) then I shall evaluate every argument immediately after I outline them.

7.5.1 Western woman and social life: A Salafist perspective

In the first argumentative thread, Alhawali tries to negatively represent social life in the West. He puts forward a set of arguments starting his argument thread with the following argument:

Figure 7.8 Argument 1 in argumentation thread 1 (Western women and social life)

ولكن المرأة الغربية المنكوية إذ تحاول الرجوع إلى البيت و طبئة الأمومة لا تسلّطيه، لأن المشكلة اتسع نطاقها إلى درجة تتحد معها العودة الحقيقية، و لى حصل شىء من ذلك فإن المجتمع الذي تعود الانحلال و الاختلاط يستنكره و بأباه، بل أصبح الزواج رغم هشاشته مصدر إزعاج للقائمين على تلك المجتمعات فقد طلحت الصحافة الغربية يوما بخبر **يقول**: "الزعجت السلطات التأبيئية فى اسكتلانا بسبب موجة الزواج التى تحصف بالمدرسات، فقد تبين أنه من خلال عام (1960م) عينت 1563 مدرسة فى سكتلنه، و فى نهاية العام الدراسى تركت ألف منهن الوطنية للزواج ، و <u>قالت</u> النظام المدرسى".

But the afflicted western woman, who tries to settle down in her home and revert to being a mother, is unable to do so. The case is so, because the problem has reached the limits that make it impossible to return to (her real role in life, i.e. being a mother). If something like that (going back to settle in a house and perform her real role as a mother) happens, the society that has accustomed to and practiced moral debasement and the mixing between men and women, would refuse it. Additionally, marriage, though not based on solid grounds, became a cause of discomfort to people in authority in those (western) societies. The western press, in one day, published a piece of news that <u>savs</u>: 'the Scottish educational authorities are annoyed because of the wave of marriages that blow away female teachers (from teaching). It has been known that during the year 1960 there were 1563 female teachers appointed in schools, and 1000 of them left their jobs to get married by the end of the Academic year. The authorities said that marriage threatens the school system'.

Debasing the values of the western societies, Alhawali claims that western society and its educational system oppress women. He starts his premise by constructing western woman as 'afflicted'. The premise stems from an unstated assumption which is a preconceived ideological Salafist perception of woman's role in life, i.e., a woman's role should only involve household chores and bringing up children. Alhawali denounces western societies warranting his argument with what the western press *says* about western societies which is a case of argumentation scheme by sign. Alhawali presents the western press as an authority, i.e. western press publishing news about western societies, thus seen by Alhawali as a reliable

source (although it is perhaps ironic that he uncritically uses western sources in order to be critical of the West). Alhawali aggregates a set of unnamed sources using the plural form 'newspapers' which, according to van Leeuwen (2008, p. 37) 'is used to regulate practice and manufacture consensus opinion'. In the above excerpt (Figure 7.8) Alhawali writes about Scottish educational authorities whom he projects as being against marriage because married women stop being teachers. Alhawali uses this as an example to show the degeneracy of western culture as being against marriage.

Alhawali uses argument 1 to establish a situation which he conceives of as being oppressive of women in order to go further and present another argument (coordinative argumentation) based on the initial unstated premise in Figure 7.8. Thus after introducing argument 1 above, he uses a second argument to further negatively represent the West as forcing women to seek revenge:

Figure 7.9 Argument 2 in argumentation thread 1 (Western women and social life)

و حيل نلك إظلم المرأة الغربية]: ماذا في رسع المرأة أن تنطى؟ بأي شيء تواجه المجتمع النك الذي يلهب ظهرها بالسياط، و يتطع عليها طريق العودة إلى قطرتها، ليس هذاك إلا أحد سبيلين: إما الانتقام من هذا المجتمع الظلم بترويعه و تحكير صفوه، كما جاء في التقرير الذي يلمب طهرها بالسياط، و يتطع عليها طريق العودة إلى قطرتها، ليس خلل عام (1960م) نحو (2014) سرقة.[...]، و الغريب أن (60%) من هذه السرقات ارتكبتها نساء جاوزن من البلوغ، و (30%) ارتكبها نكر أقل من السابعة عشرة و تقول الإحصائيات: إن كل السارقات من النساء لم يكن في حاجة للمل، نعم إنها ليست الحاجة للمال، و لكنها الزغية من نفسه بالانتخام من نفسها بالانتخار و تقول الإحصائيات: إن كل السارقات من النساء لم يكن في حاجة للمل، نعم إنها ليست الحاجة للمال، و لكنها الزغية في تعريع السخط و أما الانتخام من نفسها بالانتخار كما قطلت الممثلة الشهيرة مارلين مونرو، التي كثبت قليل التحارها نعم بنها ليست الحاجة للمال، و لكنها الزغية في تعريع السخط و أما الانتخام من نفسها بالانتخار كما قطلت الممثلة الشهيرة مارلين مونرو، التي كثبت قليل المحارها للمان، عم إنها ليست العاجة عمرة، و هذه الارض... لم أستطع أن أكون أمار.. إلى كلمان تصريحة لبنا التحارية السرائية على في في الموزية من كل من يخدعك بالأصواء... إلى أمر الحياة المائلية الممثل عان أكون أمار.. إلى أمرأة أفضل البيت... الحياة العائلية التريفة على كل شيء... إن سعادة الم أماة التعيتية في الحيزة العائلية المرأة بل الإنسانية، و تقول في النهاية، لتد ظلمني كل الناس...، و أن العمل في السينما يجعل من المرأة سلمة رخيصة تنفية مهما نالت من المجد و التبريرة الوائفة".

And to reject that [western woman being repressed]: what could a woman do? How could she face such unjust society that lashes her back fiercely [a metaphor meaning that society is oppressing her] and prevents her from going back to her original nature [marriage and fostering children]. There is no means to [face such society] but to do either of the following: To take revenge against this unjust society by terrorizing it and causing disturbances as written in <u>a report published in newspapers</u> that shoplifting from grand stores during 1960 reached around 32194 cases, [...]. It is so strange that 60% of shoplifting was committed by adult women whereas 30% of such crimes were committed by males under the age of 17. The statistics <u>says</u>: All female shoplifters weren't in need of money, but they sought revenge against society. Or they seek to inflict punishment on themselves by committing suicide as the famous actress Marilyn Monroe did. She wrote a piece of advice to her fellow women before committing suicide. She <u>says</u>: beware of glory... beware all those who try to pull your leg by the glamour of fame... I am the most miserable women on earth... I could not be a mother... I am a woman that prefers staying in my home... and the honourable family life to everything else. Truly, woman's real happiness is in the purified and honourable family life. Indeed, such family life is not only the symbol of woman's happiness but indeed it is the symbol of happiness to all humanity. She finally <u>says</u>: 'All people dealt with me unjustly..., and working in movie industry makes the woman a cheap and worthless commodity despite her glory and unreal fame'.

Warranting his arguments with news reports, statistics and a piece of advice that is attributed to the actress Marilyn Monroe, Alhawali utilizes the verbs 'written' and 'says' to introduce such warrants to conclude that western women are oppressed. Alhawali also continues to develop his argumentation thread to denounce the West. He uses some more statistics to warrant his arguments:

Figure 7.10 A report and statistics used by Alhawali as warrants (argument 3: Western women and social life)

وليس غريباً أن تؤكد الإحصائيات العالمية أن نسبة محاولات الانتحار عند النساء أكثر منها عند الرجال، **يقول** تقرير كتبه أحد الأطباء الاجتماعيين في فيينا؛ و قد لوحظ أن النساء أكثر محاولة من الرجال، فنى عام (1948م) كان عند المحاولات فى النساء (381) و هذا يوافق (5.86%) من المجموع، و فى عام (1956م) كان العدد (590) أي بنسبه (56.67%)، و فى عام (1959) كانت النسبة (55.92%)، كما لوحظ أن نسبة المحاولات فى الفتيان و القتيات الذين تتراوح أعمار هم بين (14) عاماً و (20) عاماً ترتفع باستمرار، فحد الفتيان كانت النسبة (1948م)- (5.6%)، و فى عام (1956م)، و فى عام (1959م)، و أما عند الفتيات فالتصاعد مخيف، ففى عام (1959م) كانت النسبة فى عام (1948م)- (5.6%)، و فى عام (1956م)، و فى عام (1959م)، و أما عند الفتيات فالتصاعد مخيف، ففى عام (1959م) حاولت (50) فتاة الانتحار، و هذا يشكل نسبة (7.6%) من مجموع محاولات الأنتحار فى تلك العام، و فى عام أما عند الفتيات فالتصاعد مخيف، ففى عام (1948م) حاولت (50) فتاة الانتحار، و هذا يشكل نسبة (7.6%) من مجموع محاولات الانتحار فى تلك العام، و فى عام (1956م) حاولت (89) فتاة الانتحار، و هذا يتكل نسبة (1955هم)، و فى عام (1959م)، و هن عام (1959م)، و فى عام (1950م (1956م) حاولت (89) فتاة الانتحار، و هذا يتمكن نسبة (25.8%) و فى عام (1959م) حاولت (150) فتاة الانتحار، و هذا يعلى تسعة أيام توجد ست محاولات انتحار، و هذا يتمكن نسبة (1958م) حاولت (150) فتاة الانتحار، و هذا يعنى أن كل

It is not very unfamiliar that international statistics confirms that attempts to commit suicide increased among women as compared to men. A report written by one of the 'social' physicians in Vienna <u>says</u>: and it has been noticed that women attempt to commit suicide more than men. In 1948 there were 381 suicidal attempts committed by women which constitute 55.61% of the total attempts. In 1956 the number increased to 590 attempts that mean 56.73% and in 1959 the percentage was 55.92%. It is also noticed that the percentage of suicidal attempts is increasingly growing among young males and females whose ages range from 14 to 20. The percentage of the cases of committing suicide increased among young males as it was 6.5% in 1956 and became 6.81% in 1959. However the rates of committing suicide among young females were increasing terribly. In 1948, 89 young females tried to end their lives which constitute 8.55% of the total number of suicidal attempts. Thus we can conclude that every 9 days there were 6 attempts to commit suicide where 4 of these attempts are made by young girls whereas two are made by young boys.

Last but not the least, Alhawali uses another warrant to lend support to his claims about women's status in the West based on statistics that were attributed to an unnamed 'social' physician, which again could be seen as a fallacious appeal to authority (detailed below).⁹⁰ Alhawali thus attributes some information supported with statistics (using the topos of numbers) to writing from unnamed individuals about people attempting to commit suicide.

Alhawali's projection of women's status in the West is one form of his appeal to moral justification as a kind of moral legitimation of his (Salafist) stance towards western social life. This type of argumentation permeates Alhawali's Salafist writings as I have shown in Section 7.5 above. Alhawali uses all the above three arguments to lend support to one standpoint which is about what he conceives of as the unjust western society that

^{90.} The social actor 'أحد الأطباء الاجتماعين' ('aħad 'al'ațibba' 'a'ijtmā 'iyyīn; one of the social physicians) is a strange nomination that appears in Alhawali's text which is originally taken from another writer (Mustafa Alsibaie 1915-1964 the founder of Syrian branch of Muslim Brotherhood). I tried to find the quote in the original text (Alsibaie, 1999), but I did not find the whole quote. This might be due to that edition 7 of the book is the only available one whereas Alhawali quoted it from edition 4, as indicted in his list of references, which is not available (the first edition was published in 1984). However, the appellation 'social physician' could possibly be a misnomer that refers to family physician or social worker which again is unnamed rendering his Alhawali's argument into fallacious appeal to authority (see below).

oppresses women - coordinative argumentation (van Eemeren et.al, 2002). These arguments

could be reconstructed as follows:

Figure 7.11 Reconstruction of Alhawali's arguments about the values of western societies (thread 1	es (thread 1)
--	---------------

r		Bannenis debut nie valdes of western sectories (inicia 1)			
	N	Arguments structures in Alhawali (thread 1)	Schemes		
P		Woman's primary role is limited to house chores and bringing up children (unstated premise)/ Western women cannot go			
Premises	1	back to practice their prime role/ Western educational system opposes marriage (social life in the West is decayed)	-		
	2	Western women need to do something to show their rejection of oppression			
s	3	It is not unfamiliar that western women attempt to commit suicide	1		
		Because western society refuses that women go back to practice their primary role in life (house chores and bringing up	_		
wa	1	children)/ The western press 'says'	sign /		
warrants		A report published in newspapers introduced by the verb 'says'/ Statistics (numbers and percentages)/ Social oppression			
nts	2	of women in the west causes women to commit suicide and shoplifting	causality		
	3	A report published by a 'social' physician in Vienna introduced by the verb 'says'/ Statistics (numbers and percentages)			
co		Western Society is accustomed to and practiced moral debasement and the mixing between men and women Western	Y Y		
nclu	1	women are oppressed and deprived from performing their prime role			
conclusions		Women take revenge against the western unjust society by terrorizing it and causing disturbances (shoplifting and]		
SU	2	committing suicide)			
	3	Number of women who tend to commit suicidal is double the number of men in the West			
		ptomatic relationships: 1 & 2 Western press knows better about western society, hence it is inherently reliable source. 2- S			
гel	and Marilyn Monroe 'say'. Statistics gives scientific precision and Marilyn is a famous actress who before she ended her life adviced				
relatio	women to stay at home and not to mix with men)/ 3- 'Social' physician's report in Vienna/ Statistics. Causal relationships: 1- Moral				
suc	debasement and mixing between men and women in the west cause women not to go back to practice their primary roles. 2- West				
	soci	ety oppresses women which causes them to shoplift and commit suicide.			
· · · · ·					

As appears in Figure 7.11, Alhawali puts his arguments in two argumentation schemes (by sign and causality) to represent western social life. For evaluation of these two schemes, I

shall subject them to some critical questions.

7.5.1.1 Critical evaluation of argumentation thread 1 in Alhawali

As indicated in the previous section, Alhawali utilizes two schemes: argumentation schemes from sign and by causality. In order to evaluate these schemes I use some critical questions adopted from Walton (1995, p. 142) and cited in Walton et. al. (2008, pp. 309-346). Let us start with argumentation from sign (symptomatic argumentation). In order to evaluate the argumentation scheme by sign in Alhawali, I use the questions that I have introduced earlier:

- 1. What is the strength of correlation of the sign with the event signified?
- 2. Are there other events that would more reliably account for the sign?

Answering the above questions reveals the following: Alhawali, in arguments 1 and 2 about western societies, uses the plurals 'الصحافة الغربية' ('alşaḥafah 'alġarbiyyah; western

newspapers) in argument 1 and 'lace' 'lsohof; newspapers) in argument 2 (aggregation of 'newspapers') as reliable sources of information about the West, but no specific source is mentioned though, making the report difficult to verify. Aggregation of 'newspapers' is intended to 'regulate practice and manufacture consensus opinion' (2008, p. 37). The mere aggregation of 'newspapers' is dubious as it anonymously presents western newspapers as a whole that covers one event (female teachers in Scottish schools apparently left their jobs to have children) or publishes a report about shoplifting committed by western women. In both cases, Alhawali aims to make generalizations about western society as a whole. It is worth noting that the example given to support argument 1 (western women cannot go back to their role of raising children) is counter-productive as the women in the example about Scottish schools are all apparently allowed to leave their jobs. Additionally, Alhawali, in argument 2, quotes the film actress Marilyn Monroe who committed suicide. She was an actress rather than an expert on gender relations, suggesting that there is a fallacious appeal to authority here – Monroe's fame is used to legitimate the quote, rather than any research-based qualifications to justify her being able to speak on behalf of women in general.

Also, it is questionable that Monroe actually said the words that Alhawali attributes to her.⁹¹ And in selecting a single quote from one actress, Alhawali is using a cherry-picking fallacy. He could, for example, have used other quotes attributed to Monroe (which do not fit so well with Salafi ideology) such as "I have too many fantasies to be a housewife", "Sex is a part of nature, I go along with nature" and "If I'd observed all the rules, I'd never have got anywhere."⁹² Additionally, he could have quoted from other types of women rather than cherry-picking one quote from one woman.

^{91.} Alhawali does not give the source of the Monroe quote, and when I tried to find it online, I only found it being used a few times by some Muslims who were using it in a similar way to Alhawali, in blogs and forums to claim that Western women were oppressed and unhappy. I could not find any reference to the quote in other web pages, although there are numerous websites about Monroe, many which include lists of quotes from her. This lack of verifiability of the quote suggests that this could be a fallacy of false attribution in addition to fallacious appeal to authority and cherry-picking.

^{92.} http://www.brainyquote.com/quotes/authors/m/marilyn_monroe.html

Similarly, in argument 3, Alhawali quotes a report attributed to an unnamed 'social' physician in Vienna. In addition to the fact that it is unusual to use the appellation 'social' physician, Alhawali quotes an unknown individual to draw conclusions about suicide rates among women in Vienna in the years 1948-1959. As Alhawali does not provide full references for the facts and figures he cites, it is difficult to verify their authenticity – this could be referred to as a false attribution fallacy, where an advocate appeals to an irrelevant, unqualified, unidentified, biased or fabricated source in support of an argument (see footnote 90 above). However, as well as quoting unqualified authorities such as Monroe, Alhawali relies on the topos of numbers claiming that 60% of shoplifting is carried out by women and the rate of suicidal attempts among western women was increasing (quoting a report attributed to an unnamed 'social' physician from Vienna in the years 1948, 1956, 1959).

Alhawali uses a causality argument here – claiming that the (supposedly) high rate of female shop-lifting is attributed to the oppression of women in western society. The questions that critically account for the argumentation scheme by causality are as follows:

- 1. Is the evidence cited strong enough to warrant the causal generalization?
- 2. Are there other causal factors that could interfere with the production of the effect in the given case?

There is no evidence given to demonstrate how the oppression causes shop-lifting – so, assuming the facts are true, this is a case of 'correlation not causation' or the post hoc ergo propter hoc fallacy (Reisigl and Wodak 2001, p. 73). Additionally, Alhawali does not quote figures about other sorts of crime so his focus on one type of crime (shop-lifting) is another example of the fallacy of incomplete evidence (cherry-picking).

To summarize, Alhawali, in argumentation thread 1, presents 3 arguments that constitute a conclusive defence of his standpoint vis-à-vis social life in the west (coordinative argumentation). I have demonstrated above that his arguments (in both schemes: by sign and

from causality) are fallacious. I also have pointed out that he relies on topos of numbers. The above three arguments of Alhawali could also be classified as appealing to the fallacy of hasty generalization or 'secundum quid' (Reisigl and Wodak, 2001, p. 73) as he generalises his conclusion about western social life based on an unrepresentative sample. Furthermore, Alhawali uses false attributions to legitimate his arguments by not providing full sources relating to the people he quotes. Let us now move to Alhawali's arguments about religious life and Christianity in the West.

7.5.2 The Church and the Salvation 'theory'

Unlike Ibn Baz who invites his Muslim audience to argue with the people of the Book (Christians and Jews) in a 'good manner' (see Chapter 6 Section 6.2.1.2), Alhawali, in a purely theological discussion of the Salvation theory in Christianity, denounces the Church. Utilizing an argumentation scheme by analogy, he makes a comparison between the Islamic conception of sins and the Church's concept of original sin (argumentation thread 2).

[…] فإن التصور الإسلامي يقرر ويؤكد حقيقة عظمى و قاعدة جليلة تصمنها قوله تعالى ((ألا أنزرُ وَإَنَرَةُ وَزَنُ أَخْرَى * وَ أَن لَئِينَ لِلإِنسَانِ إِلاَّ مَعَى)) [التج:38-39] فلا يؤاخذ الله تعلى أحاً بذلب غيره مهما كالت الصلة بينهما […]. إنه - حسب قاعدة العدل الرياني - لا يجوز أن يؤاخذ أحد غير أم بخطيئته، حتى و لو كان ابن التعطان الذي أغواء بالخطيئة - فضلاً عن أن يكون ابن الله- كما **تقول** الكنيسة تعالى الله عن ذلك علواً كبيراً أو أحداً من بني آدو. و بذلك خلا التصور الإسلامي من الأفكار و النظريات التي ابتدعتها الكنيسة باعتبارها من مستلزمات الخطيئة.[…]. و كان علماء المسلمين أسيقي من فلاسفة عصر التوير و الت أحدهم: (… فلسبوا الإله الذي – سبحانه - إلى ما يألف أسفط الناس و أظهر أن يقطه بعملوكه و عبده […] و نسبوه إلى أقبح التاريخي في هذا المصمار ، **يقول** أحدهم: (… فلسبوا الإله الذي – سبحانه - إلى ما يألف أسفط الناس و أظهر أن يقطه بعملوكه و عبده […] و نسبوه إلى أقبح الظلم، حين أنيوا ف أولياءه في الجنب الذي الذي الذي الله على ما يألف أساما المسلمين أسب<u>عة</u> من فلاسفة عصر التنوير و التياع مدرسة الذي التربي في هي المصمار ، **يقول** أحدهم: (… فلسبوا الإله الذي – سبحانه - إلى ما يألف أسفط الناس و أظهر أن يقطه بعملوكه و عبده […] و نسبوه إلى أقبح الظلم، حيث زعموا أنه سجن أنبياءه و رسله و أولياءه في الجزء منين الصلة أليهم و نسبوه إلى علياة الساس و أظهر أن يقطه بعملوكه و عبده […] و نسبوه إلى أقبح الظلم. و معبودها و الهما بنا بنه بدن الحملية المان و حلى أله عله إلى الذاب بتمكينه أعاده من نفسه حتى قتلوه و …](…] (…] ا

[...] and the Islamic conception of (sins) states and confirms a noble reality and a splendid principle that is included in the following Qur'anic verse, Allah <u>said</u>: {That no burdened person (with sins) shall bear the burden (sins) of another, And that man can have nothing but what he does (good or bad)}. 53:38-39. Thus Allah, the Exalted, will not punish one with the guilt of another no matter how closely they are related [...]. Verily – according to the divine justice – it is not permissible to punish any individual because of the sin of Adam, even if this individual is the son of the Satan who persuaded Adam to commit the sin let alone the son of God –as the Church <u>says</u>. Glorified and Exalted be He! High above (the great falsehood that they say) – or any of the sons of Adam (humans). Thus Islamic conception [of sins] does not include the ideas and theories that were invented by the Church which are considered to be inevitable consequences of the (original) sin [...]. Muslim scholars (religious leaders) <u>preceded (had the precedence over)</u> the philosophers of the Enlightenment and the advocates of historical criticism in this respect [criticizing the Church] attributed to the true God – the Exalted – what the most morally depraved and insignificant individual could attribute to his own slave! [...] They also attributed the worst form of injustice to Him when they claimed that he imprisoned His prophets and messengers and His close pious worshippers in Hell-fire because of their father's sin. They attributed foolishness to Him as they claimed that He granted them salvation [from the sin] by enabling His enemies to kill and crucify Him [...]). Generally we do not know any nation that execrated their Lord whom they worship like this nation [Christian nation], as Omar [the second Caliph in Islam], may Allah be pleased with him, <u>said</u>: (Verily they execrated Allah with a vilification that no single human has ever attributed to Him] [...]

Figure 7.12 The Church and the original sin (argumentation thread 2)

The analogy in the above figure is between two different and opposing concepts of sin in Christianity and Islam. Alhawali introduces how each concept is characterised in each religion by the verb 'تقول' (taqūl; it (the Church) says) trying to lend support to his argument symptomatically. First, he introduces the Islamic concept of sins by a Qur'anic verse which is seen by Muslims as inherently true. He also utilizes the words of the Muslims' second Caliph (Omar 634 – 644) as another warrant to support his argumentative denigration of the Church. However, in order to introduce the standpoint of Church that regards the salvation theory, Alhawali does not refer to any Christian sacred texts. On the contrary Alhawali utilizes the verb 'says' to generally introduce the Church's standpoint. Additionally, he quotes one of the advocates of the school of historical criticism without naming him. In these two argumentation schemes (by sign and analogy), Alhawali's rejection of Salvation theory in Christianity is not brought up to establish a dialogue with the people of the Church. It is used here to denounce the Church which explains Alhawali's exclusion of Christian social actors, instead personalizing the Church, i.e. representing the people of the Church by spatialization (for more information about spatialization and Alhawali's representation of social actors see Sections 6.2.1.2 and 6.3). This suggests that Alhawali is not addressing 'the Other' in his writings but has another type of addressees in mind (his fellow Muslims in general and more specifically Saudis). This argumentative thread could be reconstructed as follows:

N Arguments structures in Alhawali (thread 1) schemes Islamic conception of sins reflects the divine justice and is free from the ideas and theories that were invented by the Premises Church Muslim scholars (religious leaders) rejected the Church's theories of the original sin before the advocates of 1 Sign/ analogy historical criticism did/ Christians say bad things about Allah Qur'anic verse the Church says Muslim scholars (religious leaders), the philosophers of the Enlightenment and the 1 Warrants advocates of historical criticism reject the original sin/ Omar, [the second Caliph in Islam] said The Church is wrong. Islamic concept of sins is right/ Muslim religious leaders are aware of these facts and better than Conclusions 1 the philosophers of the Enlightenment and the advocates of historical criticism Symptomatic relationships: Qur'anic verses that are the words of Allah which, according to Muslims, are inherently true/ Muslim religious leaders are people of true religious knowledge thus they are considered by Muslims to be reliable / Enlightenment philosophers and the advocates of historical criticism who reject the concept of Original sin belong to the western culture and know western Christian Relations culture better, thus they are reliable sources of knowledge in this respect / Omar (the second Caliph) is reliable religious authority and a source of reliable knowledge. Relationship for arguments from analogy: Using analogy to point differences between Islamic concept of

Figure 7.13 Reconstruction of Alhawali's argumentation of the original sin (Argumentation thread 2)

The construction of Alhawali's argument concerning the original sin in Christianity shows his major aim of denouncing the West. Alhawali also continues to use the symptomatic scheme in order to negatively represent the West seeking to support his arguments via appeal to 'studies' that he presents as an authority. Within the symptomatic argumentation scheme, Alhawali introduces an argument from analogy that is deployed to undermine the efforts of western Enlightenment philosophers and advocates of historical criticism.

sin and the original sin in Christianity/ Pointing similarity; Elites in both Cultures (Islam vs. Christianity) reject the Church's theory of the

original sin/ Pointing differences: Muslim religious leaders rejected original sin theory before western elites did.

Alhawali compares two types of social actors who lived in two different historical epochs; medieval Muslim religious leaders and modern Enlightenment philosophers and advocates of historical criticism. These social actors are seen as being similar in that both rejected the Church's theory of the Original sin. Alhawali uses the evaluative hyperbolic adjective 'أسبق' ('asbaq; had the precedence over) to attribute and record the priority of the Muslim religious leaders in criticizing the concept of the Original sin. The hyperbolic adjective 'أسبق' ('asbaq; had the precedence over) is neither used to introduce the criticism of the Original sin chronologically nor to value the contribution of Western philosophers. On the contrary, it is used to undermine the efforts of the Enlightenment philosophers and the advocates of historical criticism and to positively represent Muslim religious leaders. In so doing, Alhawali tries to legitimate his negative stance toward the west. Next, I shall include

more of Alhawali's arguments about the West and what Alhawali claims to be the onslaught on the Arabian Peninsula.

7.5.3 Political enmity: Invasion of the Gulf and the Arabian Peninsula

Alhawali tries to represent the 'Other' (the West) to his readers as wanting to destroy Islam and invade the 'Muslim World' as I have presented above and in Chapters 4 and 5. Again, considering statistics and some western newspaper reports as an authentic and reliable source, Alhawali includes, in his arguments, reference to studies to support his stance of what he believes to be the evil intention of the West (represented mainly in United States) towards Saudi Arabia and the Gulf region. Below I present an argumentation thread that is used by Alhawali to support his negative stance towards the United States.

Figure 7.14 United States Invades the Gulf and the Arabian Peninsula (argument 1 in Argumentation thread 3)

3- برامج لاحتلال الخليج والجزيرة
[...] و في ظل تلك الأوضاع المتقلبة و الإدارة الأمريكية الضعيفة ظهرت دعوات قوية في أمريكا تنقد ضعف بلادها و تنذر أمتها بالخطر المحدق إن لم تعد العدة كاملة لإحتلال الخليج. كما <u>صرح</u> بعضهم أو لحماية أمن الخليج كما يعللها الأكثرون. و نشير هذا إلى بعض المترجم المتداول في الأوساط العربية منها، العدة كاملة لإحتلال الخليج. كما <u>صرح</u> بعضهم أو لحماية أمن الخليج كما يعللها الأكثرون. و نشير هذا إلى بعض المترجم المتداول في الأوساط العربية منها، العدة كاملة لإحتلال الخليج. كما <u>صرح</u> بعضهم أو لحماية أمن الخليج كما يعللها الأكثرون. و نشير هذا إلى بعض المترجم المتداول في الأوساط العربية منها، [...] العدة كاملة لإحتلال الخليج. كما <u>صرح</u> بعضهم أو لحماية أمن الخليج كما يعللها الأكثرون. و نشير هذا إلى بعض المتركية المتداول في الأوساط العربية منها، [...] المتكرية العسكري في منابع النفط، الاحتمالات و الخطط " [...] المتكري في منابع النفط، الاحتمالات و الخطط " و...] المعمون الأربية منها، منها معلكم عسكرية المتراتيجية نشرتها مجلة فورتشون الأمريكية في مايو 1979م بطوان "التدخل العسكري في منابع النفط، الاحتمالات و الخطط " و [...] المعمونية المتدارة المربية منها، منها على عنها من الذلية منها معدة عنهما منها المعادية نابعاً من الخلوفات التائمة منذ زمن حول الأراضي هو خطر حقيقي". تم تقول الدراسة :- "لقد ما ورد فيها ص 4: "إن نزاعاً بين العراق و الكويت أو السعودية نابعاً من الخلافات التائمة منذ زمن حول الأراضي هو خطر حقيقي". تم تقول الدراسة :- "لقد أوضح كل من براون و بريجينسكي - مساعدا الرئيس كارتر لشؤون الأمن القومي - مؤخراً أن الولايات المتحدة ستنخذ خطوات من بينها استخدام القوات المعكرية المعردية.

3- Scenarios to invade the Gulf and the [Arabian] Peninsula

[...] and in the light of unstable political conditions [in the Middle East] and the weak American administration, some serious proposals appeared in America criticising the American weakness and warning the American nation of the dangers if they do not prepare their troops to invade and occupy the Gulf as some of them <u>stated</u> or to protect the Gulf as most of them <u>justified</u>. I present here some of the translated reports (into Arabic): [...] 1-<u>A military strategic study</u> published by 'Fortune' in May 1979 titled 'Military intervention of Oil Wells: Possibilities and plans'. In page 4 there appears an indication that 'there could be a conflict between Iraq and Kuwait or Saudi Arabia over the long lasting problematic territories which emerge to be a real threat'. Then the study <u>says</u>: 'Both Brown and Brzezinski - Assistants to the President (Jimmy Carter) for National Security Affairs – have recently declared that the United States will take measures including military intervention to protect its interests in Saudi Arabia'.

Alhawali puts forward the following premise: United States is planning to invade Saudi Arabia. He presents some extreme proposals that he claims American leaders are making using warrants that include statements attributed to some unidentified social actors and are introduced by the words 'stated' and 'justified' that both carry the meaning 'to say' (using the symptomatic argumentation scheme – explained further below). Additionally, he uses the verb 'says' to introduce another warrant; 'military strategic study'. Alhawali warrants his argument quoting two well known individuals (Brown and Brzezinski - Assistants to the United States President (Jimmy Carter 1977–1981) for National Security Affairs). Quoting such high rank officials in the Carter's administration, as people in the position to know the plans and policies of the United States, serves Alhawali's purpose to support his claim about the intentions of the United States to occupy oil wells⁹³ (see Section 7.5.3.1 below for critical view of this and the following arguments).

Alhawali also continues to project the American and British press as collaborating in the war against Saddam Hussein (the president of Iraq 1979 – 2003) by demonizing him. Additionally, Alhawali continues to support his claims about American and Israeli war in the Middle East.

Reading through the American and British press quickly, we observe that the campaign for the war [on Iraq] is based on demonizing Saddam Hussein. And the western media started to repeat the threats that Saddam expressed towards Israel and at the same time this [western] media neglected the Israeli threats against the different Arab countries. And in another article the same magazine (EIR FEETURE – spelling in the original text) <u>savs</u>: "Israel is preparing for *another* war in the Middle East to make a final solution to the Arab problem ... This means a war, presumably against Iraq and other states, and the destruction of Jordan. It also ridicules Bush's (George H. W.) claims that he is seeking to establish peace in the region (Middle East). It <u>says</u>: "Bunk. The United States government is committed to a war in the Middle East to achieve what is called the final solution. The exact date is the onty thing that's open to question." It also <u>savs</u>: "There may be, and probably will be, new agreements between forces in Israel and Hafez al-Assad (Syrian president 1971 – 2000), things of that sort. But, the side issues aside, Israel is presently programmed for war. Those who tell you that this is not true are either foolish, or are lying. That's the nature of the situation."

Figure 7.15 American and British press and the war against Iraq (argument 2 in Argumentation thread 3) و بالنظر بشكل عابر للصحافة الأمريكية و البريطانية بلاحظ أنه قد تم بناء الأساس الدعائي للحرب من خلال تصوير صدام حسين كبعيع في المنطقة و تولت وسائل الإعلام الغربية الحرف على وتر التهديدات التي أطلقها صدام ضد إسرائيل و قامت هذه الوسائل في الوقت نفسه بالتعاضي عن التهديدات الإسرائيلية لمختلف الدول العربية. و في مقال آخر تقول المجلة نفسها (EIR FEETURE) : إسرائيل تجهز لقيام حرب *أهرم* في منطقة الشرق الأوسط لإيجاد حل نهائي للمشكلة العربية. و هذا يعني قيما يبدو الحرب ضد العراق و دول أخرى و تدمير الأردن. و تسخر من مزاعم بوتن أنه يسعى للسلام في المنطقة قائلة: هذا هراء فالولايات المتحدة ملتزمة بقيام حرب في و دول أخرى و تدمير الأردن. و تسخر من يرتحدد هو تاريخ إندلاعها، أما الإسرائيليون فهم مستحدون للحرب و إما تيمي بالحراب في العراق و دول أخرى و تدمير الأردن. و تسخر من يرتحدد هو تاريخ إندلاعها، أما الإسرائيليون فهم مستحدون للحرب و إما تيمي بالحال النهائي. و قالت ربما كان هذاك ال جديدة بين القوى في إسرائيل و حافظ الأسد، أو شيء من هذا، و إذا ما تم وضع التعمايا الهامشية جانباً فإن إسرائيل ميرمجة حالياً بالحرب و الذين يقولون غير ذلك إما أغيباء أو كاذبون، إن ذلك هو طبيعة المسائة و أصلها. الذين يقولون غير ذلك إما أغيباء أو كاذبون، إن ذلك هو طبيعة المسائة و أصلها.

^{93.} It is noteworthy here that the study was published almost 10 years before the Gulf crises after which this argument was written by Alhawali (see below).

In the argument above (Figure 7.15), Alhawali utilizes two argumentation schemes: symptomatic and argument from analogy schemes. The context of this argumentation thread is during the military invasion of Kuwait by the Iraqi regime in 1990 (for further information see below and Chapter 8). In the argument from analogy, Alhawali utilizes the American and British media to warrant his argument, though as above, not naming any specific media organization. He writes about the coverage of the American and British newspapers of the war accusing them of bias, comparing what he believes to be exaggeration of Saddam's threats to Israel. He also reports their negligence of the Israeli threats of launching wars against different Arab countries. Alhawali also uses argument from sign utilizing translated (into Arabic) articles from a magazine named 'EIR FEETURE'⁹⁴. In this argumentation scheme, Alhawali warns his readers of possible threats that feed directly into demonizing the west making association between Israel and America.

Comparing Alhawali's quotes to the source article in the 'EIR', I found that either Alhawali paraphrases the translation of the article or the translator was paraphrasing during the process of translation. The following is the first paragraph of the original article:

Figure 7.16 Screenshot of the first paragraph of the original article from the 'EIR'

Israel is preparing for war. The state of Israel is now marshaled, in preparation for a war, which, from one standpoint, might be described as Israel's attempted "final solution" to the Arab problem. This means a war, presumably against Iraq and other states, and the destruction of Jordan.

The translation of the paragraph in the above screenshot in Alhawali's quote reads:

ابسرانيل تجهز لقيام حرب *أخرى في منطقة الشرق الأوسط لإيجاد* حل نهاني للمشكلة العربية... و هذا يعني *فيما* **يبد**و الحرب ضد العراق و دول أخرى و تدمير الأردن.

^{94.} The spelling of 'EIR FEETURE' exists in the original text where this argument is found. However after I searched for this magazine I found that he refers to the Feature articles section in the 'Executive Intelligence Review' which was founded in 1974. He precisely refers to the Feature article that was issued on July 20, 1990 (see also footnote 54 in Chapter 5). The article is edited from the oral remarks made on July 2, 1990 by the political activist and the founder of 'EIR', Lyndon LaRouche. The EIR is quoted as being criticised by the editor-in-chief of the Washington Times as "an anti-Semitic potpourri of disinformation, factoids, rumor, gossip, loony tunes and an occasional fact." (de Borchgrave, Arnaud (August 27, 2002). "Thinking the unthinkable". Washington Times: p. A.15.)

The above Arabic version could be translated as follows:

Israel is preparing for *another* war *in the Middle East to make* a final solution to the Arab problem ... This means a war, *seemingly* against Iraq and other states, and the destruction of Jordan (see Figure 7.15).

The translation of the Arabic version shows an addition of the word 'أخرى' ('oḥrā; another) in the first sentence. Additionally, the underlined sentence is paraphrased as shown in my translation of the Arabic text in Figure 7.15. It is noteworthy that Alhawali introduces his argument utilizing the verb 'تقول' (taqūl; it says) which indicates a direct quote from the Article. However, quoting turns out to be adding the word 'أخرى' ('oḥrā; another) as well as paraphrasing the second sentence in the source text utilizing the phrase 'إيجاد' (li'iğād; to make) and translating the adverb 'presumably' that indicates a presumption into 'فيما يبدو' (fima yabdo; seemingly).

Despite the fact that the author of the source text (in the excerpt above) builds an argument to lend support to the major claim that appears in the title 'Israel is now preparing for a Mideast war', the translator or Alhawali tries to change what is probable into a seemingly inevitable event. Here Alhawali could be accused of committing the 'straw man fallacy' (Reisigl and Wodak, 2001, p. 73). This is done by twisting the argument in the source text by means of re-lexicalization – replacing the equivalent Arabic words that are supposed to transfer the meanings of the source text with ideologically-laden ones or introducing words that are not available in the source text in order to achieve desired (biased) effects in the target text that serve the agenda of the producer of the translation or Alhawali himself.

Figure 7.17 reconstructs and summarizes the argumentation schemes used by Alhawali in an argumentative thread about western political enmity (premises, warrants and conclusions are numbered according to the number of arguments in each thread).

Figure 7.17 Reconstruction of Alhawali's arguments about political enmity (argumentation thread 3)

		gaments about pointear emitty (argumentation tineau 5)			
	N	Arguments structures in Alhawali (thread 3)	Schemes		
Premises	1	There are some scenarios proposed by some Americans to invade the Gulf and Arab Peninsula/ Israel intends to wage a war against Iraq and other Arab countries and it intends to destroy Jordon.			
	2	The western media repeatedly published the threats that Saddam expressed towards Israel and neglected the Israeli threats against the different Arab countries? The claims of Bush (George H. W.) to establish peace in the region (Middle East) are not sincere.	<u>S:</u>		
warrants	1	A military strategic study published by 'Fortune' in May 1979 quoting statements of 'Brown and Brzezinski - Assistants to the President (Jimmy Carter) for National Security Affairs.	Sign / analogy		
	2	Reading through western press 'EIR FEETURE (spelling in the original text) says:	alogy		
conclusions	1	United States had the intention to invade Saudi Arabia before the Gulf Crisis in 1990–91(Unstated conclusion)/ Israelis are ready and willing to launch the war against Arab countries and are ready for what is called the final solution.			
	2	Western media have double standards and are biased/ Bush's claim of establishing peace in the Middle East is not true			
-	Syn	nptomatic relations: A military strategic study published by "Fortune" (which is considered authenitic source by Alhawali) quot	ting two US		
relation	senior officials, namely, Brown and Brzezinski - Assistants to the President (Jimmy Carter) for National Security Affairs who represent the				
lon	American administration (who are considered official authorities). Relation for argument from analogy: Pointing out differences in media				
x	cov	erage.			

Alhawali denounces the western media via comparing its coverage of the threats of Saddam Hussein to Israel to that of Israel to Arab countries (using the argumentation from analogy scheme). He utilizes a study in 'Fortune' quoting the assistants to the President Jimmy Carter to National Security to lend support to his arguments on the basis that these sources are true and representative of the western press (symptomatic schemes). I shall now turn to critically examine Alhawali's arguments in argumentation thread 3 that he utilizes to lend support to his claims about western political enmity to Muslim countries.

7.5.3.1 Critical evaluation of argumentation thread 3 in Alhawali

Alhawali quotes from some unreliable sources in the West (especially in the United States) to lend support to his arguments from sign and analogy. He tries to tell his audience about perceived threats to Saudi Arabia. The critical questions I use to account for these forms of schemes are mentioned in Section 7.5.1.1 and elsewhere in this Chapter. Let us start with Alhawali's arguments from sign.

The declarations of high rank officials such as the assistants to the President Jimmy Carter to National Security are significant in indicating American political views towards Saudi Arabia. However, these statements were published in 1979, i.e. almost 11 years before Operation Desert Strom launched by the American-led Coalition force against Iraq after which the text that contains this argument was published (for further explanation about the political and military contexts of the presence of the Coalition troops in Saudi Arabia see Section 8.5.2 in the next chapter). Alhawali uses such quotes to reject the presence of the troops of the United States and its allies in Saudi Arabia during Operation Desert Storm. This undermines the correlation strength of the quotes of the two American officials as they belong to Carter's Administration (Democratic Party) whereas the 1st war against Iraq took place during the administration of George H. W. Bush who was a Republican president of the United States from 1989 to 1993. Alhawali here could be accused of utilizing the fallacy of false attribution (unqualified authority) as well as using the topos of danger (Reisigl and Wodak, 2001, p. 77) in claiming that Ameican troops will invade Saudi Arabia and occupy the Gulf region. Thus the religiously packed political decision of allowing such troops to come to Saudi Arabia is claimed to be wrong and, according to Alhawali, should result in invading the 'abode of Islam'. Alhawali continues by stressing claims that Israel is backed by the West and is willing and ready to launch a war on Arab countries.

In the second symptomatic argument (Figure 7.15), Alhawali refers to the American and British press collectively (representation by aggregation) as collaborating to demonize Saddam Hussein. He claims that they only publish the threats of Saddam Hussein towards Israel whereas they neglect the threats declared by Israel towards Arab countries. Again, without naming any actual sources, Alhawali aggregates American and British newspapers 'to manufacture' a Saudi (or perhaps Muslim) negative 'consensus opinion' (van Leeuwen 2008, p. 37) towards United States and Britain. Additionally, Alhawali quotes from another source 'EIR FEETURE' of questionable authority (see footnote 93 above) to unequivocally introduce further external threats. Alhawali here as I have indicated elsewhere in this Chapter, has fallen into the fallacy of appealing to an unqualified authority as well as hasty generalization which render his arguments in argumentative thread 3 to be fallacious.

Among the threats that Alhawali's argumentation thread 3 contains is that Israel is ready to launch a war against some Arab states including its aim to destroy the north neighbouring country of Saudi Arabia; Jordan, which means that Israel will be threatening the northern borders of Saudi Arabia. He also includes a quote that ridicules Bush's (George H. W.) claims that he is seeking to establish peace in the region (Middle East) to further demonize the United States. In addition to quoting an unqualified authority, Alhawali bases his argument on the topos of danger to incite his audience against United States and its allies.

In summary, Unlike Ibn Baz, Alhawali uses most of his arguments in political argumentative topics to reject what he believes to be the Western dominance over Muslim countries as well as warning Muslims that westerners have aggressive aims towards Muslim countries. Additionally, he introduces arguments that denounce both social and religious life in the west. Alhawali claims that women in the west are oppressed trying to lend support to his arguments symptomatically as well as utilizing arguments from both analogy and causality. He also denounces the Church by means of criticising Salvation theory making the Church responsible for such teachings. Thus Alhawali uses all the three forms of schemes of Van Eemeren and Grootendorst (2004) in his arguments that are introduced by the different word forms of the verb 'to say' quoting unqualified sources as well as utilizing some other fallacious arguments such as the fallacies of incomplete evidence and hasty generalization as well as some topoi such as the topos of danger.

The argumentation schemes utilized by both authors reveal some aspects of stereotypical patterns (Walton, 2005) of the Salafist reasoning. Despite the fact the Salafi discourses as exemplified in the writings of Ibn Baz and Alhawali, belong to the same geopolitical area and are mostly based on the Salafi teachings of Mohammad Ibn AbdulWahhab who depended largely on the teachings of the medieval Salafi religious leaders, they nevertheless differ in their patterns of reasoning. On the one hand, the patterns of reasoning in Ibn Baz's arguments are mostly theological. As I have explicated above, Ibn Baz's symptomatic argumentation schemes tend to be warranted by Qur'anic verses and Prophetic traditions. The nomination of medieval religious leaders in his arguments reveals another aspect of Ibn Baz's reasoning, namely, a reincarnation of medieval religious Salafi reasoning and explications of the Sacred texts. Additionally, Ibn Baz's argument from analogy via which he compares modern mystic practices to those of pre-Islamic polytheists, is further supported by other Qur'anic verses. Regarding Ibn Baz's argument from causality (his rejection of Arab nationalism), it is also religiously justified; adoption of Arab nationalism is viewed as leading to the rejection of the Sharia (religious laws). Hence, reasoning in Ibn Baz's corpus is mostly moulded with the medieval Salafi conception of Islam as well as marked with reference to Qur'anic verses (and in some cases Prophetic traditions) in all forms of arguments; symptomatic, analogical and causal argumentation schemes.

On the other hand, Alhawali's argumentation schemes indicate a different direction of the Salafist patterns of reasoning which might also indicate an emerging change in Salafi discourse and an evolution of a new politically motivated Salafist strand within the Salafi movement in Saudi Arabia. As we have seen in Chapters 4, 5 and 6, Alhawali's semantic foci and sociosemantic categories are marked with his many negative politico-religious representations of the West, though he shares some of the religious thematic foci with Ibn Baz. This is also evident in his legitimation strategies in this chapter. Apart from his argument about the original sin (which is basically based on analogy –Islam vs. Christianity)⁹⁵, Alhawali's symptomatic schemes are based on quoting non-religious sources (most of which are reports from newspapers, magazines and military and political studies) to support his politically motivated claims. Also, Alhawali utilizes statistics and numbers (to give an impression that his arguments are scientifically based). This is also another mark that indicates a shift in Salafi discourse towards politically motivated arguments. Additionally, both authors tried to regulate practice and manufacture consensus opinion. Thus, we may conclude that stereotypical patterns of reasoning in the Salafist discourses (as exemplified in Alhawali's writings) are incorporating attempts at scientific reasoning (though not accurate and in some cases lacking in veracity) and are becoming politically oriented.

7.6 Conclusion

In this chapter I tried to quantitatively identify and critically analyse the arguments and argumentation schemes in both Salafi corpora under investigation. I utilized the keywords in both corpora in order to identify the most salient key argumentation indicators that distinguish the two Salafi authors when compared against each other. Additionally, I used analysis of frequent clusters to narrow down the analysis to a manageable number of argumentative keywords as well as focussing on analysing concordance lines, including more of the linguistic co-text, to qualitatively account for the arguments that appear therein. And in order to critically evaluate the argumentation schemes in all threads I used critical questions adopted from Walton (1995, p. 142) and cited in Walton et. al. (2008, pp. 309-346) that help to reveal the (un)acceptability of the argument and ideological stances taken by Ibn Baz and Alhawali. Additionally, I used argumentation schemes as analytical tools that could reflect the stereotypical patterns of the Salafi reasoning.

^{95.} In his argument about the original sin in Christianity, Alhawali basically compares Christianity to Islam supporting his argument from analogy with another symptomatic scheme quoting a Qur'anic verse (a warrant), and the words of the Muslims' second Caliph (as a Salafi religious authority).

Both Salafi writers present arguments to lend support to their stances in relation to their (politically oriented/religiously based) semantic foci and representations of social actors: Alhawali is more concerned with western social and religious life and political enmity, whereas Ibn Baz focuses on pure theological issues in Islam. Their arguments can be classified under the three schemes of van Eemeren and Grootendorst (2004). Their arguments from sign that are based on Qur'anic verses are related to the topics they intend to support. However, other symptomatic signs in Alhawali, as I have shown above, are fallacious, because he relies on unqualified authorities or makes hasty generalizations.

The arguments from analogy in argumentation thread 1 in Ibn Baz's corpus and in argumentation threads 2 and 3 in Alhawali's are both fallacious. Both writers make the fallacy of faulty analogy which could stem from the Salafist stereotypical perspective of 'the Other'. On the one hand, Ibn Baz, in his argument from analogy, perceives of 'the Other' (Sufis and Shiites) as practicing polytheistic rituals without addressing their arguments for carrying out such rituals. On the other hand, Alhawali's arguments from analogy, that stem from a Salafist politically oriented stereotypical perspective, are concerned with undermining the efforts of western advocates of historical criticism who criticised the Church's theory of Salvation as well as collectively accusing the western media of bias, without providing convincing evidence. Both authors also use fallacious arguments from causality in their argumentation threads (3 in Ibn Baz and 1 and 3 in Alhawali) as again, they do not provide enough evidence to support their arguments: Ibn Baz employs the fallacy of faulty causation when he tries, from a Salafist perspective, tries to denounce western social life. He also has used the fallacy of faulty causation.

Finally, the two Salafi authors (without discussing their opponents' arguments) initiate their arguments from a Salafist (political or religious) perspective in order to refute,

defame or reject their opponents' social/religious practices. In so doing, both authors try to dissuade their audience from following, accepting or adhering to what they believe to be non-Islamic teachings and practices and/or opposing what they represent as the true teachings of Islam. Such argumentative moves appear to legitimate the authors' stances vis-a-vis the semantic foci (delineated in Chapters 4 and 5) and their representations of social actors (detailed in Chapter 6). Additionally, their arguments from sign (quoting Qur'anic verses, prophetic tradition and other Salafi writings) contribute to the process of legitimizing their stances via impersonal authority vested in the quoted texts as well as help to achieve conformity (legitimation via conformity) with the medieval Salafi mainstream of writings. The argumentation schemes reflected the stereotypical patterns of reasoning of both Salafist writers. These reasoning patterns could also show the difference between the authors as well as indicate the emergence of the politically oriented Salafism.

In order to better understand why certain semantic foci, representations and arguments occur in the writings of these two authors, as we have seen in chapters 4-7, we need to move beyond corpus analysis and take into consideration political and religious contexts which constitute the major focus of the next chapter.

a server party beneficient and server products and an and a server and a server of the server of the server of The server party beneficient at specifically and the server of the server of the server of the server of the ser Server beneficient server beneficients and the server of
Part Four

CHAPTER 8: Explanations of the results in Context

8.1 Introduction

This Chapter concludes the five chapters that analyse the Salafi discourses. It seeks to answer the following research question: How can the findings in Chapters 4, 5, 6 and 7 be explained in relation to Saudi social and politico-religious contexts?

In order to address this question, I use the Discourse Historical Approach (Reisigl & Wodak, 2009) which is triangulatory in nature (see Section 3.4.9 in Chapter 3). In the DHA, the context of a situation is elaborated on by considering links between texts and discourses. In this case I link the discourses that the two Salafi text producers draw on to discourses found in other canonical, religious texts i.e. taking into account interdiscursivity and intertextuality (see Section 3.2.3.4). In addition to drawing on my own knowledge of the context of situations in Saudi Arabia, I refer to the biographies of the two text producers and their institutions which are considered as context-relevant information (see Section 1.3.7 in Chapter 1).

There are three types of analyses in my thesis, namely, semantic foci, social actor representation and argumentation analyses of the writings of Sheikh Ibn Baz and Sheikh Safar Alhawali. In section 8.2 I provide a summary of results found when analysing Ibn Baz's corpus followed by explanations of these findings in section 8.3 and its subsections (8.3.1-8.3.4). Section 8.4 gives a summary of the analysis of Alhawali's corpus followed by two subsections explaining these results in context, namely, sections 8.5.1 and 8.5.2. Section 8.6 brings together explanations of the representations of sexual morality in the West from the Salafi perspectives in both corpora. I have grouped explanations of the findings about the Salafi representations of sexual morality in the West and Islam in one separate section

because the Salafi writers have more or less the same stance towards these two themes. However they differ slightly in respect to the motives behind such representations. Before I start summarizing the findings of analysing the two Salafi corpora, it is worth noting that neither of the Salafi writers mentions each other in their writings that compose the research data. This may indicate no direct antagonism or enmity between the two, although equally, there is no obvious support of the other's position either. Let us first start with a summary of the findings in Ibn Baz's corpus.

8.2 Summary of findings 1: Ibn Baz

The themes in Ibn Baz's writings (Chapter 4) are mostly theological. There are four interrelated theological semantic foci in Ibn Baz that are salient in his writings. Firstly, the concept of divinity through which Ibn Baz has projected Allah as the powerful 'God' to whom the pure religion is to be dedicated, calling on Muslims to refer to the Qur'an and the Sunnah of His prophet (Sharia) in all disputes. Secondly, in his negative representation of polytheism, Ibn Baz recontextualizes quotes from the sacred texts and previous (medieval) Salafi writings and explications of the types of polytheism. His classification and presentation of polytheism are oriented towards rituals and aspects of worship that he conceives of as contradicting Islamic monotheistic beliefs. Ibn Baz also writes about monotheism recontextualizing the Salafi explications of the names and attributes of Allah.

Thirdly, although Ibn Baz's construction of Islam revolves around monotheistic beliefs and rituals, he occasionally writes about what he considers to be the enemies of Islam which results in an interpretation of his writings as political at times. Additionally, Ibn Baz appears to share some political views that Alhawali holds (Chapter 5) which include his rejection of Arab nationalism and colonialism. Ibn Baz's final theological theme is 'the final

258

return' through which he represents two opposing states: punishment (Hell-fire) and reward (paradise).

There are five categories of social actors in Ibn Baz, four of which belong to the ingroup whereas the social actors in the fifth category are considered 'the Other'. These categories are as follows: 'The Prophet of Islam and other prophets', two types of Salaf, namely 'The companions of the Prophet' and their successors, 'General social actors that all belong to the in-group social actors, and finally 'the Other'. It is notable that the in-group social actors outnumber those who belong to the out-group, namely of the 75 keywords that were derived from Ibn Baz's corpus, only 6.6% of them refer to 'the Other'. This may reflect Ibn Baz's interests in the in-group rather than attacking or representing 'the Other'.

In terms of 'the Other', Ibn Baz refers to Arab nationalists and the Auliya (friends of god in the Sufi and Shiite beliefs). Ibn Baz also represents social life in the West negatively claiming that moral denigration permeates the social life there without specifying the moral values that are debased (see Section 8.6 below). Additionally, he represents social actors by personalization via categorization, selecting theological terms such as polytheists and propagandists of doctrines that he conceives of as opposing Islam as well as classification via religionym and functionalization. Additionally, Ibn Baz utilizes nomination by honorification to represent medieval social actors (mainly Salafi religious leaders). These representations are deployed in Ibn Baz's corpus in different argumentation schemes (discussed in the following paragraph). All the sociosemantic categories in Ibn Baz create the difference between the ingroup social actors and 'the Other' (i.e. the 'Us vs. 'the Other').

Ibn Baz utilizes arguments (mostly argumentation from sign) to legitimate his stances towards both the semantic foci he addresses and the social actors involved. His arguments are mainly theological, relying on quoting from powerful in-group others, which are signified by clusters such as 'و قال تعالى' (wa qala ta'āla; and the Exalted said) which indicates Qur'anic

verses (arguments from sign) that, according to Muslims, are seen as inherently true. Additionally he uses arguments from analogy to refute the ritualistic practices of some Sufis and Shiites when they beseech help from dead holy men. He compares such practices to those of the early Arab polytheists. These results could be grouped under different themes that epitomize Ibn Baz's pursuit (from a Salafi perspective) towards the perfection of religious rituals and beliefs. These themes include impermissible innovation of religious practices and unifying Allah in His names and attributes.

8.3 Ibn Baz: Perfection of religious duties (Religiously based Salafism)

The semantic foci and representations in Ibn Baz's corpus could be reformulated as follows: 'Religiously impermissible innovation of ritualistic practices', 'pure ritualistic practices that include his 'refutation of associationism', 'unification of Allah's names and attributes' and the 'political stance and rejection of Arab nationalism and colonialism'. Let us take each in turn.

8.3.1 Innovation of religious rituals: Oneness of Worship vs. Associationism

According to Salafis, any form or rituals of worship should be based on Qur'anic verses or prophetic traditions. While Salafism is the principal form of Islam that is practised in Saudi Arabia, the country is also home to Sufis (who mostly live in the Hijaz area of Saudi Arabia such as Jeddah, Makkah and Medina) and Shiite factions (the Twelvers who live in eastern Saudi Arabia and Ismailis (also called Makramis) who live in Najran – in the southern province of Saudi Arabia - and another Shiite minority, living in Medina, called the Nakhawila). These minority groups have ritualistic practices that Ibn Baz and all Salafis reject and deem to be impermissible religious practices.

Some rituals of some Sufis and/or Shiites that include beseeching help from pious dead individuals are treated by Salafis as 'بدعة' (bid'ah; impermissible religious innovation)

that Muslims should avoid. Such practices did occur during the period when Ibn Baz was writing his texts as the Sufis in Saudi Arabia and other Muslim countries have never ceased celebrating the Prophet's birthday till the time of writing this thesis. There were also some calls to oppose the Salafis' rejection of the celebration of the Prophet's birthday. Ibn Baz mentions in his corpus that Sheikh Salih Mohammad Jamal,⁹⁶ on 26-12-1984 in the newspaper Al-nadwah⁹⁷ issue 7854, criticized the Salafi rejection of the Prophet's birthday.⁹⁸

There are other religious practices that Sufis and Shiite share and are deemed by Ibn Baz and other Salafis to be impermissible. These include the Shiites and Sufi's beseeching of help from Alhussein the grandson of the Prophet of Islam (the mosque of Alhussein is a famous shrine in Cairo, Egypt).

Another form of what Ibn Baz conceives of to be impermissible religious innovation as well as a form of polytheism (see Chapter 4, section 4.2.2) is the Sufi and Shiites' beseeching of help from pious individuals such as the Prophet of Islam and famous members of his descendants including his daughter Fatima and her son Alhussein. The last two individuals mark the Shiites' practice that includes veneration of Fatima and her son Alhussein as well as his father Ali Ibn Abi Talib (the cousin of the Prophet of Islam and the husband of Fatima).⁹⁹ Ibn Baz's rejection of seeking the help of such individuals collides with the Shiites' beliefs. The Salafi perspective that rejects such veneration of holy men

^{96.} Sheikh Salih Mohammad Jamal is a Saudi citizen who is from the holy city of Makkah where some of the modern Saudi Sufi minority exists.

^{97.} Al-nadwah ceased publishing on 27-11-2011 and was replaced by a newspaper called 'Makkah.

^{98.} I searched for the Article that was refuted by Ibn Baz, but as Al-nadwah newspaper ceased publishing it was so difficult to get a copy from the archives. However, another Salafi writer named Altwaijiry refuted the article of Sheikh Salih Mohammad Jamal in a book called 'Refuting the Thesis of the Misguided Writer' (Altwaijiri, 1987).

^{99.} According to the Shiites (mostly the twelvers who live in Iran and some parts of the Gulf States including Saudi Arabia), Ali Ibn Abi Talib is the heir and the legitimate Imam and successor of the Prophet of Islam. The Sunni Muslims including the Salafis hold a different view towards Ali Ibn Abi Talib which makes him the fourth Caliph (after Abu Bakr Alssiddeeq. Omar Ibn Alkhattab and Othman Ibn Affan respectively) and one of the elite companions of the Prophet of Islam.

underlines Ibn Baz's rejection of ritualistic practices of Sufis and Shiites. He conceives of such practices as violating the monotheistic belief of the unity of worship.

The existence of critiques of the Salafi rejection of celebrating the Prophet's birthday and beseeching help from Sufi and Shiites' saints as well as the presence of Sufi and other minority groups in Saudi Arabia helps to explain why Ibn Baz voices his concerns about these practices – as a Salafi he wants to dissuade others within his own country from following practices that he feels are erroneous. Another relevant piece of context which helps to explain Ibn Baz's focus on impermissible religious practices relates to the Shiite Islamic revolution in 1979 which was led by Khomeini. In this revolution, Khomeini used his religious status called 'الله' (*'ayatuallah*; AyatuAllah –the Shiite highest religious authority) to instigate the Twelvers in Iran to revolt against their Shah. There were concerns in the Arab countries that this might instigate Islamists and extend uprisings to other Arab countries including Saudi Arabia. Thus, Ibn Baz, in his role as a religious leader, including his official role as Grand Mufti, would be expected to uphold the traditions of Salafism and set an example which would promote stability and help to cement the power base of the existing leaders (at the time) in Saudi Arabia. He is also expected to warn Muslims, especially Sunnis, to beware of the Shiites' religious practices and show that these practices are erroneous.

In terms of intertextuality and interdiscursivity which include, as Reisigl and Wodak (2009) point out, explicit reference to a topic or main actor, Ibn Baz utilizes discourses of previous Salafi religious leaders such as Ibn Taymiyyah. It is worth considering whether Ibn Baz's discourses around impermissible religious innovation and polytheism are original and thus unique to himself or whether he is drawing, explicitly or not, on other sources. In fact, prohibiting innovations in ritualistic practices is evident in the writings of old and modern Salafis. Ibn Taymiyyah's (1263-1328) efforts in such prohibition are evident and had been expounded in his book 'العبودية' ('al'obodiyyah; Servitude/worship) (Ibn Taymiyyah, 2003).

Additionally, Mohammad Ibn AbdulWahhab's call in the eighteenth century is also based on rejecting veneration and beseeching help from dead holy men, including the Prophet of Islam and his descendants, which are mostly practiced by Sufi and Shiite factions in Saudi Arabia. Aljizani (a Saudi religious clerk and a professor in the Islamic University in Medina) (2010) makes such continuity clear when he proposes that prohibition of ritualistic practices is the rule unless Qur'anic verses of Prophetic traditions report the opposite. The title of his book is as follows: 'Investigating and confirming that the Basic Rule in Worshipping Practices is Prohibition', which recapitulates the Salafist conception of the Sufi and Shiites' innovations of ritualistic practices.

Thus while others have criticised the practices of Sufis and Shiites, Ibn Baz recontextualizes the same rejection in the context of being the officially sanctioned religious leader of a country that may have been under threat from an uprising which could have spread from nearby countries.

Another constituent of Islamic monotheism, according to Ibn Baz, is unifying Allah in His names and attributes. Ibn Baz tackles the belief of Allah's names and attributes refuting some Sunni factions such as the Ash'arites which is the theme of the following section.

8.3.2 Unifying Allah in His names and attributes

As in his representation of associationism (and consequently monotheism), Ibn Baz holds a Salafi stance towards the names and attributes of Allah. He tends to preserve the literal interpretations of Allah's names and attributes that are mentioned in the sacred texts. In doing so, Ibn Baz recontextualizes the Salafi writings about the attributes and names of Allah such as those of Ibn Taymiyyah and Mohammad Ibn AbdulWahhab. The Ash'arites (who nowadays live mostly in Egypt and some other Muslim countries) hold an opposing view of the names and attributes of Allah that are often seen as heretical by the Salafis. The

263

disagreement between Salafis and the Ash'arites stems from the adherence of the first faction to the literal interpretation of the sacred texts that contain Allah's names and attributes.

The Salafis and Ash'arites believe that Allah has the most excellent names and attributes as inferred from the following Qur'anic verse: { {وَبِلْمُ الْأُسْمَاءُ الْحُسْنَىٰ فَادْعُوهُ بِهَا could be translated as: {And (all) the Most Beautiful Names belong to Allah, so call on Him by them}. However they disagree in the interpretation of the names and attributes that appear in the sacred texts. The Salafis, on the one hand, adhere to the literal interpretations of these names and attributes and do not attempt to go further in analysing them, which could lead them to altering their meanings. On the other hand, the Ash'arites attempt to interpret such names and attributes as metaphors. An example of such disagreement is their interpretation of the attribute of 'Highness' in the following Qur'anic verse: {الرَّخمَنْ عَلَى الْعَرْشِ اسْتَوَىٰ}. The meaning of the this Our'anic verse, according to the Salafis, could be translated literally as {The Most Gracious (Allah) rose over (Istawa) the (Mighty) Throne (in a manner that suits His Majesty) which indicates that they believe that Allah is above His creatures. They interpreted it as 'He rose upon the throne', i.e. Allah rose over His Throne which, according to Salafis, is in the highest heaven. They say so without any further explanation using the phrase 'in a manner that suits Him'. However, the Ash'arites explicate the Qur'anic verse metaphorically as Allah has the ultimate authority which could be translated as {(Allah) Most Gracious is firmly established on the throne (of authority)}. In doing so, the Ash'arites, according to Salafis, reject the 'Highness' of Allah and thus they nullify the attribute of Allah's Highness.

As I have noted in Chapter 4, the Sunni religious leaders, including the Salafis, tend not to teach such disagreement between the Sunni factions that pertains to the beliefs in Allah's names and attributes to ordinary people. However, on 25-11-1987 (issue 3383) the newspaper Asharq Al-awsat (Middle East) published an article by Muhyi al-Din Alsafi titled newspaper (min 'ağl 'n nakūn 'aqwā 'ommah; In Order that We Become the

Strongest Nation) which was according to Ibn Baz, dedicated to outlining the idea that there is no obvious disagreement between the Salafis and the Ash'arites' beliefs in the names and attributes of Allah. Ibn Baz here considers Muhyi al-Din's attempt to disregard the Salafi beliefs. He refutes the Ash'arites' faith utilizing quotes from Ibn Taymiyyah's writings and claims that the Salafi faith is true warning against what he believes to be erroneous creed. In this example of Ibn Baz's writings, he recontextualizes the Salafi writings refocusing the attention of his audience on the Salafi faith in the names and attributes of Allah.

As described in Chapter 1, it is also notable that in the 1960s, a number of people from Egypt and other Arab countries emigrated to Saudi Arabia to find work. Some of these people had a non-Salafi religious education and held the Ash'arites' faith (some of them also held the Sufi beliefs such as the veneration of holy men). Taken within such a context then, Ibn Baz's warnings against such beliefs could be interpreted as concern that his own beliefs might be threatened if larger numbers of people coming into his country continued to hold different beliefs, and possibly influence others. He recontextualizes the writings of the medieval Salafi religious leaders in order to refute the Ash'arites. In doing so, Ibn Baz claims to protect the people's faith from what he believes to be distortion of faith by the Ash'arites.

As for Ibn Baz's political views, they can be recapitulated into two dominant themes: Opposing Arab nationalism and colonial domination over the Arab/Muslim countries and the person that Muslims should take as their ruler. Let us take each in turn.

8.3.3 Opposing foreign domination over Muslim countries and Arab nationalism

Ibn Baz also shares some of Alhawali's political views concerning colonialism and Arab nationalism. This makes his representations not purely religious, although Ibn Baz tends to frame his rejection of colonialism and Arab nationalism inside religious discourse that helps to legitimate his stances. In the case of colonialism, Ibn Baz, using a religious focus, claims that a domination of non-Islamic forces over Muslim lands would lead to a loss of religion. Domination of non-Islamic forces over Muslim lands is/was thus used by Ibn Baz as a threatevoking strategy to legitimate his negative stance towards Arab nationalism.

The Arab nationalists in Saudi Arabia were mostly the workers of the then joint American-Saudi petroleum company (ARAMCO) (Fandy 2001) who participated in political unrest which took place in Saudi Arabia in the 1950s and 1960s (see Chapter 1 section 1.3.2). They were inspired by Arab nationalism that was supported and perpetuated by Abdul Nassir of Egypt and his allies (the Baath regimes) in Iraq and Syria. The war between Arab nationalist leaders and the Saudi monarchy started in the 1950s (Commins 2006; Fandy, 2001). Ibn Baz opposed Arab nationalism claiming that the movement was based around propaganda and constituted a war against Islam. He constructed Arab nationalists negatively claiming that they deviate from what he considers to be the right path. He also claimed that they call for erroneous beliefs that oppose the 'pure teaching of Islam'. In doing so, Ibn Baz demonstrated that he is supportive of the Saudi monarchy which relied on religious forces (Hegghammer & Lacroix, 2007; Mordechai, 1993) in their wars including that against Arab nationalists.

It is historically relevant that the establishment of the first Saudi State (1744–1818) started with an alliance that was formed between the prince of Diriyah Mohammad Ibn Saud and the religious reformer Sheikh Mohammad Ibn AbdulWahhab. This type of politico-religious alliance also forms one of the cornerstones of the constitution of the second (1818–1891) and third (modern) Saudi States and was also found within the relationship between Ibn Baz and the Saudi monarchy in the 1990s. The relationship between the monarchy and religious leaders in Saudi Arabia is thus cemented in centuries of tradition. This politico-religious alliance helps to explain Ibn Baz's negative stance, politically speaking, towards Arab nationalists who oppose the Saudi monarchy as well as his rejection of colonial domination over Muslim land including Saudi Arabia. As Grand Mufti (the highest religious

authority in Saudi Arabia), Ibn Baz is expected to support the Saudi monarchy which could also explain some of the other semantic foci in Ibn Baz such as his representation of '*ilequev*' (*'alhorūb*; the wars) during the disruptive events in the years 1818-1824 and his concept of the Imam/ruler (see Chapter 5 section 5.4.2). The next section shows the Salafi concept of Imam/ruler and Ibn Baz's representation of the disruptive historical events (1818-1824) in Najd (the central region of Saudi Arabia).

8.3.4 The conception of the ruler/Imam in apolitical Islam

The concept of the Imam/ruler in Salafi Islam is crucial to understanding Ibn Baz's representation of political themes in his writings. It stems from Qur'anic verses such as:

This Qur'anic verse could be translated as follows:

{O you who believe! Obey Allah and obey the Messenger (Mohammad), and **those of you (Muslims) who are in authority**. (And) if you differ in anything amongst yourselves, refer it to Allah and His Messenger, if you believe in Allah and in the Last Day. That is better and more suitable for final determination} 4:50.

This Qur'anic verse and other Prophetic traditions call Muslims to obey rulers provided that this ruler is Muslim. They also stipulate that they should not obey the ruler if he orders them in matters that dissatisfy Allah. Accession to power by rulers in Sunni Islam should take place via (*'aššorā*; mutual counselling), i.e., people agree on one Muslim to be the ruler which is based on the Qur'an (verse 38 in Chapter 42 or 'الشورى' (*'aššorā*; mutual counselling):

{وَالَّذِينَ اسْتَجَابُوا لِرَبِّهِمْ وَأَقَامُوا الصَّلَاةَ وَأَمْرُهُمْ شُورَىٰ بَيْنَهُمْ وَمِمَّا رَزَقْنَاهُمْ يُنفِقُونَ} سورة الشورى اية ٣٨

This verse could be translated as:

{And those who answer the Call of their Lord [i.e. to believe that He is the only One Lord (Allah), and to worship none but Him Alone], and perform the five daily prayers, and who conduct their affairs by mutual consultation, and who spend of what We have bestowed on them} 42:38.

However, since the Umayyad Caliphate (661–750), accession to the throne is actually based on inheritance rather than mutual counselling where religious leaders acquiesce to the ruler who has the ultimate power. Most modern Salafis acquiesce to any Muslim ruler regardless of the way he ascends the throne. They call him the legitimate Imam. Most Salafis believe that obedience to legitimate Muslim rulers (or 'ولي الأمر' (*wali 'al'am*r; the ruler/the guardian)) is mandatory for all citizens. Most Salafis do not commit themselves to politics that, according to them, is the responsibility of the Imam/ruler. They also believe that if such a ruler makes mistakes or deals unjustly with his people, then 'نصيحة' (*nasiḥah*; advice) should be given to him privately rather than publishing critiques or denouncing him publicly which may instigate the public against the ruler.

Such concessions to the ruler could explain Ibn Baz's stance towards the Saudi monarchy which is supportive rather than contesting of the monarchy. In his representation of political matters such as Arab nationalism or the disruptive historical events (1818-1824) in Najd after the military invasion led by Mohammad Ali Pasha of Egypt ended the first Saudi State (1744-1818), Ibn Baz writes his opinion in a manner that shows his loyalty to the Imam as the legitimate ruler. For example, he recontextualizes the disruptive historical events (1818-1824) to superficially address the early stage of establishing the second Saudi State reducing it to gathering the people (of Najd) by the Imam/ruler. Here, Ibn Baz clearly shows his support of what he believes to be the legitimate Muslim ruler. Hence, Ibn Baz's political stances are theologically bound and also strongly linked to a sense of loyalty to the Saudi monarchy and desire to maintain the status quo which is not completely recognized by the politically oriented strand of Salafism, as in Alhawali's writings. In what follows, I try to tackle Alhawali's semantic foci and his representations of social actors starting with a brief summary of the findings in Chapters 5, 6 and 7.

8.4 Summary of the findings 2: Alhawali

Alhawali shares some of the theological views of Ibn Baz such as the dangers of 'Polytheism' and the existence of 'Hell-fire' for those who deviate from what he believes to be the true teachings of Islam. However, he is mostly concerned with political issues, as shown by the semantic foci of his writings. These themes tend to be concentrated on protecting Islam from what Alhawali deems to be threats or enemies who he sees as wanting to destroy Islam and invade Muslim/Arab countries. Unlike Ibn Baz, the most prominent theme in Alhawali's corpus is 'External threats' that are focused on western and Israeli military invasions of the Arab/Muslim countries (Crusades and colonial wars). The construction of the 'Past' in Alhawali's corpus is limited to a negative representation of Europe and his claims about longstanding European enmity to Islam as well as constructing an idealised, advanced image of Islamic civilization in the Middle Ages. Alhawali uses the key term 'Mind' to negatively represent the teachings of the Church, political and economic theories (such as communism) and the 18th century literary romanticism, i.e., demonizing the West. He also uses the key term 'RELIGION' to construct a negative image of Europe recalling what he believes to be the historical enmity between Europe and Islam. It is also worth noting here that Alhawali represents western religious and moral values negatively considering the West to be void of any moral value, although at times he is not very specific about those values (detailed in section 8.6 below).

Alhawali's semantic foci involve representation of social actors that mostly belong to the out-group. As I have noted above and in Chapter 6, the representation of social actors in Alhawali is mostly concerned with 'the Other' (81% of the social actors in Alhawali belong to the Other). Alhawali uses both personalization and impersonalization categories to represent social actors. He chooses sociosemantic categories that represent social actors collectively and anonymously by means of personalization via categorization (e.g. 'the

269

human' and 'mankind') and functionalization (as in 'researchers'). He also metonymically represents 'the Other' by means of objectivation via spatialization (impersonalization) such as 'Europe' and 'the West'. Alhawali also uses categorization such as 'clergymen' and classification via religionym such as 'the Jewish'. Abstraction is another means that Alhawali uses to represent social actors such as 'communism'. Finally, Alhawali uses nomination strategies to represent political and literary figures such as Saddam Hussein and Taha Hussein respectively. As is the case with Ibn Baz's referential strategies, Alhawali's sociosemantic categories serve to create the difference between the positively represented ingroup social actors (Us) from the negatively represented social actors that he deems to be 'the Other'.

Alhawali's representations are introduced with argumentation schemes that he utilizes for the purpose of legitimation. His argumentation schemes included the three schemes introduced by Van Eemeren and Grootendorst's (2004): causal argumentation, symptomatic (sign) argumentation and argumentation based on comparisons (argument from analogy). The following sections address Alhawali's different political and theological representations by considering the context within which he was writing.

8.5 The West and western enigma: Politically oriented Salafism

As I have shown in Chapters 5, 6 and 7, there are two major aspects of the West that Alhawali aims to represent negatively. These aspects are western religious and social lives as well as what Alhawali believes to be the political and military aggressions on the Muslim/Arab countries. It is helpful here to reiterate (from Chapter 1) some tenets of the political Salafi trend that emerged clearly in the 1980s and were in vogue in the 1990s.

As I have noted in Chapter 1 (section 1.3.4), some advocates of the Islamic awakening trend in Saudi Arabia have incorporated some of Sayyid Qutb's (1906-1966)

religious and political views into the Salafi Najdi backgrounds (Alotaibi 2007). Mohammad Qutb (1919–present), who is the brother of Sayyid Qutb, moved to Saudi Arabia and worked as a lecturer in Um al-Qura University in Makkah where he supervised both the MA and PhD theses of Alhawali. The brotherhood ideologies and political views are embodied in the writings of Sayyid Qutb and his brother Mohammad which include 'الحاكمية' (*alhakimiyyaha*; sovereignty of Allah) – or unifying Allah in His sovereignty – and 'الجاهلية' (*ğahiliyyah*; ignorance of the pre-Islamic days) (for more details on the tenets of Sahwah see Chapter 1 section 1.3.6). Now let us move to consider Alhawali's politically oriented representations.

8.5.1 Religious and Social life in the West

Despite the fact that Alhawali did not use many terms that constitute the cornerstones of the religious awakening movement in his writings, he nevertheless incorporated them in his representation of the West. He represented the Church and clergymen negatively. He also claimed that clergymen carried out immoral conduct. Additionally, he claimed that the Church and clergymen distorted Christianity which consequently caused people in the West to adopt secularism and political, economic and literary theories (such as communism and romanticism). These theories, according to Alhawali, are against the teachings of Islam. Social life in the West was also projected negatively by Alhawali. Western women were represented negatively, for example as undertaking acts of shoplifting and committing suicide because they are oppressed.

All the above negative representations of the West serve Alhawali's aim of demonizing the West and western thinking calling it modern 'جاهلية' (*ğahiliyyah*; ignorance of the pre-Islamic days).¹⁰⁰ But why would Alhawali focus so much of his writing in doing so? Saudi Arabia started to modernise after the rapid increase of oil revenues in the 1960s and

^{100.} The word 'modern' does not carry any positive connotation. On the contrary, it refers only to contemporary Europe.

1970s. Around this time, some Saudis (both males and females) started to study in rich western countries which had good reputations for excellence in education.

Some Saudis who studied in the West started to work in different jobs in education, industry, medicine and other professional sectors. They embarked on modernising social life in Saudi Arabia. Some of them started to call for equality between men and women, based on what they had experienced when they had studied abroad. In hospitals and the media sector (Saudi official TV channels and radio), women began to work side by side with men. There were calls that women should be allowed to drive cars which was thwarted by a fatwa issued in 1990 by the Grand Mufti (AbdulAziz Ibn Baz) that bans women from driving cars.

Alhawali also brought in his refutation of westernization of the country (Saudi Arabia) the experience of modernizing Egypt that took place in last decades of the 19th century and the beginning of the last century. He warned his audience about what he considers to be extreme modernists who were under the influence of western thinkers. Among these Arab modernists is Ahmed Lutfi El-Sayyed (Egyptian intellectual 1872-1963) who, among others, were used as examples by Alhawali to warn Saudis against modernization of their country that, according to him, is against the values of Islam and even involves changing the language that people speak.

Additionally, Alhawali claimed that Arab nationalist/socialist parties had been under the influence of the West.¹⁰¹ Thus Alhawali includes Arab nationalists in those whose call for modernizing Arab countries to be contributing to changing religious, political and social norms in Arab/Muslim countries.

Alhawali, like Ibn Baz is a religious leader, but whereas Ibn Baz is strongly linked to the Saudi monarchy, Alhawali was one of the leaders of the Committee for the Defense of

^{101.} The West, according to Alhawali, is comprised of the non-Islamic states in the West including communist countries.
Legitimate Rights (CDLR) created in 1993, which openly challenged the Saudi monarchy. Instead, Alhawali has been a leader of the 'awakening' movement in Saudi Arabia. Rather than focussing his message around warning Saudis about Shiites and others who might threaten the status quo, Alhawali would contest the current monarchy, which he associates at times with being too complicit with western modernising trends. The modernization process of Saudi Arabia (and other Arab/Muslim countries) is considered by Alhawali as a precursor of a new colonial phase and invasions of Arab and Muslim countries including Saudi Arabia. Next, I give an explanation of Alhawali's representation of external hazards that threaten Saudi Arabia and the Arab/Muslim countries at large.

8.5.2 The Crusades and the Crusader states: Protecting Islam and the abode of Islam

Alhawali's representation of the West is ambivalent. Sometimes he claims that the West rejected religion and has adopted secularism. But in other parts of his writings, he represented the West as religiously fanatic and wanting to destroy Islam. Alhawali uses the term 'Crusades' and 'Crusader states' in order to remind his readers of the historical devastating wars that, according to him, were initiated by Western countries against Muslims during the Middle Ages, as well as linking the historic concept of crusaders to modern-day western political policy (for an accessible introduction to Crusades see Runciman (1951)).

Alhawali's stance toward western countries could be explained if we take the political context in Saudi Arabia into our consideration. On the one hand Saudi Arabia, as a major oil producing country, has a close relationship with the United States of America. On the other, there is a continuous support of the United States and other western countries provided to Israel. The latter has been in an ongoing struggle with Palestine since the early 20th century, forming part of a wider Arab-Israeli conflict. Safar Alhawali, as a religious leader, is not at ease with the Saudi-United States close relations. He has thus envisaged the US support to Israel as a crusaders' onslaught on the Arab and Muslim countries.

Additionally, on the 2nd of August 1990, Saddam Hussein invaded Kuwait which forced Saudi Arabia to seek the military assistance of the United States. The presence of the American and Allied troops in huge numbers also served as a catalyst for Alhawali to represent the West as conducting a new crusade against the Arab countries in his writings. Hence, Alhawali represented the western countries as pursuing an old religious antagonism against Islam.

It is noteworthy that Ibn Baz was not concerned in his writings with Saudi diplomatic relations with the United States of America or with American support to Israel. And, as I have indicated above, Ibn Baz opposed Arab nationalists' movements and rejected their political agendas. Ibn Baz's position here stems from his Salafi conception of the Imam/ruler where individuals should acquiesce to him and support his decisions (see Section 4.3.4 above). It is also noticeable that Arab nationalists were in a war with the Saudi monarchy and, according to Ibn Baz himself, they intended to disregard the Sharia laws should they rule. Ibn Baz is part of the government as the highest religious authority in the country and he is expected to reject Arab nationalism and support the Saudi monarchy.

Alhawali adheres to the Ideas of Sayyid Qutb and his brother Mohammad who supervised the two academic theses of Alhawali (see Chapter 1 section 1.3.7.1). Sayyid Qutb's ideas are based on his rejection of what he believed to be weak and non-Islamic stances of modern rulers of the Arab countries. While the two authors are identified as Salafi, their other allegiances influence their stances, indicating that there is not a single set of political tenets that unite all Saudi Salafis. With that said, one area of agreement was in their criticism of the morality of the West. Let us consider explanations for this in the section below.

8.6 Morality in the West

While both Salafi writers have differed to an extent in terms of their foci, arguments and aims, one way that they are similar is that they have sought moral legitimation by representing morality in the West negatively. Words that indicated the West were not salient in Ibn Baz's corpus. However, they appear when I compared the uses of the key term 'WEST' in Alhawali to that in Ibn Baz's corpus. As I have pointed above and elsewhere in this thesis, while Alhawali extensively demonizes the West, Ibn Baz's representation of the West was less obvious. The latter rejected foreign domination (Western/Eastern colonialism) over Arab countries (see Section 8.3.3 above). Ibn Baz also wanted to limit sending Saudi students to Western countries only to study sciences that are not available to students in Saudi universities.

Ibn Baz claimed that Saudis studying in western universities might like and imitate what he believed to be denigrated moral values of the West. Additionally, he claimed that some western writers warned against the mixing between men and women. He also claimed that in the West strange behaviours are exhibited by young men because they lack faith. As described above, the fact that people who had studied abroad wanted to instigate sexual equality in Saudi Arabia helps to explain why Ibn Baz focussed on characterising the West as amoral in his writing. Ibn Baz as a Salafi religious leader believed he should warn his Saudi fellows not to acquire the moral values of the West as he deemed them against the teachings of Islam in order to keep the status quo of the traditional and religious social values of Saudi Arabia.

Alhawali's negative representation of the moral values of the clergymen and the medieval crusaders as well as in modern western countries is motivated by his negative stance toward secularism and modern western domination over Arab/Muslim countries. He attributed the rejection of religion by people in the West to what he believes to be the

immoral behaviours of clergymen. As I have indicated in section 8.5.1 above, the incremental modernization of Saudi Arabia following western civilization was seen by Alhawali as a precursor of westernizing and secularizing Saudi Arabia, hence the abode of Islam would be dominated by western 'non-religious' and 'جاهلي' (*ğahili*; of/relating to ignorance of the pre-Islamic days) civilization. Alhawali as a Salafi religious leader is expected to hold a negative stance towards both modernizing and 'secularizing' the abode of Islam, Saudi Arabia.

Thus while both writers disapprove of the West, there is a difference in the amount of focus they give to this subject (with Alhawali having it as a main focus). Additionally, Ibn Baz appears to be more concerned about Saudi people copying certain western ways, while Alhawali constructs the west as a potential invader of Muslim countries. This could be attributed to the fact that Alhawali, in his politically motivated negative representations of the West, was under the influence of the two Qutbis: Sayyid Qutb and his brother Mohammad who was the supervisor and mentor of Alhawali in his higher studies.

8.7 Conclusion

This chapter concludes the previous 4 chapters dedicated to analysing the Salafi discourses in terms of semantic foci, representation of social actors and argumentation schemes. The main goal of the present chapter is to provide explanations of the results obtained in Chapters 4, 5, 6 and 7 utilizing religious, political and social contexts. I have introduced explanations of the findings in three major areas; 1) the religiously based ritualistic practices that are aimed towards the perfection of the religious duties from the Salafist perspective; 2) external hazards that both Salafi writers felt threatening Islam and Arab and Muslim countries as well as religious and social lives in the West and; 3) the moral values in the west.

In their different representations, the agendas of two Salafi authors meet in wanting to protect Salafi Islam and the bode of Islam and keeping the Saudi status quo of religious and social conducts that are based on strict interpretations of the sacred texts (Qur'an and the Prophetic traditions). However, they diverge in their politically motivated representations of the West and western morals. While Ibn Baz, in his politically motivated negative representation of Arab nationalism, supported the Saudi monarchy, Alhawali rejected the Saudi decision that allowed the American-led troops to come to Saudi Arabia based on a fatwa issued by Ibn Baz. Each of the two opposing stances towards the Saudi monarchy stems from different Salafist perspectives (one focussed more on theological aspects and maintaining the status quo in Saudi Arabia, the other more openly politically-focused, contesting the status quo in Saudi Arabia and explicitly anti-western) as I have indicated above.

After this summary of the most important findings, it is time now to move to the final concluding chapter in this thesis. Chapter 9 will focus on methodological implications and limitations of the study as well as recommendations for future research in studying religious discourses critically.

Chapter 9: Conclusion

9.1 Introduction

This conclusion chapter aims to address and reflect on the findings made in the earlier analysis chapter. Section 9.2 provides brief summaries to the research questions stated in Chapter 1. Following this, in section 9.3, I consider the methodological implications raised by the work undertaken in this thesis. In section 9.4 I critically reflect on the limitations of this thesis which is followed by recommendations for future research in section 9.6. The thesis closes with some concluding remarks in section 9.5.

9.2 Saudi Salafists: Ibn Baz vs. Alhawali

It is not the intention of this section to rewrite sections 8.2 and 8.4 in Chapter 8 or the results from the previous analysis chapters. Rather, I only intend to point out the Saudi Salafis' major general differences and/or similarities that have emerged over the analysis chapters (4-8). Sketching out the differences or similarities in the writings of the two strands of the Saudis Salafism could serve as a starting point for researchers to draw the boundaries of Salafism worldwide. It was an intention of this thesis to draw attention to the fact that Salafis do not consist of a single united group with shared goals but that a more diverse set of discourses revolve around different conceptualitions of Salafism.

Before briefly summarizing the similarities/differences in the Salafi strands in Saudi Arabia, it is worth repeating that the work in this thesis is based on keyword comparisons between two corpora consisting of the writings of two Salafi leaders in Saudi Arabia. The keyword analysis was designed to contrast the most salient discursive features between the two writers. I have distinguished between three types of keywords, viz. semantic foci keywords, keywords that indicate social actors and argumentation indicators.

278

In the semantic foci chapters (Chapters 4 and 5), I compared the use of keywords for the analysis of each chapter in both Salafi corpora. I distinguished between theological foci (mainly referring to Allah, the prophet and other aspects of religion) and political foci (which involved references to countries and ideologies). Ibn Baz's keywords tended to consist of the former, while Alhawali's involved the latter. While Ibn Baz tends to recontextualize the Salafi creeds regarding the main tenets of the Salafi movement of Ibn AbdulWahhab, Alhawali's focus is mostly politically motivated towards negative representations of the West.

However, Ibn Baz shares some political views that Alhawali holds (Chapter 5) which include his rejection of both Arab nationalism and colonialism which is motivated by his support to the Saudi monarchy against Arab nationalists' regimes in Egypt, Syria Iraq and Yemen. Alhawali shares some of the theological views of Ibn Baz such as being against the concept of 'Polytheism' while emphasising the punishment of 'Hell-fire' for those who do not adhere to his religious views. However, Alhawali is mostly concerned with political issues that are shown in the semantic foci of his writings (Chapter 5). These themes are mostly focussed on protecting ('the abode of') Islam of what Alhawali deems to be the enemies in the West who want to destroy Islam and invade Muslim/Arab countries.

Regarding social actors representation, most of the social actors in Ibn Baz belong to the in-group which could be interpreted to be typical of religious discourses which nominate religious figures (the Prophet of Islam and his companions as well as the Salaf). On the contrary, the social actors in Alhawali's corpus are mostly politically motivated indicating 'the Other'.

While 81% (out of 37) of Alhawali's social actor keywords refer to 'the Other', only 6.6% (out of 75) social actor keywords indicate 'the Other' in Ibn Baz's corpus. Also, when comparing the two Salafi authors in their use of certain selected social actor keywords (as in

sections 6.2.1 and 6.3.1 in Chapter 6), both writers differ even in using the same appellations utilized to designate certain social actor. For example both Salafi writers use the words, ' $(z_{a})''$ (do ' $\bar{a}h$; teachers/propagandists) and ' $(z_{a}h)$; ' $(z_{a}h)$; the people of) (these are keywords in Ibn Baz) to represent the 'Other'. Each Salafi writer uses them to represent different social actors though. Ibn Baz's representations of the 'Other' are, by and large, related to polytheism and people who claim to be Muslims but hold what he feels are non-Islamic beliefs or engage in non-Islamic practices. On the hand, Alhawali is mostly concerned with demonizing the West as a potential military invader and coloniser (as opposed to Ibn Baz's view of the West as more implicitly transmitting unwanted cultural values to Saudis who study there). Despite the fact that the above Salafi negative representations of the 'Other' contribute more to the reification and polarization of two 'Worlds' (the West vs. 'the Muslim World'). These representations are developed in the writings of both Salafi authors utilizing argumentation schemes that I have analyzed using certain keywords as lexical indicators to locate them in the two Salafi corpora.

The keywords that serve as argumentation indicators vary in the Salafi corpora both on the linguistic level and in the significance of their meanings. On the one hand, a reliance on Qur'anic verses, Prophetic traditions and the early Salafis' writings has strongly influenced the choice of argumentation indicators in Ibn Baz's writings. On the other hand, Alhawali's argumentation-indicating keywords mostly do not bear any mark that indicates religious sacred texts or the writings of the Salaf. Additionally, Alhawali's symptomatic arguments are mostly based (perhaps surprisingly) on the writings of western individuals.

It is perhaps surprising to discover how easy it was to identify aspects of fallacious reasoning in the writings of both authors, particularly considering their status as important leaders. Across the writers I found fallacious appeals to (unqualified) authority, cherry-

280

picking, false attribution, correlation not causation, hasty generalization and the straw man fallacy. Alhawali's use of sources that were vague or of dubious authority to back up his arguments was especially noteworthy. However, perhaps it is due to the status of the writers that their writings have not received this level of scrutiny before. As leaders, their proclamations are meant to be read as truths, rather than critically questioned by their followers.

While the argumentative strategies are flawed, these texts were probably not intended to be seen as authoritative in an 'academic' sense then. Indeed, Ibn Baz's highly frequent use of recontextualising quotes from the Qur'an and other important Islamic writings automatically lends credibility to believers who are not expected to question their veracity. Instead, they function more as polemics, and perhaps work better in terms of promoting emotional rather than rational responses in their readers, especially through use of social actor representation. In terms of rhetoric, on the other hand, Alhawali's use of language employs including impersonalization, numerous 'othering' strategies spatialization and collectivization. 'The West' is thus constructed as an aggressive, invading opposition. And Alhawali recontextualises the concept of the crusader, to imply that westerners have historically always being the enemy. Rhetorically, this is a powerful strategy in that it indicates continuous and longstanding animosity. Also, unlike Ibn Baz, Alhawali appears to be relying on evidence based argument.

To sum up, the two Salafi writers share some Salafi aspects in their representations that aim to maintain the status quo of the religious beliefs and/or rituals and social norms of Saudis which is noticeable in their rejections of innovations of religious rituals such as the Shiites and Sufis' venerations and beseeching help from dead saints. Both writers also reject modernizing aspects of the society such as women driving cars.

While both writers reject colonialism and any form of foreign domination over Muslim countries, they differed in some of the politically motivated representations. While Ibn Baz was supportive of the Saudi monarchy, Alhawali contested them specifically in seeking the help of the United States of America against the Iraqi troops that invaded Kuwait in 1990.

Thus, while my approach has highlighted similarities between the two writers, it has also shown that there are important differences that could be taken to mean that the Salafi movement in Saudi Arabia is not as homogenous as deemed to be by some western and Arab writers. As I have indicated in Chapter 8, the differences between the two writers can be largely explained by considering the social position, connections and influences that each one has within Saudi Arabia, as well as the sorts of expectations assigned to people holding their respective roles within that society.

Now let us turn to consider the implications that resulted from applying the qualitative and quantitative methods used in this thesis.

9.3 Implications

I do not claim by any means that I have reinvented the synergetic approach that combines CDA with CL. However, to the best of my knowledge, this is the first large scale study that from a critical discourse perspective attempts to study religious discourse per se and more specifically Arabic religious discourse of Salafis. Additionally, what distinguishes my contribution in this thesis is that it opens the way for critical discourse analysts to embark on a mission of analysing religious discourses critically. As noted in Chapter 2 of this thesis, religious texts have tended not to be subjected to critical discourse analysis. There are most likely good reasons for this as such research is likely to be seen as blasphemous if the end result is a rejection of texts that hold sacred status in a particular religion. However, I would argue that is possible to critically examine texts that have recontextualised aspects of such sacred writings without falsifying the religion per se. Indeed, as a Muslim, this research has not changed my continued belief in the sanctity of the Qur'an and the Prophetic traditions. Thus, I argue that a CDA of religious texts can be carried out from within a religious perspective and that 'critical' is different from 'criticize'.

Bringing CDA and CL together in investigating religious discourses per se and more specifically the Salafi discourses is another contribution of this thesis as it paves the way to critical discourse analysts to utilize the synergetic approaches of CDA and CL to investigate Arabic discourses (religious or otherwise). And as I have indicated earlier, religious writings (especially in Islam) pose a problem for critical discourse analysts: they are linguistically manifested, at least in the research data of the present study, in a hybrid of manifold recontextualizations of Qur'anic verses and Prophetic traditions by different historical and modern religious leaders in different religious factions –Ash'arites vs. Salafi interpretation of the names and attributes of Allah is only one example. Without taking the side of any faction utilizing corpus linguistic tools has helped me to begin my analysis with a relatively objective set of foci – the keywords - which have allowed me to investigate the different ideological representations in the two strands of Salafism in Saudi Arabia.

In this thesis, I have also implemented the analysis of argumentation schemes as part of an eclectic CDA approach. Argumentation schemes, which are used for different authorial legitimation purposes, connect three types of analysis: semantic foci and sociosemantic categories using the triangulation in the DHA to explain the results in context. The aim of adopting the typology of argumentation schemes in Van Eemeren and Grootendorst (2004) is to critically evaluate Salafi arguments utilizing some critical questions adopted from Walton (1995, p. 142) and cited in Walton et. al. (2008, pp. 309-346) as well as to show the stereotypical patterns (Walton, 2005) of the Salafists' reasoning. And in order to locate arguments in the research data, I have utilized argumentation indicators which facilitated the search of arguments in my data utilizing corpus tools. Thus a further unique aspect of my research is in combining a set of existing methods and tools for analysis in a way that, to the best my knowledge, has not been attempted before. Such a combination could be considered as a possible method for other researchers who are interested in comparing two related ideological or religious based positions.

Finally, the focus on Arabic language presents another set of implications for research. It is still the case that both in corpus linguistics and in critical discourse analysis, the greatest amount of work takes place on the English language (or related European languages), and from within 'western' contexts. The analysis of Arabic-language texts resulted in challenges, in terms of translating texts to English, working with corpus tools that are normally used with English texts, and taking into account grammatical and morphological aspects of Arabic that differ from English. I have dealt with these challenges as they arise, although as the following section will indicate, Arabic itself perhaps poses some potential limitations for this study.

9.4 Limitations

In analysing the theological aspects of the data the eclectic synergetic approach of CDA and CL utilized in this thesis provided me with a working framework to carry out analysis, allowing me to answer the research questions that I set. However, after embarking on the analysis of the research data, it is important to acknowledge some limitations regarding the corpus linguistics tool utilized in this thesis. The nature of the research data may have played a crucial role in the emergence of such limitations. What I have dealt with in this thesis is religious discourse per se that has both theological (mostly in the writings of Ibn Baz and to a lesser degree in the writings of Alhawali) and non-theological (mostly in political writings in Alhawali's corpus) natures.

First, collocational analysis did not enable me to reveal any positive or negative semantic preference in the nomination of social actors in Alhawali's corpus (however in other cases they fit well in the analysis) but this problem is solved utilizing concordance-based

analysis which revealed more about the negative meanings that are associated with those social actors. For example, the word form 'الدعاة' ('addo 'āh; the teachers/propagandists - def. pl.) is used to represent in-group social actors which makes the meaning of this keyword in English to be 'the teachers'. This word has two meanings, the collocates tend to be used with the positive meaning of the word, rather than contributing towards the construction of 'the Other'. Therefore, a collocational analysis is not appropriate here in order to examine how Ibn Baz constructs the Other with this keyword. Instead, I have focused on concordancebased analysis and needed to differentiate the homographs by hand. The form 'الدعاة' ('addo 'āh; the teachers/propagandists – def. pl.) was identified as a keyword, used more by Ibn Baz, which resulted in me analysing it. However, once the two meanings of the words are disambiguated, it is doubtful whether the 'propaganda' sense of the word would have been key for Ibn Baz. Had I been able to use an Arabic semantic tagger which annotated words in order to disambiguate meaning, it is possible that a different set of keywords would have emerged. Similarly, a tool which could split conjoined Arabic words (such as words which contain articles in them), and then identify lemmas of words might also alter the sorts of keywords produced.

It is noteworthy here that while WordSmith 4 was able to process the right-to-left writing system of Arabic, it is not designed to recognize homographs and homophones which are only realized by speakers of Arabic (native or otherwise) and so researcher intervention was required on my behalf.

A further limitation of this study is concerned with the data collected. I have only focused on written data produced by the two authors. However, both authors also made speeches, that exist as recordings and if analysed could contribute to a clearer understanding of Salafism in Saudi Arabia and further consolidate the results obtained in this study. I did not choose to include this spoken data in my thesis as it would have required too much time in terms of transcription and preparation. Thus my findings should only be limited to the comparison of the written output of the two writers. Analysing spoken data of the two Salafi strands (that I focused on in this thesis) would be one recommendation for future research that I shall consider in the next section.

9.5 Recommendations and future Research

I would recommend the replication of this study on other religious discourses in different religions. With regards to Islam there are many religious discourses within different Islamic factions that could be identified via CDA techniques. Examples of these are the modern 'الجهاديون' ('alğidiyyūn ; Jihadi Salafists or the Jihadists) and other Salafi factions in Saudi Arabia and other Muslim countries, 'الأحباش ' ('al'aḥbāš; the Ahbash –a Sufi faction that claims to be Sunni Ash'arites in Lebanon), the different Shiite factions specifically the militant 'فالله ullah; Hizb Allah) which is a faction of the Shiite twelvers in Lebanon (this faction adheres to the teachings of the Iranian Ayatu Allah the Sistani who lives nowadays in Iraq) and the newly evolved Shiite factions; 'الحوثيون' ('alḥutiyyūn; the Huthis) in Yemen and 'المهدويون' ('almahdawyīn; the Mahdis –a faction of Shiite Twelvers who believe in 12 Imams and 12 Mahdis –Saviors) that started to appear in the late 1990s in Iraq.

Also, other non-religious discourses in Arabic (political or otherwise) need to be investigated utilizing a combination of CDA and CL methods. I would also suggest that argumentation schemes are to be considered in further investigations of discourses.

Building diachronic corpora in Arabic that reflect the different aspects of social, political or religious lives as well as literary works in the different historical epochs (pre-Islamic, early Islamic, Umayyad's or the Abbasid's eras and so on)¹⁰² are legitimate targets of investigations. Building such corpora is of prime importance to understanding the different linguistic and other phenomena (political, social, religious and so forth).

¹⁰² Umayyads (661-750) and Abbasids (750-1258) are two dynasties that ruled the Islamic empire.

9.6 Concluding remarks

Now, we have reached the final word of this conclusion chapter. In this thesis I have utilized the synergetic approach that combines both CDA and CL in addition to brining into the analytic tools the argumentation schemes from the argumentation theory. With this amalgam of approaches, I hope that I have shed light on the Salafi discourses in Saudi Arabia and pointed out the different ideological discursive strategies employed by the two Salafi authors in their representations. A major intention of this thesis was to demonstrate that, like many other religions, Salafism is not necessarily experienced in a homogenous way but is instead shaped by the interests of individuals and the political, historic and social contexts that they live in.

This thesis is also hoped to open the way for future research to investigate and compare religious discourses across different factions, both within strands of Islam or within other religions across the world. And while religious discourse poses unique challenges for critical discourse analysis research, I hope that this thesis has indicated that they are not necessarily insurmountable.



References

- Abdul-Wahhab, M. (1996). *Kitab At-tauhid "The Book of Monotheism"* (C. R. D. DAR-US-SALAM, Trans.). Riyadh: Dar-us-Salam Publications.
- Al-Farsy, F. (1990). Modernity and Tradition: The Saudi Equation. London: Kegan Paul International Ltd.
- Al-Hilali, M. T.-u.-D., & Khan, M. M. (1997). *The Noble Qur'an: English Translation of the Meanings and Commentry*. Medinah: King Fahad Complex for the Printing of the Holy Qur'an.
- al-Jaza'iri, A. B. (2003). Hatha Alhabeeb Ya Mohib (This is the Beloved One). Madenah: Aloloom wa Alhikam.
- Aldayyani, Y. (2007a). 'Fitnat alhakimiyyah', The Sedition of Devine Sovereignty. In T. e. a. Aldakeel (Ed.), *Assaroriyyah 'Sarorism'* (Vol. One, pp. 64-84). Dubai: Almesbar Studies & Reseach Centre.
- Aldayyani, Y. (2007b). Saror rahal min Burayyadah wa baqiyat afkaruhu 'Saror... Left
 Buraidah, but his Ideas Remained'. In T. e. a. Aldakeel (Ed.), Assaroriyyah 'Sarorism'
 (Vol. one, pp. 41-51). Dubai: Almesbar Studies & Reseach Centre.
- Alexander, P. S. (2008). Insider/Outsider Labelling and the Struggle for Power in Early Judaism In N. Green & M. Searle-Chatterjee (Eds.), *Religion, Language, and Power* (Vol. 10, pp. 83-100). New York: Routledge.
- Alharasi, A. (2003). min 'alhayymanah 'alƙafiyyah 'ilaa 'alttadfoe 'alwaaei: naħo ro'yah loġawiyyah naqdiyyah ɛarabiyyag; From the Hidden Hegemony to Cognizant Scurry: Towards an Arabic Critical Linguistic Perspective. Nizwa, 34(April).
- Alhawali, S. (1985a). 'žaahirat al'iraja' fi 'alɛaalam al'islami', The Phenomenon of Irja'a in the Islamic World Available from http://www.alhawali.com/index.cfm?method=home.showcontent&contentID=7
- Aljizani, M. I. H. (2010). 'dirasat wa taħqeeq qaaɛidat 'al'asl fi 'alɛibadah 'almanɛ' (Investigating and Confirming that the Basic Rule in Worshipping practices is Prohibition) Dammam: Dar Ibn Aljawzi.
- Allen, C. (2006). God's Terrorists: The Wahhabi Cult and the Hidden Roots of Modern Jihad. London: Little, Brown.
- Alnadi, A. (2007). Aldolah wa nidham alhukm fi tasawwar alikhwan almoslimeen ' The State and Constitutional law in the Perception of Muslim Brotherhood'. In T. e. a.
 Aldakheel (Ed.), *ALikhwan Almuslimoon: Altaa'sees (Muslim Brotherhood 1-foundation)* (Vol. twelve, pp. 51-74). Dubai: Almesbar Stuies & Research Centre.
- Alomar, N. (no date). *Live Transmission: Facts and Numbers*, 'albath almubashir haqa'q wa arqam ' Available from <u>http://www.almoslim.net/documents/bth.pdf</u>
- Alotaibi, A. B. (2007). Assaroriyah, Sarorism. In T. e. a. Aldakheel (Ed.), Assaroriyyah, 'Sarorism' (Vol. one, pp. 9-39). Dubai: Almesbar Studies & Research Centre.
- Alsibaie, M. (1999). 'almar'ah bain 'alfigh wa 'alqanoon (Woman between Islamic Jurisprudence and Law) (7 ed.). Beirut: Dar Alwarrak.
- Alsinaidi, I. A. (1986). dokool 'alqawaneen 'alwađeiyyah fi masr (Adoption of Man-made Laws in Egypt). Imam University, Riyadh.
- Altwaijiri, H. I. A. (1987). الرد على الكاتب المفتون (arrad 'alā 'lakātib 'almaftūn) Refuting the Thesis of the Misguided Writer. Riyadh: Dar Alliwaa for publication and distribution.
- Awaji, G. b. A. (2001). '*firaq islamiyyah mo'asirah tadda'i alislam'; Modern factions claim Islam* (fourth ed.). Jeddah: Almaktabah al'asriyyah althahabiyyah for printing, publishing and marketing.
- Baker, P. (2006). Using Corpora in Discourse Analysis. London, New York: Continuum
- Baker, P. (2010). Representations of Islam in British broadsheet and tabloid newspapers 1999-2005. Language and Politics, 9(2), 310-338.

- Baker, P., Gabrielatos, C., KhosraviNik, M., Krzyzanowski, M., McEnery, T., & Wodak, R. (2008). A useful methodological synergy? Combining critical discourse analysis and corpus linguistics to examine discourses of refugees and asylum seekers in the UK press. *Discourse and Society*, 19(3), 273-309.
- Baker, P., Gabrielatos, C., & McEnery, T. (2013). Discourse Analysis and Media Attitudes: The representation of Islam in the British Press: Cambridge University Press.
- Baker, P., Hardie, A., & McEnery, T. (2006). *A Glossary of Corpus Linguistics*. Edinburgh: EdinburghUniversity Press.
- Baxter, J. (2003). Positioning Gender in Discourse: A Feminist Methodology. Basingstoke: Palgrave Macmillan.
- Bernstein, B. (1990). The Structuring of Pedagogic Discourse. London: Routledge.
- Bodley, R. V. C. (1946). The Messenger: The Life of Mohammad. New York: Doubleday & Company.
- Brindle, A. J. (2009). A Linguistic Analysis of a White Supermacist Web Forum. PhD, Lancaster University, Lancaster.
- Burr, V. (1995). An Intoduction to Social Constructionism. London: Routledge.

Caldas-Coulthard, C. R. (2003). Cross-Cultural Representation of 'Otherness' in Media Discourse. In G. Weiss & R. Wodak (Eds.), *Critical Discourse Analysis: Theory and Interdisciplinarity* (pp. 272-296). New York: Palgrave Macmillan.

- Charteris-Black, J. (2006). Politicians and Rhetoric: The Persuasive Power of Metaphor: Palgrave Macmillan.
- Chilton, P. (2004). Analysing Political Discourse: Theory and Practice. London: Routledge.
- Chilton, P. A., & Schaffner, C. (Eds.). (2002). Politics as Text and Talk: Analytic approaches to political discourse (Vol. 4). Amsterdam/Philadelphia: John Benjamins Publishin Company.
- Christmann, A. (2008). Reclaiming Mysticism: Anti-Orientalism and the Construction of 'Islamic Sufism' in Postcolonial Egypt In N. Green & M. Searle-Chatterjee (Eds.), *Religion, Language, and Power* (Vol. 10, pp. 57-79). New York Routledge.
- Commins, D. (2006). The Wahabbi Mission and Saudi Arabia. London: I.B Tauris & Co Ltd.
- Cook, M. (2000). Commanding Right and Forbidding Wrong in Islamic Thought. Cambridge: Cambridge University Press.
- Curtin Winsor, J. (2007). Saudi Arabia, Wahhabism and the Spread of Sunni Theofascism. *Mideast Monitor, 2*(1). Retrieved from
- http://www.mideastmonitor.org/issues/0705/0705_2.htm
- Dawkins, R. (2006). The God Delusion. London: Black Swan.
- Dekmejian, R. H. (1994). The Rise of Political Islam in Saudi Arabia. *Middle East Journal,* 48(4), 672. Retrieved from <u>http://pao.chadwyck.co.uk/PDF/1234108545001.pdf</u>
- DeLong-Bas, N. J. (2004). Wahhabi Islam From Revival and Rform to Global Jihad. London: Oxford University press.
- Fairclough, N. (2000). New Labour, New Language? London: Routledge.
- Fairclough, N. (2003). Analysing Discourse: Textual analysis for social research. London and New York: Routledge.
- Fairclough, N., Mulderrig, J., & Wodak, R. (2011). Critical Discourse Analysis. In T. A. van Dijk (Ed.), Discourse Studies: A Multidisciplinary Introduction (2nd ed.). Los Angeles: Sage.
- Fandy, M. (1998). Safar AL-hawali: Saudi Islamist or Saudi Nationalist? Journal of Islam and Christian-Muslim relations, 9(1), 5-21.
- Fandy, M. (2001). Saudi Arabia and the Politics of Dissent. London: Palgrave.
- Field, D. A. (2007). Private Life and Communist Morality in Khushehev's Russia. New York: Peter Lang Publishing.
- Firth, J. R. (1957). Papers in Linguistics 1934-1951. Oxford: Oxford University Press.

- Fitzgerald, T. (2003). Playing language games and performing rituals: religious studies as ideological state apparatus. *Method & Theory in the study of Riligion, 15*, 209-254.
- Green, N. (2008). Idiom, Genre, and the Politics of Self-Description on the Peripheries of Persian. In N. Green & M. Searle-Chatterjee (Eds.), *Religion, Language, and Power* (Vol. 10, pp. 202-220). New York: Routledge.
- Green, N., & Searle-Chatterjee, M. (Eds.). (2008). Religion, Language, and Power (Vol. 10). New York London: Routledge.
- Gregory, J. (2008). Articulating Anglicanism: The Church of England and the Language of the 'Other' in the Eighteenth Century. In N. Green & M. Searle-Chatterjee (Eds.), *Routledge studies in religion* (Vol. 10, pp. 143-164). New York: Routledge.
- Habib, J. S. (1978). Ibn Sa'ud's Worriors of Islam: Ikhwan of Najd and Their Role in the Creation of the Sa'udi Kingdom, 1910-1930 (Vol. XXVII). Leiden E. J. Brill.
- Haddad, G. F. (2004). Albani and His Friends; a Goncise Guid to the Salafi Movement: Aqsa Publications.
- Hajer, M. (1997). The Politics of Environmental Discourse: Ecological Modernization and the Policy Process. . Oxford: Oxford University Press.
- Halliday, M. A. K. (2004). An Introduction to Functional Grammar (Third ed.). London: Hodder Education.
- Hanafi, H. (1988). Athar Alimam Alshaheed Sayyid Qutb Ala Alharakat Alddiniyyah Almo'asirah "The Effect of the Martyr Imam Sayyid Qutb on Modern Islamic Movements" Addeen wa Althorah fi Masr "Religion and Revolution in Egypt" (Vol. Five, pp. 167-200). Cairo: Maktabat Madbooli.
- Hardt-Mautner, G. (Ed.) (1995) UCREL Technical Paper 6. Lncaster University.
- Hegghammer, T. (2008). Islamist Violence and Regime Stability in Saudi Arabia. *international Affairs*, 84(4), 701-715.
- Hegghammer, T. (2009). Jihadi Salafis or Revolutionaries: On Religion and Politics in the Study of Islamist Militancy. In R. Meijer (Ed.), *Global Salfism: Islam's New Religious Movement* (pp. 244-266). New York: Colombia University Press.
- Hegghammer, T., & Lacroix, S. (2007). Rejectionist Islamism in Saudi Arabia: The Story of Juhayman al-'Utaybi Revisited *International Journal of Middle East Studies 39*(1), 103-122.
- Hirst, J. S. (2008). Who Are the Others? Three Moments in Sanskrit-Based Practice. In N. Green & M. Searle-Chatterjee (Eds.), *Religion, Language, and Power* (Vol. 10, pp. 101-122). New York: Routledge.
- Hitchens, C. (2007). God is Not Great: How Religion Poisons Everything. New York: Atlantic Books.
- Hjelm, T. (2013). Religion, Discourse and Power: A Contribution towards a Critical Sociology of Religion. *Critical Sociology*, 0(0), 1-17.
- Hukam Sing, P. K., & Thuraisingam, T. (2011). Language for reconciliation in religious discourse: A critical discourse analysis of contradictions in sermons explored through the activity theory framework. *Multilingua - Journal of Cross-Cultural and Interlanguage Communication 30*(3-4), 391-404.
- Hunston, S. (2002). Corpora in Applied Linguistics. Cambridge: Cambridge University Press.
- Ibn Taymiyyah, A. b. A. (2003). The Compilation of Fatwas of ibn Taymiyyah 'majmooe alfatawa ibn Taymiyya'. In A. i. M. Bin Qasim & M. i. A. Bin Qasim (Eds.) (Vol. 3). Medina, Saudi Arabia: King Fahad Complex for Printing the Glorious Qur'an.
- Ibn Taymiyyah, A. b. A. (2004). The Compilation of Fatwas of Ibn Taymiyyah 'Majmooe Alfatawa Ibn Taymiyya'. In A. i. M. Bin Qasim & M. i. A. B. Qasim (Eds.). Medina, Saudi Arabia: King Fahad Complex for Printing the Glorious Qur'an.
- KhosraviNik, M. (2010). Actor descriptions, action attributions, and argumentation: towards a systematization of CDA analytical categories in the representation of social

groups. Critical Discourse Studies, 7(1), 55-72.

- Klutz, T. (2008). Christianos: Defining the Self in the Acts of the Apostles. In N. Green & M. Searle-Chatterjee (Eds.), *Religion, Language, and Power* (Vol. 10, pp. 167-185). New York: Routledge.
- Lacroix, S. (2009). Between Revolution and Apoliticism: Nasir al-Din al-Albani and his Impact on the Shaping of Contemporary Salfism. In R. Meijer (Ed.), *Global Salfism: Islam's New Religious Movement* (pp. 58-80). New York: Colombia University Press.
- Matar, N. (1999). Turks, Moors and Englishmen in the Age of Discovery. New York: Columbia University Press.
- Mautner, G. (2007). Mining large corpora for social information: The case of *elderly*. *Language in Society*, 36(1), 51–72.
- Mautner, G. (2009). Check and Balances: How Corpus Linguistics can Contribute to CDA. In R. Wodak & M. Meyer (Eds.), *Methods of Critical Discourse Analysis* (second ed.). London: Sage.
- Meijer, R. (Ed.). (2009). *Global Salafism: Islam's New Religious Movement*. New York: Columbia University Press.
- Mittelstaedt, J. D. (2002). A framework for understanding the relationships between religions and markets. *Journal of Macromarketing*, 22(1), 6-18.
- Mordechai, A. (1993). Saudi Arabia: Government, Society and the Gulf Crisis. London and New York: Routledge.
- Naseerah, H. (2007). Sayyid Qutb min altakhyyeel ila attaa'weel "Sayyid Qutb: From Phantasy to Hermeneutics In T. e. a. Aldakeel (Ed.), *ALikhwan Almuslimoon: Altaa'sees (Muslim Brotherhood 1- foundation)* (Vol. twelve, pp. 77-97). Dubai: Almesbar Stuies & Research Centre.
- Niblock, T. (Ed.). (2006). Saudi Arabia Power, Legitimacy and Survival. London: Routledge.
- Oliveti, V. (2001). Terror's Source: The Ideology of Wahhabi-Salafism and its Concequences. Nirmingham: Amadues Books.
- Partington, A. (2010). Modern Diachronic Corpus-Assisted Discourse Studies (MD-CADS) on UK newspapers: an overview of the project. *Corpora* 5(2), 83-108.
- Perelman, C., & Olbrechts-Tyteca, L. (1969). The New Rhetoric: A Treatise on Argumentation (J. Wilkinson & P. Weaver, Trans.). Notre Dame: University of Notre Dame Press.
- Poole, E., & Richardson, J. E. (Eds.). (2006). *Muslims and the News Media*. London: I.B. Tauris.
- Qutb, M. (1980). Jahiliyyat alqarn alɛišreen (the Jahilyyiah of the Twentieth Century). Cairo: Dar Alshuruq.
- Qutb, S. (2001). Milestones (first ed.). Abdul Naeem for Islamic Book Service.
- Rea, T., & Wright, J. (1997). The Arabi-Israeli Conflict. New York: Oxford University Press.
- Reisigl, M., & Wodak, R. (2001). Discourse and discrimination : rhetorics of racism and antisemitism. London Routledge.
- Reisigl, M., & Wodak, R. (2009). The Discourse-Historical Approach (DHA). In R. Wodak
 & M. Meyer (Eds.), *Methods of Critical Discourse Analysis* (second ed., pp. 87-121).
 London: Sage Publication Ltd.
- Richardson, J. (2004). (Mis)representing Islam. Amsterdam: Benjamins.
- Richardson, J. E., & Wodak, R. (2009). Recontextualising fascist ideologies of the past: rightwing discourses on employment and nativism in Austria and the United Kingdom. *Critical Discourse Studies*, 6(4), 251–267.
- Rizwan, S. (2011). Religion, Ideology and Discourse: A Critical Discourse Analysis of Majlis-e-Hussain. *Mathal/Mashal*, 1(1), 1-35.
- Roy, O. (1994). *The Failure of Political Islam* (C. Volk, Trans.). London: Harvard University Press.

- Rubinelli, S. (2009). The Classical Technique of Constructing Arguments from Aristotle to Cicero (Vol. 15): Springer.
- Runciman, S. (1951). A History of the Crusades. Cambridge: Cambridge University Press.
- Ryding, K. C. (2005). A Reference Grammar of Modern Standard Arabic. Cambridge: Cambridge University Press.
- Sager, A. (2005). Political Opposition in Saudi Arabia. In P. AArt & G. Nonneman (Eds.), Saudi Arabia in the Balance; Political Economy, Society, Foreign Affairs (pp. 234-270). London: C. Hurst & Co.
- Said, E. (1978). Orientalism. London: Routledge & Kegan Paul.
- Salama, A. (2011). Ideological Collocation in Meta-Wahhabi Discourse Post-9/11: A Symbiosis of Critical Discourse Analysis and Corpus Linguistics. Lancaster University, Lancaster.
- Schwartz, S. (2002). The Two Faces of Islam: The House of Sa'ud from Tradition to Terror. New York: Doubleday.
- Scott, M. (1999). Wordsmith tools Help Manual (Version 3.0). Oxford: Oxford University Press.
- Scott, M. (2004). Oxford WordSmith Tools version 4. Oxford: Oxford University Press.
- Segal, R. A. (Ed.). (2006). *The Blackwell Companion to the Study of Religion*. Malden: Blackwell Publishing.
- Stubbs, M. (1996). Text and Corpus Analysis (Vol. 23). London: Blackwell.
- Stubbs, M. (1997). Whorf 's Children: Critical Comments on Critical Discourse Analysis(CDA). In A. Rayn & A. Wray (Eds.), *Evolving models of language* (pp. 110-116). Clevendon: Multilingual Matters.
- Suleiman, U. S. R. (2008). Too Many Truths... Too Many Broken Dreams: At-Taqiyya in Arab-Palestinian Political Discourse. Universitat Wien.
- Titscher, S., Meyer, M., Wodak, R., & Vetter, E. (2000). *Methods of Text and Discourse Analysis* (B. Jenner, Trans.). London: Sage Publication.
- van Der Valk, I. (2000). Parliamentary Discourse on Immigration and Nationality. In R. Wodak & T. A. Van Dijk (Eds.), *Racism at the Top: Parliamentary Discourses on Ethnic Issues in Six European States* (Vol. 2, pp. 221-260). Celovec: Drava.
- van Dijk, T. A. (1993). Elite Discourse and Racism. Newbury Park, CA: Sage.
- van Dijk, T. A. (1998). Ideology: A Multidisciplinary Approach. London: Sage.
- van Dijk, T. A. (2011). Discourse and Ideology. In T. A. van Dijk (Ed.), *Discourse Studies: A Multidisciplinary Introduction* (2nd ed., pp. 379-407). Los Angeles: Sage.
- van Eemeren, F., Grootendorst, R., & Henkemans, F. S. (2002). Argumentation Analysis, Evaluation, Presentation. New York, London: Routledge.
- van Eemeren, F., Houtlosser, P., & Henkemans, F. S. (2007). Argumentative Indicators: A Pragma-Dialectical Study (Vol. 12). Dordrecht: Springer.
- van Eemeren, F. H., & Grootendorst, R. (2004). A Systematic Theory of Argumentation The pragma-dialectical approach. Cambridge: Cambridge University Press.
- van Emeren, F. H., & Grootendorst, R. (1992). Argumentation, Communication, and Fallacies, A Pragam-Dialectical Perspective. Hillsdale. NJ: Lawrance Erlbaum Associates.
- Van Emeren, F. H., & Grootendorst, R. (2004). A Systematic Theory of Argumentation The pragma-dialectical approach. Cambridge: Cambridge University Press.
- van Leeuwen, T. (1996). The Representation of Social Actor. In C. R. Caldas-Coulthard & M. Coulthard (Eds.), *Texts and Practice: Readings in Critical Discourse Analysis* (pp. 32-70). London: Routledge.
- van Leeuwen, T. (2007). Legitimation in discourse and communication. *Discourse and Communication*, 1(1), 91-112.

- van Leeuwen, T. (2008). Discourse and Practice: New Tools for Critical Discourse Analysis. New York: Oxford University press.
- Vassiliev, A. (1998). The History of Saudi Arabia. London: Saqi books.
- Walton, D. (1995). A Pragmatic Theory of Fallacy. Tuscaloosa and London: University of Alabama press.
- Walton, D. (2005). Justification of Argument Schemes. Australian Journal of Logic, 3, 1-13.
- Walton, D., Reed, C., & Macagno, F. (2008). Argumentation Schemes. Cambridge: Cambridge University Press.
- Walton, D., & Reed, C. A. (2005). Argumentation Schemes and Enthymemes. *Synthese*, 145(3), 348-.
- Widdowson, H. G. (1995). Discourse Analysis: a Critical review. Language and Literature, 4(3), 157-172.
- Widdowson, H. G. (1998). The Theory and Practice of Critical Discourse Analysis. *Applied Linguistics*, 19(1), 136-151. Retrieved from applij.oxfordjournals.org
- Williams, P., & Chrisman, L. (Eds.). (1993). Colonial Discourse and Post-colonial Theory: A Reader. London: Longman.
- Wodak, R. (1996). The Gensis of Racist Discourse in Austria since 1989. In C. R. Caldas-Coulthard & M. Coulthard (Eds.), *Texts and Practices: Readings in Critical Discourse Analysis* (pp. 107-128). London: Routledge.
- Wodak, R. (1999). Critical Discourse Analysis at the End of the 20th Century. *Research on Language & Social Interaction*, 32(1), 185-193.
- Wodak, R. (2001a). The Discourse Historical Approach. In R. Wodak & M. Meyer (Eds.), Methods of Critical Discourse Analysis (first ed., pp. 63-94). London: Sage.
- Wodak, R. (2001b). What CDA is about a summary of its history, important concepts and its developments. In R. Wodak & M. Meyer (Eds.), *Methods of critical discourse analysis* (pp. 1-13). London: Sage.
- Wodak, R. (2007). Pragmatic and Critical Discourse Analysis: A ross-disciplinary Inquiry. Pragmatics & Cognition, 15(1), 103-255.
- Wodak, R. (2009). The Discourse of Politics in Action: Politics as Usual. New York: Palgrave Macmilan.
- Wodak, R., & Meyer, M. (2009a). Critical Discourse Analysis: History, Agenda, Theory and Methodology. In R. Wodak & M. Meyer (Eds.), *Methods of Critical Discourse Analysis* (second ed., pp. 1-33). Los Angeles: Sage.
- Wodak, R., & Meyer, M. (Eds.). (2009b). *Methods of Critical Discourse Analysis* (Second ed.). London: Sage Publications Ltd.
- Yusuf Ali, A. (2009). The Holy Qur'an: English Translation with Original Arabic Text (3rd ed.). New Delhi: Kitab Bhavan.

APPENDIX A

Table A1 Keywords in Ibn Baz¹⁰³

N	Keyword	Eng. trans.	Texts	Freq.	% F	. Alhawali	% Alhawali	Keyness	Р
	الله	Allah	5	4253	3.255337	994	0.504043996	3803.249023	1.49391E-2
	و	and	5	17867	13.6757831	17215	8.729495049	1973.226563	1.09699E-1
	سبحانه	the Glorified	5	809	0.61922586	8		1409.837524	3.07177E-1
-	عليه	upon him	5	1398	1.07005906	351	0.177987367	1181.897949	5.28908E-1
5	قال	He/he said	5	1100	0.84196347	195	0.098881871	1129.327026	6.08777E-1
5	صلى	blessings (be upon him)/ (he) prayed	5	876	0.6705091	90	0.045637786	1107.981567	6.45802E-1
7	سلم	(may Allah's) peace (be upon)	5	872	0.66744738	97	0.049187392	1075.768066	7.07582E-1
3	נצ (except/ unless	5	515	0.39419198	0		948.8884888	1.04469E-1
)	له	for/to Him/him	5	387	0.29621806	3		680.6802368	2.95816E-1
10	جل	the Majestic	5	424	0.32453864	18		648.805542	3.44252E-1
1	عز	the Exalted in Mighty	5	363	0.27784794	11		580.4790649	4.90196E-1
12	الصلاة	(may) the blessings (of Allah be upon him)/the prayer	5	289	0.22120676	18		413.4133606	1.46471E-1
3	قوله	His/his saying	5	221	0.16915811	1		395.1074524	1.69895E-1
4	بعد	after/ yet	5	209	0.15997305	1		373.1136169	2.05142E-1
5	رسول	messenger	5	270	0.20666375	28	0.014198423	339.875	2.79549E-1
16	تعالى	the Exalted	5	477	0.36510598	151	0.076570064	339.1069641	2.81665E-17
7	النبي	the prophet	4	225	0.17221979	21	0.010648817	292.0751648	4.65427E-1
8	به	with Him/him/ it	5	570	0.43629014	255	0.129307061	288.5444031	4.85064E-17
19	بن	son of	5	265	0.20283664	48	0.024340153	268.4516602	6.20769E-17
20	رسوله	His messenger	5	233	0.17834316	34	0.017240942	259.9049377	6.93925E-17
21	أهل	the people of	5	244	0.18676281	40	0.020283461	258.9854431	7.02468E-17
22	رضي	(may Allah) be pleased with him	5	182	0.13930667	18		232.3206329	1.02562E-16
.3	ما	what	5	1413	1.08154034	1189	0.602925897	222.8648834	1.18751E-16
.4	للله	to Allah	5	267	0.20436748	74	0.037524402	210.0124969	1.4671E-16
5	الرسول	the messenger	5	161	0.12323283	15		209.0583954	1.49122E-16
26	الدعوة	the call (to)	4	216	0.1653310	45	0.022818893	203.4159851	1.64523E-16
27	الآية	the Qur'anic verse/ till the end of the Qur'anic verse	4	129	0.09873934	5		199.8371887	1.75406E-16
.8	السلام	the Peace/ the peace/ the Peace/ peace / Islamic salutation	5	255	0.19518244	78	0.039552748	186.1885834	2.26957E-16
29	الأيات	the Qur'anic verses	5	138	0.10562814	14		174.8026581	2.86588E-16
30	أيضا	too	5	108	0.08266550	3		174.2516479	2.89979E-16
31	الشرك	associationism/ polytheism	5	131	0.10027019	15		159.7037048	4.02759E-16

103. The unusual numbers in column P. could be read as follows: for example 1.49391E-20 indicates that the first digit of the number (1) is located in the 20th place after the decimal point i.e. 0.0000000000000000000149391 and so forth.

32	محمد	Mohammad	5	233	0.17834316	78	0.039552748	157.878006	4.20831E-16
33	بانله	in/with Allah	4	173	0.13241788	41	0.020790547	151.0360565	4.99035E-16
34	نبينا	our Prophet	4	87	0.06659165	1		150.1987	5.09901E-16
35	أله	His family	4	102	0.07807297	7		142.8763885	6.19853E-16
36	الحق	the True/ true/ right	5	242	0.18523195	94	0.047666132	142.7099609	6.22704E-16
37	إنّ	indeed	5	108	0.08266550	10		140.45784	6.63066E-16
38	علا	the Most High	3	76	0.05817202	0		139.8762207	6.74054E-16
39	آمنوا	(they/ those who) believed	5	88	0.06735707	3		138.6416321	6.98195E-16
40	المعنى	the meaning	5	104	0.07960382	10		133.8149567	8.04458E-16
41	أيها	O you (vocative)	5	104	0.07960382	11		130.0518646	9.02761E-16
42	يا	O (vocative)	5	156	0.11940572	40	0.020283461	129.4469604	9.20034E-16
43	الأحاديث	Hadiths (the sayings of the Prophet of	5	83	0.06352996	4		124.3766632	1.08373E-15
		Islam)						121.5700052	1.005751-15
44	السنة	the Sunnah of the Prophet of Islam/ the	5	142	0 100(0002	26	0.017740000	100 0175000	1 0 1 0 2 4 5 1 5
44		Sunnis	5	142	0.10868983	35	0.017748028	120.9175339	1.21834E-15
45	الرسل	the messengers	5	98	0.07501129	11		120.246788	1.24698E-15
46	من	from	5	3948	3.02188348	4734	2.400547743	116.1447983	1.44328E-15
47	إن	if	4	77	0.05893744	4		113.9173737	1.56735E-15
49	الا	not to	5	61	0.04669070	0		112.2648544	1.66874E-15
48	أنهم	indeed they are	5	61	0.04669070	0		112.2648544	1.66874E-15
51	جميعا	all	5	65	0.04975238	1		110.2798386	1.80244E-15
50	عباده	His servants/ worshippers	5	65	0.04975238	1		110.2798386	1.80244E-15
52	أصحابه	his (the Prophet) companions	4	68	0.05204865	2		109.0186539	1.89488E-15
53	ان	if/indeed	5	58	0.04439443	0		106.7428284	2.07835E-15
54	عنه	with him	5	221	0.16915811	106	0.053751174	102.5393066	2.48468E-15
55	الهدى	the guidance/ the right path	5	67	0.05128322	3		101.5890045	2.59097E-15
56	كثيرة	many	5	149	0.11404778	51	0.025861414	98.98375702	2.91525E-15
57	الباطل	the untruth	5	71	0.05434491	5		98.87672424	2.9297E-15
58	عبد	servant/ worshiper	5	149	0.11404778	53	0.026875587	95.60716248	3.42134E-15
59	صحبه	his (the Prophet) companions	5	69	0.05281407	5		95.47122955	3.44409E-15
60	بإحسان	in a benevolent way	5	51	0.03903648	0		93.85842896	3.73004E-15
61	يعلم	know(s)	5	87	0.06659165	14		93.059021	3.88357E-15
62	عليهم	upon them	5	190	0.14543005	88	0.044623613	92.09111023	4.08095E-15
63	العباد	the servants of Allah/ His worshippers	5	67	0.05128322	5		92.0736084	4.08464E-15
64	امر	issue	5	142	0.10868983	51	0.025861414	90.30308533	4.48222E-15
65	لهم	to them	5	252	0.19288617	147	0.074541718	88.0959549	5.05354E-15
66	العظيم	the great/ the Great	5	84	0.06429538	15		85.63043976	5.81263E-15
67	فالواجب	and the obligatory duty	3	46	0.03520938	0		84.65556335	6.15472E-15
68	رواه	narrated by	4	56	0.04286359	3		82.39207458	7.05939E-15
69	دينه	His/his religion	4	73	0.05587575	11		80.31568146	8.05204E-15
70	الناس	the people	5	396	0.30310684	304	0.154154301	79.64291382	8.41337E-15
71	السماو ات	the heavens (skies)	5	48	0.03674022	1		79.5899353	8.44273E-15

12 44 Lis Prophet 5 51 0.03903648 2 78.85894775 8.86222E-15 13 404 Fils attributes 5 89 0.06812249 21 0.01064817 77.8881782 9.4633E-14 15 4.34 His lawfil to 5 83 0.06352996 18 76.37981415 1.05102E-14 16 end ay of 5 110 0.08419634 26 0.01825115 75.9294922 1.0834E-14 17 Eii impernisable is attributes 5 61 0.03903648 3 73.73503876 1.27257E-14 18										
1 0.00000000000000000000000000000000000	72			5	51	0.03903648	2		78.85894775	8.86222E-15
74 Chain His alurbluces 5 42 0.00214769 0 77.29343414 9.8588E-15 75 Lis lawful to 5 83 0.06352996 18 73.7902115 1.05102E-14 76 Lis lawful to 5 83 0.06482154 8 74.0818932 1.24235E-14 77 Eul impermission 5 63 0.04422154 8 74.0818932 1.24235E-14 78 direct (young) to do rightcous good deeds 5 51 0.03903448 3 73.73503576 1.27517E-14 80 other picus ancestors 4 60 0.04592327 7 72.2734466 1.33308E-14 81 ω_4 with them 5 91 0.06965334 25 0.012677164 71.8994293 1.50617E-14 83 ω_4 46 0.04666374 25 0.012677164 71.8994293 1.50617E-14 84 ω_4 41 0.0313827 1 66.0909816 2.04696E-14 85 ω_4 70.02820788 0.012677164 67.93598933 2.04696E-14	73	قل	you say (imperative)	5	89	0.06812249	21	0.010648817	77.88813782	9.46343E-15
75 ω_{44} is la wint to 5 83 0.06332996 18 76.37981415 1.05.102E-14 76 ω_{44} the algously 110 0.08419634 36 0.018255115 75.95204952 1.08343E-14 77 \mathcal{E}^{-14} importance practices \mathcal{I}	74	صفاته	His attributes	5	42	0.03214769	0		77.29343414	
76 μ_{sb} the day of the religiously importises practices direct (yourbs) to do right cours good deeds 5 63 0.04822154 8 74.08189392 1.24235E-14 77 E_{sb} importises practices right cours good deeds 5 51 0.03903648 3 73.73503876 1.27517E-14 79 ω_{sb} dite Nulsims 5 226 0.17298522 138 0.069977939 73.12379456 1.33588E-14 80 ω_{sb} with them 5 91 0.0696334 25 0.012677164 71.98994293 1.46021E-14 81 ω_{sb} with them 5 91 0.0696334 25 0.012677164 71.98994293 1.5047E-14 82 ω_{sb} off filte of (acc.) 4 61 0.04669070 8 70.88607788 1.5047E-14 83 ω_{sb}	75	يجوز	it is lawful to	5	83	0.06352996	18		76.37981415	
77 E-ali innovative practices innovative practices innovative practices innovative practices innovative practices innovative practices innovative practices 5 63 0.04822154 8 74.08189392 1.24235E-14 78 eta innovative practices innovative practices 5 51 0.03903648 3 73.73503876 1.27517E-14 79 innovative practices ingletosus good decds 5 21 0.03903648 3 0.069977939 73.12379456 1.33588E-14 80 ison of 5 116 0.03873887 3 0.021207164 71.9899429 1.54035E-14 81 ω ison of 5 116 0.0468070 8 70.88007788 1.5904595 1.660285325 82 ota father of (acc.) 4 61 0.04745612 9 69.26620483 1.82395E-14 83 ω indusity 5 88 0.06735707 25 0.012677164 67.99598988 2.04696E-14 84 indusity 4 78 0.05970286 19 66.90700531 2.242564E-14 85 induci 5 73 0.0535749 <t< td=""><td>76</td><td>يوم</td><td></td><td>5</td><td>110</td><td>0.08419634</td><td>36</td><td>0.018255115</td><td>75.95294952</td><td></td></t<>	76	يوم		5	110	0.08419634	36	0.018255115	75.95294952	
7^{10} 2^{22} 7^{10} <t< td=""><td>77</td><td>البدع</td><td>impermissible innovative practices</td><td>5</td><td>63</td><td>0.04822154</td><td>8</td><td></td><td>74.08189392</td><td>1.24235E-14</td></t<>	77	البدع	impermissible innovative practices	5	63	0.04822154	8		74.08189392	1.24235E-14
10 <td>78</td> <td>يصلح</td> <td>righteous good deeds</td> <td>5</td> <td>51</td> <td>0.03903648</td> <td>3</td> <td></td> <td>73.73503876</td> <td>1.27517E-14</td>	78	يصلح	righteous good deeds	5	51	0.03903648	3		73.73503876	1.27517E-14
81 ## with them 5 91 0.06965334 25 0.012677164 71.98025313 1.46021E-14 82 wi son of 5 116 0.0873887 43 0.02180472 71.58994293 1.50617E-14 84 #Ly1 the father of (acc.) 4 61 0.04669070 8 69.26620483 1.52935E-14 85 #Ly1 the father of (acc.) 61 0.04745512 9 68.09098316 2.01916E-14 86 mary (nop on him) 5 88 0.06735707 25 0.012677164 67.93598938 2.04696E-14 87 xim from you 4 41 0.03138227 1 67.01797485 2.2248E-14 88 mary (nop on him) 5 73 0.055770286 19 66.90700531 2.24504E-14 89 willim 4 0.02755516 0 66.2505188 2.38495E-14 90 iindeed 4 36 0.02755516 0 0.55272434 65.43347931 2.5758E-14 91 the worship 3 120	79			5	226	0.17298522	138	0.069977939	73.12379456	1.33588E-14
82 10002180472 110002187143 10002180472 11000217214 83 1000171414 10002180472 11000217214 11000217214 84 $1100000000000000000000000000000000000$	80	السلف	the pious ancestors	4	60	0.04592527	7		72.67344666	1.38309E-14
83636464640.0466097087.05070781.5007175411.5007175418494440.04745612969.266204831.82395E-1485620.04745612968.09098162.01916E-1486(may Allah have)5880.06735707250.01267716467.935989382.04696E-1487(may Allah have)5880.057357551666.907005312.2248E-1488Muslim4780.055725516066.20501882.3494E-1489witheglorious/the Generous/generous5730.055875751666.766006472.27415E-1490viindeed4360.02755516066.25051882.38495E-1491 ω they51870.143133781090.05527243465.433479312.5758E-1491 ω the worship31200.9185055510.02586141464.293205262.87776E-1492 ω the/tri calls to5700.053579491666.67620E-1493 ω ω the father of (nom.)4590.045159851061.9414863694 ω ω 0.02527454240.01217007661.990236913.67299E-1495 ω the father of (nom.)4590.03504261270.06439998760.316585544.38926E-1495 ω the father of (nom.)4 <t< td=""><td>81</td><td>بهم</td><td>with them</td><td>5</td><td>91</td><td>0.06965334</td><td>25</td><td>0.012677164</td><td>71.98025513</td><td>1.46021E-14</td></t<>	81	بهم	with them	5	91	0.06965334	25	0.012677164	71.98025513	1.46021E-14
84 $(\lambda)^{-1}$ the Imam562 0.04745612 9 69.26620433 $1.82395E-14$ 85 $(May Allah have)$ mercy (ugon him)588 0.06735707 25 0.012677164 67.93598938 $2.04696E-14$ 86 $(May Allah have)$ mercy (ugon him)588 0.06735707 25 0.012677164 67.93598938 $2.04696E-14$ 87 $(May Allah have)$ mercy (ugon him)441 0.03138227 1 67.01797485 $2.22248E-14$ 88 $Muslim$ 478 0.05970286 19 66.90700551 $2.24504E-14$ 89 $Muslim$ 478 0.05587575 16 66.76600647 $2.27415E-14$ 90 $(May Ching)$ 110 0.0755516 0 66.2505188 $2.38495E-14$ 91 $Muslim$ 436 0.02575516 0 66.2205188 $2.38495E-14$ 92 $Muslim$ 3120 0.0918505 51 0.02586141 65.43347931 $2.5758E-14$ 93 $usexints/worshippers$ 434 0.02602452 62.56962585 $3.43065E-14$ 94 $usexints/worshippers$ 582 0.06276454 24 0.01217076 61.90232611 $3.67992E-14$ 95 uid told590 0.01515985 10 61.46488821 $3.78821E-14$ 95 uid told5337 0.25794699 271 0.137420446 60.5727768 $4.52171E-14$ 96 uid the sound Prophe	82	ابن	son of	5	116	0.08878887	43	0.02180472	71.58994293	1.50617E-14
85 that you may 5 37 0.02832058 0 68.0909816 2.01916E-14 86 0.0275707 25 0.012677164 67.93598938 2.04696E-14 87 0.02832058 0 67.0797485 2.22248E-14 88 0.012677164 67.93598938 2.04696E-14 89 Muslim 4 78 0.05587575 16 66.07000531 2.2248E-14 80 indeed 4 36 0.0275516 0 66.2505188 2.38495E-14 91 0 the worship 3 120 0.0918505 51 0.055272434 65.43347931 2.5758E-14 92 to His 3 120 0.0918505 51 0.025272434 65.43347931 2.5758E-14 93 to His servants/worshippers 4 34 0.02602432 0 62.45573425 3.47201E-14 94 u=u (hefit) calls to 5 70 0.0357049 16 61.94148636 3.66726E-14 95 ohers 5 82 0.06276454 24 0.012170076	83	أبي	the father of (acc.)	4	61	0.04669070	8		70.88607788	1.59405E-14
86(may Allah have) mercy (upon him)5880.06735707250.01267716467.935989382.04696E-14 87 41 0.03138227167.917974852.22248E-14 88 $Muslim$ 4780.059702861966.907005312.24504E-14 89 $Muslim$ 4780.059702861966.907005312.24504E-14 90 $indeed$ 4360.02755516066.25051882.38495E-14 91 $indeed$ 4360.02755516066.25051882.38495E-14 91 $indeed$ 4360.02755516062.5051882.38495E-14 91 $indeed$ 4360.02602432062.50625262.87776E-14 91 $indeed$ 5700.053579491662.455734253.47201E-14 94 $index$ 5700.03750564561.941486363.66726E-14 94 $index$ 5820.06276453240.01217007661.990236913.67999E-14 96 $index$ 4330.02525890060.316885144.38936E-14 97 $indee$ 4330.02525890060.729190834.18921E-14 98 $inde$ 4330.02525890060.316885544.38936E-14 99 $inde$ $inde$ 53.370.25794692710.1374204660.057277684.52171E-14 99 $inde$ </td <td>84</td> <td>الإمام</td> <td>the Imam</td> <td>5</td> <td>62</td> <td>0.04745612</td> <td>9</td> <td></td> <td>69.26620483</td> <td>1.82395E-14</td>	84	الإمام	the Imam	5	62	0.04745612	9		69.26620483	1.82395E-14
a0accmercy (upon him)588 $0.06^{-3} 5/0^{-2} 23$ $0.0126^{7}/164$ $6^{7},93598938$ $2.04896c^{-1}/4$ 87 4 41 0.03138227 1 $6^{7},93598938$ $2.24248c^{-1}/4$ 88 4 78 0.055970286 19 66.90700531 $2.24248c^{-1}/4$ 89 4 36 0.0275516 0 66.2600647 $2.27415c^{-1}/4$ 90 1^{10} indeed436 0.0275516 0 66.2600188 $2.38495c^{-1}/4$ 91 a they5187 0.14313378 109 0.055272434 65.43347931 $2.5758E^{-1}/4$ 91 a they5187 0.025861414 64.29320526 $2.87776c^{-1}/4$ 92 a to His a 0.02602432 0 62.56962585 $3.43065E^{-1}/4$ 93 a a 0.02602432 0 62.56962585 $3.47201E^{-1}/4$ 94 a a 0.02602432 0 61.94148636 $3.66726E^{-1}/4$ 95 a a 0.02575784 24 0.012170076 61.99023691 $3.67999E^{-1}/4$ 96 a 97 a 98 a 99 a a a a a a	85	لعلكم	that you may	5	37	0.02832058	0		68.09098816	2.01916E-14
88Muslim478 0.05970286 19 66.90700531 $2.24504E-14$ 89the glorious/the Generous/generous573 0.05587575 16 66.90700531 $2.24404E-14$ 90the ded436 0.02755516 0 66.2505188 $2.38495E-14$ 91they5187 0.14313378 109 0.055272436 65.43347931 $2.5758E-14$ 92the worship3120 0.09185055 51 0.025861414 64.29320526 $2.87776E-14$ 93servants/worshippers434 0.02602432 0 62.56962585 $3.43065E-14$ 94servants/worshippers582 0.06276454 24 0.012170076 61.90923691 $3.67992E-14$ 95uittold549 0.03750564 5 61.94148636 $3.66726E-14$ 96uitforbade433 0.0252890 0.0212170076 61.90923691 $3.78821E-14$ 97uitthe father of (nom.)459 0.04515985 10 0.616308821 $3.78821E-14$ 97uitwith what/ wherewith5200 0.15308426 127 0.064399987 60.31658554 $4.38936E-14$ 100the sound Prophetic552 0.03980191 7 59.8303668 $4.3887E-14$ 101the sound Prophetic552 0.03980191 7 59.8303668 $4.3887E-14$ 103the sound Prophetic5<	86	رحمه		5	88	0.06735707	25	0.012677164	67.93598938	2.04696E-14
89 10° the glorious/ the Generous/generous5730.055875731666.766006472.27415E-1490 10° indeed4360.02755516066.25051882.38495E-1491 10° they51870.143133781090.05527243465.433479312.5758E-1492 10° 10° 10° 10° $2.5758E-14$ 10° $2.5758E-14$ 93 10° 10° 10° 0.025272434 65.43347931 $2.5758E-14$ 94 10° 10° 10° 0.02602432 0 62.56962585 $3.43065E-14$ 94 10° 10° 10° 0.03750564 5 61.94148636 $3.66726E-14$ 95 10° 10° 10° 0.02525890 0 0.012170076 61.90923691 $3.67999E-14$ 95 10° 10° 10° 10° 10° 10° 10° 10° 10° 96 10° 10° 10° 10° 10° 10° 10° 10° 10° 97 10° 10° 10° 10° 10° 10° 10° 10° 10° 97 10° 10° 10° 10° 10° 10° 10° 10° 98 10° 10° 10° 10° 10° 10° 10°	87	منكم	from you	4	41	0.03138227	1		67.01797485	2.22248E-14
39Generous/generous573 0.03387373 16 $067,0600647$ $2.27413E-14$ 90 $0.01140200000000000000000000000000000000$	88	مسلم	Muslim	4	78	0.05970286	19		66.90700531	2.24504E-14
91 Λ they51870.143133781090.05527243465.433479312.5758E-1492 54347931 the worship31200.09185055510.02586141464.293205262.87776E-1493 616413 61413 61413 614130 61429320526 2.87776E-1494 56473425 $3.43065E-14$ 62.56962585 $3.43065E-14$ 95 516416 570 0.05357949 16 62.45573425 $3.47201E-14$ 96 $5000000000000000000000000000000000000$	89	الكريم		5	73	0.05587575	16		66.76600647	2.27415E-14
92 نهاينا the worship 3 120 0.09185055 51 0.025861414 64.29320526 2.87776E-14 93 iservants/worshippers 4 34 0.02602432 0 62.56962585 3.43065E-14 94 iservants/worshippers 5 70 0.05357949 16 62.45573425 3.47201E-14 95 isold 5 82 0.06276454 24 0.012170076 61.94148636 3.66726E-14 96 isothers 5 82 0.06276454 24 0.012170076 61.90923691 3.67999E-14 97 ist father of (nom.) 4 59 0.04515985 10 61.64088821 3.78821E-14 98 with what/ wherewith 5 200 0.15308426 127 0.064399987 60.31658554 4.38936E-14 100 islam 5 337 0.25794699 271 0.137420446 60.05727768 4.52171E-14 101 islam 5 52 0.03980191 7 59.83603668 4.63887E-14 102 islam 5 52 0.	90	أنّ	indeed	4	36	0.02755516	0		66.2505188	2.38495E-14
93b) His servants/worshippers434 0.02602432 0 62.56962585 $3.43065E-14$ 94 y y (h/it) calls to570 0.05357949 16 62.45573425 $3.47201E-14$ 95 y y told549 0.03750564 5 61.94148636 $3.66726E-14$ 96 y y $others$ 582 0.06276454 24 0.012170076 61.90923691 $3.67999E-14$ 97 y y i the father of (nom.)459 0.04515985 10 61.6408821 $3.78821E-14$ 98 w w i forbade433 0.02525890 0 60.72919083 $4.18921E-14$ 99 w w with what/ wherewith5 200 0.15308426 127 0.064399987 60.31658554 $4.38936E-14$ 100 i w i w min what/ wherewith5 200 0.15308426 127 0.064399987 60.31658554 $4.38936E-14$ 101 k w i h e companions of the $Hadiths$ 5 52 0.03980191 7 59.83603668 $4.63887E-14$ 102 i w i h e sources ors of the c c c c c c $mprohet320.0249348058.888763435.18896E-14103iwihe sources ors of thecccccmprohet5860.06582623290.014705157.87$	91	هم	they	5	187	0.14313378	109	0.055272434	65.43347931	2.5758E-14
1313134 0.0200432 0 $0.2.0362283$ 3.43032514 14141414141414141415101010101010101610101010101010101710101010101010101710101010101010101710101010101010101810101010101010101910101010101010101910101010101010101001111111110111010101010111111011111010101010101111111111111111111111110211111111111111111111111031111111111111111111111111031111111111111111111111111110311111111111111 <td< td=""><td>92</td><td>العبادة</td><td>the worship</td><td>3</td><td>120</td><td>0.09185055</td><td>51</td><td>0.025861414</td><td>64.29320526</td><td>2.87776E-14</td></td<>	92	العبادة	the worship	3	120	0.09185055	51	0.025861414	64.29320526	2.87776E-14
95العبر1005490.03750564561.941486363.66726E-1496خیر مهothers5820.06276454240.01217007661.909236913.67999E-1497خیر he father of (nom.)4590.045159851061.640888213.78821E-1498خیر he father of (nom.)4590.045159851060.729190834.18921E-1499with what/ wherewith52000.153084261270.06439998760.316585544.38936E-14100ithe companions of the Prophet53370.257946992710.13742044660.057277684.52171E-14101ithe sound Prophetic Hadiths5520.03980191759.836036684.63887E-14103ithe resurrection day4450.03444396459.172088625.01571E-14104ithe sisues5860.06582623290.0147055157.870399485.88181E-14106ithe issues5860.06582623290.0147055157.870399485.88181E-14105ithe jaries5610.04669701356.690296176.84809E-14106ito you5610.04669701355.066453557.54762E-14107ito you5610.04669701355.06453557.54762E-14108ito you5610.04669701355.066453557.54762E-14109	93	لعباده		4	34	0.02602432	0		62.56962585	3.43065E-14
96 $\exists z_{1}$ $\exists z_{1}$ z_{2} 0.06276454 24 0.012170076 61.90923691 $3.67999E-14$ 97 z_{1} z_{1} z_{2}	94	يدعو	(he/it) calls to	5	70	0.05357949	16		62.45573425	3.47201E-14
979794966260.045159851061.640888213.78821E-149898979897	95	أخبر	told	5	49	0.03750564	5		61.94148636	3.66726E-14
98 60.0251 60.02525890 0 60.72919083 $4.18921E-14$ 99 100 with what/ where with 5 200 0.15308426 127 0.064399987 60.31658554 $4.38936E-14$ 100 101 1000 1000	96	غير هم	others	5	82	0.06276454	24	0.012170076	61.90923691	3.67999E-14
99La with what/ wherewith5200 0.15308426 127 0.064399987 60.31658554 $4.38936E-14$ 100the companions of the Prophet548 0.03674022 5 60.29718781 $4.39908E-14$ 101 μ Islam5337 0.25794699 271 0.137420446 60.05727768 $4.52171E-14$ 102the sound Prophetic Hadiths552 0.03980191 7 59.83603668 $4.63887E-14$ 103the resurrection day the successors of the companions of the Prophet432 0.02449348 0 58.88876343 $5.18896E-14$ 104Uthe issues586 0.06582623 29 0.01470551 57.87039948 $5.88181E-14$ 105Lord/ may561 0.04669070 13 56.69029617 $6.84809E-14$ 106 μ Lord/ may561 0.04669070 13 56.69029617 $6.84809E-14$ 107 μ Lord/ may561 0.04669070 13 56.69029617 $6.84809E-14$ 107 μ to you561 0.04669070 13 56.69029617 $6.84809E-14$ 108 μ μ 57.96645355 $7.54762E-14$ 111 $4.1424E-14$ 109 μ μ 57.113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 108 μ μ 58.5060808 30 0.015212596 54.91756439 $8.74091E-14$ <td>97</td> <td>أبو</td> <td>the father of (nom.)</td> <td>4</td> <td>59</td> <td>0.04515985</td> <td>10</td> <td></td> <td>61.64088821</td> <td>3.78821E-14</td>	97	أبو	the father of (nom.)	4	59	0.04515985	10		61.64088821	3.78821E-14
100 $Har mar mar mar mar mar mar mar mar mar m$	98	نهى	forbade	4	33	0.02525890	0		60.72919083	4.18921E-14
ProphetProphet543 0.05074022 5 0.0137420446 0.05074024 0.0137420446 0.05727768 $4.52171E-14$ 101 μ Islam5 337 0.25794699 271 0.137420446 60.05727768 $4.52171E-14$ 102the sound Prophetic5 52 0.03980191 7 59.83603668 $4.63887E-14$ 103the resurrection day4 45 0.03444396 4 59.17208862 $5.01571E-14$ 104the resurrection day4 32 0.02449348 0 58.88876343 $5.18896E-14$ 105companions of the Prophet4 32 0.02449348 0 58.88876343 $5.18896E-14$ 105the issues5 86 0.06582623 29 0.01470551 57.87039948 $5.88181E-14$ 106Lord/may5 61 0.04669070 13 56.69029617 $6.84809E-14$ 107 μ to you5 61 0.04669070 13 56.69029617 $6.84809E-14$ 108aaapraise5 67 0.05128322 17 55.96645355 $7.54762E-14$ 108aaapraise5 74 0.05664117 22 0.026368501 55.18482971 $8.41424E-14$ 109aaaathe sheikh5 113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 110the sheikh5 113 0.0650608 30 0.015212596 <	99	بما	with what/ wherewith	5	200	0.15308426	127	0.064399987	60.31658554	4.38936E-14
101Islam5337 0.25794699 271 0.137420446 60.05727768 $4.52171E-14$ 102the sound Prophetic552 0.03980191 7 59.83603668 $4.63887E-14$ 103the resurrection day445 0.03444396 4 59.17208862 $5.01571E-14$ 104the successors of the companions of the Prophet32 0.02449348 0 58.88876343 $5.18896E-14$ 105UracleLord/may561 0.04669070 13 56.69029617 $6.84809E-14$ 106UracleUracle561 0.04669070 13 56.69029617 $6.84809E-14$ 106UracleUracle561 0.04669070 13 56.69029617 $6.84809E-14$ 107Lord/may561 0.04669070 13 56.69029617 $6.84809E-14$ 108UracleUracle567 0.05128322 17 55.96645355 $7.54762E-14$ 108UracleUracle5113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 109AttS113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 110UracleS85 0.06506080 30 0.015212596 54.91756439 $8.74091E-14$ 111UracleUracleS85 0.02501665 3.5490044403 $8.76241E-14$	100	الصحابة		5	48	0.03674022	5		60.29718781	and the second s
Hadiths552 0.03900191 7103Hadiths5 52 0.03900191 7103the resurrection day445 0.03444396 4 59.17208862 $5.01571E-14$ 104the successors of the Prophet32 0.02449348 0 58.88876343 $5.18896E-14$ 105prophet105prophet100.0466907013 56.69029617 $6.84809E-14$ 106Lord/may561 0.04669070 13 56.69029617 $6.84809E-14$ 107praise567 0.05128322 17 55.96645355 $7.54762E-14$ 108praise567 0.05128322 17 55.96645355 $7.54762E-14$ 109with them574 0.05664117 22 0.011155903 55.20428848 $8.3911E-14$ 110the sheikh5113 0.08649261 52 0.0226368501 55.18482971 $8.41424E-14$ 110the sheikh5113 0.06506080 30 0.015212596 54.91756439 $8.76241E-14$	101	الإسلام	-	5	337	0.25794699	271	0.137420446	60.05727768	4.52171E-14
103the resurrection day445 103 11350 113 104 the successors of the companions of the Prophet32 0.02449348 0 58.88876343 $5.18896E-14$ 105 prophetrophet10 114000 10 57.87039948 $5.88181E-14$ 106 Lord/ may561 0.04669070 13 56.69029617 $6.84809E-14$ 106 Lord/ may561 0.04669070 13 56.69029617 $6.84809E-14$ 107 $praise$ 567 0.05128322 17 55.96645355 $7.54762E-14$ 108 praise567 0.05664117 22 0.011155903 55.20428848 $8.3911E-14$ 109 with them574 0.05664117 22 0.026368501 55.18482971 $8.41424E-14$ 110 the sheikh5113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 111 then He/he said585 0.06506080 30 0.015212596 54.91756439 $8.76241E-14$	102	الصحيحة		5	52		7			
104دەשף האפוןנדוא בעניcompanions of the Prophet432 0.02449348 0 58.88876343 $5.18896E-14$ 105Prophethe issues586 0.06582623 29 0.01470551 57.87039948 $5.88181E-14$ 106Lord/ may561 0.04669070 13 56.69029617 $6.84809E-14$ 107Lord/ wou561 0.04669070 13 56.69029617 $6.84809E-14$ 108praise567 0.05128322 17 55.96645355 $7.54762E-14$ 109with them574 0.05664117 22 0.011155903 55.20428848 $8.3911E-14$ 110the sheikh5113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 111then He/he said585 0.06506080 30 0.015212596 54.91756439 $8.76241E-14$	103	القيامة		4	45		4			
105نال لله issues586 0.06582623 29 0.01470551 57.87039948 $5.88181E-14$ 106 Lord/may561 0.04669070 13 56.69029617 $6.84809E-14$ 107 Lord/may561 0.04669070 13 56.69029617 $6.84809E-14$ 107 Lord/may561 0.04669070 13 56.69029617 $6.84809E-14$ 108 Image: praise567 0.05128322 17 55.96645355 $7.54762E-14$ 109 Image: with them574 0.05664117 22 0.011155903 55.20428848 $8.3911E-14$ 110 Image: with them5113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 111 Image: then He/he said585 0.06506080 30 0.015212596 54.91756439 $8.76241E-14$	104	التابعين	companions of the	4	32	0.02449348	0			
100 Lord/may 5 61 0.04009070 13 56.69029617 6.84809E-14 107 مل to you 5 61 0.04669070 13 56.69029617 6.84809E-14 108 praise 5 67 0.05128322 17 55.96645355 7.54762E-14 109 with them 5 74 0.05664117 22 0.011155903 55.20428848 8.3911E-14 110 with them 5 113 0.08649261 52 0.026368501 55.18482971 8.41424E-14 111 the sheikh 5 85 0.06506080 30 0.015212596 54.91756439 8.76241E-14	105	الأمور		5	86	0.06582623	29	0.01470551		
107to you561 0.04669070 13 56.69029617 $6.84809E-14$ 108praise567 0.05128322 17 55.96645355 $7.54762E-14$ 109with them574 0.05664117 22 0.011155903 55.20428848 $8.3911E-14$ 110with thesheikh5113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 111the sheikh585 0.06506080 30 0.015212596 54.91756439 $8.76241E-14$ 111then He/he said585 0.02016185 2 54.90044403 $8.76241E-14$	106	رب	Lord/ may	5	61					
100praise567 0.05128322 17 100 praise567 0.05128322 17 100 with them574 0.05664117 22 0.011155903 55.20428848 $8.3911E-14$ 110 the sheikh5113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 111 the sheikh585 0.06506080 30 0.015212596 54.91756439 $8.74091E-14$ 111 then He/he said585 0.02061685 2 54.90044403 $8.76241E-14$	107	لكم	to you	5	61	0.04669070				
109 25 74 0.03064117 22 0.01110500 $8.41424E-14$ 110 113 0.08649261 52 0.026368501 55.18482971 $8.41424E-14$ 111 111 113 0.06506080 30 0.015212596 54.91756439 $8.74091E-14$ 111 111 112 113 0.06506080 30 0.015212596 54.90044403 $8.76241E-14$	a state of the second	الحمد		5	67					
10 5 113 0.03049201 52 0.001201000 111 the sheikh 5 85 0.06506080 30 0.015212596 54.91756439 8.74091E-14 111 then He/he said 5 85 0.06506080 30 0.015212596 54.91756439 8.74091E-14	and the second	reic	with them	5	74					
then He/he said 5 85 0.0050000 50 50 50 50 54 90044403 8.76241E-14	BERSON -	الشيخ	the sheikh	5	113					
Imams 4 40 0.03061685 3 54.90044403 8.70241E-14 النمة النمة	125.76		then He/he said	5				0.015212596		
	112	أنمة	Imams	4	40	0.03061685	3		54.90044403	0.70241114

114		his Y and							
114	ربه الأدلة	his Lord	4	40	0.03061685	3		54.90044403	8.76241E-14
113		the proofs	4	54	0.04133275	10		54.06848145	9.89798E-14
115	الصفات	the attributes (of Allah)	4	46	0.03520938	6		53.56055069	1.06892E-13
116	ثبت	it became firm	4	46	0.03520938	6		53.56055069	1.06892E-13
117	أحسن	more suitable/ best	4	62	0.04745612	15		53.41001511	1.00892E-13
118	المبين	obvious	5	29	0.02219721	0		53.36754227	1.1012E-13
119	ذلك	that	5	990	0.75776714	1086	0.550695956	52.6259613	1.23804E-13
120	رحمة	mercy	5	56	0.04286359	12	0.550095950	51.8814621	1.23804E-13
121	كتاب	book/ the Book (the Qur'an)	5	143	0.10945525	82	0.041581094	51.36889267	1.5992E-13
122	اليه	to Him/him	5	192	0.14696089	130	0.065921247	51.14474487	1.58744E-13
123	القرآن	the Qur'an	5	102	0.07807297	47	0.023833066	49.7303009	2.05532E-13
124	منديمه	in his book of sound Hadiths	3	27	0.02066637	0	0.025055000	49.68677139	2.07247E-13
125	المستقيم	straight (way)	5	34	0.02602432	2		49.15423203	2.29873E-13
		(has) ordained/ the		5.	0.000000100	-		47.13425205	2.27075E-15
126	شرع	canonical law of Islam (Sharia)	4	47	0.03597480	8		49.00533676	2.36784E-13
127	جميع	all	4	135	0.10333187	78	0.039552748	47.90705872	2.9743E-13
128	حديث	Hadith	4	48	0.03674022	9		47.76143646	3.06972E-13
129	K	no/ not	5	1506	1.15272450	1786	0.905656576	47.59189987	3.18603E-13
130	خير	good	5	94	0.07194960	42	0.021297634	47.55235291	3.214E-13
131	العزيز	the Exalted in power/	5	46	0.03520938	8		47.48233795	3.26433E-13
132	لما	when	5	173	0.13241788	116	0.05882204	46.99156952	3.64842E-13
133	إنّ	indeed	3	25	0.01913553	0		46.00604248	4.62264E-13
134	يوفقنا	(to) grant us success	3	25	0.01913553	0		46.00604248	4.62264E-13
135	بالحق	truly/ with the truth	5	37	0.02832058	4		45.94203186	4.69738E-13
136	أنزل	sent	5	96	0.07348045	45	0.022818893	45.82517624	4.83801E-13
137	أوجب	made obligatory	4	32	0.02449348	2		45.70882416	4.98368E-13
138	الحذر	caution	4	41	0.03138227	6		45.65147018	5.05764E-13
139	الثقلين	the humans and the	3	24	0.01837011	0		44.16568756	7.61517E-13
142	امرنا	ordered us/ (our) guardian	4	24	0.01837011	0		44.16568756	7.61517E-13
141	لقول	because of his words	4	24	0.01837011	0		44.16568756	7.61517E-13
140	شينا	thing	5	31	0.02372806	2		43.99154663	8.02032E-13
143	الواجب	the compulsory (duty)	3	52	0.03980191	13		43.85921478	8.34763E-13
144	الاستغاثة	beseeching for help	3	28	0.02143179	1		43.84391403	8.38661E-13
145	النار	the Fire	5	58	0.04439443	17		43.73765182	8.66418E-13
146	الصالح	pious	4	39	0.02985143	6		42.52970123	1.28911E-12
147	القادر	Almighty	5	37	0.02832058	5		42.51114655	1.29756E-12
148	محمدا	Mohammad (acc.)	5	23	0.01760469	0		42.32534409	1.38638E-12
149	ارسل	sent	5	30	0.02296263	2		42.27819061	1.41021E-12
150	بصيرة	discernment	4	30	0.02296263	2		42.27819061	1.41021E-12
151	إخلاص	sincerity	4	27	0.02066637	1	0.110544076	42.0749855	1.51937E-12
152	الذين	who (plural)	5	261	0.19977496	218	0.110544868	41.79287338	1.69043E-12 1.71062E-12
153	غيره	other	5	91	0.06965334	44	0.022311807	41.76208878	2.27176E-12
154	الشرع	Allah's decree	4	34	0.02602432	4		41.06169891	2.27176E-12 2.27176E-12
156	كنتم	you were	4	34	0.02602432	4		41.06169891	2.27176E-12 2.27176E-12
157	وجوب	obligatory duty	4	34	0.02602432	4		41.06169891	2.27176E-12 2.93331E-12
155	انزلنا	(We have) sent down	4	22	0.01683927	0		40.48501205	2.9555112-12

1601210140,485012159نظار10Youyou4260.01990095140,485012161varning4310.02372806339.804073162vibbecause51580.120936571100.0557795239.720138163varning52250.172219701830.09279683238.845767164vause52550.01913553138.545299165obeying Him3250.01913553138.45299166obeying Him3250.01913553138.45299166obeying Him3250.01913553138.45299166arbitration3300.02296263338.151336167jath/belief51350.1033187890.0451307169jathhet guardian of4360.0255516637.899330170ith/belief3240.01837011136.784973171security3240.01837011136.784973172security3240.01837011136.301355176caycompanion of the rankind/4290.02219721336.604673173caecamartade in anlawful3300.02226800536.21875174caysamartade in anlawful3300.022298005 </th <th><u></u></th> <th></th> <th>the first and</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>	<u></u>		the first and							
1510101040.885012161warning4310.02372806339.804073162 \vee Warning4310.02372806339.804073163 \vee Ireligious knowledge52250.172219791830.09279683238.845767164 \vee religious knowledge52250.01913553138.545299165 \vee bet wo books of sound Hadiths3250.01913553138.545299166 \vee bet wo books of sound Hadiths30.02296263338.15136167 \vee bet wo books30.002296263338.15136168 \vee faith/ belief51350.1033187890.0451307170 \vee faith/ belief51350.1033082036.804370171 \vee generous/ most Generous4200.01530842036.804370172generous/ most Generous240.01837011136.784973174 \vee companion of the Prophet of Islam)4290.02219721336.604673175 \vee companion of the Hat exists)330.02525890536.21875176 \vee companion of the Hat exists)400.03061685936.02216663177 \vee companion of the Hat exists)330.022272666436.301355176 \vee companion of the Hat exists)25310.02372806436.301355 <td></td> <td>ربهم</td> <td>their Lord</td> <td>3</td> <td>22</td> <td>0.01683927</td> <td>0</td> <td></td> <td>40.48501205</td> <td>2.93331E-12</td>		ربهم	their Lord	3	22	0.01683927	0		40.48501205	2.93331E-12
161 431 0.01372806 3 $40.308/12$ 162 1002372806 3 39.84073 162 1002372806 3 39.84073 163 110 0.05577952 39.72013 164 1100 000587 110 0.02577952 165 1100 000587 110 0.022796832 166 1100 00005867 1100 0002796832 166 1100 325 0.01913553 1 167 0000916 300 002296263 3 168 10000916 300 002296263 3 169 10000916 $13000000000000000000000000000000000000$							0		40.48501205	2.93331E-12
1621580.122/19031000.0557795239.00107163religious knowledge52250.172219791830.09279683238.845767164religious knowledge52250.172219791830.09279683238.845767165religious knowledge52250.01913553138.545299166sound Hadiths3250.01913553138.545299166arbitration3300.02296263338.151336168iuiuyfaith/belief51350.10333187890.0451307169uiuthe guardian of4360.022755516637.899330170generous/most4200.01530842036.804370171generous/most200.01837011136.784973173security3240.01837011136.784973174rephet of Islam)rephet of Islam)4480.036740221436.3301351176uulawful3330.02252890536.21875177cual Hadith3330.02525890536.21875178indicated/ informed4260.0190095235.4695241178indicated/ informed4260.0190095235.4695241178indicated/ informed4260.0190095235.4695241178indicated/ informed <td></td> <td></td> <td></td> <td></td> <td></td> <td>0.01990095</td> <td>1</td> <td></td> <td>40.30871201</td> <td>3.18616E-12</td>						0.01990095	1		40.30871201	3.18616E-12
163religious knowledge5225 0.120301 110 $0.0537/952$ 39.2013 164sound Hadiths325 0.01913553 1 38.84567 165obeying Him325 0.01913553 38.545299 166 $A=11$ polytheists (acc.)330 0.02296263 3 38.151336 167 $I=01$ polytheists (acc.)330 0.02296263 3 38.151336 168 $I=01$ faith/belief5 135 0.1033187 89 0.0451307 37.908443 169 $I=020875516$ 6 37.899330 36.990222 170 $I=02877866$ 20 0.0153042 0 36.804370 172 $I=02877866$ 420 0.01837011 1 36.784973 174 $I=028778666$ $I=0287786666666666666666666666666666666666$			e			0.02372806	3		39.80407333	4.09158E-12
164the two books of sound Hadiths325 0.01913553 1 38.545299 165obeying Him325 0.01913553 1 38.545299 166arbitration330 0.02296263 38.151336 167uplytheists (acc.)330 0.02296263 38.151336 168ibeying Him5135 0.1033187 89 0.0451307 169the guardian of436 0.0275516 6 37.899330 170the guardian of420 0.01530842 0 36.804370 171generousgenerous420 0.01530842 0 36.804370 172generousgenerous324 0.01837011 1 36.784973 173Hurairah (a1 36.784973 Hurairah (a 33 0.02219721 3 36.504673 174Szamcompanion of the429 0.02219721 3 36.504673 175kascompanion of the429 0.02219721 3 36.21875 176unlawful333 0.02525890 5 36.21875 177ti was narrated in sound Hadith3 30.02226661 8 35.5554311 178is was narrated in sound Hadith 3 30.01760469 34.9640655 178is knew 5 38.02290601 8 35.5554311 178is knew 5 38.02290651 8 35.55		-			158	0.12093657	110	0.05577952	39.72013855	4.27435E-12
164sound Hadiths325 0.01913553 138.545299165obeying Him325 0.01913553 138.545299166arbitration330 0.02296263 338.151336167izyanpolytheists (acc.)330 0.02296263 338.151336168izyanfaith/belief5135 0.1033187 89 0.0451307 37.908443169izyanthe guardian of436 0.02755516 637.899330170izyangenerous/most20 0.01530842 036.804370172generous/most420 0.01837011 136.784973174recous424 0.01837011 136.784973174repropher of Islam)reprohe of Islam)429 0.02219721 336.504673175ix was narrated in sound Hadith333 0.02525890 536.21875178ix was narrated in sound Hadith333 0.02525890 536.21875178ix exw538 0.02908601 835.5554312178indicated/ informed23 0.01760469 $0.34.9640655$ 179ix exw5147 0.1454300 $0.34.9640655$ 180indicated/ informed19 0.01454300 $0.34.9640655$ 181indicated/ informed19 0.01454300 $0.34.9640655$ 182indi	53	العلم		5	225	0.17221979	183	0.092796832	38.84576797	7.03076E-12
166عائة $30,0.0229623$ $30,0.0259330$ $30,0.025929222$ $30,0.0229623$ $30,0.025929222$ $30,0.0229623$ $30,0.025929222$ $30,0.0229623$ $30,0.025929222$ $30,0.029129721$ <t< td=""><td></td><td>0</td><td>sound Hadiths</td><td>3</td><td></td><td>0.01913553</td><td>1</td><td></td><td>38.54529953</td><td>8.51927E-12</td></t<>		0	sound Hadiths	3		0.01913553	1		38.54529953	8.51927E-12
167المدركي167المدركي167المدركي168					25	0.01913553	1		38.54529953	8.51927E-12
1681681611/190173173173173174174174175175175175175175175175175176177177174175175176177173173173173173173173173173173174174174174174174175176177173173173173173173173174174174174174174174174174174174175176177177177178177178173173173173174174174174174175176176177176177177178178178177178178178177178178178178178177178 <td></td> <td></td> <td></td> <td>3</td> <td>30</td> <td>0.02296263</td> <td>3</td> <td></td> <td>38.15133667</td> <td>1.11822E-11</td>				3	30	0.02296263	3		38.15133667	1.11822E-11
169the guardian of4360.02755516637,89030170the believers4470.035974801336.990222171egenerous/most4200.01530842036.804370172security3240.01837011136,784973174iscurity3240.01837011136,784973173iscurity3240.01837011136,784973174iscurity3240.01837011136,784973175companion of the Prophet of Islam)4290.02219721336,504673175iscurityguaranic verse5310.02372806436,3301351176it was narrated in sound Hadith3330.02525890536,21875178it was narrated in sound Hadith5400.03061685936,0221366179it was narrated in sound Hadith5380.02908601835,5554313180indicated/ informed4260.0190095235,4095243181indicated/ informed4260.0190095235,4095243182istic informed4190.01454300034,96406553184ibp and pinous3190.01454300034,96406553183alone51470.112516931050.05324408834,9597055183a					30	0.02296263	3		38.15133667	1.11822E-11
170العالي170العالي171العالي171172171173173173174200.01530842036.804370174iscurity3240.01837011136.78497336.784973174iscurity3240.01837011136.784973174iscurity3240.01837011136.784973174iscurity3240.01837011136.784973175iscurity4290.02219721336.504673176indeful4480.036740221436.3301351176indeful3330.02525890536.21875178it was narrated in sound Hadith3330.02525890536.21875178it was narrated in sound Hadith3330.02525890536.21875178indicated/ informed4260.0199095235.4695243180indicated/ informed4260.0199095235.4095243181indicated/ informed3190.01454300034.9640655184ispanimous3190.01454300034.9640655184ispanimous3190.01454300034.9640655183ispanimous3190.01454300034.9640655184ispanimous3190.01454300034.9640655<		الإيمان	faith/ belief	5	135	0.10333187	89	0.0451307	37.90844345	1.33992E-11
171عوال (most Generous)4200.01530842036.804370172security3240.01837011136.784973174security3240.01837011136.784973174expanion of the 4240.01837011136.784973173prophet of Islam)prophet of Islam)336.504673175 $Prophet of Islam$ 4290.02219721336.504673176unlawful4480.036740221436.3301353176gur'anic verse5310.02372806436.2373693177sound Hadith3330.02525890536.21875178mankind/ jinn and all that exists)5400.03061685936.0221366180indicated/ informed4260.0199095235.46952431814the heaven (sky)4190.01454300034.9640655182good/ better4190.01454300034.9640655183alone51470.112516931050.05324408834.9272308183hence51470.112516930.02535432834.9272308184upsellion5233.78020820183alone51260.09644308850.04310235433.802820184upsellion3250.013553233.763166 <tr<< td=""><td>59</td><td>ولي</td><td>the guardian of</td><td>4</td><td>36</td><td>0.02755516</td><td>6</td><td></td><td>37.89933014</td><td>1.34934E-11</td></tr<<>	59	ولي	the guardian of	4	36	0.02755516	6		37.89933014	1.34934E-11
1/1 $G_{Generous}$ 420 0.01530842 0 36.804370 1/2security324 0.01837011 1 36.784973 1/2Hurairah (a $4 = 24$ 0.01837011 1 36.784973 1/4soupanion of the Prophet of Islam)29 0.02219721 3 36.504673 1/3(blessed) end429 0.02372806 4 36.3301352 1/5 v_{res} (has) forbidden/made unlawful448 0.03674022 14 36.3301352 1/6 v_{res} v_{res} 531 0.02372806 4 36.2373692 1/7 v_{res} it was narrated in sound Hadith3 33 0.02525890 5 36.221366 1/7 v_{res} knew5 38 0.02908601 8 35.5554312 1/8 v_{res} v_{res} 5 30.01760469 35.0279966 1/8 v_{res} v_{res} 31.9000955 2 35.4695242 1/8 v_{res} 31.90009552 34.96406552 34.96406552 1/8 v_{res} 31.90001760469 34.96406552 34.96406552 1/8 v_{res} 31.9000055 30.02535428 34.9272308 1/8 v_{res} s_{res} $51.7000000000000000000000000000000000000$	70	المؤمنين		4	47	0.03597480	13		36.99022293	2.99212E-11
Hurairah (a ecompanion of the Prophet of Islam)LOttorstortIDerivity of tecompanion173فریر 3 companion of the Prophet of Islam)4240.01837011136.784973173(blessed) end4290.02219721336.504673175 a_{cd} (has) forbidden/ made unlawful4480.036740221436.3301352176Cur'anic verse5310.02372806436.2373692177cit was narrated in sound Hadith mankind/ jinn and all that exists)5400.03061685936.0221366178indicated/ informed4260.01990095235.4695242180curindicated/ informed4260.0190095235.4695242181magnanimous3190.01454300034.9640655184jgood/ better4190.01454300034.9640655185alone5910.06965334500.02535432834.957753183alone5910.06965334500.02535432834.9272308187curreligion51260.09644308850.04310235433.8020820183alone5910.06965334500.02535432834.9272308184jaolne5910.06965334500.02535432834.9272308185jaolne5910.06965	71	کریم		4	20	0.01530842	0		36.80437088	3.6273E-11
174ديرةcompanion of the Prophet of Islam)4240.01837011136.784973173نجاب(blessed) end4290.02219721336.504673175 c_{abs} forbidden/made unlawful4480.036740221436.3301357176 c_{als} Qur'anic verse5310.02372806436.2373692177 c_{ass} it was narrated in sound Hadith3330.02525890536.21875178mankind/ jinn and all that exists)5400.03061685936.0221366179knew5380.02908601835.5554313180indicated/ informed4260.01990095235.4695243181indicated/ informed4260.01454300034.9640655182good/ better4190.01454300034.9640655184jgood/ better4190.01454300034.9640655185alone51470.112516931050.05324408834.9597053183alone5910.06965334500.0235432833.8020820184jalone51260.09644308850.04310235433.8020820183alone51260.0964338850.04310235433.8020820184jalone51260.0964334850.04310235433.80208201	72	السلامة		3	24	0.01837011	1		36.78497314	3.70362E-11
173(blessed) end429 0.02219721 3 36.504673 175(has) forbidden/made unlawful448 0.03674022 14 36.3301352 176Qur'anic verse531 0.02372806 4 36.2373692 177it was narrated in sound Hadith that exists)3 30.02525890 5 36.21875 178mankind/ jinn and all that exists)540 0.03061685 9 36.0221366 179 ω_{xw} 538 0.02908601 8 35.5554312 180indicated/ informed426 0.0199095 2 35.4695242 181 ω_{xw} 323 0.01760469 1 35.0279966 182etis/his path/way323 0.01454300 34.96406552 184 ω_{xw} 5147 0.11251693 105 0.053244088 185 ω_{xy} ω_{xy} 19 0.0454300 0.025354328 34.9272308 185 ω_{xy} ω_{xy} 126 0.0964334 50 0.025354328 34.9272308 187 ω_{xy} ω_{xy} 126 0.0964334 50 0.025354328 34.9272308 188 ω_{xy} ω_{xy} 126 0.0964334 50 0.025354328 34.9272308 189 ω_{xy} ω_{xy} 126 0.0964334 50 $0.033.73631666$ 189 ω_{xy} ω ω ω ω ω ω ω </td <td>74</td> <td>هريرة</td> <td>companion of the</td> <td>4</td> <td>24</td> <td>0.01837011</td> <td>1</td> <td></td> <td>36.78497314</td> <td>3.70362E-11</td>	74	هريرة	companion of the	4	24	0.01837011	1		36.78497314	3.70362E-11
173 2 unlawful448 0.03674022 14 36.330135 176Qur'anic verse531 0.02372806 4 36.2373692 177 2 it was narrated in sound Hadith333 0.02525890 5 36.21875 178mankind/ jinn and all that exists)5 40 0.03061685 9 36.0221366 179 2 knew5 38 0.02908601 8 35.5554312 180 310 indicated/ informed4 26 0.0199095 2 35.4695242 181 418 419 0.01454300 0 34.9640655 182magnanimous3 19 0.01454300 0 34.9640655 186 12 good/ better4 19 0.01454300 0 34.9640655 186 12 good/ better4 19 0.01454300 0.025354328 34.957053 183 $alone$ 5 91 0.06965334 50 0.025354328 34.9272308 187 2 religion 5 126 0.09644308 85 0.043102354 33.8020820 188 40 0.03061685 10 33.7363166 33.7363166 189 14 40 0.03061685 10 33.7363166 189 14 40 0.03061685 10 33.7363166 180 40 35 0.02678974 7 33.6779896 181 40 0.02678974 <td< td=""><td>73</td><td>العاقبة</td><td>(blessed) end</td><td>4</td><td>29</td><td>0.02219721</td><td>3</td><td></td><td>36.504673</td><td>5.0914E-11</td></td<>	73	العاقبة	(blessed) end	4	29	0.02219721	3		36.504673	5.0914E-11
177 i was narated in sound Hadith333 0.02525890 5 36.21875 178 i mankind/jinn and all that exists)5 40 0.03061685 9 36.0221366 179 i knew5 38 0.02908601 8 35.5554312 180 i indicated/ informed4 26 0.01990095 2 35.4695243 181 i his path/way3 23 0.01760469 1 35.0279966 182 i his path/way3 23 0.01454300 0 34.9640655 184 i good/ better419 0.01454300 0 34.9640655 186 i good/ better419 0.01454300 0 34.9640655 186 i good/ better419 0.01454300 0 34.9640655 185 i hence5 147 0.11251693 105 0.053244088 34.9272308 183 i alone5 91 0.06965334 50 0.02354328 34.9272308 187 i alone5 126 0.09644308 85 0.043102354 33.8020820 188 i and this4 40 0.03061685 10 33.7363166 189 i and this4 40 0.03061685 10 33.7363166 189 i and this4 40 0.02678974 7 33.67798907 180 i and this 4 40 0.02678974 7 33.677989767 <td></td> <td></td> <td>unlawful</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>36.33013535</td> <td>6.32334E-11</td>			unlawful						36.33013535	6.32334E-11
177عن3333 0.02323890 536.21875178sound Hadith that exists)540 0.03061685 9 36.0221360 179that exists)538 0.02908601 8 35.5554313 180indicated/ informed426 0.01990095 2 35.4695243 181this path/way323 0.01760469 1 35.0279960 182etainthe heaven (sky)419 0.01454300 0 34.9640655 184jagoad/better419 0.01454300 0 34.9640655 186jagoad/better419 0.01454300 0 34.9640655 185good/ better419 0.01454300 0 34.9640655 185alone5147 0.11251693 105 0.053244088 34.9272308 187is religion5126 0.09644308 85 0.043102354 33.8020820 188is religion5126 0.09644308 85 0.043102354 33.8020820 189is and this440 0.03061685 10 33.7363166 190it he Jinn335 0.02678974 7 33.6779899	76	ايات		5	31	0.02372806	4		36.23736954	7.14294E-11
178 that exists) 5 40 0.05001085 5 50.0221500 179 فراف knew 5 38 0.02908601 8 35.5554313 180 indicated/informed 4 26 0.01990095 2 35.4695243 181 unup His/his path/way 3 23 0.01760469 1 35.0279960 182 etain the heaven (sky) 4 19 0.01454300 0 34.9640655 184 sgood/ better 4 19 0.01454300 0 34.9640655 186 good/ better 4 19 0.01454300 0 34.9640655 185 etain hence 5 147 0.11251693 105 0.053244088 34.9597053 183 alone 5 91 0.06965334 50 0.025354328 34.9272308 187 religion 5 126 0.09644308 85 0.043102354 33.8020820 188 and this 4 40 0.03061685 10 33.7363166	77	صح	sound Hadith	3	33	0.02525890	5		36.21875	7.32428E-11
180دلت50506101100011181indicated/ informed426 0.01990095 2 35.4695242 181His/his path/way323 0.01760469 1 35.0279960 182etainthe heaven (sky)419 0.01454300 0 34.9640652 184 90001454300 9 9001454300 0 34.9640652 186 $9000000000000000000000000000000000000$		العالمين		5					36.02213669	9.67517E-11
181His/his path/way323 0.01760469 1 35.0279960 182His/his path/way323 0.01760469 1 35.0279960 182etainthe heaven (sky)419 0.01454300 0 34.9640655 184 9.001454300 9.001454300 9.001454300 9.001454300 9.001454300 9.001454300 186 9.001454300 9.001454300 9.001454300 9.001454300 9.001454300 9.0053244088 9.9570553 185 9.001454300 9.006965334 50 0.025354328 9.9272308 183 9.2000 9.1000000 9.009644308 8.5 0.043102354 9.8020820 187 9.0000000 9.009644308 8.5 0.043102354 9.8020820 188 9.0000000 9.000000000 9.00000000000 $9.000000000000000000000000000000000000$				5						2.12587E-10
182المتاب المتاب19190.01454300034.9640655184 \rightarrow magnanimous3190.01454300034.9640655186 \rightarrow good/ better4190.01454300034.9640655185 \rightarrow hence51470.112516931050.05324408834.9597053183 \rightarrow alone5910.06965334500.02535432834.9272308187 \rightarrow religion51260.09644308850.04310235433.8020820188 \rightarrow religion3250.01913553233.7802085189 \rightarrow and this4400.030616851033.7363166190 \rightarrow the Jinn3350.02678974733.6779899			indicated/ informed	4	26		2			2.52092E-10
184190.01454300034.9640655186190.01454300034.9640655186190.01454300034.96406551851851470.112516931050.0532440881831081090.06965334500.0253543281831091260.09644308850.0431023541871261260.09644308850.0431023541871261260.09644308850.0431023541881901260.030616851033.78020851891134400.030616851033.73631661901461903350.02678974733.6779895			His/his path/way	3						7.36403E-10
186190.01454300034.9640655185good/better4190.01454300034.9640655185hence51470.112516931050.05324408834.9597053183هناalone5910.06965334500.02535432834.9272308187ندينreligion51260.09644308850.04310235433.8020820188jand this3250.01913553233.7802085189ia and this4400.030616851033.7363166190ithe Jinn3350.02678974733.6779899		السَّمَاءِ	the heaven (sky)	4						4.33766E-10
185 197 100 10115160 105 185 105 105 0.053244088 34.9597053 183 105 0.053244088 34.9597053 183 105 0.025354328 34.9272308 187 126 0.09644308 85 0.043102354 187 126 0.09644308 85 0.043102354 188 126 0.09644308 85 0.043102354 189 116 100 33.7802085 189 116 100 33.7363166 190 116 100 33.6779899 190 116 100 33.6779899 190 116 100 33.6779899 190 116 100 33.6779899 190 116 100 33.6779899 190 116 100 33.6779899 190 116 100 33.6779899 116 116 100 116 </td <td>34</td> <td>جواد</td> <td>magnanimous</td> <td>3</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>4.33766E-10</td>	34	جواد	magnanimous	3						4.33766E-10
183الجان 147 0.1125105 100 183 alone591 0.06965334 50 0.025354328 34.9272308 187 religion5126 0.09644308 85 0.043102354 33.8020820 188 polytheists (nom.)325 0.01913553 2 33.7802085 189 and this440 0.03061685 10 33.7363166 190 the Jinn335 0.02678974 7 33.6779899			good/ better	4						4.33766E-10
187 دین religion 5 126 0.09644308 85 0.043102354 33.8020820 188 polytheists (nom.) 3 25 0.01913553 2 33.7802085 189 and this 4 40 0.03061685 10 33.7763166 190 the Jinn 3 35 0.02678974 7 33.6779899		بذلك	hence	5	147					4.41294E-10
188 polytheists (nom.) 3 25 0.01913553 2 33.7802085 189 and this 4 40 0.03061685 10 33.6779899 190 the Jinn 3 35 0.02678974 7 33.6779899		وحده	alone	5	91					4.97905E-10
189 المسركون عاد المسركون عاد المسركون 33.7363166 190 ناما the Jinn 3 35 0.02678974 7 33.6779899 290 ناما الجن 35 0.02678974 7 33.6779899	17	دين	religion	5	126			0.043102354		3.17665E-09
190 4 40 0.0507100 7 33.6779895 190 the Jinn 3 35 0.02678974 7 33.6779895	8	المشركون	polytheists (nom.)	3	25					3.24564E-09
	9	فهذا	and this	4	40	0.03061685	10			3.38644E-09
191 (b) indicated/informed 5 27 0.02066637 3 33.2312278	0	الجن	the Jinn	3	35	0.02678974	7			3.57854E-09
	11	دل	indicated/ informed	5	27	0.02066637	3			5.25789E-09
evil 5 56 0.02500001 5	12	شر	evil	5	38	0.02908601	9		33.17198181	5.51103E-09
193 know(s) better 3 31 0.02372806 5 33.1188583	13	اعلم	know(s) better	3	31	0.02372806	5		33.11885834	5.74468E-09
194 teachers/ 3 44 0.03367853 13 32.9771423 propagandists	14	الدعاة		3	44	0.03367853	13		32.97714233	6.40018E-09
195 to make (something) 4 55 0.04209817 21 0.010648817 32.9692344 evident	5	بيان	to make (something)	4	55	0.04209817			32.96923447	6.43818E-09
196 دعوة call 5 55 0.04209817 21 0.010648817 32.9692344	16	دعوة		5	55				32.96923447	6.43818E-09
198 His/his book 5 95 0.07271502 56 0.028396847 32.6166191	8	كتابه		5	95	0.07271502	56	0.028396847	32.61661911	8.30038E-09
197 the call to the true 3 24 0.01837011 2 32.0968284 بالدعوة religion	7		the call to the true	3	24	0.01837011			32.09682846	1.1743E-08
199 is 10 the hereafter 5 63 0.04822154 28 0.014198423 52.0049032	9	الأخرة	-	5	63	0.04822154	28	0.014198423	32.06490326	1.1986E-08
200 4 His creatures 4 43 0.03291311 13 31.6596698	0				43	0.03291311	13		31.65966988	1.54451E-08

201	حجة	plea against/argument/ pilgrimage	3	26	0.01990095	3	31.60539055	1.59657E-08
202	التواصى	recommend one another to	3	17	0.01301216	0	31.2834816	1.93722E-08
.03	الحافظ	Alhafidh (a higher status of the scholars of Hadith)	4	17	0.01301216	0	31.2834816	1.93722E-08
06	أنصاره	his patrons	3	17	0.01301216	0	31.2834816	1.93722E-08
04	المنكر	the evil	4	33	0.02525890	7	30.74622345	2.64829E-08
05	الأصنام	idols	3	23	0.01760469	2	30.41987801	3.18697E-08
07	بعدهم	after them	3	23	0.01760469	2	30.41987801	3.18697E-08
10	لأهل	to the people of	4	23	0.01760469	2	30.41987801	3.18697E-08
09	شاء	will (that something be done)	4	42	0.03214769	13	30.35300827	3.30901E-08
08	علمه	taught him	4	29	0.02219721	5	30.05491447	3.90736E-08
11	رسله	His messengers	4	25	0.01913553	3	29.98752403	4.05588E-08
12	إذا	if	4	27	0.02066637	4	29.91103363	4.23084E-08
13	هدى	guidance	4	34	0.02602432	8	29.80052185	4.49616E-0
14	عمر	Omar	4	37	0.02832058	10	29.60105133	5.01514E-08
15	الأوزاعي	Alawzaee	3	16	0.01224674	0	29.44320297	5.46549E-08
16	جميعا	all	4	16	0.01224674	0	29.44320297	5.46549E-08
19	رْسُولًا	a messenger (acc.)	4	16	0.01224674	0	29.44320297	5.46549E-08
18	فبين	and (He/he) clarified	3	16	0.01224674	0	29.44320297	5.46549E-08
17	كلام	speech	5	48	0.03674022	18	29.28578568	5.95274E-08
20	عليكم	upon you	3	41	0.03138227	13	29.05761337	6.73327E-08
21	مالك	Malik	4	30	0.02296263	6	28.86632919	7.46246E-08
222	الأولباء	friends of God (religious status in Sufi orders)	3	22	0.01683927	2	28.74989891	7.94295E-08
223	أتباعهم	their followers	5	22	0.01683927	2	28.74989891	7.94295E-08
225	طاعة	obedience	3	42	0.03214769	14	28.54162598	8.87809E-08
224	اكمل	made perfect/ continued	4	33	0.02525890	8	28.38845825	9.63306E-08

Table A2 Keywords that indicate semantic foci in Ibn Baz

and and and and	Key word	Eng. trans.	Texts	Freq.	%	F. Alhawali	% Alhawali	Keyness	Р
1	الله	Allah	5	4253	3.25534	994	0.50404	3803.249	1.49E-20
2	سبحانه	the Glorified	5	809	0.61923	8		1409.8375	3.07E-19
3	جل	the Majestic	5	424	0.32454	18		648.80554	3.44E-18
1	عز	the Exalted in Mighty	5	363	0.27785	11		580.47906	4.90E-18
5	الصلاة	(may) the blessings (of Allah be upon him)/ the prayer	5	289	0.2212	18		413.41336	1.47E-17
5	تعالى	the Exalted	5	477	0.36511	151	0.07657	339.10696	2.82E-17
7	ىڭە	to Allah	5	267	0.20437	74	0.03752	210.0125	1.47E-16
3	الدعوة	the call (to)	4	216	0.16533	45	0.02281	203.41599	6.23E-16
9	السلام	the Peace/ the peace/ Islamic salutation	5	255	0.19518	78	0.03955	186.18858	2.27E-16
0	الشرك	associationism/ polytheism	5	131	0.10027	15		159.7037	4.03E-16
1	بالله	in/with/by Allah	4	173	0.13242	41	0.02079	151.03606	4.99E-16
2	الحق	the True/ right	5	242	0.18523	94	0.04766	142.70996	2.59E-15
3	علا	the Most High	3	76	0.05817	0		139.87622	6.74E-16
4	الهدى	the guidance/ the right path	5	67	0.05128	3		101.589	8.05E-15
5	الباطل	the falsehood	5	71	0.05434	5		98.876724	2.93E-15
6	بإحسان	followed them exactly (in faith)	5	51	0.03904	0		93.858428 9	3.73E-15
7	العظيم	the great/ the Great	5	84	0.0643	15		85.63044	5.81E-15
8	دينه	His/his religion	4	73	0.05587	11		80.315681	9.86E-15
9	صفاته	His attributes	5	42	0.03215	0		77.293434	9.86E-15
.0	البدع	the religiously impermissible innovative practice and beliefs	5	63	0.04822	8		74.081894	1.24E-14
21	العبادة	the worship	3	120	0.09185	51	0.02586	64.293205	2.88E-14
22	الإسلام	Islam	5	337	0.25794	271	0.13742	60.057278	2.30E-13
23	القيامة	the (Day of) Resurrection	4	45	0.03444	4		59.172089	5.02E-14
24	رب	Lord, may	5	61	0.04669	13		56.690296	6.85E-14
25	ربه	his Lord	4	40	0.03062	3		54.900444	8.76E-14
26	الصفات	the attributes (of Allah)	4	46	0.03521	6		53.560551	1.07E-13
27	المستقيم	straight (way)	5	34	0.02602	2		49.154232	2.37E-13
28	شرع	(Allah has) ordained	4	47	0.03597	8		49.005337 47.482338	2.27E-12 3.26E-13
29	العزيز	the Exalted in power	5	46	0.03521	8		47.482338	2.93E-12
80	بالحق	truly/ with the truth	5	37	0.02832	4		43.843914	2.93E-12 8.39E-13
31	الاستغاثة	beseeching for help	3	28	0.02143	1		43.843914	8.66E-13
32	النار	the Hell-fire	5	58	0.04439	17		43.737032	4.70E-13
33	الشرع	the Sharia Law	4	34	0.02602	4		40.485012	7.03E-12
34	شهادة	testimony	3	22	0.01683	0		39.804073	4.09E-12
35	التحذير	warning	4	31	0.02372	3	0.00000		
36	العلم	knowledge (about Allah)	5	225	0.17221	183	0.09279	38.845768	8.52E-12

37	طاعته	obeying Him	3	25	0.01913	1		38.5453	8.88E-08
38	التحاكم	arbitration (according to Islamic Law)	3	30	0.02296	3		38.151337	1.34E-11
39	ولي	Wali (Lord/ God/ Protector/ Helper)	4	36	0.02756	6		37.89933	1.35E-11
40	کريم	the most Generous	4	20	0.01531	0		36.804371	3.63E-11
41	السلامة	security	3	24	0.01837	1		36.784973	5.09E-11
42	العاقبة	(blessed) end	4	29	0.02219	3		36.504673	7.36E-10
43	سبيله	His/his path/way	3	23	0.0176	1		35.027996	3.18E-09
44	جواد	Magnanimous	3	19	0.01454	0		34.964066	4.34E-10
45	وحده	(He) Alone	5	91	0.06965	50	0.02535	34.927231	4.98E-10
46	دين	religion	5	126	0.09644	85	0.0431	33.802082	6.44E-09
47	شر	evil	5	38	0.02908	9		33.171982	5.51E-09
48	دعوة	call (to Allah)	5	55	0.04209	21	0.01064	32.969234	1.17E-08
49	بالدعوة	via the call (to Allah)	3	24	0.01837	2		32.096828	3.91E-08
50	الأخرة	the Hereafter	5	63	0.04822	28	0.01419	32.064903	1.20E-08
51	المنكر	Al-Munkar (polytheism and disbelief and all that is wrong and Islam has forbidden)	4	33	0.02525	7		30.746223	2.65E-08
52	الأصنام	idols	3	23	0.0176	2		30.419878	3.19E-08
53	هدى	guidance	4	34	0.02602	8		29.800522	2.88E-14
54	طاعة	obedience (submission to Allah)	3	42	0.03214	14		28.541626	1.12E-11

Table A3 Keywords that indicate social actors in Ibn Baz

N	Key word	Eng. trans.	Texts	F.	<u>ai actors ii</u> %	F. Alhawali	% Alhawali	V	n
		(may) the blessings (of	10/(15	<u>_</u>	/0	r. Alnawan	% Alhawali	Keyness	Р
1	الصلاة	Allah be upon him)/ the	5	289	0.221206	18		413.41336	1.46E-17
		prayer				10		415.41550	1.401/17
2	رسول	messenger	5	270	0.206663	28	0.01419	339.875	2.80E-17
3	النبي	the prophet	4	225	0.172219	21	0.01064	292.07516	4.65E-17
4	بن	son of	5	265	0.202836	48	0.02434	268.45166	6.21E-17
5	رسوله	His messenger	5	233	0.178343	34	0.01724	259.90494	6.94E-17
6	اهل	the people of	5	244	0.186762	40	0.02028	258.98544	7.02E-17
7	رضي	(may Allah) be pleased with	5	182	0.139306	18		232.32063	1.03E-16
		him							
8	الرسول	the messenger	5	161	0.123232	15		209.0584	1.49E-16
9	السلام	(may) peace (of Allah be	5	255	0.195182	78	0.03955	186.18858	2.27E-16
10	محمد	upon him) Mohammad	5	233	0.178343	78	0.03955	157.87801	4.21E-16
10	نبينا	our Prophet	4	233 87	0.066591	1	0.03755	150.1987	5.10E-16
12	<u> </u>	his family	4	102	0.078073	7		142.87639	6.20E-16
13	أمنوا	(they, those who) believed	5	88	0.067357	3		138.64163	6.98E-16
	ايها	O, (those, people)	5	104	0.079603	11		130.05186	9.03E-16
14		(vocative)							
15	يا	O, (vocative)	5	156	0.119405	40	0.02028	129.44696	9.20E-16
16	الرسل	the messengers	5	98	0.075011	11		120.24679	1.25E-15
17	أنهم	indeed they are	5	61	0.04669	0		112.26485	1.67E-15
18	جميعا	all	5	65	0.049752	1		110.27984	1.80E-15
19	عباده	His servants, worshippers	5	65	0.049752	1		110.27984	1.80E-15
20	أصحابه	his (the Prophet) companions	4	68	0.052048	2		109.01865	1.89E-15
21	عنه	with him	5	221	0.169158	106	0.05375	102.53931	2.48E-15
22	عبد	servant, worshiper	5	149	0.114047	53	0.02687	95.607162	3.42E-15
		his (the Prophet)	5	69	0.052814	5		95.47123	3.44E-15
23	صحبه	companions	5	09			0.04460	02 00111	4.08E-15
24	عليهم	upon them	5	190	0.14543	88	0.04462	92.09111	
25	العباد	the servants of Allah, His worshippers	5	67	0.051283	5		92.073608	4.08E-15
26	لهم	to them	5	252	0.192886	147	0.07454	88.095955	5.05E-15
27	الناس	the people	5	396	0.303106	304	0.15415	79.642914	8.41E-15
28	-		5	51	0.039036	2		78.858948	8.86E-15
29	نبيه	His Prophet	5	226	0.172985	138	0.06997	73.123795	1.34E-14
30	المسلمين	the Muslims	4	60	0.045925	7		72.673447	1.38E-14
	السلف	the pious ancestors	5	91	0.069653	25	0.01267	71.980255	1.46E-14
31	بهم	with them	-		0.088788	43	0.0218	71.589943	1.51E-14
32	ابن ،	son of	5	116 61	0.088788	8		70.886078	1.59E-14
33	أبي	the father of (acc.)	4		0.04009	9		69.266205	1.82E-14
34	الإمام	the Imam	5	62		0		68.090988	2.02E-14
35	لعلكم	that you may	5	37	0.02832		0.010/7		2.05E-14
36	رحمه	(may Allah) have mercy (upon) him	5	88	0.067357	25	0.01267	67.935989 67.017975	2.03E-14 2.22E-14
37	منكم	from you	4	41	0.031382	1			2.22E-14 2.25E-14
38	مسلم	Muslim	4	78	0.059702	19		66.907005	
39			5	187	0.143133	109	0.05527	65.433479	2.58E-14
40	هم	they	4	34	0.026024	0		62.569626	3.43E-14
	لعباده	to His servants/worshippers		82	0.062764	24	0.01217	61.909237	3.68E-14
41	غيرهم	others	5	82 59	0.045159	10		61.640888	3.79E-14
42	أبو	the father of (nom.)	4	57	0.010107				

43	الصحابة	the companions of the Prophet	5	48	0.03674	5		60.297188	4.40E-14
44	التابعين	the successors of the companions of the Prophet	4	32	0.024493	0		58.888763	5.19E-14
45	عنهم	with them	5	74	0.056641	22	0.01115	55.204288	8.39E-14
46	الشيخ	the sheikh	5	113	0.086492	52	0.02636	55.18483	8.41E-14
47	أئمة	Imams	4	40	0.030616	3		54.900444	8.76E-14
48	الثقلين	the humans and the jinn	3	24	0.01837	0		44.165688	7.62E-13
49	الصالح	pious	4	39	0.029851	6		42.529701	1.29E-12
50	محمدا	Mohammad (acc.)	5	23	0.017047	0		42.325344	1.39E-12
51	الذين	who (plural)	5	261	0.199775	218	0.11054	41.792873	1.69E-12
52	غيره	other	5	91	0.069653	44	0.0223	41.762089	1.71E-12
53	كنتم	you were	4	34	0.026024	4		41.061699	2.27E-12
54	اليك	to You/you	4	26	0.019901	1		40.308712	3.19E-12
55	المشركين	polytheists (acc.)	3	30	0.022962	3		38.151337	1.12E-11
56	المؤمنين	the believers	4	47	0.035974	13		36.990223	2.99E-11
57	هريرة ا	Hurairah	4	24	0.01837	1		36.784973	3.70E-11
58	العالمين	mankind, jinn and all that exists	5	40	0.030616	9		36.022137	9.68E-11
59	المشركون	polytheists (nom.)	3	25	0.019135	2		33.780209	3.25E-09
60	الجن	the Jinn	3	35	0.026789	7		33.67799	3.58E-09
61	الدعاة	teachers, propagandists	3	44	0.033678	13		32.977142	6.40E-09
62	خلقه	His creatures	4	43	0.032913	13		31.65967	1.54E-08
62	حلفه الحافظ	Alhafidh (a higher status of the scholars of Hadith)	4	17	0.013012	0		31.283482	1.94E-08
64	انصاره	his patrons, advocates	3	17	0.013012	0		31.283482	1.94E-08
	-	•	3	23	0.017604	2		30.419878	3.19E-08
65	بعدهم	after them	4	23	0.017604	2		30.419878	3.19E-08
66	لأهل	to the people of		25	0.019135	3		29.987524	4.06E-08
67	رسله	His messengers	4	37	0.02832	10		29.601051	5.02E-08
68	عمر	Omar	4		0.02832	0		29.443203	5.47E-08
69	رَسُولًا	a messenger (acc.)	4	16		0		29.443203	5.47E-08
70	الأوزاعي	Alawzaee	3	16	0.012246	•		29.443203	5.47E-08
71	جَمِيعًا	all	4	16	0.012246	0		29.057613	6.73E-08
72	عليكم	upon you	3	41	0.031382	13		28.866329	7.46E-08
73	مالك	Malik	4	30	0.022962	6			
74	الأولياء	friends of God (religious status in Sufi orders)	3	22	0.016839	2		28.749899	7.94E-08 7.94E-08
75	أتباعهم	their followers	5	22	0.016839	2		28.749899	7.740-00

the market was allowing

Table A4 Argumentation indicators (keywords)

N	Keyword	Eng. trans.	Texts	F.	%	F. Alhawali	% Alhawali	Keyness	Р
1	قال	He/he said	5	1100	0.842	195	0.0989	1129.3	6.09E-19
2	قوله	His/his saying	5	221	0.1692	1		395.11	1.7E-17
3	الآية	the Qur'anic verse/ till the end of the Qur'anic verse	4	129	0.0987	5		199.84	1.75E-16
4	الأيات	the Qur'anic verses	5	138	0.1056	14		174.8	2.87E-16
5	الأحاديث	Hadiths (the sayings of the Prophet of Islam)	5	83	0.0635	4		124.38	1.08E-15
6	السنة	the Sunnah of the Prophet of Islam	5	142	0.1087	35	0.0177	120.92	1.22E-15
7	فالواجب	therefore it is obligatory (on) to	3	46	0.0352	0		84.656	6.15E-15
8	رواه	(this Hadith is) narrated by	4	56	0.0429	3		82.392	7.06E-15
9		then (He/he) said	5	85	0.0651	30	0.0152	54.918	8.74E-14
10	الأدلة	the indications (from Qur'an and Prophetic traditions)/ proofs	4	54	0.0413	10		54.068	9.9E-14
11	ثبت	it became firm (that the Prophet said)	4	46	0.0352	6		53.561	1.07E-13
12	حديث	Hadith	4	48	0.0367	9		47.761	3.07E-13
13	لقول	because (He/he) said	4	24	0.0184	0		44.166	7.62E-1 3
14	لأن	because	5	158	0.1209	110	0.0558	39.72	4.27E-12
15	الصحيحين	the two most authentic books Hadiths	3	25	0.0191	1		38.545	8.52E-12
-16	أيات	Qur'anic verse	5	31	0.0237	4		36.237	7.14E-11
17	صح	it was narrated in a sound Hadith	3	33	0.0253	5		36.219	7.32E-11
18	دلت	indicated (fem. Sing. and pl.)	4	26	0.0199	2		35.47	2.52E-10
19	بذلك	hence/ by that means	5	147	0.1125	105	0.0532	34.96	4.41E-10
20	دل	indicated (masc. sing.)	5	27	0.0207	3		33.231	5.26E-09

になったとれていたので、ことになってきたなからないないないないであると、

Table A5 Word clusters of the key term 'ALLAH' and the keywords 'سبحانه' (subħanahu; the Glorified) and 'سبحانه' (taɛaala; the Exalted) in Ibn Baz

	N	Word clusters	Eng. trans.		
2-	1	قال تعالى	the Exalted said	Freq.	Texts
2-word clusters	2	<u>قال سبحانه</u>		194	5
dc	3	قوله تعالى	the Glorified said	157	5
lus	4	قوله سبحاني	,	58	3
ters	5		His words the Glorified	57	4
		يقول سبحانه	the Glorified says	27	4
	6	فبين سبحانه	then the Glorified showed	12	3
	7	هو مبدانه	He, the Glorified	10	3
	8	فقال سبحانه	and the Glorified said	9	3
	9	فهو سبحانه	because He, the Glorified	9	3
	10	له سبحانه	,	9	3
	11	أوضح سبحانه	the Glorified explicated	7	3
	12	فرد الله	Allah has referred () to/ Allah has replied	7	3
ω	13	نسأله سبحانه	we pray to Him, the Glorified	6	4
3-word clusters	14	الله تعالى في	Allah, the Exalted in	7	3
ord	15	الله سبحانه في	Allah, the Glorified in	14	3
clu	16	الله سيحانه قد	indeed, Allah, the Glorified has	13	4
Iste	17	الله سبحانه و	Allah, the Glorified and	76	5
S	18	إلى الله سبحانه	to Allah, the Glorified	21	3
	19	أمر الله سيحانه	Allah, the Glorified, ordered	9	3
	20	أن الله سبحانه	that Allah, the Glorified	28	4
	21	بالله سبحانه و	in Allah, the Glorified and	5	3
	22	بأن الله سبحانه	is that Allah the Glorified	7	3
[23	به سبحانه و	in/with Him, the Glorified and	6	4
	24	ثم قال سبحانه	then the Glorified said	5	3
	25	سبحانه ذَلِكَ بِأَنَّ	the Glorified that (something) is	5	3
	26	سبحانه في كتابه	the Glorified, in His Book	7	3
	27	سبحانه في هذه	the Glorified in this	17	4
	28	سبحانه و تعالى	the Glorified and the Exalted	114	5
Γ	29	سبحانه و قد	the Glorified, and (something) has indeed	6	4
	30	سبحانه و لا	the Glorified and not	7	3
	31	سبدانه و لقد	the Glorified and indeed	7	3
	32	سبحانه و ما	the Glorified and what	23	4
	33	سبحانه یا أیها	the Glorified: (O Those	13	3
	34	فان الله سيحانه	and then indeed Allah, the Glorified	5	3
	35	فبين سيحانه في	and then the Glorified showed in	5	3
	36	في قوله تعالى	in His words, the Exalted	11	3
	37	قال الله تعالى	Allah, the Exalted said	22	3
ſ	38	قال الله سيحانه	Allah, the Glorified, said	22	5
	39	قال تعالى و	the Exalted said: (and	92	4
	40	قال تعالى يا	the Exalted said: (O	8	3
	41	قال سبحانه في	the Glorified said in	12	3
	42	قال سبحانه و	the Glorified (and the Exalted) said	80	5
	43	قوله تعالى و	in His words, the Exalted and	24	3
	44	قوله سبحانه و	in His words the Glorified and	27	3
	45	قوله سبحانه يا	in His words the Glorified: (O	5	5
	46	كما قال الله	as Allah has said	30	4
	47	كما قال تعالى	as the Exalted said	49	5
-	48	كما قال سبحانه	as the Glorified said	14	4
	49	لأن الله سبحانه	because Allah, the Glorified	14	

	50	لله سبحانه و	to Allah, the Glorified and		2
	51	له «بحانه و	to Him the Glorified and	6	3
	52	من الله سبحانه	from Allah, the Glorified	7	3
	53	نسأل الله العاقية	we seek refuge in Allah	9	3
	54	نسال الله أن	we pray to Allah that	7	
	55	نساله سبحانه ان	we pray to Him, the Glorified, that	6	3
	56	و الله سيحانه	and Allah, the Glorified	13	4
	57	و أسال الله	and I pray to Allah	16	3
	58	و قال تعالى	and the Exalted said	123	4
	59	و قال سبحانه	and the Glorified said	82	4
	60	و قوله تعالى	and in His words the Exalted	13	
	61	و قوله سبحانه	and in His words the Glorified	24	3
	62	و يقول سبحانه	and the Glorified says	13	4
	63	يقول سبحانه و	the Glorified (and the Exalted) says	17	4
1	64	الآية و قال تعالى	the Qur'anic verse and the Exalted said	7	
	65	الطّاغُوتَ و قال تعالى			4
			(taguut), viz., idols and Jinni. And the Exalted said	5	3
	66	الله سبحانه و تعالى	Allah, the Glorified and the Exalted	39	4
	67	إلى الله سبحانه و	to Allah, the Exalted and	9	3
	68	بِالله و الْيَوْمِ الْآخِرِ	in Allah and the Last Day	5	3
	69	تعالى و ما أرْسَلْنَا	the Glorified (says/said) and whosoever (from among the messengers) We sent	7	4
	70	تعالى يا أيها الناس	the Exalted (says/said) O All people	6	3
	71	سبحانه ذَلِكَ بِأَنَّ الله	the Glorified that Allah is	5	3
	72	سبحانه في هذه الآيات	the Glorified in these Qur'anic verses	5	3
	73	سبحانه في هذه الآية	the Glorified in this Qur'anic verse	12	3
	74	سبحانه و تعالى في	the Glorified and the Exalted in	11	4
	75	سبحانه و تعالى و	the Glorified and the Exalted and	21	3
	76	سبحانه يا أيها الذين	the Glorified: (O Those who	10	3
	77	فبين سبحانه في هذه	and the Glorified showed in this	5	3
	78	في قوله تعالى و	in His words the Exalted and	8	3
	79	قال تعالى و لقد	the Exalted said: (indeed	11	4
	80	قال تعالى و ما	the Exalted said: (and whatsoever	22	4
	81	قال تعالى و من	the Exalted said: (and whosoever	8	4
	82	قال سبحانه و تعالى	the Glorified and the Exalted said	17	3
	83	قال سبحانه و لقد	the Glorified said: Indeed	6	3
	84	قال سبدانه و ما	the Glorified said: Whatsoever/whosoever	11	3
	85	قوله تعالى و من	His words the Exalted: And those	5	3
	86	كما قال الله سيحانه	as Allah, the Glorified, said	14	5
	87	كما قال تعالى و	as the Exalted said and	27	4
	88	كما قال سبحانه و	as the Glorified and (the Exalted) said	23	5
	89	و قال تعالى و	and the Glorified said and	58	4
	90	و قال تعالى يا	and the Exalted said: O	5	3
	91	و قال سبحانه و	and the Glorified and (the Exalted) said	47	4
	92	و قوله سبحانه و	and His words the Glorified and	17	3
	93	و يقول سبحانه و	and the Glorified and (the Exalted) says	7	3

No - Manufacture and some

Table A6 Collocates of the key term 'ALLAH' in Ibn Baz's corpus

N	keyword	Eng. trans.	F.	Collocates	Eng. trans.	F.	Texts	Joint	MI
1				لْيَنْصُرَنَّ	(Allah) will certainly aid	7	4	7	5.6525679
2				رسول	messenger	270	5	269	5.4243755
3				لِيُقَرِّبُونَا	that they may bring us near (to Allah)	10	3	16	5.3306398
4				شُفَعَاؤُنًا	our intercessors (with Allah)	10	3	15	5.2375307
5			لكتاب	to the book of	9	4	11	4.9420748	
6				بسم	in the name of	5	3	6	4.9156027
7			لشرع	the Sharia of (Allah)	6	3	7	4.8749604	
8			عند	with/near (Allah)	13	3	15	4.8590188	
9			معاذ	Muaath (a companion of the Prophet of Islam)	8	3	9	4.8224931	
10				أصحاب	the people of/ the companion of (the Prophet of Allah)	30	4	33	4.7900715
11				صح	it was narrated in a sound Hadith	33	3	36	4.7780991
12				بحمد	glory to (Allah)	16	3	17	4.7400308
13				رضي	(may Allah) be pleased (with him/her)	182	5	177	4.6839323
14				رحمه	may (Allah have) mercy (upon)	88	5	89	4.66887
15				أن	indeed, that	18	4	18	4.6525679
6		Allah	4253	اسال	pray that	18	4	18	4.6525679
7				يَحْكُمْ	(Allah will) judge between them	18	3	18	4.6525679
8				أغبدوا	(you pl.)worship (v. imperative)	15	4	15	4.6525679
9				بكتاب	with the Book (of Allah)	13	3	13	4.6525679
:0	الله			هَوْلاءِ	these (people)	13	3	13	4.6525679
21		Anan	4255	فَاعْبُدِ	and (you sing.) worship (imperative)	12	4	12	4.6525679
2				فنسأل	hence we pray to Allah (that)	11	3	11	4.6525679
23				نَعْبُدُهُمْ	we worship them	10	3	10	4.6525679
.4				تتصروا	(if you) help (in the cause of Allah)	9	3	9	4.6525679
5				يَبْغُونَ	seek	9	3	9	4.6525679
6				يَخْتَلِفُونَ	(they) differed	9	3	9	4.6525679
7				بِأَنَّ	that	8	4	8	4.6525679
8				فحكمه	the decision thereof (is with Allah)	8	3	8	4.6525679
9				بَعْبُدُونَ	they worship	8	3	8	4.6525679
0				يَتْصُدُهُ	aid Him	8	4	8	4.6525679
1				أتُنَبِّنُونَ	do you inform Allah (that)	7	3	7	4.6525679
2				رَسُولُ	messenger (nom.)	7	3	7	4.6525679
3				ليَقُولُنَّ	they will surely say	7	3	7	4.6525679
4				رضوان	(may Allah) be pleased (with them)	6	3	6	4.6525679
5				سمى	(Allah has) named (that)	6	3	6	4.6525679
6				صراط	the (straight) way (of Allah	6	3	6	4.6525679
7				وفقه	(may Allah) guide him	6	3	6	4.6525679
8				اتَبَعُو هُمْ	followed them (exactly in Faith)	5	3	5	4.6525679
19				الأمثال	(So put not forward) similitudes (for Allah)	5	3	5	4.6525679
40				الأمرز	the affairs	5	3	5	4.6525679
41				بإحسان	(followed them) exactly in Faith	5	3	5	4.6525679

42	بشرع	with the Sharia (of Allah)	5	3	5	4.6525679				
43	تَضْرِبُوا	(So) put (not) forward (similitudes	5	3	5	4.6525679				
44	رَضِيَ	for Allah) (Allah) is well-pleased (with them)	3	5						
45	فَسَيَةُولُونَ	they will say	5	3	5 5	4.6525679 4.6525679				
46	مىلى	(may Allah) send his blessings (upon him)	876	5	872	4.6459656				
47	نسال	we pray (to Allah that)	30	3	29	4.6036587				
48	النبي	the Prophet	225	4	212	4.5667071				
49	تَنَازَ عَتُم	(you pl.) differ	15	3	14	4.5530324				
50	فَرُدُوهُ	refer it (to Allah)	15	3	14	4.5530324				
51	اتَقُوا	fear (v. pl. imperative)	13	3	12	4.5370908				
52	دُونِ	besides (Allah)	22	4	20	4.5150647				
53	ذلك	that	11	4	10	4.5150647				
54	لقول	because (Allah/the Prophet) has said	24	3	21	4.4599228				
55	رَسُولًا	a messenger)acc.)	16	3	14	4.4599228				
56	فرد	then (Allah) has replied	8	3	7	4.4599228				
57	معبود	worthy of worship	8	3	7	4.4599228				
58	كتاب	Book	143	5	125	4.4584808				
59	عانشة	Aysha (the youngest wife of Prophet Mohammad)	7	4	6	4.4301758				
60	لنبيه	to His Prophet	7	3	6	4.4301758				
61	اوجب	(Allah) made obligatory	32	4	27	4.4074554				
62	هريرة	Hurairah (a companion opf the Prophet and a famous reporter of Hadith	24	3	20	4.3895335				
63	رحمهم	(may Allah) have mercy upon then	18	3	15	4.3895335				
64	بعثه	(Allah) has sent him	12	4	10	4.3895335				
65	فطر	(Allah's Fitrah -Allah's Islamic Monotheism- with which He) has	6	3	5	4.3895335				
	يُدَبِّرُ	created (mankind) dispose (all affairs)	6	3	5	4.3895335				
66 67	يدبر أمَة	nation	17	3	14	4.3724604				
68	عبد	the servant of/ the first of a compound the name 'Abdu Allah'/	149	5	122	4.3641367				
(0	رحمة	worshipped (v.) mercy	56	5	45	4.3370662				
69 70	رحمه المنه	(the) way (of Allah)/ year/ the Sunnah of the Prophet	125	5	92	4.2103457				
71	خَبِير	(Allah is) Well-Acquainted (with what you do)	7	3	5	4.1671414				
72	يُرِيدُ	(He/he) wants	7	3	5	4.1671414				
73	يري حرم	prohibited	48	4	34	4.1550684				
74	روي	it was reported that	17	3	12	4.1500678				
75	ان ا	indeed	108	5	74	4.1071339				
76	بي جواد	Most Magnanimous	19	3	13	4.1050801				
77	دين	religion	126	4	86	4.101553				
78	بعث	sent	46	4	31	4.0832024				
78 79	بت	Alkhattab (the second name of a companion(s) of the Prophet of Islam)	12	3	8	4.0676055				
81 العلم (vital) 9 3 6 4.0676052 82 Shelkh 20 3 13 4.0310798 84 Shelkh 20 3 13 4.0310798 84 Shelkh 20 3 13 4.0310798 84 Shelkh 20 3 13 4.0310798 85 Shelkh 20 3 14 4.0220909 86 Stemson, workspeers 24 4 14 3.379504 87 Stemson, workspeers 24 4 14 3.379504 89 J.J. Stemson, workspeers 24 4 14 3.379504 91 God 74 3 40 3.3705504 92 J. O. (vocative) 156 5 79 3.570428 94 O. (vocative) 156 3 5 11.421431 96 J.S. Stem of Clay) 15 3 11.421431 97 Allah 173 3 14 7.8316326	80			1	زُلْقَى	No. 4 (11) 15				
---	---------------------------------------	-------	-------------	------	---------------	--	-----	---	-----	-----------
82 سابل (b) سابل (b) سابل (b) 20 4 13 40310798 83 سابل (b) (b) (b) (b) (b) (c) (c) <td< td=""><td></td><td></td><td></td><td></td><td></td><td>Near to (Allah)</td><td>9</td><td>3</td><td>6</td><td>4.0676055</td></td<>						Near to (Allah)	9	3	6	4.0676055
83 الجنوب العالي العال	State and				-			4	13	4.0310798
84 العامر العاري 100 point 51 53 34 4.024397 85 The messenger 161 3 104 4.022090 86 ٤ The two books of sound Hadiths 25 3 16 4.0087118 86 ٤ the Sharia of (Allah) 46 5 3 3.8749604 87 servank, worshippers 24 4 14 3.8294458 89 J S S S S S 90 S God 74 3 40 3.6502627 91 God 74 3 40 3.6502627 92 U O (vocative) 155 5 79 3.6709466 93 Allah 173 3 14 8.6872425 117 3.473842 94 O Vocative) 15 3 5 11.421431 95 Mise ison Strength (and no power save by Allah) 173	1995-17-52						20	3	13	4.0310798
85 الماد المعنوب المالي 101 3 104 4.022009 86 المعنوب الدي الدي المحلوب المعنوب المالي The two books of sound Hadiths 3 104 4.022109 87 المعنوب المع				24.7			51	5	33	4.0245371
86 المالة مالة	and the second second							3	104	4.0220909
87 المحسسة (a) 40 5 30 4.004809 88 المحسة (a) servans, worshipers 24 4 14 3.8749604 89 المحسة (a) God 23 4 13 3.829458 89 المحسة (a) God 74 3 40 3.795526 91 (a) God 74 3 40 3.7656428 92 (b) O. (vocative) 156 5 79 3.6709466 93 (b) O. (vocative) 156 5 105 3.026277 94 (b) (b) (c) (vocative) 15 3 5 11.421431 96 (a) (b) (c) (b) (c) 3 14 8.6872425 98 (a) (a) (a) (c) (c) <td>S. March 19</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>25</td> <td>3</td> <td>16</td> <td>4.0087118</td>	S. March 19						25	3	16	4.0087118
88 العلم المسجور المسلم (ac.) 23 4 14 3.849404 89 العلم (ac.) 23 4 13 3.829453 90 الحمدية sent down 96 4 53 3.795526 91	12.1.1				-	· ,	46	3	30	4.0048699
89 المسحلة التعلي	1000						24	4	14	3.8749604
العلي العلي المحلة الحلي							23	4	13	3.8294458
91 المالي المرابع المالي المرالي المالي المرابع المالي المرابي	·王治·北京						96	4	53	3.795526
92 10 10 10 10 10 10 100	20415-348						48	4	26	3.7680454
93 $u = u_1$ His messenger23351051055.050207494 $u = u_1$ His messenger23351053.502627495 $u = u_1$ 96 $u = u_1$ 97 $u = u_1$ 98 $u = u_1$ 99 $u = u_1$ 100 $u = u_1$ 101 $u = u_1$ 101 $u = u = u_1$ $u = u_1$ 102 $u = u = u_1$ $u = u_1$ 103 $u = u = u_1$ $u = u_1$ 104 $u = u = u_1$ $u = u = u_1$ $u = u_1$ 105 $u = u = u_1$ $u = u_1$ $u = u_1$ $u = u_1$ <					إله	God	74	3	40	3.7650428
94 ω the son of26551073.473084295 $in/with/by$	2010					O (vocative)	156	5	79	3.6709466
الله الله الله الله الله الله الله الله	2 Martin Contraction			-	رسوله	His messenger	233	5	105	3.5026274
96						the son of	265	5	117	3.4730842
97 المالي in/with/by Allah 173 3 14 8.6872425 98 in/with/by Allah 173 3 14 8.6872425 99 (there is no strength and no) power (save by Allah) 173 3 14 7.8316326 99 (there is no strength and no) power (save by Allah) 173 3 14 7.8316326 100 (there is no strength and no) power (save by Allah) 173 3 14 6.0027442 101 (the purc (religion)) 11 4 14 6.0027442 103 (So put not forward) similitudes (for Allah) 5 3 5 8.6461372 104 (So put not forward) similitudes (for Allah) 5 3 5 8.6461372 105 (So put not) forward (similitudes (for Allah) 12 4 11 8.520606 106 (for Allah's sake only) 12 4 11 8.520606 107 (ding religious deeds) sincerely (for Allah's sake only) 12 4 11 8.43061224 108 to Allah 267 (ding religion all kinds of good deeds) 3 8	95					(the Last) Day	6	3	5	12.74336
98 in/with/by Allah 173 3 14 8.68/2425 99 (here is no strength and no) power (save by Allah 173 3 14 7.8316326 99 (here is no strength and no) power (save by Allah 173 3 14 7.8316326 100 (here is no strength and no) power (save by Allah 173 3 14 7.8316326 101 11 4 14 6.0027442 101 173 3 9 5.4087071 101 173 3 23 4.7873383 102 (bo put not forward) similitudes (for Allah) 5 3 5 8.6461372 103 (bo put not forward) similitudes (for Allah) 5 3 5 8.6461372 104 105 iaige 67 5 63 8.5573282 106 Allah uch (who abstain form evill deeds and perform all kinds of good deeds) 9 3 8 8.4762125 107 iaige (her eligion 13 4 11 8.4051294 108 io Allah 267 iiii the religion 13	96				الأخر	the Last (Day)	15	3	5	11.421431
98 سرم Allah 173 3 14 7.8316326 99 (nare by Allah 173 3 14 7.8316326 99 (nare by Allah 173 3 14 7.8316326 100 (nare by Allah 173 3 14 7.8316326 101 173 3 9 5.4087071 101 173 3 23 4.7873383 102 103 (So put not forward) similitudes (for Allah) 5 3 5 8.6461372 104 (So put not) forward) similitudes (for Allah) 5 3 5 8.6461372 105 106 11 4 11 8.520606 106 107 Allah much (who abstain from evil deeds and perform all kinds of good deeds) 9 3 8 8.4762125 108 109 ito Allah 267 itil itil 11 8.4396868 110 ito Allah 267 itil 13 8 8.1607103 110 ito Allah 267 itil itil 3	97		in/with/hy/		قوة		173	3	14	8.6872425
100 السرافي السرافي 101 173 3 9 5.4087071 101 173 3 23 4.7873383 102 نهاس الخاليين the pure (religion) 11 4 11 8.6461372 103 نهاس الخاليين (So put not forward) similitudes for Allah) 5 3 5 8.6461372 104 istic (So put not forward (similitudes for Allah) 5 3 5 8.6461372 105 istic (Gor Allah) for Allah 5 3 5 8.6461372 106 istic istic (for Allah) istic 12 4 11 8.520606 107 istic (for Allah) istic		بالله		173	حول	(save by Allah	173	3	14	7.8316326
101 ١٥ ١٩ ١٦ 3 23 4.7873383 102 نهای الله وست (religion) 11 4 11 8.6461372 103 نگایش (So put not forward) similitudes (for Allah) 5 3 5 8.6461372 104 is active (for Allah) (So) put not forward (similitudes for Allah) 5 3 5 8.6461372 105 is active (for Allah) is active (for Allah) is active (for Allah) 5 3 5 8.6461372 106 is active (for Allah)'s sake only) 12 4 11 8.520606 107 Al-Muttagen (those who fear almoute (who abstain from evil good deeds) 9 3 8 8.4762125 108 alogo deeds) it he religion 13 4 11 8.4051294 109 it oo Allah 267 it i	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				الإيمان	faith	173	4	14	6.0027442
102 نوابت الغاليين 103 (So put not forward) similitudes for Allah) 5 3 5 8.6461372 104 (So put not forward) similitudes for Allah) 5 3 5 8.6461372 105 (So) put (not) forward (similitudes for Allah) 5 3 5 8.6461372 105 praise 67 5 63 8.5573282 106 (Allah) 12 4 11 8.520606 107 (Allah uch (who abstain from evil deeds and perform all kinds of good deeds) 9 3 8 8.4762125 108 uwaat to Allah 267 (the religion 13 4 11 8.4396868 109 to Allah 267 (the religion 13 4 11 8.4396868 110 to Allah 267 (the code of the religion 13 4 11 8.4396868 110 the code of the religion 13 4 11 8.4396868 111 the code of the religion 13 4 11 8.1607103 111 the lamin (mankind, jinn and	100				الشرك	associationism,/polytheism	173	3	9	5.4087071
103 نائلتان (So put not forward) similitudes (for Allah) 5 3 5 8.6461372 104 (So) put (not) forward (similitudes for Allah) 5 3 5 8.6461372 105 acation praise 67 5 63 8.5573282 106 if acation if acation if acation if acation 12 4 11 8.520606 107 Allah much (who abstain from evil good deeds) 9 3 8 8.4762125 108 acation allah much (who abstain from evil good deeds) 9 3 8 8.4396868 109 ato Allah 267 if and performal kinds of good deeds) 3 13 8.4396868 109 ato Allah 267 if adot if acation 13 4 11 8.4051294 110 if adot if acation if adot if acation if adot if acation 3 5 8.1607103 111 if adot if acation if adot if acation if adot if acation 7 3 5 7.78685298 113 if adot if acation if adot if acation	101					but	173	3	23	4.7873383
103 103 104 105 105 106 106 (for Allah) 5 3 5 8.6461372 105 105 106 107 108 111 8.520606 107 108 104 111 8.520606 108 109 104 111 8.4396868 109 107 103 113 8.4396868 109 107 103 8.4396868 111 8.4396868 109 108 103 4 111 8.4396868 109 105 103 4 111 8.4396868 110 107 103 8 8.1867056 111 114 267 104 111 3 8 8.1867056 111 113 8 8.1867056 111 3 5 7.78685298 1112 114 114 114 114 114 114 114 114 114 114 114 114 114 115 115 116 116 114	102				الْخَالِصُ		11	4	11	8.6461372
10410510610710710710810610610710610710611118.520606107107108107108	103				الأمثال	(for Allah)	5	3	5	8.6461372
106 المتعن المعني	1. 1. 1. 1.				تَضْرِبُوا	for Allah)				
106 (for Allah's sake only) 12 4 11 0.320000 107 Al-Muttaqeen (those who fear Allah much (who abstain from evil deeds and perform all kinds of good deeds) 9 3 8 8.4762125 108 المتقين د 15 3 13 8.4396868 109 it to Allah 267 itune 11 3 8 8.4051294 110 ith condition 13 4 11 8.4051294 110 ith condition 13 4 11 8.4051294 110 ith condition 7 3 5 8.1607103 111 ithe Lord of 11 3 8 8.1607103 112 ithe Alamin (mankind, jinn and all that exists) (part of a Qur'anic verse) 12 3 7 7.8685298 113 ithe Alamin (mankind, jinn and all that exists) 14 <td>105</td> <td></td> <td></td> <td></td> <td>الحمد</td> <td></td> <td>67</td> <td>5</td> <td>63</td> <td>8.5573282</td>	105				الحمد		67	5	63	8.5573282
107 انتقان Allah much (who abstain from evil deeds and perform all kinds of good deeds) 9 3 8 8.4762125 108 ion Allah 267 ion	106				مخلِصًا	(for Allah's sake only)	12	4	11	8.520606
106 100 100 100 100 100 100 110 13 4 11 8.4051294 110 110 110 11 3 8 8.1867056 111 110 110 110 3 8 8.1867056 111 110 110 110 3 8 8.1867056 111 110 110 110 3 8 8.1867056 112 110 110 110 110 110 3 8 112 110 110 110 110 110 110 110 113 110 110 110 110 110 110 114 110 110 110 110 110 110 115 110 110 110 110 110 110 116 110 110 110 110 110 110 116 110 110 110 110 110 110 110 110 110 110 110 110 110 111 110 110 110 110 110 110 111 110 110 110 110 110 110 111 110 110 110 110 110 110 111 110 110 110 110 110 110 111 110 110 110 110 110 110 111 110 </td <td>107</td> <td></td> <td></td> <td></td> <td>لامتقين</td> <td>Allah much (who abstain from evil deeds and perform all kinds of</td> <td>9</td> <td>3</td> <td>8</td> <td></td>	107				لامتقين	Allah much (who abstain from evil deeds and perform all kinds of	9	3	8	
109 Δa to Allah267 D_{cl} D_{cl} D_{cl} D_{cl} D_{cl} 110 110 110 111 3 8 8.1867056 111 111 3 8 8.1867056 111 111 111 3 8 8.1867056 112 111 111 3 5 8.1607103 112 111 1111 1111 1111 11111 112 11111 111111 1111111 1111111111111 113 $111111111111111111111111111111111111$	108				لرسوله	to His messenger	15	3	13	
110 (1) (1) (1) (1) (1) (1) (1) (1) (1) 111 (1) (1) (1) (1) (1) (1) (1) (1) (1) 111 (1) (1) (1) (1) (1) (1) (1) (1) (1) 112 (1) (1) (1) (1) (1) (1) (1) (1) (1) 113 (1) (1) (1) (1) (1) (1) (1) (1) (1) 113 (1) (1) (1) (1) (1) (1) (1) (1) (1) 114 (1) (1) (1) (1) (1) (1) (1) (1) (1) 114 (1) (1) (1) (1) (1) (1) (1) (1) (1) 114 (1) (1) (1) (1) (1) (1) (1) (1) (1) 114 (1) (1) (1) (1) (1) (1) (1) (1) (1) 115 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1) (1) 116 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1) 116 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1) 116 (1) (1) (1) (1) </td <td>109</td> <td>لله</td> <td>to Allah</td> <td>267</td> <td>الدّينُ</td> <td>the religion</td> <td>13</td> <td>4</td> <td>11</td> <td></td>	109	لله	to Allah	267	الدّينُ	the religion	13	4	11	
111 أينا العاليين 112 أينا العاليين 112 أينا العاليين 113 أينا في	110		to ritiun		رَبٌ		11	3	8	8.1867056
112 نا الغاليين 113 that exists) (part of a Qur'anic 12 3 7 7.8685298 113 غقبة end 9 3 5 7.7981405 114 غقبة end 40 3 22 7.7836409 114 the Alamin (mankind, jinn and all that exists) 40 3 22 7.7836409 115 performing worship practices sincerely (for Allah's sake alone) 27 3 14 7.6986046 116 هحمه 91 4 46 7.6619048 116 هحمه 91 2 5 7.3831029	111				نَهَوُا	do	7	3	5	8.1607103
113 ألب العالمين 113 ألب العالمين 114 ألب العالمين 114 ألب العالمين 114 ألب العالمين 115 أوب المالي 115 أوب المالي 116 أوب المالي ألب المالي أوب المالي 116 أوب المالي أوب المالي أوب المالي أوب الما	112				الْعَالَمِينَ	that exists) (part of a Qur'anic	12	3	7	
114 the Alamin (mankind, jinn and all that exists) 40 3 22 7.7836409 115 that exists) performing worship practices sincerely (for Allah's sake alone) 27 3 14 7.6986046 116 Alone 91 4 46 7.6619048	113				عَاقدَةُ		9	3	5	7.7981405
115 performing worship practices sincerely (for Allah's sake alone) 27 3 14 7.6986046 116 وحده 91 4 46 7.6619048 116 وحده 91 2 5 7.3831029						the Alamin (mankind, jinn and all that exists)	40	3	22	7.7836409
116 91 4 46 7.6619048 116 وحده 91 7.6619048	115				إخلاص	performing worship practices	27	3		
	116				وحده		91			
Vice Vice	117				المُنْكَر	vice	12	3	5	7.3831029

118	رب	Lord of	61	3	22	7.1748314
119	العاقبة	the (blessed) end	29	3	8	6.7881565
120	العبادة	the worship	120	3	30	6.6461372
121	الا	not to	61	4	11	6.1748314
122	الصبلاة	blessings/ prayer	289	4	36	5.6411366
123	السلام	the Peace/ peace / Islamic salutation	255	4	28	5.4591389
124	فلا	do not	97	3	7	4.8535795
125	العبادة	the worship	120	3	8	4.7392468
126	يعلم	knows	87	3	5	4.5251217
127	الدين	the religion	222	4	12	4.4366841
128	اما	and then	184	3	7	3.9299302
129	سبحانه	the Glorified	809	5	30	3.8930321
130	له	to Him/him	387	4	13	3.7503872

N Word form Eng. trans. Freq. Collocates Eng. trans. F. Texts Joint MI 1 الْخَالِصُ the pure 11 4 11 9.19973 فَاعْبُد 2 So worship (Allah) 12 4 11 9.0742 مخلصنا 3 sincerely 12 11 9.0742 4 (Allah) makes him know (the 4 4,001 7 3 6 8.97733 religion) 5 خيرا (to be) good/faithful 19 3 6 7.53676 6 dav 28 110 4 7.22572 يوم 7 بإحسان followed them exactly (in faith) 51 3 12 7.11226 8 VI Surely 6.72842 61 4 11 9 له to Him/him 387 4 30 5.51043 الدين the religion 222 10 after 209 3 5.19281 بعد 13 11 لله to Allah 267 4 12 4.72399 12 بل rather 140 3 6 4.65541 فقد indeed 138 3 5 13 4.41313 الذي 14 that 365 3 12 4.27294 المسامين Muslims 3 7 4.1869 15 226 إلى to Him/him 1104 4 33 4.1356 16 كان 17 289 3 8 4.0248 was 570 3 18 به with him 13 3.74535 703 3 16 3.74235 ail 19 this 4 27 337 6.37515 20 الإسلام Islam 242 3 15 6.00489 21 126 الحق the true (guidance) دين religion 22 الله Allah 4253 4 86 4.38886 raise the prestige of (His word 6 3 5 10.5413 يعلى 23 and make it) superior 3 8 10.4824 كلمته His word 10 24 (may Allah) render (His 16 3 7 9.61167 25 ينصر religion) victorious 7 29 3 8.75369 jurisprudence 4 ail 26 His/his 73 دينه 9 3 religion Muslims 226 6.15407 المسلمين 27 3 6 حتى 234 5.51892 till 28 1605 3 25 4.79982 أن that 29 1398 3 17 4.44263 30 عليه on him 570 3 6 4.23446 with it 31 به 234 3 6 4.73146 حتى till 32 religion 126 دين 226 3 7 6.73339 Muslims المسلمين 33 396 3 7 5.92421 الناس the people 34 their ۳۸ دينهم 3 8 3.94937 1779 على on him 35 religion 827 3 6 4.63943 about 36 عن

Table A7 Collocates of the key term 'RELIGION' in Ibn Baz's corpus

	N	Word clusters	Eng. trans.	Freq.	Texts
3-w	1	في الدين و	in the religion and	27	3
3-word clusters	2	في دينه و	in His/his religion	23	3
clust	3	في دين الله	in the religion of Allah	18	3
ters	4	الدين ألا لله	religious deeds sincerely for Allah's sake	11	4
	5	لله الذينُ الْخَالِصُ	the pure religion is for Allah only	11	4
	6	له الدين ألا	religion is for (Allah alone). Surely	11	4
	7	لهم دِينَهُمُ الذي	to them their religion that	11	4
	8	مُخْلِصًا له الدين	by doing religious deeds sincerely for Allah's sake	11	4
	9	في دينهم و	in their religion and	9	3
	10	و دين الحق	and the religion of the truth	8	3
	11	الفقه في دينه	comprehending the religion	7	3
	12	أن ينصر دينه	(may Allah) render His religion victorious	7	3
	13	دين الإسلام و	the religion of Islam and	7	3
	14	ينصر دينه و	(may Allah) render His religion victorious and	7	3
	15	يفقهه في الدين	makes him comprehend the religion	6	3
	16	دين الحق و	the religion of truth and	5	3
4-word clusters	17	دينه و يعلى	His religion and raise the prestige	5	3
	18	من دين الإسلام	from the religion of Islam	5	3
	19	ألا ش الدّينُ	Surely (the pure) religion is for Allah (only)	11	4
	20	لله الدِّينُ الْحَالِصُ	the pure religion is for Allah only	11	4
	21	الذينُ الْخَالِصُ و	the pure religion and	9	4
	22	الدين ألا بنه الدِّينُ	(by doing) religious deeds (sincerely for Allah's sake). Surely the (pure) religion	11	4
d clus	23	الله مُخَاصًا له الدين	(worship) Allah (Alone) by doing religious deeds sincerely for Allah's sake	11	4
ters	24	بإحسان إلى يوم الدين	followed them exactly (in faith) till the Day of judgement	11	3
	25	له الدين ألا ش	to Him the (pure) religion. Surely (the pure religion) is for Allah (Alone)	11	4
	26	مُخْلِصًا له الدين ألا	by doing religious deeds sincerely for Allah's sake. Surely	11	4
	27	ان ينصر دينه و	may (Allah) render His religion victorious and	7	3
	28	خير ا يفقهه في الدين	(if Allah wants to do) good (to a person), He makes him comprehend the religion	6	3
	29	دينه و يعلي كلمته	(victorious) and raise the prestige of His word and make it superior	5	3
	30	ينصر دينه و يعلي	(may Allah) render His religion victorious and raise the prestige (of His word) and make (it) superior	5	3
	31	دِينَهُمُ الذي ارْتَضَى لهم	their religion which He has chosen for them	8	4
	32	ألا شه الدِّينُ الْخَالِصُ	Surely the pure religion is for Allah only	11	4
	33	لله الدِّينُ الْخَالِصُ و	the pure religion is for Allah only and	9	4
7-word	34	الله مُخْلِصًا له الدين ألا لله الدِّينُ	(worship) Allah (Alone) by doing religious deeds sincerely for Allah's sake only. Surely (the pure) religion is for Allah (only)	11	4

Table A8 Clusters of the key term 'RELIGION' in Ibn Baz

35	فَاعْبُدِ الله مُخْلِصًا له الدين ألا ش	(worship) Allah (Alone) by doing religious deeds sincerely for Allah's sake only. Surely (the pure religion) is for Allah (only)	11	4
36	مُخْلِصًا له الدين ألا شر الدِّينُ الْخَالِصُ	by doing religious deeds sincerely for Allah's sake only. Surely the pure religion is for Allah only	11	4
37	له الدين ألا شه الدِّينُ الْحَالِصُ و	for Allah's sake only. Surely the pure religion is for Allah only and	9	4
38	إلا لِيَعْبُدُوا الله مُخْلِصِينَ له الدين حُنَّفَاءَ	(And they were commanded) not, but that they should worship Allah, and worship none but Him Alone (abstaining from ascribing partners to Him	7	1
39	أُمِرُوا إلا لِيَعْبُدُوا الله مُخْلِصِينَ له الدين	(And) they were commanded not, but that they should worship Allah, and worship none but Him Alone (abstaining from ascribing partners to Him	7	1
40	في دينه و الثبات عليه و أن	(comprehending His/his religion and sincere adherence (to its teachings)	7	2
41	النوم المملت لكم دينكم و انْمَمْتُ عَانِيْكُمْ	This day, I have perfected your religion for you, completed (My Favour) upon you	6	1
42	أَكْمَلْتُ لكم دِينَكْمْ و أَثْمَمْتُ عَلَيْكُمْ نِعْمَتِي	I have perfected your religion for you, completed My Favour upon you	6	1
43	ا نِعْمَتِي و رَضِيتُ لَكُمُ الْإِسْلامَ دِينًا و	(I have) completed My Favour upon you, and have chosen for you Islam as your religion. But	6	2
44	يرد الله به خيرا يفقهه في الدين	(if Allah) wants to do good to a person, He makes him comprehend the religion	6	3
45	الدين ألا لله الذِّبنُ الْحَالِصُ و الذين	religious deeds (sincerely for Allah's sake only). Surely the pure religion is for Allah only and those who	5	2
46	المسلمين للفقه في دينه و الثبات عليه	All Muslims comprehending the religion	5	2
47	ان ينصر دينه و يعلي كلمته و	(may Allah) render His religion victorious and raise the prestige of His word and make it superior and	5	3
48	دِينَكُمْ و أَتَمَمْتُ عَلَيْكُمْ نِعْمَتِي و رَضِيتُ	(I have perfected) your religion for you, completed My Favour upon you, and have chosen	5	1
49	عَلَيْكُمْ نِعْمَتِي و رَضِيتُ لَكُمُ الْإِسْلامَ دِينًا	(I have) completed My Favour upon you, and have chosen for you Islam as your religion	5	1
50	عز و جل الْيَوْمَ أَكْمَلْتُ لَكُم دِينَكْمُ	the Exalted in Mighty and the Majestic (said): This day, I have perfected your religion for you	5	2
51	لكم دِينَكُمْ و أَنْمَعْتُ عَلَيْكُمْ نِعْمَتِي و	(I have perfected) your religion for you, completed My Favour upon you, and	5	1
52	للفقه في دينه و الثبات عليه و	to comprehending His/his religion and sincere adherence to its teachings and	5	2
53	لله الدِّينُ الْخَالِصُ و الذين اتَّخَذُوا من	the pure religion is for Allah only. And those who take (Auliya' -protectors, helpers, lords, gods- besides) (Him)	5	2
54	من يَبْتَغ غَيْرَ الإسلام دِينًا قَلْن يُقْبَلَ	And whoever seeks a religion other than Islam, it will never be accepted (of him)	5	2

Table A9 Collocates	of the key term 'SA'	Y' in Ibn Baz's corpus
----------------------------	----------------------	------------------------

N	word form	Eng. trans.	F.	collocate	Eng. trans.	F.	Joint	MI
1			51	فاعبدون	therefore worship Me	10	8	6.281617
2				أنزلناه	We have sent down	9	7	6.240975
3				لقد	verily	12	9	6.188508
4				أوحِيَ	(what) has been revealed (to you (O Mohammad) from your Lord)	8	6	6.188508
5				أرْسَلْنَا	We have sent	19	14	6.162973
6				كِتَابٌ	A Book (revelation)	14	10	6.118119
7				لينصرن	verily (Allah) will aid	7	5	6.118119
8				ألغ	(did) you not	7	5	6.118119
9				سَأَلْتَهُمْ	(if) you (were to) ask them	7	5	6.118119
10				حَقًا	(it was) incumbent (upon Us)	7	5	6.118119
11				عَلَيْنَا	upon Us	7	5	6.118119
12				لَيَقُولُنَّ	they will certainly say	7	5	6.118119
13				نَصْرُ	help	7	5	6.118119
14				ذَلِكَ	that	11	7	5.951469
15	-			لذين	if (you) were to	10	6	5.86658
16				الحافظ	Alhafidh (a higher status of the scholars of Hadith)	17	10	5.838011
17		a started		الطَّاغُوتَ	false deities	16	9	5.77347
18				تُرْحَمُونَ	(that) you (may) obtain mercy	9	5	5.755548
19		10.0		عَاقِبَةُ	end	9	5	5.755548
20		the set		الْمَوْمِنِينَ	believers (acc.)	11	6	5.729076
21	قال	He/he said	1100	السميع	All-Hearer	11	6	5.729076
22				هَذَا	this	21	11	5.67066
23				لعلكم	that you may	37	19	5.64202
24				تعالى	the Exalted	477	242	5.624563
25				خافت	I (Allah) created	14	7	5.603545
26				تسليما	with full submission	11	5	5.466042
27				اغبُدُوا	you (mankind) worship	16	7	5.4109
28				نعم	yes/ best	14	6	5.381153
29				الْجِنَّ	Jinn (genies)	14	6	5.381153
30		11-11-11		قلت	I said	19	8	5.355618
31				يشاء	(Allah) wills (that something be done)	12	5	5.340511
32		-13-5		كما	as	514	206	5.284421
33				رحمه	(may Allah) have mercy on him	88	35	5.273397
34				عز	the Exalted in Mighty	363	130	5.122087
35				أبو	father of	59	21	5.11322
36				قل	you say (imperative)	89	31	5.082008
37			1.20	جل	the Majestic	424	145	5.055534
38	54			مالك	Malik	30	10	5.018583
30 39				لقد	verily	76	24	4.94058
39 40		- /			O (vocative)	156	48	4.903106
40	100			پ شيخ	Sheikh	20	6	4.86658
41				رَبِّكَ	your Lord	20	6	4.86658

43		121252.00		حيث	as	47	13	4.749396
44		12		ايها	O (another form of vocative)	104	26	4.603545
45				علا	the Most High	76	18	4.525543
46				سبحاته	the Glorified	809	187	4.490444
47				انه	that He/he	348	80	4.48253
48				ابن	son of	116	25	4.389421
49				إنَ	indeed	108	23	4.37222
50				الآية	the Qur'anic verse/ till the end of the Qur'anic verse	129	27	4.347206
51				إليك	to you	26	5	4.225034
52				لهذا	for this reason	48	9	4.188508
53				رضي	(may Allah) be pleased with him	182	32	4.095751
54				النبي	the Prophet	225	37	3.999218
55		199		سواہ	other than Him/him	37	6	3.979054
56			_	الإمام	the Imam	62	10	3.971277
57				أمنوا	(those) who believed	88	14	3.951469
58				الذين	those	261	38	3.823567
59				رسول	messenger	270	38	3.774657
60				عنه	with him	221	31	3.769839
61				سلم	(may Allah send) His peace (upon)	872	117	3.705726
62				يا	O (vocative)	156	12	6.597003
63				له	to him	387	9	4.871178
64	فقال	then He/he said	85	سبحانه	the Glorified	809	9	3.807371
65		The file salu		رسول	messenger	270	5	4.542555
66			in the second	سلم	(may Allah send) His peace (upon)	872	10	3.851186
67				أيها	O (vocative)	104	14	6.025846
68				تعالى	the Exalted	477	64	6.021085
69		0.02515		يا	O (vocative)	156	16	5.633528
70				أمنوا	(those) who have faith	88	9	5.629424
71				جل	the Majestic	424	33	5.235404
72	1.1			سبحانه	the Glorified	809	57	5.091825
73				عز	the Exalted in Mighty	363	24	5.000067
74		His/his		الذين	those	261	14	4.69838
75	قوله	saying	221	تُغبُدُوا	you worship	19	5	6.992931
76	12			معنى	the meaning (of)	46	8	6.395369
77				منها	from it/them	129	14	5.715058
78		1962		الا	not to	61	6	5.573156
79				الآية	till the end of the Qur'anic verse	129	9	5.077629
80				الآيات	the Qur'anic verse	138	7	4.617761
31				ذلك	that	990	31	3.921842
82	-			كما	as	514	15	3.820197

	N	Eng. trans	Eng. trans	Freq.	Texts
2	1	الله قال	() Allah said	17	3
-wo	2	إذ يقول	because He/he said	5	3
ord o	3	ان قال	(till) He/he said	11	3
2-word clusters	4	ان يقال	(it) could be said	13	3
ers	5	ان بقول	(till) He/he says	12	3
	6	انه قال	that He/he said	74	3
	7	بقوله تعالى	in His saying the Exalted	6	3
	8	ثم قال		21	5
	9	حيث قال		11	3
	10	ذلك بقوله	that in His/his saying	6	3
	11	ذلك قوله		13	3
	12	سلم قال		33	4
	13	فقال سبحانه		9	3
	14	فقال له	and he said to him	9	3
	15	في قوله	in His/his saying	34	4
	16	قال الإمام	the Imam said	10	3
	17	قال الحاقظ	Alhafidh (a higher status of the scholars of Hadith) said	10	3
	18	قال الله	Allah said	75	5
	19	قال النبي	the Prophet said	27	3
	20	قال إن	He/he said: Indeed	11	3
	21	قال أبو	the father of () said	14	3
	22	قال أهل	the people of () said	6	3
	23	قال تعالى	the Exalted said	194	5
	24	قال جل	the Majestic (and the Most High) said	11	3
	25	قال سبحانه	the Glorified said	157	5
	26	قال صلى	(the Prophet/messenger may Allah) send His blessings (and peace upon him) said	20	3
	27	قال عز	the Exalted in Mighty (and the Majestic) said	113	5
	28	قال عليه	(the Prophet/messenger may Allah's blessings and peace be)upon him said	13	3
	29	قال في	He/he said in	30	3
	30	قال لا	He/he said: No	16	3
-	31	قال له	he said to him	12	3
	32	قال من	He/he said: whosoever	43	4
	33	قال و	He/he said: And	20	3
	34	قول الله	the saying of Allah	9	4
	35	قوله تعالى	the saying of the Exalted	58	3
	36	قوله سيحانه	the saving of the Glorified	57	4
	37	قوله عز	the saying of the Exalted in Mighty (and the Most High)	24	3
	38	قوله و	His/his saying and	18	4
	39	كما قال	as He/he said	203	5
	40	لقول الله	because Allah has said	11	3
	41	لقول النبى	because the Prophet has said	7	3
	42	لهذا قال	because of that He/he said	9	3

Table A10 Clusters of the key term 'SAY' in Ibn Baz's corpus

	43	ما قاله	inde no salu	12	3
	44	ما يقول		5	3
	45	معنى قوله	the meaning of His/his saying	8	3
	46	من القول	including saying	6	3
	47	منها قوله	including His/his saying	14	3
	48	و قال		438	
	49	و قالوا			5
	50	و قوله		14	3
	51	و بقول		76	4
	52	و بيون و يقولون		39	4
				6	3
	53	يقول الله	y =	7	3
	54	يقول سبحانه	the Glorified says	27	4
	55	يقول في	He/he says in	8	3
	56	يقول و	He/he says: and	8	3
,	57	و قال تعالى	and the Exalted said	123	4
-	58	قال عز و	the Exalted in Mighty and Said	111	5
	59	قال تعالى و	the Exalted said and	92	4
	60	و قال سبحانه	and the Glorified said	82	4
	61	قال سبحانه و	the Glorified and (the Exalted) said	80	5
	62	سلم أنه قال	that (the Prophet may Allah send) His peace (upon him) said		
	63			55	3
		كما قال سبحانه	as the Glorified said	54	5
	64	كما قال تعالى	as the Exalted said	49	4
	65	كما قال عز	as the Exalted in Mighty said	41	4
	66	و سلم قال	(may Allah send His blessings) and peace (upon him) said	33	4
	67	كما قال الله	as Allah said	30	5
	68	قوله سبحانه و	His saying, the Glorified and	27	3
	69	قوله تعالى و	His saying, the Exalted and	24	3
	70	و قوله سبحانه	and His saying the Glorified	24	3
	71	-	that He/he said: whosoever	23	3
	72	قوله عز و	His saying, the Exalted in Mighty and (the Majestic)	23	3
	73 74	قال الله تعالى	Allah, the Exalted, said	22 22	3 5
1	75	قال الله سبحانه قال النبی صلی	Allah, the Glorified, said the Prophet (may Allah send His) blessings (on him) said	22	3
	76	كان اللبي صلى الأية و قال	till the end of the Qur'anic verse and He said	20	4
	77	قال صلى الله	he (may) Allah (send His) blessings (upon him) said	20	3
	78	و قال صلى	he (may Allah send His) blessings (upon him) said	17	3
	79	يقول سبحانه و	the Glorified and (the Exalted) said	17	4
	80	قال عليه الصلاة	he may the blessings (and peace of Allah be) upon him) said	13	3
	81	و قوله تعالى	and His saying, the Exalted	13	3
N CLASS	82	و يقول سبحانه	and the Glorified says	13	4
Course of the	83		the Glorified said in	12	3
1	84	و قال عليه	and he, (may the blessings and peace of Allah be) upon him, said	12	3
100000	85	في قوله تعالى	in His saying, the Exalted,	11	3
	86	قال جل و	the Majestic and (the Most High) said	11	3
	87	و قوله عز	and His saying, the Exalted in Mighty,	10	3
1	88	و جل قل	(Allah the Exalted in Mighty) and the Majestic said	9	3

	89	و لهذا قال	set this reason flor her her salu	9	3
	90	الله و قال	Allah and said	8	3
	91	رسول الله قال	Set of Finan Said	8	3
	92 93	قال تعالى يا		8	3
	93	الآية و قوله ذلك و قال	and this saying	7	3
	94			7	4
	95	لقول النبي صلى	because the Prophet (may Allah send) His blessings (and peace upon him) said	7	3
	96	و قال و	and He/he said and	6	3
	97	و كما قال	and as He/he said	6	3
	98 99	انه قال إن	that He/he said verily	5	3
	99	ثم قال سبحانه	then the Glorified said	5	3
	100	سلم كما قال	(may Allah send His (blessings and) peace (upon him) as He/he said	5	3
	101	قوله سبحانه يا	His saying (i.e. Qur'anic verses), the Glorified, O ya (vocative)	5	3
	102	قال عز و جل	the Exalted in Mighty and the Majestic said	111	5
4-w	103	و قال عز و	and the Exalted in Mighty and (the Majestic) said	62	4
ord	104	و قال تعالى و	and the Exalted said: And		
clu	101			58	4
4-word clusters	105	و سلم أنه قال	that (the Prophet/messenger may Allah send His blessings) and peace (upon him) that he said	55	3
	106	و قال سيدانه و	and the Glorified and (the Exalted) said	47	4
	107	كما قال عز و	as the Exalted in Mighty and (the Majestic) said	40	4
	108	كما قال تعالى و	as the Exalted said: And	27	4
	109	قوله عز و جل	His saying, the Exalted in Mighty and the Majestic	23	3
	110	كما قال سبحانه و	as the Glorified and (the Exalted) said	23	5
	111	قال النبي صلى الله	the Prophet, may Allah send His blessings and peace upon him, said	22	3
	112	قال تعالى و ما	the Exalted said: And whatsoever	22	4
	113	سلم أنه قال من	that (the Prophet/messenger may Allah) send His peace (upon him) said: Whosoever	21	3
	114	قال صلى الله عليه	he (the Prophet), may Allah send His blessings (and peace upon him) said	20	3
	115	قال سبحانه و تعالى	the Glorified and the Exalted said	17	3
	116	و قال صلى الله	and (the Prophet), may Allah send His blessings (and peace upon him), said	17	3
	117	و قوله سيحانه و	and in His saying, the Glorified and (the Exalted)	17	3
	118	قال الله عز و	Allah the Exalted in Mighty and the Majestic, said	15	3
	119	قال عليه الصلاة و	he (the Prophet of Islam), may Allah's blessings and (peace) be upon him, said	13	3
	120	و قال عليه الصلاة	and he (the Prophet of Islam), may Allah's blessings (and peace) be upon him, said	12	3
	121	قال تعالى و لقد	the Exalted said: And verily	11	4
	122	قال جل و علا	the Majestic and the Most High said	11	3
	123	قال سبحانه و ما	the Glorified said: And whatsoever/whosoever	11	3
	124	كما قال الله عز	as Allah, the Exalted in Mighty (and the Majestic), said	9	3
	125	و قوله عز و	and His saying (i.e. Qur'anic verses), the Exalted in Mighty and (the Majestic),	9	3
	126	في قوله تعالى و	in His saying, the Exalted: And	8	3
10.000 million	127	قال تعالى و من	the Exalted said: And whosoever	8	4
-	128	كما قال سيدانه و	as the Glorified and (the Exalted) said	8	4
	129	لقول النبي صلى الله	because the Prophet, may Allah send His blessings (and peace	7	3
		لغون النبي صلى الم	occurse me reperty of		-

1			upon him) said		1
	130	و يقول سبحانه و	and the Glorified (and the Exalted) says	7	3
	131	قال سبحانه و لقد	the Glorified said: Indeed	6	3
	132	الطَّاغُوتَ و قال تعالى	(taguut), viz., idols and Jinni. And the Exalted said	5	3
	133	سلم أنه قال إن	that (the Prophet (may Allah) send His peace upon him said: Indeed	5	3
	134	قوله تعالى و من	in His saying, the Glorified: And whosoever	5	3
	135	و سلم كما قال	(the Prophet/messenger may Allah) send His peace (upon him) as He/he said	5	3
1.1	136	و قال تعالى يا	and the Exalted said: O	5	3
	137	يا رسول الله قال	O messenger of Allah, he said	5	3
	138	النبي صلى الله عليه و سلم قال	the Prophet, may Allah send His peace and blessings upon him, said	23	4
7-	139	قال النبي صلى الله عليه و سلم	(said) the Prophet, may Allah send His peace and blessings upon him	22	3
WOF	140	الله عليه و سلم أنه قال من	(may) Allah send His blessings and peace upon him said	21	3
7-word clusters	141	و قال صلى الله عليه و سلم	and (the Prophet), may Allah send His peace and blessings upon him, said	17	3
ers	142	لقول النبي صلى الله عليه و سلم	because of the saying of the Prophet, may Allah send His blessings and peace upon him (i.e. Hadith)	7	3
	143	قال تعالى و ما أَرْسَلْنَا من قَبْلِكَ	The Exalted said: And We sent not before you (as Messengers) any but	5	3

APPENDIX B

Table B1 Keywords in Alhawali

V	Keyword	Eng. trans.	Texts	F.	%	F. Baz	% Baz	Keyness	Р
1	كما	as	5	583	0.2956	1		580.49	5E-18
2	كانت	(she/it) was/were	5	600	0.3043	49	0.0375	353.31	2E-17
1	الإنسان	the human	4	368	0.1866	19	0.0145	257.73	7E-17
1	التي	which	5	1230	0.6237	329	0.2518	250.18	8E-17
5	کان	(he/it) was	5	1113	0.5644	289	0.2212	237.6	9E-17
5	اوروبا	Europe	5	263	0.1334	4		233.32	1E-16
1	الحياة	life	5	394	0.1998	33	0.0253	229.16	1E-16
	۲ن	indeed, if	5	750	0.3803	192	0.147	163.56	4E-10
)	الخليج	the Gulf	5	161	0.0816	- 1		153.4	5E-10
0	الغرب	the west	5	181	0.0918	6		142.03	6E-16
1	المنطقة	the region	3	135	0.0685	0		137.28	7E-10
2	هي	she/it	5	635	0.322	169	0.1294	129.97	9E-1
3	الأمريكية	the American/ (Unites States) of America	5	137	0.0695	1		129.31	9E-10
4	النظرية	the theory	3	120	0.0609	0		122.03	1E-1
5	الثورة	the revolution	4	115	0.0583	0		116.94	1E-1
5	النطور	the development	3	113	0.0573	0		114.91	2E-1
7	المسيح	Christ	3	151	0.0766	6		113.65	2E-1
8	أي	or	5	333	0.1689	61	0.0467	111.29	2E-1
9	أمريكا	America	4	137	0.0695	4		110.29	2E-1
)	العالم	the world	5	305	0.1547	52	0.0398	109.48	2E-1
1	البشرية	mankind	4	123	0.0624	2		108.25	2E-1:
2	ان	indeed	5	3287	1.6668	1605	1.2285	105.16	2E-1
3	أيضبأ	too	5	148	0.075	8		102.11	3E-1:
4	الذي	which/that/ who	5	995	0.5046	365	0.2794	101.53	3E-1:
5	لكن	but	5	315	0.1597	60	0.0459	101	3E-1:
6	الفكر	the intellect/ the thinking	5	120	0.0609	3		99.336	3E-15
7	دول	governments/ countries	5	96	0.0487	0		97.616	3E-1
8	الدينية	the religious	4	104	0.0527	1		96.293	3E-1
9	الغربية	the western	4	110	0.0558	2		95.468	3E-1
)	المتحدة	the United/ United	3	93	0.0472	0		94.565	4E-1:
1	الأوروبية	the European	3	92	0.0467	0		93.548	4E-15
2	تلك	that	5	226	0.1146	33	0.0253	92.961	4E-15
3	الإنسانية	The human	3	90	0.0456	0		91.514	4E-15
4	جديدة	new	4	98	0.0497	1		90.309	4E-15

35	القرن	century	3	134	0.0679	8		89.418	5E-15
36	الشيو عية	communism	3	124	0.0629	6		88.502	5E-15
37	الأن	now/ the present time	5	123	0.0624	6		87.58	5E-15
38	لهلا	all	5	95	0.0482	1		87.32	5E-15
39	فكرة	idea	4	117	0.0593	5		86.433	6E-15
40	في	in	5	5531	2.8047	2982	2.2825	86.019	6E-15
41	مصر	Egypt	4	108	0.0548	4		82.665	7E-15
42	الواقع	occurring/ actuality	3	150	0.0761	14	0.0107	82.619	7E-15
43	العراق	Iraq	4	122	0.0619	7		82.528	7E-15
44	النظام	the system/ regime	3	88	0.0446	1		80.353	8E-15
45	يكن	be	5	149	0.0756	15	0.0115	78.774	9E-15
46	الولايات	the (United) States of America	3	77	0.039	0		78.293	9E-15
47	يمكن	may	5	154	0.0781	17	0.013	77.139	1E-14
48	إذ	suddenly/ as/ when	4	157	0.0796	19	0.0145	74.147	1E-14
49	بل	rather,/even/ however	5	458	0.2322	140	0.1072	72.498	1E-14
50	ظل	remain, endure, continue (masculine)	5	91	0.0461	3		71.478	2E-14
51	منذ	since	4	121	0.0614	10		70.768	2E-14
52	لاسيما	particularly	4	78	0.0396	1		70.423	2E-14
53	نفسه	itself/himself	5	180	0.0913	28	0.0214	70.209	2E-14
54	الطبيعي	the natural	3	83	0.0421	2		69.123	2E-14
55	الحرب	the war	5	112	0.0568	9		66.357	2E-14
56	رجال	men	4	123	0.0624	12		66.161	2E-14
57	الصراع	the conflict	5	65	0.033	0		66.09	2E-14
58	خلال	during, through	4	85	0.0431	3		65.779	2E-14
59	الدول	the governments,/ countries	5	115	0.0583	10		65.642	3E-14
60	نفسها	itself/herself	5	115	0.0583	10		65.642	3E-14
61	البشري	the human	4	64	0.0325	0		65.073	3E-14
62	بين	between	5	606	0.3073	220	0.1684	63.582	3E-14
63	الفكرة	the idea	4	71	0.036	1		63.492	3E-14
64	خاصة	specially	5	116	0.0588	11		63.354	3E-14
65	حين	when	5	123	0.0624	13		63.234	3E-14
66	لذلك	therefore/because of that/based on that/(to/for) that	4	136	0.069	17	0.013	62.827	3E-14
67	الشرق	the East	4	99	0.0502	7		61.971	4E-14
68	الأمريكي	the American	3	69	0.035	1		61.515	4E-14
69	الوجود	presence	3	74	0.0375	2		60.424	4E-14
and the second second		And the second							

70	الكبر ى	the greater	3	67	0.034	1		59.539	5E-14
71	عصر	era	3	83	0.0421	4		59.303	5E-14
72	اصبح	he/it became (masculine)	4	87	0.0441	5		58.814	5E-14
73	السياسة	politics	3	82	0.0416	4		58.381	6E-14
74	العقل	mind	3	116	0.0588	13		57.56	6E-14
75	الأولى	the first/ the early	5	132	0.0669	18	0.0138	57.263	6E-14
76	تمامأ	exactly	4	56	0.0284	0		56.938	7E-14
77	الجديد	the new	3	64	0.0325	1		56.579	7E-14
78	تعد	consider,/deem/ prepare	5	64	0.0325	1		56.579	7E-14
79	الأوروبي	European	3	55	0.0279	0		55.921	8E-14
80	أية	any	5	54	0.0274	0		54.904	9E-14
81	نظام	system,/regime	3	82	0.0416	5		54.304	1E-13
82	الحقيقة	the truth,/ fact	5	128	0.0649	18	0.0138	54.232	1E-13
83	السياسية	the politics	4	67	0.034	2		53.698	1E-13
84	الغربي	the western	3	67	0.034	2		53.698	1E-13
85	أصبحت	she/it became (fem.)	5	72	0.0365	3		53.536	1E-13
86	كذلك	also, as well	5	133	0.0674	20	0.0153	53.385	1E-13
87	التاريخ	the history	5	107	0.0543	12		53.069	1E-13
88	أمام	facing, in front of	3	71	0.036	3		52.601	1E-13
89	الفرنسية	the French	4	51	0.0259	0		51.854	1E-13
90	الحال	the case	5	59	0.0299	1		51.656	1E-13
91	الوسطى	the mid/ middle	4	59	0.0299	1		51.656	1E-13
92	قانلأ	saying that	3	59	0.0299	1		51.656	1E-13
93	حركة	movement	4	58	0.0294	1		50.673	2E-13
94	مرحلة	a stage	4	49	0.0248	0		49.82	2E-13
95	الحركة	the movement	4	57	0.0289	1		49.691	2E-13
96	فحسب	only	3	57	0.0289	1		49.691	2E-13
97	قبل	before	5	177	0.0898	38	0.0291	49.338	2E-13
98	الاجتماعي	the social	3	56	0.0284	1		48.709	3E-13
99	إسرانيل	Israel	3	83	0.0421	7		48.077	3E-13
100	أكثر	more	5	185	0.0938	42	0.0321	47.962	3E-13
101	الأقل	(at) least/ the least	4	47	0.0238	0		47.786	3E-13
102	العسكرية	the military	3	47	0.0238	0		47.786	3E-13
103	المعاصر	modern	3	47	0.0238	0		47.786	3E-13
104	مع	in addition,/and	4	513	0.2601	195	0.1493	47.028	4E-13
105	العربي	the Arab,/Arabic	5	65	0.033	3		47.018	4E-13
106	واحدة	one	5	89	0.0451	9		46.927	4E-13

107	الحديثة	the modern	4	54	0.0274	1		46.748	4E-13
108	يبدو	seem	3	54	0.0274	1		46.748	4E-13
109	الماضي	the past	3	44	0.0223	0		44.736	6E-13
110	النفسية	the psychological inclination	3	44	0.0223	0		44.736	6E-13
111	اليهودي	the Jewish	4	44	0.0223	0		44.736	6E-13
112	الديني	the religious	4	57	0.0289	2		44.165	8E-13
113	الصليبية	pertaining to the Crusades	4	51	0.0259	1		43.81	8E-13
114	مثلأ	for example	5	51	0.0259	1		43.81	8E-13
115	الدين	the religion	5	554	0.2809	222	0.1699	42.785	1E-12
116	لها	for it/her	5	310	0.1572	101	0.0773	42.706	1E-12
117	البابا	the Pope	4	42	0.0213	0		42.702	1E-12
118	حرية	freedom	4	42	0.0213	0		42.702	1E-12
119	العالمية	world	5	55	0.0279	2		42.272	1E-12
120	مبدأ	principle	4	41	0.0208	0		41.686	2E-12
121	يستطيع	can, have the ability to	3	79	0.0401	8		41.62	2E-12
122	تاريخ	history	5	59	0.0299	3		41.484	2E-12
123	ليست	not	3	92	0.0467	12		41.237	2E-12
124	العصور	ages	3	48	0.0243	1		40.88	2E-12
125	السوفيتي	Soviet	3	40	0.0203	0		40.669	3E-12
126	المعاصرة	Modern,/contemporary	5	40	0.0203	0		40.669	3E-12
127	فرنسا	France	3	40	0.0203	0		40.669	3E-12
128	معظم	most	3	40	0.0203	0		40.669	3E-12
129	انها	indeed she/it is	5	208	0.1055	57	0.0436	40.445	3E-12
130	الخاصة	the special	3	53	0.0269	2		40.384	3E-12
131	نتيجة	result	4	70	0.0355	6		40.232	3E-12
132	هذه	this	5	868	0.4402	395	0.3023	40.084	4E-12
133	الأحداث	the incidents	4	39	0.0198	0		39.652	4E-12
134	المدياسي	the political/ the politician	3	39	0.0198	0		39.652	4E-12
135	ظلت	remain, endure,/continue (fem.)	3	39	0.0198	0		39.652	4E-12
136	يتحدث	speak(s) (about)	4	38	0.0193	0		38.635	8E-12
137	ظهرت	appeared	4	60	0.0304	4		38.439	9E-12
138	الدولة	the government/ country	3	112	0.0568	20	0.0153	38.393	9E-12
139	جهة	side/perspective	4	81	0.0411	10		37.734	2E-11
140	العملية	practical	3	37	0.0188	0		37.618	2E-11
141	موقف	stand point/ point of view	5	50	0.0254	2		37.562	2E-11
142	منطقة	region	3	49	0.0248	2		36.625	4E-11

143	الاتحاد	the union/ Union	3	36	0.0183	0		36.602	5E-11
144	المرحلة	the stage	5	36	0.0183	0		36.602	5E-11
145	عسكرية	military	3	36	0.0183	0		36.602	5E-11
146	عملية	operation/ process	3	36	0.0183	0		36.602	5E-11
147	وضع	put	5	65	0.033	6		35.999	1E-10
148	الجديدة	the new	3	48	0.0243	2		35.689	2E-10
149	العسكر ي	the military	3	35	0.0177	0		35.585	2E-10
150	مشكلة	problem	3	35	0.0177	0		35.585	2E-10
151	صورة	picture	3	60	0.0304	5		34.95	5E-10
152	الحكومة	the government	3	56	0.0284	4		34.906	5E-10
153	القوة	the power/the force	3	56	0.0284	4		34.906	5E-10
154	القديمة	the old	5	47	0.0238	2		34.754	8E-10
155	أثر	preferred	3	47	0.0238	2		34.754	8E-10
156	الباحثين	researchers (acc.)	4	34	0.0172	0		34.568	1E-09
157	ناحية	side	3	34	0.0172	0		34.568	1E-09
158	تعد	consider/ prepare/ promise	5	41	0.0208	1		34.074	2E-09
159	فقط	only	5	55	0.0279	4		34.028	3E-09
160	مهما	whatsoever	5	55	0.0279	4		34.028	3E-09
161	حد	extent	5	66	0.0335	7		33.855	3E-09
162	الوحيد	the only	3	46	0.0233	2		33.822	3E-09
163	حدث	happened	5	46	0.0233	2		33.822	3E-09
164	عندما	when	4	79	0.0401	11		33.724	3E-09
165	جزءأ	part	3	33	0.0167	0		33.551	4E-09
166	دينية	religious	5	33	0.0167	0		33.551	4E-09
167	لم	not, never	5	808	0.4097	377	0.2886	32.941	7E-09
168	يعيش	live(s)	4	32	0.0162	0		32.534	9E-09
169	هناك	there	5	141	0.0715	35	0.0268	32.18	1E-08
170	تحت	under	4	100	0.0507	19	0.0145	32.128	1E-08
171	تطور	developed (v.)/ development	3	31	0.0157	0		31.518	2E-08
172	شيئاً	thing	5	85	0.0431	14	0.0107	31.495	2E-08
173	تكن	is/are	5	63	0.0319	7		31.423	2E-08
174	العلاقات	the relationships	3	38	0.0193	1		31.174	2E-08
175	بان	that/ indeed	5	216	0.1095	69	0.0528	31.096	2E-08
176	فقد	already	5	356	0.1805	138	0.1056	30.672	3E-08
177	النصر انية	Christianity	5	51	0.0259	4		30.543	3E-08
178	فكان	and (something/someone) was/were	4	51	0.0259	4		30.543	3E-08

179	الأزمة	the crises	4	30	0.0152	0		30.501	3E-08
180	المقدس	the holy	3	30	0.0152	0		30.501	3E-08
181	الوطنية	patriotism	3	30	0.0152	0		30.501	3E-08
182	المادية	materialistic/ materialism	3	42	0.0213	2		30.11	4E-08
183	التاريخي	the historical	3	29	0.0147	0		29.484	5E-08
184	الكنائس	the churches/ Churches	3	29	0.0147	0		29.484	5E-08
185	انا	Ι	3	29	0.0147	0		29.484	5E-08
186	تركيا	Turkey	3	29	0.0147	0		29.484	5E-08
187	مجلس	council	3	29	0.0147	0		29.484	5E-08
188	تقول	(she/it) says	3	57	0.0289	6		29.368	6E-08
189	لدى	at/ upon	5	57	0.0289	6		29.368	6E-08
190	الفكر ي	intellectual	4	36	0.0183	1		29.247	6E-08
191	جانب	side	3	41	0.0208	2		29.188	6E-08
192	البحث	research	3	53	0.0269	5		29.021	7E-08
193	الحضارة	civilization	3	45	0.0228	3		28.828	8E-08
194	حسين	Hussein	3	45	0.0228	3		28.828	8E-08
195	الوقت	time	4	81	0.0411	14	0.0107	28.674	8E-08
196	الإسلامي	Islamic	4	178	0.0903	54	0.0413	28.586	9E-08
197	يرى	consider	5	75	0.038	12		28.529	9E-08
198	تقريباً	almost	5	28	0.0142	0		28.467	9E-08
199	جزء	part	3	28	0.0142	0		28.467	9E-08

Ν	Keyword	Eng. trans.	Texts	Freq.	%	F. Baz	% Baz	Keyness	Р
1	أوروبا	Europe	5	263	0.1334	4		233.32	1.00E-16
2	الخليج	the Gulf	5	161	0.0816	1		153.4	5.00E-16
3	الغرب	the West	5	181	0.0918	6		142.03	6.00E-16
4	النظرية	the theory	3	120	0.0609	0		122.03	1.00E-15
5	الثورة	the revolution	4	115	0.0583	0		116.94	1.00E-15
6	الفكر	the intellect/ the thinking	5	120	0.0609	3		99.336	3.00E-15
7	الدينية	the religious (def. adj. fem. pl)	4	104	0.0527	1		96.293	3.00E-15
8	الحرب	the war	5	112	0.0568	9		66.357	2.00E-14
9	الصراع	the conflict	5	65	0.033	0		66.09	2.00E-14
10	عصر	era	3	83	0.0421	4		59.303	5.00E-14
11	السياسة	politics	3	82	0.0416	4		58.381	6.00E-14
12	العقل	mind	3	116	0.0588	13		57.56	6.00E-14
13	الأوروبي	European (sing. adj. masc.)	3	55	0.0279	0		55.921	8.00E-14
14	السياسية	the political (adj. pl.)	4	67	0.034	2		53.698	1.00E-13
15	الغربي	the western (pl. adj. masc.)	3	67	0.034	2		53.698	1.00E-13
16	التاريخ	the history (def.)	5	107	0.0543	12		53.069	1.00E-13
17	الوسطى	the middle (ages)	4	59	0.0299	1		51.656	1.00E-13
18	حركة	movement	4	58	0.0294	1		50.673	2.00E-13
19	الحركة	the movement	4	57	0.0289	1		49.691	2.00E-13
20	الاجتماعي	the social (contract)	3	56	0.0284	1		48.709	3.00E-13
21	إسرانيل	Israel	3	83	0.0421	7		48.077	3.00E-13
22	العسكرية	the military (def. adj. fem.)	3	47	0.0238	0		47.786	3.00E-13
23	المعاصر	contemporary	3	47	0.0238	0		47.786	3.00E-13
24	الحديثة	the modern	4	54	0.0274	1		46.748	4.00E-13
25	الماضى	the past	3	44	0.0223	0		44.736	6.00E-13
26	الدينى	the religious (def. adj. masc. pl.)	4	57	0.0289	2		44.165	8.00E-13
27	الصليبية	pertaining to the Crusades	4	51	0.0259	1		43.81	8.00E-13
28	الدين	the religion	5	554	0.2809	222	0.1699	42.785	1.00E-12
29	العالمية	the () World War/ the international	5	55	0.0279	2		42.272	1.00E-12
30		history (indef.)	5	59	0.0299	3		41.484	2.00E-12
31	العصور		3	48	0.0243	1		40.88	2.00E-12
	المعاصرة	ages the modern	5	40	0.0203	0		40.669	3.00E-12
32	المعاصرة	the political (masc. adj.) the	5	10					
33	السياسي	politician	3	39	0.0198	0		39.652	4.00E-12
34	منطقة	region	3	49	0.0248	2		36.625	4.00E-11
35	عسكرية	military (indef. adj. fem.)	3	36	0.0183	0		36.602	5.00E-11
36	العسكر ي	the military (def. adj. masc.)	3	35	0.0177	0		35.585	2.00E-10
37	دينية	religious (indef. adj. fem. pl.)	5	33	0.0167	0		33.551	4.00E-09
38	النصرانية	Christianity	5	51	0.0259	4		30.543	3.00E-08
39	الوطنية	patriotism	3	30	0.0152	0		30.501	3.00E-08
40	المادية	materialistic/materialism	3	42	0.0213	2		30.11	4.00E-08
41	التاريخي	the historical	3	29	0.0147	0		29.484	5.00E-08
42	الفكري	intellectual	4	36	0.0183	1		29.247	6.00E-08
43	الحضارة	civilization	3	45	0.0228	3		28.828	8.00E-08

Table B2 Keywords that indicate semantic foci in Alhawali's corpus

Table B3 Keywords that indicate social actors in Alhawali's corpus

Ν	Keyword	Eng. trans.	Texts	F.	%	F. Baz	% Baz	Keyness	Р
1	الإنسان	the human	4	368	0.1866	19	0.014543	257.73474	7.14E-17
2	أوروبا	Europe	5	263	0.13336	4		233.32214	1.01E-16
3	الخليج	the Gulf	5	161	0.08164	1		153.40109	4.70E-16
4	الغرب	the west	5	181	0.09178	6		142.03424	6.35E-16
5	الأمريكية	the American (fem.)/(Unites States) of America	5	137	0.06947	1		129.30927	9.24E-16
6	المسيح	Christ	3	151	0.07657	6		113.65211	1.58E-15
7	أمريكا	America	4	137	0.06947	4		110.28928	1.80E-15
8	العالم	the world	5	305	0.15466	52	0.0398019	109.48082	1.86E-15
9	البشرية	mankind	4	123	0.06237	2		108.24721	1.96E-15
10	دول	governments/ countries	5	96	0.04868	0		97.615776	3.11E-15
11	الغربية	the western	4	110	0.05577	2		95.467979	3.45E-15
12	المتحدة	the United/ United	3	93	0.04715	0		94.56472	3.60E-15
13	الأوروبية	the European	3	92	0.04665	0		93.547707	3.79E-15
14	الشيو عية	communism/ communist	3	124	0.06287	6		88.502434	4.94E-15
15	مصر	Egypt	4	108	0.05476	4		82.665268	6.94E-15
16	العراق	Iraq	4	122	0.06186	7		82.527977	7.00E-15
17	النظام	the regime	3	88	0.04462	1		80.35331	8.03E-15
18	الو لايات	the (United) States of America/states	3	77	0.03904	0		78.29303	9.21E-15
19	رجال	men	4	123	0.06237	12		66.161026	2.41E-14
20	الدول	the governments/ the countries	5	115	0.05831	10		65.642349	2.53E-14
21	الأمريكي	the American	3	69	0.03498	1		61.514648	3.84E-14
22	الأوروبي	European	3	55	0.02788	0		55.921146	7.60E-14
23	الغربي	the western	3	67	0.03397	2		53.697906	1.05E-13
24	إسرائيل	Israel Arab/Arab	3	83	0.04208	7		48.0769	2.87E-13
25	العربي	(world)/Arabic/Arabian (Gulf)	5	65	0.03296	3		47.018147	3.63E-13
26	اليهودي	the Jewish	4	44	0.02231	0		44.735939	6.47E-13
27	البابا	the Pope	4	42	0.02129	0		42.702316	1.21E-12
28	السوفيتي	Soviet	3	40	0.02028	0		40.668713	2.70E-12
29	فرنسا	France	3	40	0.02028	0		40.668713	2.70E-12
30	الدولة	the government/the country	3	112	0.05679	20	0.0153084	38.392731	9.44E-12
31	الاتحاد	the union/ Union	3	36	0.01825	0		36.601551	4.54E-11
32	الباحثين	researchers (acc.)	4	34	0.01724	0		34.567993	1.19E-09
33	الكنائس	the churches/ Churches	3	29	0.0147	0		29.484173	5.35E-08
34	انا	I	3	29	0.0147	0		29.484173	5.35E-08
35	تركيا	Turkey	3	29	0.0147	0		29.484173	5.35E-08
36	حسين	Hussein	3	45	0.02281	3		28.828354	7.62E-08
37	الإسلامي	Islamic/Muslim (World)	4	178	0.09026	54	0.0413328	28.586121	8.67E-08

Table B4 Keywords that serve as argumentation indicators in Alhawali's corpus

N	Keyword	Eng. trans.	Texts	F.	%	F. Baz	% Baz	Keyness	Р
1	!ن	indeed, if	5	750	0.3803	192	0.147	163.56	4.00E-16
2	لكن	but	5	315	0.1597	60	0.0459	101	3.00E-15
3	يمكن	may	5	154	0.0781	17	0.013	77.139	1.00E-14
4	بل	rather,/even/ however	5	458	0.2322	140	0.1072	72.498	1.00E-14
5	لذلك	therefore/because of that/based on that/(to/for) that	4	136	0.069	17	0.013	62.827	3.00E-14
6	نتيجة	result	4	70	0.0355	6		40.232	3.00E-12
7	موقف	stand point/ point of view	5	50	0.0254	2		37.562	2.00E-11
8	الباحثين	researchers (acc.)	4	34	0.0172	0		34.568	1.00E-09
9	أنا	I	3	29	0.0147	0		29.484	5.00E-08
10	تقول	(she/it) says	3	57	0.0289	6		29.368	6.00E-08

Table B5 Collocates of the key term 'ALLAH' in Alhawali's corpus

N	keyword	Eng. trans.	F.	Collocates	Eng. trans.	F.	Texts	Joint	MI
1	الله	Allah	994	بإذن	with His/his permission	8	3	8	7.623642445
2				رحمه	(may Allah have) mercy (upon him)	25	4	25	7.623642445
3				صلى	(may Allah send His) blessings (upon him)	90	5	90	7.623642445
4				رضي	may Allah be) pleased (with him)	18	3	14	7.261072159
5				النبى	the Prophet	21	3	15	7.138215542
6				ابن	son of	43	3	23	6.720939636
7				رحمة	mercy	12	3	6	6.623642445
8				عبد	the servant of/ the first of a compound the name 'Abdu Allah'/ worshipped (v.)	53	3	16	5.895721912
9				ر سوله	His messenger	34	5	10	5.858107567
10				ذكر	(he) mentioned	35	3	9	5.664284229
11				محمد	Mohammad	78	3	17	5.425703049
12				كتاب	Book/book	82	5	15	5.172980785
13				سنة	(the) way (of Allah)/ year/ the Sunnah of the Prophet	160	5	25	4.945570469
14				قال	He/he said	195	3	25	4.660168171
15				امر	He/he ordered	51	3	6	4.536179543
16				الا	except	514	4	52	4.318457603
17				عند	near/as	137	4	13	4.2260499
18				تعالى	the Exalted	151	4	14	4.192592621
19	ىللە		74	الحمد	praise	17	4	9	10.45375347
20	بالله		41	Y	no	1787	3	6	4.004830837

Tubi	the 20 constants of and hey term sentext in minawait's corpus											
N	key term	Eng. trans	<u> </u>	collocates	Eng. trans	F.	Texts	Joint	MI			
1	القرن	the century	134	مطلع	(the) beginning of	16	3	9	9.684579849			
2				الماضىي	last	44	2	17	9.142686844			
3				هذا	this	1058	2	13	4.167971134			
4				الميلادي	the Gregorian	7	2	6	10.29226303			
5				عشر	tenth	88	2	62	10.00941944			
6				التاسع	ninth	37	2	30	10.2120924			
7				الثامن	eighth	23	2	20	10.31302166			
8				السابع	seventh	17	2	10	9.749120712			
9				العشرين	twentieth	20	2	19	10.44065475			
10				قد	indeed	690	1	8	4.084202766			
11				الرابع	fourth	22	1	8	9.055223465			
12	القرون	centuries	57	الوسطى	the middle	59	3	37	11.07466507			
13				الأولى	first	132	2	7	7.510815144			
14				كانت	(they/it) were	600	2	5	4.84096384			
15				طيلة	all	13	1	7	10.85476971			
16				على	on	2563	1	12	4.009199142			
17				کان	(it) was	1113	1	7	4.434971333			

Table B6 Collocates of the key term 'CENTURY' in Alhawali's corpus

Appendix C

Transliteration¹⁰⁴

Arabic letter shape	Transliteration Symbol
ç	,
ų	b
ب ت ث	t
ث	<u>t</u>
う	<u>t</u> <u>ğ</u>
<u>ح</u> خ	<u> </u>
<u></u> ζ	<u> </u>
٢	d
i	<u>d</u>
ر ز	r
	Z
س ش ص ط	<u> </u>
ش	š
ص .	<u>Ş</u>
ص	d t
<u>م</u> ظ	t
	Z.
ع ع غ ق ك	<u>ģ</u> f
رة	q
ك	k
J	1
م	m
ن	n
ھـ	h
و	W
ى	У

Short vowels:¹⁰⁵

fatḥah	Ó	а	
kasrah	<u> </u>	i	
ḍammah	ं	0	

Long vowels:

ā	
ī	
ū	

 ¹⁰⁴. I follow the DIN 31635 standard for the transliteration of the Arabic alphabet (available on the following link: <u>http://transliteration.eki.ee</u>).
 ¹⁰⁵. The round characters resemble the Arabic letters that take the diacritics around it to indicate short vowels.

Appendix D

Concordance D1 Expanded Concordance lines of Argumentation threads in Ibn Baz

N	Expanded Concordance lines (argumentation threads 1-4)	
	من العقائد المضادة للحق ما يعتقده بعض المتصوفة : من أن بعض من يسمونهم بالأولياء يشاركون الله في التدبير ، ويتصرفون في	
	العالم، ويسمونهم بالأقطاب و الأوتاد و الأغواث ، و غير اذلك من الأسماء التي اخترعوها لألهتهم ، و هذا من أقبح الشرك	
هو شر من شرك حاهلية العرب؛ لأن كفار العرب لم يشركوا في الربوبية و إنما أشركوا في العبادة ، و كان شركهم		
	إ في حال الرخاء ؛ أما في حال الشدَّة فنخاصون بله العبادة ؛ كما قال الله سيحانه : (فاذا رَكَبُوا في الفلك دعوًا الله مُخلصين له الدين قلمًا ا	
	ا نَحَاهُمُ إِلَى الْأَرَ هُمُ يُشْرُ كُونَ) أما إلا بوينة فكانوا معتر فين بها لله وحده ، كما قال سيحانه : (و لنن سالتهم من خلفهم ليفولن الله) ، في	
	قال تعالى بير إلى المدير المدير في المدير المدين المرابع المائي السَّمْعَ وَ الأَبْصَارَ و مَن يُخْرِجُ الْحَيَّ مَن الْمَيَّتِ و يُخْرِجُ الْمَيَّتَ من الْحَيَّ و	
	مَنْ يُدَبِّرُ أَلْأَمْرُ فَسَيَقُولُونَ الله فَقُلْ أفلا تَتَقُونَ) و الآيات في هذا المعنى كثيرة. أما المشركون المتأخرون فزادوا علي الأولين من جهتين	
	، إحداهما : شرك بعضهم في الربوبية ، و الثانية : شركهم في الرخاء و الشدة ، كما يعلم ذلك من خالطهم و سبر أحوالهم ، و رأى ما	
	يفعلون عند قبر الحسين و البدوي و غير هما في مصر ، وعند قبر العيدروس في عدن ، و الهادي في اليمن و ابن عربي في الشام ،	
	[] وغير ها من القبور المشهورة التي غلت فيها العامة و صرفوا لما الكثير من حق الله عز و جُل []	
	Some other beliefs that oppose the true guidance of Allah are that of some Sufis: [Some Sufis]	
	belief that the so called 'Aulliya' (friends of God) share Allah in disposing the affair of all things in the	
	world. They name them 'the poles', the pillars', 'the aid providers' (ranks of Sufi saints) and other names they invented for their gods, which is more repugnant form of polytheism (that opposes the	
	Unity of Lordship). It is even more wicked than the polytheistic beliefs of the ignorant Arabs in the	
	pre-Islamic era. The case is so, <u>because</u> those Arab infidels (in the pre-Islamic period) did not	
	practice polytheism that opposes the Unity of Lordship, but practiced the form of polytheism that	
1	opposes the Unity of worship. In addition, they practiced polytheism during times of comfort, but	
-	during times of hardships they sincerely worship Allah making their faith pure for Him Alone, as	
	Allah, the Glorified, said: {And when they embark on a ship, they invoke Allah, making their Faith	
	pure for Him only: but when He brings them safely to land, behold, they give a share of their worship	
	to others}. Also, they do not contradict the Unity of Lordship as they confirm that Allah is their Lord,	
	as the Glorified said: { And if you ask them who created them, they will surely say: "Allah"}, and the	
	Exalted said: {Say (O Muhammad): "Who provides for you from the sky and the earth? Or who owns	
	hearing and sight? And who brings out the living from the dead and brings out the dead from the	
	living? And who disposes the affairs?" They will say: "Allah." Say: "Will you not then be afraid of	
	Allah's punishment (for setting up rivals in worship with Allah)?"}. And there are plenty of Qur'anic	
	verses that indicate such meanings. However, modern-day polytheists added two aspects to the old	
	form of polytheism: they associate others with Allah in His Lordship and commit polytheism in times	
	of both comfort and hardship. This is known about them (modern-day polytheists] to those who live	
	with them and explored their style of worship and saw what they did near the grave of 'Alhussein', and	
	Albadawi and other graves of Sufi saints in Egypt, Alaidarouse in Aden, Alhadi in Yemen, Ibn Arabi	
	in Cham (Damascus), [] and some other famous graves that common people exceeded the limits and	
	perform rituals of worship that are His right (i.e. these rituals are supposed to be conducted to Allah	
	Alone) []	
	و العبادة : هي طاعته سبحانه و طاعة رسوله محمد صلى الله عليه و سلم ، بفعل ما أمر الله به و رسوله ، و ترك ما نهى الله عنه و رسوله و عن إيمان بالله و رسوله ، و إخلاص لله في العمل ، مع غاية الحب لله. و كمال الذل له وحده كما قال تعالى : {و قَضَى رَبُكَ أَلا تَعْبُدُوا إِلاَ إِيَّاهُ} أي أمر	
	المسلم المسلم المسلم المعالي المعالي المعالي المعالية المحب الله وتحمل الذل له وحدة حما قان تعالى : {و قصلى ربك الا تعبدوا إلا إيه } اي امر و أوصي بان يعبد وحدو ق قال تعالى : {الْحَمَّدُ للهُ رَبَّ الْعَالَمِينَ * الرَّحْمَنِ الرَّحِيمِ * مَالِكِ يَوْمِ الدِّينِ * إِيَّاكَ نَعْبُدُو و إِيَّاكَ نَسْتَعِينُ } أبان سبحانه بهذه الأولسلي ال	
	الإياث أنه هو المستحق لأن يعبد وحده ، وانستعان به وحده ، واقال عز واحل ، إفاغتُد الله مُخْاصًا له الدين * إلا لله الذينُ الْخَاصُ، ما قال توال	
2	{ فادعوا الله مُخْلِصِينَ له الدين و لو كرة الكافِرُونَ } و قال تعالى : {و أنَّ الْمُسَاجِدَ لله فلا تَدْعُوا مَعَ الله أَجَدًا } و الآبات في هذا المعنى كثير قرء و	
-	ا ملها تدل على : وجوب إفراد الله بالعباده ، و معلوم أن الدعاء بانواعه من العبادة ، فلا يجوز لاحد من الناس أن يدعو الأريه ، و لا يستعين و لا	
	يستغيث إلا به ، عملا بهذه الآيات الكريمة ، و ما جاء في معناها و هذا فيما عدا الأمور العادية ، و الأسباب الحسبة ، التي يقدر عليها المخلوق	

كلها لذل على : وجوب إفراد الله بالعباده ، و معلوم أن الدعاء بالواعة من العبادة ، فلا يجوز لاحد من الناس أن يدعو إلا ربه ، و لا يستعين و ا يستغيث إلا به ، عملا بهذه الآيات الكريمة ، و ما جاء في معناها و هذا فيما عدا الأمور العادية ، و الأسباب الحسية ، التي يقدر عليها المخلوق الحي الحاضر ، فإن تلك ليست من العبادة ، **بل يجوز بالنص (القرآن و السنة) و الإجماع (علماء السلف)** أن يستعين الإنسان بالإنسان الحي القادر ، في الأمور العادية التي يقدر عليها ، [...]

And the worship is obeying, Him the Glorified, and obeying His messenger Muhammad, may Allah send His blessings and peace upon him. It is doing what Allah and His messenger ordered us to do and refrain from committing what Allah and His messenger forbid holding faith in Allah and His messenger. It is the sincere worship of Allah making faith pure to Him Alone. Worship is also the ultimate devotion of love to Allah and the complete submissiveness to Him Alone as the Exalted: {And your Lord has decreed that you worship none but Him} which means that He ordered and enjoined on us to worship Him Alone, and the Exalted said: {So worship Allah (Alone) by doing religious deeds sincerely for Allah's sake only. Surely the religion (i.e. the worship and the obedience) is for Allah only}, and the Exalted said: {So, call you (O Muhammad and the believers) upon (or invoke) Allah making (your) worship pure for Him (Alone) (by worshipping none but Him and by doing religious deeds sincerely for Allah's sake only and not to show off and not to set up rivals with Him in worship), however much the disbelievers (in the Oneness of Allah) may hate (it)}, and the Exalted said: {And the mosques are for Allah (Alone): so invoke not anyone along with Allah}, and the Qur'anic verses about this issue are many that indicate the obligation of worshipping Allah Alone. And it is known that all types of supplications are part of worship, thus, according to these Qur'anic verses, it is impermissible for all people to supplicate to and neither seek nor beseech the help from other than Allah in order to fulfil the obligation indicated by the previously mentioned Qur'anic verses and other verses that bear the same meanings. However, it is not considered as a form of worship if someone takes proper means to seek the help of living human beings who are present in normal issues they can do. On the contrary it is permissible as indicated by both sacred texts (Qur'an and Prophetic traditions) and the consensus of the (Salafi) religious leaders [...]

الوجه الرابع : من الوجوه الدالة على بطلان الدعوة إلى القومية العربية أن يقال : إن الدعوة إليها و التكتل حول رايتها **يفضي** بالمجتمع و لا بد إلى رفض حكم القرآن؛ لأن القوميين غير المسلمين لن يرضوا تحكيم القرآن ، فيوجب ذلك لز عماء القومية أن يتخذوا أحكاما وضعية تخالف حكم القرآن ، حتى يستوي مجتمع القومية في تلك الأحكام ، و قد صرح الكثير منهم بذلك كما سلف ، وهذا هو الفساد العظيم ، و الكفر المستبين و الردة السافرة ، كما قال تعالى : (فلا و ربَّكَ لا يُؤْمِنُونَ حتى يُحكم التررَبَيَنَهُمْ ثم لا يَجدُوا في أَنْفُسِهِمْ حَرَجًا مِمًا قَضَيْت و يُسَلَّمُوا تَسْلِيمًا) و قال تعالى : (فلا و ربَّكَ لا يُؤْمِنُونَ حتى يُحَمُّوكَ فيما شَجَرَ بَيْنَهُمْ ثم لا يَجدُوا في أَنْفُسِهِمْ حَرَجًا مِمًا قَضَيْتَ و يُسَلَّمُوا تَسْلِيمًا) و قال تعالى : (فلا و ربَّكَ لا يؤ و قال تعالى : (و من لم يَحْكُمْ بما أنزل الله فَأُولَئِكَ هم الْعَافِرُونَ) و قال تعالى : (و من لم يَحْكُمُ بما أنزل الله فَأُولَئِكَ هم الطَّالِمُونَ) و قال تعالى : (و من لم يَحْكُمْ بما أنزل الله فَأُولَئِكَ هم الْمَافِرُونَ) و قال تعالى : (مَن لم يَحْكُمُ بما أنزل الله فَأُولَئِكَ هم الطَّالِمُونَ)

The fourth viewpoint: One of the viewpoints that also indicate the rejection of the nationalists' call is that it could be said that calling to it (Arab nationalism) and gathering people under its banner will definitely lead [Muslim] society to reject arbitration to Qur'an (Sharia laws), because the [Arab] nationalists who are not Muslims will not be satisfied with the Sharia laws. Hence, [pan Arab] nationalism leaders will take man-made laws that contradict Sharia in order to make all people in their community equal. Verily a lot of them [leaders of Arab national parties] declared that they will rule according to man-made laws as indicated earlier. Doing so is a tremendous corruption, obvious infidelity and blatant apostasy, as the Exalted said: {But no, by your Lord, they can have no Faith, until they make you (O Muhammad) judge in all disputes between them, and find in themselves no resistance against your decisions, and accept (them) with full submission} and the Exalted said: {Do they then seek the judgement of (the days of) Ignorance? And who is better in judgement than Allah for a people who have firm Faith}and the Exalted said: {And whosoever does not judge by what Allah has revealed, such are the Kafirun (i.e. disbelievers - of a lesser degree as they do not act on Allah's Laws)} and the Exalted said: {And whosoever does not judge by that which Allah has revealed, such are the Zalimun (polytheists and wrong-doers - of a lesser degree)} and the Exalted said: { And whosoever does not judge by what Allah has revealed (then) such (people) are the Fasiqun [the rebellious i.e. disobedient (of a lesser degree)] to Allah}. و هكذا الإيمان بالكتب يجب الإيمان أجمالاً بأن الله سبحانه أنزل كتباً على أنبيائه و رسله ، لبيان حقه و ألدعوة إليه ، كما قال تعالى

3

4

: (لَقَدْ أَرْسَلْنَا بِالْبَيْنَاتِ وَ أَنزَلنا مَعَهُمُ الكتاب و المِيزَانَ لِيَقُومَ الناس بِالْقِسْطِ) الآية ، و قال تعالى : (كَانَ الناس أُمَّةُ و احدَةً فَبَعَتَ الله النَّبِيِّينَ مُبَشَرِينَ و مُنْذِرِينَ و أنزل مَعَهُمُ الكتاب بِالْحَقَّ لِيَحْكَمَ بَيْنَ النَّاسِ فَيما الْخَتَلَفُوا فيه) الآية . و نومن على سبيل التفصيل بما سمى الله منها كالتوراة و الإنجيل و الزيور و القرآن [...] and in the same manner is the <u>belief in Scriptures. It is obligatory (on Muslims) to generally</u> <u>believe that Allah, the Glorified, sent down Books to His prophets and messengers</u> to state His rights (of true worship) and to call people to them, <u>as the exalted said</u> { Indeed We have sent Our Messengers with clear proofs, and revealed with them the Scripture and the Balance (justice) that mankind may keep up justice... till the end of the Qur'anic verse, <u>and the Exalted said</u>: {Mankind were one community and Allah sent Prophets with glad tidings and warnings, and with them He sent down the Scripture in truth to judge between people in matters wherein they differed till the end of the Qur'anic verse. <u>We should specifically believe</u> in those <u>Books that Allah named (in the</u> <u>Qur'an) such Torah, Gospel, Psalms (the Holy Book of Dawud (David)) and the Qur'an</u>

Concordance D2 Expanded Concordance lines of Argumentation threads in Alhawali

	N	Expanded Concordance lines (argumentation threads 1-3)	
		و لكن المرأة الغربية المنكوبة إذ تحاول الرجوع إلى البيت و وظيفة الأمومة لا تستطيع، لأن المشكلة اتسع نطاقها إلى درجة نتعذر	
		معها العودة الحقيقية، و لو حصل شيء من ذلك فإن المجتمع الذي تعود الانحلال و الاختلاط يستنكر ه و يأباه، بل أصبح الزواج ر غم هشاشته مصدر إز عاج للقائمين على تلك المجتمعات فقد طلعت الصحافة الغربية يوما بخبر يقول : "انز عجت السلطات التعليمية في	
		اسكتلندا بسبب موجة الزواج التي تعصف بالمدرسات، فقد تبين أنه من خلال عام (١٩٦٠م) عينت ١٥٦٣ مدرسة في سكتلندا، و في	
		نهاية العام الدراسي تركت ألف منهن الوظيفة للزواج، و قالت السلطات إن الزواج يهدد النظام المدرسي".	
	1	But the afflicted western woman, who tries to settle down in her home and revert to being a mother, is	
		unable to do so. The case is so, because the problem has reached the limits that make it impossible to return to (her real role in life, i.e. being a mother). If something like that (going back to settle in a	
Argumentation thread		house and perform her real role as a mother) happens, the society that has accustomed to and	
		practiced moral debasement and the mixing between men and women, would refuse it. Additionally,	
		marriage, though not based on solid grounds, became <u>a cause of discomfort to people in authority in</u> those (western) societies. <u>Western newspapers</u> , in one day, published a piece of news that <u>says</u> : 'the	
		Scottish educational authorities are annoyed because of the wave of marriages that blow away female	
		teachers (from teaching). It has been known that during the year 1960 there were 1563 female	
		teachers appointed in schools, and 1000 of them left their jobs to get married by the end of the	
ead		Academic year. The authorities <u>said</u> that marriage threatens the school system'.	
		و حيال ذلك إظلم المرأة الغربية]: ماذا في وسع المرأة أن تفعل؟ بأي شيء تواجه المجتمع النكد الذي يلهب ظهرها بالسياط، و يقطع	
		عليها طريق العودة إلى فطرتها، ليس هناك إلا أحد سبيلين: إما الانتقام من هذا المجتمع الطالم بترويعة و تعكير صفوه، كما جاءً في التقرير الذي نشرته الصحف من أنه بلغت عدد سرقات المتاجر الكبيرة في إنجلترا خلال عام (١٩٦٠م) نحو (٣٢١٩٤) سرقة، []،	
		ا و الغريب أن (٦٠%) من هذه السرقات ارتكبتها نساء جاوزن سن البلوغ، و (٣٠%) ار تكبها ذكور أقل من السابعة عشرة، و تقول	
	2	ا الإحصائيات: إن كل السارقات من النساء لم يكنُّ في حاجة للمال، نعم إنها ليست الحاجة للمال، و لكنها الر غية في الانتقام و تفريغ	
		السخط . و أما الانتقام من نفسها بالانتحار كما فعلت الممثلة الشهيرة مارلين مونرو، التي كتبت قبيل انتحارها نصيحة لبنات جنسها يقول فروا: "احذر ميال حد ماجذر مرمن كارين بذر والإسلاني الميانية الشهيرة مارلين مونرو، التي كتبت قبيل انتحارها ن	
		تقول فيها: "احذري المجداحذري من كل من يخدعك بالأضواءإنى أتعس امرأة على هذه الأرض لم استطع أن أكون أمأ إني امرأة أفضل البيت الحياة العائلية الشريفة على كل شيء إن سعادة المرأة الحقيقية في الحياة العائلية الشريفة الطاهرة، بل إن هذه	
		ا الحياة العائلية لهي رمز سعادة المراة بل الإنسانية، و تقول في النهاية: لقد ظلمني كل الناس، و أن العمل في السينما بجعل من المرأة	
		سلعة رخيصة تافيهة مهما نالت من المجد و الشهرة الزائفة".	

such unjust society that lashes her back fiercely [a metaphor meaning that society is oppressing her] and prevents her from going back to her original nature [marriage and fostering children]. There is no means to [face such society] but to do either of the following: To take revenge against this unjust society by terrorizing it and causing disturbances as written in <u>a report published in newspapers</u> that shoplifting from grand stores during 1960 reached around 32194 cases, []. It is so strange that 60% of shoplifting was committed by adult women whereas 30% of such crimes were committed by males under the age of 17. The statistics <u>savs</u> : All female shoplifters weren't in need of money, but they sought revenge against society. Or they seek to inflict punishment on themselves by committing suicide as the famous actress Marilyn Monroe did. She wrote a piece of advice to her fellow women before committing suicide. She <u>savs</u> : beware of glory beware all those who try to pull your leg by the glamour of fame I am the most miserable women on earth I could not be a mother I am a woman that prefers staying in my home and the honourable family life to everything else. Truly, woman's real happiness is in the purified and honourable family life. Indeed, such family life is not only the symbol of woman's happiness but indeed it is the symbol of happiness to all humanity. She finally <u>says</u> : 'All people dealt with me unjustly, and working in movie industry makes the woman a cheap and worthless commodity despite her glory and unreal fame'.
و ليس غريباً أن تؤكد الإحصائيات العالمية أن نسبة محاولات الانتحار عند النساء أكثر منها عند الرجال، يقول تقرير كتبه أحد

و ليس غريبا أن تؤكد الإحصائيات العالمية أن نسبة محاولات الانتحار عند النساء اكثر منها عند الرجال، يقول تقرير كتبه احد الأطباء الاجتماعيين في فيينا: و قد لوحظ أن النساء أكثر محاولة من الرجال، ففي عام (١٩٤٨م) كان عدد المحاولات في النساء (٣٨١) و هذا يوافق (٣٨.٥%) من المجموع، و في عام (١٩٥٦م) كان العدد (٩٥٠) أي بنسبه (٣٨.٢%)، و في عام (١٩٥٩) كانت النسبة (٢٩.٥%)، كما لوحظ أن نسبة المحاولات في الفتيان و الفتيات الذين تتر اوح أعمار هم بين (١٤) عاماً و (٢٠) عاماً ترتفع باستمرار، فعند الفتيان كانت النسبة في عام (١٩٤٨م)- (٥,٦%)، و في عام (١٩٥٦م) أي بنسبه (٢.٤٣%)، و في عام (١٩٥٩ ر ١٨.٦%)، و أما عند الفتيان كانت النسبة في عام (١٩٤٨م)- (٥,٦%)، و في عام (١٩٥٦م) (٣٠.٦%)، و في عام (١٩٥٩م) مرارم عند الفتيات فالتصاعد مخيف، ففي عام (١٩٤٨م)- (٥,٠%)، و في عام (١٩٥٠م) (٣.٥ مجموع محاولات الانتحار في ذلك العام، و في عام (١٩٥٦م) حاولت (٥٠) فتاة الانتحار، و هذا يشكل نسبة (١٩٠٩%) من مجموع محاولات الانتحار في ذلك العام، و في عام (١٩٥٦م) حاولت (٥٠) فتاة الانتحار، و هذا يشكل نسبة (٣٠,٥%)، و في عام أربع منها من جانب الفتيات، و التحار، و هذا يعني نسبة (١٩٤٦م) حاولت (٥٠) فتاة الانتحار، و هذا يشكل نسبة (٣٠,٥%) و في عام أربع منها من جانب الفتيات، و النتحار، و هذا يعني نسبة (١٩٥٠م) حاولت (٥٠) فتاة الانتحار، و هذا يشكل نسبة (٣٠,٥%) و في عام مجموع محاولات (١٠٠) فتاة الانتحار، و هذا يعني نسبة (٢٠٤، ٢٤ %) و هذا يعني أن كل نسعة أيام توجد ست محاولات انتحار،

It is not very unfamiliar that international statistics confirms that attempts to commit suicide increased among women as compared to men. A report written by one of the 'social' physicians in Vienna <u>says</u>: and it has been noticed that women attempt to commit suicide more than men. In 1948 there were 381 suicidal attempts committed by women which constitute 55.61% of the total attempts. In 1956 the number increased to 590 attempts that mean 56.73% and in 1959 the percentage was 55.92%. It is also noticed that the percentage of suicidal attempts is increasingly growing among young males and females whose ages range from 14 to 20. The percentage of the cases of committing suicide increased among young males as it was 6.5% in 1956 and became 6.81% in 1959. However the rates of committing suicide among young females were increasing terribly. In 1948, 89 young females tried to end their lives which constitute 8.55% of the total number of suicidal attempts and in 1959 another 150 young females tried to commit suicide which means 14.20% of the total suicidal attempts. Thus we can conclude that every 9 days there were 6 attempts to commit suicide where 4 of these attempts are made by young girls whereas two are made by young boys.

3

Argumentation thread

N

4

[...] فإن التصور الإسلامي يقرر و يؤكد حقيقة عظمى و قاعدة جليلة تضمنها **قوله** تعالى: ((ألاَ تَزَرُ وَازِرَةٌ وَزَرَ أَخَرَى * و أَن لَيْسَ لِلإِنسَانِ إِلاَّ مَا سَعَى)) [النجم: ٣٨-٣٩] فلا يؤاخذ الله تعالى أحداً بذنب غيره مهما كانت الصلة بينهما [...]. إنه - حسب قاعدة العدل الرباني - لا يجوز أن يؤاخذ أحد غير أدم بخطيئته، حتى و لو كان ابن الشيطان الذي أغواه بالخطيئة - فضلاً عن أن يكون ابن الله - كما تقول الكنيسة تعالى الله عن ذلك علواً كبيراً أو أحداً من بني آدم. و بذلك خلا التصور الإسلامي من الأفكار و النظريات التي ابتدعتها الكنيسة باعتبار ها من مستلزمات الخطيئة،[...]. و كان علماء المسلمين أسبق من فلاسفة عصر التنوير و اتباع مدرسة التو في هذا المضمار، يقول أحدهم: (... فنسبوا الإله الحق – سبحانه - إلى ما يأف أسقط الناس و أقلهم أن يفعله بمملوكه و عبده [...] و نسبوه إلى أقبح الظلم، حيث ز عموا أنه سجن أنبياءه و رسله و أولياءه في الجحيم بسبب خطيئة أبيهم، و نسبوه إلى غاية السفه حيث نصبوه إلى أقبح الظلم، حيث ز عموا أنه سجن أنبياءه و رسله و أولياءه في الجحيم بسبب خطيئة أبيهم، و نسبوه إلى غاية السفه حيث خاصهم من العذاب بتمكينه أعداءهم نفسه حتى قتلوه و صلبوه [...](. و بالجملة فلا نعلم أنهم أن يوا معبوه إلى غاية السفه حيث به هذه الأمة، كما أنه من معبودها و أنه سجن أنبياءه و رسله و أولياءه في الم عليف أستوم الناس و أقلهم أن يفعله بمملوكه و عبده بسبوه إلى أقبح الظلم، حيث ز عموا أنه سجن أنبياءه و رسله و أولياءه في الجحيم بسبب خطيئة أبيهم، و نسبوه إلى غاية السفه حيث بله هذه الأمة، كما قال عمر رضي الله عنه: [[أنهم سبوا الله مسبة ما سبه إياها أحد من البشر] [...]

[...] and the Islamic conception of (sins) states and confirms a noble reality and a splendid principle that is included in the following Qur'anic verse, Allah said: {That no burdened person (with sins) shall bear the burden (sins) of another, And that man can have nothing but what he does (good or bad)}. 53:38-39. Thus Allah, the Exalted, will not punish one with the guilt of another no matter how closely they are related [...]. Verily - according to the divine justice - it is not permissible to punish any individual because of the sin of Adam, even if this individual is the son of the Satan who persuaded Adam to commit the sin let alone the son of God -as the Church says, Glorified and Exalted be He! High above (the great falsehood that they say) – or any of the sons of Adam (humans). Thus Islamic conception [of sins] does not include the ideas and theories that were invented by the Church which are considered to be inevitable consequences of the (original) sin [...]. Muslim scholars (religious leaders) preceded (had the precedence over) the philosophers of the Enlightenment and the advocates of historical criticism in this respect [criticizing the Church's concepts of the original sin]. One of them [the philosophers and the advocates of historical criticism] says: (... they [the people of the church] attributed to the true God – the Exalted – what the most morally depraved and insignificant individual could attribute to his own slave! [...] They also attributed the worst form of injustice to Him when they claimed that he imprisoned His prophets and messengers and His close pious worshippers in Hell-fire because of their father's sin. They attributed foolishness to Him as they claimed that He granted them salvation [from the sin] by enabling His enemies to kill and crucify Him [...]). Generally we do not know any nation that execrated their Lord whom they worship like this nation [Christian nation], as Omar [the second Caliph in Islam], may Allah be pleased with him, said: (Verily they execrated Allah with a vilification that no single human has ever attributed to Him] [...]

٣- برامج لاحتلال الخليج والجزيرة
[...] و في ظل تلك الأوضاع المتقلبة و الإدارة الأمريكية الضعيفة ظهرت دعوات قوية في أمريكا تنقد ضعف بلادها و تنذر أمتها بالخطر المحدق إن لم تعد العدة كاملة لإحتلال الخليج. كما <u>صرح</u> بعضهم أو لحماية أمن الخليج كما يعللها الأكثرون. و نشير هذا إلى بعض المُتَرجم المتداول في الأوساط العربية منها، [...] ١- <u>دراسة عسكرية استراتيجية</u> نشرتها مجلة فورتشون الأمريكية في مايو ١٩٧٩م بعنوان "التدخل العسكري في منابع النفط، الاحتمالات و الخطط " و مما ورد فيها ص ٤: "إن نزاعاً بين العراق و الكويت أو السعودية نابعاً من الخلافات القائمة منذ زمن حول الأراضي هو خطر حقيقي". ثم <u>تقول</u> الدراسة :- "لقد أوضح كل من براون و الكويت بريجينسكي - مساعدا الرئيس كارتر لشؤون الأمن القومي - مؤخراً أن الولايات المتحدة ستتخذ خطوات من بينها استخدام القوات العسكرية لحماية مصالحنا في العربية السعودية".

3- Scenarios to invade the Gulf and the [Arab] Peninsula

[...] and in the light of unstable political conditions [in the Middle East] and the weak American administration, some serious proposals appeared in America criticising the American weakness and warning the American nation of the dangers if they do not prepare their troops to invade and occupy the Gulf as some of them <u>stated</u> or to protect the Gulf as most of them <u>justified</u>. I present here some of the translated reports (into Arabic): [...] 1-<u>A military strategic study</u> published by the American 'Fortune' in May 1979 titled 'Military intervention in Oil Wells: Probabilities and plans'. In page 4 there appears an indication that 'there could be a conflict between Iraq and Kuwait or Saudi Arabia over the long lasting problematic territories which emerge to be a real threat'. Then the study <u>says</u>: 'Both Brown and Brzezinski - Assistants to the President (Jimmy Carter) for National Security Affairs – have recently declared that the United States will take measures including military intervention to protect our interests in Saudi Arabia'.

ω

四人一個種 有多能留法的法國多路的法律律

و بالنظر بشكل عابر للصحافة الأمريكية و البريطانية يلاحظ أنه قد تم بناء الأساس الدعائي للحرب من خلال تصوير صدام حسين كبعبع في المنطقة و تولت وسائل الإعلام الغربية العزف على وتر التهديدات التي أطلقها صدام ضد إسرائيل و قامت هذه الوسائل في الوقت نفسه بالتغاضي عن التهديدات الإسرائيلية لمختلف الدول العربية. و في مقال آخر تقول المجلة نفسها (EIR FEETURE) : إسرائيل تجهز لقيام حرب/ تعرى في منطقة الشرق الأوسط لإيجاد حل نهائي للمشكلة العربية... و هذا يعني فيما يبدو الحرب ضد العراق و دول أخرى و تدمير الأردن. و تسخر من مزاعم بوش أنه يسعى للسلام في المنطقة قائلة: هذا هراء فالولايات المتحدة ملتزمة بقيام حرب في الشرق الأوسط و ربما كان الشيء الوحيد الذي لم يتحدد هو تاريخ إندلاعها، أما الإسرائيليون فهم مستعدون الحرب و لما يُسمى بالحل النهائي. و قالت ربما كانت هناك أو ستكون على الأرجح إتفاقات جديدة بين القوى في إسرائيل و حافظ الأسد، أو شيء من هذا، و إذا ما تم وضع القضايا الهامشية جانباً فإن إسرائيل مبر مجة حالياً باتجاه الحرب و الفري إما أخبياء أو كاذبون، إن ذلك هو طبيعة القصايا الهامشية جانباً فإن إسرائيل مبر مجة حالياً باتجاه الحرب و الفرين

Reading through the American and British press quickly, we observe that the campaign for the war [on Iraq] is based on demonizing Saddam Hussein. And the western media started to repeat the threats that Saddam expressed towards Israel and at the same time this [western] media neglected the Israeli threats against the different Arab countries. And in another article the same magazine (EIR FEETURE – spelling in the original text) <u>says</u>: "Israel is preparing for *another* war in the Middle East to make a final solution to the Arab problem ... This means a war, presumably against Iraq and other states, and the destruction of Jordan. It also ridicules Bush's (George H. W.) claims that he is seeking to establish peace in the region (Middle East to achieve what is called the final solution. The exact date is the onty thing that's open to question." It also <u>says</u>: "There may be, and probably will be, new agreements between forces in Israel and Hafez aI-Assad (Syrian president 1971 – 2000), things of that sort. But, the side issues aside, Israel is presently programmed for war. Those who tell you that this is not true are either foolish, or are lying. That's the nature of the situation."

Appendix F

Glossary

Islamic terms and figures	Explanations
Al-Albani (' <i>al'albani</i> ; الألباني)	Muhammad Nasiruddin al-Albani (1914-1999) is Syrian (Albanian origin) Hadith scholar who was a staff member of the Islamic University at Medina. He later was considered to be the founder of a movement called Al-Albani's movement which adheres to the medieval school of Hadith.
Alhakimiyyah (الحاكمية;	A religious concept developed by Said Qutb which means Sovereignty of
'alḥakimiyyah)	God
Alhisbah ('alħsibah;الحسبة)	The Salafi Group that practices surveillance (vigilantism) and enforcement of practicing the principles of religion, but without anticipating any worldly rewards, which was later developed as official institution called the religious police or the so called المنكر' (Hay'at 'al'amr bilma 'arūf wa 'annahi 'an 'amlonkar; the Committee for Commanding Right and Forbidding Vice).
Aljahiliyyah (الجاهلية; 'alğāhliyyah)	Ignorance or the religious state of Arabs before Islam
Ash'arites (الأشاعرة; 'al'aša'irah)	Adherents of Abu Alhasan Alashari (874-936) who was a follower of Mu'tazilite school (see Mu'tazilites below). Abu Alhasan Alashari rejected his ideas before death. However, his followers till modern days still seek to understand the basics of Islamic creed (specially in the Muslims creed of the names and attributes of Allah) based on dialectical and speculative theology (i.e. rationalisation and reasoning).
Brethren of Najd (إخوان نجد; 'iḱwan najd)	The Bedouin who depended on plundering as a means of living in the days of draught and later accepted the fundamentals of Orthodox Islam of the Hanbali School as preached by Abd-al-Wahhab, they fought with AbdulAziz in his wars in Arabia and revolted against him, but they were subdued in the Battle of Sibila in 1929.
Dahriyyah (الدهرية; 'addahriyyah)	a pre-Islamic atheistic doctrine
Divine religions (الأديان السماوية; 'al'adyān 'assamāwiyyah)	The religions that Muslims believe they were sent by Allah to humans which are mostly used to indicate the three monotheistic (Abrahamic) religions i.e. Judaism, Christianity and Islam
Hadith (الحديث; 'alḥadīt)	The Prophetic Traditions
Hanbali School of jurisprudence (المذهب الحنبلي; 'alma <u>d</u> hab 'alḥanbali)	one of major four canonical schools of jurisprudence in Sunni Islam; the other three are 'حنفي')šafi'i; Shafi'i), 'مالكي' (maliki; Maliki) and 'حنفي' (ħanafi; Hanafi
Ibn Hanbal, Ahmad	The founder of a religious-legal school called Hanbalism in the ninth century (see Hanbali School of jurisprudence).
Ibn Taymiyya, Ahmad	A medieval scholar of religion (d. 1328) who adhered to the Hanbali School of jurisprudence. He is sometimes referred to by as the Sheik of Islam.
Irjā' (ارجاء)	The concept of ''Irjā'' is linked to 'إيمان' (<i>'iman</i> ; faith) in Islam. It means both delay and hope. Faith in Islam is a tripartite concept. It is composed of the following elements; believe with intention, vocalizing believes (i.e. accompanied with utterance/s) and action or performance of duties of worship –prayers, observing fasting during the days of Ramadan etc Morjia'ah (the plural of Morj'a) are those who believe that it suffices for an individual to be a believer if s/he accomplishes the first two elements.

Islamic awakening (الصحوة الإسلامية; alṣahwah al'islamiyyah	Islamic general awakening (revivalist) movement started in the 1970s and became manifest in the 1980s after thwarting Juhaiman's revolt. It is also called Sahwah.
Maturidites (الماتريدية; ' <i>almaturidiyyah</i>)	The followers of Abu Mansur Almatudidi (853-944 in Uzbakistan). They (as well as the Ash'arites) base their beliefs on speculative theology.
Monotheism (التوحيد; 'attawḥīd)	Unity of Lordship, Unity of Worship and Unity of Allah's Names and Attributes –understanding these concepts is essential to understanding the different Sunni factions, because they were a source of clashes and disputes between the different Islamic sects and factions both in the past and modern times.
Mu'tazilites (المعتزلة; 'almu'tazilah)	An Islamic school of theology (flourished in the 8 th and 10 th centuries) who sought to understand Islamic creed based on reason and later relied on logic. There are two schools of Islamic theology who are considered the descendants of the Mu'tazilites; 'الأشاعرة' ('al'aša'irah; Ash'arites) and 'الماتريدية'
Particles of faith	Faith in the Sunni Islam is to belief in Allah, His Angels, His Books, His messengers, the Last Day and the Divine Preordainment - good and bad.
Pilgrimage (حج; ḥağ)	The fifth of pillar of Islam which is performed during the twelfth month of the Hijra calendar called 'ذو الحجة' (<u>dol</u> hiğğah).
Pillars of Islam	In the Sunni Islam the pillars of Islam are as follows: 1- Testimony; to witness that there is no god but Allah and Mohammad is his messenger, 2- Performing the five daily prayers, 3- Fasting during the days of the month of Ramadan (of the Muslim calendar), 4- Alms-giving (zakat), and 5- Performing pilgrimage by bodily and financially able adult Muslims.
Qutbis	Adherents of Said Qutb (1906-1966) who is an Egyptian literary and religious figure.
Qutbism	A general term given to the modern political Islamic movements that are based on the teachings of Sayyid Qutb
Sahawists	Religious figures of the Islamic Awakening trend such as Safar Alhawali whose efforts and opposition to the Saudi government were salient in the 1990s.
Salafi	The followers of the early pious predecessors (superior in religious status) who belong to the three generations of the early Muslims.
Sayyid Qutb	The spiritual founder of Qutbism that has profound influence on most modern political Islamic movements.
Shiites	The second largest faction of Islam (approximately 10-13% of Muslim population) most of which are 'الإثني عشرية' ('al'ithay 'ašriyyah; the Twelvers) who mostly live in Iran, some parts of Uraq and some Gulf States suchcas Kuwait and Saudi Arabia. The Twelvers believe in 12 infallible Imams -divinely granted freedom from sins and errors. The rest are 'الزيدية' ('azzaydiyyah; Zaidis (in Yemen) and other Shiites factions such as 'الزيدية' ('al'isma 'iliyyah; Ismailis) (in Najran, Saudi Arabia) and the newly evolved Shiite factions; 'الحوثيون' ('alḥuṯiyyūn; the Huthis) in Yemen and 'المهدويون' ('almahdawyīn; the Mahdis –a faction of Shiite Twelvers who believe in 12 Imams and 12 Mahdis –Saviors) that started to appear in the late 1990s in Iraq. Although the Shhite factoins differ to a large extent all of them believe that Ali Ibn Abi Talib (the cousin and the son in law of the Prophet of Islam) is the legitimate successor of the Prophet of Islam.

Sufis/Sufism (الصوفية; 'aṣofiyyah)	A group of mystic orders (congeregations gatherd around a master) praying to reach what they call purity. They venerate and take Auliya (friends of Allah according to the Sufi's interpretations of Auliya) as intercessors with Allah.
Sunnah/Sunnis	The sayings and doings of the Prophet of Islam as reported in the books of Hadith (Prophetic traditions) is called the Sunnah of the Prophet or simply the Sunnah. However, the appellation 'sunnah' is used to refer to the Sunnis or the people of Sunnah that constitute the largest branch of Islam (as opposed to the Shiites).
The people of Hadith (أهل الحديث ; hlulḥad <u>t</u>)	People of the Prophetic traditions that opposes the use of reason in religious ruling, calling for direct and exclusive reliance on the Qur'an and the Sunnah